# **University of Strathclyde**

# **Department of Accounting and Finance**

# The Time-Variation in Style Effects in the UK Stock Market

Hong Li

A thesis submitted for the award of Ph.D.

May 2013

## **Declaration of Authenticity and Author's Rights**

'This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has lead to the award of a degree.'

'The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.'

Signed:

Date:

## Abstract

The thesis extends most previous studies on static version of style effects in the overall period to their time-varying properties in the dynamic macroeconomic conditions and market states in the UK Stock Market. It deals with four research questions on style effects in the UK Stock Market in four empirical chapters, respectively. Firstly, the thesis uses two indicators, long/short return  $R_{lms}$  and style coefficient  $\gamma$ , to examine whether the time-variation in style effects with short-/medium-/long-term horizon exists in the UK Stock Market. The research finds stronger momentum effect with 6/12 months' formation horizon and 6/12 months' holding horizon from 1956 to 2008, but its volatile pattern over time with the strongest in the 1990s and the weakest in the 1970s. Empirical evidence presents no existence of significant size effect from 1979 to 2008 due to the rotation between (marginally) significant small-cap effect in the 1980s and marginally significant large-cap effect in the 1990s. It finds significant value effect, strengthening from 5 months' to 24 months' holding horizon, from 1980 to 2008 due to persistent outperformance of shares with low price-to-book ratio over time.

Secondly, the thesis applies time-series regression to investigate what factors can significantly explain time-varying style effects in the overall period from 1980 to 2008 and three subperiods. It finds that macroeconomic variables, such as annual change of GDP(GDP(Y)), unexpected annual change of TBILL(UEXTBILL(Y)), annual change of CPI(CPI(Y)), and lagged 8-year's market return (MR(-8)), can offer significant explanatory power for the time-variation in momentum effect (12/6,

12/12), size effect (12), and value effect (24) stronger than corresponding style effects with other holding horizons from 1980 to 2008. The decline in GDP(Y) and the increase in UEXTBILL(Y) imply strong momentum effect and large-cap effect. The decline in CPI(Y) and high MR(-8) means strong value effect. The sensitivity to the same economic forces can explain the findings on correlation among style effects. The thesis shows the variation in significant macroeconomic variables to explain style effects through holding horizons and over time.

Thirdly, the thesis discusses whether the dynamic style effects can be indeed predictable by using some lagged macroeconomic variables from 1980 to 2008. It uses statistics PIS, POOS, and PSS to test the performance of recursive in-sample regression model and out-of-sample forecasting model relative to sample mean model. With the model test, the research examines the reliability/or pitfall of in-sample predictability. Different from momentum effect (12/6, 12/12) and size effect (12), out-of-sample forecasting model for value effect (24) performs poorly relative to the historical mean model even though its in-sample regression model does better. The test shows that the recursive out-of-sample forecasting model offers successful signal to capture momentum effect (12/6, 12/12) in higher percentage of all observations, followed by size effect (12) and value effect (24).

Finally, the thesis explores whether active trading strategies with filter threshold and signal from out-of-sample forecasting model can capture style rotation from 1980 to 2008. Empirical work shows the significant gains to timing momentum effect. Active trading strategy on any equal-weighted decile momentum portfolio (12/6) and value-

weighted Winner/Loser portfolio (12/6) significantly outperforms passive trading strategy which always hold Winner/Loser portfolio all the time. Although active trading strategy on size effect (12) improves passive strategy from negative average monthly excess return to positive one, it might becomes unprofitable after considering trading costs. Persistent performance of return spread between 'Low' and 'High' portfolios and poor performance of out-of-sample forecasting model lead to the failure of active trading strategy on value strategy.

## Acknowledgements

I should like to acknowledge my indebtedness and gratitude to my supervisors, Professor Jonathan Fletcher and Dr Leilei Tang. Professor Jonathan Fletcher provided important assistance throughout the PhD process. More than 40 meetings and discussions, as well as comments on my reports and drafts, paved the way to the completion of PhD programme. His insightful ideas and suggestions contributed a great deal to my thesis. Dr Leilei Tang played important role in my application of Matlab in empirical analysis and proof reading of my thesis. Without their encouragement and support, this thesis would never have been completed.

I should like to express my thanks to other members of the Department of Accounting and Finance, especially Professor Krishna Paudyal whose very helpful comments expanded and improved my empirical work. My thanks must go out to Professor Dick Davies who offered valuable suggestions in annual PhD presentation. I am especially grateful to Donna McDougall and Donald Campbell for giving me support of data collection which saved me much time. I would like to extend my heartfelt gratitude to Lorna Carlaw for her never-ending kindness and support in my academic life. I thank other PhD students for their help and encouragement. In addition, I should show my gratitude to Professor Krishna Paudyal and Dr Frankie Chau who offered important suggestions for improvements in my thesis.

My final gratitude is dedicated to my father and my wife who provided consistent understanding, encouragement, and support in four-year's PhD programme.

## Dedication

In loving memory of my mother

## Contents

Declaration of Authenticity and Author's Rights	ii
Abstract	iii
Acknowledgements	vi
Dedication	vii
Contents	viii
List of Tables and Figures	xii
List of Abbreviations	xix
Chapter 1 Introduction to the Research	1
1.1 Motivation for the research	
1.1.1 Research background	3
1.1.2 Underlying empirical studies	5
1.1.3 Market characteristics	9
1.2 Research questions	14
1.3 Structure of the thesis	17
1.4 Contributions and implications of the research	
1.4.1 Contributions of the research	20
1.4.2 Implications of the research	22
Appendix: tables and figures in Chapter 1	24
Chapter 2 Literature Review	28
2.1 Cross-sectional return predictability	
2.1.1 Cross-sectional return predictability in the US Stock Market	
2.1.2 Cross-sectional return predictability in the international and	
UK Stock Market	
2.1.3 Theoretical arguments for return anomalies	35
2.2 Time-series return predictability	
2.2.1 Time-series return predictability in the US Stock Market	
2.2.2 Time-series return predictability in the international	
and UK Stock Market	46
2.3 The time-variation in return anomalies	49

2.3.1 The time variation in return anomalies in the US Stock Market4	9
2.3.2 The time variation in return anomalies in the international and UK Stock	
Market5	52
2.4 Concluding Remarks	6
Chapter 3 Data and Portfolio Formation5	8
3.1 Data	8
3.1.1 Data source	58
3.1.2 Firm characteristics6	51
3.1.3 Holding portfolio returns	4
3.2 Portfolio formation	57
3.2.1 Momentum portfolio formation	57
3.2.2 Size portfolio formation	59
3.2.3 Value portfolio formation	4
Appendix: tables and figures in Chapter 3	6
Chapter 4 The Time-variation in Style Effects	4
4.1 Research methods	4
4.1.1 The estimation of style coefficient γ8	6
4.1.2. The estimation of long/short return R <sub>lms</sub>	8
4.2 Momentum effect	39
4.2.1 Momentum effect based on 6 months' formation length8	39
4.2.2 Momentum effect based on 12 months' formation length9	)7
4.2.3 Robustness test on momentum effect10	5
4.3 Size effect	9
4.4 Value effect	7
4.5 The correlation between style effects	3
4.6 Long/Short return $R_{lms}$ and style coefficient $\gamma$	5
4.7 Style effects in business cycles and market conditions12	7
4.7.1 Long/Short return R <sub>lms</sub> in business cycles12	27
4.7.2 Long/Short return R <sub>lms</sub> in market conditions12	9
4.8 Conclusion	1
Appendix: tables and figures in Chapter 413	35

Chapter 5 The Impact of Macroeconomic Variables on the Time-variation Style Effects	in 59
5.1 Macroeconomic variables	59
5.1.1 Derived primary macroeconomic variables1	62
5.1.2 Three kinds of macroeconomic variables	66
5.2 Research methods	71
5.2.1 Univariate regression to extract explanatory variables	72
5.2.2 Multiple regression	76
5.3 The impact of macroeconomic variables on the time-variation in	
momentum effect1	78
5.3.1 Momentum effect based on 6 months' formation length1	78
5.3.2 Momentum effect based on 12 months' formation length1	84
5.4 The impact of macroeconomic variables on the time-variation in seffect	ize 90
5.5 The Impact of macroeconomic variables on the time-variation in valueffect	lue 96
5.6 Conclusion	02
5.6.1 Explanatory power of macroeconomic variables for time-varying style	
effects2	.02
5.6.2 Significant macroeconomic variables for time-varying style effects20	)4
5.6.3 The variation in significant macroeconomic variables for time-varying style	le
effects	07
Appendix: tables and figures in Chapter 52	10
Chapter 6 The Recursive out-of-sample Forecasting Model on Style Effects2	79
6.1 The recursive out-of-sample forecasting model	80
6.2 Research methods	83
6.2.1 Recursive out-of-sample forecasting model2	83
6.2.2 Statistic test	86
6.3 The test on performance of recursive in-sample regression	90
6.3.1 Series adjusted R <sup>2</sup> 2	.90
C 2 2 Series test indian (DIS)	<b>0</b> 4

6.3.2 Series test statistics, 'PIS'	294
6.4 The test on performance of recursive out-of-sample forecasting model	298
6.4.1 Series test statistics, 'POOS'	298
6.4.2 Test statistics, 'PSS'	301

6.5 The series performance of macroeconomic variables in recursive in-sample				
regression	306			
6.6 Conclusion	309			
Appendix: tables and figures in Chapter 6	312			

Chapter 7 Active Trading Strategy on Style Rotation					
7.1 Research methods					
7.1.1 Active trading strategy on individual style effect					
7.1.2 Active trading strategy on combined style effects	332				
7.2 Active trading strategy on individual style effect	335				
7.2.1 Active trading strategy in single direction					
7.2.2 Active trading strategy in double directions					
7.2.3 Active trading strategy on any decile portfolio	349				
7.3 Active trading strategy on combined two/or three style effects	361				
7.3.1 Active trading strategy on combined three style effects	361				
7.3.2 Active trading strategy on combined two style effects					
7.4 Conclusion					
Appendix: tables and figures in Chapter 7	371				

Chapter 8 Conclusions to the Research	
8.1 Summary of findings	
8.2 Contributions and implications of the research	
8.2.1 Contributions of the research	
8.2.2 Implications of the research	
8.3 Limitations and future research	
8.3.1 Limitations of the research	
8.3.2 Future research	

References	
------------	--

## List of Tables and Figures

## Chapter 1 Introduction to the Research

Table 1.1 The holding return of MSCI Value Index and Growth Index in US, UK, and Japan Stock Market in subperiods and overall period from 1975 to 2009 (100/%)

#### **Chapter 3 Data and Portfolio Formation**

Table 3.1 The performance of average monthly Winner/Loser return R<sub>wml</sub> based on lagged 6 months' formation: skipping one month (i-2 to i-6) and unskipping one Table 3.2 The performance of average monthly Winner/Loser return R<sub>wml</sub> based on lagged 12 months' formation: skipping one month (i-2 to i-12) and unskipping one Table 3.3 The average number of shares in the sample period in the investigation of Table 3.4 The percentage of market value of decile size portfolios based on number Figure 3.1 The percentage of market value of the 1<sup>st</sup> (small-cap) portfolio among overall market capitalization over years with portfolio formation based on number of Figure 3.2 The percentage of market value of the 10<sup>th</sup> (large-cap) portfolio among overall market capitalization over years with portfolio formation based on number of Table 3.5 The number of shares in decile size portfolios with Dimson and Marsh 

## **Chapter 4 The Time-variation in Style Effects**

Table 4.17 Statistics on Long/Short return $R_{lms}$ and style coefficient $\gamma$ 155
Table 4.18 The performance of monthly Long/Short return $R_{lms}$ in positive and negative annual change of GDP from 1980 to 2008
Table 4.19 The performance of monthly Long/Short return $R_{lms}$ in positive and negative lagged one-year's market return from 1980 to 2008157
Figure 4.1 Equal-weighted monthly Small/Large return $R_{sml}$ and Low/High return $R_{lmh}$ with 12 months holding horizon from 1980 to 2008158

# **Chapter 5 The Impact of Macroeconomic Variables on the Time-variation in Style Effects**

Table 5.1 Macroeconomic variables for the time-variation in style effects .......210 Table 5.2 The autocorrelation and the unit root test of primary macroeconomic Table 5.3 The autocorrelation of (unexpected) change of primary macroeconomic Table 5.4 The correlation coefficient of annual (unexpected) change of Table 5.5 Univariate regression of momentum effect on individual lagged Table 5.6 Univariate regression of momentum effect on individual lagged Table 5.7 Univariate regression of momentum effect on individual lagged Table 5.8 Univariate regression of momentum effect on individual lagged Table 5.9 Univariate regression of momentum effect on individual lagged Table 5.10 Multivariate regression of momentum effect on the combination of lagged Table 5.11 Multivariate regression of momentum effect on the combination of lagged Table 5.12 Multivariate regression of momentum effect on the combination of lagged Table 5.13 Multivariate regression of momentum effect on the combination of lagged 

Table 5.14 Multivariate regression of momentum effect on the combination of lagged

 Table 5.17 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/6)
 232

 Table 5.18 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/12)
 234

Table 5.21 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables (Momentum-12 months' formation length) ......240

Tabl varia	le 5.27 U ables (Si	Univariate ize-3)	regression	of size	effect	on	individual	lagged	macroeco	onomic 247
Tabl	le 5.28 U	Univariate	regression	of size	effect	on	individual	lagged	macroeco	onomic

 Table 5.32 Multivariate regression of size effect on the combination of lagged macroeconomic variables

 257

Table 5.36 Multivariate regression of size effect on the combination of lagged macroeconomic variables in subperiods (Size: 18 months' holding length) .......261

 Table 5.41 Univariate regression of value effect on individual lagged macroeconomic variables (Value-18)
 269

 Table 5.42 Univariate regression of value effect on individual lagged macroeconomic variables (Value-24)
 271

 Table 5.43 Multivariate regression of value effect on the combination of lagged macroeconomic variables
 273

Table 5.44 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 3 months' holding length) .......274

Table 5.45 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 6 months' holding length) .......275

Table 5.46 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 12 months' holding length) ......276

Table 5.47 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 18 months' holding length) .....277

Table 5.48 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 24 months' holding length) ......278

## **Chapter 6 The Recursive out-of-sample Forecasting Model on Style Effects**

Table 6.1 The beginning date on in-sample period and out-of-sample period whenlong/short portfolios are hold for 6 months312

Figure 6.2 The performance of series adjusted  $R^2$  in recursive in-sample regression of momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) on lagged macroeconomic variables from December 1989 to December 2007: Momentum effect (12/12) .....315

Figure 6.3 The performance of series adjusted  $R^2$  in recursive in-sample regression of size coefficient  $\gamma_{mv}$  (Small/Large return  $R_{sml}$ ) on lagged macroeconomic variables from December 1989 to December 2007: Size effect (12) ......316

Figure 6.7 'PIS', 'POOS', 'Sign', the performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to 2008: Size effect (12)

## **Chapter 7 Active Trading Strategy on Style Rotation**

Figure 7.1 Equal-weighted monthly Small/Large return R<sub>sml</sub> from 1980 to 2008 ...371

## **List of Abbreviations**

 $R_{lms}$ : Long/short return, the return spread between top/bottom decile portfolios sorted by firms' characteristics.

R<sub>wml</sub>: the return spread between 'Winner' portfolio and 'Loser' portfolio.

R<sub>sml</sub>: the return spread between 'Small' portfolio and 'Large' portfolio

R<sub>lmh</sub>: the return spread between 'Low' portfolio and 'High' portfolio.

 $\gamma$ : Style coefficient, the slope coefficient of the regression of decile portfolio returns in the holding horizon on their mean values of firms' characteristics in the formation horizon.

 $\gamma_{mom}$ : momentum coefficient, the slope coefficient of the regression of decile portfolios' holding returns on their formation returns;

 $\gamma_{mv}$ : size coefficient, the slope coefficient of the regression of decile portfolios' holding returns on their log mean market values at the formation time;

 $\gamma_{pb}$ : value coefficient, the slope coefficient of the regression of decile portfolios' holding returns on their price-to-book ratios at the formation time;

MEV: macroeconomic variable;

UEXMEV: unexpected macroeconomic variable;

GDP(Y): the previous annual change of GDP;

UEXGDP(Y): the unexpected annual change of GDP;

IP(i): the change of Industrial production over previous the i<sup>th</sup> month;

UEXIP(i): the unexpected change of Industrial production over previous i months;

IPM(i): the change of Industrial Production Index – Manufacturing over previous the  $i^{th}$  month;

UEXIPM(i): the unexpected change of Industrial Production Index – Manufacturing over previous i months;

M0(i): the change of Money Supply M0 over previous the i<sup>th</sup> month;

UEXM0(i): the unexpected change of Money Supply M0 over previous i months;

UNE(i): the change of Unemployment Rate over previous the i<sup>th</sup> month;

UEXUNE(i): the unexpected change of Unemployment Rate over previous i months;

CPI(i): the change of Consumer Price Index over previous the i<sup>th</sup> month;

UEXCPI(i): the unexpected change of Consumer Price Index over previous i months;

TBILL (i): the change of monthly return on a 90-day Treasury Bill over previous the  $i^{th}$  month;

UEXTBILL (i): the unexpected change of TBILL over previous i months;

TERM (i): the change of TERM over previous the  $i^{th}$  month. TERM: difference in the annualized yield on the long-term UK government bonds and the yield on the three-month Treasury Bill.

UEXTERM (i): the unexpected change of TERM over previous i months;

DY (i): the change of Dividend Yield over previous the i<sup>th</sup> months. DY is estimated based on value-weighted market index of all companies;

UEXDY (i): the unexpected change of Dividend Yield over previous i months;

EMR/VMR(-y): equal-(value-) weighted y years' compounded return; y=1,2,...10;

PIS: the statistics to measure the performance of recursive in-sample regression model relative to sample mean model;

POOS: the statistics to measure the performance of recursive out-of-sample forecasting model relative to sample mean model;

PSS: the percentage having same sign between actual and predicted style coefficient  $\gamma$  (Long/Short return  $R_{lms}$ ), which is estimated from out-of-sample forecasting model.

## **Chapter 1 Introduction to the Research**

Traditionally, technical analysis and fundamental analysis are two main investment strategies in the stock market. Technical analysis focuses on movements of share prices, and uses this data to forecast their movements in the future. It is based on the idea that share prices reflect all information in firms' fundamentals. On the other hand, fundamental analysis pays attention to economic factors in firms' financial statements to determine their intrinsic values. Long position is allocated to shares whose prices are below their intrinsic values, and short position is allocated to shares whose prices are above their intrinsic values.

Different from technical analysis and fundamental analysis, style investment in the stock market focuses on a particular class of stocks that share common characteristics, such as past return, market value, and price-to-book ratio. Dimson and Nagel (2001) propose that comovement of prices within a distinct group of stocks contributes to the definition of a style, and that a broad classification of shares with similar characteristics into styles may help investors to ease their burden in processing enormous information flow and simplify problems of choice. As Barberis and Shleifer (2003) argue, the grouping of shares into categories based on some similarity among them is called 'styles', and this process itself, namely allocating funds among styles rather than among individual shares, is known as 'style investing.' The categorization of style investing is attractive to institutional investors who follow fundamental rules of portfolio allocation in asset management.

Stocks can be classified into different categories, for example, winner versus loser, small versus large, value versus growth. Since the later 1970s some academic studies have focused on return difference between them, and found that stocks with high past 6/or 12 months' return outperform stocks with low past 6/or 12 months' return, small-cap stocks outperform large-cap stocks, and stocks with high book-to-market ratio outperform stocks with low book-to-market ratio. Cross-sectional variation in share returns based on these firm-specific characteristics is defined as return anomalies or style effects.

The thesis investigates the time-variation in style effects in the UK Stock Market, which include momentum, size, and value effect. Different from most previous studies on style effects, the thesis explores the dynamic pattern of cross-sectional variation in share returns in the UK Stock Market in terms of momentum, size, and value effect, and what factors drive this time-varying pattern. Furthermore, the thesis tests recursive in-sample regression and out-of-sample forecasting model, and explores the profitability of active trading strategy on style effects.

As the beginning part of the thesis, Chapter 1 consists of four sections, which introduces motivation for the research, research questions, structure of the thesis, and contributions and implications of the research, respectively.

## **1.1Motivation for the research**

The research investigates the time-variation in style effects in the UK Stock Market due to three reasons, which are associated with research background, underlying empirical studies, and market characteristics. It is motivated by the gap between the academic derivation on consistent style effects in the long run in most existing studies and their dynamic patterns in the market. The research aims to provide insights into time-varying cross-sectional variation in share returns, examine explanatory factors for this dynamic patterns, test the reliability/or pitfall of insample predictability, and further explore how to develop active trading strategies to capture style rotation in the UK Stock Market.

## 1.1.1 Research background

As return anomalies in the stock market, cross-sectional variation in share returns has been eye-caught in the academic field since the later 1970s. The dispersion in specific firm characteristics, such as past share return, size, and book-to-market ratio, are associated with cross-sectional variation in share returns. The majority of studies focus on the static version of style effects in the overall period, but pay relatively little attention to time-varying property of style effects. These studies only investigate whether the sample mean of the variation in cross-sectional share returns is significantly different from zero in the overall period, but do not mention any time series properties of return anomalies other than the mean and t-ratio. They propose that at any time point there is a linear and positive (or negative) relationship between firms' specific characteristics and cross-sectional share returns. The static style effect is derived by examining return anomalies in only one period. In the view of the thesis, this is not a rational derivation. In the investing community, stocks live for many periods, which are associated with dynamic macroeconomic conditions and market states. Therefore, in the empirical examination of style effects, it is instructive to make certain assumption that style effects are volatile over time. The investigation on time-varying style effects can provide insights into dynamic characteristics of style effects and the practical implication for the adjustment of portfolio allocation over time.

In examining the source of value premium and its dynamics, Santos and Veronesi (2005) argue that quantitative conclusion on the cross-sectional pattern may be quite misleading if the explanation for cross-sectional share returns is independent of the time-series properties of the market portfolio. In reviewing methodologies and empirical evidence on cross-sectional variation in share returns, Goyal (2012) presents that there are relatively few literature for the case dealing with time-varying characteristics although firm characteristics that vary over time is always employed in practice. He argues that using time-varying characteristics in cross-sectional regression leads to biased and inconsistent estimators if the expected returns depend on average characteristics. Therefore, based on the rational derivation that style effects are always not in the time-invariant case, the thesis examines the time-variation in style effects in the UK Stock Market, and the conditions under which dynamic style effects arise or persist.

The research on time-varying style effects is based on underlying empirical studies on the source of the variation in cross-sectional share returns, time-varying firms' fundamentals, and time-varying risk preference. The interaction between crosssectional dispersion in cash-flow risks of firms and time-varying firms' fundamentals/and risk preference contributes to the time-variation in style effects. Santos and Veronesi (2005) propose a theoretical model and empirical work and find that cross-sectional dispersion in cash-flow risk between value stocks and growth stocks is enough large to capture value premium, and that its interaction with series variation in risk preference leads to dynamic value premium.

#### The source of the variation in cross-sectional share returns

The present value model proposes that share prices reflect series present discounted expected future dividends, and that their movements reflect the change in expected dividends and discount rate. The change in expectations of dividends in the future is an important factor determining share returns. The present value model justifies that cross-sectional variation in share returns results from the difference in expected future dividends, which are associated with cash flow. Some papers document this rational intuition. For instance, Cohen, Polk, and Vuolteenaho (2003) decompose cross-sectional variance of book-to-market ratio (BE/ME) into three components: (a) covariance of future stock returns with past BE/ME ratios; (b) covariance of future profitability with past BE/ME ratios, and (c) persistence of BE/ME ratios. They find that about 20 to 25 percent of the value spread is explained by expected stock

returns, and 75 to 80 percent of the value spread is explained by expected profitability (cash flows). Using cash flow risk measured by aggregate consumption and cash flow growth rates, Bansal, Dittmar, and Lundblad (2005) argue that cash flow risk is an important factor to capture the difference in risk premia across shares. They find that the risk measure can explain more than 60 percent of cross-sectional variation in mean returns of momentum, size, and value sorted portfolios. Hansen, Heaton, and Li (2008) show that value shares' cash flows positively covary in the long run with aggregate consumption growth more than growth shares, which implies that two kinds of shares are characterized by different cash flow risks due to dynamic macroeconomic growth. Campbell, Polk, and Vuolteenaho (2010) explore the economic origins of systematic risks for value shares and growth shares. They find that cash flows of value/growth shares are particularly sensitive to permanent/temporary movements, driven by shocks to aggregate cash flows/market discount rates.

## Time-varying firms' fundamentals

The common expectation is that the fundamental variation in cyclical behaviour should exist across firms with specific characteristics through different stages in business cycle. The impacts of business expansion or contraction on small firms and large firms are not identical. Tamari (1984) uses three indicators, such as sales, profitability, and current liabilities, to examine the reaction of small firms and large firms to the change in US economic conditions. He finds that small firms are affected adversely to a greater degree than large firms in three recession periods from later 1950s to mid-1970s. Sales of small firms suffered more than large firms during periods of reduced demand, which was followed by the greater fall in their incomes and profits. Gertler and Gilchrist (1994) present empirical evidence on the cyclical behaviour of US small and large manufacturing firms, and on their differential response to monetary transmission. They argue that the rise in interest rate, due to tightening monetary policy, directly hits balance sheets by reducing cash flows and by lowering the value of collateral assets, which magnifies the effect of monetary policy on borrowers' costs. Using three variables, such as sales, inventories, and short-term debt, they find that small firms contracted much more than large firms after tightening monetary policy and during recessions from 1960 to 1991. Ehrmann (2000) finds the same asymmetric effect of tightening monetary policy on small and large firms in Germany. Grieb and Reyes (2002) find that information to UK Smallcap Index negatively impacts its next period correlation with UK Large-cap Index from 1955 to 1996, which reflects the difference in the sensitivity of small-cap and large-cap stock prices to the same macroeconomic variables. For example, small firms, as net importers, gain from the pound appreciation. Conversely, large firms, as net exporters, suffer from this movement in exchange rates.

## Time-varying risk preference

It is widely accepted that the price of risk changes over time with business cycle. Fama and French (1989), Ferson and Harvey (1991) empirically document that equity risk premia exhibits countercyclical pattern, higher at business cycle troughs and lower at business cycle peak. With consumption-based asset pricing model, Campbell and Cochrane (1999), Lettau and Ludvigson (2001) present economic explanation for the fundamental source of the price of risk and its time-varying pattern. They argue that investors' risk preference (or aversion) is closely associated with their consumption habit. In Campbell and Cochrane (1999) slow-moving habit model, consumption booms exist in the period when consumption rises above habit, followed by the decline in risk aversion and in turn greater demand for risky assets. The consumption-wealth ratio in Lettau and Ludvigson (2001) is related to investors' risk preference. They analyze its economic intuition as: Investors increase (decrease) consumption out of asset wealth and labour income when they expect higher (lower) excess return in the future. In this way, they maintain a smooth consumption path over time and keep future consumption from the time-variation in expected returns. The process in which investors adjust consumption-wealth ratio to optimize consumption behaviour always precedes the movement in share returns, reflecting the existence of time-varying risk preference. Smith and Whitelaw (2009) find that risk premia varies counter-cyclically, and risk aversion increases on the course of economic contractions.

As these studies assume the change in consumption habit as the source of timevarying risk preference in the rational dimension of financial theory, others view psychological traits in investors as rational explanation for the variation. As a competing paradigm against the traditional financial theory, behavioural finance integrates insights from psychology with neo-classical financial theory. As Shleifer (2000) and Montier (2007) state, five psychological traits inherent in most investors, such as overoptimistic, overconfidence, conservatism bias, anchoring, and representativeness heuristic, influence investors' risk preference to both aggregate market and individual share. Investors become confident and overoptimistic in economic expansion and bull market, which leads to the increase in risk preference and the decline in risk premia. Conversely, investors become overpessimistic in economic contraction and bear market, which leads to the decline in risk preference and the increase in risk premia. Daniel, Hirshleifer, and Subrahmanyam (1998), Hong and Stein (1999) present that behavioural or cognitive biases can explain the variation in risk preference, and further momentum effect and reversal effect in share returns. Some studies document that irrational sentiment of investors may lead to common movement of share prices that is not related to firm fundamentals. Barberis, Shleifer, and Wurgler (2005) find that shares added to the S&P500 exhibit substantial and significant increase in their beta with the S&P500 and substantial and significant decrease in their betas with the rest of the market, and that the change in S&P betas was quantitatively larger after 1988. Greenwood (2005) presents similar results on the redefinition of the Nikkei 225 Index in Japan Stock Market.

#### 1.1.3 Market characteristics

As academic research focused on cross-sectinal predictability, the rapid development of style investing in the market has driven the introduction of style index, such as small-/large-cap index and value/growth index. MSCI launched a series of international and regional Value/Growth Indices based on large-cap and mid-cap in December 1974. The value attribute for index construction is defined by three measures, such as book-to-price ratio, 12-months forward earnings to price ratio, and dividend yield. The growth attribute for index construction is defined by five measures, such as long-term forward earnings per share (EPS) growth rate, shortterm forward EPS growth rate, current internal growth rate, long-term historical EPS growth trend, long-term historical sales per share growth trend. MSCI further introduced international and regional Value/Growth Indices based on large-cap, mid-cap, and small-cap, Large-cap/Small-cap Indices in December 1992, and Mid-cap Index in May 1994. Similarly, the FTSE indices series established FTSE UK Small-cap Index in 1986, and FTSE UK Value Index and Growth Index in 1993. The Russell indices introduced the full set of style indices, which represent the performance of value/growth shares in small-cap, mid-cap, and large-cap categories in 1986. Advanced Quantitative Research Capital Management introduced the AQR Momentum Index in 2009.

## The performance of Value/Growth Index in US, UK, and Japan Stock Market

Style index uses clear and consistent sets of attributes and rigorous methodological framework to categorize shares. It can help us to observe dynamic performance of style effects over time. The research collects MSCI Value/Growth Indices (Large+Mid Cap) in US, UK, and Japan Stock Market from 1975 to 2009, which are plotted in Figure 1.1. The first row in Figure 1.1 clearly show outperformance of Value Index (solid line) over Growth Index (dotted line) in US Stock Market before 1990, and the opposite afterwards. The second and third row in Figure 1.1 shows that Value Index (solid line) always performs above Growth Index (dotted line) in UK and Japan Stock Market from 1975 to 2009.

[Figure 1.1]

Table 1.1 shows that from 1975 to 2009 US Growth Index outperforms US Value Index by 396 percent, UK Value Index outperforms UK Growth Index by 540 percent, and Japan Value Index dominantly outperforms Japan Growth Index by 2726 percent. If splitting the whole sample period into 7 five-year's subperiods, Value Index vs Growth Index is volatile over subperiods. Different from the pattern in overall period, Value Index outperforms Growth Index in the 1<sup>st</sup>, 2<sup>nd</sup>, and 6<sup>th</sup> subperiods in US Stock Market, Growth Index outperforms Value Index in the 2<sup>nd</sup>, 5<sup>th</sup>, and 7<sup>th</sup> subperiods in UK Stock Market, Growth Index outperforms Value Index in the 5<sup>th</sup> subperiods in Japan Stock Market. If we observe the return spread between Value Index and Growth Index each month as shown in Table 1.2, we can clearly find no obvious difference in the percentage having positive and negative monthly long/short return between two styles in US and UK Stock Market. In particular, the magnitude of average monthly negative return spread is slightly higher than positive one. Even though value effect dominates in Japan Stock Market, the return spread yields average negative return of 2.30 per month, slightly 14 basis points lower than positive return spread. Growth Index outperforms Value Index in 42.62 percent of 420 months.

## [Table 1.1, 1.2]

#### The performance of Small-cap/Large-cap Index in US, UK, and Japan Stock Market

The research collects MSCI Small-cap/Large-cap indices for US, UK, and Japan Stock Market from May 1994 to December 2009, which is plotted in Figure 1.2. The first row in Figure 1.2 shows that Large-cap Index (dotted line) performs above

Small-cap Index (solid line) during 1994 to early 2000, and vice versa after then until 2009 in US Stock Market. The middle row shows that Small-cap Index (solid line) performs above Large-cap Index (dotted line) from early the 2000s until financial crisis happened in UK Stock Market. The bottom row shows that Large-cap Index (dotted line) performs above Small-cap Index (solid line) in most time of the period in Japan Stock Market.

## [Figure 1.2]

When observing the holding return in Table 1.3, we find the difference in index return between overall period and 3 five-year's subperiods. In US Stock Market, although Small-cap Index yields return 124 percent higher than Large-cap Index from May 1994 to December 2009, its performance is poor relative to Large-cap Index by 128 percent from May 1994 to 1999. Similar pattern exist in UK Stock Market. Although in Japan Stock Market Large-cap Index outperforms Small-cap Index in the overall period, Small-cap Index performs well in the second subperiod from 2000 to 2004. Table 1.4 shows that if we observe the return spread of Small-cap Index and Large-cap Index each month, we can clearly find no obvious difference between two styles in US, UK and Japan Stock Market, with almost same percentage having positive and negative Small/Large return. Except for US Stock Market, the magnitude of average monthly positive and negative Small/Large return is almost the same in UK and Japan Stock Market.

## [Table 1.3, 1.4]

The introduction of various style indices in the market further motivates academic research on style index. Using US Russell style indices, Kao and Shumaker (1999) show that from 1979 to 1997 Large-cap/Value Index outperformed the Growth Index in 62 percent of the time in US Stock Market. To construct style indices in UK Stock Market, Levis and Liodakis (1999) find that Small-cap Index performs well in 183 months (or 53% of the time), while Large-cap Index is better off the rest of the 348 months (or 47% of the time) from 1968 to 1997. Large-cap Index earns average annual return higher than Small-cap Index does from July 1988 to June 1997 after the small-cap premium during two decades. They conclude that the value/growth return spread is positive in 232 months (or 67% of the total period), and negative in 116 months (or 33% of the total period). Chan, Karceski, and Lakonishok (2000), Barberis and Shleifer (2003) come to similar conclusion in US Stock Market. Chen and De Bondt (2004) note that for large US companies in the S&P-500 Index between 1977 and 2000 investors could benefit from chasing style indices that were successful over the previous 3 to 12 months. Berger, Israel, and Moskowitz (2009) present that the AQR Momentum Index outperforms Value Index and Growth Index. By comparing style indices, they mention that momentum effect and growth effect are positively correlated, and that momentum effect and value effect are negatively correlated.

## **1.2 Research questions**

The interaction of cross-sectional variation in cash-flow risk of firms and timevarying firms' fundamentals/and investors' risk preference leads to one implication that style effects reasonably depend on dynamic nature of business conditions and market states at any time point and thus vary over time. The thesis attempts to measure and interpret whether the time-variation in style effects based on past share return, market value, and price-to-book value relates to macroeconomic variables in the UK Stock Market. Another purpose is to investigate active trading strategy to time style rotation. The thesis specifically centres on following four issues, which are empirically analysed in Chapter 4 to Chapter 7:

Firstly, whether does the pattern of time-variation in style effects exist? The research examines momentum, size, and value effect with 3, 6, 12, 18, and 24 months' holding horizons, to deepen understanding of style effects with short-/medium-/long-term horizons. Furthermore, the research investigates the correlation among style effects given the assumption that some macroeconomic variables might have same impact on different style effects.

Secondly, what factors are associated with the time-variation in style effects and how they capture the dynamic properties in overall sample period and subperiods? The research examines the correlation between the time-variation in style effects and lagged macroeconomic variables given the previous empirical findings on timevarying firms' fundamentals and time-varying risk preference exposure to the volatility in economic conditions and market states, which drives the time-variation in style effects.

Thirdly, if some lagged macroeconomic variables can explain the time-variation in style effects, it is necessary to ask whether this dynamics can be indeed predictable by using recursive in-sample regression and out-of-sample forecasting model. Some studies find pitfalls of in-sample predictability of variables such as dividend yield, earning price ratio, book-to-market ratio, and short-term interest rate. Although insample predictability exists, out-of-sample forecasting model performs poorly. With out-of-sample forecasting model, the research tests explanatory power of macroeconomic variables on time-varying style effects over time. Meanwhile, recursive in-sample regression and out-of-sample forecasting model can help to find whether explanatory power and forecasting power of macroeconomic variables are stable or volatile. Furthermore, the test result based on recursive out-of-sample model can be applied in the foundation of exploring active trading strategy on style rotation.

Finally, how to establish recursive out-of-sample forecasting model to benefit a variety of active trading strategies on style effects in the market? Due to dynamic economic conditions and market states, time-varying style effects lead to the failure in traditional passive strategy on long/short return all the time. For example, many studies find that small-cap premium disappeared after the 1980s, and value effect underperformed growth effect in the later 1990s in some major stock markets. So, to establish forecasting model to time the significant period of individual style effect

would be rather important in practical investment. It is instructive to examine whether investors can actually use forecasts of macroeconomic variables in making investment decision on style rotation.
## **1.3 Structure of the thesis**

The thesis consists of eight chapters, with the first chapter on introduction and the final chapter on overall conclusion. The second and third chapters mainly offer description on literature review, data and portfolio formation, respectively. The following four chapters contain corresponding research methods and empirical analysis on four research questions, respectively, which are raised in the previous subsection.

Chapter 2 reviews previous studies on cross-sectional predictability, time-series return predictability, and the time-variation in cross-sectinal predictability, respectively. This chapter focuses on relevant studies on return anomalies in terms of momentum, size, and value effect, the correlation between aggregate market return and some macroeconomic variables, and the time-variation in return anomalies in the UK Stock Market as well as other international markets.

Chapter 3 describes data and portfolio formation in the research. It details data source, firm characteristics, holding portfolio returns, and approaches to construct momentum portfolio, size portfolio, and value portfolio.

Chapter 4 concentrates on the first research question on how momentum, size, and value effect perform in the UK Stock Market. Using two indicators, such as style coefficient  $\gamma$  and long/short return R<sub>lms</sub>, it investigates time-varying characteristics of individual style effect with 3, 6, 12, 18, and 24 months' holding lengths in overall

sample period and subperiods. Style coefficient  $\gamma$  is estimated by regressing stylesorted decile portfolio returns on the value of firm characteristics, and so includes the information of all shares' firm-specific characteristics and returns. Long/short return  $R_{lms}$  is estimated by the difference in top/bottom portfolio return. In addition, the research discusses the correlation among style effects.

Chapter 5 deals with the second research question on the correlation between timevarying style effects and lagged macroeconomic variables in overall sample period and subperiods. Using time-series regression, the research explores what lagged economic forces and market variables capture the variation in style coefficient  $\gamma$  and long/short return R<sub>lms</sub>, and how they explain the volatility of style effects with 3, 6, 12, 18, and 24 months' holding horizons.

Chapter 6 discusses the third research question on whether the time-variation in style effects is indeed predictable by some lagged macroeconomic variables. It uses statistic indicators to test the performance of recursive in-sample regression and out-of-sample forecasting model on time-varying style effects. The empirical work only focuses on momentum effect with 12 months' formation length and 6/12 months' holding length, size effect with 12 months' holding length, and value effect with 24 months' holding length, the most strongest or stronger effect among the cases with five holding horizons as found in Chapter 4. Meanwhile, relative to style effects with other holding horizons, their time-varying properties are captured much more by macroeconomic variables as documented in Chapter 5.

Chapter 7 deals with the fourth research question on how to improve recursive outof-sample forecasting model to benefit a variety of active trading strategies for timing style rotation. Following empirical findings in Chapter 6, it uses the signal from recursive out-of-sample forecasting model and filter thresholds, and explores the possibility of performance enhancement of active trading strategies relative to traditional passive trading strategies with long/short position all the time. This exploration on trading strategies is based on individual style effect and combined style effects.

## **1.4 Contributions and implications of the research**

The thesis focuses on four research questions on the time-variation in style effects in the UK Stock Market. The empirical findings make contributions to studies on style effects, and offer implications for researchers in academic field and style-oriented investors in the market.

## 1.4.1 Contributions of the research

Firstly, although some existing studies have empirically examined the time-variation in style effects since the year 2000, they always focus on individual style effect, or two style effects with one certain holding horizon, such as 6 or 12 months. Because the thesis examines three style effects in one empirical research, it can help us to find the correlation between momentum effect and size effect, between momentum effect and vale effect, and between size effect and vale effect. In addition, because the thesis examines style effects with 3, 6, 12, 18, and 24 months' holding horizons, it can offer us one approach to observe the difference in style effects with short-/medium-/long-term horizon, and the variation in significant explanatory factors for style effects through holding horizons.

Secondly, besides traditional indicator long/short return  $R_{lms}$  in previous studies, the thesis uses style coefficient  $\gamma$  to investigate time-varying style effects. Different from long/short return  $R_{lms}$  based on top/bottom portfolios, style coefficient  $\gamma$  is estimated by cross-sectional regression of decile portfolios' returns in the holding period on

their average values of firm characteristics in the formation period. It contains the information of all shares in decile portfolios in terms of share return and firm characteristics. So, style coefficient  $\gamma$  could capture much more information on the relation between share return and firm characteristics than long/short return R<sub>lms</sub>. Meanwhile, the test on the impact of lagged macroeconomic variables on the time-variation in style coefficient  $\gamma$  helps us to diagnose the efficiency of active trading strategy which allocates long position and short position on any decile portfolio according to portfolio returns predicted from out-of-sample forecasting model.

Thirdly, all previous studies only investigate the time-variation in style effects and explain their dynamic property in the context of macroeconomic conditions and market states in the overall sample period. The thesis undertakes these studies in subperiods as well as the overall period. The empirical studies in subperiods help us to find the stability/or volatility of significant macroeconomic variables to explain the time-variation in style effects.

Fourthly, Goyal and Welch (2003, 2008) find that many predictor variables for aggregate market return perform poorly in out-of-sample test relative to the prevailing sample mean. Although some previous studies find significant explanatory power of macroeconomic factors for the time-variation in style effects, they do not undertake the test on out-of-sample forecasting model. The thesis uses statistic measures PIS, POOS, and PSS to test the recursive in-sample regression and out-of-sample forecasting model on style effects, and verify whether macroeconomic factors

indeed offer predictive power for time-varying style effects. The test helps us to find the reliability/or pitfall of in-sample predictability on style effects.

#### 1.4.2 Implications of the research

Academics could benefit from the empirical findings in the thesis. The thesis finds the variation in significant macroeconomic variables for one style effect with different holding horizons and over time. This implies that researchers should examine empirical work over different holding horizons and subperiods when they undertake studies on series return predictability. In addition, the thesis finds that outof-sample forecasting model on value effect (24) fails to outperform the sample mean model although its in-sample regression model has good performance. This means that it is necessary for researchers to conduct out-of-sample model test with two statistics, POOS and PSS, in order to examine the reliability/or pitfall in-sample predictability.

Empirical findings have implications for practitioners who favour style investing. Due to the existence of biased and inconsistent significant macroeconomic variables for the time-variation in style effects with short-/medium-/long-horizons and over time, style-oriented investors should be cautious to use forecasting variables to time style rotation according to their holding horizons and macroeconomic conditions. In addition, the thesis develops a variety of active trading strategies on long/short portfolio for individual style effect and combined style effects. Active trading strategy with either any equal-weighted decile momentum portfolio (12/6) or valueweighted Winner/Loser portfolio (12/6) seems to be one good choice for styleoriented investors in the UK Stock Market.

# Appendix: tables and figures in Chapter 1





Note: The figure is plotted by series monthly MSCI Value Index and Growth Index from the MSCI Global Equity Indices.

Table 1.1 The holding return of MSCI Value Index and Growth Index in US, UK, and Japan Stock Market in subperiods and overall period from 1975 to 2009 (100/%)

	US		U	UK		ban
Subperiod	Value	Growth	Value	Growth	Value	Growth
1975-1979	0.6352	0.3210	2.3564	1.8887	1.3784	0.7446
1980-1984	0.5475	0.4665	0.1836	0.4193	1.2114	0.8554
1985-1989	0.8959	1.2359	2.3633	1.3479	5.3946	3.7740
1990-1994	0.2450	0.4121	0.2592	0.2527	-0.0369	-0.3274
1995-1999	1.5551	3.3143	1.0471	1.2839	0.0183	0.1135
2000-2004	-0.0351	-0.3805	-0.0102	-0.2346	-0.0567	-0.5113
2005-2009	-0.1464	0.0111	-0.1922	0.0500	0.0176	-0.2214
Overall period:1975-2009	11.5699	15.5291	26.5378	21.1339	30.6621	3.4037

Note: The holding return over 5-year's subperiods and overall period is the percentage by which the value of Value/Growth Index has grown for a particular period. Series monthly MSCI Value Index and Growth Index in US, UK, and Japan Stock Market is collected from the MSCI Global Equity Indices.

Table 1.2 The average monthly return spread of MSCI Value Index and Growth Index in US, UK, and Japan Stock Market from 1975 to 2009

	Return spread	Number	Percentage (%)	Average monthly return (%)
US	+	210	50.00	1.87
	-	210	50.00	-2.06
UK	+	217	51.67	2.11
	-	203	48.33	-2.14
Japan	+	241	57.38	2.44
	-	179	42.62	-2.30

Note: The average monthly return spread of MSCI Value Index and Growth Index in US, UK, and Japan Stock Market is estimated by data from the MSCI Global Equity Indices.

Figure 1.2 MSCI Small-cap Index and Large-cap Index in US, UK, and Japan Stock Market from May 1994 to December 2009



Note: The figure is plotted by series monthly MSCI Small Index and Large Index from the MSCI Global Equity Indices.

Table 1.3 The holding return of MSCI Small-cap Index and Large-cap Index in US, UK, and Japan Stock Market in subperiods and overall period from May 1994 to 2009 (100/%)

	US		UK		Japan			
Subperiod	Small-cap	Large-cap	Small-cap	Large-cap	Small-cap	Large-cap		
1994/5-1999	1.1299	2.4088	0.6106	1.4904	-0.5158	0.2853		
2000-2004	0.6911	-0.2648	0.3149	-0.1680	0.1975	-0.4496		
2005-2009	-0.0094	-0.0710	-0.0299	-0.0708	-0.1924	-0.1064		
Overall period: 1994/5-2009	2.5681	1.3281	1.0544	0.9253	-0.5318	-0.3679		

Note: The holding return over 5-year's subperiods and overall period is the percentage by which the value of Small-cap/Large-cap Index has grown for a particular period. Series monthly MSCI Small-cap Index and Large-cap Index in US, UK, and Japan Stock Market is collected from the MSCI Global Equity Indices.

Table 1.4 The average monthly return performance of MSCI Small-cap Index minusLarge-cap Index in US, UK, and Japan Stock Market from May 1994 to December 2009

	Return spread	Number	Percentage (%)	Average monthly return (%)
US	+	94	50.27	3.12
	-	93	49.73	-2.53
UK	+	92	49.20	1.18
	-	95	50.80	-1.14
Japan	+	92	49.20	3.11
	-	95	50.80	-3.17

Note: The average monthly return spread of MSCI Small-cap Index and Large-cap Index in US, UK, and Japan Stock Market is estimated by data from the MSCI Global Equity Indices.

## **Chapter 2 Literature Review**

The Capital Asset Pricing Model of Sharpe (1964), Lintner (1965), and Black (1972) has long shaped the way that academics and practitioners think about average share returns and risk. The central argument of the model is that invested market portfolio is mean-variance efficient. The efficiency of market portfolio implies that expected security return is a positive linear function of its market beta (the slope in the regression of the security return on the market return), and market beta suffices to describe the cross-sectional variation in share returns. The theory of CAPM identifies market portfolio return as the only important factor in the pricing of risky assets, and assumes that betas remain constant over time.

Since the Efficient Market Hypothesis (EMH) was introduced in Fama (1970), many studies on the EMH has often used the CAPM to test market efficiency. As further and deep academic discussion on EMH, more and more research work has cast doubt on the capability of Sharpe-Lintner-Black (SLB) model to explain cross-sectional variation in share returns. Fama and French (1992) investigate the static pattern of CAPM and find flat relation between market beta and average return. Fama and French (1992), Jagannathan and Wang (1996) present evidence that the static CAPM fails to explain cross-sectional share returns based on size/beta portfolios. Santos and Veronesi (2005) find the failure of the CAPM to generate cross-sectional variation in their risk estimators across book-to-market sorted decile portfolios. Other empirical studies find that the source of risk in CAPM does not completely account for the cross-sectional expected returns, and present that one or more additional factors may

capture the performance of expected returns. They show that average returns on common stocks are correlated to firm characteristics and financial ratios, such as share size, earnings/price, cash flow/price, book-to-market equity, past sales growth, long-(short-) term past return. Because these patterns in average returns apparently are not explained by the CAPM, they are called 'return anomalies'.

Early in the past century, statisticians noticed that the change in stock prices seems to follow a fair game pattern. In 1900, French mathematician Louis Bachelier proposed the Random Walk Hypothesis (RWH) in his PhD thesis, which states that the movement in stock prices is random, like the steps taken by a drunk, and therefore is unpredictable. Similar to the idea of the RWH, the Efficient Market Hypothesis (EMH) in Fama (1970) seeks to explain the random walk hypothesis by positing that only new information will move stock prices significantly, and since new information is presently unknown and occurs at random, future movements in stock prices are also unknown and thus move randomly. The EMH states that financial markets are efficient, and that share prices already reflect all known information in the market or rapidly adjust to any new information. Hence, the EMH implies that share prices must follow a random walk, and that it is impossible to outperform the market by picking undervalued and overvalued stocks in the efficient market. However, two hypothetical theories of RWH and EMH have been challenged by many empirical findings since the early 1980s that share prices seem not to completely reflect all known information in the market, and that their movements seem not to be unpredictable. For example, stocks with low price to earnings, cash flow, or book value outpeform the others. More studies of linking economic

variables to aggregate stock market return have been motivated after Ross (1976) introduced the Arbitrage Pricing Theory (APT). As an alternative to the CAPM, APT is more general than CAPM and allows multiple risk factors to capture asset returns besides market portfolio. The change in economic forces could be viewed as one underlying risk factor for stock market return. Some economic and market indicators, such as industrial production, inflation, interest rate, and dividend yield, are significantly correlated to forthcoming aggregate stock return.

Obviously, in the context of classical finance theories, such as CAPM, RWH, and EMH, stocks always move at their fair price, neither undervalued nor overvalued. It should be impossible for any professional investment to perform well than the aggregate market by using stock selection and market timing. This inference has been contradicted and challenged by empirical work on return predictability in terms of cross-sectional and time-series dimensions. The research on cross-sectional return predictability is concerned about return anomalies in terms of momentum, size, and value effect, and tests of conditional CAPM to capture cross-sectional variation in share returns. The research on time-series return predictability focuses on the forecasting power of macroeconomic variables and market variables on subsequent aggregate market return. Therefore, the developing process of modern finance theory sheds some light on return predictability in the market.

## 2.1 Cross-sectional return predictability

The research on examination of cross-sectinal return predictability stems from academic tests of CAPM. Since the later 1970s, a large amount of studies have found cross-sectional return predictability in documenting the shortcomings of SLB model and EMH. They argue that cross-sectional share returns could be associated with some firms' specific characteristics.

## 2.1.1 Cross-sectional return predictability in the US Stock Market

Many empirical papers have documented the existence of cross-sectinal return predictability in the US Stock Market. They conduct cross-sectional research with regarding to the prediction of firm-specific characteristics on expected returns. For example, Basu (1977) shows that price/earnings ratio (P/E) has explanatory power on the cross-sectional variation in share returns during the period of April 1957 to March 1971. The low P/E portfolio earned higher absolute and risk-adjusted return than the high P/E portfolio. Basu (1983) further supports the P/E effect after controlling the difference in firm size from 1963 to 1979. It is found in Banz (1981) that small firms had significantly larger risk-adjusted returns than large firms from 1926-1975. Rosenberg, Reid, and Lanstein (1985) present that the return of strategy, which buys stocks with high book-to-market ratio and sells ones with low ratio, is significantly positive from 1973 to 1984. Fama and French (1993) find that size and BE/ME capture the cross-sectional variation in average stock returns associated with other variables such as earnings/price, cash flow/price, and sales growth from 1963

to 1991. They present that cross-sectional return variance could be captured by a three-factor model of market return, the spread in the returns on portfolios of small stocks and large stocks, and the spread in the returns on portfolios of high BE/ME stocks and low BE/ME stocks. Fama and French (1996) further show that the three-factor model captures long-term return reversal. Asness (1997) proposes that value variables, such as book-to-market ratio and dividend-price ratio are positively associated with future expected returns. Consistent with findings on 25 size/value sorted portfolios by Fama and French (1993), Hodrick and Zhang (2001) presents a monotonic increase in average returns as book-to-market ratio increases within a size quintile, and the average return on the smallest firms higher than the average return on the largest firms within a book-to-market quintile, except for the lowest book-to-market one, from 1952 to 1997. Goyal (2012) finds up-to-date evidence of positive but statistically unsignficant small-cap premium, and positive and statistically significant value premium over the sample period of 1946 to 2010.

Other empirical studies focus on return anomalies based on share price levels in different horizons. De Bondt and Thaler (1985, 1987) state that prior "losers" in past three to five years are found to outperform prior "winners". Jegadeesh (1990) documents that monthly returns on individual stocks exhibit significantly positive serial correlation at three-month, six-month, and one-year lags, which is designated as momentum effect. Jegadeesh and Titman (1993, 2001), Chan, Jegadeesh, and Lakonishok (1996), and Asness (1997) further document return momentum. Figelman (2007a, 2007b), Goyal (2012) systematically offer the existence of short-term reversal effect, intermediate-term momentum effect, and long-term reversal

effect. It is found in Heston and Sadka (2008) that positive return of winner/loser strategy with one-year's holding horizon is concentrated around financial announcements by firms.

2.1.2 Cross-sectional return predictability in the international and UK Stock Market

Besides in the US Stock Market, the cross-sectional variation in share returns has been found in other stock markets. Chan, Hamao, and Lakonishok (1991) find that book-to-market ratio and cash flow yield have significantly positive impact on expected returns in the Japan Stock Market. Brouwer, Put, and Veld (1996) conclude that in France, Germany, the Netherlands and the UK Stock Market among four value variables such as earnings-to-price (E/P) ratio, cash-flow-to-price (CF/P) ratio, book-to-market (B/M) ratio, and dividend yield, the remarkable force to drive the cross-sectional variation in share returns stems from the CF/P ratio. Fama and French (1992, 1996, 1998), Lakonishok, Shleifer, and Vishny (1994), Davis, Fama, and French (2000) further test the relation between expected returns and book-to-market ratio, and find value premium in other major markets. Asness, Moskowitz, and Pedersen (2009), Chui, Titman, and Wei (2010) show positive momentum returns in 37 share markets. Fama and French (2011) present significant value premium in Europe, Japan, and Asia Pacific, and significant momentum premium in North America, Europe, and Asia Pacific. Asness (2011) finds significant value long/short return in the UK and Japan Stock Market and significant momentum long/short return in the UK and Europe Stock Market from 1981 to 2010. Hou, Karolyi, and Kho (2011) find that the cross-sectional variation in share returns across 49 markets might be explained by firm characteristics, such as size, dividend yield, earnings yield, book-to-market equity, cash flow-to-price, and share returns.

There is similar empirical research on cross-sectional return predictability in the UK Stock Market. Poon and Taylor (1991) report average annual return premium of 6.5 percent for smaller firms from 1958 to 1982. But Clare and Thomas (1994) find that large firms outperform small firms from 1983 to 1990. Levis and Liodakis (1999) show that average equal-(value-) weighted annual return of value portfolio is 23.58% (22.31%), over 11 percentage points higher than the return of growth portfolio from 1968 to 1997. After classifying companies whose market values make up the bottom 10% (the top 80%) of the total equity capitalization into the small-cap (large-cap) portfolio, they find that average equal-(value-) weighted annual return of small-cap portfolio is 17.51% (15.63%), 83 (77) basis points higher (lower) than the return of large-cap portfolio. In Dimson and Marsh (2001), the micro-cap, low-cap, and highcap indices cover 1%, 9%, and 90% of the total equity capitalization, respectively. They come to the similar conclusion that although micro-cap premium and low-cap premium from 1955 to 1989 exist, they go into reverse from 1989 to 1999. To examine naive extrapolation hypothesis in the behavioural finance, Levis and Liodakis (2001) find the superior performance of value strategies, based on book-toprice, earnings-to-price, cash-flow-to-price, and three years' past EPS growth, from 1968 to 1997. Ahmed, Lockwood, and Nanda (2002) present that average annual value/growth return spread based on earnings-to-price ratio is 4.03 percent, and small-/large-cap return spread is -0.57 percent from 1981 to 1997. Antoniou,

Galariotis, and Spyrou (2003) indicate that short-term contrarian strategies are profitable and more pronounced for extreme market capitalization stocks. Dimson, Nagel, and Quigley (2003) document the value (measured by BE/ME and dividend yield) premium among small-capitalization and large-capitalization stocks from 1955 to 2001. It is found in Siganos (2004) that momentum profits are significant in London Stock Exchange. Antonious, Lam, and Paudyal (2007) present momentum effect from 1977 to 2002. Fletcher (2007) shows significantly positive excess returns on HML (high B/M minus low B/M) and WML (winner minus loser) factors from 1979 to 2005. Fletcher (2011) finds size effect and value effect based on DY from 1957 to 2009.

## 2.1.3 Theoretical arguments for return anomalies (correction 7, 8)

There has been discussion about the underlying reasons behind return anomalies. Academics offer either risk-based or behavioural-based explanations for anomalies. The argument in some studies, that return anomalies are reward for risk, is based on the present value model. As Goyal (2012) presents, the common denominator in market capitalization and book-to-market ratio is the use of market price, which is associated with expected future cash flows and discount rates, thus reveals the difference in expected returns. Campbell and Vuolteenaho (2004) argue that risk factor can be decomposed into two components: cash flow and discount rate. They find that risk component from cash flows is priced in the cross-section of size and book-to-market sorted portfolio returns, and that return premium on value shares and small-cap shares is due to their higher cash–flow betas than growth shares and largecap shares. Bansal, Dittmar, and Lundblad (2005) find that cash-flow risk can capture more than 60 percent of cross-sectional variation in momentum, size, and value sorted portfolio returns. Santos and Veronesi (2005) find that the difference in cash-flow risk between value shares and growth shares can explain value premium. In another rational perspective derived from cash-flow risk, return anomalies are priced in macroeconomic conditions. Chordia and Shivakumar (2002) propose that momentum profit can be explained by business cycles. Petkova (2006) shows that a factor model, including term structure and default spread, captures returns of 25 size and book/market sorted portfolios in Fama and French (1993). Liu and Zhang (2008) find that a factor measured by growth rate of industrial production captures more than half momentum premium.

An alternative argument is based on the role of behavioural bias in explaining return anomalies. In contrast to risk-based argument, behavioural theories attribute return anomalies to mispricing due to behavioural bias. DeBondt and Thaler (1985) argue that reversal premium comes from mispricing when contrarian investors make profits by buying value shares that na we investors overreact due to misplaced pessimism and by shorting growth shares that are in-favour. Return reversal is a reflection of the cognitive bias of investors' overreaction. Jegadeesh and Titman (1993) claims that positive autocorrelation in share returns is caused by investors' delayed reaction to firm-specific information. Lakonishok, Shleifer, Vishny (1994) argue that investors' overreaction can explain value effect. When shares with high B/M ratios generally fall in tough time, investors irrationally extrapolate their worse performance, which leads to value shares undervalued and value premium. Daniel, Hirshleifer, and Subrahmanyam (1998) propose a theory with two psychological biased traits: investor overconfidence in the precision of their private information and selfattribution bias of investment outcomes. This implies that investors overreact to private information and underreact to public information. They show that short-term momentum can be a result of continuing overreaction, and followed by long-term reversal. Barberis, Shleifer, and Vishny (1998) use other two psychological traits, such as conservatism and representative heuristic, to offer an alternative explanation for behavioural characteristics in financial markets. Investors change their belief too slowly and underreact to news, which leads to momentum effect. On the other hand, they overweight recent experience or data and overreact to consistent good or bad news, which leads to reversal effect.

The debate between risk-based and behavioural-based argument for return anomalies comes from the test of value effect in the recession periods. Lakonishok, Shleifer, and Vishny (1994) present that if value shares are fundamentally riskier, they must underperform growth shares in some 'bad' sates of world where the marginal utility of wealth is high, making value shares unattractive to risk-averse investors. However, they find the opposite evidence that value shares outperform growth shares in economic recessions and downward markets. Using stochastic dominance approach, Abhyankar, Ho, and Zhao (2008) find the same evidence that contradicts the risk-based explanations for value premium in recession periods. They argue that value premium may reflect the missing behavioural components from value investors. These findings cast doubt on risk-based explanations that value effect simply results from risk factors in asset pricing models.

## 2.2 Time-series return predictability

Different from the research on cross-sectional return predictability, a number of academic studies investigate the relation between macroeconomic variables and aggregate market return. With the present value model, stock prices can be estimated as the sum of present discounted expected future dividend flows. Clearly, macroeconomic forces influence firms' cash flows, dividend payouts, and discount rates, which further affect the movement in stock prices. Using the Arbitrage Pricing Theory (APT) model in Ross (1976), Chen, Roll, and Ross (1986) measure risk factors associated to various macroeconomic indicators to capture stock market returns, and offer the foundation for the belief on the relation between share returns and macroeconomic forces. Some papers examine the impact of business cycle on stock returns in the US Stock Market. Keim and Stambaugh (1986), Campbell and Shiller (1988), Fama and French (1988a, 1988b, 1989), and Hodrick (1992) document that expected stock returns are driven by four risk factors, such as the dividend-price ratio, the short-term interest rate, the term spread, and the credit spread. In addition, other studies centre on the impact of market conditions on expected aggregate market performance. For example, Keim and Stambaugh (1986), Campbell, Grossman, and Wang (1993) explore the correlation between lagged share prices, trading volume and their subsequent movements.

## 2.2.1 Time-series return predictability in the US Stock Market

#### Forecasting economic variables

It is well known that expected excess returns on common stocks vary countercyclically in the US Stock Market, and that risk premia are higher in recessions but lower in expansions. Fama and French (1989), Ferson and Harvey (1991) plot fitted values of the expected risk premium on the aggregate stock market and find that it increases during economic contractions and peaks near business cycle troughs. Harrison and Zhang (1999), Campbell and Diebold (2009) come to the same conclusion on the countercyclical pattern of expected share returns. Cochrane (2011) makes a summary that expected returns are low and prices are high in 'good times' when consumption, output, and investment are strong, unemployment is low, and interest rates are high, and vice versa.

A more direct way of testing the time-variation in share returns is to explicitly forecast excess returns with predetermined conditioning variables. The empirical asset pricing studies have focused on the relation between series share returns and macroeconomic information, such as money supply, inflation, the levels of shortterm and long-term interest rates measured by short-term interest rates and term spreads, default spreads, industrial production, and proxies for the consumptionwealth ratio (CAY). The relevant literature on predictive economic variables for expected return is summarized as below: <u>Money supply:</u> Brunner (1961), Friedman and Schwartz (1963) establish the relation between the change in money supply and the change in asset prices. It is generally concluded that the increase or decrease in the growth of money supply leads to the adjustment for the proportion of money and other assets in investors' portfolios, and further the change in asset prices. Cooper (1974) shows significant relation between Standard and Poor 500 Index and money supply at lagged eight months from 1947 to 1970. However, Rozeff (1974) finds that share returns are unrelated to past change in money supply, and do not include information on money supply.

Inflation and Short interest rate: An increase in expected inflation rate is likely to lead to tightening economic policies, and further negatively influence firm fundamentals. Meanwhile, the rising inflation pushes up the nominal risk-free rate and the discount rate in the present value model. Fama and Schwert (1977) argue that unlike government bonds and real estate, stocks were not hedges against inflation. Fama (1981) notes that one important determinant of stock returns stems from real activity. When nominal quantity of money in the market does not move sufficiently with real activity, a negative relation exists between inflation and real activity. The same conclusion on the relation between inflation and share returns is drawn by Chen, Roll, and Ross (1986). In the present value model, real interest rate that investors use to discount future dividends is negatively associated with stock prices. Chen (1989) presents negative relation between expected share returns and TBILL rate. Chen (1991) further finds that the forecasting power of TBILL is limited to the next quarter. Hodrick (1992), Lettau and Ludvigson (2001) conclude that short-term interest rates, often measured as a 'relative TBILL rate' (e.g., the 30-day Treasurybill rate minus its 12-month moving average) predicts share returns.

Term spread and default spread: Kessel (1965) presents that yields on long-term Treasury bonds rise less during business expansions and fall less during business contractions than yields on short-term bills. Thus the yield spread of long-term Treasury bonds over short-term Treasury bills has a clear countercyclical pattern. Keim and Stambaugh (1986) find weaker forecasting power of term spread, defined as the difference between yields on long-term under-BAA-rated corporate bonds and short-term one month U.S. Treasury bill, on monthly risk premiums. The t-statistics on term structure are typically less than conventional significance level in the first subperiod from 1928 to 1952, and the estimates themselves are sometimes negative in the second subperiod from 1953 to 1978. Fama and French (1988a) study the forecasting power for share returns with the 10-year Treasury bond yield minus the one-year Treasury bond yield (a measure of the term spread) and the BAA corporate bond yield minus the AAA corporate bond yield (a measure of the default spread). Fama and French (1989) extend the test of Keim and Stambaugh (1986) on monthly returns to longer horizon returns, and present that predictable variation in common stock returns is tracked by the default spread and term spread in bond returns. The default spread as well as dividend yields seem to be related to long-term business episodes that span several business cycles. They forecast high returns when business conditions are weak and low returns when business conditions are strong. The term spread is more closely related to the short-term business cycles, performing with low around measured business cycle peaks and high near troughs. It is shown in Chen (1991) that the default spread has forecasting power for real and excess market returns and its power starts to fade beyond the first year, and term structure only offers forecasting power less than 3 quarters.

<u>Industrial production</u>: Chen, Roll, and Ross (1986) show that from 1958 to 1984 monthly growth rate in industrial production with positive risk coefficient, combining with unanticipated change in the default premium and term structure, is significant in explaining expected stock returns. Hodrick and Zhang (2001) present that industrial production has significant predictive power for expected monthly return, but explains 1% of return variation from 1952 to 1997. Humpe and Macmillan (2007) find significantly positive and negative influence of industrial production and inflation, respectively, on US stock prices from 1965 to 2005.

<u>Consumption-wealth ratio (CAY)</u>: Lettau and Ludvigson (2001) use a log-linear approximation of a representative investor's consumption-wealth ratio (CAY) as one variable and show that the deviation from a cointegrating relation for log consumption  $c_t$ , log asset wealth  $a_t$ , and log labor income  $y_t$ , is a potential predictor of real and excess stock market returns. Hodrick and Zhang (2001) show that CAY has significant predictive power for expected quarterly return, and explains 11% of return variation from 1952 to 1997. It is found in Lettau and Ludvigson (2010) that over one-quarter horizon, the 'CAY' and the relative-bill rate have statistically significant predictive power for excess returns from 1952 to 2000 when dividend-price ratio, term spread, default spread, and one lag of the market return have little forecasting power for quarterly excess returns. This single variable is capable of

predicting 34% of the variability in three-year's excess return. When sample data after 1995 are included, dividend-price ratio has no ability to forecast excess stock returns at horizons ranging from 1 to 24 quarters. The second half of the 1990s saw an extraordinary surge in stock prices relative to dividends, weakening the tight link between the dividend-yield and future returns that has been found in early sample period. However, the forecasting power of the 'CAY' seems to have been less affected by this episode. The consumption-wealth ratio (CAY) should have more stable forecasting power for common share returns in changing macroeconomic risk than other macroeconomic variables.

#### • Forecasting market variables

Other studies investigate the impact of market indicators on expected share returns. The empirical asset pricing literature has focused on the relation between series variation in share returns and stock market valuation ratios, such as price-dividend and price-earnings ratios, dividend-payment ratio, and book-market ratio. Fama and French (1988a), Campbell and Shiller (1988), and Hodrick (1992) find that the ratio of price to dividend or earnings has predictive power for excess returns. Fama and French (1988a) show that the explanatory power of dividend yield on returns strengthens as the horizon extends longer. Dividend yield typically explains less than 5% of monthly or quarterly return variations, but often explains more than 25 % of two- to four-year's return variations from 1927 to 1986. Similarly, using Standard and Poor Composite Index, Campbell and Shiller (1988) present that 3.9% of one-year's real return variations is explained by the log dividend-price ratio from 1871 to

1987. As return horizon extends to 3 and 10 years, the fraction of the variation explained by the log dividend-price ratio increases to 11.0% and 26.6%. Fama and French (1989) conclude that although D/P and E/P capture similar components of expected returns, t-stats and regression R-square suggest that D/P makes better forecasts of stock returns. Chen (1991) finds that the dividend yield D/P forecasts real and excess market returns over the next 2 years and its forecasting power diminishes toward the end of the second year. Cochrane (2008, 2011) draw the same conclusion on explanatory power of dividend yield for share returns over horizons.

Price and volume are simultaneously determined in equilibrium. The investors' trading process generates volatility in both price and volume. The real investing market might contain some information which is not included in macroeconomic variables. Chen, Roll, and Ross (1986) find that stock market index, although explaining a significant portion in time-series volatility of share returns, has an unsignificant influence on expected returns when it is combined with macroeconomic variables such as industrial production growth, the default premium, and the term structure in the model. But, Keim and Stambaugh (1986) find that when the seasonality is taken into account, the level of stock prices contains ex post information about the change in expected risk premia. Campbell and Hamao (1992) use lagged excess market return as well as macroeconomic variables, such as dividend-price ratio, relative short rate, and long-short yield spread, to investigate predictable stock returns in US and Japan Stock Market.

Return momentum and reversal show the existence of series autocorrelation in share prices, and imply the impact of lagged share return performance in different horizons on their subsequent returns. Daniel, Hirshleifer, and Subrahmanyam (1998), Hong and Stein (1999) apply investors' behavioural or cognitive biases to explain short-term price momentum in Jegadeesh and Titman (1993) and long-term price reversal in DeBondt and Thaler (1985, 1987). Daniel, Hirshleifer, and Subrahmanyam (1998) show that investors' unrational emotional movement leads to price momentum and reversal. Hong and Stein (1999) conclude that investors' initial underreaction and subsequent overreaction to information contribute to price momentum and reversal. Investors become overconfidence and overoptimistic in the rising market, which further drive common stock prices in the rising direction, vice versa.

In addition, at least two theoretical papers suggest that past trading volume may provide valuable information about the movement in share prices. Campbell, Grossman, and Wang (1993) present a model in which trading volume proxies for the aggregate demand of liquidity traders, and that price changes accompanied by high volume will tend to be reversed. Blume, David, and Maureen (1994) establish a model in which traders can learn valuable information about a security by observing both past price and past volume information, and conclude that volumes offer information that can not be deduced from the price performance. Empirical research supports the relation between trading volume and expected returns. Datar, Naik and Radcliffe (1998) show that shares with low (high) volume earn higher (lower) future returns. Lee and Swaminathan (2000) present that the interaction between past returns and trading volume can predict both the magnitude and the persistence of future price momentum.

## 2.2.2 Time-series return predictability in the international and UK Stock Market

Time series return predictability in other stock markets has been studied since the later 1980s. Campbell and Hamao (1992) find positive and negative impact of dividend-price ratio and the relative short interest rate (the difference between the current short interest rate and its one year backward moving average) on excess stock return in the Japan Stock Market from 1971 to 1990. Mukherjee and Naka (1995) use vector error correction model (VECM) to find a cointegrating relation between Tokyo Stock Exchange Index and six Japanese macroeconomic variables, such as exchange rate, money supply, inflation, industrial production, long-term government bond rate, and call money rate from 1971 to 1990. Nasseh and Strauss (2000) show significant long-run correlation between stock prices and domestic macroeconomic activity in six Europe countries, such as France, Italy, Netherlands, Switzerland, Germany, and UK. Domestic macroeconomic variables, including short-term interest rates, long-term interest rates, consumer prices, manufacturing orders, and real industrial production, can forecast 37% to 82% of the variation in four-year's share returns in these markets. Maysami and Koh (2000) find significant relation between the change in interest rate, exchange rate, and stock market level in the Singapore Stock Market. Bordo and Wheelock (2006) point out that most booms in ten developed stock markets were procyclical in the 20th century. The booms took place during the period of above-average GDP growth and below-average or falling inflation, and ended in the period of rising inflation and tightening monetary policy. Humpe and Macmillan (2007) present that Japan stock prices are positively associated to industrial production but negatively to money supply from 1965 to 2005. The unexpected money supply forecasts may be explained by Keynesian liquidity trap during the late 1990s and early 2000s, the period when increasing money supply and falling interest rates failed to boost Japanese economy and prevent its stock market from the decline.

Beenstock and Chan (1988) find that four risk factors, such as treasury bill rate, money supply, fuel and material costs, and inflation, significantly explain about 33% of the variation in expected share returns in the UK Stock Market from 1977 to 1983. Clare and Thomas (1994) show corporate default and retail price index are priced in the UK Stock Market from 1983 to 1990 when beta-sorted portfolios and size-sorted portfolios are tested by the model of Chen, Roll and Ross (1986). They find that market return contains information, which is not included in two macroeconomic forces, and captures significant risk premium in the size-sorted portfolios. In the test of CAPM and APT, Soufian (2001) finds that the explanatory power of market portfolio becomes unsignificant when some economic variables are regarded as risk factors, and that the change in yearly industrial production and unexpected inflation have significant impact on expected return in the overall period from 1980 to 1997 and two subperiods. Fletcher (2002) shows that the performance of the CAPM improves sharply when a proxy for labour income growth is added to the stock market index from 1975 to 2000. Fletcher and Hillier (2002) present that the combination of market index and other factors, such as term structure, monthly

percentage change in industrial production and inflation, and the difference between risk-free return and inflation, can offer good predictability on UK stock returns. Gunsel and Cukur (2007) find that seven economic variables, such as term structure, unanticipated inflation, unanticipated industrial production, risk premium, exchange rate, money supply, unanticipated dividend yield, explain 28% to 79% of the variation in portfolio returns of 10 UK industries from 1980 to 2003.

## 2.3 The time-variation in return anomalies

Some studies argue that time-varying betas exist in the explanation of size effect and value effect. Jagannathan and Wang (1996), Lettau and Ludvigson (2001), Santos and Veronesi (2005) show that the betas of small-cap stocks and value stocks vary over business cycle. Brandt, Clara, and Valkanov (2009) note that the relation between firm characteristics (such as market capitalization, book-to-market ratio, lagged twelve-month return) and the joint distribution of returns is time-variant in estimating the coefficients of portfolio policy.

#### 2.3.1 The time variation in return anomalies in the US Stock Market

By extending longer sample period and investigating pattern in subperiods, some empirical papers show that return anomalies are volatile, rather than stable, over time in the US Stock Market. Banz (1981) shows that size effect was not very stable through time, stronger from 1936 to 1945 and weaker from 1946 to 1955. Fraser (1995) further shows that prior to mid-1989 smaller companies consistently outperformed market portfolio but since then to October 1991 small-cap premium disappeared. The volatility also exists in value effect. It is found in Asness, Friedman, Krail, and Liew (2000) that value stocks sustain periods of poor performance although they on average beat growth stocks. Since 1982 there have been three major bear markets for value stocks: 1989-1990, 1995-1996, and 1998-1999. Ahmed, Lockwood, and Nanda (2002), Lucas, Dijk, and Kloek (2002) present that the impact of firm-specific characteristics like size and book-to-price on stock returns varies

considerably over time. Teo and Woo (2004) find strong evidence for reversal and momentum at the style level in terms of large value and small growth. Cooper, Gulen and Rau (2005) point out that the variation in style effects over time is one important reason for style-originated investors to allocate risky assets into styles based on their past performance.

Since small-cap premium, value premium, and momentum premium were found in the later 1970s to the early 1990s, some studies have begun to link dynamic performance of return anomalies to different macroeconomic backgrounds and market conditions. Fama and French (1993) present that distressed firms, based on size and book-to-market ratio, may be more sensitive to certain business cycle factors. It is found in Jensen, Johnson, and Mercer (1998) that small-cap premium and value premium are stronger in expansionary monetary periods, but weak or negative in tightening ones. Coggin (1998) finds that it is possible to predict style index returns by conditioning such predictions on some outside macroeconomic information besides monthly returns. Perez-Quiros and Timmermann (2000) test the impact of macroeconomic variables, such as short interest rate, default spread, and money growth, on decile-sized portfolios, and conclude that excess return on the smallest portfolio is most strongly related with these variables. Lucas, Dijk, and Kloek (2002) show that investment styles have different performance in economic recession and expansion, and that over recessions investors load less on momentum stocks and more on small-cap stocks. Chordia and Shivakumar (2002) document that momentum payoffs are negative during recessions and positive during expansions, and that macroeconomic variables such as lagged dividend yield, default preimium,

term structure, and three-month T-bill yield can explain momentum profit. However, with the decomposed buy-and-hold method, Liu and Strong (2008) replicate the research by Chordia and Shivakumar (2002), and find the difference in momentum payoffs between expansionary and contractionary periods is insiginificant (0.774%, t=1.05). Cohen, Polk and Vuolteenaho (2003) present that the expected value-minus-growth return is high when the value spread is wide. Vassalou (2003) comes to the conclusion that the mimicking portfolio related to future GDP growth is an important factor for explaining the cross-sectional returns of book-to-market and size portfolio. Asness, Moskowitz, and Pedersen (2009) report that value premium and momentum premium are both positively related to long-run consumption growth and both negatively related to recessions. Liu and Zhang (2008) document that the value spread (the log book-to-market of value stocks minus the log book-to-market of growth stocks) negatively correlates with default premium (-0.41) and short-term T-bill rate (-0.51).

Cooper, Gutierrez, and Hameed (2004) show that momentum profits depend on the lagged state of the market in the US Stock Market from 1929 to 1995, the significant positive (negative) profit following positive (negative) market return in past one to three years. Avramov and Chordia (2006), Avramov, Chordia, Jostova, and Philipov (2007) come to the same conclusion. L'Her, Mouakhar, and Roberge (2007) compare the predictive power of three approaches, which are recursive partitioning (RP), neural networks (NN), and genetic algorithm (GA), to correctly time the size premium, and find that with RP approach the most important variable (the first split criterion) in timing small-cap premium is the US Conference Board Coincident

Economic Indicators Index (COIN), and the second important factors (the next split criterion) are Dividend Yield of the S&P500 (DIV) and Six-Month momentum of the S&P500 (MOM). COIN and MOM are also the factors in NN approach that play the most important role in timing small-cap premium. L'Her, Mouakhar, and Roberge (2007) present that some information on market states, such as six-month index performance and one-month volume change in NYSE, can offer prediction on shifting time between large-cap stocks and small-cap ones.

# 2.3.2 The time variation in return anomalies in the international and UK Stock Market

Bird and Whitaker (2004) examine value and momentum effects in major European markets from 1990 to 2002, and find their large upward movement followed by a significant correction. They find that momentum effect occurred during the 1990s' running up market, and value effect was confined in the correction period. This shows pro-cyclical momentum effect and counter-cyclical value effect. Using 59 MSCI industry indices for the UK like the US and Japan, Babameto and Harris (2008) note that term spread significantly predicts the variation in momentum portfolio returns with a negative coefficient, the aggregate book-to-market ratio offers statistically significant explanatory power for the performance in value strategy with a positive coefficient. Hou, Karolyi, and Kho (2011) find that book-to-market effect in shares across 27 emerging countries is weaker in the period from 1992 to 2003 than the period from 1981 to 1992, and the opposite pattern for effects in terms of cash flow-to-price (C/P), dividend yield (D/P), and earnings yield (E/P).
The time-variation in return anomalies also exists in the UK Stock Market. As shown in Levis and Liodakis (1999), no obvious small-cap premium in the whole sample period from 1968 to 1997 results from quite different return performance of small-cap shares over periods. As opposite performance in the previous period, the large-cap portfolio yielded an average equal- (value-) weighted return of 13.28% (14.35%), 4.30 (6.30) percentage points higher than the return of the small-cap portfolio in the period from 1988 to 1997. Similarly, Dimson and Marsh (2001) show that although micro-cap equities had highest return in the UK from 1955 to 1999, the percentage of years in which large-cap equities outperformed micro-cap equities and small-cap equities is 22 percent, and small-cap premium disappeared from the 1980s to the 1990s. It is found in Dimson, Nagel, and Quigley (2003) that annual premium returns on the SMB (small minus big) and the HML (high BE/ME minus low BE/ME) zero-investment portfolio in the UK Stock Market have been volatile from 1955 to 2001.

The similar research investigates the relation between macroeconomic variables and time-varying return anomalies in the UK Stock Market. Fraser (1995) points out that the disappearance of small-cap premium in the London Stock Market in the early 1990s might be caused by UK economy's deep recession in this period. Levis and Liodakis (1999) show that small-cap stocks benefit from lower inflation rate and high dividend yield ratio, and that value stocks benefit from lower inflation rate and a fall in the monthly £/\$ exchange rate from 1968 to 1997. One-month lagged value/growth return spread and monthly change in consumer price index are two significant variables on annual value/growth spread in the UK from 1968 to 1999.

Dimson and Marsh (2001) find that dividend yield and dividend growth explain the time-variation in size effect. Small-cap shares' higher dividend yield and dividend growth in 1955 to 1988 drive their return premium. In the following decade smallcap shares' declining dividend yield and dividend growth lead to the disappearance of small-cap effect. Fletcher (2001) examines the impact of lagged market performance and macroeconomic variables on decile size portfolio returns, and finds that lagged one month excess return of the Financial Times All Share Index and lagged one month risk-free return have statistically positive and negative significance impact on smaller-cap portfolio return more than large-cap portfolio return from 1955 to 1995. Although lagged dividend yield on the Financial Times All Share Index is generally unsignificant in 9 out of 10 portfolios, it has increasingly positive impact on the return from the first (the smallest market capitalization) portfolio to the tenth (the largest market capitalization) portfolio. The lagged FTA excess return has significantly positive impact on smaller-cap portfolio return more than large-cap portfolio return. Levis and Tessaromatis (2004) investigate the role of macroeconomic variables in growth-value spread based on FTSE100 and FTSE250 indices from 1987 to 2001, and document negative relation between default premium, term structure and growth-value return spread, and positive relation between dividend yield and the return spread. Dimson, Nagel, Quigley (2003) find that correlation in annual return between HML (high minus low BE/ME) portfolios and IMC (high minus low dividend yield) portfolios from 1956 to 2001 is 0.82. The reason for this high correlation is that although dividend yield in the UK Stock Market has declined since the mid-1980s, about 75 percent of listed shares still paid dividends in 2001. The market value of shares with dividend payment made up 95

percent of total market capitalization. Dividend yield explains the time-variation in cross-sectional return of HML. Zhang, Hopkins, Satchell, and Schwob (2009) offer the existence of positive relationship between SMB/HML and unexpected higher GDP, short term interest rates, and term spread, and negative (positive) relationship between SMB (HML) and unexpected inflation.

# 2.4 Concluding remarks

There was the voluminous research on the pattern of share prices in past one century. The early work based on the Random Walk Hypothesis (RWH), the Efficient Market Hypothesis (EMH), and the Capital Asset Pricing Model (CAPM) before the 1980s argues that the movement in share prices can not be predicted, and that it is impossible to outperform the aggregate market by picking stocks in cross-sectional dimension and time-taking in time-series dimension.

The theories of RWH, EMH, and CAPM have begun to face challenge from large number of empirical findings and the tests of CAPM since the later 1970s. Many studies have found the existence of return anomalies, such as momentum effect, size effect, and value effect in the US Stock Market and other markets. This means that cross-sectional share returns are related to firms' characteristics, and that investors could earn return premium by allocating positions on shares with specific characteristics.

In another dimension of share prices, empirical work in the 1960s and 1970s found the relation between money supply, inflation and aggregate market return. With the introduction of the Arbitrage Pricing Theory (APT) model in 1976, more studies have focused on the impact of macroeconomic forces on series share returns. This offers the possibility in time-taking by using macroeconomic variables associated with business cycles. Since the mid-1990s, empirical work has found the unstable pattern of return anomalies as the sample period extends longer. Small-cap premium disappeared after the 1980s, and value stocks underperformed growth stocks in the later 1990s. In this context, it is accepted that small-cap premium and value premium is not invariant all the time. Since the 2000, some studies have begun to investigate time-varying style effects, and explored what factors lead to the variation. This drives the research on the combination of cross-sectional and series properties of expected returns.

# **Chapter 3 Data and Portfolio Formation**

The research consists of four aspects of empirical work, that is to investigate the time-variation in relation between share returns and firm characteristics in the UK Stock Market, the impact of lagged macroeconomic variables on the variation, the performance of recursive in-sample regression model and out-of-sample forecasting model, and the application of active trading strategy on style effects. Followed by next four chapters on empirical analysis, this chapter introduces data description and portfolio formation used in the empirical work. The relevant data on macroeconomic variables is described in Chapter 5. The remainder of the chapter proceeds as followings. Section 3.1 is concerned on data, which describes data source, firm characteristics, and holding portfolio returns. Section 3.2 introduces portfolio formation for momentum effect, size effect, and value effect.

#### **3.1Data**

#### 3.1.1 Data source

The research investigates time-varying style effects in the UK Stock Market. As other studies, the research classifies style effects into momentum effect, size effect, and value effect, which are relation between share returns and firm characteristics, such as lagged share return, market value, and price-to-book ratio. In practical application, some database, such as MSCI, FTSE, Russell, and Datastream, have published value/growth index and small/large index since the 1970s. A series of style-originated mutual funds in terms of small-/mid-/large-cap, value/growth have been launched. As a long-known style, momentum index has been edited and published by AQR Capital Management since 2009. Although a large number of funds based on value/growth and small/large-cap effects have been established for a long period, the first momentum fund was launched by AQR Capital Management in 2009.

The research uses the London Share Price Database (LSPD) to collect shares' monthly returns, share prices, share capitals, monthly market values, and delisting and suspended information. LSPD, which is maintained by the Institute of Finance and Accounting at London Business School, covers 9412 listed stocks in the UK Stock Market from 1955 to 2008. The return is adjusted for the change in capital structure and dividends. The research uses Datastream to collect shares' price-to-book values. Datastream is compiled by Thomson Financial Limited, and covers UK companies since 1965.

With LSPD, the research collects 9412 shares' monthly returns since 1955 to form decile momentum portfolios, and monthly market values of all shares after January 1979 onwards to form decile size portfolios. With Datastream, the research collects companies' price-to-book ratios to form decile value portfolios. LSPD and Datastream have different identification approaches for companies quoted on London Stock Exchange. LSPD allocates a unique identification number in a sequence from

1 for them since 1955. Datastream allocates a unique identification code with 6 characters for them since 1965. The research matches all companies' names in LSPD with ones in Datastream by hand. (10) After January 1980 onwards, 92.8 percent of company names in LSPD could be matched with ones in Datastream. Few companies in LSPD has corresponding price-to-book ratio in Datastream prior to January 1980. Therefore, in order to include shares as much as possible in the empirical analysis, the research investigates value effect at the beginning of the 1980.

The research splits the whole sample period from January 1955 to December 2008 into three subperiods, and establishes decile momentum portfolios according to relevant criteria in three subperiods. Dimson, Nagel, and Quigley (2003) note that in the first subperiod before 1975 the LSPD does not have full coverage of all companies, so the research uses random one-in-three of companies to form the portfolios prior to 1975. In the second subperiod from 1975 to 1979 and the third subperiod from 1980 to 2008, portfolio formation is based on the complete sample. In three subperiods, investment trusts are excluded to avoid double counting and ensure a purer measure on return performance in UK equities because investment trusts invest in other equities and other assets. In the third subperiod, three kinds of shares, which are secondary shares, odd foreign mining and banking shares, Irish, Scottish, and odd companies, are excluded by using data in LSPD archive file.

#### 3.1.2 Firm characteristics

The research uses firm characteristics in terms of lagged share return, market value, and price-to-book ratio to construct decile style portfolios, and examines the relation between firm characteristics and share returns in holding horizons.

lagged share return in momentum portfolios

The research examines momentum effect, which is measured by share returns in past 6/12 months and their subsequent holding returns. Lehmann (1990) and Jegadeesh (1990) examine the performance of trading strategies based on one week and one month return, respectively, and find that these short horizon strategies yield contrarian profits over the next one week and one month. The research uses two approaches, namely unskipping one month and skipping one month between the formation period and the holding period, to construct momentum portfolios, and to observe the impact of monthly price reversal on momentum effect as robustness test. In the case of unskipping one month, the research uses equation (3.1) and (3.2) to calculate all shares' compounded return  $r_{.1 \sim .6}$  in the month t-12 to form decile momentum portfolios. In the case of skipping one month, the research uses equation (3.1) and (3.2) to calculate all shares' compounded return  $r_{.2 \sim .12}$  in the month t-2 to the month t-2 to the month t-6, and compounded return  $r_{.2 \sim .12}$  in the month t-2 to the month t-12 to form decile momentum portfolios.

$$r_{i,-1\sim-6} = \prod_{t=-1}^{-6} (1+r_{i,t}) - 1$$
(3.1)

$$r_{i,-2\sim-6} = \prod_{t=-2}^{-6} (1+r_{i,t}) - 1$$
(3.1)

$$r_{i,-1\sim-12} = \prod_{t=-1}^{-12} (1+r_{i,t}) - 1$$
(3.2)

$$r_{i,-2 \sim -12} = \prod_{t=-2}^{-12} (1+r_{i,t}) - 1$$
(3.2)

The research conducts empirical work on momentum effect with 6/12 months' formation return and five holding horizons. It compares the performance of average monthly Winner/Loser return  $R_{wml}$  between skipping and unskipping one month in the formation period. Table 3.1 and Table 3.2 show that over five holding horizons, Winner/Loser return  $R_{wml}$  based on the approach of skipping one month in portfolio formation is higher than one without skipping one month, and that the percentage of observations with positive Winner/Loser return  $R_{wml}$  in the skipping case is higher than one in the unskipping case. The difference in Winner/Loser return  $R_{wml}$  and the percentage of  $R_{wml}>0$  between the skipping case and the unskipping case becomes small as holding length extends from 3 months to 24 months. The impact by skipping

one month in momentum formation declines when the holding length becomes longer. In addition, in Chapter 4 the research finds high correlation in empirical result between the skipping approach and the unskipping approach, with more than 0.92 for most 'Winner' return and 'Loser' return, more than 0.82 for most Winner/Loser Return  $R_{wml}$ , and more than 0.87 for most momentum coefficient  $\gamma_{mom}$ , as shown in Table 4.8. Therefore, in order to obtain conservative estimation on momentum effect, the research does not skipping one month between the formation period and the holding period to construct momentum portfolios in the empirical analysis in the following chapters.

#### [Table 3.1, 3.2]

• Market value in size effect and value-weighted portfolios

The research uses the archive of monthly share market capitalization in LSPD database to form decile size portfolios after January 1979 onwards, and to examine monthly-by-monthly size effect. In addition, shares' market values are used to construct value-weighted momentum portfolios, size portfolios, and value portfolios from January 1979 to December 2008.

• Price-to-book ratio in value effect

The research collects all shares' price-to-book ratios from Datastream to investigate value effect after January 1980 onwards. Because of different share codes for

Datastream and LSPD, the research matches all shares' names in two databases. After the year of 1980, 92.8 percent of shares in LSPD could be matched in Datastream. The research excludes shares with negative price-to-book ratio in empirical work.

#### 3.1.3 Holding portfolio returns

The research makes two corrections to shares' missing monthly returns in LSPD. Firstly, Shumway (1997) finds that most of missing delisting returns in CRSP database correspond to surprise delists due to bankruptcy and other negative reasons. They are often worthless when they delist, and yield average delisting return of -30 percent in actual US Share Market. He documents that introducing -1 to missing delisting returns leads to the large decline in the return of small-cap portfolio in declie size portfolios and loser portfolio in decile momentum portfolios. This implies that empirical results on size effect and momentum effect might be biased if missing delisting returns are omitted. Following the approach in Dimson, Nagel, and Quigley (2003), the research uses an upper bound for the delisting bias to assign -1 for missing delisting returns in LSPD at the death event date where the LSPD code indicates valueless death. Secondly, except shares with missing return due to valueless death, some shares have missing returns with death or a temporary suspension. Following the approach of Liu and Strong (2008), the research assigns a zero value for these shares returns when they have missing return observation. With two reasons, the research investigates the relation between firm characteristics and share returns in five holding horizons, such as 3, 6, 12, 18, and 24 months. Firstly, style-originated investors in the market always have a preference for different holding horizons, including short-/medium-/long-term, because of the variation in their trading psychology and behaviour, risk exposure, and asset patterns. It is rational to explore the variation in style effects with different holding horizons. Secondly, the forecasting power of lagged informative variables for share returns with different holding horizons is not invariant. For example, some macroeconomic variables, such as the dividend yield and the term structure, have stronger explanatory power in longer holding period than in shorter one because they are largely measures of long-term business cycles. Fama and French (1989) find that the R-square for regression of expected stock returns on the dividend yield and the default spread tends to increase with longer holding period. It is typically less than 0.1 for monthly and quarterly returns, but is often greater than 0.3 for one- to fouryear's return. Cochrane (2008, 2011) further supports the same conclusion on the dividend yield. With the equation (3.3), the research calculates shares' compounded buying-and-holding returns in five holding horizons.

$$r_{i,1\sim3(6,12,18,24)} = \prod_{t=1}^{3(6,12,18,24)} (1+r_{i,t}) - 1$$
(3.3)

The research calculates equal-weighted portfolio holding returns and value-weighted portfolio holding returns, respectively. Equal-weighted portfolio returns are weighted toward small shares, and value-weighted ones are affected more by large shares. Two kinds of holding returns offer a better way to examine the performance of share returns as a function of share size, especially in momentum effect and value effect. With monthly market value since 1979 from LSPD archive, the research calculates all shares' value weighting to examine value-weighted momentum, size, and value effect January 1979 afterwards.

The investigation on value-weighted momentum effect from 1955 to 1978 is limited by the collection of monthly share market value. LSPD database offers the archive of annual market value of all shares. Market values of many shares in this archive are set to be 0 or 1 because of rounding off and the expression with million. This leads to the same market values for some shares, and makes impossible to form valueweighted decile momentum portfolios by number of shares. For example, at the end of 1955, among 1073 shares, 564 shares have market value of 0, and 252 shares have market value of 1. To deal with this issue, the research multiplies annual share capital by monthly share price in LSPD as monthly market value. This approach ignores the capital change of some shares during one year, so the result on valueweighted momentum effect prior to 1979 might be biased.

# 3.2 Portfolio formation

The research examines the time-variation in style effects based on firm characteristics in the UK Stock Market. The research uses shares' individual firm characteristics to establish decile portfolios, and then examines the variation in relation between firm characteristics and equal/value-weighted holding portfolio returns with five holding horizons of 3, 6, 12, 18, and 24 months.

### 3.2.1 Momentum portfolio formation

LSPD offer monthly share returns after January 1955 onwards, the research investigates monthly-on-monthly momentum effect with 6 and 12 months' formation lengths, respectively, in the UK Stock Market from 1955 to 2008. For momentum effect with 6 months' formation length, depending on magnitudes of their lagged 6 months' compounded returns in the formation period, all shares are ranked from the lowest to the highest each month, and then decile momentum portfolios with the same number of shares are formed. The first portfolio contains shares with lowest formation return, and is designated as 'Loser' portfolio. The tenth portfolio contains shares with highest formation return, and is designated as 'Winner' portfolio. Similar approach is used to form decile momentum portfolios based on lagged 12 months' compounded share returns.

In the overall sample period, there are 648 months from January 1955 to December 2008. After excluding the first month with no share returns, the research uses 6 months from February 1955 to July 1955 to form the first decile momentum portfolios in the sample period. It rolls this portfolio formation forward each month, and uses last 3, 6, 12, 18, and 24 months as five holding horizons for the last decile momentum portfolios at the end of the sample period. The research has 639, 636, 630, 624, and 618 overlapping sample observations for momentum effect with 6 months' formation length and 3, 6, 12, 18, and 24 months' holding horizon, respectively, until the end of 2008. When momentum portfolios are based on lagged 12 months' shares returns, the first decile momentum portfolios are formed in January 1956, and there are 633, 630, 624, 618, and 612 overlapping sample observations for momentum effect with 12 months' formation length and five holding horizons, respectively. Table 3.3 shows that there are an average 1662 to 1682 (1615 to 1638) shares to construct equal-(value-) weighted momentum portfolios based on lagged 6 months' share returns and five holding horizons, and an average 1572 to 1587 (1540 to 1558) shares to construct equal-(value-) weighted momentum portfolios based on lagged 12 months' share returns and five holding horizons.

[Table 3.3]

LSPD offers the archive of monthly market values of all shares after January 1979 onwards, so the research examines size effect monthly-on-monthly from 1979 to 2008. Each month all shares are ranked from the lowest to the highest according to their market values. The first and tenth portfolios include the smallest companies and the largest companies, and are designated as 'Small' portfolio and 'Large' portfolio, respectively. The first decile size portfolios are formed in the end of January 1979, and there are 357, 354, 348, 342, and 336 overlapping sample observations for size effect with five holding horizons of 3, 6, 12, 18, and 24 months until the end of 2008. Table 3.3 shows that there are an average 1785 to 1822 shares to construct equal-(value-) weighted size portfolios with five holding horizons.

#### Two approaches to form decile size portfolios

In academic and practical field, there are two approaches to establish decile size portfolios on number of shares and market capitalization, respectively. After ranking all shares' market values from the lowest to the highest, CRSP (Center for Research in Security Prices) defines decile portfolios with same number of shares. Similarly, Ibbotson Assocites uses the number of shares to construct a set of size indices. The breakpoints for size portfolios are defined by grouping the largest 20 percent shares for large-cap index, the next 30 percent for mid-cap index, the next 30 percent for small-cap index, and the smallest 20 percent for micro-cap index. In contrast,

Dimson and Marsh (2001) apply market capitalization deciles to establish micro-cap, small-cap, and large-cap indices.

With Dimson and Marsh (2001) approach, the research establishes decile size portfolios based on cumulative market capitalization, rather than number of shares, for three reasons. Firstly, FTSE small-cap index is based on the bottom 10% by market capitalization of each market. To construct decile size portfolios on market capitalization, the research maximizes its comparability with FTSE cap indices. Secondly, because empirical data shows that most companies in the UK Stock Market have low market value, it is necessary to apply the approach of market capitalization decile to classify small companies into the 1<sup>st</sup> and the 2<sup>nd</sup> portfolios, which captures small-cap effect. Thirdly, it is found in Table 3.4 that decile size portfolios constructed by equal number of shares leads to dynamic distribution in the market value between the 1<sup>st</sup> (small-cap) portfolio and the 10<sup>th</sup> (large-cap) portfolio. Especially, the 2<sup>nd</sup> and 11<sup>th</sup> columns in Table 3.4 show that the percentage of smallcap portfolio among overall market capitalization increases from 0.034% in 1969 to 0.158% in 1989, and that the percentage of large-cap portfolio among overall market capitalization declines from 86.93% to 79.01%. After 1989, they have opposite variation over time. Figure 3.1 and Figure 3.2 reflect the opposite trendline in the distribution of small-cap portfolio and large-cap portfolio among overall market capitalization in the period of 1969 to 1989 and afterwards. The unstable distribution in market value of small-cap portfolio and large-cap portfolio over time might lead to biased empirical result on size effect if same number of shares is allocated in decile size portfolios.

## [Table 3.4]

### [Figure 3.1, 3.2]

With Dimson and Marsh (2001) approach, decile size portfolios are constructed according to the fixed % market value banding after ranking all shares' market values from the lowest to the highest each month. The fixed % market value banding for the 1<sup>st</sup>, 2<sup>nd</sup>, 3<sup>rd</sup> through 8<sup>th</sup>, 9<sup>th</sup> and 10<sup>th</sup> portfolios are 1%, 9%, 10%, 15% and 15%, respectively.

#### • Adjusted Dimson and Marsh (2001) approach to form decile size portfolios

However, Dimson and Marsh (2001) approach raises another issue, that is not enough number of shares in some portfolios. Table 3.5 shows that there are less than 10 shares in the 5<sup>th</sup> to 10<sup>th</sup> portfolios in some years. Especially, there are only 1 or 2 shares in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolios. In order to overcome this limitation, the research adjusts Dimson and Marsh (2001) approach to ensure each formation: at least 10 large shares and more in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio; small-cap shares in the 1<sup>st</sup> and the 2<sup>nd</sup> portfolios with fixed market value banding. Because the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio must include at least 10 large shares, it has flexible market value banding over time. Therefore, with adjusted Dimson and Marsh (2001) approach, the research constructs decile size portfolios with fixed and flexible % market value banding from the 1<sup>st</sup> portfolio to the 10<sup>th</sup> portfolio.

# [Table 3.5]

As shown in Table 3.6, when 10 large shares are allocated in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio, its average % market value banding is 7.3%, 11.5%, 37.5%, respectively. If the fixed % market value banding for the 1<sup>st</sup> and 2<sup>nd</sup> portfolio is 1% and 9% as ones in Dimson and Marsh (2001), the 1<sup>st</sup>, 2<sup>nd</sup>, 8<sup>th</sup>, 9<sup>th</sup> and 10<sup>th</sup> portfolios account for 66.3 percent of overall market capitalization. Then only 33.7 percent of overall market capitalization is allocated to the 3<sup>rd</sup> to 7<sup>th</sup> portfolio, average 6.7% market value banding for them. This leads to % market value banding of the 2<sup>nd</sup> portfolio higher than the 3<sup>rd</sup> to 7<sup>th</sup> portfolio. Therefore, the research further reduces % market value banding of the 2<sup>nd</sup> portfolio from 9% to 4%, and allocates 38.7 percent of overall market capitalization to the 3<sup>rd</sup> to 7<sup>th</sup> portfolio, average 7.7% market value banding for them.

Table 3.6 shows the comparison in % market value banding between Dimson and Marsh (2001) approach and adjusted Dimson and Marsh (2001) approach. Dimson and Marsh (2001) approach applies the fixed 10% and 15% market value banding to form the 3<sup>rd</sup> to 10<sup>th</sup> portfolio, which leads to 1 or 2 shares in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio. To adjust Dimson and Marsh (2001) approach, the research increases average % market value banding of the 10<sup>th</sup> (large-cap) portfolio and decrease average % market value banding of the 2<sup>nd</sup> to 9<sup>th</sup> portfolio in order to ensure at least 10 shares in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio each month. Because the percentage of the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio among overall market capitalization volatiles over time, % market value banding of the 3<sup>rd</sup> to 7<sup>th</sup> portfolio is flexible in the allocation of at least

10 shares in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio. Therefore, decile size portfolios with adjusted Dimson and Marsh (2001) approach consists of the 1<sup>st</sup> and 2<sup>nd</sup> portfolio with fixed 1% and 4% market value banding, and the  $3^{rd}$  to  $10^{th}$  portfolio with flexible % market value banding.

#### [Table 3.6]

As shown in Table 3.7, decile size portfolios with adjusted Dimson and Marsh (2001) approach are summarized as following criteria:

- Fixed 1% and 4% market value banding in the 1<sup>st</sup> and the 2<sup>nd</sup> portfolio each month; average 41.3% of all shares included in the 1<sup>st</sup> portfolio and average 29.8% of all shares included in the 2<sup>nd</sup> portfolio, which reflects that most shares in the UK Stock Market have low market value;
- average 7.3% to 8.1% flexible market value banding in the 3<sup>rd</sup> through the 8<sup>th</sup> portfolio (similar to 10% market value banding in the 3<sup>rd</sup> through the 8<sup>th</sup> portfolio in Dimson and Marsh (2001) approach), minimum % market value banding between 4.3% to 4.7% and maximum % market value banding between 10.1% to 11.5%;
- average 11.5% and 37.5% flexible market value banding in the 9<sup>th</sup> and 10<sup>th</sup> portfolio to ensure at least 10 large shares in the 8<sup>th</sup>, 9<sup>th</sup>, and 10<sup>th</sup> portfolio each portfolio formation. 10 largest shares in the 10<sup>th</sup> portfolio account for 22.9 to 56.7 percent of overall market capitalization over time.

## [Table 3.7]

The research collects shares' price-to-book ratio (pb) from Datastream. Because there are few companies with price-to-book ratio prior to January 1980, the research explores monthly-on-monthly value effect in the UK Stock Market from 1980 to 2008. It excludes shares with negative pb. After ranking all shares' price-to-book ratios according to ascending order at the end of months, the research establishes decile value portfolios with same number of shares. The shares on the binding between two portfolios, especially for 1<sup>st</sup> to 3<sup>rd</sup> portfolios with lower price-to-book ratios, have the same pb. The research assigns the share with the same pb in the i<sup>th</sup> portfolio into the (i-1)<sup>th</sup> portfolio on the binding between two portfolios, which leads to slight difference in number of shares each value portfolio. For instance, Table 3.8 shows monthly distribution in the number of shares for decile value portfolios with 12 months' holding horizon when the first (last) decile portfolios are constructed in January 1980 (December 2007). The table presents slightly different number of shares from the 1<sup>st</sup> portfolio to the 10<sup>th</sup> portfolio each month. The first portfolio with low price-to-book ratio is designed as 'Low' portfolio. Conversely, the tenth portfolio with high price-to-book ratio is designed as 'High' portfolio.

# [Table 3.8]

There are 348 months in the overall sample period from January 1980 to December 2008. The first decile value portfolio is formed in the end of December 1979. The research rolls decile value portfolio formation forward each month, and uses last 3, 6,

12, 18, and 24 months as five holding horizons for the last decile value portfolios until the end of 2008. The research has 345, 342, 336, 330, and 324 overlapping sample observations for value effect with 3, 6, 12, 18, and 24 months' holding horizon, respectively. Table 3.3 shows that there are an average 1528 to 1647 (1148 to 1191) shares to construct equal-(value-) weighted value portfolios with five holding horizons.

# Appendix: tables and figures in Chapter 3

Table 3.1 The performance of average monthly Winner/Loser return  $R_{wml}$  based on lagged 6 months' formation: skipping one month (i-2 to i-6) and unskipping one month (i-1 to i-6)

Panel A Equal-weighted									
		$R_{wml}$ (%)		% R <sub>wml</sub> >0					
Holding months	Skipping	Unskipping	Difference	Skipping	Unskipping	Difference			
3	0.80	0.02	0.78	66.7	53.4	13.3			
6	0.93	0.54	0.39	73.0	64.0	9.0			
12	0.73	0.56	0.17	75.7	70.2	5.5			
18	0.39	0.21	0.18	67.3	62.0	5.3			
24	0.31	0.12	0.19	62.9	57.6	5.3			

Panel B Value-weighted									
		$R_{wml}$ (%)		% R <sub>wml</sub> >0					
Holding months	Skipping	Unskipping	Difference	Skipping	Unskipping	Difference			
3	1.27	0.76	0.51	65.6	59.9	5.7			
6	1.18	0.94	0.24	71.1	64.0	7.1			
12	0.93	0.82	0.11	72.1	70.8	1.3			
18	0.46	0.34	0.12	63.6	62.2	1.4			
24	0.41	0.41	0.00	62.9	61.3	1.6			

Note: The table reports the the performance of average monthly Winner/Loser return  $R_{wml}$  with lagged 6 months' share returns: skipping one month and unskipping one month between the formation period and the holding period. Winner/Loser return  $R_{wml}$  is estimated by the return spread between 'Winner' portfolio and 'Loser' portfolio. '%  $R_{wml} > 0$ ' denotes the percentage having positive Winner/Loser return  $R_{wml}$  in all observations.

Panel A Equal-weighted									
			% R <sub>wml</sub> >0						
Holding months	Skipping	Unskipping	Difference	Skipping	Unskipping	Difference			
3	1.06	0.51	0.55	70.6	61.9	8.7			
6	1.01	0.74	0.27	74.9	70.5	4.4			
12	0.50	0.40	0.10	70.2	67.5	2.7			
18	0.26	0.16	0.10	64.1	62.8	1.3			
24	0.03	-0.09	-0.09 0.12		54.6	2.1			

Table 3.2 The performance of average monthly Winner/Loser return  $R_{wml}$  based on lagged 12 months' formation: skipping one month (i-2 to i-12) and unskipping one month (i-1 to i-12)

#### Panel B Value-weighted $R_{wml}$ (%) $\ \ R_{wml} > 0$ Holding months Skipping Unskipping Difference Skipping Unskipping Difference 3 1.33 0.97 0.36 67.6 61.1 6.5 6 1.16 0.98 0.18 71.7 66.3 5.4 0.57 66.3 12 0.55 0.02 66.2 0.10.28 0.23 0.05 62.0 -0.3 18 61.7 0.12 0.11 0.01 55.2 -1.5 24 56.7

Note: The table reports the the performance of average monthly Winner/Loser return  $R_{wml}$  with lagged 12 months' share returns: skipping one month and unskipping one month between the formation period and the holding period. Winner/Loser return  $R_{wml}$  is estimated by the return spread between 'Winner' portfolio and 'Loser' portfolio. '%  $R_{wml} > 0$ ' denotes the percentage having positive Winner/Loser return  $R_{wml}$  in all observations.

Table 3.3 The average number of shares in the sample period in the investigation of momentum, size, and value effect

	Momentun	n effect (6)	Momentum effect (12)		Size effect		Value effect	
Holding months	EW	VW	EW	VW	EW	VW	EW	VW
3	1682	1638	1587	1558	1822	1822	1647	1191
6	1680	1635	1584	1555	1817	1817	1630	1186
12	1673	1628	1579	1549	1807	1807	1597	1175
18	1668	1621	1575	1544	1795	1795	1563	1161
24	1662	1615	1572	1540	1785	1785	1528	1148

Note: The table reports the average number of shares in the sample period in the investigation of momentum, size, and value effect with five holding horizons. Momentum effect(6) and Momentum effect(12) stand for momentum portfolios based on shares' returns in lagged 6 and 12 months, respectively. EW and VW stands for equal-weighed portfolio and value-weighted portfolio, respectively.

capitan										
Year	Small	$2^{nd}$	3 <sup>rd</sup>	$4^{\text{th}}$	$5^{\text{th}}$	$6^{\text{th}}$	7 <sup>th</sup>	$8^{\rm th}$	9 <sup>th</sup>	Large
1956	0.00086	0.00207	0.00369	0.00618	0.00924	0.01385	0.02226	0.03807	0.07696	0.82683
1957	0.00088	0.00203	0.00349	0.00551	0.00808	0.01168	0.01957	0.03391	0.07043	0.84443
1958	0.00088	0.00225	0.00375	0.00602	0.00846	0.01236	0.02014	0.03535	0.07024	0.84055
1959	0.00066	0.00177	0.00321	0.00524	0.00817	0.01224	0.02074	0.03627	0.07837	0.83333
1960	0.00063	0.00182	0.00343	0.00561	0.00866	0.01341	0.02164	0.03664	0.07835	0.82981
1961	0.00079	0.00203	0.00395	0.00630	0.00955	0.01462	0.02309	0.03952	0.08112	0.81905
1962	0.00063	0.00191	0.00381	0.00585	0.00928	0.01395	0.02228	0.03803	0.08060	0.82368
1963	0.00062	0.00191	0.00390	0.00614	0.00985	0.01528	0.02369	0.04040	0.08067	0.81754
1964	0.00066	0.00214	0.00438	0.00711	0.01069	0.01664	0.02599	0.04264	0.07625	0.81350
1965	0.00068	0.00248	0.00481	0.00749	0.01097	0.01701	0.02648	0.04208	0.07566	0.81235
1966	0.00070	0.00224	0.00446	0.00704	0.01046	0.01599	0.02516	0.04089	0.07466	0.81840
1967	0.00056	0.00188	0.00376	0.00598	0.00898	0.01431	0.02210	0.03571	0.06659	0.84013
1968	0.00048	0.00157	0.00317	0.00513	0.00773	0.01229	0.02000	0.03318	0.06392	0.85253
1969	0.00034	0.00131	0.00267	0.00315	0.00677	0.01229	0.01752	0.02859	0.05850	0.86932
1970	0.00038	0.00140	0.00275	0.00429	0.00630	0.01046	0.01705	0.02894	0.06164	0.86678
1971	0.00030	0.00140	0.00273	0.00429	0.00643	0.01019	0.01646	0.02894	0.06337	0.86554
1972	0.00047	0.00181	0.00200	0.00425	0.00043	0.01291	0.02043	0.02000	0.00337	0.83924
1972	0.00032	0.00101	0.00393	0.00545	0.00860	0.01201	0.02043	0.03436	0.07055	0.84028
1973	0.00005	0.00227	0.00373	0.00628	0.00885	0.01303	0.01990	0.03430	0.07033	0.83854
1075	0.00070	0.00257	0.00427	0.00028	0.00365	0.01313	0.01577	0.02649	0.07230	0.85034
1975	0.00070	0.00150	0.00338	0.00303	0.00730	0.01158	0.01021	0.02049	0.00713	0.83384
1970	0.00000	0.00163	0.00276	0.00480	0.00707	0.01136	0.01700	0.03435	0.00304	0.83304
1977	0.00005	0.00103	0.00270	0.00444	0.00707	0.01130	0.01742	0.03120	0.07892	0.84449
1978	0.00070	0.00193	0.00347	0.00580	0.00930	0.01475	0.02290	0.04193	0.10393	0.79510
1979	0.00111	0.00208	0.00433	0.00089	0.01085	0.01033	0.02437	0.04344	0.10112	0.78800
1980	0.00112	0.00231	0.00419	0.00073	0.01049	0.01338	0.02383	0.04133	0.09399	0.79777
1901	0.00083	0.00191	0.00326	0.00521	0.00809	0.01201	0.02088	0.04033	0.09002	0.81003
1982	0.00085	0.00193	0.00550	0.00327	0.00859	0.01529	0.02203	0.04200	0.10547	0.79870
1903	0.00074	0.00170	0.00293	0.00408	0.00713	0.01178	0.02009	0.03770	0.09938	0.01505
1904	0.00093	0.00209	0.00300	0.00534	0.00848	0.01279	0.02041	0.04074	0.09979	0.80301
1965	0.00084	0.00194	0.00330	0.00533	0.00799	0.01227	0.01980	0.03900	0.09493	0.81452
1980	0.00088	0.00208	0.00343	0.00555	0.00780	0.01250	0.01985	0.03049	0.09024	0.81339
1987	0.00107	0.00238	0.00379	0.00559	0.00815	0.01255	0.02129	0.03882	0.09282	0.81354
1966	0.00140	0.00303	0.00403	0.00009	0.00980	0.01487	0.02443	0.04303	0.09848	0.79347
1989	0.00158	0.00323	0.00499	0.00720	0.01056	0.01640	0.02522	0.04310	0.09757	0.79016
1990	0.00115	0.00235	0.00359	0.00548	0.00783	0.01191	0.01953	0.03539	0.09049	0.82228
1991	0.00064	0.00144	0.00233	0.00368	0.00573	0.00879	0.01536	0.03146	0.08359	0.84698
1992	0.00049	0.00114	0.00202	0.00342	0.00546	0.00886	0.01/33	0.03520	0.09567	0.83041
1993	0.00035	0.00081	0.00155	0.00270	0.00439	0.00774	0.01552	0.03300	0.09474	0.83940
1994	0.00053	0.00123	0.00234	0.00406	0.00649	0.01064	0.018/6	0.03575	0.09880	0.82141
1995	0.00068	0.00163	0.00294	0.00483	0.00764	0.01227	0.01959	0.03/56	0.09798	0.81487
1996	0.00055	0.00133	0.00253	0.00429	0.00686	0.01134	0.01882	0.03501	0.09196	0.82/31
1997	0.00057	0.00136	0.00251	0.00421	0.00656	0.01085	0.01788	0.03419	0.08570	0.83618
1998	0.00043	0.00104	0.00195	0.00330	0.00512	0.00856	0.01499	0.02823	0.0/1/6	0.86463
1999	0.00031	0.00074	0.00135	0.00220	0.00348	0.00616	0.01147	0.02153	0.05211	0.90066
2000	0.00032	0.00076	0.00133	0.00229	0.00389	0.00/17	0.01295	0.02439	0.06052	0.88639
2001	0.00027	0.00065	0.00115	0.00202	0.00332	0.00613	0.01119	0.02216	0.05463	0.89849
2002	0.00019	0.00050	0.00093	0.00151	0.00264	0.00449	0.00929	0.01922	0.05157	0.90967
2003	0.00018	0.00049	0.00089	0.00144	0.00247	0.00465	0.00900	0.01981	0.05296	0.90811
2004	0.00020	0.00056	0.00111	0.00197	0.00336	0.00590	0.01156	0.02430	0.06383	0.88720
2005	0.00021	0.00061	0.00116	0.00206	0.00350	0.00617	0.01141	0.02252	0.06349	0.88887
2006	0.00018	0.00051	0.00110	0.00195	0.00311	0.00539	0.01000	0.01980	0.05701	0.90094
2007	0.00026	0.00076	0.00140	0.00234	0.00391	0.00674	0.01199	0.02246	0.06292	0.88723
2008	0.00026	0.00070	0.00133	0.00237	0.00385	0.00629	0.01135	0.02128	0.05414	0.89843
Average	0.00065	0.00167	0.00298	0.00474	0.00725	0.01139	0.01868	0.03371	0.07862	0.84032

 Table 3.4 The percentage of decile size portfolios based on number of shares in overall market capitalization

Note: At the end of each year decile size portfolios are formed with the same number of shares after ranking all shares' market values from the smallest to the largest



Figure 3.1 The percentage of the 1<sup>st</sup> (small-cap) portfolio among overall market capitalization over years with portfolio formation based on number of shares

Note: The figure plots annual percentage of the 1<sup>st</sup> (small-cap) portfolio among overall market capitalization from 1968 to 2008. At the end of each year decile size portfolios are formed with the same number of shares after ranking all shares' market values from the lowest to the highest.



Figure 3.2 The percentage of the 10<sup>th</sup> (large-cap) portfolio among overall market capitalization over years with portfolio formation based on number of shares

Note: The figure plots annual percentage of the 10<sup>th</sup> (large-cap) portfolio among overall market capitalization from 1968 to 2008. At the end of each year decile size portfolios are formed with the same number of shares after ranking all shares' market values from the lowest to the highest.

Table 3.3	, The I	lumper	of shares h	i ueche	size por u	tonos wi	III DIIII50	n anu iv	1ai sii (20	01) approac
Year	Small	$2^{nd}$	3 <sup>rd</sup>	$4^{\text{th}}$	$5^{\text{th}}$	$6^{th}$	$7^{\text{th}}$	$8^{\text{th}}$	$9^{\text{th}}$	Large
1956	855	118	48	25	9	5	1	1	1	1
1957	879	108	43	17	8	3	1	0	1	1
1958	846	112	43	20	9	5	1	1	1	1
1959	829	101	41	22	9	4	2	1	1	1
1960	808	101	41	23	10	5	2	1	1	2
1961	768	107	42	23	12	6	4	2	1	1
1962	783	107	30	23	12	7	4	2	1	2
1063	785	112	40	21	13	7	4	2	1	2
1903	710	112	40	21	12	0	2	2	1	1
1904	740	120	43	22	12	0 7	3	2	1	1
1903	700	101	40	22	12	7	4	1	1	1
1900	702	124	44	20	12	/	3	2	1	1
1967	113	109	34	17	9	5	3	1	1	1
1968	/69	91	30	15	8	5	3	1	1	1
1969	757	73	26	11	6	3	2	0	1	1
1970	719	12	25	12	7	4	1	1	1	1
1971	694	67	21	12	7	5	1	1	1	1
1972	633	80	26	15	9	5	3	2	1	1
1973	608	78	25	13	7	5	3	1	1	1
1974	583	73	25	13	9	5	4	2	1	1
1975	591	54	20	12	6	4	2	2	1	1
1976	1854	196	76	41	26	17	10	7	4	2
1977	1761	173	69	37	23	16	9	5	2	2
1978	1580	196	82	47	30	19	12	8	4	2
1979	1501	213	87	49	31	19	11	7	4	3
1980	1452	201	81	44	30	17	11	7	3	2
1981	1361	160	68	39	23	16	11	7	2	2
1982	1365	163	72	40	24	17	11	8	4	2
1983	1389	145	65	37	22	15	10	6	3	3
1984	1374	161	69	40	24	15	11	7	4	3
1985	1411	160	64	37	24	16	10	6	4	2
1986	1442	157	65	34	23	16	11	6	5	3
1987	1435	172	67	36	24	17	11	6	4	3
1988	1421	201	76	42	26	19	13	9	5	3
1989	1430	214	81	41	20	20	14	9	5	3
1990	1480	162	63	35	23	17	10	7	5	3
1991	1438	124	51	28	19	14	9	6	4	3
1992	1320	116	53	30	20	11	8	5	4	3
1993	1276	102	49	30	17	11	8	6	4	3
1994	1270	117	51	32	21	14	9	6	5	3
1005	1215	136	57	32	21	14	10	7	5	1
1995	1/22	120	59	22	23	14	10	7	5	4
1990	1432	139	50	21	21	14	9	7	5	2
1997	1515	140	39	20	21	13	9	5	4	5
1996	1505	155	47	20	19	15	07	2	2	5
1999	1385	8/	30	21	15	9	7	3	3	2
2000	1453	101	37	22	15	9	5	3	2	2
2001	1554	93	38	20	12	7	4	2	1	2
2002	1535	84	35	19	11	5	3	2	1	2
2003	1478	82	31	16	9	6	3	2	1	2
2004	1411	95	38	20	11	6	4	2	1	2
2005	1591	105	43	22	12	7	4	3	1	2
2006	1846	112	44	23	11	7	4	3	2	2
2007	1935	134	51	27	14	8	6	3	3	2
2008	1985	133	48	23	9	7	5	4	2	2

Table 3.5 The number of shares in decile size portfolios with Dimson and Marsh (2001) approach

Note: With Dimson and Marsh (2001) approach, decile size portfolios are formed according to the fixed % market value banding after ranking all shares' market values from the lowest to the highest each year. The fixed % market value banding for the 1<sup>st</sup>, 2<sup>nd</sup>, 3<sup>rd</sup> through 8<sup>th</sup>, 9<sup>th</sup> and 10<sup>th</sup> portfolios are 1%, 9%, 10%, 15% and 15%, respectively.

Portfolio	DM(2001) approach	adjusted DM(2001) approach
Small-cap	1.0	1.0
$2^{nd}$	9.0	4.0
3 <sup>rd</sup>	10.0	7.7
$4^{th}$	10.0	7.7
5 <sup>th</sup>	10.0	7.7
$6^{\text{th}}$	10.0	7.7
$7^{\text{th}}$	10.0	8.1
8 <sup>th</sup>	10.0	7.3
$9^{\text{th}}$	15.0	11.5
Large-cap	15.0	37.5

Table 3.6 The comparison in % market value banding between Dimson and Marsh(2001) approach and adjusted Dimson and Marsh (2001) approach.

Note: With Dimson and Marsh (2001) approach, decile size portfolios are formed according to the fixed % market value banding after ranking all shares' market values from the lowest to the highest each month. The fixed % market value banding for the  $1^{st}$ ,  $2^{nd}$ ,  $3^{rd}$  through  $8^{th}$ ,  $9^{th}$  and  $10^{th}$  portfolios are 1%, 9%, 10%, 15% and 15%, respectively. With adjusted Dimson and Marsh (2001) approach, the fixed % market value banding for the  $1^{st}$  and  $2^{nd}$  is 1% and 4%, average 7.3 to 8.1 % flexible market value banding is arranged for the  $3^{rd}$  through the  $8^{th}$  portfolios. 10 companies are arranged in the  $8^{th}$ ,  $9^{th}$ , and  $10^{th}$  portfolios with flexible market value banding.

Portfolio	Average number of	Average % MV	Max % MV	Min % MV
	shares (%)	banding	banding	banding
Small-cap	617 (41.3)	1.0	1.0	1.0
$2^{nd}$	432 (29.8)	4.0	4.0	4.0
3 <sup>rd</sup>	225 (15.3)	7.7	11.0	4.3
$4^{\text{th}}$	78 (5.6)	7.7	10.9	4.3
$5^{\text{th}}$	39 (2.8)	7.7	11.0	4.4
$6^{th}$	24 (1.7)	7.7	10.9	4.3
$7^{\rm th}$	17 (1.2)	8.1	11.5	4.7
$8^{th}$	10 (0.8)	7.3	10.1	4.6
9 <sup>th</sup>	10 (0.8)	11.5	16.1	8.8
Large-cap	10 (0.8)	37.5	56.7	22.9

Table 3.7 The summary statistics on average number of shares(%) and % market value (MV) banding for decile size portfolios by adjusted Dimson and Marsh (2001) approach

Note: With adjusted Dimson and Marsh (2001) approach, decile size portfolios are formed by market capitalization after ranking all shares' market values from the lowest to the highest each month. The fixed % market value banding for the  $1^{st}$  and  $2^{nd}$  portfolios is 1% and 4%. Average 7.3% to 8.1 % flexible market value banding is arranged for the  $3^{rd}$  through the  $8^{th}$  portfolio. 10 shares are arranged in the  $8^{th}$ ,  $9^{th}$ , and  $10^{th}$  portfolios with flexible market value banding.

			(1-1	nonens	norums	10112011)				
Date	Low	2 <sup>nd</sup>	3 <sup>rd</sup>	4 <sup>th</sup>	5 <sup>th</sup>	6 <sup>th</sup>	7 <sup>th</sup>	8 <sup>th</sup>	9 <sup>th</sup>	High
1980/1	25	26	24	21	25	23	26	22	24	29
1980/2	24	27	22	26	21	27	21	24	24	30
1980/3	27	23	24	22	24	27	21	24	24	31
1980/4	25	27	20	28	23	24	21	24	24	31
1980/5	26	23	23	24	24	28	20	24	24	31
1980/6	24	24	25	25	23	23	24	24	25	30
1980/7	25	25	23	23	25	24	23	24	24	32
1980/8	26	23	24	24	27	22	22	24	25	31
1980/9	24	24	25	25	26	21	23	24	24	32
1980/10	26	22	26	24	25	21	24	24	24	32
1980/11	24	25	23	24	25	23	25	23	24	32
1980/12	30	22	26	22	25	27	23	26	24	25
1981/1	27	25	24	26	23	25	25	25	25	30
1981/2	26	27	22	28	22	25	26	24	25	31
1981/3	25	25	26	29	21	25	24	26	24	31
1981/4	26	26	24	24	26	25	24	25	25	31
1981/5	26	25	25	24	25	25	27	23	25	32
1981/6	25	25	25	26	25	26	23	25	25	32
1981/7	27	27	23	29	19	25	25	26	24	33
1981/8	27	25	23	25	27	26	23	24	25	33
1981/9	25	29	21	27	24	24	25	26	24	33
1981/10	29	21	27	24	24	27	23	25	25	34
1981/11	26	26	28	26	24	27	25	26	26	28
1981/12	28	26	25	25	26	29	23	27	25	28
2006/1	292	294	295	286	287	289	291	287	289	293
2006/2	294	298	288	291	292	294	291	288	293	299
2006/3	296	300	300	294	293	297	298	291	295	304
2006/4	303	295	310	288	308	293	301	294	299	300
2006/5	304	302	309	290	301	302	301	301	299	305
2006/6	311	307	295	306	306	299	305	304	305	304
2006/7	311	310	302	301	311	304	305	306	305	312
2006/8	311	308	308	305	303	308	306	309	307	313
2006/9	311	308	317	300	315	303	315	305	307	314
2006/10	319	316	295	315	306	312	310	311	306	318
2006/11	312	320	305	313	317	307	310	312	312	320
2006/12	320	313	317	317	311	313	314	315	315	319
2007/1	322	321	322	324	318	321	324	318	319	325
2007/2	331	317	324	327	320	325	318	322	323	328
2007/3	336	331	311	329	331	318	329	325	324	328
2007/4	335	326	329	323	335	325	326	325	329	328
2007/5	332	330	330	332	327	331	326	324	330	334
2007/6	338	324	332	333	334	331	326	331	330	333
2007/7	338	330	337	326	329	332	332	332	334	337
2007/8	334	350	317	343	322	332	339	331	329	341
2007/9	337	340	330	343	331	339	326	334	337	335
2007/10	341	344	331	337	337	335	335	338	335	338
2007/11	342	334	344	339	333	341	337	334	338	339
2007/12	340	338	346	343	330	337	339	339	339	347

 Table 3.8 The number of shares of decile value portfolios each month from 1980 to 2007

 (12 months' holding horizon)

Note: After ranking all shares' price to book values (pb) according to ascending order at the end of months, decile value portfolios with 12 months' holding horizon are formed. The research assigns the share with the same pb in the i<sup>th</sup> portfolio into the (i-1)<sup>th</sup> portfolio on the binding between two portfolios. 'Low' portfolio includes shares with low price-to-book ratio. 'High' portfolio includes shares with high price-to-book ratio.

# **Chapter 4 The Time-variation in Style Effects**

As the beginning of empirical analysis, this chapter mainly focuses on the first research question in the thesis of how the time-variation in style effects in the UK Stock Market exists. The research classifies firm characteristics into three kinds in terms of lagged share return, market value, and price-to-book ratio, and examines the time-variation in their impact on cross-sectional share returns, respectively. The remainder of the chapter proceeds as followings. Section 4.1 describes research methods for investigating the performance of time-varying style effects. Section 4.2 details data analysis on the time-variation in momentum effect with 6 and 12 months' formation lengths, respectively. Section 4.3 and Section 4.4 report data analysis on the time-variation in size effects. Section 4.6 discusses the correlation between style effects. Section 4.7 presents style effects in business cycles and market conditions. The last section comes to conclusion with a summary of results. The empirical analysis is based on the examination of equal-(value-) weighted style effects with 3, 6, 12, 18, and 24 months' holding horizons.

# 4.1 Research Methods

The research uses two indicators, long/short return  $R_{lms}$  and style coefficient  $\gamma$ , to measure time-varying style effects. Besides traditional indicator long/short return

 $R_{lms}$  in the majority of literature on style effects, the research uses style coefficient  $\gamma$  to investigate time-varying style effects. Two indicators are estimated after decile portfolios are constructed by firms' characteristics, such as lagged share returns, market value, and price-to-book ratio. Long/short return  $R_{lms}$  is return spread between top/bottom portfolios. It contains the information of shares in top and bottom portfolios in terms of firm characteristics and share return, and causes a loss of relevant information of shares in the  $2^{nd}$  to  $9^{th}$  portfolio. Style coefficient  $\gamma$  is the slope coefficient of cross-sectional regression of decile portfolios' returns in the holding period on their average values of firm characteristics in the formation period. It contains the information of all shares across decile portfolios. So, it is expected that style coefficient  $\gamma$  could capture much more information on the relation between share return and firm characteristics than long/short return  $R_{lms}$ .

Style coefficient  $\gamma$  has two implications in the empirical work. Firstly, Chapter 5 focuses on the impact of lagged macroeconomic variables on time-varying style effects. Because some macroeconomic variables might explain cross-sectional variation in returns of all shares, the research expects that they would have explanatory power for style coefficient  $\gamma$  stronger than long/short return R<sub>lms</sub>. Secondly, the test on the impact of lagged macroeconomic variables on the time-variation in style coefficient  $\gamma$  helps to diagnose the efficiency of active trading strategy on any decile portfolio, which is discussed in Chapter 6 and Chapter 7. Besides passive and active trading strategies on top and bottom portfolios, the research explores the profitability of active trading strategy on any decile portfolios.

coefficient  $\gamma$  offers analysis on the return forecast of any decile portfolio, and the profitability of holding long/short any decile portfolio.

#### 4.1.1 The estimation of style coefficient $\gamma$

The research uses Fama-Macbeth (1973) approach to estimate series average style coefficient  $\gamma$  of momentum, size, and value effect in terms of short-/medium-/long-term of 3, 6, 12, 18, and 24 months in the UK Stock Market, and then to examine their time-varying pattern.

Firstly, at the end of each month the research ranks all shares by firm characteristics in terms of past share return, market value, price-to-book ratio, individually, from the lowest to the highest, and classifies them into decile portfolios. The relevant portfolio formation is described in Chapter 3.

Secondly, the research treats mean lagged 6/12 months' compounded returns, log mean market values, mean price-to-book ratios of decile portfolios in the formation period t-1 as independent variables, and equal-(value-) weighted portfolio returns in following holding period as dependent variables. With equation (4.1), the research applies univarite ordinary least squares regression to estimate style coefficient  $\gamma$  each month.

$$R_{p,t} = \gamma_{0,t} + \gamma_{1,t} * FC_{p,t-1} + \varepsilon_{p,t}$$

$$t = 1, 2, ..., N, p = 1, 2, ..., 10,$$
(4.1)

where  $R_{p,t}$  is equal-(value-) weighted decile portfolio returns in the holding period t , FC<sub>p,t-1</sub> is the mean value of firm characteristics, such as lagged share return, market value, price-to-book ratio, individually, of decile portfolios in the formation period t-1.  $\gamma_{1,t}$  is the regression coefficient, measuring the impact of portfolios' firm characteristics in the formation period t-1 on portfolio returns in the holding period t, which is defined as style coefficient  $\gamma$ . To repeat this step each month, the research estimates series style coefficient  $\gamma$  in the sample period.

Thirdly, the research uses Fama-MacBeth (1973) approach to control unreliable regression statistics OLS standard errors due to cross-sectional correlation. It regards 24 months as one testing period to average 24 style coefficient  $\gamma$ . To repeat this step each month, the research collects series mean  $\gamma$  over the whole sample period.

Fourthly, Fama-MacBeth (1973) approach assumes no series autocorrelation, which leads to undervalued standard deviation in the parameter test and over-rejection of the null hypothesis of mean  $\gamma = 0$ . To adjust underestimated standard errors due to series autocorrelation, the research uses a finite period adjustment through Newey-West (1987) approach. Newey-West (1987) suggests the estimation of the variance by using a long-run variance matrix instead of the standard deviation of series cross-sectional regression coefficients in Fama-MacBeth (1973) approach. This approach includes the estimation of series covariance across time. With Newey-West adjustment approach, the research calculates series t-statistics for mean  $\gamma$  in one testing period monthly by monthly.

How wide a window k should be used in the Newer-West estimator? The window k is "..too short values of k, that is significantly autocorrelated, and you do not correct for correlation that might be there in the errors. Too long a value of k, together with a series that does not have much autocorrelation, and the performance of the estimate and test deteriorates." (Cochrane, 2001, p.222). In addition, too long window in the research might highlight long-term overall pattern in style effect but smooth out its short/medium-term fluctuations. It might not reflect actual timing-varying characteristics in style coefficient  $\gamma$ . Therefore, the research repeats above process monthly by monthly to estimate series 24-month's average style coefficient  $\gamma$  and their series t-statistics, and then to examine time-varying style effects.

#### 4.1.2 The estimation of long/short return R<sub>lms</sub>

Following other studies, the research constructs zero-investment portfolio to estimate long/short return  $R_{lms}$ , that goes long/short position in top/bottom portfolio. After ranking all shares with their lagged share return, market value, and price-to-market (PB) ratio, to form decile portfolios, the research estimates monthly-by-monthly return spread between 'Winner' portfolio and 'Loser' portfolio, between 'Small' portfolio and 'Large' portfolio, and between 'Low' portfolio and 'High' portfolio, respectively. The variation through time of three kinds of series return spreads represents for the dynamitic performance of momentum, size, and value effect. The research regards the performance of long/short return  $R_{lms}$  as the second indicator to investigate the time-variation in style effects.
# **4.2 Momentum effect**

This section focuses on the investigation of the time-variation in momentum effect in the UK Stock Market from 1956 to 2008. As mentioned in Chapter 3, the empirical analysis on momentum effect in two subsections is based on shares' past 6 and 12 months' share returns and following 3, 6, 12, 18, and 24 months' holding returns, respectively. In the third subsection, the research compares the result in two approaches of skipping/unskipping one month between the formation period and the holding period to construct momentum portfolios as robustness test.

### 4.2.1 Momentum effect based on 6 months' formation length

In this subsection, the research examines the relation between shares' lagged 6 months' formation returns and following 3, 6, 12, 18, and 24 months' holding returns, respectively, in the UK Stock Market from 1955 to 2008. It presents empirical analysis on time-varying momentum effect in different holding horizons in terms of the performance of decile momentum portfolio returns, Winner/Loser return  $R_{wml}$ , and momentum coefficient  $\gamma_{mom}$ .

#### Decile momentum portfolio returns

The research investigates the performance of decile momentum portfolios in terms of average monthly excess return and the percentage of one portfolio with highest and lowest return among decile portfolios over the whole sample period. Monthly excess return of decile portfolios is the difference between monthly portfolio return and monthly risk-free interest rate, measured by 90-day Treasury Bill (TBILL). TBILL is collected from LSPD. Portfolio P1 consists of stocks with the lowest ranking lagged 6 months' formation returns and is represented as 'Loser' portfolio. Portfolio P10 consists of stocks with the highest ranking lagged 6 months' formation returns and is represented as 'Winner' portfolio. To investigate six-month's reversal effect in some time points, the research estimates the percentage of one portfolio with highest monthly excess return to reflect the percentage of 'Loser' portfolio outperforming other decile portfolios in the whole sample period. Conversely, the research estimates the percentage of one portfolio with lowest monthly excess return to reflect the percentage of 'Winner' portfolio underperforming other decile portfolios.

# [Table 4.1]

The performance of momentum portfolios' excess returns reflects the existence of momentum effect based on lagged 6 months' share returns in the UK Stock Market. Panel A1 (B1) of Table 4.1 presents the performance of decile portfolios' returns and standard deviations over different holding horizon. It shows when equal-(value-) weighted decile portfolios are established according to lagged 6 months' share returns and hold for following 3, 6, 12, 18, 24 months, 'Winner' portfolio yields the highest (the second highest for value-weighted 'Winner' portfolio based on 24 months' holding horizon) average monthly excess return. Conversely, equal-weighted 2<sup>nd</sup> portfolio with the second lowest lagged 6 months' share returns underperforms other portfolios, and value-weighted 'Loser' portfolio exhibits worst

performance among decile portfolios. For the case of five holding horizons, the standard deviation of 'Loser' portfolio return is highest, which means the volatility in 'Loser' portfolio return. The standard deviation of decile portfolio returns exhibit U pattern through the 1<sup>st</sup> to 10<sup>th</sup> portfolio. The standard deviation of 'Winner' portfolio return and 'Loser' portfolio return monotonically falls as holding horizon grows.

'Winner' ('Loser') portfolio's excess return over five holding horizons indicates the strongest (or stronger) momentum effect with 12 months' holding period. Panel A1 (B1) of Table 4.1 shows equal-(value-) weighted 'Winner' portfolio with 12 months' holding horizon yields 1.50% (0.97%) average monthly excess return, the highest among the cases with five holding horizons. 'Loser' portfolio with 12 months' holding horizon yields 0.94% (0.15%) of excess return per month, the (second) third lowest return among the case with five holding horizons. For different holding horizons, the power of 'Winner'('Loser') portfolio based on 12 months' holding horizon to maintain their previous return performance is the strongest (stronger). 'Winner' portfolio return rises over 3 to 12 months' holding horizon and then falls over 18 and 24 months' holding horizon. 'Loser' portfolio return becomes higher from 6 months' to 24 months' holding horizon.

'Winner' ('Loser') portfolio's high (low) holding returns are supported by its percentage with highest and lowest return among decile portfolios in all observations. Panel A2 and A3 (Panel B2 and B3) of Table 4.1 show that the percentage of each portfolio with highest and lowest monthly excess return among decile portfolios over the whole sample period exhibits the U-pattern. Firstly,

although it outperforms other portfolios, it is more likely for 'Winner' portfolio to have the poor performance, following 'Loser' portfolio and the 2<sup>nd</sup> portfolio. Similarly, although its return is lower or lowest among decile portfolios for all holding horizons, it is more likely for 'Loser' portfolio to yield the highest portfolio return than the 2<sup>nd</sup> to 9<sup>th</sup> portfolio. This reflects the existence of 6-month's price reversal in some time points. Secondly, equal-weighted 'Winner' portfolio has highest percentage of all observations to yield the highest excess return over decile portfolios, much higher than the percentage of all observation to yield the lowest excess return. This leads to 'Winner' portfolio's superior performance. Conversely, although equal-weighted 'Loser' portfolio with five holding horizons yields the highest excess return in 16.0 to 26.9 percentage of all observations, following the percentage of 'Winner' portfolio, it is more likely to earn the lowest return. Panel A3 shows that equal-weighted 'Loser' portfolio with five holding horizons has worst return performance among decile portfolios in 20.2 to 35.7 percentage of all observations, which counterbalances its higher return in good time. Thirdly, when equal-weighted 'Winner' portfolio is hold for 12 months, it yields the highest monthly excess return in decile portfolios in 31.3 percentage of all observations, but the lowest excess return in 6.7 percentage of all observation. This big spread in percentage between highest and lowest excess return plays role in its good return performance of 1.5% per month. Conversely, When equal-weighted 'Loser' portfolio is hold for 12 months, its percentage having highest return among decile portfolios is 16.0 percent in all observations, much lower than 35.7 percentage with lowest return. The big spread in percentage between lowest and highest excess return makes 'Loser' return down to 0.94% per month. The same pattern is for value-weighted case.

#### Winner/Loser return R<sub>wml</sub>

The research designates Winner/Loser return  $R_{wml}$  as the difference in average monthly excess return between the top (P10) and bottom (P1) portfolios, and investigates the performance of Winner/Loser return  $R_{wml}$  for five holding horizons in the overall sample period and subperiods.

# [Table 4.2]

Winner/Loser return  $R_{wml}$  based on 6 and 12 months' holding period is higher. Panel A1 (B1) of Table 4.2 indicates that t-statistics of equal-weighted Winner/Loser return  $R_{wml}$  is 2.97 and 3.17 when momentum long/short portfolio is hold for 6 and 12 months, and t-statistics of value-weighted Winner/Loser return  $R_{wml}$  ranges 2.12 to 4.19 when momentum long/short portfolio is hold for five horizons, which means existence of Winner/Loser return  $R_{wml}$  reliably different from zero. Equal-(value-) weighted Winner/Loser return  $R_{wml}$ , when long/short position is hold for 6 and 12 months, is 0.54 and 0.56 (0.94 and 0.82) percent per month, which is higher than ones with other three horizons. Their outperformance is supported by the percentage of positive/negative Winner/Loser return  $R_{wml}$  in the sample period. As shown in Panel A2 (B2), equal-(value-) weighted Winner/Loser position with 6 and 12 months' holding horizon yields positive return in 64.0 and 70.2 (64.0 and 70.8) percentage of all observations, higher than ones with other three horizons, higher than ones with other three horizon yields positive return in 64.0 and 70.2 (64.0 and 70.8) percentage of all observations, higher than ones with other three holding horizons.

Winner/Loser return  $R_{wml}$  exhibits the variation over time. Although Winner/Loser position with 6 and 12 months yields higher return, it exhibits the variation over subperiods. For example, as shown in Panel A1 of Table 4.2, when equal-weighted Winner/Loser position is hold for 6 months, it yields significant positive return after 1980, but (marginally) significant negative return from sample beginning time to 1959 and in the 1970s. The variation in Winner/Loser return  $R_{wml}$  over time is reflected in the variation in the percentage of positive Winner/Loser return  $R_{wml}$  over subperiods. Panel A2 shows that Winner/Loser position with 6 months' holding horizon yields positive return in more than 71.7 percentage of all observations since 1980, but only 27.1 percent from sample beginning time to 1959 and 44.2 percent in the 1970s. Similarly, Panel B1 shows that when value-weighted Winner/Loser position is hold for 6 months, it earns significant positive return in the 1960s and afterwards 1980, but unsignificant positive return from sample beginning time to 1959 and in the 1970s. This return variation is consistent with the variation in the percentage of positive Winner/Loser return R<sub>wml</sub> as shown in Panel B2.

Winner/Loser return  $R_{wml}$  is highest in the 1990s, but lowest in the 1970s. Panel A1 (B1) of Table 4.2 shows that equal-(value-) weighted zero-investment momentum portfolios based on all holding horizons yield significant positive return in the 1990s, more than 1.02(0.84) % per month and higher than corresponding return in most other subperiods. This can be reflected in Panel A2 (B2) that more than 70.8 (63.3) percentage of observations earns positive Winner/Loser return  $R_{wml}$  in the 1990s. Conversely, in the 1970s equal-weighted Winner/Loser return  $R_{wml}$  with five

horizons is negative due to more than 44 percentage of observations with negative return as shown in Panel A3, and value-weighted Winner/Loser return  $R_{wml}$  is negative or less than 0.09 due to low percentage of positive Winner/Loser return  $R_{wml}$  as shown in Panel B2.

# <u>Momentum coefficient y<sub>mom</sub></u>

With the equation (4.1), the research conducts monthly-on-monthly regression of equal-(value-) weighted decile portfolios' holding returns on their formation returns, applies Fama-Macbeth (1973) approach to adjust cross-sectional correlation, and Newey-West(1987) approach to adjust standard errors due to serial correlation from overlapping data in examination of momentum effect. With Fama-Macbeth (1973) approach, the research averages 24 regression coefficient  $\gamma_{mom}$  as one testing period, uses Newey-West (1987) approach to adjust standard errors and estimates t-statistics, and finds number (percentage) of significant momentum/short reversal effect among all observations. Confidence level for t-test is 95%. Different from long-term return reversal with three to five years' formation and holding horizons in De Bondt and Thaler (1985, 1987), the research examines return reversal based on lagged 6 months' share returns.

# [Table 4.3]

Monthly-on-monthly cross-sectional regression of portfolios' holding returns on their 6 months' formation returns reflects the strong momentum effect with 6 and 12 months' holding horizon. Panel A1 and A2 (Panel B1 and B2) of Table 4.3 show that equal-(value-) weighted portfolios based on 6 and 12 months' holding horizon have 35.4 and 43.7 (42.6 and 43.2) percentage of observations with significantly positive mean momentum coefficient  $\gamma_{mom}$  in the overall sample period, which are higher than ones with other three holding horizons, and only 2.9 and 2.6 (0 and 0.2) percentage of observations with significantly negative mean momentum coefficient  $\gamma_{mom}$ , the lowest among cases with five holding horizons.

Momentum coefficient  $\gamma_{mom}$  exhibits the variation over time. When observing the performance of momentum coefficient  $\gamma_{mom}$  in subperiods, except the 1950s due to limited observations after averaging 24 regression coefficient  $\gamma_{mom}$  as one test, it is found in Panel A1 (B1) of Table 4.3 that equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  is significant positive in more than 64.2 (30.0) percentage of all observations in the 1990s, higher than ones in other subperiods for corresponding holding horizons. But the corresponding percentage is no more than 12.5 (30.0) percent in the 1970s, lower than one in other subperiods. This pattern is consistent with findings on Winner/Loser return R<sub>wml</sub>.

In short, the return performance of momentum portfolios, Winner/Loser return  $R_{wml}$ , and momentum coefficient  $\gamma_{mom}$  reflect the existence of momentum effect based on lagged 6 months' formation in the UK Stock Market from 1956 to 2008. The performance of 'Winner', 'Loser' (or second 'Loser') and 'Winner/Loser' portfolios in five holding horizons confirms UK momentum effect in Antonious, Lam, and Paudyal (2007), Fletcher (2007) and Asness (2011). It is found that momentum effect on 6 and 12 months' holding horizon is stronger than ones with other holding horizons, and that momentum effect becomes weaker as holding horizon extends longer. Momentum effect exhibits the variation over time, strongest in the 1990s, but weakest in the 1970s. Following 'Winner' ('Loser') portfolio, 'Loser' ('Winner') portfolio yields the highest (lowest) return among decile portfolios more often than other eight portfolios.

# 4.2.2 Momentum effect based on 12 months' formation length

In this subsection, the research examines the impact of lagged 12 months' share returns on following 3, 6, 12, 18, and 24 months' returns, respectively, in the UK Stock Market from 1955 to 2008. As the previous subsection, the research presents empirical analysis on time-varying momentum effect in terms of return performance of decile momentum portfolios, Winner/Loser return  $R_{wml}$ , and momentum coefficient  $\gamma_{mom}$ .

#### Decile momentum portfolio returns

The research investigates the performance of decile momentum portfolios in terms of average monthly excess return and the percentage of one portfolio with highest and lowest return among decile portfolios over the whole sample period. Portfolio P1 includes shares with the lowest ranking 12 months' formation returns and is designated as 'Loser' portfolio. Portfolio P10 includes shares with the highest ranking 12 months' formation returns and is designated as 'Winner' portfolio. To investigate twelve-month's reversal effect in some time points, the research estimates the percentage of one portfolio with highest monthly excess return to reflect the percentage of 'Loser' portfolio outperforming other decile portfolios in the whole sample period. Conversely, the research estimates the percentage of one portfolio with lowest monthly excess return to reflect the percentage of 'Winner' portfolio underperforming other decile portfolios.

# [Table 4.4]

The performance of momentum portfolios' excess returns reflects the existence of momentum effect based on lagged 12 months' share returns in the UK Stock Market. Panel A1 (B1) of Table 4.4 presents the performance of decile momentum portfolios' returns and their standard deviation over different holding horizons. It shows when equal-(value-) weighted decile portfolios are formed according to previous 12 months' share returns and hold for next 3, 6, 12, and 18 months (3, 6, and 12 months), 'Winner' portfolio has the highest average monthly excess return among decile portfolios. When value-weighted portfolio is hold for next 18 or 24 months, the 9<sup>th</sup> portfolio with second highest lagged 12 months' share returns outperforms other portfolios. Conversely, 'Loser' portfolio exhibits poor return performance. Especially value-weighted 'Loser' portfolio yields the lowest average monthly excess return among decile portfolios, except for the case with 3 months' holding horizon. When equal-weighted portfolio is hold for 24 months, 'Loser' portfolio yields the highest average monthly excess return over decile portfolios, meaning that one-year's return reversal might take place when holding horizon extends to a certain extent. As momentum portfolios with 6 months' formation length, return on 'Loser'

portfolio has the highest standard deviation, followed by the 2<sup>nd</sup> portfolio or 'Winner' portfolio. The volatility of decile portfolio returns exhibit U pattern through the 1<sup>st</sup> to 10<sup>th</sup> portfolio. The standard deviation of 'Winner' portfolio return and 'Loser' portfolio return monotonically falls as holding horizon extends.

'Winner' ('Loser') portfolio's return over five holding horizons indicates strong momentum effect with 6 and 12 months' holding horizon. Panel A1 (B1) of Table 4.4 shows that average monthly excess return on equal-(value-) weighted 'Winner' portfolio monotonically falls from 1.52% to 1.30% (0.96% to 0.74%) when holding horizon is from 6 months to 24 months, and 'Loser' portfolio return monotonically increases from 0.78% to 1.39% (-0.02% to 0.63%). The opposite variation over holding horizons means that 'Winner' and 'Loser' portfolios based on 6 and 12 months' holding horizon have the strong power to maintain their previous 12 months' return performance, and that momentum effect becomes weaker as holding horizon extends.

'Winner' ('Loser') portfolio's high (low) holding returns is supported by its percentage of highest (lowest) return among decile portfolios in all observations. Panel A2 (B2) of Table 4.4 tabulates the percentage of observations where each portfolio yields the highest monthly excess return among decile portfolios over all observations, and indicates that equal-(value-) weighted 'Winner' portfolio based on 3 to 18 months' holding horizon takes up 23.5 to 35.4 (17.2 to 22.1) percent among all observations with highest monthly return, higher than other nine portfolios. When equal-(value-) weighted 'Winner' portfolio is hold for 6 and 12 months, its percentage having highest return among decile portfolios is 35.4 and 29.8 (22.1 and 20.5) in all observations, the highest or higher among five holding horizons. As indicated in Panel A3 (B3), equal-(value-) weighted 'Winner' portfolio with 6 and 12 months' holding horizon yields lowest return among decile portfolios in 6.7 and 8.5 (8.9 and 10.6) percentage of observations, the lowest and the second lowest among the cases with five horizons. The big spread between the percentage with highest return and the percentage with lowest return contributes to high return on 'Winner' portfolio. Conversely, When equal-(value-) weighted 'Loser' portfolio is hold for 6 and 12 months, its percentage having highest return among decile portfolios is 17.9 and 17.3 (13.2 and 14.4) in all observations, the lowest or second lowest among five holding horizons. It earns lowest return among decile portfolios in 33.0 and 37.3 (32.5 and 31.3) percentage of observations, the highest or higher among the cases with five horizons. The large spread between the percentage with lowest return and the percentage of observations, the highest or higher among five holding horizons. The large spread between the percentage with lowest return and the percentage with highest return contributes to low return or 'Loser' portfolio.

The percentage of decile momentum portfolios' highest monthly return over the whole sample period exhibits the U-pattern. Panel A2 (B2) of Table 4.4 shows that although 'Loser' portfolio return is lower or lowest among decile momentum portfolios for all holding horizons, it yields the highest portfolio return more often than the 2<sup>nd</sup> to 9<sup>th</sup> portfolio. Equal-(value-) weighted 'Loser' portfolio with highest monthly return has 17.3 to 24.0 (13.2 to 16.8) percentage of all observations for five holding horizons, following corresponding percentage of 'Winner' portfolio. This means the existence of twelve-month's reversal effect in 'Loser' portfolio in some time points.

# Winner/Loser return R<sub>wml</sub>

The research designates Winner/Loser return  $R_{wml}$  as the spread in average monthly excess return between the top (P10) and bottom (P1) portfolios, and investigates the performance of Winner/Loser return  $R_{wml}$  in the overall sample period and subperiods for five holding horizons. Panel A1 (B1) of Table 4.5 shows that equal-(value-) weighted Winner/Loser return  $R_{wml}$  in the overall sample period is from 0.41 to 0.74 (0.55 to 0.98) per month when long/short position is hold for 3 to 12 months. Its adjusted t-stats by Newey-West (1987) approach is more than 2.04, which means that Winner/Loser return  $R_{wml}$  is reliably different from zero. When Winner/Loser portfolio is hold for 18 and 24 months, the return spread is unsignificant positive or negative.

# [Table 4.5]

Winner/Loser return  $R_{wml}$  based on 6 months' holding period is highest. Panel A1 (B1) of Table 4.4 indicates that as holding horizon extends, equal-(value-) weighted 'Winner' portfolio's average monthly excess return monotonically descends from 1.52% (0.96%) for 6 months' holding horizon to 1.30% (0.74%) for 24 months' holding horizon. Conversely, 'Loser' portfolio return monotonically ascends from 0.78 to 1.39 (-0.02 to 0.63). 'Winner' and 'Loser' portfolio returns' converse variation over holding horizons leads to the highest Winner/Loser return  $R_{wml}$  for the case with 6 months' holding period, and the declining pattern of Winner/Loser return  $R_{wml}$  over holding horizons. The converse variation is consistent with the

performance of their percentage having highest and lowest returns as indicated in Panel A2 and A3 (Panel B2 and B3) of Table 4.4.

The declining pattern of Winner/Loser return  $R_{wml}$  over holding horizons is mainly caused by 'Loser' portfolio return. Panel A1 (B1) of Table 4.4 shows that equal-(value-) weighted 'Winner' portfolio's average monthly return over 6 to 24 months' holding horizon falls by 0.22 from 1.52 to 1.30 (0.22 from 0.96 to 0.74), the amount less than the rising magnitude of 'Loser' portfolio return, which is 0.61 from 0.78 to 1.39 (0.65 from -0.02 to 0.63). This result is different from the findings by Jegadeesh and Titman (2001) that both winners and losers contribute about equally to Winner/Loser return  $R_{wml}$  in the US Stock Market.

As Winner/Loser return  $R_{wml}$  performs, Panel A2 (B2) of Table 4.5 shows that the percentage of positive equal-(value-) weighted Winner/Loser return  $R_{wml}$  among all observations monotonically declines from 70.5 to 54.6 (66.3 to 56.7) percentage over 6 to 24 months' holding horizon. In addition, equal-(value-) weighted zero-investment portfolio based on 6 months' holding horizon yields positive Winner/Loser return  $R_{wml}$  in more than 59.2 (60.8) percentage of observations each subperiod. Conversely, equal-(value-) weighted zero-investment portfolios based on 24 months' holding horizon takes up below 55 percentage of observations with having positive Winner/Loser return  $R_{wml}$  in most subperiods. Panel A3 (B3) shows that in the 1960s, 1970s, 1980s, and 2000s the percentage of observations with negative equal-(value-) weighted Winner/Loser return  $R_{wml}$  ranges from 45.4 to 70.8

(38.3 to 49.2), almost the same or higher than the percentage with positive Winner/Loser return  $R_{wml}$ .

*Momentum effect is strongest in the 1990s, and weakest in the 1970s.* Panel A1 (B1) of Table 4.5 shows that equal-(value-) weighted zero-investment portfolios based on all holding horizons yield positive Winner/Loser average monthly return in the 1990s, which are more than 0.85% (0.66%) per month and reliably different from zero. In the 1970s equal-weighted Winner/Loser return  $R_{wml}$  with almost all holding horizons is negative, and the lowest among all subperiods. Similarly, in the 1970s value-weighted Winner/Loser return  $R_{wml}$  with all holding horizons is unsignificant positive except for the case with 18 months' holding horizon, and the lowest or lower among all subperiods.

# <u>Momentum coefficient γ<sub>mom</sub></u>

With equation (4.1), the research conducts monthly-on-monthly regression of equal-(value-) weighted decile portfolios' holding returns on their lagged 12 months' formation returns. With Fama-Macbeth (1973) approach, the research averages 24 regression coefficient  $\gamma_{mom}$  as one testing period. Then, using Newey-West (1987) approach to adjust standard errors and estimate t-statistics, the research finds the number of observations with significant momentum/short reversal effect and corresponding percentages among all observations. Confidence level for t-test is 95%.

[Table 4.6]

The performance of momentum coefficient  $\gamma_{mom}$  in overall period and subperiods supports the stronger momentum effect when portfolios are hold for 6 and 12 months. Panel A1 (B1) of Table 4.6 shows that equal-(value-) weighted portfolios based on 6 and 12 months' holding period have 42.5 and 39.9 (35.3 and 33.4) percentage of observations with significantly positive mean momentum coefficient  $\gamma_{\rm mom}$ , the highest percentage among five holding horizons. As shown in Panel A2 (B2), they have only 0.8 and 4.0 (0.0 and 3.3) percentage of observations with significantly negative mean momentum coefficient  $\gamma_{mom}$ , the lowest or lower percentage among five holding horizon. As the holding horizon extends from 6 months to 24 months, the percentage of observation with significantly positive mean momentum coefficient  $\gamma_{mom}$  monotonically declines from 42.5 to 22.6 (35.3 to 18.3), the percentage of observation with significantly negative mean momentum coefficient  $\gamma_{mom}$  rises from 0.8 to 18.5 (0 to 13.9). The similar varying pattern over holding horizons is in most subperiods. This indicates that the longer is holding horizon, the weaker is momentum effect. Especially, when portfolios are hold for 18 and 24 months, equal-weighted portfolios yield significantly negative mean momentum coefficient  $\gamma_{mom}$  in 31.7 and 58.3 percentage of observations in the 1970s, which is consistent with the performance of significant negative Winner/Loser return R<sub>wml</sub> in Panel A1 of Table 4.5.

In summary, there are same findings on the performance of series share returns between two momentum portfolios based on 6 and 12 months' formation length. The return performance of momentum portfolios, Winner/Loser return  $R_{wml}$ , and

momentum coefficient  $\gamma_{mom}$  reflects the existence of momentum effect based on lagged 12 months' formation in the UK Stock Market from 1956 to 2008. It is found that momentum effect with 6 and 12 months' holding horizon is stronger than ones with other holding horizons, and that momentum effect declines as holding horizon extends. Momentum effect exhibits the variation over time, strongest in the 1990s, but weakest in the 1970s. Following 'Winner' ('Loser') portfolio, 'Loser' ('Winner') portfolio yields the highest (lowest) return among decile portfolios more often than other eight portfolios.

#### 4.2.3 Robustness test on momentum effect

To avoid the effect of short-term reversal effect documented in Lehmann (1990) and Jegadeesh (1990), the research uses the approach with skipping one month between the formation period and the holding period to construct momentum portfolios, and compares empirical results with the findings in previous two subsections in which momentum portfolios are constructed without skipping one month. Momentum portfolios with skipping one month are constructed in the month t according to share returns in the months from t-2 to t-6, and t-2 to t-12. Table 4.7 presents the difference in average monthly return of 'Winner' portfolio, 'Loser' portfolio, and 'Winner/Loser' portfolio, respectively, between skipping one month approach and unskipping one month approach in the case of 6 months' and 12 months' formation lengths and corresponding five holding horizons. This return difference is denoted as ' $R_s$ - $R_{us}$ '.

#### [Table 4.7]

As shown in the 4<sup>th</sup> and 7<sup>th</sup> columns of Panel A and Panel B, Winner/Loser  $R_s$ - $R_{us}$  is positive over five holding horizons, which reflects that Winner/Loser return  $R_{wml}$ based on the approach with skipping one month in portfolio formation is higher than the approach without skipping one month; In the same holding horizon, the difference for the formation with lagged 6 months' share return is larger than the formation with lagged 12 months' share return; In the same formation length and holding horizon, the difference for equal-weigthed portfolio is larger than valueweigthed portfolio.

This difference in the performance of Winner/Loser return  $R_{wml}$  between two approaches of skipping and unskipping one-month in momentum formation becomes small as holding horizon extends from 3 months to 24 months. The difference is obvious in short-term holding case. For example, when equal-(value-) weighted Winner/Loser portfolio is constructed according to lagged 6 months' share returns and is hold for 3 and 6 month, average monthly holding return in the formation approach with skipping one month is 0.77 and 0.39 ( 0.51 and 0.24) larger than one in the formation approach without skipping one month. The corresponding difference for momentum effect with 12 months' formation period is 0.56 and 0.27 (0.36 and 0.18). The effect of skipping one month in formation declines or even disappears when the holding horizon becomes longer. For example, the difference is no more than 0.20 when Winner/Loser portfolio is hold for 12, 18, and 24 months. The difference in Winner/Loser return  $R_{wml}$  between two approaches of skipping and unskipping one month in momentum formation is mainly caused by 'Loser' portfolio. For example, the 2<sup>nd</sup> and 5<sup>th</sup> columns in Panel A show that equal-(value-) weighted 'Winner' portfolio with skipping one month in lagged 6 months' formation length yields average monthly return 0.15 and 0.08 (0.08 and 0.04) higher than one with unskipping pattern when 'Winner' portfolio is hold for 3 and 6 months. The 3<sup>rd</sup> and 6<sup>th</sup> columns show that equal-(value-) weighted 'Loser' portfolio with skipping one month in the formation yields average monthly return 0.62 and 0.31 (0.43 and 0.20) lower than one with unskipping pattern. The monthly price reversal in 'Loser' portfolio is stronger than one in 'Winner' portfolio, which mainly contributes to the difference in Winner/Loser return R<sub>wml</sub> between two approaches. The similar result is for the case of 12 months' formation in Panel B.

Table 4.8 reports the correlation coefficient of time series 'Winner' return, 'Loser' return, 'Winner/Loser' return  $R_{wml}$ , and momentum coefficient  $\gamma_{mom}$  between two approaches of skipping and unskipping one month in momentum portfolio formation. For 'Winner' return and 'Loser' return with five holding horizons, most correlation coefficients are higher than 0.92. For 'Winner/Loser' portfolio return  $R_{wml}$ , most correlation coefficients are higher than 0.82. For momentum coefficient  $\gamma_{mom}$ , most correlation coefficients are higher than 0.87.

[Table 4.8]

In summary, as robustness test, the research investigates momentum effect based on formation approach with skipping one month between the formation period and the holding period. It finds the existence of one-month's price reversal, which is mainly caused by 'Loser' portfolio. Empirical comparison shows high correlation coefficient in series 'Winner' return, 'Loser' return, 'Winner/Loser' return  $R_{wml}$ , and momentum coefficient  $\gamma_{mom}$  between two formation approaches with skipping and unskipping one month. To obtain conservative estimation on momentum effect, the research uses momentum indicators from formation approach with unskipping one month to conduct empirical analysis in following chapters.

# **4.3 Size Effect**

This section concentrates on the relation between shares' market values and their returns with five holding horizons, such as 3, 6, 12, 18, and 24 months, in the UK Stock Market from 1979 to 2008. It presents empirical analysis on time-varying size effect over different holding horizons in terms of decile size portfolio returns, Small/Large return  $R_{sml}$ , and size coefficient  $\gamma_{mv}$ .

#### Decile Size Portfolio returns

The research investigates the performance of decile size portfolios in terms of average monthly excess return and the percentage of one portfolio with highest/lowest return among decile portfolios from 1979 to 2008. It uses fixed % and flexible % market value banding in adjusted Dimson and Marsh (2001) approach to form decile size portfolios, which is described in Chapter 3. Portfolio P1 includes shares with the smallest ranking market value at the end of each month, and is defined as 'Small' portfolio. Portfolio P10 includes shares with the largest ranking market value at the end of each month, and is defined as 'Large' portfolio. Because small-cap effect and large-cap effect may exist when they outperform/or underperform other portfolios, the research estimates the percentage of one portfolio with highest and lowest monthly return to examine the percentage of small-cap premium and large-cap premium in the whole sample period.

[Table 4.9]

The return performance of equal-weighted size portfolios reflects existence of smallcap premium in the UK Stock Market. Panel A1 (B1) of Table 4.9 presents the performance of equal-(value-) decile size portfolios' average monthly excess returns and standard deviations over different holding horizons. It shows when equalweighted size portfolios are formed, 'Small' portfolio has the highest average monthly excess return among portfolios for five holding horizons. Value-weighted 'Small' portfolio return performs in the second highest position when it is hold for 12 to 24 months. 'Small' portfolio return over five holding horizons indicates that the portfolio based on 24 months' holding period exhibits the strongest small-cap premium. Panel A1 (B1) shows that equal-(value-) weighted 'Small' portfolio return monotonically increases from 0.56% to 1.13% (0.35% to 0.92%) as holding horizon extends from 3 months to 24 months, but its standard deviation monotonically decreases. The standard deviation of portfolio returns monotonically declines through the 1<sup>st</sup> to the 10<sup>th</sup> portfolio. 'Small' portfolio return is more volatile than other portfolios, and has highest standard deviation. The return on the 8<sup>th</sup>, 9<sup>th</sup>, and 'Large' portfolio, which include the largest and larger shares, perform steadily than other seven portfolios, and have lower standard deviation.

Because small-cap shares always have wider bid-ask spread as well as low trading volume, they might not actually reflect market trading prices. If the research compares portfolio return among  $2^{nd}$  to  $10^{th}$  ('Large-cap') portfolio, it is found that equal-(value-) weighted  $2^{nd}$  portfolio with the second smallest market share yields the lowest or second lowest average monthly return, and that equal-(value-) weighted  $8^{th}$  portfolio earns the highest or second highest return. This means the existence of

medium-cap effect, rather than small-cap effect and large-cap effect, in the UK Stock Market.

Higher 'Small' portfolio return is supported by its percentage of highest return among decile portfolios in all observations. Panel A2 (B2) of Table 4.9 offers the percentage of one portfolio yielding the highest monthly excess return among decile size portfolios over all observations, and indicates that equal-(value-) weighted 'Small' portfolio based on five holding horizon yields the highest monthly excess return in 22.7 to 37.5 (19.3 to 36.3) percentage of all observations, the highest among decile portfolios, except for value-weighted 'Small' portfolio with 3 months' holding horizon. This percentage exhibits monotonic increase as holding horizon extends from 3 months to 24 months. When equal-(value-) weighted 'Small' portfolio is hold for 24 months, its percentage having highest return among decile portfolios is 37.5 (36.3) in all observations, the highest among five holding horizons. Given bid-ask spread and low trading volume in small-cap shares discussed above, the research observes the percentage of 2<sup>nd</sup> to 10<sup>th</sup> portfolio with highest return among all observations, and finds that equal-(value-) weighted 8<sup>th</sup> portfolio with five holding horizons earns the highest return in 14.1 to 21.7 (14.6 to 21.7) percentage of observations, more than most other portfolios. This provides explanation for the outperformance of the 8<sup>th</sup> portfolio return.

The volatility of 'Small' portfolio return is supported by its percentage of highest and lowest return among decile portfolios in all observations. Although 'Small' portfolio yields the highest or higher average monthly excess return, and has the highest percentage with highest return among decile size portfolios over the sample period, it is likely to exhibit the worst return performance in some periods. Panel A3(B3) indicates that equal-(value-) weighted 'Small' portfolio yields the lowest return in 16.1 to 23.2 (17.9 to 24.3) percentage of all observations, higher than or almost same with the corresponding percentage for 'Large' portfolio. Especially, for equal-(value-) weighted 'Small' portfolio with 3 to 6 months' holding horizon, the percentages of observation with highest and lowest return are almost same. This could offer explanation for high standard deviation of 'Small' portfolio return.

Similarly, 'Large' portfolio exhibits higher frequency in the best and worst return performance among decile portfolios. As indicated in Panel A2 (B2), equal-(value-) weighted 'Large' portfolio yields the highest return among decile portfolios in 10.4 to 17.6 (10.7 to 20.2) percentage of all observations, following 'Small' portfolio, the 8<sup>th</sup> and 9<sup>th</sup> portfolios. Panel A3 (B3) shows that 'Large' portfolio yields the lowest return in 15.8 to 19.0 (18.2 to 25.7) percentage of all observations, following, following/or heading 'Small' portfolio. The extreme performance in return of 'Small' and 'Large' portfolio means the rotation between small-cap effect and large-cap effect.

#### <u>Small/Large return R<sub>sml</sub></u>

The research designates Small/Large return  $R_{sml}$  as the difference in average monthly excess return between the bottom ('Small') and top ('Large') portfolios, and investigates the performance of Small/Large return  $R_{sml}$  with five holding horizons in the overall sample period from 1979 to 2008 and three subperiods. Table 4.10 Panel A1(B1) shows that equal-(value-) weighted Small/Large return  $R_{sml}$  in overall sample period is from 0.10 to 0.44 (-0.06 to 0.30) per month when Small/Large position is hold for five horizons. Its adjusted t-stats by Newey-West (1987) approach is less than 1.15, which means that Small/Large return  $R_{sml}$  is unreliably different from zero.

#### [Table 4.10]

Small/Large return  $R_{sml}$  based on 24 months' holding period is highest. Panel A1(B1) of Table 4.9 indicates that as holding horizon extends, average monthly return on equal-(value-) weighted 'Small' portfolio monotonically increases from 0.56% (0.35%) for 3 months' holding horizon to 1.13% (0.92%) for 24 months' holding horizon. 'Large' portfolio return becomes higher from 0.46% to 0.69% (0.41% to 0.62%). Although two portfolio returns rise as holding horizon extends, 'Small' portfolio return has the rising magnitude of 0.57% (0.57%) per month, more than 'Large' portfolio's corresponding magnitude of 0.23% (0.21%). This leads to the highest Small/Large return  $R_{sml}$  when long/short position is hold for 24 months. The rising pattern of Small/Large return  $R_{sml}$  over holding horizons is mainly caused by 'Small' portfolio return.

Panel A2 (B2) of Table 4.10 shows that the percentage of positive equal-(value-) weighted Small/Large return  $R_{sml}$  in all observations rise from 45.8 to 61.3 (44.8 to 61.0) as holding horizon extends from 3 months to 24 months, which is consistent with rising Small/Large return  $R_{sml}$  over holding horizons. In the 1980s and the 2000s the percentage exhibits monotonically rise over holding horizons. Panel A3

(B3) presents that equal-(value-) weighted 'Small/Large' portfolio with five holding horizons yields negative average monthly return from 38.7 to 54.2 (39.0 to 55.2) percentage of observation. No obvious difference in the percentage of zeroinvestment portfolio having positive and negative Small/Large return  $R_{sml}$  leads to unsignificant positive return spread in the overall sample period.

*Small/Large return*  $R_{sml}$  *rotates over three subperiods.* Panel A1(B1) of Table 4.10 indicates that during the 1980s, equal-(value-) weighted Small/Large return  $R_{sml}$  is (marginally) positive with rising magnitude over holding horizons. Its strong performance is supported by more than 59.7% (52.7%) of observations having positive zero-investment return in the 1980s. In the 1990s Small/Large return  $R_{sml}$  is marginally negative and rises as holding horizon grows. It yields negative return in more than 60 percentage of observations. In the 2000s 'Small/Large' return  $R_{sml}$  is (marginally) significantly positive in 18 and 24 holding horizon.

# <u>Size coefficient $\gamma_{mv}$ </u>

With equation (4.1), the research conducts monthly-on-monthly regression of equal-(value-) weighted decile portfolio returns on log mean portfolios' market values at the end of formation month. Due to ascending order of share market values in portfolio formation, negative regression coefficient  $\gamma_{mv}$  means that small-cap shares outperform large-cap shares, and vice versa. (In order to be consistent with the sign of Small/Large return  $R_{sml}$ , the research takes negative size coefficient  $\gamma_{mv}$  as dependent variable in series regression in following chapters.) With Fama-Macbeth (1973) approach, the research averages 24 regression coefficient  $\gamma_{mv}$  as one testing period. Then, using Newey-West (1987) approach to adjust standard errors and to calculate t-statistics, the research estimates the number of observations (percentages) with significant small-cap/large-cap effect among all observations. Confidence level for t-test is 95%.

#### [Table 4.11]

Monthly-on-monthly cross-sectional regression of portfolio holding returns on their log mean market values indicates strong rotation between small-cap/large-cap effect when holding horizon is 18 and 24 months. Panel A1 and A2 (Panel B1 and B2) of Table 4.11 shows that the percentage of equal-(value-) weighted portfolios with significant negative and positive average slope coefficient  $\gamma_{mv}$  rises over holding horizons. For the case of 3, 6, 12 months' holding horizon, the percentage of significant negative average size coefficient  $\gamma_{mv}$  is lower than the positive one. The portfolios based on 18 and 24 months' holding horizons have 45.8 and 41.2 (46.7 and 39.9) percentage of observations with significant negative average size coefficient  $\gamma_{mv}$ . The difference in the percentage between significant negative and positive  $\gamma_{mv}$  is not so large that Small/Large return  $R_{sml}$  is unsignificant in the overall sample period as shown in Table 4.10.

In the 1980s equal-(value-) weighted portfolios based on 18 and 24 months' holding horizon have significant negative average size coefficient  $\gamma_{mv}$  in more than 61.2 percentage of observations, and significant positive average  $\gamma_{mv}$  in less than 18.8 percentage of observations. The large difference in percentage with significant negative and positive average  $\gamma_{mv}$  leads to significant 'Small/Large' return  $R_{sml}$  in the period, as showing in Panel A1 (B1) of Table 4.10. The same condition exists in the 2000s. Conversely, in the 1990s equal-(value-) weighted portfolios obtain 38.3 to 62.5 percentage of observations having significant positive average size coefficient  $\gamma_{mv}$  over five holding horizon, and less than 17.5 percentage of observations having significant negative average  $\gamma_{mv}$ . This large difference causes marginally negative 'Small/Large' return  $R_{sml}$  in the period as indicated in Table 4.10 Panel A1 (B1).

In brief, the performance of decile size portfolio returns, Small/Large return  $R_{sml}$ , and size coefficient  $\gamma_{size}$  reflects no existence of obvious size effect in the UK Stock Market from 1979 to 2008. This pattern is mainly caused by the rotation in return premium between small-cap and large-cap portfolio over subperiods. It is consistent with the findings of Levis and Liodakis (1999) that size spread is unconsistent across different subperiods in the UK Stock Market from 1968 to 1997, and that the average annual return for small-cap index performs higher than one for large-cap index before 1988, but conversely thereafter. Similarly, Fama and French (2011) present no size effect in four regions of North America, Europe, Japan, and Asia Pacific from 1991 to 2011. After excluding 'Small' portfolio due to the possibility of its wider bid-ask spread and low trading volume, the research finds the equal-(value-) weighted 2<sup>nd</sup> portfolio with the second smallest market value yields the lowest or second lowest average monthly return, and that equal-(value-) weighted 8<sup>th</sup> portfolio

# 4.4 Value Effect

In this section, the research investigates the relation between shares' price-to-book ratios (PB) at the end of formation month and share returns in following 3, 6, 12, 18, and 24 months, respectively, in the UK Stock Market from 1980 to 2008. It presents empirical analysis on time-varying value effect in different holding horizons in terms of decile portfolio returns, Low/High return  $R_{lmh}$ , and value coefficient  $\gamma_{pb}$ .

### Decile value portfolio returns

The research examines the performance of decile value portfolios in terms of average monthly excess return and the percentage of one portfolio with highest/lowest return among decile portfolios from 1980 to 2008. Portfolio P1 includes shares with the lowest ranking price-to-book ratio (PB) at the end of formation month, and is defined as 'Low' portfolio. Portfolio P10 includes shares with the highest ranking PB at the end of formation month, and is defined as 'High' portfolio. In order to observe the possibility of 'High' portfolio to outperform other portfolios, the research estimates the percentage of one portfolio with highest monthly excess return among decile value portfolios to examine the percentage of high PB premium (growth effect). Similarly, the research estimates the percentage of one portfolio with lowest monthly excess return to find out the underperformance of low PB portfolio over others.

[Table 4.12]

The return performance of value decile portfolios reflects the existence of low PB premium (value effect) in the UK Stock Market. Panel A1 (B1) of Table 4.12 presents the performance of decile value portfolios' average monthly excess returns and standard deviations when portfolios are hold for five holding horizons. It shows that 'Low' portfolio yields the highest average monthly excess return, and 'High' portfolio yields the lowest or second/third lowest return among decile portfolios over different holding horizon. Decile portfolio returns exhibit declining pattern from 'Low' portfolio to 'High' portfolio in which price-to-book ratio increases through portfolios. Average monthly excess return on equal-(value-) weighted 'Low' portfolio increases from 1.17% to 1.35% (1.24% to 1.50%) as holding horizon extends from 3 months to 24 months. This indicates 'Low' portfolio based on 24 months' holding horizon exhibits the strongest low PB premium. With higher standard deviation, 'Low' portfolios, becomes lower as holding horizon extends from short-term to long-term.

The return performance of 'Low' and 'High' portfolios is supported by its percentage of highest and lowest return among decile portfolios in all observations. Panel A2 (B2) of Table 4.12 shows that equal-(value-) weighted 'Low' portfolio with five holding horizons yields the highest return in 32.8 to 37.1 (26.4 to 37.0) percentage of all observations, the highest percentage among decile portfolios. Panel A3 (B3) indicates that equal-(value-) weighted 'Low' portfolio with five holding horizons yields the lowest return in 5.9 to 9.4 (6.5 to 13.2) percentage of all observations. The large spread between two percentages of 'Low' portfolio

contributes to its return premium. Conversely, equal-(value-) weighted 'High' portfolio with five holding horizons earns highest return in only 3.7 to 9.0 (7.9 to 11.9) percentage of all observations, much lower than the percentage of 39.4 to 45.2 (15.2 to 26.9) with lowest return. This causes the worst return performance of 'High' portfolio.

# Low/High return R<sub>1mh</sub>

The research designates Low/High return  $R_{lmh}$  as the difference in average monthly excess return between the bottom (low price-to-book ratio) and top (high price-to-book ratio) portfolios, and investigates its performance in the overall sample period and subperiods for five holding horizons. Panel A1(B1) of Table 4.13 shows that equal-(value-) weighted Low/High return  $R_{lmh}$  in the overall sample period is from 0.97 to 1.25 (0.93 to 1.04) per month when Low/High position is hold for five horizons. Its adjusted t-stats by Newey-West (1987) approach is more than 2.86, which means that Low/High return  $R_{lmh}$  is reliably different from zero.

### [Table 4.13]

*E qual-(value-) weighted Low/High return*  $R_{lmh}$  *descends (ascends) over holding horizons.* Panel A1 (B1) of Table 4.13 indicates the descending equal-weighted Low/High return  $R_{lmh}$  from 1.25% to 0.97% per month, but the ascending value-weighted Low/High return  $R_{lmh}$  from 0.93% to 1.04% per month as holding horizon grows from 3 months to 18 months. This difference is caused by the performance of the return on 'Low' portfolio and 'High' portfolio through holding horizon. For the

equal-weighted case, Panel A1 of Table 4.12 shows that the return on 'Low' portfolio increases less than the return on 'High' portfolio as holding horizon extends from 3 months to 18 months, which leads to the descending return spread between equal-weighted 'Low' portfolio and 'High' portfolio through holding horizon. Conversely, for the value-weighted case, Panel B1 of Table 4.12 shows that the return on 'Low' portfolio increases more than the return on 'High' portfolio as holding horizon grows from 3 months to 18 months, which leads to the ascending return spread between value-weighted 'Low' portfolio and 'High' portfolio as holding horizon.

*Low/High return*  $R_{lmh}$  *keeps positive over subperiods.* Panel A1 (B1) indicates that Low/High returns  $R_{lmh}$  with five holding horizon are significantly positive in overall sample period. What is more striking is that positive Low/High return  $R_{lmh}$  is consistent in three subperiods. Equal-weighted Low/High return  $R_{lmh}$  is (marginally) significantly positive in the 1980s and the 1990s over five holding horizons and in the 2000s over 3 to 12 months' holding horizon. Value-weighted Low/High return  $R_{lmh}$  is significantly positive in the 1980s and the 2000s over all holding horizons, and positive but unreliably different from zero in the 1990s. The persistent good performance of Low/High return  $R_{lmh}$  is supported by the percentage of its positive return in the whole sample period and subperiods. A2 (B2) of Table 4.13 shows that equal-(value-) weighted Low/high return  $R_{lmh}$  is positive in more than 75.2 (60.6) percentage of observations in overall sample period and 64.2 (50.0) in three subperiods.

# *Value coefficient* $\gamma_{pb}$

With equation (4.1), the research conducts monthly-on-monthly regression of equal-(value-) weighted value portfolio returns on mean portfolios' price-to-book values at the end of formation month. Due to ascending order of share's price-to-book ratio in portfolio formation, negative regression coefficient  $\gamma_{pb}$  means that low price-to-book shares outperform high ones, and vice versa. (In order to be consistent with the sign of Low/High return R<sub>lmh</sub>, the research takes negative  $\gamma_{pb}$  as dependent variable in series regression in following chapters.) With Fama-Macbeth (1973) approach, the research averages 24 regression coefficient  $\gamma_{pb}$  as one testing period, uses Newey-West (1987) approach to adjust standard errors and estimates t-statistics, and further finds the number of significant value effect/growth effect and corresponding percentages among all observations. Confidence level for t-test is 95%.

# [Table 4.14]

The monthly-on-monthly cross-sectional regression of portfolio holding returns on their mean price-to-book ratios indicates strong value effect when value portfolios are hold for 24 months. Panel A1 and A2 (Panel B1 and B2) of Table 4.14 shows that the percentage of equal-(value-) weighted portfolios with significantly negative average value coefficient  $\gamma_{pb}$  rises from 59.3 (40.4) to 74.1 (57.1) over five holding horizons in the overall period, and that the percentage having significantly positive  $\gamma_{pb}$  is below 6.5 (13.3). This means that the value effect becomes stronger as holding horizon grows from 3 to 24 months, which explains rising 'Low' portfolio return shown in Table 4.12 Panel A1 (B1).

For equal-weighted value portfolios, the percentage with significantly negative  $\gamma_{pb}$  is high in three subperiods, most of them higher than 50 percent. For valuel-weighted value portfolios, this percentage is higher than 50 percent in the 1980 and in the 2000s, but low in the 1990s. Their patterns are consistent with the performance in corresponding Low/High return R<sub>lmh</sub> over three subperiods. Most equal-weighted Low/High returns are significant in three subperiods, and value-weighted ones are significant in the 1980s and 2000s.

In short, the performance of decile value portfolio returns, Low/High return  $R_{Imh}$ , and value coefficient  $\gamma_{pb}$  reflects strong value effect in the UK Stock Market from 1979 to 2008. Low price-to-book premium rises as holding horizon extends from 3 months to 24 months. The similarly rising pattern is for the percentage of observations with significantly negative value coefficient  $\gamma_{pb}$ . Value effect is significant in three subperiods, except for value-weighted one in the 1990s. This finding is consistent with other studies on value effect in the UK Stock Market. For example, Levis and Liodakis (1999) explores the value/growth performance in the UK Stock Market from 1968 to 1997, and find that positive value/growth return spread persisted for all three subperiods. Fletcher (2007) shows significantly positive excess return spread between high B/M and low B/M portfolios from 1979 to 2005.

# 4.5 The correlation between style effects

Style effects in stock market are not necessarily independent with each other. It is reasonable to consider that some economic factors might lead to the existence of two significant style effects in the market at the same time point/or period, which will be examined in detail in next chapter. Asness (1997) shows that value variable ((BV/MV)) and momentum variable (PAST(2,12)) are positively associated with future expected returns but negatively associated with each other. Value strategy is significantly stronger over loser shares than over winner shares. Momentum strategy is significantly stronger over expensive shares than over cheap shares. Asness (2011) finds negative correlation between momentum effect and value effect in USA, Europe, and Japan Stock Market. In this section, the investigation of the correlation between two style effects is based on two measures, namely Long/Short return  $R_{ims}$  and style coefficient  $\gamma$  estimated in previous three sections. Momentum effect is based on lagged 12 months' share returns.

Table 4.15 reports correlation coefficient among style effects in terms of Long/Short return  $R_{lms}$  in overall sample period and three subperiods. Panel A shows negative correlation in Long/Short return  $R_{lms}$  between momentum effect and size effect, negative correlation in Long/Short return  $R_{lms}$  between momentum effect and value effect, and positive correlation in Long/Short return  $R_{lms}$  between size effect and value effect from 1980 to 2008. This reflects that it is likely momentum effect, large-cap effect, and growth effect to take place in the same period. As shown in Panel B, C, D in Table 4.15, correlation coefficients between momentum effect and size effect,

and between momentum effect and value effect are negative in three subperiods of the 1980s, the 1990s, and the 2000s.

However, Table 4.15 shows the variation in correlation between size effect and value effect over three subperiods. Particularly, correlation coefficients in the equalweighted case with five horizons change from more than positive 0.54 in the 1980s to less than negative 0.36 in the 2000s. This reflects that positive correlation between size effect and value effect is strong in the 1980s, 1990s, but disappears after the 2000. This variation is mainly caused by high-tech bubble that emerged in the market from 1997 to 2000. Figure 4.1 plots the performance of monthly Small/Large return R<sub>sml</sub> and Low/High return R<sub>lmh</sub> from 1981 to 2008 when long/short return is hold for 12 months. It is found that small effect is strong during four periods, such as 1983 to 1988, 1993 to 1994, 1999 to 2000, and 2003 to 2004. Small effect and value effect cover three periods except for the third period. Small-cap shares yielded return premium during 1999 to 2000 when growth shares significantly outperformed value shares due to crazy speculation in technology stocks. If the research excludes the period of 1999 to 2002 when small effect and growth effect simultaneously happened, correlation coefficient between equal-weighted R<sub>sml</sub> and R<sub>lmh</sub> rises from 0.23 to 0.46 in the 1990s, and -0.56 to 0.33 in the 2000s.

The same pattern is found in the correlation in style coefficient  $\gamma$  between two style effects, which is shown in Table 4.16.

[Table 4.15, 4.16], [Figure 4.1]
#### 4.6 Long/Short return $R_{lms}$ and style coefficient $\gamma$

Table 4.17 reports summary statistics on series Long/Short return  $R_{lms}$  and style coefficient  $\gamma$  for three style effects, such as momentum effect, size effect, and value effect. It shows the decline in standard deviation of series Long/Short return  $R_{lms}$  and the increase in standard deviation of style coefficient  $\gamma$  for three style effects as holding horizon grows from 3 months to 24 months. The value of maximum and/or minimum Long/Short return  $R_{lms}$  declines as holding horizon grows, which leads to its lower volatility. The value of maximum and/or minimum style coefficient  $\gamma$  increases as holding horizon grows, which leads to its higher volatility.

#### [Table 4.17]

Panel A, B, and C in Table 4.17 show that Long/Short return  $R_{lms}$  and style coefficient  $\gamma$  of momentum effect and size effect have high correlation coefficient over five holding horizons, ranging from 0.84 to 0.92 and from 0.94 to 0.98, respectively. This means that two indicators could explain similar pattern on the relation between share returns and lagged price performance, market value, respectively. However, Panel D shows that correlation coefficient of Low/High return  $R_{lmh}$  and value coefficient  $\gamma_{pb}$  over five holding horizons is lower, ranging from 0.63 to 0.81. This reflects the difference in two indicators to measure value effect, which could be explained by the percentage of observations for each portfolio

with highest and lowest return among decile value portfolios. Table 4.12 shows that the 2<sup>nd</sup>, 3<sup>rd</sup>, and 4<sup>th</sup> portfolio, besides 'Low' portfolio, yields highest return in higher percentage of observations, and that the 7<sup>th</sup>, 8<sup>th</sup>, and 9<sup>th</sup> portfolio, besides the 'High' portfolio, yields lowest return in higher percentage of observations. Low/High return  $R_{lmh}$  is return spread between the 1<sup>st</sup> and the 10<sup>th</sup> portfolio. It only captures the distribution in highest/lowest return of two portfolios, but ignores its distribution of the 2<sup>nd</sup> to 9<sup>th</sup> portfolio. Correspondingly, value coefficient  $\gamma_{pb}$  is the slope coefficient of cross-sectional regression of value decile portfolios' returns in the holding period on their average values of price-to-book ratio in the formation period. It includes the information of all shares in decile value portfolios, and can capture the distribution in highest/lowest return through them.

#### 4.7 Style effects in business cycles and market conditions

In Section 4.2, 4.3, and 4.4, the research finds the time-variation in style effects in the UK Stock Market, measured by Long/Short return  $R_{lms}$  and style coefficient  $\gamma$ . In this section, the research further investigates the performance of style effects in business cycles and market conditions from 1980 to 2008. Because the previous section shows high or higher correlation between two indicators, the research observes the performance of Long/Short return  $R_{lms}$  in economic expansionary and contractionary periods, and upward and downward markets, respectively. The investigation is based on momentum effect with 12 month's formation horizon, size effect, and value effect when long/short portfolio is hold for 12 months, which is indicated as momentum effect (12/12), size effect (12), and value effect (12).

#### 4.7.1 Long/Short return $R_{lms}$ in business cycles

Time-varying cross-sectional risk premium in momentum, size, and value sorted portfolios might be exposure to business cycles, such as economic expansionary and contractionary periods. The research uses equation (4.2) to calculate the previous annual change of GDP as an indicator of the current economic condition, which is denoted as GDP(Y):

$$GDP(Y)_{t-L} = GDP_{t-L} / GDP_{t-L-4} - 1$$
 (4.2)

where GDP<sub>t-L</sub> is the level of GDP for quarter t-L. 'L' stands for lag length of two quarters to ensure GDP known as economic indicator in advance in the decisionmaking on allocation in Winner/Loser, Small/Large-cap, and Low/High portfolios. GDP is collected from ESDS IFS (The Economic and Social Data Service, International Financial Statistics). The research classifies series GDP(Y) into positive and negative groups, and regards them as economic expansionary and contractionary periods, respectively.

#### [Table 4.18]

Table 4.18 shows the performance of Long/Short return  $R_{lms}$  of style effects in the period with positive and negative GDP(Y) from 1980 to 2008. For momentum effect (12/12), average Winner/Loser return  $R_{wml}$  is negative 0.11% (0.14%) per month and positive in 60.9 (54.2) percent of observations in the period with positive GDP(Y) when equal-(value-) weighted Winner/Loser portfolio is hold for 12 months. Average  $R_{wml}$  is significantly positive 1.40% (1.23%) per month and positive in 87.7 (76.7) percent of observations in the period with positive GDP(Y), which are obviously higher than corresponding values in the period with positive GDP(Y). This means that momentum effect is significant and stronger in the declining annual change of GDP, but disappears in the increasing annual change of GDP.

For size effect (12), average Small/Large return  $R_{sml}$  is significantly positive 0.68% (0.52%) per month and positive in 52.5 (56.9) percent of observations in the period with positive GDP(Y), but negative 0.07% (0.14%) per month and positive in 41.8

(42.4) percent of observations in the period with negative GDP(Y) when equal-(value-) weighted Small/Large portfolio is hold for 12 months. This reflects significant small-cap effect in the increasing annual change of GDP, but no size effect in the declining annual change of GDP.

For value effect (12), there is no obvious difference in the performance of Low/High return  $R_{lmh}$  in the period with positive and negative GDP(Y) when Low/High portfolio is hold for 12 months. Equal-(value-) weighted Low/High portfolio yields significantly positive monthly  $R_{lmh}$  with small difference of 0.18% (0.24%) no matter GDP(Y) rises or falls.  $R_{lmh}$  is positive in almost same percentage of observations in two economic conditions.

In summary, momentum effect and size effect have different performance in business cycles as measured by GDP(Y). Small-cap (Loser) shares outperform large-cap (Winner) shares in economic expansionary period, and vice versa. However, value effect seems not to be associated with economic conditions.

#### 4.7.2 Long/Short return $R_{lms}$ in market conditions

The time-variation in style effects might be associated with dynamic market states. The research groups one-year's market return into positive and negative category, and observes the performance of Long/Short return  $R_{lms}$  of style effects with 12 months' holding horizon in lagged upward market and downward market from 1980

to 2008. Table 4.19 reports the performance of Long/Short return  $R_{lms}$  of style effects in the period with positive and negative lagged one-year's market return, which is denoted as MR(-1). MR(-1) is calculated by monthly FTSE ALL SHARE PRICE from LSPD.

#### [Table 4.19]

When Long/Short portfolio is hold for 12 months, average monthly Winner/Loser return  $R_{wml}$  and Small/Large return  $R_{sml}$  are positive in lagged one-year's upward market and downward market. Equal-(Value-) weighted Winner/Loser return  $R_{wml}$  is average 0.10% (0.30%) per month in the rising market lower than in the declining market, and positive in almost same percentage of the observations in two market conditions. Small/Large return  $R_{wml}$  has similar pattern no matter whether the market rises or falls in past one year. For value effect (12), equal-(value-) weighted Low/High portfolio yields significantly positive Low/High return  $R_{lmh}$  in either upward or downward market in past one year. It earns average monthly return of 0.35% (0.46%) in upward market lower than in downward market, but has almost same percentage of observations with positive return in two market conditions.

In summary, momentum effect, size effect, and value effect seems not to be related to lagged short-term market conditions. Long/Short return  $R_{lms}$  of three style effects do not have obvious difference in either magnitude or the percentage of observations with positive return when the market rises or falls in past one year.

#### 4.8 Conclusion

In this chapter, the research uses two indicators, such as long/short return  $R_{lms}$  and style coefficient  $\gamma$ , to investigate the performance of time-varying style effects in the UK Stock Market. It classifies firm characteristics into lagged share return, market value, and price-to-book ratio, and examines the relation between three firm characteristics and share returns with 3, 6, 12, 18, and 24 months' holding horizon, respectively, in the overall sample period and subperiods.

#### • Momentum effect

When momentum portfolios are formed on lagged 6 and 12 months' share returns and hold for five holding horizons with 3, 6, 12, 18, and 24 months, there are common patterns in holding returns. Firstly, the performance of decile momentum portfolios' average monthly excess returns, Winner/Loser return  $R_{wml}$ , and momentum coefficient  $\gamma_{mom}$  reflects the existence of momentum effect in the UK Stock Market from 1956 to 2008. Secondly, when portfolios are hold for 6 and 12 months, Winner/Loser return  $R_{wml}$  are highest or second/third highest among the cases with five holding horizons, and are positive in the highest or second highest percentage of all observations. Meanwhile, their momentum coefficient  $\gamma_{mom}$ , which reflects information of decile momentum portfolios, is significantly positive in the highest and second highest percentage of all observations. These reflect that momentum effect based on 6 and 12 months' holding horizons is strong. As holding horizon grows more than 12 months, momentum effect becomes weaker and unsignificant. Thirdly, Winner/Loser return  $R_{wml}$  and momentum coefficient  $\gamma_{mom}$  exhibit the variation over time. Their patterns show that momentum effect is strongest in the 1990s, but weakest in the 1970s. Fourthly, following 'Winner' portfolio, 'Loser' portfolio has higher percentage to yield highest return among decile momentum portfolio. It is not the case that 'Winner/Loser' return  $R_{wml}$  is nearly always positive. It is positive in 53-71 percent and negative in 29-47 percent of observations with five holding horizons. This pattern motivates the investigation on what factors lead to the time-varying momentum effect, and the exploration to capture the rotation between 'Winner' and 'Loser' portfolios by using forecasting model, which is detailed in next chapters.

#### • Size Effect

The empirical work indicates no existence of obvious size effect in the UK Stock Market from 1979 to 2008 due to size rotation in three subperiods. Firstly, 'Small' portfolio yields lowest return among decile size portfolios in higher percentage of all observations while it outperforms other portfolios in high percentage. Its high volatility leads to no obvious small-cap premium in the overall sample period. The similar pattern exists in 'Large' portfolio. Secondly, there is rotation between smallcap effect and large-cap effect over three subperiods. Small/Large return  $R_{sml}$  is (marginally) significant positive in the 1980s, and marginally significant negative in the 1990s, and positive afterwards. Thirdly, this size rotation is consistent with the performance of the variation in the percentage of observations with significantly positive and negative size coefficient  $\gamma_{mv}$  over three subperiods. Given higher volatility in return of 'Small' and 'Large' portfolio, it is instructive to examine what factors explain the volatility, and whether the forecasting model times the rotation in size effect, which is analyzed in next chapters.

#### • Value Effect

The empirical work on value effect with the measurement of shares' price-to-book ratios in the UK Stock Market from 1980 to 2008 indicates, firstly, 'Low' portfolio and 'High' portfolio are much likely to yield highest return and lowest return than other portfolios, respectively. Their opposite pattern supports significantly positive Low/High return  $R_{lmh}$  as well as low price-to-book return premium. Secondly, value effect based on 24 months' holding horizon is the strongest as compared to ones with short or medium horizons, which is supported by highest return of 'Low' portfolio and highest percentage of observations with significantly negative value coefficient  $\gamma_{pb}$ . Thirdly, the performance of Low/High return  $R_{lmh}$  and value coefficient  $\gamma_{pb}$  reflects persistent value effect in three subperiods. Given invariant value effect over time as well as high (low) percentage of 'Low' portfolio with highest (lowest) return among decele portfolios, it is less likely to succeed to time the rotation in value effect by using forecasting model, which is discussed in Chapter 7.

#### • The correlation between two style effects

The empirical work finds negative correlation between momentum effect and size effect, between momentum effect and value effect from 1980 to 2008 as well as three subperiods of the 1980s, the 1990s, and the 2000 afterwards. There exists the variation in correlation between size effect and value effect over time. Strong positive correlation in the 1980s, but disappears after the 2000, namely strong

negative in the equal-weighted case and weak positive in the value-weighted case due to high-tech bubble from 1997 to 2000 when small-cap effect and growth effect emerged.

#### • Style effects in business cycles and market conditions

Momentum effect and size effect have different performance in business cycles as measured by GDP(Y), annul change of GDP. Small-cap (Loser) shares outperform large-cap (Winner) shares when economic expansion accelerates or economic contraction slows down, and vice versa. However, value shares and growth shares have the same return performance no matter how economic conditions change. Meanwhile, it is found that three style effects seem not to be related to lagged oneyear's market performance.

### Appendix: tables and figures in Chapter 4

				Panel A Eq	ual-weighte	d portfolio					
		,	A1 Average	e monthly ex	cess return	and standar	d deviati	ion			
Holding months	Loser		D3	P/	P5	P6	P7	D8	PQ	Winner	<b>P</b> .
3	1 17	0.62	0.62	0.63	0.71	0.76	0.82	0.88	1 00	1 19	0.02
5	(4.99)	(4.17)	(3.80)	(3.60)	(3.46)	(3.34)	(3.30)	(3.31)	(3.47)	(4.05)	(3.51)
6	0.82	0.63	0.73	0.76	0.83	0.90	0.98	1.02	1.15	1.36	0.54
	(3.91)	(3.28)	(3.03)	(2.85)	(2.77)	(2.68)	(2.63)	(2.67)	(2.78)	(3.21)	(2.35)
12	0.94	0.83	0.87	0.92	1.00	1.04	1.09	1.15	1.28	1.50	0.56
	(3.29)	(2.63)	(2.43)	(2.29)	(2.20)	(2.12)	(2.06)	(2.12)	(2.20)	(2.45)	(1.83)
18	1.21	1.09	1.10	1.12	1.14	1.15	1.18	1.20	1.27	1.41	0.20
	(2.76)	(2.31)	(2.17)	(2.05)	(1.93)	(1.87)	(1.86)	(1.90)	(1.93)	(2.16)	(1.62)
24	1.31	1.19	1.18	1.25	1.25	1.21	1.26	1.26	1.31	1.43	0.12
	(2.43)	(2.18)	(1.90)	(1.90)	(1.87)	(1.70)	(1.73)	(1.76)	(1.77)	(1.99)	(1.59)
		A2 P	ercentage o	f observatio	ns for each	portfolio wi	th highe	st return			
Holding months	Loser	P2	P3	P4	P5	P6	th highes	P7	P8	P9	Winner
3	26.9	8.8	5.6	4.5	3.6	3.3		7.5	8.5	10.6	20.7
6	20.9	5.8	5.2	4.7	4.7	4.2		7.1	8.6	13.5	25.2
12	16.0	6.3	5.1	4.1	5.4	4.8		6.3	7.1	13.5	31.3
18	19.4	6.3	7.4	5.8	7.5	5.4		6.9	7.5	12.0	21.8
24	22.7	6.1	7.8	6.0	8.1	6.0		6.3	6.6	11.8	18.6
		A3 F	Percentage of	of observatio	ons for each	portfolio wi	th lowes	st return			
Holding months	Loser	P2	P3	P4	P5	P6		P7	P8	P9	Winner
3	20.2	16.4	11.0	8.9	6.6	7.0		6.3	4.5	8.0	11.1
6	32.1	14.5	9.3	7.9	6.1	6.8		4.6	4.9	5.2	8.8
12	35.7	14.0	9.7	7.5	6.7	5.1		4.9	5.6	4.3	6.7
18	31.3	13.3	8.8	6.9	5.9	6.6		5.3	5.4	5.4	11.1
24	29.8	13.8	7.1	6.1	6.5	6.3		6.1	6.3	6.0	12.0
27	27.0	15.0	/11	0.1	0.0	0.0		011	0.0		12.0
27	27.0	15.0	,,,,	Domal D. Va	lua maiahta	d montfolio		011	0.0		12.0
	27.0	10.0	D1 A	Panel B Va	lue-weighte	d portfolio					12.0
	27.0	]	B1 Average	Panel B Va e monthly ex	lue-weighte	d portfolio and standar	d deviati	ion	DO		
Holding months	Loser	P2	B1 Average P3	Panel B Va e monthly ex P4	lue-weighte ccess return P5	ed portfolio and standar P6	d deviati	ion P8	P9	Winner	R <sub>wml</sub>
Holding months 3	Loser 0.08	P2 0.25 (4 (7))	B1 Average P3 0.37	Panel B Va e monthly ex P4 0.40	lue-weighte ccess return P5 0.48	ed portfolio and standar P6 0.53 (2.51)	d deviati P7 0.60	ion P8 0.66	P9 0.69	Winner 0.84	R <sub>wml</sub> 0.76
Holding months 3	Loser 0.08 (5.57)	P2 0.25 (4.67) 0.26	B1 Average P3 0.37 (4.12) 0.44	Panel B Va e monthly ex P4 0.40 (3.91) 0.48	lue-weighte ccess return P5 0.48 (3.76) 0.50	d portfolio and standar P6 0.53 (3.51) 0.62	d deviati P7 0.60 (3.46)	ion P8 0.66 (3.62) 0.60	P9 0.69 (3.57) 0.70	Winner 0.84 (3.82)	R <sub>wml</sub> 0.76 (4.87)
Holding months 3 6	Loser 0.08 (5.57) -0.04 (3.89)	P2 0.25 (4.67) 0.26 (3.25)	B1 Average P3 0.37 (4.12) 0.44 (3.02)	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81)	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76)	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60)	d deviati P7 0.60 (3.46) 0.66 (2.65)	ion P8 0.66 (3.62) 0.69 (2.64)	P9 0.69 (3.57) 0.79 (2.67)	Winner 0.84 (3.82) 0.90 (2.84)	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.10)
Holding months 3 6	Loser 0.08 (5.57) -0.04 (3.89) 0.15	P2 0.25 (4.67) 0.26 (3.25) 0.35	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74	ion P8 0.66 (3.62) 0.69 (2.64) 0.73	P9 0.69 (3.57) 0.79 (2.67) 0.89	Winner 0.84 (3.82) 0.90 (2.84) 0.97	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82
Holding months 3 6 12	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86)	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35)	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21)	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11)	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94)	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95)	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94)	P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89)	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94)	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12)	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35)
Holding months 3 6 12 18	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76	P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34
Holding months 3 6 12 18	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32)	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92)	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80)	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67)	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61)	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62)	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65)	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64)	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64)	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83)	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98)
Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73	lue-weighte <u>xcess return</u> P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81	P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31
Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85)	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55)	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47)	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45)	lue-weighte <u>xcess return</u> P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40)	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40)	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45)	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44)	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49)	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62)	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64)
Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85)	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P2	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40)	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highesit	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49)	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62)	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64)
Holding months 3 6 12 18 24 Holding months	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85)	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) Percentage o P3	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62)	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner
Holding months 3 6 12 18 24 Holding months 3	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8
Holding months 3 6 12 18 24 Holding months 3 6	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4 5.0	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highes	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7
Holding months 3 6 12 18 24 Holding months 3 6 12	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4 5.0 5.2	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0
Holding months 3 6 12 18 24 Holding months 3 6 12 18	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4 5.0 5.2 6.1	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0 8.4	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4 5.0 5.2 6.1 5.0	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Lagrandow Lagrandow Lagrando	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2 B3 F	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0 8.4 Percentage o	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5	$\begin{array}{c} \text{lue-weighte} \\ \text{ccess return} \\ \hline \text{P5} \\ 0.48 \\ (3.76) \\ 0.50 \\ (2.76) \\ 0.56 \\ (1.94) \\ 0.65 \\ (1.61) \\ 0.68 \\ (1.40) \\ \hline \text{ns for each} \\ \hline \text{P5} \\ 6.4 \\ 5.0 \\ 5.2 \\ 6.1 \\ 5.0 \\ \hline \text{ons for each} \\ \end{array}$	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8 portfolio wi	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest th lowest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9 st return	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7 Loser	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2 B3 F P2	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0 8.4 Percentage o P3	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5 of observatio P4	$\begin{array}{c} \text{lue-weighte} \\ \text{ccess return} \\ \hline \text{P5} \\ 0.48 \\ (3.76) \\ 0.50 \\ (2.76) \\ 0.56 \\ (1.94) \\ 0.65 \\ (1.61) \\ 0.68 \\ (1.40) \\ \hline \text{ns for each} \\ \hline \text{P5} \\ 6.4 \\ 5.0 \\ 5.2 \\ 6.1 \\ 5.0 \\ \hline \text{ons for each} \\ \hline \text{P5} \end{array}$	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8 portfolio wi P6	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest th lowest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9 st return P7	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1 P8	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2 P9	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2 Winner
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 4 Holding months 3	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7 Loser 27.2	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2 B3 F P2 10.6	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0 8.4 Percentage o P3 8.9	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5 of observatio P4 6.9	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4 5.0 5.2 6.1 5.0 ms for each P5 6.7	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8 portfolio wi P6 6.4	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest th lowest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9 st return P7 6.3	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1 P8 7.2	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2 P9 8.1	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2 Winner 11.6
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7 Loser 27.2 31.8	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2 B3 F P2 10.6 12.7	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0 8.4 Percentage o P3 8.9 7.4	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5 of observatio P4 6.9 7.2	lue-weighte ccess return P5 0.48 (3.76) 0.50 (2.76) 0.56 (1.94) 0.65 (1.61) 0.68 (1.40) ns for each P5 6.4 5.0 5.2 6.1 5.0 ons for each P5 6.7 6.7	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8 portfolio wi P6 6.4 6.8	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest th lowes	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9 st return P7 6.3 6.0	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1 P8 7.2 5.7	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2 P9 8.1 5.5	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2 Winner 11.6 11.6
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7 Loser 27.2 31.8 36.8	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2 B3 F P2 10.6 12.7 11.1	B1 Average P3 0.37 (4.12) 0.44 (3.02) 0.49 (2.21) 0.67 (1.80) 0.71 (1.47) ercentage o P3 8.6 7.4 7.0 8.0 8.4 Percentage o P3 8.9 7.4 7.6	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5 of observatio P4 6.9 7.2 5.9	$\begin{array}{c} \text{lue-weighte} \\ \hline \text{ccess return} \\ P5 \\ 0.48 \\ (3.76) \\ 0.50 \\ (2.76) \\ 0.56 \\ (1.94) \\ 0.65 \\ (1.61) \\ 0.68 \\ (1.40) \\ \hline \text{ns for each} \\ \hline P5 \\ 6.4 \\ 5.0 \\ 5.2 \\ 6.1 \\ 5.0 \\ \hline \text{ons for each} \\ P5 \\ 6.7 \\ 6.1 \\ 6.5 \\ \end{array}$	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8 portfolio wi P6 6.4 6.8	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest th lowest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9 8.1 8.8 7.9 st return P7 6.3 6.0 6.0 6.0	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1 P8 7.2 5.7 5.2	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2 P9 8.1 5.5 6.3	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2 Winner 11.6 11.6 8.6
Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24 Holding months 3 6 12 18 24	Loser 0.08 (5.57) -0.04 (3.89) 0.15 (2.86) 0.49 (2.32) 0.54 (1.85) Loser 17.8 14.2 12.5 15.1 14.7 Loser 27.2 31.8 36.8 30.9	P2 0.25 (4.67) 0.26 (3.25) 0.35 (2.35) 0.58 (1.92) 0.65 (1.55) B2 P P2 11.3 8.8 6.5 8.5 9.2 B3 F P2 10.6 12.7 11.1 11.4	B1         Average           P3         0.37           (4.12)         0.44           (3.02)         0.49           (2.21)         0.67           (1.80)         0.71           (1.47)            Percentage o         P3           8.6         7.4           7.0         8.0           8.4            Percentage o         P3           8.9         7.4           7.6         8.3	Panel B Va e monthly ex P4 0.40 (3.91) 0.48 (2.81) 0.53 (2.11) 0.68 (1.67) 0.73 (1.45) f observatio P4 6.6 7.7 5.4 5.1 6.5 of observatio P4 6.9 7.2 5.9 3.0	$\begin{array}{c} \text{lue-weighte} \\ \hline \text{ccess return} \\ P5 \\ 0.48 \\ (3.76) \\ 0.50 \\ (2.76) \\ 0.56 \\ (1.94) \\ 0.65 \\ (1.61) \\ 0.68 \\ (1.40) \\ \hline \text{ns for each} \\ \hline P5 \\ 6.4 \\ 5.0 \\ 5.2 \\ 6.1 \\ 5.0 \\ \hline \text{ns for each} \\ \hline P5 \\ 6.7 \\ 6.1 \\ 6.5 \\ 6.7 \\ 6.1 \\ 6.5 \\ 6.7 \\ 6.1 \\ 6.5 \\ 6.7 \\ 6.1 \\ 6.5 \\ 6.7 \\ \hline \end{array}$	d portfolio and standar P6 0.53 (3.51) 0.62 (2.60) 0.66 (1.95) 0.71 (1.62) 0.74 (1.40) portfolio wi P6 6.4 6.9 7.0 6.4 6.8 portfolio wi P6 6.4 6.8 portfolio wi P6 6.4 6.9 7.0 6.4 6.8	d deviati P7 0.60 (3.46) 0.66 (2.65) 0.74 (1.94) 0.76 (1.65) 0.81 (1.45) th highest th lowest	ion P8 0.66 (3.62) 0.69 (2.64) 0.73 (1.89) 0.77 (1.64) 0.80 (1.44) 0.80 (1.44) st return P7 6.3 7.9 8.1 8.8 7.9 8.1 8.8 7.9 st return P7 6.3 6.0 6.0 6.0 6.9	P9 0.69 (3.57) 0.79 (2.67) 0.89 (1.94) 0.83 (1.64) 0.87 (1.49) P8 6.6 7.5 7.1 8.7 8.1 P8 7.2 5.7 5.2 6.4	Winner 0.84 (3.82) 0.90 (2.84) 0.97 (2.12) 0.83 (1.83) 0.85 (1.62) P9 12.2 12.9 17.1 14.7 16.2 P9 8.1 5.5 6.3 7.5	R <sub>wml</sub> 0.76 (4.87) 0.94 (3.19) 0.82 (2.35) 0.34 (1.98) 0.31 (1.64) Winner 17.8 21.7 24.0 18.6 17.2 Winner 11.6 11.6 8.6 12.8

Table 4.1 Statistics on decile momentum portfolios based on 6 months' formation length

Note: The table reports statistics on decile momentum portfolios based on ranking lagged 6 months' share returns from the lowest to the highest and 3, 6, 12, 18, and 24 months' holding horizons from 1956 to 2008. The research designates 'Winner' and 'Loser' as the top (P10) and bottom (P1) portfolios. Winner/Loser return  $R_{wml}$  is estimated by the spread in average monthly excess return between 'Winner' portfolio and 'Loser' portfolio. Panel A1 shows equal-weighted decile momentum portfolios' average monthly excess returns, with standard deviation in parenthesis. Panel A2 and A3 show the percentage of equal-weighted portfolio with highest return and lowest return among decile portfolios, respectively, over the sample period. Panel B shows the corresponding statistics on value-weighted decile momentum portfolios.

			Panel A Ec	qual-weighted p	ortfolio			
		A1 Aver	age monthly V	Vinner/Loser re	turn R <sub>wml</sub> and t-	-stats		
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	639	-1.33	-0.51	-1.58	0.59	1.11	1.21	0.02
		(-4.87)	(-2.37)	(-2.92)	(1.70)	(2.53)	(1.78)	(0.12)
6	636	-0.66	0.15	-0.65	0.90	1.53	1.34	0.54
		(-2.72)	(0.77)	(-1.65)	(2.69)	(3.94)	(2.32)	(2.97)
12	630	-0.05	0.33	-0.11	0.61	1.46	0.73	0.56
		(-0.22)	(1.30)	(-0.36)	(1.65)	(4.50)	(1.24)	(3.17)
18	624	0.19	-0.04	-0.31	0.07	1.02	0.31	0.20
		(1.16)	(-0.12)	(-1.23)	(0.22)	(3.25)	(0.69)	(1.32)
24	612	0.79	-0.26	-0.65	0.08	1.24	0.24	0.12
		(4.33)	(-1.16)	(-2.59)	(0.24)	(4.05)	(0.52)	(0.89)
	A2 The r	number (percentage) of	positive avera	age monthly Wi	nner/Loser retu	rn R <sub>wml</sub> (mome	entum effect)	
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	639	13	48	44	76	85	75	341
		(25.5)	(40.0)	(36.7)	(63.3)	(70.8)	(69.4)	(53.4)
6	636	13	71	53	86	103	81	407
		(27.1)	(59.2)	(44.2)	(71.7)	(85.8)	(75.0)	(64.0)
12	630	24	77	67	85	106	83	442
		(57.1)	(64.2)	(55.8)	(70.8)	(88.3)	(76.9)	(70.2)
18	624	22	57	53	76	108	71	387
		(61.1)	(47.5)	(44.2)	(63.3)	(90.0)	(65.7)	(62.0)
24	618	25	53	44	66	108	66	362
		(83.3)	(44.2)	(36.7)	(55.0)	(90.0)	(61.1)	(58.6)
	A3 The	number (percentage) o	of negative ave	erage monthly V	Winner/Loser re	turn R <sub>wml</sub> (rev	ersal effect)	
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	639	38	72	76	44	35	33	298
		(74.5)	(60.0)	(63.3)	(36.7)	(29.2)	(30.6)	(46.6)
6	636	35	49	67	34	17	27	229
		(72.9)	(40.8)	(55.8)	(28.3)	(14.2)	(25.0)	(36.0)
12	630	18	43	53	35	14	25	188
		(42.9)	(35.8)	(44.2)	(29.2)	(11.7)	(23.1)	(29.8)
18	624	14	63	67	44	12	37	237
		(38.9)	(52.5)	(55.8)	(36.7)	(10.0)	(34.3)	(38.0)
24	612	5	67	76	54	12	42	256
		(16.7)	(55.8)	(63.3)	(45.0)	(10.0)	(38.9)	(41.4)
			Panel B Va	alue-weighted p	orttolio			
		B1 Avera	age monthly W	/inner/Loser ret	turn R <sub>wml</sub> and t-	stats		
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	639	-0.28	0.31	-0.71	0.85	1.41	2.56	0.76
		(-0.58)	(1.12)	(-1.04)	(1.96)	(2.21)	(2.75)	(2.82)
6	636	0.14	0.69	0.03	0.94	1.68	1.78	0.94
		(0.33)	(2.64)	(0.06)	(2.60)	(2.86)	(2.21)	(4.17)
12	630	1.26	0.88	0.08	0.73	1.38	0.90	0.82
		(2.53)	(3.33)	(0.23)	(2.35)	(3.72)	(1.14)	(4.19)
18	624	1.16	0.16	0.01	0.17	0.84	0.29	0.34
		(1.81)	(0.75)	(0.05)	(0.55)	(2.31)	(0.49)	(2.12)
		(1.01)	()	· · ·	· /	. ,		· /
24	618	0.72	0.29	0.04	0.21	1.02	0.42	0.31

Table 4.2 Statistics on Winner/Loser return  $R_{\rm wml}$  based on 6 months' formation length

#### Continued

B2 The number (percentage) of positive average monthly Winner/Loser return R<sub>wml</sub> (momentum effect)

Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	639	26	65	68	75	76	73	383
		(5.1)	(54.2)	(56.7)	(62.5)	(63.3)	(67.6)	(59.9)
6	636	22	77	65	79	86	78	407
		(45.8)	(64.2)	(54.2)	(65.8)	(71.7)	(72.2)	(64.0)
12	630	28	86	76	85	93	78	446
		(66.7)	(71.7)	(63.3)	(70.8)	(77.5)	(72.2)	(70.8)
18	624	27	66	65	76	87	67	388
		(75.0)	(55.0)	(54.2)	(63.3)	(72.5)	(62.0)	(62.2)
24	618	20	66	66	66	92	69	379
		(66.7)	(55.0)	(55.0)	(55.0)	(76.7)	(63.9)	(61.3)

B3 The number (percentage) of negative average monthly Winner/Loser return R<sub>wml</sub> (reversal effect)

Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	639	25	55	52	45	44	35	256
		(49.0)	(45.8)	(43.3)	(37.5)	(36.7)	(32.4)	(40.1)
6	630	26	43	55	41	34	30	229
		(54.2)	(35.8)	(45.8)	(34.2)	(28.3)	(27.8)	(36.0)
12	630	14	34	44	35	27	30	184
		(33.3)	(28.3)	(36.7)	(29.2)	(22.5)	(27.8)	(29.2)
18	624	9	54	55	44	33	41	236
		(25.0)	(45.0)	(45.8)	(36.7)	(27.5)	(38.0)	(37.8)
24	618	10	54	54	54	28	39	239
		(33.3)	(45.0)	(45.0)	(45.0)	(23.3)	(36.1)	(38.7)

Note: The table reports statistics on Winner/Loser return  $R_{wml}$ , the spread in average monthly excess return between top and bottom momentum portfolios which are formed on previous 6 months' share returns and hold for 3, 6, 12, 18, and 24 months from 1956 to 2008. Panel A1 shows equal-weighted Winner/Loser return  $R_{wml}$  and its t-statistics in parenthesis in overall period and subperiods. t-statistics is adjusted by Newey-West (1987) approach. Lag length is holding horizon minus one. Panel A2 and A3 show the number (percentage) of positive and negative equal-weighted Winner/Loser return  $R_{wml}$ , respectively, in overall period and subperiods. Panel B shows the corresponding statistics on value-weighted Winner/Loser return  $R_{wml}$ .

$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$				Panel A Eq	ual-weighted po	ortfolio			
			A1 The number (per	centage) of sig	gnificant positiv	e momentum co	efficient $\gamma_{mom}$	1	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Holding months	Observation Number	Sample beginning -1959	1960s	1970s	1980s	1990s	2000-2008	Overall period
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	3	616	0	0	0	42	81	66	189
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	5	010	(0 <sup>°</sup> 0)	(0,0)	(0,0)	(35.0)	(67.5)	(61.1)	(30.7)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	6	613	0	(0.0)	5	57	85	63	217
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Ū.	010	(0,0)	(58)	(42)	(47.5)	(70.8)	(58.4)	(35.4)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	12	607	0	57	15	60	88	45	265
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		007	(0,0)	(47.5)	(12.5)	(50.0)	(73.3)	(41.7)	(43.7)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	18	601	3	21	10	55	81	41	211
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			(23.1)	(17.5)	(8.3)	(45.8)	(67.5)	(38.0)	(35.1)
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	24	595	7	21	4	38	77	45	192
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$			(100.0)	(17.5)	(3.3)	(31.7)	(64.2)	(41.7)	(32.3)
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$			. ,	. ,		. ,		. ,	
			A2 The number (per	centage) of sig	nificant negativ	e momentum co	befficient $\gamma_{mon}$	n	
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	months	Number	-1959						period
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	3	616	26	23	6	0	0	0	55
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$			(92.9)	(19.2)	(5.0)	(0.0)	(0.0)	(0.0)	(8.9)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	6	613	1	0	17	0	0	0	18
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			(4.0)	(0.0)	(13.3)	(0.0)	(0.0)	(0.0)	(2.9)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	12	607	0	0	16	0	0	0	16
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			(0.0)	(0.0)	(14.2)	(0.0)	(0.0)	(0.0)	(2.6)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	18	601	0	2	39	8	0	4	53
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			(0.0)	(1.7)	(32.5)	(6.7)	(0.0)	(3.7)	(8.8)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	24	595	0	10	49	6	1	9	75
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$			(0.0)	(8.3)	(40.8)	(5.0)	(0.8)	(8.3)	(12.6)
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$									
B1 The number (percentage) of significant positive momentum coefficient $\gamma_{mom}$ Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-2008         Overall period           3         616         0         25         12         34         36         39         146           (0.0)         (20.8)         (10.0)         (28.3)         (30.0)         (36.1)         (23.7)           6         613         2         67         29         53         65         45         261           (8.0)         (55.8)         (24.2)         (44.2)         (54.2)         (41.7)         (42.6)           12         607         1         58         36         56         69         42         262           (5.3)         (48.3)         (30.0)         (46.7)         (57.5)         (38.9)         (43.2)           18         601         0         61         36         28         63         39         227           (0.0)         (50.8)         (30.0)         (23.3)         (52.5)         (36.1)         (37.8)           24         595         0         53         22				Panel B Va	lue-weighted p	ortfolio	6 <b>6</b>		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$			BI The number (per	centage) of sig	gnificant positiv	e momentum co	efficient $\gamma_{mom}$	1	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	months	Number	-1959					•	period
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	3	616	0	25	12	34	36	39	146
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	-	(10)	(0.0)	(20.8)	(10.0)	(28.3)	(30.0)	(36.1)	(23.7)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	6	613	2	67	29	53	65	45	261
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	10	<0 <b>7</b>	(8.0)	(55.8)	(24.2)	(44.2)	(54.2)	(41.7)	(42.6)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	12	607	1	58	36	56	69	42	262
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	10	<b>601</b>	(5.3)	(48.3)	(30.0)	(46.7)	(57.5)	(38.9)	(43.2)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	18	601	0	61	36	28	63	39	227
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	24	505	(0.0)	(50.8)	(30.0)	(23.3)	(52.5)	(36.1)	(37.8)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	24	595	0	53	22	33	68	45	221
B2 The number (percentage) of significant negative momentum coefficient γ <sub>mom</sub> Holding       Observation       Sample beginning       1960s       1970s       1980s       1990s       2000-2008       Overall period         months       Number       -1959       -1959       -1970s       1980s       1990s       2000-2008       Overall			(0.0)	(44.2)	(18.3)	(27.5)	(56.7)	(41.7)	(37.1)
HoldingObservationSample beginning1960s1970s1980s1990s2000-2008OverallmonthsNumber-1959period			B2 The number (per	centage) of sig	nificant negativ	e momentum co	oefficient V		
months Number -1959 period	Holding	Observation	Sample beginning	19600	1970c	1980	1990s	2000-2008	Overall
nonuis Number -1757 period	months	Number		17003	17705	17003	17705	2000-2008	period
	3	616	-1757	0	1	0	Ο	Ο	1
	5	010	(0,0)	(0,0)	(0.8)	(0,0)	(0,0)	(0,0)	(02)
6  613  0  0  0  0  0  0  0  0  0	6	613	0	0	0	0	0	0	0
	0	015	(0,0)	(n) m	ທັກ	ທັກ	00	(0,0)	ທັກ
	12	607	0	0	0	0	0	1	1
(0.0) $(0.0)$ $(0.0$	1 2	007	(0,0)	<u>(0</u> )	<u>(0</u> )	00	00	(0,9)	$(0^{2})$
18  601  0  0  10  2  11  6  29	18	601	0	0	10	2	11	6	29
(0.0)  (0.0)  (8.3)  (1.7)  (9.2)  (5.6)  (4.8)	10	001	<u>(0,0)</u>	<u>(0,0)</u>	(83)	(1.7)	(9.2)	(5 6)	(4.8)
24 595 0 0 20 11 8 10 49	24	595	0	0	20	11	8	10	49
(0.0)  (0.0)  (16.7)  (9.2)  (6.7)  (9.3)  (8.2)			(0.0)	(0.0)	(16.7)	(9.2)	(6.7)	(9.3)	(8.2)

Table 4.3 Statistics on momentum coefficient  $\gamma_{mom}$  based on 6 months' formation length

Note: The table reports statistics on momentum coefficient  $\gamma_{mom}$ , estimated by regressing decile momentum portfolios' holding returns on their formation returns. Decile momentum portfolios are formed on previous 6 months' shares returns and hold for 3, 6, 12, 18, and 24 months from 1956 to 2008. The research averages 24 regression coefficient  $\gamma_{mom}$  as one testing period, uses Newey-West (1987) approach to adjust standard errors and estimates t-statistics. Lag length is holding horizon minus one. The confidence level for t-test is 95%. Panel A1 and A2 show number (percentage) of significant positive and negative momentum coefficient  $\gamma_{mom}$  for equal-weighted momentum portfolios in overall sample period and subperiods. Panel B shows the corresponding statistics on value-weighted momentum coefficient  $\gamma_{mom}$ .

Table 4.4 Statistics on decile momentum portfolios based on 12 months' formation length

				Panel A Ec	qual-weighte	ed portfolio					
			A1 Average	e monthly e	xcess return	and standa	rd devia	tion			
Holding months	Loser	P2	P3	P4	P5	P6	P7	P8	P9	Winner	R <sub>wml</sub>
3	0.97	0.52	0.56	0.60	0.72	0.75	0.89	0.98	1.13	1.48	0.51
	(4.80)	(4.15)	(3.88)	(3.58)	(3.47)	(3.37)	(3.34)	(3.26)	(3.43)	(3.92)	(3.23)
6	0.78	0.67	0.69	0.73	0.84	0.88	1.00	1.07	1.21	1.52	0.74
	(3.90)	(3.32)	(3.01)	(2.84)	(2.74)	(2.66)	(2.63)	(2.6)	(2.71)	(3.11)	(2.31)
12	1.03	0.96	0.97	0.93	1.02	1.05	1.09	1.17	1.21	1.43	0.40
	(3.24)	(2.69)	(2.49)	(2.25)	(2.17)	(2.14)	(2.06)	(2.12)	(2.16)	(2.51)	(1.91)
18	1.22	1.17	1.16	1.12	1.17	1.17	1.19	1.23	1.25	1.37	0.16
	(2.70)	(2.29)	(2.19)	(2.00)	(1.96)	(1.91)	(1.83)	(1.90)	(1.95)	(2.21)	(1.68)
24	1.39	1.30	1.28	1.29	1.26	1.24	1.24	1.26	1.26	1.30	-0.09
	(2.52)	(2.01)	(1.95)	(2.03)	(1.77)	(1.73)	(1.70)	(1.77)	(1.78)	(2.00)	(1.83)
	()	(=:*=)	(130)	()	()	(11.12)	(	(1117)	(11.0)	(=:::;)	(1100)
		A2 P	Percentage of	of observation	ons for each	portfolio w	ith highe	est return			
Holding months	Loser	P2	P3	P4	P5	P6		P7	P8	P9	Winner
3	24.0	60	39	2.8	49	35		52	70	123	30.3
6	17.9	49	3.8	3.5	3.8	3.5		5.9	8.4	12.9	35.4
12	17.3	5.6	4.2	43	5.0	43		7.5	9.5	12.9	29.4
12	10.3	8.1	5.8	J 6.6	5. <del>4</del> 6.0	5.2		7.3	8.6	07	22.0
24	22.5	10.3	5.0 6.4	59	83	5.2 7.8		6.9	8.2	9.5	14.1
24	22.1	10.5	0.4	5.7	0.5	7.0		0.9	0.2	).5	14.1
		<b>43</b> I	Dercentage (	of observatio	ons for each	portfolio w	vith lowe	st return			
Holding months	Loser	D2	P3	P/	P5	Portiono w		P7	D8	PQ	Winner
3	23.0	167	117	88	9.0	68		51	16	65	7.0
5	23.7	15.6	0.2	7.5	5.0	6.7		5.2	4.0	5.2	67
12	33.0	13.0	9.2	7.5	0.3	0.7		J.2 A.C	4.0	5.2	0.7
12	37.3	13.8	8.2	7.1	6.1	4.8		4.0	3.2	0.4	8.5 12.0
10	33.2	9.5	8.9	6.6	6.8	4.5		6.0	5.7	6.0	12.8
24	31.2	7.2	7.4	6.9	6.5	5.7		1.2	5.9	6.5	15.5
				Denal D. W.	.1						
				Panel B Va	alue-weighte	ed portfolio					
		-	BI Average	e monthly e	xcess return	and standar	rd devia	tion			
Holding months	Loser	P2	P3	P4	P5	P6	P7	P8	P9	Winner	$R_{wml}$
3	0.03	-0.11	0.22	0.20	0.56	0.58	0.68	0.77	0.84	1.00	0.97
	(5.81)	(4.72)	(4.24)	(3.99)	(3.80)	(3.64)	(3.58)	(3.52)	(3.64)	(4.08)	(5.15)
6	-0.02	0.06	0.35	0.40	0.56	0.63	0.72	0.78	0.88	0.96	0.98
	(4.14)	(3.34)	(3.03)	(2.97)	(2.78)	(2.67)	(2.69)	(2.66)	(2.74)	(2.99)	(3.42)
12	0.33	0.34	0.57	0.59	0.68	0.66	0.70	0.77	0.85	0.88	0.55
	(3.04)	(2.37)	(2.16)	(2.07)	(1.98)	(1.93)	(1.94)	(2.05)	(2.04)	(2.26)	(2.55)
18	0.56	0.56	0.70	0.66	0.74	0.71	0.76	0.77	0.85	0.79	0.23
	(2.66)	(1.84)	(1.78)	(1.64)	(1.65)	(1.65)	(1.69)	(1.81)	(1.83)	(1.93)	(2.41)
24	0.63	0.71	0.77	0.76	0.80	0.78	0.77	0.77	0.88	0.74	0.11
	(1.92)	(1.49)	(1.46)	(1.41)	(1.42)	(1.43)	(1.49)	(1.61)	(1.61)	(1.67)	(1.80)
		B2 P	Percentage o	f observatio	ons for each	portfolio wi	ith highe	est return			
Holding months	Loser	P2	P3	P4	P5	P6		P7	P8	P9	Winner
3	15.3	7.0	7.9	5.4	7.4	6.0		8.5	10.6	12.5	19.4
6	13.2	7.0	7.9	6.3	6.8	5.9		7.3	8.4	15.1	22.1
12	14.4	6.1	6.9	5.6	7.5	7.5		7.2	9.8	14.4	20.5
18	15.7	7.8	7.3	5.8	7.9	8.4		7.3	8.6	14.1	17.2
24	16.8	10.9	8.2	8.3	7.7	8.5		6.4	8.2	11.6	13.4
		B2 H	Percentage of	of observation	ons for each	portfolio w	ith lowe	st return			
Holding months	Loser	P2	P3	P4	P5	P6		P7	P8	P9	Winner
3	25.4	17.2	11.5	9.0	5.5	3.6		4.6	5.4	6.8	10.9
6	32.5	17.3	8.3	7.3	4.9	4.9		5.4	5.7	4.8	8.9
12	31.3	15.1	8.5	5.4	5.1	6.3		6.3	5.0	6.6	10.6
18	28.3	12.9	8.4	6.5	5.7	6.6		6.1	6.6	6.8	12.0
24	27.1	11.4	7.5	7.2	4.4	6.5		6.4	7.4	6.7	15.4
<b>_</b> ·			-		-						

Note: The table reports statistics on decile momentum portfolios based on ranking lagged 12 months' share returns from the lowest to the highest and 3, 6, 12, 18, and 24 months' holding horizons from 1956 to 2008. The research designates 'Winner' and 'Loser' as the top (P10) and bottom (P1) portfolios. Winner/Loser return  $R_{wml}$  is estimated by the spread in average monthly excess return between 'Winner' portfolio and 'Loser' portfolio. Panel A1 shows equal-weighted decile momentum portfolios' average monthly excess returns, with standard deviation in parenthesis. Panel A2 and A3 show the percentage of equal-weighted portfolio with highest return and lowest return among decile portfolios, respectively, over the sample period. Panel B shows the corresponding statistics on value-weighted decile momentum portfolios.

$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	<sup>3</sup> Overall period 0.51 (2.65) 0.74 (4.11) 0.41 (2.04) 0.16 (0.86)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	<ul> <li>3 Overall period</li> <li>0.51</li> <li>(2.65)</li> <li>0.74</li> <li>(4.11)</li> <li>0.41</li> <li>(2.04)</li> <li>0.16</li> <li>(0.86)</li> </ul>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 0.51\\ (2.65)\\ 0.74\\ (4.11)\\ 0.41\\ (2.04)\\ 0.16\\ (0.86)\end{array}$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 0.51\\ (2.65)\\ 0.74\\ (4.11)\\ 0.41\\ (2.04)\\ 0.16\\ (0.86)\end{array}$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} (2.03) \\ 0.74 \\ (4.11) \\ 0.41 \\ (2.04) \\ 0.16 \\ (0.86) \end{array}$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 0.74 \\ (4.11) \\ 0.41 \\ (2.04) \\ 0.16 \\ (0.86) \end{array}$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$(4.11) \\ 0.41 \\ (2.04) \\ 0.16 \\ (0.86)$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} 0.41 \\ (2.04) \\ 0.16 \\ (0.86) \end{array}$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(2.04) 0.16 (0.86)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0.16 (0.86)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(0.86)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	
(10.16) (-1.81) (-3.82) (-0.55) (3.44) (-0.57)	-0.09
	(-0.53)
A2 The number (percentage) of positive average monthly Winner/Loser return $R_{y}$ (momentum effect)	
Holding Observation Council a local and local	Orverell
Holding Observation Sample beginning 1960s 1970s 1980s 1990s 2000-2008	, Overall
months Number -1959	period
3   633   23   61   65   76   93   74	392
(51.1) $(50.8)$ $(54.2)$ $(63.3)$ $(77.5)$ $(68.5)$	(61.9)
6 630 25 83 71 82 104 79	444
(59.5) (69.2) (59.2) (68.3) (86.7) (73.1)	(70.5)
12 624 28 66 68 84 103 72	421
(77.8) $(55.0)$ $(56.7)$ $(70.0)$ $(85.8)$ $(66.7)$	(67.5)
18 618 28 57 50 76 109 68	388
(93.3) $(47.5)$ $(41.7)$ $(63.3)$ $(90.8)$ $(63.0)$	
24 612 24 51 35 61 104 59	(62.8)
(100) $(42.5)$ $(29.2)$ $(50.8)$ $(86.7)$ $(54.6)$	(62.8) 334
	(62.8) 334 (54.6)
	(62.8) 334 (54.6)
A. The number (noreentege) of negative everge monthly Winner/Leger return P (reverse) affect)	(62.8) 334 (54.6)
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)	(62.8) 334 (54.6)
A3 The number (percentage) of negative average monthly Winner/Loser return Rwml (reversal effect)HoldingObservationSample beginning1960s1970s1980s1990s2000-2008	(62.8) 334 (54.6)
A3 The number (percentage) of negative average monthly Winner/Loser return R <sub>wnl</sub> (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-2008           months         Number         -1959         -1959         -1970s         -1970s         -1970s         1980s         1990s         2000-2008	(62.8) 334 (54.6) 3 Overall period
A3 The number (percentage) of negative average monthly Winner/Loser return Rwml (reversal effect)HoldingObservationSample beginning1960s1970s1980s1990s2000-2008monthsNumber-19593633225955442734	(62.8) 334 (54.6) 3 Overall period 241
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1)
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5)
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5)
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4)
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4)
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4)
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wnl}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200s           months         Number         -1959         -         <	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4)
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200s           months         Number         -1959         -         <	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4)
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding months         Observation Number         Sample beginning -1959         1960s         1970s         1980s         1990s         2000-200s           3         633         22         59         55         44         27         34           6         630         17         37         49         38         16         29           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (6.7)         (52.5)         (58.3)         (36.7)         (9.2)         (37.0)           24         612         0         69         85         59         16         49           (0.0)         (57.5)         (70.8)         (49.2)         (13.3)         (45.4)	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding months         Observation Number         Sample beginning -1959         1960s         1970s         1980s         1990s         2000-200s           3         633         22         59         55         44         27         34           (48.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)           6         630         17         37         49         38         16         29           (40.5)         (30.8)         (40.8)         (31.7)         (13.3)         (26.9)           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (6.7)         (52.5)         (58.3)         (36.7)         (9.2)         (37.0)           24         612         0         69         85         59         16         49           (0.0)         (57.5)         (70.8)         (49.2) <td>(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period</td>	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wnl}$ (reversal effect)           Holding months         Observation         Sample beginning 1959         1960s         1970s         1980s         1990s         2000-200s           3         633         22         59         55         44         27         34           (48.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)           6         630         17         37         49         38         16         29           (40.5)         (30.8)         (40.8)         (31.7)         (13.3)         (26.9)           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (6.7)         (52.5)         (58.3)         (36.7)         (9.2)         (37.0)           24         612         0         69         85         59         16         49           (0.0)         (57.5)         (70.8)         (49.2)	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97
A.5 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding months         Observation Number         Sample beginning -1959         1960s         1970s         1980s         1990s         2000-200s           3         633         22         59         55         44         27         34           (48.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)           6         630         17         37         49         38         16         29           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (6.7)         (52.5)         (58.3)         (36.7)         (9.2)         (37.0)           24         612         0         69         85         59         16         49           (0.0)         (57.5)         (70.8)         (49.2)         (13.3)         (45.4)           Panel B Value-weighted portfolio <td< td=""><td>(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34)</td></td<>	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34)
As The number (percentage) of negative average monthly Winner/Loser return $R_{wnl}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200s           3         633         22         59         55         44         27         34           (48.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)           6         630         17         37         49         38         16         29           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (6.7)         (52.5)         (58.3)         (36.7)         (9.2)         (37.0)           24         612         0         69         85         59         16         49           (0.0)         (57.5)         (70.8)         (49.2)         (13.3)         (45.4)           Panel B Value-weighted portfolio           B1 Avera	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98
As The number (percentage) of negative average monthly Winner/Loser return $R_{wnal}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200t           nonths         Number         -1959         -	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81)
A.3 The number (percentage) of negative average monthly Winner/Loser return $R_{wnl}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200t           months         Number         -1959         -         -         -         3         633         22         59         55         44         27         34           (48.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)         6         630         17         37         49         38         16         29           (40.5)         (30.8)         (40.8)         (31.7)         (13.3)         (26.9)           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           Panel B Value-weighted portfolio           Bale beginning         1960s         1970s         1980s         1990s         2000-2008           months         Number         -1959	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81) 0.55
A.5 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200t           months         Number         -1959         -         -         448.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)           6         630         17         37         49         38         16         29           (40.5)         (30.8)         (40.8)         (31.7)         (13.3)         (26.9)           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           24         612         0         69         85         59         16         49           Marel B Value-weighted portfolio           Panel B Value-weighted portfolio           Observation         Sample beginning         1960s         1970s         1980s         1990s         2	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81) 0.55 (2.27)
A3 The number (percentage) of negative average monthly Winner/Loser return $R_{wnl}$ (reversal effect)           Holding months         Observation Number         Sample beginning -1959         1960s         1970s         1980s         1990s         2000-2000           3         633         22         59         55         44         27         34           (48.9)         (49.2)         (45.8)         (36.7)         (22.5)         (31.5)           6         630         17         37         49         38         16         29           (40.5)         (30.8)         (40.8)         (31.7)         (13.3)         (26.9)           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (0.0)         (57.5)         (70.8)         (49.2)         (13.3)         (45.4)           Panel B Value-weighted portfolio           BI Average monthly Winner/Loser return $R_{wnl}$ and t-stats           Holding         Observation         Sample beginni	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81) 0.55 (2.27) 0.22
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81) 0.55 (2.27) 0.23 (1.15)
A.5 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding         Observation         Sample beginning         1960s         1970s         1980s         1990s         2000-200i           months         Number         -1959         -	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81) 0.55 (2.27) 0.23 (1.15) (.15)
A.3 The number (percentage) of negative average monthly Winner/Loser return $R_{wml}$ (reversal effect)           Holding months         Observation         Sample beginning -1959         1960s         1970s         1980s         1990s         2000-200t           3         633         22         59         55         44         27         34           6         630         17         37         49         38         16         29           12         624         8         54         52         36         17         36           (22.2)         (45.0)         (43.3)         (30.0)         (14.2)         (33.3)           18         618         2         63         70         44         11         40           (6.7)         (52.5)         (58.3)         (36.7)         (9.2)         (37.0)           24         612         0         69         85         59         16         49           (0.0)         (57.5)         (70.8)         (49.2)         (13.3)         (45.4)           Panel B Value-weighted portfolio           Independent of the part o	(62.8) 334 (54.6) 3 Overall period 241 (38.1) 186 (29.5) 203 (32.5) 230 (37.2) 278 (45.4) 3 Overall period 0.97 (3.34) 0.98 (3.81) 0.55 (2.27) 0.23 (1.15) 0.11 0.11

 Table 4.5
 Statistics on Winner/Loser return R<sub>wml</sub> based on 12 months' formation length

 Panel A Equal-weighted portfolio

#### Continued

Holding months	Observation Number	Sample beginning -1959	1960s	1970s	1980s	1990s	2000-2008	Overall period
3	633	25	70	70	74	78	70	387
		(55.6)	(58.3)	(58.3)	(61.7)	(65.0)	(64.8)	(61.1)
6	630	26	78	73	75	93	73	418
		(61.9)	(65.0)	(60.8)	(62.5)	(77.5)	(67.6)	(66.3)
12	624	25	80	81	71	85	71	413
		(69.4)	(66.7)	(67.5)	(59.2)	(70.8)	(65.7)	(66.2)
18	618	15	73	77	63	91	64	383
		(50.0)	(60.8)	(64.2)	(52.5)	(75.8)	(59.3)	(62.0)
24	612	6	74	66	61	83	57	347
		(25.0)	(61.7)	(55.0)	(50.8)	(69.2)	(52.8)	(56.7)

Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	633	20	50	50	46	42	38	246
		(44.4)	(41.7)	(41.7)	(38.3)	(35.0)	(35.2)	(38.9)
6	630	16	42	47	45	27	35	212
		(38.1)	(35.0)	(39.2)	(37.5)	(22.5)	(32.4)	(33.7)
12	624	11	40	39	49	35	37	211
		(30.6)	(33.3)	(32.5)	(40.8)	(29.2)	(34.3)	(33.8)
18	618	15	47	43	57	29	44	235
		(50.0)	(39.2)	(35.8)	(47.5)	(24.2)	(40.7)	(38.0)
24	612	18	46	54	59	37	51	265
		(75.0)	(38.3)	(45.0)	(49.2)	(30.8)	(47.2)	(43.3)

Note: The table reports statistics on Winner/Loser return  $R_{wml}$ , the spread in average monthly excess return between top and bottom momentum portfolios which are formed on previous 12 months' share returns and hold for 3, 6, 12, 18, and 24 months from 1956 to 2008. Panel A1 shows equal-weighted Winner/Loser return  $R_{wml}$  and its t-statistics in parenthesis in overall period and subperiods. t-statistics is adjusted by Newey-West (1987) approach. Lag length is holding horizon minus one. Panel A2 and A3 show the number (percentage) of positive and negative equal-weighted Winner/Loser return  $R_{wml}$ , respectively, in overall period and subperiods. Panel B shows the corresponding statistics on value-weighted Winner/Loser return  $R_{wml}$ .

Table 4.6 Statistics on momentum coefficient  $\gamma_{mom}$  based on 12 months' formation length

			Panel A Equ	ual-weighted po	ortfolio			
		A1 The number (perce	entage) of sig	nificant positiv	e momentum co	befficient $\gamma_{mom}$		
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	610	0	4	7	57	85	53	206
		(0.0)	(3.3)	(5.8)	(47.5)	(70.8)	(49.1)	(33.8)
6	607	5	44	11	59	89	50	258
		(26.3)	(36.7)	(9.2)	(49.2)	(74.2)	(46.3)	(42.5)
12	601	9	38	11	50	80	52	240
		(69.2)	(31.7)	(9.2)	(41.7)	(66.7)	(48.1)	(39.9)
18	595	7	17	) 0	36	74	46	180
		(100.0)	(14.2)	(0.0)	(30.0)	(61.7)	(42.6)	(30.3)
24	589	1	8	) 0	31	70	23	133
		(100.0)	(6.7)	(0.0)	(25.8)	(58.3)	(21.3)	(22.6)
						. ,	X	. ,
		A2 The number (perce	ntage) of sign	nificant negativ	e momentum co	Defficient $\gamma_{mom}$		
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	610	0	0	6	0	0	0	6
		(0.0)	(0.0)	(5.0)	(0.0)	(0.0)	(0.0)	(1.0)
6	607	0	0	5	0	0	0	5
		(0.0)	(0.0)	(4.2)	(0.0)	(0.0)	(0.0)	(0.8)
12	601	0	0	19	5	0	0	24
		(0.0)	(0.0)	(15.8)	(4.2)	(0.0)	(0.0)	(4.0)
18	595	0	10	38	26	2	5	81
		(0.0)	(8.3)	(31.7)	(21.7)	(1.7)	(4.6)	(13.6)
24	589	0	10	70	12	4	13	109
		(0.0)	(8.3)	(58.3)	(10.0)	(3.3)	(12.0)	(18.5)
						()		
			Panel B Va	lue-weighted p	ortfolio			
		B1 The number (perce	entage) of sig	nificant positiv	e momentum co	befficient $\gamma_{mom}$		
Holding	Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
months	Number	-1959						period
3	610	5	20	26	27	57	49	184
		(22.7)	(16.7)	(21.7)	(22.5)	(47.5)	(45.4)	(30.2)
6	607	6	36	21	44	67	40	214
		(31.6)	(30.0)	(17.5)	(36.7)	(55.8)	(37.0)	(35.3)
12	601	0	43	38	29	60	31	201
		(0.0)	(35.8)	(31.7)	(24.2)	(50.0)	(28.7)	(33.4)
18	595	0	29	21	23	55	29	157
10	0,00	(0,0)	(24.2)	(17.5)	(19.2)	(45.8)	(26.9)	(26.4)
24	589	0	14	2	23	47	22	108
21	507	(0,0)	(11.7)	(1.7)	(19.2)	(39.2)	(20.4)	(18.3)
		(0.0)	(111)	()	(1)/	(0))	(=0)	(10.0)
		B? The number (norse	ntage) of size	nificant pagativ	a momentum a	pefficient V		
Uoldina	Observation	Sample beginning	1060a			1000	2000 2009	Overall
months	Number		19008	19/08	19808	19908	2000-2008	ported
months	number	-1939						perioa

Observation	Sample beginning	1960s	1970s	1980s	1990s	2000-2008	Overall
Number	-1959						period
610	0	0	0	0	0	0	0
	(0.0)	(0.0)	(0.0)	(0.0)	(0.0)	(0.0)	(0.0)
607	0	0	0	0	0	0	0
	(0.0)	(0.0)	(0.0)	(0.0)	(0.0)	(0.0)	(0.0)
601	0	0	7	9	0	4	20
	(0.0)	(0.0)	(5.8)	(7.5)	(0.0)	(3.7)	(3.3)
595	0	0	13	17	5	6	41
	(0.0)	(0.0)	(10.8)	(14.2)	(4.2)	(5.6)	(6.9)
589	0	19	14	14	12	23	82
	(0.0)	(15.8)	(11.7)	(11.7)	(10.0)	(21.3)	(13.9)
	Observation Number 610 607 601 595 589	Observation         Sample beginning           Number         -1959           610         0           (0.0)         607           601         0           (0.0)         601           601         0           (0.0)         595           (0.0)         589           0         (0.0)	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	ObservationSample beginning1960s1970sNumber $-1959$ $0$ $0$ $0$ $610$ $0$ $0$ $0$ $0$ $607$ $0$ $0$ $0$ $0$ $607$ $0$ $0$ $0$ $0$ $601$ $0$ $0$ $7$ $(0.0)$ $(0.0)$ $(0.0)$ $(5.8)$ $595$ $0$ $0$ $13$ $(0.0)$ $(0.0)$ $(10.8)$ $589$ $0$ $19$ $14$ $(0.0)$ $(15.8)$ $(11.7)$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$

Note: The table reports statistics on momentum coefficient  $\gamma_{mom}$ , estimated by regressing decile momentum portfolios' holding returns on their formation returns. Decile momentum portfolios are formed on previous 12 months' share returns and hold for 3, 6, 12, 18, and 24 months from 1956 to 2008, respectively. The research averages 24 regression coefficient  $\gamma_{mom}$  as one testing period, uses Newey-West (1987) approach to adjust standard errors and estimates t-statistics. Lag length is holding horizon minus one. The confidence level for t-test is 95%. Panel A1 and A2 show number (percentage) of significant positive and negative momentum coefficient  $\gamma_{mom}$  for equal-weighted momentum portfolios in overall sample period and subperiods. Panel B shows the corresponding statistics on value-weighted momentum coefficient  $\gamma_{mom}$ .

Table 4.7 The diference in average monthly return of 'Winner', 'Loser', and 'Winner/Loser' portfolio between two approaches of skipping and unskipping one month in momentum portfolio formation ( $R_s$ - $R_{us}$ )

Panel A Momentum formation with lagged 6 months' share returns								
		Equal-weighted	1		Value-weighted	ł		
	Winner	Loser	Winner/Loser	Winner	Loser	Winner/Loser		
Holding months	$R_s-R_{us}$	$R_s-R_{us}$	$R_s-R_{us}$	$R_s-R_{us}$	$R_s-R_{us}$	$R_s-R_{us}$		
3	0.15	-0.62	0.77	0.08	-0.43	0.51		
6	0.08	-0.31	0.39	0.04	-0.20	0.24		
12	0.02	-0.16	0.18	0.02	-0.09	0.11		
18	0.04	-0.14	0.18	0.03	-0.08	0.11		
24	0.06	-0.13	0.19	0.05	-0.06	0.11		
	Panel B	Momentum form	nation with lagged 12	2 months' share	returns			
		Equal-weighted	1		Value-weighted	ł		
	Winner	Loser	Winner/Loser	Winner	Loser	Winner/Loser		
Holding months	$R_s-R_{us}$	$R_s - R_{us}$	$R_s-R_{us}$	$R_s-R_{us}$	$R_s - R_{us}$	$R_s-R_{us}$		
3	0.10	-0.46	0.56	0.06	-0.30	0.36		
6	0.04	-0.23	0.27	0.05	-0.13	0.18		
12	-0.01	-0.10	0.11	0.01	-0.01	0.02		
18	0.00	-0.10	0.10	0.02	-0.03	0.05		
24	0.01	-0.11	0.12	0.01	0.00	0.01		

Note: The table reports the difference in average monthly return of 'Winner', 'Loser', and 'Winner/Loser' portfolio between two approaches of skipping and unskipping one month in momentum portfolio formation from 1956 to 2008, which are denoted as 'Winner  $R_s$ - $R_{us}$ ', 'Loser  $R_s$ - $R_{us}$ ', and 'Winner/Loser  $R_s$ - $R_{us}$ '

Table 4.8 The correlation in series return of 'Winner', 'Loser', 'Winner/Loser' portfolio, and momentum coefficient  $\gamma$  between two approaches of skipping and unskipping one month in momentum portfolio formation

Panel A Momentum formation with lagged 6 months' share returns									
Equal-weighted						Value	e-weighted		
Holding months	Winner	Loser	Winner/Loser	$\gamma_{\rm mom}$	Winner	Loser	Winner/Loser	$\gamma_{\rm mom}$	
3	0.9805	0.9833	0.9192	0.9533	0.9288	0.9317	0.8518	0.9078	
6	0.9859	0.9871	0.9203	0.9476	0.9308	0.9409	0.8983	0.9010	
12	0.9855	0.9877	0.9195	0.9417	0.9375	0.9446	0.8714	0.9062	
18	0.9807	0.9722	0.8639	0.9230	0.9326	0.9168	0.8270	0.8834	
24	0.9232	0.9047	0.6503	0.7250	0.9348	0.9237	0.8290	0.8749	
	P	anel B Mome	ntum formation wi	ith lagged 1	2 months' sh	are returns			
		Equal	-weighted			Value	e-weighted		
Holding months	Winner	Loser	Winner/Loser	$\gamma_{\rm mom}$	Winner	Loser	Winner/Loser	$\gamma_{\rm mom}$	
3	0.9880	0.9873	0.9432	0.9710	0.9528	0.9549	0.9085	0.9361	
6	0.9898	0.9907	0.9468	0.9681	0.9679	0.9643	0.9212	0.9510	
12	0.9903	0.9903	0.8476	0.9654	0.9599	0.9634	0.9175	0.9391	
18	0.9873	0.9799	0.9204	0.9379	0.9572	0.9619	0.9235	0.9233	
24	0.9726	0.8772	0.7063	0.6909	0.9593	0.9409	0.8929	0.9201	

Note: The table reports correlation coefficients of series return of 'Winner', 'Loser', and 'Winner/Loser' portfolio, and momentum coefficient  $\gamma_{mom}$  between two approaches of skipping and unskipping one month in momentum portfolio formation from 1956 to 2008.

|--|

			P	anel A Equ	al-weighted	l portfolio					
		A1	Average	monthly exe	cess return a	and standa	rd devia	tion			
Holding months	Small	P2	P3	P4	P5	P6	P7	P8	P9	Large	R <sub>sml</sub>
3	0.56	0.21	0.35	0.43	0.37	0.48	0.46	0.48	0.49	0.46	0.10
	(3.80)	(3.53)	(3.35)	(3.16)	(3.09)	(2.84)	(2.91)	(2.90)	(2.85)	(2.85)	(3.50)
6	0.66	0.29	0.39	0.46	0.42	0.48	0.50	0.51	0.53	0.48	0.18
	(3.21)	(2.62)	(2.38)	(2.16)	(2.13)	(1.96)	(1.99)	(1.96)	(1.91)	(2.00)	(2.78)
12	0.88	0.43	0.48	0.55	0.52	0.60	0.53	0.71	0.62	0.57	0.31
	(2.64)	(1.84)	(1.61)	(1.47)	(1.50)	(1.39)	(1.40)	(1.37)	(1.36)	(1.44)	(2.33)
18	1.04	0.55	0.55	0.64	0.62	0.68	0.58	0.84	0.72	0.64	0.41
	(2.24)	(1.53)	(1.33)	(1.23)	(1.24)	(1.21)	(1.18)	(1.23)	(1.13)	(1.19)	(1.95)
24	1.13	0.64	0.60	0.70	0.69	0.72	0.64	0.96	0.79	0.69	0.44
	(2.04)	(1.32)	(1.15)	(1.08)	(1.11)	(1.08)	(1.04)	(1.11)	(1.06)	(1.08)	(1.78)
		A2 Pero	centage of	observation	s for each p	ortfolio w	ith highe	st return			
Holding months	Small	P2	P3	P4	P5	Pé	5	P7	P8	P9	Large
3	22.7	3.9	5.0	4.5	5.9	7.6	5	6.2	14.8	11.8	17.6
6	25.4	3.7	3.1	2.5	5.9	7.9	)	7.9	14.1	15.3	14.1
12	29.9	2.0	0.6	3.4	6.3	7.2	2	5.7	16.7	13.5	14.7
18	35.1	1.2	0.3	5.0	4.4	8.2	2	2.6	19.3	13.5	10.5
24	37.5	0	0.3	4.5	1.5	2.7	7	4.8	21.7	16.7	10.4
			_								
TT 11' (1	0 11	A3 Per	centage of	observation	ns for each p	portfolio w	ith lowe	st return	DO	DO	T
Holding months	Small	P2	P3	P4	P5	Pe	)	P/	P8	P9	Large
3	21.0	10.9	6.2	3.4	8.4	6.4	4	6.2	12.0	8.1	1/.4
6	23.2	13.8	5.4	3.4	5.9	4.8	5	5.1	11.9	10.2	16.4
12	22.7	15.5	3.2	2.9	4.0	0.5	<b>1</b>	1.2	12.1	9.5	18.1
18	18.1	15.8	4.1	4.1	2.6	/.5	- -	1.5	10.8	10.2	19.0
24	16.1	15.8	5.1	4.5	2.4	8.0	3	10.4	8.3	13.1	15.8
			P	anel B Vali	ue-weighted	1 portfolio					
		B1	Average	monthly exe	cess return a	and standa	rd deviat	ion			
Holding months	Small	P2	P3	P4	P5	P6	P7	P8	P9	Large	Rem
3	0.35	0.23	0.36	0.43	0.36	0.47	0.47	0.47	0.46	0.41	-0.06
	(3.68)	(3.51)	(3.34)	(3.15)	(3.09)	(2.83)	(2.90)	(2.90)	(2.86)	(2.78)	(3.24)
6	0.48	0.30	0.40	0.46	0.41	0.47	0.51	0.51	0.50	0.42	0.06
	(3.02)	(2.60)	(2.36)	(2.15)	(2.12)	(1.96)	(1.98)	(1.97)	(1.89)	(1.94)	(2.54)
12	0.69	0.43	0.49	0.56	0.51	0.59	0.54	0.71	0.60	0.49	0.20
	(2.37)	(1.80)	(1.58)	(1.46)	(1.49)	(1.39)	(1.40)	(1.38)	(1.34)	(1.41)	(2.05)
18	0.84	0.55	0.56	0.64	0.60	0.67	0.59	0.85	0.69	0.56	0.28
	(1.97)	(1.51)	(1.31)	(1.22)	(1.24)	(1.21)	(1.18)	(1.24)	(1.12)	(1.17)	(1.71)
24	0.92	0.63	0.62	0.70	0.68	0.71	0.64	0.98	0.77	0.62	0.30
	(1.77)	(1.30)	(1.13)	(1.07)	(1.11)	(1.08)	(1.03)	(1.13)	(1.05)	(1.07)	(1.55)
					с I						
Holding of the	C	B2 Perc	centage of	observation	s for each p	ortiolio w	ith highe	st return	Do	DO	Len
Holding months	Small	P2	P3	P4	P5	P6	)	P/	P8	P9	Large
3	19.3	4.8	5.6	3.9	6.2	/.(	)	6./	14.6	11.8	20.2
6	24.3	3.4	3.4	4.5	5.1	8.2	2	8.5	15.5	14.4	13.0
12	28.2	1.4	1.1	4.6	5.7	8.9	<i>.</i> ,	6.0	17.5	12.9	13.5
18	32.5	0.9	0.3	5.6	4./	1.3	5	3.5	19.6	12.9	12.9
24	30.3	0	0	5.7	1.8	Ζ.	l	5.4	21.7	10.4	10.7
		B3 Per	centage of	observatior	ns for each r	ortfolio w	ith lowe	st return			
Holding months	Small	P2	P3	P4	P5	Pé	5	P7	P8	P9	Large
3	23.2	8.1	5.9	3.1	8.1	6.	7	5.0	11.2	7.8	20.7
6	24.3	9.6	4.2	3.1	6.8	5.9	9	4.5	11.3	10.7	19.5
12	23.3	10.6	3.7	2.6	3.2	6.0	)	6.0	11.2	10.9	22.4
18	18.7	11.7	3.5	5.0	2.6	6.7	7	5.3	10.2	10.5	25.7
24	17.9	11.9	4.5	3.9	2.4	9.8	3	8.3	7.1	16.1	18.2
							1 7	1 / 1		1 1 .	4 - 41

Note: The table reports statistics on decile size portfolios based on ranking shares market values from the lowest to the highest at the end of formation time and 3, 6, 12, 18, and 24 months' holding horizons from 1979 to 2008. The research designates 'Small' and 'Large' as the bottom (P1) and the top (P10) portfolios. Small/Large return  $R_{sml}$  is estimated by the spread in average monthly excess return between 'Small' portfolio and 'Large' portfolio. Panel A1 shows equal-weighted decile size portfolios' average monthly excess returns, with standard deviation in parenthesis. Panel A2 and A3 show the percentage of equal-weighted portfolio with highest return and lowest return among decile portfolios, respectively, over the sample period. Panel B shows the corresponding statistics on value-weighted decile size portfolios.

		Panel A Equal-weig	ghted portfolio					
A1 Average monthly Small/Large return R <sub>sml</sub> and t-stats								
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period			
	Number	-1989						
3	357	0.79	-0.62	0.06	0.10			
		(1.76)	(-1.34)	(0.13)	(0.33)			
6	354	0.90	-0.68	0.28	0.18			
		(1.77)	(-1.40)	(0.55)	(0.56)			
12	348	1.10	-0.75	0.63	0.31			
		(1.88)	(-1.38)	(1.20)	(0.88)			
18	342	1.35	-0.72	0.67	0.41			
		(2.29)	(-1.26)	(1.62)	(1.10)			
24	336	1.53	-0.74	0.66	0.44			
		(2.46)	(-1.32)	(2.22)	(1.15)			

Table 4.10 Statistics on Small/Large return  $R_{\rm sml}$ 

A2 The number (percentage) of positive average monthly Small/Large return  $R_{\text{sml}}$  (small-cap effect)

Holding months	Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period
3	357	77	47	47	171
		(59.7)	(39.2)	(43.5)	(47.9)
6	354	79	41	42	162
		(62.7)	(34.2)	(38.9)	(45.8)
12	348	79	39	48	166
		(65.8)	(32.5)	(44.4)	(47.7)
18	342	89	44	65	198
		(78.1)	(36.7)	(60.2)	(57.9)
24	336	87	44	75	206
		(80.6)	(36.7)	(69.4)	(61.3)

A3	The number	(percentage)	of negative	average	monthly S	Small/Large	return R <sub>sml</sub>	(large-cap	effect)
----	------------	--------------	-------------	---------	-----------	-------------	-------------------------	------------	---------

Holding months	Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period
3	357	52	73	61	186
		(40.3)	(60.8)	(56.5)	(52.1)
6	354	47	79	66	192
		(37.3)	(65.8)	(61.1)	(54.2)
12	348	41	81	60	182
		(34.2)	(67.5)	(55.6)	(52.3)
18	342	25	76	43	144
		(21.9)	(63.3)	(39.8)	(42.1)
24	336	21	76	33	130
		(19.4)	(63.3)	(30.6)	(38.7)

Panel B Value-weighted portfolio

B1 Average monthly Small/Large return R <sub>sml</sub> and t-stats							
Holding months	Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period		
3	357	0.48	-0.68	-0.02	-0.06		
		(1.17)	(-1.52)	(-0.04)	(-0.23)		
6	354	0.63	-0.74	0.28	0.06		
		(1.42)	(-1.55)	(0.61)	(0.20)		
12	348	0.81	-0.82	0.65	0.20		
		(1.79)	(-1.58)	(1.43)	(0.63)		
18	342	1.03	-0.84	0.72	0.28		
		(2.46)	(-1.54)	(1.96)	(0.85)		
24	336	1.21	-0.86	0.68	0.30		
		(2.85)	(-1.61)	(2.56)	(0.89)		

#### Continued

B2 The number (percentage) of positive average monthly Small/Large return R <sub>sml</sub> (small-cap effect)							
Holding months	Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period		
3	357	68	45	47	160		
		(52.7)	(37.5)	(43.5)	(44.8)		
6	354	76	39	49	164		
		(60.3)	(32.5)	(45.4)	(46.3)		
12	348	79	36	59	174		
		(65.8)	(30.0)	(54.6)	(50.0)		
18	342	85	40	77	202		
		(74.6)	(33.3)	(71.3)	(59.1)		
24	336	91	35	79	205		
		(84.3)	(29.2)	(73.1)	(61.0)		

B2 The number (percentage) of negative average monthly Small/Large return  $R_{sml}$  (large-cap effect)

					•
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
	Number	-1989			
3	357	61	75	61	197
		(47.3)	(62.5)	(56.5)	(55.2)
6	354	50	81	59	190
		(39.7)	(67.5)	(54.6)	(53.7)
12	348	41	84	49	174
		(34.2)	(70.0)	(45.4)	(50.0)
18	342	29	80	31	140
		(25.4)	(66.7)	(28.7)	(40.9)
24	336	17	85	29	131
		(15.7)	(70.8)	(26.9)	(39.0)

Note: The table reports statistics on Small/Large return  $R_{sml}$ , the spread in average monthly excess return between bottom ('Small') and top ('Large') size portfolios which are formed on shares' market value at the end of formation time and hold for 3, 6, 12, 18, and 24 months from 1979 to 2008. Panel A1 shows equal-weighted Small/Large return  $R_{sml}$  and its t-statistic in parenthesis in overall period and subperiods. t-statistics is adjusted by Newey-West (1987) approach. Lag length is holding horizon minus one. Panel A2 and A3 show the number (percentage) of positive and negative equal-weighted Small/Large return  $R_{sml}$ , respectively, in overall period and subperiods. Panel B shows the corresponding statistics on value-weighted Small/Large return  $R_{sml}$ .

Panel A Equal-weighted portfolio								
Al	A1 The number (percentage) of significant negative size coefficient $\gamma_{mv}$ (Small-cap effect)							
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period			
	Number	-1989						
3	334	28	0	0	28			
		(26.4)	(0.0)	(0.0)	(8.4)			
6	331	29	2	0	31			
		(28.2)	(1.7)	(0.0)	(9.4)			
12	325	51	6	12	69			
		(52.6)	(5.0)	(11.1)	(21.2)			
18	319	67	21	58	146			
		(73.6)	(17.5)	(53.7)	(45.8)			
24	313	55	13	61	129			
		(64.7)	(10.8)	(56.5)	(41.2)			

Table 4.11	Statistics	on size	coefficient	γ <sub>mv</sub>
------------	------------	---------	-------------	-----------------

A2 The number (percentage) of significant positive size coefficient  $\gamma_{mv}$  (Large-cap effect)

Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
	Number	-1989			
3	334	0	46	21	67
		(0.0)	(38.3)	(19.4)	(20.1)
6	331	0	49	37	86
		(0.0)	(40.8)	(34.3)	(26.0)
12	325	0	52	32	84
		(0.0)	(43.3)	(29.6)	(25.8)
18	319	8	75	30	113
		(8.8)	(62.5)	(27.8)	(35.4)
24	313	16	72	22	110
		(18.8)	(60.0)	(20.4)	(35.1)

Panel B Value-weighted portfolio

B1 The number (percentage) of significant negative size coefficient $\gamma_{mv}$ (Small-cap effect)								
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period			
	Number	-1989						
3	334	25	0	2	27			
		(23.6)	(0.0)	(1.9)	(8.1)			
6	331	29	0	0	29			
		(28.2)	(0.0)	(0.0)	(8.8)			
12	325	50	6	16	72			
		(51.5)	(5.0)	(14.8)	(22.2)			
18	319	67	21	58	149			
		(73.6)	(17.5)	(53.7)	(46.7)			
24	313	52	12	61	125			
		(61.2)	(10.0)	(56.5)	(39.9)			

B2 The number (percentage) of significant positive size coefficient  $\gamma_{mv}$  (Large-cap effect)

Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
	Number	-1989			
3	334	0	46	9	55
		(0.0)	(38.3)	(8.3)	(16.5)
6	331	0	48	16	64
		(0.0)	(40.0)	(14.8)	(19.3)
12	325	0	54	28	82
		(0.0)	(45.0)	(25.9)	(25.2)
18	319	8	75	30	108
		(8.8)	(62.5)	(27.8)	(33.9)
24	313	O Í	73	22	<b>9</b> 5
		(0.0)	(60.8)	(20.4)	(30.4)

Note: The table reports statistics on size coefficient  $\gamma_{mv}$ , estimated by regressing decile size portfolios' holding returns on their log mean market values at the formation time. Decile size portfolios are formed on shares' market values at the end of formation time and hold for 3, 6, 12, 18, and 24 months from 1979 to 2008. The research averages 24 regression coefficient  $\gamma_{mv}$  as one testing period, uses Newey-West (1987) approach to adjust standard errors and estimates t-statistics. Lag length is holding horizon minus one. The confidence level for t-test is 95%. Panel A1 and A2 show number (percentage) of significant negative and positive size coefficient  $\gamma_{mv}$  for equalweighted size portfolios in overall sample period and subperiods. Panel B shows the corresponding statistics on value-weighted size coefficient  $\gamma_{mv}$ .

Table 4.12 Statistics	on decile	value	portfolios
-----------------------	-----------	-------	------------

				Panel A E	qual-weight	ted portfoli	0				
			Al Averag	ge monthly	excess retur	n and stand	lard devi	ation			
Holding months	Low	P2	P3	P4	P5	P6	P7	P8	P9	High	R <sub>lmh</sub>
3	1.17	0.90	0.71	0.60	0.49	0.41	0.28	0.23	0.20	-0.09	1.26
	(3.26)	(3.00)	(3.07)	(2.97)	(2.88)	(2.92)	(2.81)	(2.94)	(3.24)	(3.48)	(2.64)
6	1.15	0.96	0.80	0.68	0.54	0.47	0.37	0.31	0.23	0.03	1.12
10	(2.59)	(2.31)	(2.37)	(2.29)	(2.15)	(2.15)	(2.19)	(2.26)	(2.44)	(2.75)	(2.18)
12	1.10	1.07	0.90	0.77	0.66	0.59	0.45	0.39	(1.72)	0.18	(1.96)
10	(2.14)	(1.85)	(1.79)	(1.70)	(1.68)	(1.00)	(1.60)	(1./1)	(1./3)	(2.05)	(1.86)
18	(2.07)	1.14	1.02	(1.52)	(1.46)	(1.51)	(1.26)	(1.27)	(1,41)	(1.74)	(1,77)
24	(2.07)	(1.00)	(1.70)	(1.55)	(1.40)	(1.31)	(1.50)	(1.57)	(1.41)	(1.74)	(1.//)
24	(2.04)	(1.56)	(1.46)	(1.42)	(1.46)	(1.26)	(1.26)	(1, 21)	(1, 22)	(1.50)	(1.55)
	(2.04)	(1.30)	(1.40)	(1.43)	(1.40)	(1.50)	(1.20)	(1.21)	(1.22)	(1.50)	(1.55)
		A2	Percentage	of observati	ons for each	n portfolio v	with high	nest return			
Holding months	Low	P2	P3	P4	P5	P6	5	P7	P8	P9	High
3	32.8	18.3	7.8	5.5	6.4	4.6	5	4.1	4.1	7.5	9.0
6	37.1	19.0	7.0	6.4	4.4	4.1	l	4.7	4.4	5.6	7.3
12	34.5	17.9	8.3	7.1	5.4	5.7	7	4.5	4.2	4.5	8.0
18	35.8	16.4	8.2	7.9	7.3	6.1	l	3.0	3.6	3.0	8.8
24	34.9	17.0	10.8	7.4	7.7	6.2	2	2.5	5.2	4.6	3.7
		A3 I	Percentage of	of observation	ons for each	portfolio v	vith lowe	est return			
Holding months	Low	P2	P3	P4	P5	P6	5	P7	P8	P9	High
3	8.7	4.9	6.7	4.6	4.6	5.8	3	4.1	6.1	15.1	39.4
6	9.4	4.4	3.8	4.4	4.1	4.4	1	7.0	5.6	13.7	43.3
12	9.2	2.4	3.9	3.6	5.4	2.1		4.8	7.7	15.8	45.2
18	9.4	1.2	3.6	2.7	3.3	3.0	)	6.1	8.8	18.8	43.0
24	5.9	0.6	1.9	2.8	3.1	5.9	)	3.7	10.5	22.2	43.5
				Panel B Va	alue-weighte	ed portfolio	)				
		]	B1 Average	e monthly ex	xcess return	and standa	ard devia	tion			
Holding months	Low	P2	P3	P4	P5	P6	P7	P8	P9	High	$R_{lmh}$
3	1.24	0.78	0.69	0.66	0.46	0.40	0.28	0.31	0.29	0.31	0.93
	(4.05)	(3.42)	(3.38)	(3.31)	(3.14)	(2.98)	(3.00)	(2.82)	(3.02)	(2.86)	(3.69)
6	1.31	0.77	0.82	0.66	0.52	0.44	0.34	0.37	0.33	0.35	0.96
	(3.00)	(2.37)	(2.55)	(2.37)	(2.19)	(2.14)	(2.00)	(1.93)	(2.13)	(1.99)	(2.79)
12	1.36	0.83	0.91	0.68	0.60	0.55	0.42	0.44	0.37	0.35	1.01
	(2.29)	(1.72)	(1.86)	(1.70)	(1.58)	(1.52)	(1.30)	(1.33)	(1.42)	(1.40)	(2.16)
18	1.45	0.90	1.01	0.78	0.71	0.63	0.49	0.51	0.40	0.41	1.04
	(1.95)	(1.28)	(1.61)	(1.45)	(1.38)	(1.31)	(1.11)	(1.17)	(1.15)	(1.21)	(1.90)
24	1.50	0.95	1.10	0.91	0.85	0.72	0.55	0.56	0.49	0.47	1.03
	(1./5)	(1.10)	(1.44)	(1.37)	(1.24)	(1.13)	(1.01)	(1.01)	(1.07)	(1.13)	(1.73)
		B2 F	Percentage o	f observatio	ons for each	portfolio w	ith high	est return			
Holding months	Low	P2	P3	P4	P5	P6	- )	P7	P8	P9	High
3	26.4	12.5	9.0	9.0	7.0	7.0	)	4.1	6.1	7.2	11.9
6	32.5	13.5	9.9	7.6	3.2	3.5	5	3.2	8.8	6.7	11.1
12	32.4	14.0	12.2	7.4	3.3	6.3	3	3.0	6.3	6.0	9.2
18	37.0	13.6	10.0	9.4	2.4	5.2	2	3.9	6.1	4.5	7.9
24	36.4	14.2	11.7	9.3	4.0	2.8	3	1.9	5.9	3.4	10.5
		D.2. 1	Damas (	£ _1				_4			
Holding and	T	B3 1	ercentage (	DI ODSERVATIO	JIS IOT each	portiolio w	vith lowe	p7	DQ	DO	11:-1-
Holding months	LOW	P2	P3	P4	P5	P6	)	r/	P8 97	P9 10 /	High
5	9.9	10.4	8./	1.2	6.4	11.	0	9.0	ð./	10.4	18.5
6	13.2	9.1	9.4	4./	5.8	10.	2	9.9	8.8	15./	15.2
12	9.8	8.6	7.7	4.2	6.5	8.9	<i>i</i> 7	10.1	9.2	14.3	20.5
18	10.0	5.8	1.5	6.4	5.5	9.7	/	10.9	10.9	12./	20.9
24	0.3	/.4	9.3	0.8	5.2	/.1	ι ,	10.3	11.4	9.0	20.9

Note: The table reports statistics on decile value portfolios based on ranking shares' price-to-book ratios (PB) from the lowest to the highest at the end of formation time and 3, 6, 12, 18, and 24 months' holding horizons from 1980 to 2008. The research designates 'Low' and 'High' as the bottom (P1) and top (P10) portfolios. Low/High return  $R_{lmh}$  is estimated by the spread in average monthly excess return between 'Low' portfolio and 'High' portfolio. Panel A1 shows equal-weighted value decile portfolios' average monthly excess returns, with standard deviation in parenthesis. Panel A2 and A3 show the percentage of equal-weighted portfolio with highest return and lowest return among decile portfolios, respectively, over the sample period. Panel B shows the corresponding statistics on value-weighted decile value portfolios.

Panel A Equal-weighted portfolio									
	A1 Average monthly Low/High return R <sub>lmh</sub> and t-stats								
Holding months	Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period				
3	345	1.32 (3.53)	0.90 (2.83)	1.57 (4.03)	1.25 (5.78)				
6	342	1.27 (3.29)	0.76 (2.12)	1.36 (3.21)	1.12 (4.68)				
12	336	1.35 (2.89)	0.65 (1.58)	0.98 (2.04)	0.98 (3.48)				
18	330	1.71 (3.15)	0.63 (1.61)	0.65 (1.25)	0.97 (3.10)				
24	324	1.88 (2.80)	0.60 (1.83)	0.62 (1.44)	0.99 (3.01)				

Table 4.13 Statistics on Low/High return  $R_{\rm lmh}$ 

A2 The number (percentage) of positive average monthly Low/High return  $R_{\textrm{lmh}}$  (value effect)

Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
	Number	-1989			
3	345	88	80	92	260
		(75.2)	(66.7)	(85.2)	(75.4)
6	342	89	77	96	262
		(78.1)	(64.2)	(88.9)	(76.6)
12	336	84	87	95	266
		(77.8)	(72.5)	(88.0)	(79.2)
18	330	78	80	90	248
		(76.5)	(66.7)	(83.3)	(75.2)
24	324	73	96	95	264
		(76.0)	(80.0)	(88.0)	(81.5)

A3 The number (percentage) of negative average monthly Low/High return  $R_{lmh}$  (growth effect)

Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period
345	29	40	16	85
	(24.8)	(33.3)	(14.8)	(24.6)
342	25	43	12	80
	(21.9)	(35.8)	(11.1)	(23.4)
336	24	33	13	70
	(22.2)	(27.5)	(12.0)	(20.8)
330	24	40	18	82
	(23.5)	(33.3)	(16.7)	(24.8)
324	23	24	13	60
	(24.0)	(20.0)	(12.0)	(18.5)
	Observation Number 345 342 336 330 324	Observation Number         Sample beginning           345         29           (24.8)         (24.8)           342         25           (21.9)         (21.9)           336         24           (22.2)         (23.5)           324         23           (24.0)         (24.0)	$\begin{array}{c ccccc} \hline Observation & Sample beginning & 1990s \\ \hline Number & -1989 \\ 345 & 29 & 40 \\ & (24.8) & (33.3) \\ 342 & 25 & 43 \\ & (21.9) & (35.8) \\ 336 & 24 & 33 \\ & (22.2) & (27.5) \\ 330 & 24 & 40 \\ & (23.5) & (33.3) \\ 324 & 23 & 24 \\ & (24.0) & (20.0) \\ \hline \end{array}$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

Panel B Value-weighted portfolio

B1 Average monthly Low/High return R <sub>imh</sub> and t-stats							
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period		
	Number	-1989					
3	345	1.14	0.39	1.29	0.93		
		(2.92)	(0.86)	(1.99)	(3.16)		
6	342	1.14	0.45	1.34	0.96		
		(2.89)	(0.92)	(2.10)	(3.17)		
12	336	1.25	0.38	1.47	1.01		
		(2.86)	(0.72)	(2.33)	(3.11)		
18	330	1.42	0.29	1.52	1.04		
		(2.57)	(0.58)	(2.86)	(3.07)		
24	324	1.54	0.25	1.45	1.03		
		(2.24)	(0.64)	(2.69)	(2.86)		

#### Continued

B2 7	B2 The number (percentage) of positive average monthly Low/High return $R_{lmh}$ (value effect)							
Holding months	Observation Number	Sample beginning -1989	1990s	2000-2008	Overall period			
3	345	75	63	71	209			
		(64.1)	(52.5)	(65.7)	(60.6)			
6	342	82	60	70	212			
		(71.9)	(50.0)	(64.8)	(62.0)			
12	336	89	66	74	229			
		(82.4)	(55.0)	(68.5)	(68.2)			
18	330	85	65	80	230			
		(83.3)	(54.2)	(74.1)	(69.7)			
24	324	75	67	87	229			
		(78.1)	(55.8)	(80.6)	(70.7)			

B3 The number (percentage) of negative average monthly Low/high return R<sub>lmh</sub> (growth effect)

	-			-	
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
	Number	-1989			
3	345	42	57	37	136
		(35.9)	(47.5)	(34.3)	(49.4)
6	342	32	60	38	130
		(28.1)	(50.0)	(35.2)	(38.0)
12	336	19	54	34	107
		(17.6)	(45.0)	(31.5)	(31.8)
18	330	17	55	28	100
		(16.7)	(45.8)	(25.9)	(30.3)
24	324	21	53	21	95
		(21.9)	(44.2)	(19.4)	(29.3)

Note: The table reports statistics on Low/High return  $R_{lmh}$ , the spread in average monthly excess return between bottom and top value portfolios which are formed on shares' price-to-market ratios at the end of formation time and hold for 3, 6, 12, 18, and 24 months from 1980 to 2008. Panel A1 shows equal-weighted Low/High return  $R_{lmh}$  and its t-statistic in parenthesis in overall period and subperiods. t-statistics is adjusted by Newey-West (1987) approach. Lag length is holding horizon minus one. Panel A2 and A3 show the number (percentage) of positive and negative equal-weighted Low/High return  $R_{lmh}$ , respectively, in overall period and subperiods. Panel B shows the corresponding statistics on value-weighted Low/High return  $R_{lmh}$ 

1	A1 The number (per	centage) of significant neg	gative value coeff	icient $\gamma_{ph}$ (Value effect	t)				
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period				
3	Number 322	-1989 54	54	83	101				
5	522	(57.4)	(45.0)	(76.9)	(59.3)				
6	319	48	56	82	186				
0	517	(52.7)	(467)	(75.9)	(58.3)				
12	313	56	(40.7)	(73.5)	203				
12	515	(65.9)	(57.5)	(72)	(64.9)				
18	307	59	(57.5)	(72.2)	210				
10	507	(74.7)	(63.3)	(60.4)	(68.4)				
24	301	56	(03.3)	(0).4)	223				
24	501	(767)	(767)	(69.4)	(74.1)				
fiolding months	Number	-1989	17703	2000-2000	Overall period				
A2 The number (percentage) of significant positive value coefficient $\gamma_{pb}$ (Growth effect)									
2	Number	-1989	0	0	0				
3	322	0	0	0	0				
<i>,</i>	210	(0.0)	(0.0)	(0.0)	(0.0)				
6	319	0	0	0	0				
10		(0.0)	(0.0)	(0.0)	(0.0)				
12	313	0	0	1	l				
10	207	(0.0)	(0.0)	(0.9)	(0.3)				
18	307	8	3	9	20				
		(10.1)	(2.5)	(8.3)	(6.5)				
24	301	4	2	13	19				
		(5.5)	(1.7)	(12.0)	(6.3)				
Panel B Value-weighted portfolio									
B1 The number (percentage) of significant negative value coefficient $\gamma_{\rm nb}$ (Value effect)									
TT 11' (1		0 1 1 1 1	1000	-	0 11 1				

Table 4.14 Statistics on value coefficient  $\gamma_{pb}$ 

	BI The number (per	centage) of significant ne	gative value coeff	icient $\gamma_{pb}$ (Value effect	t)
Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
0	Number	-1989			•
3	322	52	17	61	130
		(55.3)	(14.2)	(56.5)	(40.4)
6	319	58	13	62	133
		(63.7)	(10.8)	(57.4)	(41.7)
12	313	61	23	59	143
		(71.8)	(19.2)	(54.6)	(45.7)
18	307	57	40	66	163
		(72.2)	(33.3)	(61.1)	(53.1)
24	301	54	44	74	172
		(74.0)	(36.7)	(68.5)	(57.1)

B2 The number (percentage) of significant positive value coefficient  $\gamma_{pb}$  (Growth effect)

Holding months	Observation	Sample beginning	1990s	2000-2008	Overall period
	Number	-1989			
3	322	0	0	3	3
		(0.0)	(0.0)	(2.8)	(0.9)
6	319	0	12	1	13
		(0.0)	(10.0)	(0.9)	(4.1)
12	313	0	14	13	27
		(0.0)	(11.7)	(12.0)	(8.6)
18	307	4	13	17	34
		(5.1)	(10.8)	(15.7)	(11.1)
24	301	8	15	17	40
		(11.0)	(12.5)	(15.7)	(13.3)

Note: The table reports statistics on value coefficient  $\gamma_{pb}$ , estimated by regressing decile value portfolios' holding returns on their price-to-book ratios at the formation time. Decile value portfolios are formed on shares' price-to-book ratios at the end of formation time and hold for 3, 6, 12, 18, and 24 months from 1980 to 2008. The research averages 24 regression coefficient  $\gamma_{pb}$  as one testing period, uses Newey-West (1987) approach to adjust standard errors and estimates t-statistics. Lag length is holding horizon minus one. The confidence level for t-test is 95%. Panel A1 and A2 show number (percentage) of significant negative and positive value coefficient  $\gamma_{pb}$  for equalweighted value portfolios in overall sample period and subperiods. Panel B shows the corresponding statistics on value-weighted value coefficient  $\gamma_{pb}$ .

Figual-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.4843-0.17680.0002-0.2158-0.51910.35016-0.3860-0.44250.0574-0.3264-0.51210.350312-0.5147-0.45290.1560-0.4687-0.52940.503524-0.4485-0.39270.4516-0.303-0.42930.4949Panel B_i====================================	Panel A Overall period from 1980 to 2008									
Holding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.4843-0.17680.002-0.21S8-0.51910.34016-0.3860-0.44260.0574-0.3264-0.51210.358012-0.5147-0.45290.1560-0.4685-0.52940.503524-0.4485-0.32270.4516-0.3803-0.42930.4949Panel B_arurrey byso-becerberValue-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.4178-0.30340.5414-0.1962-0.31670.44596-0.4518-0.53340.6090-0.2641-0.4100.590212-0.6570-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.41190.565924-0.2854-0.44440.6343-0.1888-0.27890.433118-0.3606-0.50400.0673-0.2758-0.47320.27486-0.5606-0.50400.1034-0.4968-0.45920.404719-Deserberberberberberberberberberberberberbe			Equal-weighted			Value-weighted				
3-0.4843-0.17680.0002-0.2158-0.50190.34016-0.3860-0.44260.0574-0.3264-0.51210.358012-0.5147-0.45290.1560-0.4687-0.58930.450018-0.4728-0.39270.4516-0.3803-0.42930.494924-0.4485-0.39270.4516-0.3803-0.42930.4949Equal-weightedValue-weightedEqual-weightedValue-weightedMon/ValueSize/Value3-0.4178-0.30940.5414-0.1962-0.31670.44596-0.4518-0.53340.6090-0.2641-0.42100.590212-0.6370-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.188-0.27890.483918-0.3807-0.49440.6733-0.47320.27486-0.5666-0.05940.1034-0.4962-0.41320.27486-0.5666-0.6940.1034-0.4939-0.45920.424718-0.6649-0.64320.5149-0.4339-0.45990.466224-0.5113-0.50950.5237-0.3981-0.36950.404512-0.7107-0.43140.2344-0.5175-0.47320.424718-0.6649-0.6945 <td>Holding months</td> <td>Mom/Size</td> <td>Mom/Value</td> <td>Size/Value</td> <td>Mom/Size</td> <td>Mom/Value</td> <td>Size/Value</td>	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value			
6-0.3860-0.44260.0574-0.3264-0.51210.358012-0.5147-0.45290.1560-0.4687-0.58930.450018-0.4728-0.41720.2808-0.4545-0.52940.503524-0.4485-0.39270.4516-0.3803-0.42930.4949Panel B Jarrer 1980-December 1989Equal-weightedMom/YalueSize/ValueMom/YalueMom/YalueSize/Value3-0.4178-0.30940.5414-0.1962-0.31670.44596-0.4518-0.53340.6090-0.2641-0.42100.569112-0.65470.6373-0.6474-0.54760.617918-0.3810-0.44710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.483918-0.3606-0.50940.1034-0.4680.51540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.3693-0.45990.466224-0.5113-0.50950.5237-0.3981-0.36950.404512-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.3987-0.65710.265124-0.5113-0.50950.5237-0.3981-0.36950.319312-0.2513-	3	-0.4843	-0.1768	0.0002	-0.2158	-0.5019	0.3401			
12-0.5147-0.45290.1560-0.4687-0.58930.450018-0.4728-0.41720.2808-0.4545-0.52940.503524-0.4485-0.39270.4516-0.3803-0.42930.494924-0.4485-0.39270.4516-0.3803-0.42930.4949belin under state st	6	-0.3860	-0.4426	0.0574	-0.3264	-0.5121	0.3580			
18-0.4728-0.41720.2808-0.4545-0.52940.503524-0.4485-0.39270.4516-0.3803-0.42930.494924-0.4485-0.39270.4516-0.3803-0.42930.4949Fanel B Janury 1980-December 1989Equal-weightedMom/SizeMom/ValueSize/ValueMom/ValueSize/Value10Mom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.4178-0.30940.5414-0.1962-0.31670.44596-0.4518-0.53340.6090-0.2641-0.42100.590212-0.6370-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.483918-0.5897-0.39400.0673-0.2788Mom/ValueSize/Value19Mom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value12-0.7107-0.43140.2344-0.5175-0.47320.27486-0.5066-0.50940.1034-0.4968-0.55540.313112-0.7107-0.64320.5149-0.4339-0.45990.466224-0.5113-0.50590.5237-0.3981-0.36950.304312-0.7107-0.64320.5149-0.4339-0.45990	12	-0.5147	-0.4529	0.1560	-0.4687	-0.5893	0.4500			
24-0.4485-0.39270.4516-0.3803-0.42930.4949Panel B Jauret J980-December J980Holding monthsKom/SizeValue-weightedValue-weightedHolding months0.4178-0.30940.5414-0.1962-0.31670.44596-0.4518-0.53340.6090-0.2641-0.42100.590212-0.6370-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.4839Panel C Jauret Jupo-December JupoEqual-weightedValue-weightedNom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value10Mom/SizeMom/ValueSize/ValueMom/Size0.673-0.47320.27486-0.5666-0.50940.1034-0.4968-0.55540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.4339-0.45990.466224-0.5113-0.50550.5237-0.3981-0.36950.319312-0.7107-0.6312-0.6227-0.13554.03940.316312-0.5113-0.5051-0.3284-0.5175-0.47520.424713-0.6549-0.6227-0.13550.3193-0.5	18	-0.4728	-0.4172	0.2808	-0.4545	-0.5294	0.5035			
Parel B J=	24	-0.4485	-0.3927	0.4516	-0.3803	-0.4293	0.4949			
Equal-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.4178-0.30940.5414-0.1962-0.31670.44596-0.4518-0.53340.6090-0.2641-0.42100.590212-0.6370-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.4839Panel C January 1990-December 1999Equal-weightedMom/SizeMom/ValueSize/ValueHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value10Mom/SizeMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value11-0.5066-0.50940.1034-0.4968-0.55540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.4339-0.36950.4045Panel D Januar 2000-December 2008Equal-weightedValue-weightedValue-weightedIndia monthsMom/SizeMom/SizeMom/SizeAuguste interveightedValue-weightedSize/ValueMom/SizeMom/SizeJanuar 2000-December 2008<			Panel B Jai	nuary 1980-Decen	nber 1989					
Holding months         Mom/Value         Size/Value         Mod/Value         Size/Value         Mom/Value </td <td></td> <td></td> <td>Equal-weighted</td> <td>5</td> <td></td> <td>Value-weighted</td> <td></td>			Equal-weighted	5		Value-weighted				
.         .0.4178         .0.3094         0.5414         .0.1962         .0.3167         0.4459           6         .0.4518         .0.5334         .0.6090         .0.2641         .0.4210         .0.5902           12         .0.6370         .0.5477         .0.6734         .0.5478         .0.5476         .0.6179           18         .0.3810         .0.4971         .0.5790         .0.4924         .0.419         .0.5659           24         .0.2854         .0.4444         .0.6343         .0.1888         .0.2789         .0.4839           Panel C Jauveighted	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value			
6-0.4518-0.53340.6090-0.2641-0.42100.509212-0.6370-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.4839Panel C Jarrey Ty90-DecemberValue-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.5897-0.39400.0673-0.2758-0.47320.27486-0.5606-0.50940.1034-0.4968-0.55540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.4339-0.45990.466224-0.5113-0.50950.5237-0.3981-0.36950.4045Fanel D Jarrey 2000-December 2008Equal-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/Value3-0.47410.0796-0.6227-0.1935-0.58990.31936-0.1768-0.312-0.6004-0.2417-0.52000.207012-0.2585-0.3894-0.5674-0.3897-0.65710.265118-0.3267-0.2623-0.5746-0.3777-0.56070.293124-0.44	3	-0.4178	-0.3094	0.5414	-0.1962	-0.3167	0.4459			
12-0.6370-0.54770.6734-0.5478-0.54760.617918-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.4839Panel C Jaury 1990-DecemberFuel C Jaury 1990-DecemberValue-weightedValue-weighted3-0.5897-0.39400.0673-0.2758-0.47320.27486-0.5606-0.50940.1034-0.4968-0.55540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.3981-0.36950.405124-0.5113-0.50950.5237-0.3981-0.36950.40453-0.47410.0796-0.6227-0.1935-0.58890.31936-0.1768-0.3312-0.6004-0.2417-0.52000.207012-0.2585-0.3894-0.5674-0.3897-0.56070.293124-0.2585-0.3894-0.5674-0.3877-0.56070.293124-0.416-0.3058-0.5746-0.3777-0.56070.2931	6	-0.4518	-0.5334	0.6090	-0.2641	-0.4210	0.5902			
18-0.3810-0.49710.5790-0.4924-0.44190.565924-0.2854-0.44440.6343-0.1888-0.27890.4839Panel C Jaurey Pyo-DecemberFunder Staurey Pyo-DecemberHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.5897-0.39400.0673-0.2758-0.47320.27486-0.5060-0.50940.1034-0.4968-0.55540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.4339-0.45990.466224-0.5113-0.50950.5237-0.3981-0.36950.4045Panel D Jaure Store	12	-0.6370	-0.5477	0.6734	-0.5478	-0.5476	0.6179			
24         -0.2854         -0.4444         0.6343         -0.1888         -0.2789         0.4839           Panel C January 1990-December 1999           Fqual-weighted         Value-weighted           Value-weighted         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.5897         -0.3940         0.0673         -0.2758         -0.4732         0.2748           6         -0.5606         -0.5094         0.1034         -0.4968         -0.5554         0.3131           12         -0.7107         -0.4314         0.2344         -0.5175         -0.4752         0.4247           18         -0.6649         -0.6432         0.5149         -0.4339         -0.4599         0.4662           24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Panel D Januer 2000-December 2008           Equal-weighted         Mom/Size         Mom/Value         Size/Value           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227	18	-0.3810	-0.4971	0.5790	-0.4924	-0.4419	0.5659			
Panel C January 1990-December 1999           Equal-weighted         Value-weighted         Mom/Value         Size/Value           3         -0.5897         -0.3940         0.0673         -0.2758         -0.4732         0.2748           6         -0.5606         -0.5094         0.1034         -0.4968         -0.5554         0.3131           12         -0.7107         -0.4314         0.2344         -0.5175         -0.4752         0.4247           18         -0.6649         -0.6432         0.5149         -0.4339         -0.4599         0.4662           24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Size         Value-weighted           Fanel D January 2000-December 2008           Equal-weighted         Mom/Size         Mom/Value         Size/Value           Adom/Size         Mom/Value         Size/Value           Gold - 0.1768         -0.3312         -0.6004         -0.2417         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417	24	-0.2854	-0.4444	0.6343	-0.1888	-0.2789	0.4839			
Panel C January 1990-December 1999           Equal-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.5897         -0.3940         0.0673         -0.2758         -0.4732         0.2748           6         -0.5606         -0.5094         0.1034         -0.4968         -0.5554         0.3131           12         -0.7107         -0.4314         0.2344         -0.5175         -0.4752         0.4247           18         -0.6649         -0.6432         0.5149         -0.4339         -0.4599         0.4062           24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Panel D January 2000-December 2008           Equal-weighted           Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070 <t< td=""><td></td><td></td><td></td><td>1000 D</td><td>1 1000</td><td></td><td></td></t<>				1000 D	1 1000					
Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.5897         -0.3940         0.0673         -0.2758         -0.4732         0.2748           6         -0.5606         -0.5094         0.1034         -0.4968         -0.5554         0.3131           12         -0.7107         -0.4314         0.2344         -0.5175         -0.4752         0.4247           18         -0.6649         -0.6432         0.5149         -0.4339         -0.4599         0.4062           24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Fanel D January 2000-December 2008           Equal-weighted           Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070           12         -0.2585         -0.3894         -0.5674			Panel C Jai	nuary 1990-Decen	nber 1999					
Holding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.5897-0.39400.0673-0.2758-0.47320.27486-0.5606-0.50940.1034-0.4968-0.55540.313112-0.7107-0.43140.2344-0.5175-0.47520.424718-0.6649-0.64320.5149-0.4339-0.45990.466224-0.5113-0.50950.5237-0.3981-0.36950.4045Panel D January 2000-December 2008Equal-weightedValue-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.47410.0796-0.6227-0.1935-0.58890.31936-0.1768-0.3312-0.6004-0.2417-0.52000.207012-0.2585-0.3894-0.5674-0.3897-0.65710.265118-0.3267-0.2623-0.5746-0.3777-0.56070.293124-0.4416-0.3058-0.3675-0.4229-0.46180.3044			Equal-weighted	<i></i>		Value-weighted				
3         -0.5897         -0.3940         0.0673         -0.2758         -0.4732         0.2748           6         -0.5606         -0.5094         0.1034         -0.4968         -0.5554         0.3131           12         -0.7107         -0.4314         0.2344         -0.5175         -0.4752         0.4247           18         -0.6649         -0.6432         0.5149         -0.4339         -0.4599         0.4662           24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Fanel D January 2000-December 2008           Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070           12         -0.2585         -0.3894         -0.5674         -0.3897         -0.6571         0.2651           18         -0.3267         -0.2623         -0.5746         -0.3777         -0.4618         0.3044	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value			
6       -0.5606       -0.5094       0.1034       -0.4968       -0.5554       0.3131         12       -0.7107       -0.4314       0.2344       -0.5175       -0.4752       0.4247         18       -0.6649       -0.6432       0.5149       -0.4339       -0.4599       0.4662         24       -0.5113       -0.5095       0.5237       -0.3981       -0.3695       0.4045         Panel D January 2000-December 2008         Value-weighted         Holding months       Mom/Size       Mom/Value       Size/Value       Mom/Size       Mom/Value       Size/Value         3       -0.4741       0.0796       -0.6227       -0.1935       -0.5889       0.3193         6       -0.1768       -0.3312       -0.6004       -0.2417       -0.5200       0.2070         12       -0.2585       -0.3894       -0.5674       -0.3897       -0.65571       0.2651         18       -0.3267       -0.2623       -0.5746       -0.3777       -0.5607       0.2931         24       -0.4416       -0.3058       -0.3675       -0.4229       -0.4618       0.3044	3	-0.5897	-0.3940	0.0673	-0.2758	-0.4732	0.2748			
12       -0.7107       -0.4314       0.2344       -0.5175       -0.4752       0.4247         18       -0.6649       -0.6432       0.5149       -0.4339       -0.4599       0.4662         24       -0.5113       -0.5095       0.5237       -0.3981       -0.3695       0.4045         Panel D January 2000-December 2008         Equal-weighted       Value-weighted         Holding months       Mom/Size       Mom/Value       Size/Value       Mom/Size       Mom/Value       Size/Value         3       -0.4741       0.0796       -0.6227       -0.1935       -0.5889       0.3193         6       -0.1768       -0.3312       -0.6004       -0.2417       -0.5200       0.2070         12       -0.2585       -0.3894       -0.5674       -0.3897       -0.6571       0.2651         18       -0.3267       -0.2623       -0.5746       -0.3777       -0.5607       0.2931         24       -0.4416       -0.3058       -0.3675       -0.4229       -0.4618       0.3044	6	-0.5606	-0.5094	0.1034	-0.4968	-0.5554	0.3131			
18         -0.6649         -0.6432         0.5149         -0.4339         -0.4599         0.4662           24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Panel D January 2000-December 2008           Value-weighted           Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070           12         -0.2585         -0.3894         -0.5674         -0.3897         -0.6571         0.2651           18         -0.3267         -0.2623         -0.5746         -0.3777         -0.5607         0.2931           24         -0.4416         -0.3058         -0.3675         -0.4229         -0.4618         0.3044	12	-0.7107	-0.4314	0.2344	-0.5175	-0.4752	0.4247			
24         -0.5113         -0.5095         0.5237         -0.3981         -0.3695         0.4045           Panel D January 2000-December 2008           Fequal-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070           12         -0.2585         -0.3894         -0.5674         -0.3897         -0.6571         0.2651           18         -0.3267         -0.2623         -0.5746         -0.3777         -0.5607         0.2931           24         -0.4416         -0.3058         -0.3675         -0.4229         -0.4618         0.3044	18	-0.6649	-0.6432	0.5149	-0.4339	-0.4599	0.4662			
Panel D Januer 2000-December 2008           Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070           12         -0.2585         -0.3894         -0.5674         -0.3897         -0.6571         0.2651           18         -0.3267         -0.2623         -0.5746         -0.3777         -0.5607         0.2931           24         -0.4416         -0.3058         -0.3675         -0.4229         -0.4618         0.3044	24	-0.5113	-0.5095	0.5237	-0.3981	-0.3695	0.4045			
Equal-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.4741         0.0796         -0.6227         -0.1935         -0.5889         0.3193           6         -0.1768         -0.3312         -0.6004         -0.2417         -0.5200         0.2070           12         -0.2585         -0.3894         -0.5674         -0.3897         -0.6571         0.2651           18         -0.3267         -0.2623         -0.5746         -0.3777         -0.5607         0.2931           24         -0.4416         -0.3058         -0.3675         -0.4229         -0.4618         0.3044			Panel D Ja	nuary 2000-Decen	nber 2008					
Holding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.47410.0796-0.6227-0.1935-0.58890.31936-0.1768-0.3312-0.6004-0.2417-0.52000.207012-0.2585-0.3894-0.5674-0.3897-0.65710.265118-0.3267-0.2623-0.5746-0.3777-0.56070.293124-0.4416-0.3058-0.3675-0.4229-0.46180.3044			Equal-weighted			Value-weighted				
3       -0.4741       0.0796       -0.6227       -0.1935       -0.5889       0.3193         6       -0.1768       -0.3312       -0.6004       -0.2417       -0.5200       0.2070         12       -0.2585       -0.3894       -0.5674       -0.3897       -0.6571       0.2651         18       -0.3267       -0.2623       -0.5746       -0.3777       -0.5607       0.2931         24       -0.4416       -0.3058       -0.3675       -0.4229       -0.4618       0.3044	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value			
6-0.1768-0.3312-0.6004-0.2417-0.52000.207012-0.2585-0.3894-0.5674-0.3897-0.65710.265118-0.3267-0.2623-0.5746-0.3777-0.56070.293124-0.4416-0.3058-0.3675-0.4229-0.46180.3044	3	-0.4741	0.0796	-0.6227	-0.1935	-0.5889	0.3193			
12-0.2585-0.3894-0.5674-0.3897-0.65710.265118-0.3267-0.2623-0.5746-0.3777-0.56070.293124-0.4416-0.3058-0.3675-0.4229-0.46180.3044	6	-0.1768	-0.3312	-0.6004	-0.2417	-0.5200	0.2070			
18-0.3267-0.2623-0.5746-0.3777-0.56070.293124-0.4416-0.3058-0.3675-0.4229-0.46180.3044	12	-0.2585	-0.3894	-0.5674	-0.3897	-0.6571	0.2651			
24 -0.4416 -0.3058 -0.3675 -0.4229 -0.4618 0.3044	18	-0.3267	-0.2623	-0.5746	-0.3777	-0.5607	0.2931			
	24	-0.4416	-0.3058	-0.3675	-0.4229	-0.4618	0.3044			

Table 4.15 Correlation coefficient among style effects in terms of Long/Short return  $R_{\rm lms}$ 

Note: The table reports correlation coefficient among style effects in terms of long/short return  $R_{lms}$  from 1980 to 2008, and three subperiods. Long/short return  $R_{lms}$  are estimated from return spread of top/bottom decile portfolios in terms of momentum, size, and value effect.

Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.3661         -0.3074         0.2065         -0.3729         -0.3454         0.3716           6         -0.4498         -0.3238         0.2384         -0.4183         -0.3445         0.3716           12         -0.5175         -0.3648         0.2461         -0.519         -0.3832         0.4463           18         -0.5263         -0.3545         0.4457         -0.555         -0.2475         0.5429           24         -0.5344         -0.5327         0.4457         -0.2475         0.5429           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           10         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           11         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           12         -0.3143         -0.6249         0.5677         -0.2732         -0.5422         0.5242           12         -0.5173         -0.6465         0.5489         -0.3933		Panel A Overall period from 1980 to 2008										
Holding monthsMom/YaizeMom/YaizeSize/ValueMom/SizeMom/YaizeSize/Value3-0.3661-0.30740.2065-0.3729-0.32690.34056-0.4498-0.32380.2384-0.4133-0.34320.416312-0.5753-0.36480.2461-0.5139-0.38320.416318-0.5263-0.36450.2451-0.4555-0.24750.542924-0.5364-0.35450.4457-0.4555-0.24750.542924-0.5364-0.35450.4457-0.4555-0.24750.542925-0.3844-0.53270.5162-0.2734-0.46300.49766-0.4314-0.64590.5677-0.2732-0.54220.524218-0.3605-0.64550.5489-0.3936-0.49220.529218-0.3605-0.56330.5507-0.3736-0.36380.572524-0.3148-0.54570.5247-0.1701-0.09820.564819-0.318-0.54570.5247-0.1701-0.09820.564812-0.512-0.24300.1704-0.6384-0.46100.31686-0.6050-0.24300.1704-0.6384-0.461070.475912-0.6203-0.24300.1704-0.58390.44570.45512-0.6203-0.24300.1704-0.6384-0.46370.475918-0.6202-0.36110.4320-0.5544 <t< td=""><td></td><td></td><td>Equal-weighted</td><td></td><td></td><td>Value-weighted</td><td></td></t<>			Equal-weighted			Value-weighted						
3-0.3661-0.07240.2065-0.3729-0.32690.34056-0.4498-0.32380.2384-0.4183-0.34450.371012-0.5175-0.36480.2061-0.5139-0.38320.416318-0.5263-0.36560.3594-0.5256-0.38090.494124-0.5364-0.35450.4457-0.4555-0.24750.5429Panel B Januer J980-December J989Equal-weightetValue-weightedNom/SizeMom/YalueSize/Value3-0.3844-0.53270.5162-0.2734-0.46300.49766-0.4314-0.62490.5677-0.2732-0.54220.542412-0.5173-0.64650.5489-0.3993-0.49220.529218-0.3605-0.56330.5507-0.3736-0.36380.572524-0.3148-0.54570.5247-0.1701-0.09820.5648Holding monthsMom/SizeMom/ValueSize/Value3-0.5122-0.34170.0456-0.4314-0.30136-0.6030-0.24300.1704-0.6384-0.44370.376512-0.6203-0.24300.1704-0.6384-0.44370.475913-0.6203-0.24300.1704-0.6384-0.44370.475914-0.6203-0.24300.1704-0.6364-0.48590.515912-0.6203-0.2430 <td>Holding months</td> <td>Mom/Size</td> <td>Mom/Value</td> <td>Size/Value</td> <td>Mom/Size</td> <td>Mom/Value</td> <td>Size/Value</td>	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value					
6-0.4498-0.32380.2384-0.4183-0.34450.371012-0.5175-0.36480.2461-0.5139-0.38320.416318-0.5263-0.36560.3594-0.5256-0.38090.494124-0.5364-0.35450.4457-0.4555-0.24750.5429Holding monthsMom/SizeMom/YalueSize/ValueMom/SizeMom/ValueSize/Value3-0.3844-0.5270.5162-0.2734-0.46300.49766-0.4314-0.62490.5677-0.2732-0.54220.529218-0.3605-0.56330.5507-0.3736-0.36380.572524-0.3148-0.54570.5247-0.1701-0.09820.5648Holding monthsMom/SizeMom/ValueSize/ValueMom/ValueSize/ValueMom/ValueSize/Value40-0.3148-0.54570.5247-0.1701-0.09820.564824-0.3148-0.4300.0297-0.5893-0.46000.376512-0.6203-0.24300.1704-0.6384-0.44370.475918-0.6202-0.38610.4320-0.5542-0.58390.548524-0.4343-0.31290.4190-0.4650-0.47590.515918-0.6203-0.24300.1704-0.6384-0.41750.58170.518718-0.6203-0.2459-0.44516-0.3626-0.58170.1084 </td <td>3</td> <td>-0.3661</td> <td>-0.3074</td> <td>0.2065</td> <td>-0.3729</td> <td>-0.3269</td> <td>0.3405</td>	3	-0.3661	-0.3074	0.2065	-0.3729	-0.3269	0.3405					
12         -0.5175         -0.3648         0.2461         -0.5139         -0.3832         0.4163           18         -0.5263         -0.3555         0.4557         -0.5256         -0.3809         0.4941           24         -0.5364         -0.3555         0.4457         -0.4555         -0.2475         0.5429           24         -0.5364         -0.3545         0.4457         -0.4555         -0.2475         0.5429           24         -0.3844         -0.5327         0.5162         -0.2734         -0.4630         0.4976           6         -0.4314         -0.6249         0.5677         -0.2732         -0.5422         0.5292           18         -0.3055         -0.6665         0.5489         -0.3976         -0.3628         0.5725           24         -0.3148         -0.5457         0.5247         -0.1701         -0.0982         0.5648           12         -0.3148         -0.5457         0.5247         -0.1701         -0.0982         0.5648           140ding months         Mom/Size         Mom/Value         Size/Value         Mom/Value         Size/Value           12         -0.6105         -0.2403         0.0297         -0.5893         -0.4600         0.3765 <td>6</td> <td>-0.4498</td> <td>-0.3238</td> <td>0.2384</td> <td>-0.4183</td> <td>-0.3445</td> <td>0.3710</td>	6	-0.4498	-0.3238	0.2384	-0.4183	-0.3445	0.3710					
18-0.5263-0.36560.3594-0.5256-0.38090.494124-0.5364-0.35450.4457-0.4555-0.24750.542924-0.5364-0.35450.4457-0.4555-0.24750.5429Fanel B Janury 1980-December 1989Equal-weightedMom/ValueSize/ValueMom/ValueSize/ValueMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.3844-0.53270.5162-0.2734-0.46300.49766-0.4314-0.62490.5677-0.2732-0.54220.529218-0.3605-0.56330.5507-0.3736-0.36380.572524-0.3148-0.54570.5247-0.1701-0.09820.5648Panel C Janury 1990-December 1999Equal-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.5122-0.34170.0456-0.4729-0.36110.30186-0.6050-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24030.1704-0.6384-0.47590.515918-0.6202-0.38610.4190-0.4096-0.47590.515918	12	-0.5175	-0.3648	0.2461	-0.5139	-0.3832	0.4163					
24-0.5364-0.35450.4457-0.4555-0.24750.5429Panel B Jarrer 1980-December 1980Holding monthsMom/SizeKau-weightedValue-weightedHolding months0.3844-0.53270.5162-0.2734-0.46300.49766-0.4314-0.62490.5677-0.2732-0.54220.529218-0.3605-0.56330.5507-0.3736-0.36380.572524-0.3148-0.54570.5247-0.1701-0.09820.5648Panel C Jarrer 1990-December 1990Equal-weightedMom/SizeMom/ValueSize/ValueHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/Value3-0.5122-0.34170.0456-0.4729-0.36110.30186-0.6050-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24030.0297-0.5893-0.46000.376513-0.6202-0.38610.4120-0.5154-0.47590.515914-0.6203-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24030.0297-0.5893-0.46000.376514-0.6202-0.38610.4120-0.47590.515915	18	-0.5263	-0.3656	03594	-0.5256	-0.3809	0.4941					
Panel B J=1000000000000000000000000000000000000	24	-0.5364	-0.3545	0.4457	-0.4555	-0.2475	0.5429					
Equal-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.3844         -0.5327         0.5162         -0.2734         -0.4630         0.4976           6         -0.4314         -0.6249         0.5677         -0.2732         -0.5422         0.5292           12         -0.5173         -0.6465         0.5489         -0.3993         -0.4922         0.5292           18         -0.3605         -0.5633         0.5507         -0.3736         -0.3638         0.5725           24         -0.3148         -0.5457         0.5247         -0.1701         -0.0982         0.5648           Panel C January 1990-December 1999           Equal-weighted         Mom/Size         Mom/Value         Size/Value           Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.5122         -0.3417         0.0456         -0.4729         -0.3611         0.3018           6         -0.6050         -0.2403         0.0297         -0.5893         -0.4600         0.3765           12			Panel B Ja	nuary 1980-Decen	1989 1989							
Holding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3 $-0.3844$ $-0.5327$ $0.5162$ $-0.2734$ $-0.4630$ $0.4976$ 6 $-0.4314$ $-0.6249$ $0.5677$ $-0.2732$ $-0.5422$ $0.5424$ 12 $-0.5173$ $-0.6465$ $0.5489$ $-0.3993$ $-0.4922$ $0.5292$ 18 $-0.3605$ $-0.5633$ $0.5507$ $-0.3736$ $-0.3638$ $0.5725$ 24 $-0.3148$ $-0.5457$ $0.5247$ $-0.1701$ $-0.0982$ $0.5648$ Panel C Januer JP90-December J990Equal-weightedValue-weightedMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/ValueMom/SizeMom/Value3 $-0.5122$ $-0.3417$ $0.0456$ $-0.4729$ $-0.3611$ $0.3018$ 6 $-0.6020$ $-0.2403$ $0.297$ $-0.5893$ $-0.4600$ $0.3765$ 12 $-0.6202$ $-0.3861$ $0.4320$ $-0.5544$ $-0.5839$ $0.5485$ 24 $-0.4943$ $-0.3129$ $0.4190$ $-0.4096$ $-0.4759$ $0.5159$ Holding monthsMom/SizeMom/ValueSize/ValueHolding monthsMom/SizeMom/ValueSize/ValueAugust colspan="4">August colspan="4">			Equal-weighted	-		Value-weighted						
3.0.3844.0.5327.0.5162.0.2734.0.4630.0.49766.0.4314.0.6249.0.5677.0.2732.0.5422.0.542412.0.5173.0.6465.0.5489.0.3993.0.4922.0.529218.0.3605.0.5633.0.5507.0.3736.0.3638.0.572524.0.3148.0.5457.0.5247.0.1701.0.0982.0.5648Panel C Jaurure JPO-December JPO-Fanel C Jaurure JPO-December JPO-Value-weightedValue-weightedMom/SizeMom/ValueSize/Value6.0.6050.0.2430.0.297.0.5893.0.4600.0.376512.0.6203.0.2430.0.1704.0.6384.0.4437.0.475918.0.6202.0.3861.0.4320.0.5544.0.6393.0.5159Equal-weightedValue-weightedPanel D Jaurure JOO-December Jobs12.0.6203.0.2430.0.1704.0.6384.0.4437.0.475918.0.6202.0.3861.0.4320.0.5544.0.6393.0.5169.0.5191Holding monthsMom/SizeMom/ValueSize/ValueAugure ighted.0.2870.0.2870.0.1846.0.2870.0.2459.0.4170.0.6384.0.4630.0.5191Jaurure ighted.0.2870.0.2870.0.2870.0.2870.0.2870<	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value					
6-0.4314-0.62490.5677-0.2732-0.54220.524212-0.5173-0.64650.5489-0.3933-0.49220.529218-0.3605-0.56330.5507-0.3736-0.36380.572524-0.3148-0.54570.5247-0.1701-0.09820.5648Panel C Jaruer J990-December J999Value-weightedValue-weightedMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.5122-0.34170.0456-0.4729-0.36110.30186-0.6050-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24300.1704-0.6384-0.44370.475918-0.6202-0.38610.4320-0.5544-0.58390.548524-0.4943-0.31290.4190-0.4096-0.47590.5159Holding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/Value6-0.2670-0.38610.4320-0.5544-0.58390.548524-0.4943-0.31290.4190-0.4096-0.47590.5191Holding monthsMom/SizeMom/ValueSize/ValueMom/ValueSize/Value3-0.2013-0.2600-0.4876-0.3626-0.58170.10846-0.2870-0.2459-0.4516-0.3443-0.4870-0.008012<	3	-0.3844	-0.5327	0.5162	-0.2734	-0.4630	0.4976					
12 $-0.5173$ $-0.6465$ $0.5489$ $-0.3993$ $-0.4922$ $0.5292$ 18 $-0.3605$ $-0.5633$ $0.5507$ $-0.3736$ $-0.3638$ $0.5725$ 24 $-0.3148$ $-0.5457$ $0.5247$ $-0.1701$ $-0.0982$ $0.5648$ Panel C January 1990-December 1999Value-weightedValue-weightedMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3 $-0.5122$ $-0.3417$ $0.0456$ $-0.4729$ $-0.3611$ $0.3018$ 6 $-0.6050$ $-0.2403$ $0.0297$ $-0.5893$ $-0.4600$ $0.3765$ 12 $-0.6203$ $-0.2430$ $0.1704$ $-0.6384$ $-0.4437$ $0.4759$ 18 $-0.6202$ $-0.3861$ $0.4320$ $-0.5444$ $-0.5839$ $0.5485$ 24 $-0.4943$ $-0.3129$ $0.4190$ $-0.4096$ $-0.4759$ $0.5159$ Panel D January 2000-December 2008Value-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/Value3 $-0.2013$ $-0.2600$ $-0.4876$ $-0.3626$ $-0.5817$ $0.1084$ 6 $-0.2870$ $-0.2459$ $-0.4516$ $-0.3443$ $-0.4870$ $-0.0080$ 12 $-0.3489$ $-0.2446$ $-0.4785$ $-0.4218$ $-0.5282$ $0.0847$ 18 $-0.3640$ $-0.1227$ $-0.4599$ $-0.4475$ $-0.3979$ $0.1842$ <td>6</td> <td>-0.4314</td> <td>-0.6249</td> <td>0.5677</td> <td>-0.2732</td> <td>-0.5422</td> <td>0.5424</td>	6	-0.4314	-0.6249	0.5677	-0.2732	-0.5422	0.5424					
18 $-0.3605$ $-0.5633$ $0.5507$ $-0.3736$ $-0.3638$ $0.5725$ 24 $-0.3148$ $-0.5457$ $0.5247$ $-0.1701$ $-0.0982$ $0.5648$ Panel C Janury 1990-December 1999Value-weightedValue-weightedMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3 $-0.5122$ $-0.3417$ $0.0456$ $-0.4729$ $-0.3611$ $0.3018$ 6 $-0.6050$ $-0.2403$ $0.0297$ $-0.5893$ $-0.4600$ $0.3765$ 12 $-0.6203$ $-0.2430$ $0.1704$ $-0.6384$ $-0.4437$ $0.4759$ 18 $-0.6202$ $-0.3861$ $0.4320$ $-0.5544$ $-0.5839$ $0.5485$ 24 $-0.4943$ $-0.3129$ $0.4190$ $-0.4096$ $-0.4759$ $0.5159$ Panel D Janury 2000-December 2008Value-weightedHolding monthsMom/SizeMom/ValueSize/Value16 $-0.2870$ $-0.2459$ $-0.4876$ $-0.3626$ $-0.5817$ $0.1084$ 6 $-0.2870$ $-0.2459$ $-0.4516$ $-0.3443$ $-0.4870$ $-0.0080$ 12 $-0.3489$ $-0.2446$ $-0.4785$ $-0.4218$ $-0.5282$ $0.0847$ 18 $-0.3640$ $-0.1227$ $-0.4599$ $-0.4475$ $-0.3979$ $0.1842$ 24 $-0.5091$ $-0.0727$ $-0.1949$ $-0.4060$ $-0.2255$ $0.3435$	12	-0.5173	-0.6465	0.5489	-0.3993	-0.4922	0.5292					
24-0.3148-0.54570.5247-0.1701-0.09820.5648Panel C January 1990-December 1999Equal-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.5122-0.34170.0456-0.4729-0.36110.30186-0.6050-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24300.1704-0.6384-0.44370.475918-0.6202-0.38610.4320-0.5544-0.58390.548524-0.4943-0.31290.4190-0.4096-0.47590.5159Panel D January 2000-December 2008Equal-weightedValue-weightedHolding monthsMom/SizeMom/ValueSize/Value3-0.2013-0.2600-0.4876-0.3626-0.58170.10846-0.2870-0.2459-0.4516-0.3443-0.4870-0.008012-0.3489-0.2466-0.4785-0.4218-0.52820.084718-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435	18	-0.3605	-0.5633	0.5507	-0.3736	-0.3638	0.5725					
Panel C January 1990-December 1999           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.5122         -0.3417         0.0456         -0.4729         -0.3611         0.3018           6         -0.6050         -0.2403         0.0297         -0.5893         -0.4600         0.3765           12         -0.6203         -0.2430         0.1704         -0.6384         -0.4437         0.4759           18         -0.6202         -0.3861         0.4320         -0.5544         -0.5839         0.5485           24         -0.4943         -0.3129         0.4190         -0.4096         -0.4759         0.5159           Panel D January 2000-December 2008           Equal-weighted         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2600         -0.4876         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.00880           12         -0.3489         -0.2460         -0.4785         -0.4218         -0.5817         0.1084           6 </td <td>24</td> <td>-0.3148</td> <td>-0.5457</td> <td>0.5247</td> <td>-0.1701</td> <td>-0.0982</td> <td>0.5648</td>	24	-0.3148	-0.5457	0.5247	-0.1701	-0.0982	0.5648					
Fance Called (1997)           Equal-weighted         Value-weighted         Value-weighted           3         -0.5122         -0.3417         0.0456         -0.4729         -0.3611         0.3018           6         -0.6050         -0.2403         0.0297         -0.5893         -0.4600         0.3765           12         -0.6203         -0.2430         0.1704         -0.6384         -0.4437         0.4759           18         -0.6202         -0.3861         0.4320         -0.5544         -0.5839         0.5485           24         -0.4943         -0.3129         0.4190         -0.4096         -0.4759         0.5159           Panel D January 2000-December 2008           Equal-weighted           Holding months         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2600         -0.4876         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.0080           12         -0.3489         -0.2446         -0.4785         -0.4218         -0.5282         0.0847           18         -0.3640         -0.1227 </td <td colspan="11">Panal C January 1990 December 1999</td>	Panal C January 1990 December 1999											
Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.5122         -0.3417         0.0456         -0.4729         -0.3611         0.3018           6         -0.6050         -0.2403         0.0297         -0.5893         -0.4600         0.3765           12         -0.6203         -0.2430         0.1704         -0.6384         -0.437         0.4759           18         -0.6202         -0.3861         0.4320         -0.5544         -0.5839         0.5485           24         -0.4943         -0.3129         0.4190         -0.4096         -0.4759         0.5159           Fanel D January 2000-December 2008           Equal-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2600         -0.4876         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.0080           12         -0.3489         -0.2446         -0.4785         -0.42			Faual-weighted	indary 1990-Decen		Value-weighted						
Holding infoldingMont/SitzeMont/ValueSitz/ValueMont/SitzeMont/ValueSitz/Value3-0.5122-0.34170.0456-0.4729-0.36110.30186-0.6050-0.24030.0297-0.5893-0.46000.376512-0.6203-0.24300.1704-0.6384-0.44370.475918-0.6202-0.38610.4320-0.5544-0.58390.548524-0.4943-0.31290.4190-0.4096-0.47590.5159Panel D January 2000-December 2008Equal-weightedValue-weightedValue-weightedSize/ValueMom/SizeMom/ValueSize/Value3-0.2013-0.2600-0.4876-0.3626-0.58170.10846-0.2870-0.2459-0.4516-0.3443-0.4870-0.008012-0.3489-0.2446-0.4785-0.4218-0.52820.084718-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value					
5         10.5122         10.5417         10.0430         10.4427         10.5011         10.5016           6         -0.6050         -0.2403         0.0297         -0.5893         -0.4600         0.3765           12         -0.6203         -0.2430         0.1704         -0.6384         -0.4437         0.4759           18         -0.6202         -0.3861         0.4320         -0.5544         -0.5839         0.5485           24         -0.4943         -0.3129         0.4190         -0.4096         -0.4759         0.5159           Panel D January 2000-December 2008           Equal-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2459         -0.4516         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.0080           12         -0.3489         -0.2446         -0.4785         -0.4218         -0.5282         0.0847           18         -0.3601         -0.1227         -0.4599         -0.4475         -0.3979	3	-0 5122	-0 3417	0.0456	-0.4729	-0 3611	0 3018					
12       -0.6203       -0.2430       0.1704       -0.6384       -0.4437       0.4759         18       -0.6202       -0.3861       0.4320       -0.5544       -0.5839       0.5485         24       -0.4943       -0.3129       0.4190       -0.4096       -0.4759       0.5159         Panel D January 2000-December 2008         Equal-weighted         Mom/Size       Mom/Value       Size/Value         3       -0.2013       -0.2600       -0.4876       -0.3626       -0.5817       0.1084         6       -0.2870       -0.2459       -0.4516       -0.3443       -0.4870       -0.0080         12       -0.3489       -0.2459       -0.4516       -0.3443       -0.4870       -0.0080         12       -0.3489       -0.2459       -0.4516       -0.3443       -0.4870       -0.0080         12       -0.3489       -0.2459       -0.4516       -0.4218       -0.5282       0.0847         18       -0.3640       -0.1227       -0.4599       -0.4475       -0.3979       0.1842         24       -0.5091       -0.0727       -0.1949       -0.5409       -0.2225       0.3435	5	-0.5122	-0.2403	0.0297	-0.5893	-0.4600	0.3765					
12       -0.0203       -0.2450       -0.1104       -0.0504       -0.0304       -0.4457       -0.4457         18       -0.6202       -0.3861       0.4320       -0.5544       -0.5839       0.5485         24       -0.4943       -0.3129       0.4190       -0.4096       -0.4759       0.5159         Panel D January 2000-December 2008         Equal-weighted       Value-weighted         Holding months       Mom/Size       Mom/Value       Size/Value       Mom/Size       Mom/Value       Size/Value         3       -0.2013       -0.2600       -0.4876       -0.3626       -0.5817       0.1084         6       -0.2870       -0.2459       -0.4516       -0.3443       -0.4870       -0.0080         12       -0.3489       -0.2446       -0.4785       -0.4218       -0.5282       0.0847         18       -0.3640       -0.1227       -0.4599       -0.4475       -0.3979       0.1842         24       -0.5091       -0.0727       -0.1949       -0.5409       -0.2225       0.3435	12	-0.6203	-0.2430	0.1704	-0.6384	-0.4437	0.4759					
16       10.0202       10.0001	12	-0.6203	-0.2450	0.4320	-0.5544	-0.5839	0.5485					
24         -0.4743         -0.5123         -0.4730         -0.4733         -0.5133           Panel D January 2000-December 2008           Fqual-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2600         -0.4876         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.0080           12         -0.3489         -0.2446         -0.4785         -0.4218         -0.5282         0.0847           18         -0.3640         -0.1227         -0.4599         -0.4475         -0.3979         0.1842           24         -0.5091         -0.0727         -0.1949         -0.5409         -0.2225         0.3435	24	-0.0202	-0.3129	0.4190	-0.3544	-0.4759	0.5159					
Panel D January 2000-December 2008           Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2600         -0.4876         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.0080           12         -0.3489         -0.2446         -0.4785         -0.4218         -0.5282         0.0847           18         -0.3640         -0.1227         -0.4599         -0.4475         -0.3979         0.1842           24         -0.5091         -0.0727         -0.1949         -0.5409         -0.2225         0.3435		0.1915	0.312)	0.4190	0.4070	0.4759	0.0107					
Equal-weighted         Value-weighted           Holding months         Mom/Size         Mom/Value         Size/Value         Mom/Size         Mom/Value         Size/Value           3         -0.2013         -0.2600         -0.4876         -0.3626         -0.5817         0.1084           6         -0.2870         -0.2459         -0.4516         -0.3443         -0.4870         -0.0080           12         -0.3489         -0.2446         -0.4785         -0.4218         -0.5282         0.0847           18         -0.3640         -0.1227         -0.4599         -0.4475         -0.3979         0.1842           24         -0.5091         -0.0727         -0.1949         -0.5409         -0.2225         0.3435			Panel D Jar	nuary 2000-Decen	nber 2008							
Holding monthsMom/SizeMom/ValueSize/ValueMom/SizeMom/ValueSize/Value3-0.2013-0.2600-0.4876-0.3626-0.58170.10846-0.2870-0.2459-0.4516-0.3443-0.4870-0.008012-0.3489-0.2446-0.4785-0.4218-0.52820.084718-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435			Equal-weighted			Value-weighted						
3-0.2013-0.2600-0.4876-0.3626-0.58170.10846-0.2870-0.2459-0.4516-0.3443-0.4870-0.008012-0.3489-0.2446-0.4785-0.4218-0.52820.084718-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435	Holding months	Mom/Size	Mom/Value	Size/Value	Mom/Size	Mom/Value	Size/Value					
6-0.2870-0.2459-0.4516-0.3443-0.4870-0.008012-0.3489-0.2446-0.4785-0.4218-0.52820.084718-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435	3	-0.2013	-0.2600	-0.4876	-0.3626	-0.5817	0.1084					
12-0.3489-0.2446-0.4785-0.4218-0.52820.084718-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435	6	-0.2870	-0.2459	-0.4516	-0.3443	-0.4870	-0.0080					
18-0.3640-0.1227-0.4599-0.4475-0.39790.184224-0.5091-0.0727-0.1949-0.5409-0.22250.3435	12	-0.3489	-0.2446	-0.4785	-0.4218	-0.5282	0.0847					
24 -0.5091 -0.0727 -0.1949 -0.5409 -0.2225 0.3435	18	-0.3640	-0.1227	-0.4599	-0.4475	-0.3979	0.1842					
	24	-0.5091	-0.0727	-0.1949	-0.5409	-0.2225	0.3435					

Table 4.16 Correlation coefficient among style effects in terms of style coefficient  $\gamma$ 

Note: The table reports correlation coefficient among style effects in terms of style coefficient  $\gamma$  from 1980 to 2008, and three subperiods. Style coefficients are estimated from monthly-on-monthly cross-sectional regression of decile portfolio returns on firm characteristics such as lagged 12 months' share return, market value, and price-to-book ratio.

	Panel A Momentum effect (6 months' formation length)										
	A1 Equal-weighted										
HoldingWinner/Loser Return $R_{wml}$ Momentum coefficient $\gamma_{mom}$ Corr											
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient		
3	0.02	3.51	24.43	-22.86	0.0089	0.0629	0.4055	-0.9246	0.9034		
6	0.54	2.35	14.47	-10.85	0.0415	0.0785	0.6707	-1.1212	0.9209		
12	0.56	1.83	5.14	-7.81	0.0844	0.1985	0.5865	-1.2591	0.9093		
18	0.20	1.62	7.10	-7.60	0.0581	0.2690	1.4133	-1.4823	0.9091		
24	0.12	1.59	7.64	-17.69	0.0505	0.3392	2.0411	-2.8178	0.8944		

### Table 4.17 Statistics on Long/Short return $R_{\text{lms}}$ and style coefficient $\gamma$

	A2 Value-weighted										
Holding	W	inner/Loser	Return R <sub>wml</sub>		Me	omentum co	efficient $\gamma_{mo}$	m	Correlation		
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient		
3	0.76	4.87	17.68	-24.45	0.0208	0.1375	0.4913	-1.2283	0.9062		
6	0.94	3.19	11.95	-17.03	0.0512	0.1777	0.6194	-1.1378	0.9063		
12	0.82	2.35	7.02	-13.45	0.1026	0.2637	0.7737	-1.4299	0.8949		
18	0.34	1.98	7.91	-10.18	0.0775	0.3207	1.6109	-1.4482	0.8808		
24	0.31	1.63	7.95	-7.36	0.0890	0.3551	2.0833	-1.2139	0.8783		

## Panel B Momentum effect (12 months' formation length)

				B1 Ec	qual-weighte	d			
Holding	W	inner/Loser	Return R <sub>wml</sub>		M	omentum co	efficient $\gamma_{mo}$	m	Correlation
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient
3	0.51	3.23	15.35	-18.80	0.0159	0.0708	0.1908	-0.7627	0.8778
6	0.74	2.31	7.22	-11.33	0.0366	0.0993	0.3016	-0.8494	0.8982
12	0.41	1.91	4.78	-8.31	0.0443	0.1482	0.4528	-1.0752	0.8834
18	0.16	1.68	7.98	-9.64	0.0285	0.1986	1.1202	-1.1304	0.8940
24	-0.09	1.83	10.55	-18.01	-0.0062	0.2774	2.0580	-2.8956	0.9244

#### B2 Value-weighted Winner/Loser Return R<sub>wml</sub> Momentum coefficient $\gamma_{mom}$ Holding Correlation months coefficient Mean Std Max Min Mean Std Max Min 3 5.15 20.71 0.0224 0.0982 0.4729 -0.7601 0.8914 0.97 -24.88 6 0.98 3.42 10.55 -16.20 0.0433 0.1239 0.4010 -0.6086 0.9053 12 2.55 0.0547 0.1803 0.5590 -0.8794 0.8848 0.55 7.06 -13.53 18 0.0447 0.8538 0.23 2.41 6.70 -25.33 0.2218 0.9478 -1.4038 0.0341 24 0.11 1.80 5.23 -13.35 0.2461 1.1327 -0.9351 0.8426

#### Continued

	Panel C Size effect										
	C1 Equal-weighted										
Holding	Small/Large Return $R_{sml}$ Size coefficient $\gamma_{mv}$										
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient		
3	0.10	3.49	20.15	-8.65	0.0003	0.0137	0.0314	-0.0629	0.9511		
6	0.18	2.78	12.39	-5.86	0.0001	0.0218	0.0495	-0.0928	0.9577		
12	0.31	2.33	10.53	-3.97	0.0015	0.0359	0.0670	-0.1539	0.9719		
18	0.41	1.94	7.85	-4.00	0.0036	0.0459	0.0869	-0.1739	0.9788		
24	0.44	1.78	7.96	-3.08	0.0052	0.0548	0.1073	-0.2126	0.9802		

	C2 Value-weighted											
Holding	S	mall/Large	Return R <sub>sml</sub>			Size coeffi	cient $\gamma_{mv}$		Correlation			
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient			
3	-0.06	3.24	12.54	-8.82	0.0006	0.0129	0.0324	-0.0487	0.9465			
6	0.06	2.54	9.38	-5.65	0.0006	0.0203	0.0507	-0.0799	0.9543			
12	0.20	2.05	8.62	-4.12	0.0001	0.0326	0.0673	-0.1311	0.9683			
18	0.28	1.71	5.23	-3.59	0.0013	0.0415	0.0856	-0.1310	0.9750			
24	0.30	1.55	5.79	-3.05	0.0018	0.0495	0.1071	-0.1604	0.9788			

	Panel D Value effect										
	D1 Equal-weighted										
Holding	]	Low/High R	eturn R <sub>lmh</sub>			Value coef	ficient y <sub>pb</sub>		Correlation		
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient		
3	1.25	2.64	12.44	-12.39	0.0015	0.0069	0.0348	-0.0350	0.7922		
6	1.12	2.17	6.46	-12.38	0.0028	0.0112	0.0575	-0.0331	0.7724		
12	0.98	1.86	7.66	-7.61	0.0062	0.0168	0.0782	-0.0788	0.8038		
18	0.97	1.77	7.33	-7.84	0.0103	0.0229	0.0860	-0.0881	0.8059		
24	0.99	1.55	6.31	-4.23	0.0156	0.0303	0.0708	-0.1399	0.8089		

	D2 Value-weighted										
Holding		Low/High R	eturn R <sub>lmh</sub>			Value coef	ficient $\gamma_{pb}$		Correlation		
months	Mean	Std	Max	Min	Mean	Std	Max	Min	coefficient		
3	0.93	3.69	12.27	-11.71	0.0011	0.0072	0.0409	-0.0361	0.6529		
6	0.96	2.78	8.99	-6.89	0.0021	0.0126	0.0706	-0.0463	0.6303		
12	1.01	2.16	10.05	-4.61	0.0054	0.0166	0.0742	-0.0583	0.6802		
18	1.04	1.90	6.62	-3.33	0.0092	0.0223	0.0650	-0.0888	0.7158		
24	1.03	1.72	6.44	-2.40	0.0134	0.0282	0.0438	-0.1191	0.7562		

Note: The table reports statistics on Long/Short return  $R_{lms}$  and style coefficient  $\gamma$  of three style effects with 3, 6, 12, 18, and 24 months' holding horizons. Long/Short return  $R_{lms}$  is return spread between top/bottom portfolios sorted by lagged 6 and 12 months' return, market value, and price-to-book ratio. Style coefficient  $\gamma$  is the slope coefficient of cross-sectional regression of decile portfolios' returns in the holding period on their average values of firm characteristics in the formation period. Correlation coefficient is estimated by series Long/Short return  $R_{lms}$  and style coefficient  $\gamma$ .

Equal-weighted										
		+GDP(Y)		-GDP(Y)						
	Mean (%)	t-stat	+%	Mean (%)	t-stat	+%				
Momentum effect (12/12)	-0.11	-0.54	60.9	1.40	3.39	87.7				
Size effect(12)	0.68	3.86	52.5	-0.07	-0.38	41.8				
Value effect(12)	1.09	2.99	79.3	0.91	3.15	80.7				
		¥7.1 '. 1								
Value-weighted										
	+GDP(Y)			-GDP(Y)						
	Mean (%)	t-stat	+%	Mean (%)	t-stat	+%				
Momentum effect (12/12)	-0.14	-0.63	54.2	1.23	2.65	76.7				
Size effect(12)	0.52	3.46	56.9	-0.14	-0.80	42.4				
Value effect(12)	1.13	3.18	68.2	0.89	2.98	67.6				

# Table 4.18 The performance of monthly Long/Short return R<sub>lms</sub> in positive and negative annual change of GDP from 1980 to 2008

Note: The table reports average monthly Long/Short return  $R_{lms}$ , t-statistics, and the percentage of observations with positive return in positive and negative annual change of GDP from 1980 to 2008. Long/Short return  $R_{lms}$  is return spread between top/bottom portfolios sorted by lagged 12 months' return, market value, and price-to-book ratio when long/short portfolio is hold for 12 months.

# Table 4.19 The performance of monthly Long/Short return $R_{ims}$ in positive and negative lagged one-year's market return from 1980 to 2008

		Equal-weigh	ited				
	+MR(-1)			-MR(-1)			
	Mean (%)	t-stat	+%	Mean (%)	t-stat	+%	
Momentum effect (12/12)	0.58	1.38	73.4	0.68	1.85	73.9	
Size effect(12)	0.30	1.21	52.1	0.35	1.38	47.2	
Value effect(12)	0.91	3.01	79.4	1.26	3.58	78.3	
		Value-weigh	ited				
	+MR(-1)			-MR(-1)			
	Mean (%)	t-stat	+%	Mean (%)	t-stat	+%	
Momentum effect (12/12)	0.22	0.40	63.8	0.52	1.78	64.0	
Size effect(12)	0.19	1.52	49.4	0.43	1.81	57.9	
Value effect(12)	0.92	3 56	67.8	1 38	3 10	69.6	

Value effect(12)0.923.5667.81.383.1969.6Note: The table reports average monthly Long/Short return  $R_{lms}$ , t-statistics, and the percentage of observations<br/>with positive return in lagged one-year's upward market and downward market from 1980 to 2008. Long/Short<br/>return  $R_{lms}$  is return spread between top/bottom portfolios sorted by lagged 12 months' return, market value, and<br/>price-to-book ratio when long/short portfolio is hold for 12 months. One-year's market return is calculated by<br/>monthly FTSE ALL SHARE PRICE from LSPD.

Figure 4.1 Equal-weighted monthly Small/Large return  $R_{\rm sml}$  and Low/High return  $R_{\rm lmh}$  with 12 months holding horizon from 1980 to 2008





Note: The figure plots series equal-weighted monthly Small/Large return  $R_{sml}$  and Low/High return  $R_{lmh}$  from 1980 to 2008, which is estimated by the spread in 12 months' holding return between 'Small-cap' portfolio and 'Large-cap' portfolio, between 'Low' portfolio and 'High' portfolio.

# Chapter 5 The Impact of Macroeconomic variables on the Time-Variation in Style Effects

Based on decile momentum portfolios, decile size portfolios, and decile value portfolios, the previous chapter uses two measures, long/short return  $R_{lms}$  and style coefficients  $\gamma$ , to find time-varying momentum effect, size effect, and value effect in the UK Stock Market. This chapter mainly investigates what factors lead to the time-variation in style effects, and how they can capture these dynamic properties. It explores in more depth macroeconomic conditions and market states under which dynamic style effects occur over time in the UK Stock Market. The remainder of this chapter is structured as followings. Section 5.1 offers the description on macroeconomic variables, which are assumed to have influence on time-varying style effects in the UK Stock Market. Section 5.2 reviews research methods. Section 5.3, 5.4, and 5.5 presents data analysis on the impact of macroeconomic variables on the time-variation in momentum effect, size effect, and value effect, respectively. Section 5.6 concludes empirical work.

#### **5.1 Macroeconomic variables**

As reviewed in Chapter 2, some previous researchers pay attention to the different performance of firms/ and shares with specific characteristics in business cycle. For example, Tamari (1984), Gertler and Gilchrist (1994) find that small firms contracts

much more than large firms in economic recessions and tightening monetary policy. Jensen, Johnson, and Mercer (1998) find stronger (weaker) small-cap premium and value premium due to expansionary (tightening) monetary policy. Chordia and Shivakumar (2002), Antonious, Lam, and Paudyal (2007) conclude positive momentum payoffs during economic expansions and negative ones during economic recessions. Cooper, Gutierrez, and Hameed (2004), Avramov and Chordia (2006) present the positive relation between momentum profits and market return in lagged one to three years. Asness, Moskowitz, and Pedersen (2009) show value premium and momentum premium are positively linked to long-run consumption growth and negatively linked to economic recessions. In Chapter 4, the research finds the different performance in Winner/Loser return  $R_{wml}$  and Small/Large return  $R_{sml}$  between positive and negative annual change of GDP when long/short position is hold for 12 months These findings suggest that cross-sectional risk premium in momentum, size, and value sorted portfolios are exposure to business cycle.

The research assumes that time-varying style effects may be captured by some macroeconomic variables, which are associated with economic conditions and market states. They are measured by Growth of Domestic Production (GDP), Industrial Production/Manufacturing (IP/IPM), Money Supply (M0), Unemployment Rate (UNE), Inflation Rate (CPI), 3-month Short-term Rate (TBILL), Term Structure (TERM), Dividend Yield (DY), and Market Return (MR). The rationale for using these macroeconomic variables to capture the time-variation in style effects is that they are quantitative indicators straightforwardly measuring economic conditions and market states. Based on the findings in previous studies, the research hypothesizes
the relation between macroeconomic variables and the time-variation in style effects in the UK Stock Market, which is shown in Panel B of Table 5.1. The time-variation in momentum effect is hypothesized to have positive relation to GDP, IP, IPM, and MR(-y). The time-variation in size effect and value effect is hypothesized to have positive relation to GDP, IP, IPM and M0, and negative relation to CPI and TBILL.

The research collects series GDP and Term Structure (TERM) from ESDS IFS (The Economic and Social Data Service, International Financial Statistics), series Industrial Production (IP) and Industrial Production Index – Manufacturing (IPM), Money Supply (M0), Unemployment Rate (UNE), and CPI from Datastream. Because the research investigates the impact of macroeconomic variables on time-varying style effects at the beginning of January 1980 and Datastream offers data on money supply M1 and M2 since September 1986 and June 1982, it collects M0 as an indicator of money supply. The correlation between M0 and M1 (M2) is 0.9932 (0.9940) since September 1986, so M0 can reflect the pattern of money supply M1 and M2. 3-month Short-term Interest Rate (TBILL) and Dividend Yield (DY) are available from LSPD. Market Return (MR) in the previous horizons from one year to eight years is calculated by monthly FTSE ALL SHARE PRICE from LSPD.

When the research investigates the impact of macroeconomic variables on timevarying style effects, it collects them in corresponding lag length in order to ensure the publication of their data be available in advance when portfolios are established. The lag length for GDP is two quarters, for M0, CPI, TBILL, TERM, DY, and EMR/VMR is one month, for IP, IPM, and UNE is two months. Panel A of Table 5.1 summaries the description on macroeconomic variables, their lagged length, and data sources.

### [Table 5.1]

#### 5.1.1 Derived primary macroeconomic variables

There are two reasons to derive innovations in primary macroeconomic variables in terms of their previous change and unexpected change over different horizons when the research investigates the impact of macroeconomic variables on the timevariation in style effects.

Firstly, Ferson, Sarkissian, and Simin (2003) point out that some standard economic variables, such as dividend yield, short-term interest rate, term structure, and default spread, behave as persistent or highly autocorrelated time series. Highly persistent series are more likely to be found significant in the search for predictor variables. They cast doubt on the significance of term spread in Fama and French (1989) on the basis of either t-stats or R-square of the regression, the significance of dividend yield in Fama and French (1988b) on the basis of one but not both statistics. The simulation by Ferson, Sarkissian, and Simin (2003, 2006) shows that many regressions in the literature may be spurious due to individual predictor variable with highly-autocorrelation. They find that the extent of the spurious regression bias depends on

the parameters of the first-order autocorrelation. Therefore, the research uses differenced data of macroeconomic variables to avoid the complications associated with spurious regressions.

Panel A of Table 5.2 shows the first to tenth order autocorrelation of primary macroeconomic variables in the research, which include GDP, Industrial Production/Manufacturing (IP/IPM), Money Supply (M0), Unemployment Rate (UNE), Inflation Rate (CPI), Short-term Rate (TBILL), Term Structure (TERM), and Dividend Yield (DY). All of these macroeconomic variables have first order autocorrelations of 0.95 or higher, suggesting a high degree of persistence. In order to lower the risk of a spurious regression caused by highly autocorrelated macroeconomic variables, the research uses previous change and unexpected change of these primary macroeconomic variables over different horizons.

# [Table 5.2]

With the Augmented Dickey-Fuller (ADF) test, Panel B of Table 5.2 reports the unit root statistics for primary macroeconomic variables. The ADF statistics for macroeconomic variables, except for CPI, are below the 5% critical value. The ADF test fails to reject the null hypothesis of unit root in the levels for most macroeconomic variables. It rejects the null hypothesis of non-stationary in the levels for CPI. The unit root test presents that most primary macroeconomic variables are I (1) process. Secondly, it is rational to examine the relation between the innovations in primary macroeconomic variables and the time-variation in style effects. Because the research investigates the reason for time-varying style effects with different holding horizons, such as 3, 6, 12, 18, and 24 months, it is rational to assume that the previous change or unexpected change of macroeconomic variables over different horizons have different explanatory power for style coefficient  $\gamma$  and style long/short return R<sub>lms</sub> with different holding horizons. For example, if expected annual change of TBILL might offer significant impact on time-varying size effect with 24 months' holding length, it might not have the same explanatory power for size effect with 3 and 6 months' holding period.

The research uses the differenced natural logarithms in equation (5.1) or equation (5.1') to calculate the previous change of primary macroeconomic variables (MEV) over past the  $1^{st}$ ,  $3^{rd}$ ,  $6^{th}$ , and  $12^{th}$  month as an indicator of the current economic condition and market state:

$$MEV(i)_{t-L} = \log[MEV_{t-L} / MEV_{t-L-i}]$$
(5.1)

or

$$MEV(i)_{t-L} = MEV_{t-L} / MEV_{t-L-i} - 1$$
 (5.1')

where MEV(i) is the change of macroeconomic variable over past i months.  $MEV_{t-L}$  is the level of primary macroeconomic variables for month t-L (quarter t-L for GDP). 'L' stands for lag length of macroeconomic variables to ensure them known as explanatory variables in advance in the forecasting model. Except for lag length of 2 for GDP, IP, IPM, and UNE, lag length for other variables is set to 1. 'i' stands for 1, 3, 6, and 12 months. The research uses equation (5.1) to estimate the change of macroeconomic variables over time, except for GDP and TERM. Because the value of GDP and TERM is negative at some time points, the research uses equation (5.1') to estimate their change over time. For instance, in the sample period from 1980 to 2008, UK economy experienced two periods of negative growth from the 4<sup>th</sup> quarter in 1980 to the 1<sup>st</sup> quarter in 1982, and from the 2<sup>nd</sup> quarter in 1991 to the 4<sup>th</sup> quarter 1992. Term Structure is negative in five periods, including 1980, March 1985 to September 1986, September 1988 to October 1992, November 1997 to June 2001, and February 2005 to December 2008.

The test on monthly data by Fama and Gibbons (1984) indicates that expected real return on one-month treasury bill is estimated as its simple average of the twelve most recent realized real returns. The research follows Fama and Gibbons (1984) methodology, which is used in Chen, Roll, and Ross (1986), Hamao (1988), Zhang, Hopkins, Satchell, and Schwob (2009), to derive unexpected macroeconomic variables. It uses equation (5.2) to estimate the difference between macroeconomic variable (MEV) at month t-L and its moving average in past 3, 6, and 12 months (4 quarters for GDP), and regards it as a proxy for the unexpected macroeconomic variables (UEXMEV).

$$UEXMEV(i)_{t-L} = MEV_{t-L} - (\sum_{j=1}^{N} MEV_{t-L-j}) / N$$
(5.2)

where UEXMEV(i) is unexpected macroeconomic variables with i months. N stands for 3, 6, and 12 months (4 quarters for GDP). 'L' is lag length described as in equation (5.1).

#### 5.1.2 Three kinds of macroeconomic variables

The research assumes that three kinds of macroeconomic variables might drive the time-variation in momentum, size, and value effect in the UK Stock Market. They are associated with economic states, the bond market, and the stock market, respectively.

• (Unexpected) change of macroeconomic variables associated with economic states

Some studies focus on relation between economic forces and stock returns based on one special firm characteristics. Jensen, Johnson, and Mercer (1998) show that premiums of small-cap stocks and value stocks are closely dependent to the monetary environment. Perez-Quiros and Timmermann (2000) find that the return on small-cap portfolio is most strongly associated with economic variables, such as short interest rate, default premium, and money growth, among decile-sized portfolios. Asness, Moskowitz, and Pedersen (2008) examine different performance of momentum strategy and value strategy in business cycle. Levis and Liodakis(1999) find the impact of inflation level on small-cap stocks and value stocks. Based on the findings of previous studies, the research assumes the link between the time-variation in style effects and some derived primary economic variables. They include GDP(Y) (the annual previous change of GDP), UEXGDP(Y) (unexpected annual change of GDP), IP(i) (the change of Index of all Industrial production over previous the i<sup>th</sup> month), UEXIP(i) (unexpected change of IP over previous i months), IPM(i) (the change of Industrial Production Index – Manufacturing over previous the i<sup>th</sup> month), UEXIPM(i) (unexpected change of IPM over previous i months), M0(i) (the change of M0 over previous the i<sup>th</sup> month), UEXM0(i) (unexpected change of M0 over previous i months), UNE(i) (the change of Unemployment Rate over previous the i<sup>th</sup> months), UEXUNE(i) (unexpected change of UNE over previous i months), CPI(i) (the change of Consumer Price Index over previous the i<sup>th</sup> month), and UEXCPI(i) (unexpected change of CPI over previous i months). They are regarded as the first kind of macroeconomic variables which mainly reflects current and unexpected change of economic conditions.

• (Unexpected) change of macroeconomic variables associated with the bond market

Fama and French (1989) finds that term spread always exhibits low around businesscycle peaks and high near troughs, and that its explanatory power for stock returns tends to increase with the return horizon out to one or two years. Chen (1991) finds that short-term interest rates and term structure are associated with the quarterly growth rate of GNP in the next 4 and 5 quarters, respectively, and further have explanatory power for the next 1 and 2 quarterly stock returns. Based on these findings, the research assumes that time-varying style effects might be influenced by the second kind of macroeconomic variables, which are associated with the bond market. They consists of TBILL(i) (the change of TBILL over previous the i<sup>th</sup> month), UEXTBILL(i) (unexpected change of TBILL over previous i months), TERM(i) (the change of TERM over pervious the i<sup>th</sup> month), and UEXTERM(i) (the unexpected change of TERM over previous i months). Term structure (TERM) is the difference in the annualized yield on the long-term UK government bonds and the yield on the three-month Treasury TBILL.

• (Unexpected) change of macroeconomic variables associated with the stock market

Fama and French (1988a, 1989) find dividend yield is related to long-term business episodes that span several measured business cycles. Chen ((1991) shows the relation between dividend yield and the next quarterly growth of GNP, and further presents the explanatory power of dividend yield for following expected return in different horizons. So, the research assumes the impact of dividend yield, which is based on value-weighted market index of all companies, on the time-variation in style effects. It classifies DY(i) (the change of DY over pervious the i<sup>th</sup> month), and UEXDY(i) (unexpected change of DY over previous i months) into the third kind of macroeconomic variables, which are associated with the stock market.

Keim and Stambaugh (1986) concludes that although security returns are significantly impacted by a number of economic factors, the market return remains the dominant factor and plays a major role in pricing the US and UK Stock Market. The market return seems to incorporate most of information in the underlying factors. Levis and Liodakis(1999), Fletcher (2001) use lagged market excess return to test expected returns. Cooper, Gutierrez, and Hameed (2004) find that six-month momentum profit is positively related to the lagged market return at its low level, and this relation diminishes at its high level. In addition, cognitive biases inherent in investors implies that risk preference is volatile in the changing aggregate stock market, high in upward price trend but low in downward price trend. As Chapter 1 mentions, the interaction of cross-sectional variation in cash-flow risks of firms and time-varying risk preference leads to dynamic style effects. Therefore, the research uses EMR/VMR (equal/value-weighted market return) to reflect the present price level of aggregate stock market relative to its past level in different horizons. It uses monthly share returns from LSPD to estimate equal-(value-) weighted n months' compounded return (n=12, 24, 36, 48, 60, 72, 84, 96, 108, and 120) as EMR(y)/VMR(-y) in different horizons, where y denotes the number of year. The correlation coefficients between monthly EMR, VMR and monthly return of FTSE ALL SHARE, which is estimated with series values of FTSE ALL SHARE in LSPD, are 0.78 and 0.99, respectively.

Simulation results in Ferson, Sarkissian, and Simin (2003) find that spurious regression bias does not arise to any serious degree if first autocorrelation is 0.90 or less. Table 5.3 shows that after excluding variables with UNE, 85 percent of (unexpected) change of macroeconomic variables have the first-order autocorrelation below 0.90 or lower. After using (unexpected) change of macroeconomic variables,

the research lowers the first-order autocorrelation of explanatory variables in the regression on time-varying style effects, and the risk of spurious regression.

# [Table 5.3]

Table 5.4 reports correlation of annual (unexpected) change of macroeconomic variables from 1980 to 2008.

[Table 5.4]

# **5.2 Research methods**

Maddala (1977) presents one case of varying-parameter models that explanatory variables for changes in the parameters are known. He argues that in actual practice if  $z_t$  is a policy variable and if a relation like (5.3) is obtained by the maximization behaviour of firms or individuals, then  $z_t$  should not enter equation (5.3) additively. It should be entering as a determining variable for the parameters as in equation (5.4).

$$y_t = \beta_t * x_t + u_t \tag{5.3}$$

$$\beta_t = \alpha + \delta * z_t + \upsilon_t \tag{5.4}$$

Similar to the spirit in Maddala (1977), other studies investigate time-varying regression parameter alpha or beta in conditional asset pricing model. Shanken (1990) specifies the relation between market  $beta(\beta_m)$  and short-term rate(TB), its volatility(TBV), and a dummy variable with  $\beta_m = c_0 + c_{1*}TB + c_{2*}TBV + c_{3*}JAN$ . Ferson and Schadt (1996) use the model  $\beta_{pm}$  ( $Z_t$ )= $b_{0p}+b_p*Z_t$  to measure fund strategy and performance in changing economic conditions, where  $\beta_{pm}$  is portfolio beta, and  $b_p$  are the response coefficients of the conditional beta with respect to the macroeconomic variables  $Z_t$ . Jagannathan and Wang (1996) investigates the time-variation in betas to explain the cross-sectional returns, and find that size effect and statistical rejection of CAPM model specifications become weaker. Christopherson, Ferson, and Glassman (1998) use time-varying intercept or conditional alpha, which is a function of public

macroeconomic variables Z as  $\alpha_t = d_0 + d_{1*}Z_t$ , to measure abnormal performance of mutual funds.

In real investing field, style-oriented investors are indeed maximizing the profitability of their portfolios by using public macroeconomic variables in their decisions of which kind of stocks with specific firm characteristics are included in their portfolios. Macroeconomic variables should be entering the model not in an additive fashion but as determinants of time-varying style effects in the model (5.4). Thus, the research explores the relation between some macroeconomic variables and time-varying style effects in the UK Stock Market.

# 5.2.1 Univariate regression to extract explanatory variables

In pervious chapter, the research estimates style coefficient  $\gamma$  and long/short return  $R_{lms}$  to measure the time-variation in style effects in the UK Stock Market. In this chapter, the research assumes that a set of (unexpected) change of lagged macroeconomic variables might have explanatory power for time-varying style effects. Because 68 innovations with different horizons are estimated from 10 primary macroeconomic variables, it is rational to use univariate regression to screen some most significant innovations for the time-variation in style effects. Univariable regression is used to measure the relation between style coefficient  $\gamma$ , long/short return  $R_{lms}$  and each (unexpected) change of lagged macroeconomic variable. The research uses the innovations screened in univariate regression as explanatory

variables in the following multiple regression for  $\gamma$  and R<sub>lms</sub>. These explanatory variables can be assumed to capture the common movements in prices of shares with similar firm characteristics. The research uses univariate regression model (5.5) and (5.5') to extract most important explanatory factors for time-varying style coefficient  $\gamma$  and long/short return R<sub>lms</sub> from January 1980 to December 2008.

$$\gamma_t = \lambda_0 + \lambda_{IV} * MEV(i)_{t-L} + e_t \tag{5.5}$$

$$\gamma_t = \lambda_0 + \lambda_{IV} * UEXMEV(i)_{t-L} + e_t$$

and

$$R_{lms,t} = \lambda_0 + \lambda_{IV} * MEV(i)_{t-L} + e_t$$
(5.5')

$$R_{lms,t} = \lambda_0 + \lambda_{IV} * UEXMEV(i)_{t-L} + e_t$$

t = 1, 2, ..., n

where  $\gamma_t$  is style coefficient from monthly-by-monthly cross-sectional regression of portfolios' holding returns on their average value of firm characteristics in the formation period, in terms of lagged share return, market value, and price-to-book ratio, respectively. R<sub>lms</sub> is the difference in return between 'Winner' portfolio and 'Loser' portfolio, between 'Small' portfolio and 'Large' portfolio, and between 'Low' portfolio and 'High' portfolio. They are estimated in Chapter 4. MEV(i) and UEXMEV(i) are the previous change and unexpected change of primary macroeconomic variable at monthly, quarterly, half-yearly, and annual horizons, which are related to macroeconomic conditions, the bond market, and the stock market. 'L' denotes lag length for macroeconomic variables available in advance when portfolios are constructed. Primary macroeconomic variables include GDP, IP(Index of all Industrial production), IPM(Industrial Production Index – Manufacturing), M0(Money Supply), UNE (Unemployment Rate), CPI(Consumer Price Index), TBILL(monthly 3 months' interest rate), TERM(Term Structure), DY(Dividend Yield), and EMR/VMR(equal-/value-weighted market return). The estimation of (unexpected) change of primary macroeconomic variables at different horizons and corresponding lag length is described in the section 5.1 and Table 5.1. 'n' denotes corresponding the number of observations for style effect with five holding horizons from 1980 to 2008. The research uses the sample period from 1980 to 2008 because value effect is examined at the beginning of 1980, and all macroeconomic variables are available after 1980.

The research uses t-stats to judge which (unexpected) change of macroeconomic variable is statistically significant to explain the time-variation in style coefficient  $\gamma$  and long/short return R<sub>lms</sub>. T-stats is adjusted by Newey-West (1987) approach for heteroscedasticity and autocorrelation. In addition, following Granger and Newbold (1974), Ferson, Sarkissian, and Simin (2003, 2006) interpret a spurious regression as one in which the 't-ratios' in the regression is likely to indicate a significant relation when the variables are really independent. The problem may come from the denominator, the standard error, of the t-ratio. Because the spurious regression problem is driven by biased estimates of the standard error, the choice of standard

error estimator is crucial. The research uses the method of automatic lag selection in Ferson, Sarkissian, and Simin (2003) to estimate an efficient unbiased estimator of standard error. The number of lags is chosen by computing the autocorrelations of the estimated residuals and truncating the lag length when the sample autocorrelations become insignificant at longer lags. Specifically, the research calculates autocorrelations of the estimated residuals and compares the value with a cutoff at two approximate standard errors:  $2/T^{0.5}$ , where T is the sample size.

The volatility in explanatory power of forecasting variables is documented in Welch and Goyal (2008). It is in one case that the lower value of t-stats and adjusted  $R^2$  in overall period is caused by higher value in one subperiod and lower value in other subperiods. In another case, although macroeconomic variable is significant in two subperiods, the sign of its regression coefficient is opposite, which leads to lower value of t-stats and adjusted  $R^2$  in overall period. To control biased and inconsistent estimator in series regression model, the research uses the absolute magnitude of tstats and adjusted  $R^2$  in the univariate regression to extract (unexpected) change of macroeconomic variables to (marginally) significantly explain time-varying style effects. In the process of screening macroeconomic variables, the research sets up two conditions to select same significant macroeconomic variables for equal-(value-) weighted style coefficient  $\gamma$  and long/short return R<sub>lms</sub> with five holding horizons in order to simplify data analysis in the multivariate regression. Firstly, when slope coefficient of one macroeconomic variable at different time horizons has significant t-stats, the research selects one with the highest or higher adjusted  $R^2$  because t-stats might be biased due to its adjustment. Secondly, the research tests the significance of macroeconomic variables for the time-variation in style effects, measured by equal-(value-) weighted long/short return  $R_{lms}$  and style coefficient  $\gamma$ , to identify significant ones and use them in the multiple regression. It might be the case when one macroeconomic variable is significant at 5% or 10% significant level for one of two indicators in one style effect. To keep the same macroeconomic variable tested in the multiple regression of two equal-(value-) weighted style indicators, the research regards one macroeconomic variable with the value of t-stats more than 1.2 for one style indicator in univariate regression as one forecasting variable in multiple regression when it is significant for another style indicator at 5% or 10% significant level.

#### 5.2.2 Multiple regression

After extracting explanatory variables MEV(i) and UEXMEV(i) from univariate regression, the research uses multiple regression (5.6) and (5.6') to investigate the relation between style coefficient  $\gamma$ , long/short return R<sub>lms</sub> and the combination of these variables, respectively, and judge how well the combination of macroeconomic variables can explain time-varying style effects in overall sample period and subperiods. It uses t-stats to evaluate which (unexpected) change of macroeconomic variable is statistically significant to explain the time-variation in  $\gamma$  and R<sub>lms</sub>. T-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003).

$$\gamma_t = \lambda_0 + \lambda_{EV} * EV_{t-L} + e_t \tag{5.6}$$

and

$$R_{lms,t} = \lambda_0 + \lambda_{EV} * EV_{t-L} + e_t$$
(5.6')

t = 1, 2, ..., n

where  $\gamma_t$  and  $R_{lms}$  is style coefficient and long/short return in the holding period t, EV<sub>t-L</sub> is a (n\*1) vector representing (unexpected) change of lagged macroeconomic variables MEV(i) and UEXMEV(i) which are statistically significant in univariate regression. 'L' denotes lag length for macroeconomic variables available in advance when portfolios are established. 'n' denotes corresponding the number of observations for style effect with five holding horizons from 1980 to 2008.

# 5.3The impact of macroeconomic variables on the time-variation in momentum effect

This section investigates the impact of macroeconomic variables on time-varying momentum effect based on 6 and 12 months' formation lengths in the UK Stock Market from 1980 to 2008, respectively. It finds what macroeconomic variables have significant explanatory power for time-varying momentum effect, and how they explain the variation when momentum portfolios are based on share returns in past 6 and 12 months, and then hold for 3, 6, 12, 18, and 24 months, respectively. Time-varying momentum effect is measured by two indicators: momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$ , which are estimated in Chapter 4. The first momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  and 24 months' holding lengths are estimated in April 1980, July 1980, January 1981, July 1981, and January 1982, respectively. There are 345, 342, 336, 330, and 324 observations for momentum effect with five holding horizons.

#### 5.3.1 Momentum effect based on 6 months' formation length

In this subsection, the research investigates the impact of macroeconomic variables on time-variation in momentum effect based on 6 months' formation length and 3, 6, 12, 18, and 24 months' holding lengths.

#### A. Univariate regression to extract explanatory variables

With the model 5.5 and 5.5', the research regresses equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on 68 lagged (unexpected) change of macroeconomic variables, individually, for momentum portfolios with 3, 6, 12, 18, and 24 months' holding lengths from 1980 to 2008. Table 5.5, 5.6, 5.7, 5.8, and 5.9 report slope coefficient  $\lambda$ , t-stats, and adjusted  $R^2$  from the univariate regression in the case of five holding horizons, respectively. Panel A and B report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ and Winner/Loser return  $R_{wml}$ , respectively.

As Table 5.5 and 5.6 present, when equal-(value-) weighted momentum portfolios are hold for 3 and 6 months, UEXTBILL(Y), DY(H), and CPI(Q) have significant explanatory power for the time-variation in momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$ . Table 5.7, 5.8, and 5.9 show when momentum portfolios are hold for 12, 18, and 24 months, GDP(Y) and UEXTBILL(Y) are significant explanatory variables for the time-variation in momentum effect. Different from the findings in the US Stock Market by Cooper, Gutierrez, and Hameed (2004), the research finds no significant link between lagged 1 to 10 years' market return and momentum effect with 6 months' formation length in the UK Stock Market. Using the model 5.6 and 5.6', and (marginally) significant explanatory variables screened in univariate regression, the research regresses equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on the combination of them to investigate how these macroeconomic variables capture the time-variation in momentum effects based on 6 months' formation length and 3, 6, 12, 18, and 24 months' holding length, respectively.

# [Table 5.10]

Table 5.10 Panel A and B show that the combination of three derived macroeconomic variables, UEXTBILL(Y), DY(H), and CPI(Q) could explain less than 10 percentage of the variation in equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  when momentum portfolios are hold for 3 and 6 months' lengths. As shown in Panel C, D, and E, the explanatory power of GDP(Y) and UEXTBILL(Y) for the variation are less than 10 percentage, except for 10.05 to 12.86 percentage for equal-weighted  $\gamma_{mom}$ , when portfolios are hold for 12, 18, and 24 months. This indicates that macroeconomic variables offer weak explanatory power for time-varying momentum effect with 6 months' formation length.

UEXTBILL(Y) is significantly positive in the explanation for most  $\gamma_{mom}$  and  $R_{wml}$ , especially when portfolios are hold for 12, 18, and 24 months. One cannot reject the hypothesis that the coefficient of UNTBILL(Y) is different from zero. This indicates

strong momentum effect under the expectation of higher annual short-term interest rate. Besides UEXTBILL(Y), DY(H) (DY(H) and CPI(Q)) are significant positive (positive and negative) explanatory variables for equal-(value-) weighted timevarying momentum effect with 3 months' holding length. The rising DY in past 6 months and the declining CPI in past 3 months significantly strengthen return momentum over 3 months' horizon. Combination with UEXTBILL(Y), CPI(Q) plays determinant role in value-weighted momentum effect with 3 and 6 months' holding lengths.

Besides UEXTBILL(Y), GDP(Y) is significantly negatively associated with  $\gamma_{mom}$  and  $R_{wml}$  when holding 12, 18, and 24 months' momentum portfolios. This shows that low previous annual change of GDP at the end of formation month forecasts strong momentum effect with medium and long holding horizons. The findings extend the conclusion by Lucas, Dijk, and Kloek (2002) that over recessions investors load less on momentum stocks, and the conclusion by Asness, Moskowitz, and Pedersen (2008) that momentum premium is negatively related to recessions. The research offers empirical findings that economic condition offers forecasting information on current return momentum. Momentum investors might use two indicators, the previous annual change of GDP and unexpected annual change of TBILL, to make a decision on holding momentum portfolios with medium/long horizon.

Different from findings by Chordia and Shivakumar (2002), Cooper, Gutierrez, and Hameed (2004) that macroeconomic model based on lagged dividend yield, default spread, term structure, and TBILL yield cannot explain 6 months' momentum profits, the research find forecasting power of macroeconomic variables, UEXTBILL(Y), DY(H), CPI(Q), and GDP(Y) for time varying momentum effect in the UK Stock Market when momentum portfolios are formed on previous 6 months' share returns and hold for short-/medium-/long-term period.

#### C. Multivariate regression in subperiods

The research further tests the relation between the combination of derived macroeconomic variables and momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  in the 1980s, the 1990s, and January 2000 to December 2008, respectively. The test in subperiods helps the research to find whether macroeconomic variables have stable explanatory power for time-varying momentum effect. Table 5.11 to 5.15 report adjusted  $R^2$ , slope coefficient  $\lambda$ , and its t-statistics from multivariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on derived macroeconomic variables in three subperiods for five holding lengths.

#### [Table 5.11, 5.12, 5.13, 5.14, 5.15]

As indicated in Table 5.11 to 5.15, when momentum portfolios are hold for five horizons, adjusted  $R^2$  of multivariate regression in some subperiods are higher/or lower than corresponding one in overall period. This means the same deprived macroeconomic variables have volatile explanatory power for momentum effect over time. For example, most adjusted  $R^2$  in the 1980s ranges between 0.1174 and 0.2741 through momentum effect with five holding horizons. In the 1990s, it ranges between

0.1102 and 0.2425 through momentum effect with 12 to 24 months' holding length as well as equal-weighted momentum effect with 3 and 6 months' holding length. In 2000 through 2008, adjusted  $R^2$  for value-weighted momentum with 3 and 6 months' horizons is between 12.98 and 39.49, but it is below 0.02 for most other cases.

Similar to adjusted  $R^2$ , there exists the variation in significant macroeconomic variables on time-varying momentum effect through holding horizons and over subperiods. Firstly, significant macroeconomic variables, that affect time-varying momentum effects, vary with five holding lengths in the same subperiod. For example, in the 1990s when momentum portfolios are hold for 3 and 6 months, DY(H) is significant variable to explain the time-variation in momentum effect during this decade. When momentum portfolios are hold for 12 months, GDP(Y) as one significant variable accounts for time-varying momentum effect. For momentum portfolios with 18 months' holding lengths, UEXTBILL(Y) and GDP(Y) become significant explanatory variables for time-varying momentum effect. This might be caused by the difference in investment decision-making by momentum investors with different holding horizons although they face the same economic condition in the same period. In the 1990s, investors, who favour short-term momentum strategy, prefer to the change in dividend yield over past 6 months in their decision-making. Investors with medium momentum strategy prefer to unexpected annual change of TBILL and the annual change of GDP. Secondly, for momentum effect with one certain holding horizon, different macroeconomic variables significantly explain time-varying momentum effect over three subperiods. For example, when valueweighted portfolios are hold for 6 months, with higher adjusted  $R^2$  in three

subperiods, significant macroeconomic variables in the 1980s, the 1990s, and the 2000s are UEXTBILL(Y), DY(H), and CPI(Q), respectively. This might be caused by the existence of the variation in economic background in three subperiods. In the 1970s UK experienced higher inflation with CPI more than 20 percent, and following the obvious rise in TBILL rate. This leads to investors' more attention to unexpected change in TBILL in their practical decision in the 1980s. The volatile explanatory power of macroeconomic variables is consistent with the findings in Welch and Goyal (2008) that many explanatory variables, such as dividend yield, earning price ratio, book-to-market ratio, term spread, treasury bill rate, and inflation, are not robustly and statistically significant over the sample period.

#### 5.3.2 Momentum effect based on 12 months' formation length

In this subsection, the research investigates the impact of macroeconomic variables on the time-variation in momentum effect based on 12 months' formation length and 3, 6, 12, 18, and 24 months' holding lengths.

#### A. Univariate regression to extract information variables

Using regression model 5.5 and 5.5', the research regresses equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on 68 lagged (unexpected) change of macroeconomic variables, individually, for momentum portfolios with 3, 6, 12, 18, and 24 months' holding lengths from 1980 to 2008. Table 5.16, 5.17, 5.18, 5.19, and 5.20 report slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> from the univariate regression of momentum effect with 12 months' formation length and five holding horizons on each lagged (unexpected) change of macroeconomic variable. Panel A and B report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub>, respectively.

# [Table 5.16, 5.17, 5.18, 5.19, 5.20]

As Table 5.16, 5.17, 5.18, 5.19, and 5.20 indicate, for the equal-(value-) weighted momentum portfolios with five holding horizons, four macroeconomic variables, such as GDPG(Y), UEXTBILLG(Y), DY(Q), and EMR(-1)/CPI(H), have significant explanatory power for time-variation in equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$ , respectively. Similar to the findings by Cooper, Gutierrez, and Hameed (2004) that 6 months' momentum profit follows lagged market states in the US Stock Market, the research finds significantly negative link between lagged one-year's market return and momentum effect when equal-weighted portfolios are hold for 3 to 12 months in the UK Stock Market. When equal-weighted portfolios are hold for 18 and 24 months, lagged one-year's market return is marginally significantly associated with holding returns.

# B. Multivariate regression in overall period

Using regression model 5.6 and 5.6', and (marginally) significant explanatory variables in univariate regression, the research regresses equal-(value-) weighted

momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$ , respectively, on the combination of them to investigate how these macroeconomic variables to explain the time-variation in momentum effects with 12 months' formation length and five holding horizons.

#### [Table 5.21]

Table 5.21 shows that the combination of macroeconomic variables, GDP(Y), UEXTBILL(Y), DY(Q), and EMR(-1)/CPI(H), could explain 15.74 to 24.05, 16.11 to 22.96 percentage of the variation in equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  when momentum portfolios are hold for 6 and 12 months' lengths, which are higher than the case with 3, 18, and 24 months' holding lengths.

For momentum portfolios with five holding horizons, GDP(Y) is (marginally) significantly negative explanatory variable for  $\gamma_{mom}$  and  $R_{wml}$ . This means that low annual change of GDP at the end of formation month forecasts strong momentum effects with different holding horizons. When momentum portfolios are hold for 3, 6, and 12 months or value-weighted portfolios for 18 and 24 months, UEXTBILL(Y) is (marginally) significantly positive in the explanation of  $\gamma_{mom}$  and  $R_{wml}$ , which indicates strong momentum effect when short-term interest rate is higher than expected. DY(Q), previous quarterly growth in dividend yield, is (marginally) significantly and positively associated with most  $\gamma_{mom}$  and  $R_{wml}$ . When value-weighted portfolios are hold for 3 to 18 months, the change in CPI in past six months,

CPI(H), has negative impact on momentum effect. For equal-weighted momentum portfolios, previous one year's market return loses explanatory power for momentum effect although it is significant in the univariate analysis.

As the findings on momentum effect with 6 months' formation length in previous subsection, empirical work presents the relation between lagged macroeconomic variables and momentum effect with 12 months' formation length. When comparing the magnitude of adjusted  $R^2$  of multivariate regression in Table 5.10 and Table 5.21, it is found that the combination of lagged macroeconomic variables offers explanatory power for momentum effect with 12 months' formation length stronger than momentum effect with 6 months' formation length through five holding horizons.

# C. Multivariate regression in subperiods

The research further tests the impact of the combination of macroeconomic variables on momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  in the 1980s, the 1990s, and January 2000 to December 2008, respectively, when momentum portfolios are formed on previous 12 months' share returns and hold for 3, 6, 12, 18, and 24 months. Table 5.22, 5.23, 5.24, 5.25, 5.26 report three regression parameters, adjusted  $R^2$ , slope coefficient  $\lambda$ , and its t-statistics in three subperiods for the case with five holding lengths.

[Table 5.22, 5.23, 5.24, 5.25, 5.26]

When momentum portfolios are hold for 6 and 12 months, adjusted  $R^2$  in subperiods ranges between 0.0821 to 0.6152, and more than 0.20 in most subperiods. Its strong explanatory power in subperiods is consistent with its strong performance in the overall period. When portfolios are hold for 18 months, most of adjusted  $R^2$  in the 1980s and 1990s are more than 0.16. When portfolios are hold for 3 months and 24 months, adjusted  $R^2$  in three subperiods are less than 0.15 and 0.16 in many cases, which indicates the declining impact of macroeconomic variables on momentum effect in the short and long holding length.

As momentum effect with 6 months' formation length, adjusted  $R^2$  in subperiods is not invariant when momentum portfolios are formed on previous 12 months' share returns and hold for five horizons. When equal-weighted portfolios are hold for 3 to 18 months, adjusted  $R^2$  exhibits the declining pattern over three subperiods, the highest explanatory power of macroeconomic variables on momentum effect in the 1980s. When value-weighted portfolios are hold for 3 to 24 months, explanatory power of macroeconomic variables for momentum effect exhibits the 'U' pattern, the strongest in the 2000s.

As the performance of adjusted  $R^2$ , there exists obvious variation in significant macroeconomic variables for time-varying momentum effect with 12 months' formation length over holding horizons and over subperiods. *Firstly*, macroeconomic variables, which are significant to affect time-varying momentum effect in the same subperiod, are different for the cases with five holding lengths. For example, in the 1980s when equal-weighted portfolios are hold for 3 months, UEXTBILL(Y) and EMR(-1) are two significant macroeconomic variables to explain the time-variation in momentum effect during this decade. When the portfolios are hold for 18 months, GDP(Y) and DY(Q) are two significant variables to explain time-varying momentum effect. This might be caused by the difference in determinant factors in the makingdecision of momentum investors with different horizons. *Secondly*, for momentum effect with the same holding length, significant macroeconomic variables are different over three subperiods. For example, when equal-weighted portfolios are hold for 6 months, with higher adjusted R<sup>2</sup> in three subperiods, significant macroeconomic variables in the 1980s and the 1990s are UEXTBILL(Y) and EMR(-1), GDP(Y), respectively, combining with DY(Q). This reflects that momentum investors with same holding horizon might pay attention to different macroeconomic variables in their making-decision over time due to the variation in economic backgrounds in two subperiods. This pattern is consistent with the findings in momentum effect with 6 months' formation length.

# 5.4 The impact of macroeconomic variables on the time-variation in size effect

In this section, the research investigates the impact of macroeconomic variables on time-varying size effect with 3, 6, 12, 18, and 24 months' holding length in the UK Stock Market from 1980 to 2008, respectively. It finds what macroeconomic variables offer explanatory power for time-varying size effect, and how they affect the variation. Time-varying size effect is measured by two indicators: size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$ , which are estimated in Chapter 4. The first size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with 3, 6, 12, 18, and 24 months' holding lengths are estimated in April 1980, July 1980, January 1981, July 1981, and January 1982, respectively. There are 345, 342, 336, 330, and 324 observations for size effect with five holding horizons.

#### A. Univariate regression to extract macroeconomic variables

The research regresses equal-(value-) weighted size coefficient  $\gamma_{size}$  and Small/Large return  $R_{sml}$  on 68 lagged (unexpected) change of macroeconomic variables, individually, for size effect based on five holding horizons. Table 5.27, 5.28, 5.29, 5.30, and 5.31 report slope coefficient  $\lambda$ , t-stats, and adjusted  $R^2$  from the univariate regression of size effect with five holding horizons on each lagged (unexpected) change of macroeconomic variable. Panel A and B report these parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$ , respectively.

As shown in Table 5.27, 5.28, 5.29, 5.30, and 5.31, CPI(Y), UEXTBILL(Y) or UEXCPI(Y), UEXTBILL(Q) are two significant explanatory variables for the time-variation in size effect with five holding horizons. The findings are similar to the conclusion by Levis and Liodakis (1999) that annual return in UK's small/large spread from 1968 to 1977 is associated with some lagged economic variables, such as the monthly change in consumer price index and monthly change in the three-month Treasury bill.

Besides two common explanatory variables, UEXTERM(Y) significantly impacts on short-/or medium-term time-varying size effect with 3, 6, and 12 months' holding length. GDP(Y) or UEXGDP(Y) plays significant role in medium-/or long-term time-varying size effect with 12, 18, and 24 months' holding length.

#### B. Multivariate regression in overall period

Using (marginally) significant explanatory variables in univariate regression, the research regresses equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$ , respectively, on the combination of them to investigate how these macroeconomic variables have impact on the time-variation in size effects over five holding horizons.

## [Table 5.32]

Table 5.32 shows that adjusted  $R^2$  from multivariage regression of size effect with 12 months' holding length is higher than one with shorter and longer holding lengths. The combination of macroeconomic variables, GDP(Y), CPI(Y), UEXTBILL(Y), and UEXTERM(Y), could explain 14.93 to 17.64 percentage of the variation in equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  when size portfolios are hold for 12 months' length. For the time-variation in size effect with 6 and 18 months' holding lengths, the explanatory power of the combination of macroeconomic variables ranges between 10.31 and 12.72, between 10.75 and 13.26. When size portfolios are hold for 3 and 24 months, their explanatory power falls below 10 percentage.

For size portfolios with 3, 6, and 12 months' holding lengths, UEXTBILL(Y) combined with IP(M), independently, and combined with GDP(Y) are significant variables to explain time-varying size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$ , respectively. CPI(Y) and UEXTERM(Y) are unsignificant because of high correlation between them and UEXTBILL(Y), 0.6187 and -0.7387, respectively, as indicated in Table 5.4. The negative slope coefficient of UEXTBILL(Y) shows that the increase in unexpected annual change of TBILL at the end of formation month forecasts weak small-cap effects over 3 to 12 months, and vice versa. This is indirectly supported by the findings of Zhang, Hopkins, Satchell, and Schwob (2009) on negative relationship between SMB (small minus big) and unexpected inflation, closely related to unexpected short-term interest rate, in the US and UK Stock

Market. The return pattern of small-cap shares is consistent with the fundamental performance of small firms in the findings by Gertler and Gilchrist (1994) and Ehrmann (2000) that although all firms are hit by tightening monetary policy, small firms shrinks much more than large firms after the rise in interest rate due to tightening money policy in US and Germany. The positive slope coefficient of IP(M)and GDP(Y) indicates that the increase in previous monthly industrial production and annual GDP growth benefits following 3 and 12 months' small-cap effect, respectively. When size portfolios are hold for 18 and 24 months, time-varying size effect is driven by GDP(Y), and UEXGDP(Y) and UEXCPI(Y), respectively. Annual (unexpected) change of GDP positively impacts on small-cap effect in the long holding length up to 18 to 24 months. This pro-cyclical small-cap effect is consistent with the conclusion by Fraser (1995) that the disappearance of small-cap premium in the London Stock Market in the early 1990s might be caused by UK's deep recession in this period. It is similar to the findings by Zhang, Hopkins, Satchell, and Schwob (2009) on positive relationship between SMB (small minus big) and unexpected higher GDP in the US and UK Stock Market.

# C. Multivariate regression in subperiods

Based on the findings in the whole sample period, the research further explores the impact of the combination of macroeconomic variables on size effect in the 1980s, the 1990s, and January 2000 to December 2008, respectively. Table 5.33, 5.34, 5.35, 5.36, and 5.37 report adjusted  $R^2$ , slope coefficient  $\lambda$ , and its t-statistics from multivariate regression of size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  on the

combination of macroeconomic variables in three subperiods for five holding horizons.

[Table 5.33, 5.34, 5.35, 5.36, 5.37]

Higher adjusted  $R^2$  of multivariate regression of size effect with 12 months' holding horizon in subperiods is consistent with its performance in overall period. Among the cases with five holding horizons, adjusted  $R^2$  in subperiods is highest or second highest when size portfolios are hold for 12 months, ranging between 0.1738 and 0.4481. Especially, the combination of macroeconomic variables explains 38.82 to 44.81 percentage of the variance in size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$ in the 1990s and afterwards. Following size effect with 12 months' holding length, one with 6 months' holding length obtains stronger explanatory power from macroeconomic variables in subperiods, adjusted  $R^2$  from 0.2016 to 0.3209 in the 1980s and 2000s. For size portfolios hold for 18 months, macroeconomic variables offer 15 to 38 percentage of the variation in size effect in the 1990s and 2000s, but less than 10 percentage in the 1980s. When size portfolios are hold for 3 and 24 months, the explanatory power is generally lower than 15 percentage and volatile over subperiods.

The empirical work finds the variation in significant macroeconomic variables for time-varying size effect over holding horizons and over three subperiods. *Firstly*, macroeconomic variables, that significantly affect time-varying size effects, are different across the cases with five holding horizons in the same subperiod. For example, in the 1980s when Small/Large positions are hold for 3 to 12 months, UEXTBILL(Y) or CPI(Y), which is closely associated with inflation, are two significant variables to explain the time-variation in short-/medium-run size effect during this decade. When Small/Large portfolio is hold for 18 months, GDP(Y) significantly accounts for time-varying long-run size effect. This reflects that in the 1980s short- or medium-term size effect is more associated with inflation and shortterm interest rate than other variables, and long-term size effect is more associated with previous annual change of GDP than other variables. Secondly, significant macroeconomic variables to explain time-varying size effect change over three subperiods. For example, when size decile portfolios are hold for 12 months, with higher adjusted  $R^2$  in three subperiods and overall period, significant macroeconomic variable for size coefficient  $\gamma_{mv}$  in the 1980s, the 1990s, and from 2000 to 2008 is CPI(Y), GPD(Y), UEXTBILL(Y). This indicates that medium-term size effect is affected by different macroeconomic variables over time. Specifically, the same macroeconomic variable has different explanatory power for size effect over time. For example, size effect with 12 months' holding length is marginally negatively sensitive to CPI(Y) in the 1980s, but positive though unsignificant since the 2000.

# 5.5 The impact of macroeconomic variables on the time-variation in value effect

In this section, the research concentrates on the impact of macroeconomic variables on time-varying value effect with 3, 6, 12, 18, and 24 months' holding length in the UK Stock Market from 1980 to 2008, respectively. It explores what macroeconomic variables capture time-varying value effect and how they affect the variation. Timevarying value effect is measured by two indicators: value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub>, which are estimated in Chapter 4. The first value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> with 3, 6, 12, 18, and 24 months' holding lengths are estimated in April 1980, July 1980, January 1981, July 1981, and January 1982, respectively. There are 345, 342, 336, 330, and 324 observations for value effect with five holding horizons.

#### A. Univariate regression to extract macroeconomic variables

The research regresses equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$  on 68 lagged (unexpected) change of macroeconomic variables, individually, for value portfolios with 3, 6, 12, 18, and 24 months' holding lengths from 1980 to 2008. Table 5.38, 5.39, 5.40, 5.41, and 5.42 report slope coefficient  $\lambda$ , t-stats, and adjusted  $R^2$  from the univariate regression of value effect with five holding horizons on each lagged (unexpected) change of macroeconomic variable. Panel A and B report these parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$ , respectively.
Table 5.38, 5.39, 5.40, 5.41, and 5.42 show, for equal-(value-) weighted value portfolios with 3, 6, and 12 months' holding length, UEXTBILL(Y), UEXDY(H), and MR(-8) are three significant explanatory variables for the time-variation in value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub>, respectively. Supporting and extending the findings by Levis and Liodakis (1999) that the monthly change in three-month Treasury bill has impact on the annual change in value/growth spread in the UK from 1968 to 1977, the research finds unexpected annual TBILL significantly explains time-varying value effect with 3, 6, and 12 months' holding lengths since the 1980s. When value portfolios are hold for 18 months, GDP(Y) and MR(-8) are significant variables to explain the volatility of value effect. When value portfolios are hold for 24 months, GDP(Y), MO(Y), CPI(Y), and MR(-8) are significant ones.

## B. Multivariate regression in overall period

With (marginally) significant explanatory variables extracted in univariate regression, the research regresses equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub>, respectively, on the combination of them to investigate how these macroeconomic variables explain the time-variation in value effects over five holding horizons.

[Table 5.43]

Table 5.43 shows that the combination of lagged macroeconomic variables could explain 16.53 to 40.87 and 5.01 to 28.44 percentage of the variation in equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High R<sub>lmh</sub>, respectively. For the same holding horizon, the explanatory power of the regression is much stronger for value effect  $\gamma_{pb}$  than for Low/High return R<sub>lmh</sub>. In Chapter 4, it is found in Panel B2 and B3 of Table 4.12 that the second and third 'Low' value-weighted portfolios have higher percentage (9% to 15%) of observations with highest return among decile value portfolios, following 'Low' portfolio, and that the 6<sup>th</sup> to 9<sup>th</sup> value-weighted portfolio has higher percentage (7% to 15%) of observations with lowest return, following 'High' portfolio. Value coefficient  $\gamma_{pb}$  from the regression of value decile portfolios' holding returns on shares' price-to-book ratios at the end of formation month contains the information of all shares, but Low/High return R<sub>lmh</sub> only includes corresponding information of shares in two portfolios with low and high price-to-book ratio. So, the combination of macroeconomic variables offers explanatory power for value coefficient  $\gamma_{pb}$  stronger than for Low/High return R<sub>lmh</sub>.

Explanatory power of the combination of macroeconomic variables on value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$  rises as holding length extends, except for value-weighted Low/High return  $R_{lmh}$ . When value portfolios are hold for 24 months, the fraction of variation predicted by macroeconomic variables increases to higher level, ranging between 28.44 and 40.87.

Although some macroeconomic variables lose significance in multivariate regression of value effect, MR(-8) keeps statistically significant in most cases as its performance in univariate regression. This reflects that market state in past eight years is dominant explanatory variable for value effect, and contains some information in economic factors. The positive slope coefficient of MR(-8) means that high market return in past eight years forecasts strong value effect. This findings is consistent with conclusion by Bird and Whitaker (2004) on counter-cyclical value effect. They find that value effect is confined in the correction period after a large upward market movement in major European market from 1990 to 2002. Besides MR(-8) as common explanatory variable for value coefficient  $\gamma_{pb}$ , UEXDY(H) and GDP(Y) with t-stats of more than 2.0 are significant variables when value portfolios are hold for 3/6 months and 18 months. Different from the findings by Levis and Liodakis (1999) of no significant relation between dividend yield ratio and value spread, the research presents significant explanatory power of unexpected half-yearly change of dividend yield on short-term value effect. This conclusion also supports the findings by Cohen, Polk and Vuolteenaho (2003) that about 75 to 80 percent of value spread is explained by unexpected profitability (cash flows). The performance of macroeconomic variables indicates that the time-variation in short-term value effect is associated with market variable such as unexpected half-yearly change of dividend yield, and long-term value effect is influenced by the annual change of GDP.

### C. Multivariate regression in subperiods

The research further investigates the combination of macroeconomic variables on time-varying value effect in the 1980s, the 1990s, and January 2000 to December

2008, respectively. Table 5.44, 5.45, 5.46, 5.47, and 5.48 report adjusted  $R^2$ , slope coefficient  $\lambda$ , and its t-statistics from multivariate regression of value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$  on macroeconomic variables in three subperiods for five holding horizons.

## [Table 5.44, 5.45, 5.46, 5.47, 5.48]

As indicated in Table 5.44, 5.45, 5.46, 5.47, and 5.48, when value portfolios are hold for 3 to 18 months, most adjusted  $R^2$  of multivariate regression in the 1990s and the 2000s are much lower than in the 1980s and in the overall sample period. This means that the explanatory power of macroeconomic variables for value effect volatiles through three subperiods. When value portfolios are hold for 24 months, the combination of macroeconomic variables could explain 32.45 to 59.18 percentage of the variation in value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$  in three subperiods, higher than corresponding one in the overall period and ones in subperiods when value portfolios are hold for 3 to 18 months. High adjusted  $R^2$  in subperiods for 24 months' value effect is consistent with its performance in the overall period.

Significant macroeconomic variables for time-varying value effect are not invariant through five holding horizons and over subperiods. *Firstly*, significant macroeconomic variables are different for value effect with five holding horizons in the same subperiod. For example, although the combination of macroeconomic variables offers explanatory power higher up to 19 percentage of volatility in value effect with five holding horizons in the 1980s, significant explanatory variables are different for short-term value effect to long-term value effect. When holding value portfolios for 3 and 6 months, UEXDY(H) and MR(-8) significantly explain timevarying value-weighted  $\gamma_{pb}$ . When holding ones for 18 months, GDP(Y) combining MR(-8) does. Secondly, macroeconomic variables have different impact on timevarying value effect with one holding horizon over three subperiods. For example, as shown in Table 5.48, when value portfolios are hold for 24 months, with higher adjusted R<sup>2</sup> in three subperiods, the same macroeconomic variable offers different explanatory power for time-varying value effect over time. The signs of slope coefficient of CPI(Y) is opposite in three subperiods. In the 1980s and the 1990s low CPI(Y) forecasts strong value effect, but vice versa in the 2000s. The similar pattern happens in M0(Y) in the 1990s and the 2000s.

## 5.6 Conclusion

In this chapter, the research focuses on the relation between lagged macroeconomic variables and the time-variation in style effects with five holding horizons in the UK Stock Market from 1980 to 2008. The empirical work finds that some lagged macroeconomic variables, which are associated with economic conditions or market states, have significant explanatory power for time-varying momentum, size, and value effect. The relevant conclusion is described as following three aspects:

## 5.6.1 Explanatory power of macroeconomic variables for time-varying style effects

## • Momentum effect

As the empirical work indicates, lagged macroeconomic variables explain the timevariation in momentum effect with 12 months' formation length stronger than one with 6 months' formation length from 1980 to 2008, with adjusted R<sup>2</sup> in multivariate regression more than 0.10 for the former and less than 0.10 for the latter in most cases. Specially, the explanatory power is stronger, which ranges between 15% and 25%, for time-varying momentum effect with 6 and 12 months' holding lengths. The explanatory power of macroeconomic variables is volatile over time, which exhibits declining (rising) pattern for equal-(value-) weighted momentum effect over the 1980s, the 1990s, and from 2000 to 2008. It is consistent with the findings by Welch and Goyal (2008) on aggregate market return that the performance of predicting model based on some macroeconomic variables is volatile over time in the US Stock Market.

### • Size effect

The explanatory power of lagged macroeconomic variables for time-varying size effect with five holding horizons exhibits ' $\Lambda$ ' pattern from 1980 to 2008. When decile size portfolios are hold for 12 months, the combination of macroeconomic variables captures 14.93 to 17.64 percentage of the time-variation in size effect. Adjusted R<sup>2</sup> of the multivariate regression ranges between 10.31 and 13.26 percentage for the time-variation in size effect with 6 and 18 months' holding lengths. It falls below 10 percentage when size portfolios are hold for 3 and 24 months. High adjusted R<sup>2</sup> of multivariate regression for 12 months' size effect in overall period is consistent with its performance in subperiods when it ranges between 17.38% and 22.11% in the 1980s, and between 38.82% and 44.81% afterwards.

#### • Value effect

There is obvious difference in explanatory power of macroeconomic variables for time-varying value effect between two measures, which are value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$ . Through five holding horizons, lagged macroeconomic variables explain 16.53 to 40.87 percentage of the variation in value coefficient  $\gamma_{pb}$  from 1980 to 2008, but only 5.01 to 28.44 percentage of the variation in Low/High return  $R_{lmh}$ . High explanatory power of macroeconomic variables for value coefficient  $\gamma_{pb}$  is due to information of all firms' price-to-book ratios and share returns included in  $\gamma_{pb}$ . Following 'Low' and 'High' portfolios, other decile value portfolios have higher percentage of observations with highest return and lowest return. This pattern is captured by value coefficient  $\gamma_{pb}$ , but overlook by Low/High return  $R_{lmh}$ . In addition, as holding length extends from 5 months to 24 months, the

percentage of the variation in value effect driven by lagged macroeconomic variables increases to higher level, ranging 28.44% and 40.87%, except for value-weighted Low/High  $R_{lmh}$ . Higher adjusted  $R^2$  of multivariate regression for 24 months' value effect is also found in three subperiods when it ranges between 0.3245 and 0.5918.

## 5.6.2 Significant macroeconomic variables for time-varying style effects

As concluded in subsection 5.6.1, the combination of lagged macroeconomic variables offers explanatory power for momentum effect with 12 months' formation length and 6/12 months' holding length, size effect with 12 months' holding length, and value effect with 24 months' holding length stronger than corresponding style effects with other formation length or holding lengths. They are denoted as momentum effect (12/6, 12/12), size effect (12), and value effect (24), respectively. The research finds that (marginally) significant macroeconomic variables in the multivariate regression of momentum effect (12/6, 12/12), size effect (12/6, 12/12), size effect (12), and CPI(H); GDP(Y) and UEXBILL(Y); CPI(Y) and MR(-8), respectively. The sign of information coefficient  $\lambda$  in multivariate regression indicates that the previous annual decline in GDP and unexpected annual rise in TBILL might forecast strong mid-term momentum effect and mid-term large-cap effect, and that previous annual decline in CPI and higher market return over past 8 years might forecast strong long-term value effect.

Three style effects are closely associated with macroeconomic conditions. Momentum effect (12/6, 12/12) is stronger in declining economic growth and unexpected rise of interest rate. Winner shares always consist of firms, which have good fundamentals and resist the economic contraction and tightening monetary policy. Conversely, loser shares always consist of firms, which have worse fundamentals and are easily hit by deteriorating macroeconomic conditions. The different response of two kinds of firms to the economic conditions broadens the return spread between winner shares and loser shares. The findings are different from the hypothesis based on previous studies that momentum effect exists in economic expansionary period.

Conversely, strong small-cap effect (12) exists in economic expansion and unexpected decline of interest rate. Faster economic growth and loosing monetary policy benefit small firms' business much more than large firms, and cause the outperformance of small-cap shares over large-cap shares. These findings are supported by the research on firm size by Tamari (1984), Gertler and Gilchrist (1994), and Ehrmann (2000). They are consistent with the hypothesis based on the findings of empirical work on size effect by Jensen, Johnson, and Mercer (1998), Levis and Liodakis (1999), Perez-Quiros and Timmermann (2000).

As the findings in Levis and Liodakis (1999), the research shows strong value effect (24) in the declining CPI. Value firms always have large market capitalization, and benefit the decline in business cost due to lower inflation much more than growth firms, which leads to the rise in their profits. It is consistent with the hypothesis

based on previous studies on value effect by Jensen, Johnson, and Mercer (1998). On the other hand, the research finds that strong value effect (24) exists when the share market experiences upward movement in long horizon. In this case, the price of most shares, especially growth shares, is much higher than their firm fundamentals. Investors allocate more position in value shares and less position in growth shares to avoid the risk of downward market, which causes high value premium.

Obviously, momentum effect (12/6, 12/12) and size effect (12) are commonly significantly sensitive to the previous annual change of GDP and unexpected annual change of TBILL. The coefficient signs of GDP(Y) and UEXTBILL(Y) in the multivariate regression of momentum effect (12/6, 12/12) are opposite to corresponding ones in the regression of size effect (12), which means momentum effect and size effect are significant in the opposite economic background. In addition, Table 5.4 shows 0.6187 of correlation coefficient between CPI(Y) and UEXTBILL(Y). Momentum effect (12/6, 12/12) and size effect (12), which is significantly influenced by UEXTBILL(Y), are associated with CPI(Y). This means that momentum effect (12/6, 12/12) and value effect (24) are significant in the opposite inflation condition, and size effect (12) and value effect (24) are significant in the same inflation condition. This offers economic explanation for the empirical findings in Chapter 4 on negative relation between momentum effect, and positive relation between size effect and value effect in the UK Stock Market.

5.6.3 The variation in significant macroeconomic variables for time-varying style effects

Significant macroeconomic variables for time-varying style effects are different over five holding horizons and over subperiods. The reason for this variation is that styleoriginated investors use different macroeconomic variables to make decision on share allocation depending on investing horizons, dynamic economic conditions and market states.

*Firstly*, macroeconomic variables that significantly affect time-varying style effect in the same subperiod are different for five holding horizons. For example, in the 1980s when equal-weighted momentum portfolios are constructed with 12 months' formation length, significant variables to explain the time-variation in momentum effect with 3 and 18 months' holding length are UEXBILL(Y) and EMR(-1), GDP(Y) and DY(Q), respectively. When Small/Large position is hold for 3 to 12 months, UEXTBILL(Y) or CPI(Y) are two significant variables to explain the time-variation in Small/Large return  $R_{sml}$ . In the 18 months' long-run horizon, GDP(Y) significantly accounts for time-varying  $R_{sml}$ . When value portfolios are hold for 3 and 6 months, UEXDY(H) and MR(-8) significantly explains time-varying value-weighted  $\gamma_{pb}$ . When they are hold for 18 months, GDP(Y) combining with MR(-8) does so.

Secondly, significant macroeconomic variables to explain time-varying style effect with same holding horizon are time-variant over three subperiods. For example, UEXTBILL(Y) and EMR(-1), GDP(Y), respectively, combining with DY(Q), significantly explain the variation in equal-weighted momentum effect (12/6) in the 1980s and the 1990s. For size effect (12), significant macroeconomic variable for size coefficient  $\gamma_{mv}$  in the 1980s, the 1990s, and from 2000 to 2008 is CPI(Y), GDP(Y), UEXTBILL(Y), respectively.

Thirdly, the same macroeconomic variable has opposite explanatory power for style effect over time. For example, size effect (12) is marginally negatively sensitive to CPI(Y) in the 1980s, but positive though unsignificant since the 2000. Value effect (24) is significantly negatively associated to M0(Y) and CPI(Y) in the 1990s, but vice versa since 2000. The same pattern is found in other studies on aggregate market return. Campbell and Hamao (1992) show that the estimate of dividend-price ratio coefficient for predicting returns in Japan Stock Market switches sign from the positive in the 1970s to the negative in the 1980s. Soufian (2001) finds that unexpected inflation has significantly negative and positive impact on expected return in the UK Stock Market in the 1980s and the 1990s, respectively. Humpe and Macmillan (2007) find that the CPI does not have a significant influence over Japan Stock Market from 1965 to 2005 due to volatile macroeconomic conditions. Japan experienced high inflation and stable growth in industrial production from 1971 to 1990, and strong disinflation in the early 1990s and deflation but volatile industrial production in the late 1990s and early 21<sup>st</sup> century. During the period of low inflation, its influence on stock price may be 'cancelled' by other economic indicators with high volatility, such as industrial production and money supply. Similarly, high inflation with more than 10 percent in the early 1980s and low inflation with less

than 4 percent in most period after the 2000 contribute to the different influence of CPI(Y) on size effect (12) and value effect (24) in the UK Stock Market.

## Appendix: tables and figures in Chapter 5

Panel A Macroeconomic variables											
Macroeconomic Variables	Description	Lagged length	Sources								
GDP	Growth of Domestic Production	2 quarters	ESDS IFS								
IP	Industrial Production	2 months	Datastream								
IPM	Industrial Production Index – Manufacturing	2 months	Datastream								
M0	Money Supply M0	1 month	Datastream								
UNE	Unemployment Rate	2 month	Datastream								
СРІ	Consumer Price Index	1 month	Datastream								
TBILL	monthly return on a 90-day Treasury Bill	1 month	LSPD								
TERM	the difference in the annualized yield on the long-term UK government bonds and the yield on the three-month Treasury Bill.	1 month	ESDS IFS								
DY	Dividend Yield based on value-weighted market index of all companies	1 month	LSPD								
EMR/VMR(-y)	equal-(value-) weighted y years' compounded return; $y=1,2,10$	1 month	LSPD								

## Table 5.1 Macroeconomic variables for the time-variation in style effects

Panel B The hypothesis on	relationship between	style effects and	l macroeconomic	variables.
i unei B ine nypouneono on	renationship occureen	begre erreets and	· maero e e o monne	141401001

Macroeconomic variables	Momentum effect	Small-cap effect	Value effect
GDP	+	+	+
IP	+	+	+
IPM	+	+	+
M0	0	+	+
UNE	0	0	0
CPI	0	-	-
TBILL	0	-	-
TERM	0	0	0
DY	0	0	0
EMR/VMR(-y)	+	0	0

Note: Panel A in the table reports the description on macroeconomic variables, their lagged length, data sources. Panel B offers the hypothesis on relation between style effects and macroeconomic variables. The symbols '+' and '-' are marked as positive and negative relation between style effects and macroeconomic variables, respectively.

	Panel A The autocorrelation of primary macroeconomic variables												
	1st	2nd	3rd	4th	5th	6th	7th	8th	9th	10th			
GDP	0.96	0.92	0.88	0.84	0.81	0.77	0.73	0.68	0.64	0.58			
IP	0.99	0.99	0.98	0.98	0.97	0.96	0.96	0.95	0.94	0.93			
IPM	0.99	0.99	0.98	0.98	0.97	0.96	0.95	0.94	0.93	0.92			
M0	0.99	0.98	0.97	0.96	0.95	0.94	0.93	0.92	0.91	0.90			
UNE	0.99	0.99	0.98	0.97	0.97	0.96	0.94	0.93	0.92	0.91			
CPI	0.98	0.94	0.90	0.86	0.82	0.77	0.72	0.68	0.65	0.61			
TBILL	0.98	0.96	0.94	0.92	0.90	0.88	0.86	0.84	0.82	0.79			
TERM	0.95	0.91	0.86	0.81	0.77	0.72	0.67	0.61	0.56	0.51			
DY	0.98	0.97	0.95	0.93	0.91	0.89	0.87	0.84	0.82	0.80			

 Table 5.2 The autocorrelation and the unit root test of primary macroeconomic variables

D 1D	<b>T</b> 1 •		c			
Panel R	The linit	root statistics	of ni	rımarv	macroeconomic	variables
I uner D	The unit	100t stutistics	or pr	i i i i i i i i i i i i i i i i i i i	macrocconomic	variables

	AĽ	OF (without tre	nd)		ADF (trend)			
D-lag	2	1	0	2	1	0		
GDP	-2.68	-2.65	-2.63	-2.75	-2.72	-2.70		
IP	-0.78	-0.82	-0.96	-1.14	-1.29	-1.88		
IPM	-0.29	-0.32	-0.56	-2.48	-2.51	-2.77		
M0	0.98	0.19	0.48	0.91	0.63	0.71		
UNE	-1.21	-0.74	-0.22	-3.15	-3.22	-3.85		
CPI	-4.27	-4.26	-4.68	-3.83	-3.78	-3.65		
TBILL	-2.31	-2.22	-2.21	-2.83	-2.59	-2.48		
TERM	-2.56	-2.74	-2.71	-2.92	-2.79	-2.87		
DY	-2.07	-2.06	-2.06	-2.10	-2.05	-2.12		
Significant level 5%		-2.87			-3.42			

Note: Panel A of the table reports the first to tenth order autocorrelation of nine primary macroeconomic variables from 1980 to 2008. Panel B reports their unit root statistics. GDP (the Growth of Domestic Product) and TERM (Term Structure) are collected from ESDS IFS (The Economic and Social Data Service, International Financial Statistics). IP (Industrial Production), IPM (Industrial Production Index – Manufacturing), M0 (Money Supply), UNE (Unemployment Rate), and CPI (Consumer Price Index) are collected from Datastream. TBILL (3-month's Interest Rate) and DY (Dividend Yield) are collected from LSPD.

	1.	0 1	2.1	4.1	5.1	<u> </u>	7.1	0.1	0.1	104
655 (I)	Ist	2nd	3rd	4th	Sth	oth	/tn	8th	9th	10th
GDP(Y)	0.89	0.77	0.66	0.60	0.54	0.48	0.41	0.34	0.27	0.15
UEXGDP(Y)	0.89	0.77	0.66	0.58	0.51	0.44	0.36	0.28	0.20	0.10
IP(M)	-0.23	0.00	0.14	-0.09	0.17	-0.07	0.03	0.10	-0.08	-0.04
IP(Q)	0.58	0.34	0.08	0.13	0.17	0.10	0.11	0.04	-0.05	-0.11
IP(H)	0.76	0.67	0.60	0.48	0.38	0.16	0.17	0.13	0.02	0.00
IP(Y)	0.88	0.83	0.76	0.67	0.61	0.52	0.45	0.37	0.27	0.21
UEXIP(Q)	0.39	0.15	0.06	0.07	0.14	0.07	0.12	0.09	-0.03	-0.10
UEXIP(H)	0.65	0.53	0.44	0.30	0.26	0.15	0.16	0.12	0.02	-0.03
UEXIP(Y)	0.84	0.76	0.69	0.59	0.53	0.43	0.36	0.29	0.19	0.13
IPM(M)	-0.20	0.01	0.18	-0.06	0.09	0.08	0.04	0.07	0.00	-0.08
IPM(Q)	0.63	0.40	0.13	0.19	0.23	0.22	0.22	0.12	0.02	-0.05
IPM(H)	0.82	0.75	0.68	0.58	0.50	0.35	0.34	0.28	0.18	0.15
IPM(Y)	0.91	0.86	0.81	0.75	0.69	0.63	0.57	0.50	0.41	0.34
UEXIPM(O)	0.46	0.22	0.12	0.11	0.17	0.19	0.19	0.11	0.02	-0.07
UEXIPM(H)	0.71	0.60	0.53	0.40	0.35	0.19	0.17	0.11	0.12	0.05
UEXIPM(Y)	0.88	0.82	0.55	0.10	0.62	0.56	0.48	0.40	0.12	0.03
MO(M)	0.00	0.02	0.70	0.00	0.02	0.06	0.40	0.40	0.01	0.24
M0(0)	0.12	-0.13	-0.01	0.00	0.05	0.00	-0.02	0.03	0.00	0.04
MO(Q)	0.08	0.27	0.01	0.09	0.15	0.12	0.00	0.04	0.04	0.04
	0.85	0.08	0.30	0.44	0.50	0.13	0.12	0.12	0.10	0.11
MU(Y)	0.92	0.84	0.78	0.72	0.67	0.60	0.54	0.48	0.43	0.38
UEXM0(Q)	0.71	0.40	0.30	0.38	0.42	0.40	0.35	0.33	0.32	0.34
UEXM0 (H)	0.88	0.75	0.67	0.62	0.57	0.54	0.53	0.52	0.51	0.53
UEXMO(Y)	0.95	0.90	0.86	0.84	0.81	0.79	0.76	0.75	0.74	0.74
UNE (M)	0.55	0.53	0.47	0.49	0.49	0.45	0.40	0.42	0.39	0.36
UNE (Q)	0.92	0.82	0.72	0.70	0.68	0.65	0.61	0.59	0.55	0.51
UNE (H)	0.97	0.94	0.90	0.85	0.80	0.75	0.71	0.67	0.63	0.58
UNE (Y)	0.99	0.97	0.95	0.92	0.89	0.85	0.80	0.76	0.71	0.66
UEXUNE(Q)	0.92	0.83	0.76	0.73	0.70	0.67	0.64	0.62	0.58	0.54
UEXUNE(H)	0.97	0.93	0.88	0.84	0.79	0.75	0.72	0.68	0.64	0.59
UEXUNE(Y)	0.99	0.97	0.94	0.91	0.88	0.84	0.80	0.75	0.71	0.66
CPI (M)	0.16	0.04	0.16	0.00	0.11	0.08	-0.05	-0.02	-0.07	0.00
CPI (Q)	0.78	0.48	0.23	0.18	0.16	0.09	0.00	-0.07	-0.10	-0.16
CPI (H)	0.90	0.77	0.63	0.46	0.30	0.13	0.00	-0.10	-0.20	-0.28
CPI (Y)	0.93	0.84	0.75	0.63	0.50	0.36	0.22	0.09	-0.03	-0.13
UEXCPI(O)	0.75	0.42	0.19	0.15	0.14	0.09	0.00	-0.05	-0.06	-0.08
UEXCPI(H)	0.88	0.71	0.53	0.38	0.24	0.11	0.00	-0.07	-0.12	-0.16
UEXCPI(Y)	0.93	0.83	0.70	0.57	0.43	0.28	0.13	0.02	-0.06	-0.13
TBILL (M)	0.05	0.00	0.00	0.02	0.02	0.01	0.05	-0.06	0.06	-0.06
TBILL $(0)$	0.69	0.10	0.00	0.02	0.02	0.01	0.05	0.00	0.00	0.00
TBILL (Q)	0.05	0.40	0.00	0.05	0.05	0.04	0.05	0.01	0.02	-0.05
TBILL (II) $TBILL (V)$	0.03	0.71	0.55	0.57	0.24	0.07	0.07	0.05	0.00	0.05
$\mathbf{I} \mathbf{I} \mathbf{I} \mathbf{I} \mathbf{I} \mathbf{I} \mathbf{I} \mathbf{I} $	0.75	0.00	0.70	0.07	0.07	0.47	0.07	0.20	0.17	0.08
UEXTBILL(Q)	0.01	0.28	0.04	-0.01	-0.02	-0.02	0.02	-0.03	0.02	-0.04
$UEAIDILL(\Pi)$	0.80	0.39	0.58	0.20	0.07	0.00	0.00	-0.05	-0.02	-0.07
$\frac{\text{UEAIDILL}(1)}{\text{TEDM}(M)}$	0.90	0.78	0.03	0.35	0.40	0.29	0.10	0.07	0.01	-0.07
IERM(M)	-0.05	0.05	0.02	-0.01	0.01	0.00	0.11	-0.13	0.01	-0.21
TERM(Q)	0.66	0.37	0.04	0.03	0.05	0.03	0.03	-0.12	-0.15	-0.21
TERM(H)	0.83	0.70	0.53	0.36	0.17	-0.02	-0.05	-0.13	-0.15	-0.20
TERM(Y)	0.91	0.83	0.73	0.62	0.50	0.39	0.29	0.16	0.07	-0.04
UEXTERM(Q)	0.57	0.26	0.04	0.02	0.03	0.04	0.04	-0.11	-0.14	-0.19
UEXTERM(H)	0.77	0.58	0.39	0.22	0.09	0.00	-0.03	-0.12	-0.15	-0.19
UEXTERM(Y)	0.88	0.76	0.64	0.50	0.38	0.26	0.14	0.01	-0.07	-0.15
DY(M)	-0.01	0.09	0.06	-0.03	0.01	-0.03	0.09	0.00	0.10	-0.06
DY(Q)	0.70	0.43	0.10	0.04	0.03	0.03	0.10	0.10	0.09	-0.10
DY(H)	0.85	0.70	0.55	0.41	0.29	0.15	0.10	0.01	-0.06	-0.14
DY(Y)	0.92	0.82	0.72	0.62	0.53	0.44	0.36	0.26	0.17	0.08
UEXDY(Q)	0.57	0.27	0.07	0.02	0.02	0.05	0.10	0.07	0.07	-0.07
UEXDY(H)	0.77	0.57	0.42	0.27	0.18	0.13	0.11	0.02	-0.06	-0.15
UEXDY(Y)	0.89	0.78	0.68	0.56	0.44	0.33	0.22	0.09	-0.01	-0.10

Table 5.3 The autocorrelation of (unexpected) change of primary macroeconomic variables

Note: Table reports the first to tenth order autocorrelation of (unexpected) change of macroeconomic variables on monthly, quarterly, half-yearly, and annual horizons from 1980 to 2008, which is denoted in M, Q, H, and Y in parentheses.

	GDP	UEXGDP	IP	UEXIP	IPM	UEXIPM	M0	UEXM0	UNE	UEXUNE	CPI	UEXCPI	TBILL	UEXTBILL	TERM	UEXTERM	DY	UEXDY
GDP	1.00																	
UEXGDP	0.95	1.00																
IP	0.56	0.59	1.00															
UEXIP	0.46	0.52	0.89	1.00														
IPM	0.50	0.48	0.88	0.77	1.00													
UEXIPM	0.47	0.50	0.83	0.87	0.91	1.00												
M0	-0.13	-0.14	0.04	0.04	0.10	0.11	1.00											
UEXM0	-0.03	-0.04	-0.12	-0.11	0.02	0.01	0.57	1.00										
UNE	-0.29	-0.22	-0.59	-0.48	-0.71	-0.60	-0.34	-0.29	1.00									
UEXUNE	-0.31	-0.26	-0.63	-0.54	-0.76	-0.67	-0.33	-0.25	0.97	1.00								
CPI	0.06	0.03	0.20	0.10	0.31	0.22	0.24	0.12	-0.37	-0.40	1.00							
UEXCPI	0.10	0.10	0.39	0.26	0.52	0.40	0.44	0.21	-0.52	-0.55	0.73	1.00						
TBILL	-0.08	-0.13	0.16	0.06	0.25	0.13	0.14	-0.02	-0.36	-0.34	0.67	0.57	1.00					
UEXTBILL	-0.02	-0.04	0.27	0.18	0.38	0.29	0.36	0.09	-0.41	-0.43	0.62	0.74	0.83	1.00				
TERM	-0.04	0.05	-0.27	-0.16	-0.33	-0.20	-0.17	-0.01	0.49	0.47	-0.56	-0.56	-0.76	-0.71	1.00			
UEXTERM	-0.13	-0.06	-0.32	-0.24	-0.41	-0.31	-0.20	-0.02	0.47	0.47	-0.50	-0.58	-0.66	-0.74	0.88	1.00		
DY	-0.16	-0.25	-0.25	-0.33	-0.15	-0.24	0.24	0.18	-0.03	0.01	0.19	0.24	0.22	0.25	-0.18	-0.14	1.00	
UEXDY	0.01	-0.08	-0.09	-0.23	0.03	-0.11	0.25	0.19	-0.18	-0.13	0.28	0.41	0.31	0.36	-0.28	-0.23	0.88	1

Table 5.4 The co	orrelation coefficient	of annual (u	nexpected) cha	inge of macroecono	mic variables

Note: The table reports correlation coefficient of annual (unexpected) change of macroeconomic variables from 1980 to 2008.

Panel A Momentum coefficient $\gamma_{mom}$							Panel B Winner/Loser return R <sub>wml</sub>						
	Panel A	A1 Equal-weig	ghted	Panel A	2 Value-wei	ghted	Panel B	1 Equal-weig	ghted	Panel B	Panel B2 Value-weighted		
Macroeconomic variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	-0.0039	-1.1409	0.0060	-0.0074	-1.4854	0.0095	-0.0031	-1.0126	0.0003	-0.0088	-1.8050	0.0094	
UEXGDP(Y)	-0.0060	-1.1891	0.0060	-0.0123	-1.6742	0.0119	-0.0047	-1.0186	0.0003	-0.0140	-1.9577	0.0105	
IP(M)	-0.2352	-0.2800	-0.0028	-0.6027	-0.4140	-0.0025	0.7963	0.7288	-0.0019	0.1330	0.0806	-0.0029	
IP(Q)	-0.0808	-0.0885	-0.0029	0.1218	0.0823	-0.0029	1.0547	0.9499	0.0008	0.7896	0.4793	-0.0019	
IP(H)	-0.1247	-0.1979	-0.0027	-0.8984	-0.9163	0.0010	0.3857	0.5328	-0.0019	-0.8034	-0.7499	-0.0007	
IP(Y)	0.1826	0.4206	-0.0020	-0.5222	-0.8769	0.0002	0.4882	1.0687	0.0011	-0.7101	-1.0859	0.0010	
UEXIP(Q)	-0.0004	-0.0673	-0.0029	-0.0003	-0.0327	-0.0029	0.0079	1.0192	0.0016	0.0046	0.3910	-0.0022	
UEXIP(H)	-0.0001	-0.0244	-0.0029	-0.0029	-0.3327	-0.0023	0.0062	0.9013	0.0013	-0.0002	-0.0173	-0.0029	
UEXIP(Y)	0.0001	0.0189	-0.0029	-0.0049	-0.8040	0.0006	0.0036	0.7694	-0.0001	-0.0054	-0.8118	0.0001	
IPM(M)	-0.5221	-0.6563	-0.0021	-0.7620	-0.6335	-0.0023	0.4609	0.4364	-0.0026	-0.4187	-0.2809	-0.0028	
IPM(Q)	-0.0765	-0.0906	-0.0029	0.0619	0.0494	-0.0029	0.9866	0.9737	0.0008	0.3257	0.2356	-0.0027	
IPM(H)	0.0917	0.1526	-0.0028	-0.5341	-0.6133	-0.0013	0.6281	0.9192	0.0005	-0.5985	-0.6617	-0.0015	
IPM(Y)	0.2419	0.6329	-0.0006	-0.1813	-0.3735	-0.0024	0.4366	1.1715	0.0015	-0.3806	-0.6898	-0.0013	
UEXIPM(Q)	-0.0013	-0.2373	-0.0027	-0.0012	-0.1411	-0.0028	0.0061	0.8251	0.0000	0.0007	0.0699	-0.0029	
UEXIPM(H)	0.0002	0.0455	-0.0029	-0.0021	-0.2708	-0.0025	0.0061	0.9620	0.0019	-0.0012	-0.1503	-0.0028	
UEXIPM(Y)	0.0009	0.2716	-0.0025	-0.0022	-0.4507	-0.0020	0.0036	0.9139	0.0007	-0.0037	-0.6822	-0.0011	
M0 (M)	0.6687	0.2432	-0.0027	2.5565	0.6430	-0.0018	2.0466	0.3702	-0.0019	7.4335	1.1583	0.0036	
M0 (Q)	0.3479	0.1734	-0.0027	1.0711	0.3659	-0.0023	-2.2526	-0.6056	0.0012	-0.2933	-0.0629	-0.0029	
M0 (H)	1.2890	0.7378	0.0020	0.7841	0.3227	-0.0022	0.1317	0.0476	-0.0029	-0.9440	-0.2854	-0.0022	
M0 (Y)	1.4925	1.2477	0.0133	0.8119	0.5038	-0.0010	0.8537	0.4671	0.0002	0.0368	0.0162	-0.0029	
UEXM0(Q)	0.0000	0.6054	-0.0005	0.0001	1.4478	0.0094	0.0000	-0.0505	-0.0029	0.0001	1.0521	0.0079	
UEXM0 (H)	0.0000	0.7729	0.0017	0.0001	1.4441	0.0103	0.0000	-0.0164	-0.0029	0.0001	0.8593	0.0044	
UEXM0 (Y)	0.0000	1.0125	0.0052	0.0000	1.4915	0.0120	0.0000	0.2193	-0.0024	0.0000	0.9256	0.0051	
UNE (M)	-1.2676	-1.3415	0.0067	-0.8370	-0.5286	-0.0013	-1.8650	-1.7967	0.0093	-0.5203	-0.3253	-0.0025	
UNE (Q)	-0.5661	-1.2627	0.0087	-0.2236	-0.3392	-0.0022	-0.8807	-1.9877	0.0136	-0.1779	-0.2480	-0.0026	
UNE (H)	-0.3002	-1.2369	0.0082	-0.0277	-0.0820	-0.0029	-0.4246	-1.8007	0.0101	0.0736	0.2012	-0.0027	
UNE (Y)	-0.2121	-1.7758	0.0164	-0.1125	-0.6489	-0.0008	-0.2470	-2.2370	0.0124	-0.0473	-0.2519	-0.0027	
UEXUNE(Q)	-0.0424	-1.1042	0.0065	-0.0089	-0.1665	-0.0028	-0.0624	-1.7084	0.0090	0.0049	0.0870	-0.0029	
UEXUNE(H)	-0.0251	-1.0738	0.0061	0.0003	0.0109	-0.0029	-0.0360	-1.6666	0.0079	0.0096	0.2862	-0.0025	
UEXUNE(Y)	-0.0158	-1.2483	0.0081	-0.0022	-0.1230	-0.0028	-0.0199	-1.7740	0.0073	0.0044	0.2370	-0.0027	
CPI (M)	-0.0829	-0.8631	-0.0007	-0.3139	-1.8489	0.0099	-0.1247	-0.8909	0.0001	-0.4560	-1.8177	0.0159	
CPI (Q)	-0.0394	-0.6287	-0.0010	-0.1630	-1.6596	0.0099	-0.1198	-1.2027	0.0074	-0.3324	-2.0761	0.0344	
CPI (H)	-0.0289	-0.5292	-0.0004	-0.0935	-1.2467	0.0075	-0.0810	-1.1039	0.0086	-0.1868	-1.7824	0.0261	
CPI (Y)	0.0014	0.0347	-0.0029	-0.0282	-0.5924	-0.0007	-0.0128	-0.2518	-0.0022	-0.0676	-1.0669	0.0059	
UEXCPI(Q)	0.0064	0.7854	-0.0001	-0.0024	-0.1922	-0.0028	0.0021	0.2356	-0.0027	-0.0099	-0.6873	-0.0011	

 Table 5.5 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 6/3)

Continued												
UEXCPI(H)	0.0061	1.0402	0.0021	-0.0016	-0.1653	-0.0028	0.0023	0.3445	-0.0025	-0.0083	-0.7563	-0.0003
UEXCPI(Y)	0.0051	1.3560	0.0059	0.0003	0.0497	-0.0029	0.0036	0.8477	-0.0004	-0.0031	-0.4496	-0.0020
TBILL(M)	0.0925	1.1201	0.0008	0.0058	0.0484	-0.0029	0.1155	1.1494	0.0005	0.0327	0.2300	-0.0028
TBILL (Q)	0.0748	1.4967	0.0054	0.0208	0.2809	-0.0027	0.0884	1.4388	0.0039	0.0136	0.1383	-0.0028
TBILL (H)	0.0682	1.8967	0.0118	0.0544	1.1174	0.0008	0.0710	1.5907	0.0064	0.0425	0.6342	-0.0013
TBILL (Y)	0.0695	2.3548	0.0299	0.0714	1.7942	0.0108	0.0697	1.7469	0.0164	0.0758	1.5320	0.0079
UEXTBILL(Q)	16.3102	1.8663	0.0095	9.8760	0.7897	-0.0011	20.1170	2.1122	0.0081	12.8594	0.9479	-0.0008
UEXTBILL(H)	16.7139	2.3077	0.0197	13.8369	1.3712	0.0032	19.2747	2.4445	0.0147	14.5147	1.2405	0.0018
UEXTBILL(Y)	14.8326	2.7613	0.0307	12.9689	1.7375	0.0072	15.0528	2.6134	0.0173	11.9432	1.3396	0.0031
TERM(M)	-1.4885	-1.3746	0.0038	-1.7581	-1.1985	0.0008	-1.4735	-1.3221	0.0009	-1.5068	-0.9441	-0.0010
TERM(Q)	-1.0896	-1.3996	0.0073	-0.7711	-0.7340	-0.0009	-1.0051	-1.1984	0.0022	-0.5213	-0.4346	-0.0023
TERM(H)	-0.3983	-0.7149	-0.0001	-0.2275	-0.3157	-0.0025	-0.2795	-0.4683	-0.0021	0.0145	0.0166	-0.0029
TERM(Y)	-0.5722	-1.5508	0.0093	-0.7410	-1.5163	0.0052	-0.5130	-1.1142	0.0028	-0.6287	-1.0311	0.0011
UEXTERM(Q)	-1.6128	-1.4569	0.0087	-1.4088	-0.9594	0.0006	-1.5202	-1.3180	0.0031	-1.1517	-0.6979	-0.0013
UEXTERM(H)	-1.0369	-1.1764	0.0051	-0.7323	-0.6362	-0.0013	-0.8724	-0.9286	0.0004	-0.3684	-0.2693	-0.0026
UEXTERM(Y)	-0.7237	-1.1207	0.0043	-0.7089	-0.8577	-0.0002	-0.5510	-0.7629	-0.0005	-0.4009	-0.3980	-0.0023
DY(M)	0.1893	1.3977	0.0023	0.0056	0.0294	-0.0029	0.1074	0.6144	-0.0019	-0.1701	-0.7740	-0.0018
DY(Q)	0.2744	2.1819	0.0316	0.2080	1.2453	0.0049	0.3043	1.8197	0.0219	0.1862	0.8480	0.0015
DY(H)	0.2677	2.5720	0.0739	0.3080	2.2310	0.0373	0.3210	2.1165	0.0617	0.3571	2.1184	0.0347
DY(Y)	0.0772	1.3614	0.0120	0.1003	1.3246	0.0070	0.0533	0.8466	0.0012	0.1012	1.0856	0.0042
UEXDY(Q)	8.0493	2.9349	0.0250	4.1975	1.1069	0.0001	7.5615	2.1876	0.0115	1.9027	0.4181	-0.0025
UEXDY(H)	9.1051	3.5602	0.0605	7.9863	2.1840	0.0164	9.6266	3.0132	0.0386	7.7363	1.7226	0.0097
UEXDY(Y)	6.0069	3.3907	0.0554	5.8504	2.2993	0.0189	5.7236	2.7977	0.0280	5.6933	1.8821	0.0115
EMR(-1)/VMR(-1)	-0.0368	-1.1835	0.0112	0.0148	0.1976	-0.0026	-0.0359	-0.7573	0.0050	0.0388	0.4262	-0.0013
EMR(-2)/VMR(-2)	-0.0061	-0.2870	-0.0020	-0.0257	-0.6302	-0.0001	-0.0032	-0.1234	-0.0028	-0.0291	-0.5439	-0.0004
EMR(-3)/VMR(-3)	-0.0028	-0.1886	-0.0025	0.0015	0.0609	-0.0029	-0.0010	-0.0583	-0.0029	0.0058	0.1806	-0.0027
EMR(-4)/VMR(-4)	-0.0042	-0.4906	-0.0008	-0.0031	-0.2105	-0.0027	-0.0054	-0.5508	-0.0009	-0.0076	-0.4002	-0.0020
EMR(-5)/VMR(-5)	-0.0031	-0.5227	-0.0004	0.0023	0.2226	-0.0027	-0.0043	-0.6271	-0.0001	-0.0039	-0.2862	-0.0024
EMR(-6)/VMR(-6)	-0.0046	-1.2614	0.0077	-0.0044	-0.6475	-0.0011	-0.0049	-1.1954	0.0043	-0.0073	-0.8823	0.0006
EMR(-7)/VMR(-7)	-0.0025	-0.9523	0.0024	-0.0033	-0.6461	-0.0011	-0.0028	-0.9727	0.0012	-0.0056	-0.9374	0.0008
EMR(-8)/VMR(-8)	-0.0007	-0.3537	-0.0022	-0.0024	-0.5789	-0.0012	-0.0011	-0.4405	-0.0020	-0.0055	-1.1616	0.0033
EMR(-9)/VMR(-9)	-0.0002	-0.1656	-0.0028	-0.0011	-0.3588	-0.0023	0.0001	0.0299	-0.0029	-0.0021	-0.6140	-0.0013
EMR(-10)/VMR(-10)	-0.0007	-0.6467	-0.0002	0.0000	-0.0028	-0.0029	-0.0006	-0.4188	-0.0018	-0.0010	-0.3867	-0.0023

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with shares returns in past 6 months, and hold for 3 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal/or value-weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return R<sub>wml</sub>.

Panel A Momentum coefficient $\gamma_{mom}$							Panel B Winner/Loser return R <sub>wml</sub>							
	Panel A	A1 Equal-weig	ghted	Panel A	2 Value-wei	ghted	Panel	Panel B1 Equal-weighted				Panel B2 Value-weighted		
Macroeconomic variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>		
GDP(Y)	-0.0106	-1.4968	0.0255	-0.0143	-1.7168	0.0225	-0.0095	-1.4537	0.0137	-0.0166	-2.0315	0.0212		
UEXGDP(Y)	-0.0135	-1.3426	0.0166	-0.0202	-1.5956	0.0188	-0.0118	-1.1734	0.0080	-0.0234	-1.8767	0.0177		
IP(M)	-0.9386	-0.8651	-0.0019	-0.5420	-0.3119	-0.0028	-0.1435	-0.1043	-0.0029	0.0161	0.0073	-0.0029		
IP(Q)	-1.0612	-0.8410	-0.0002	-0.9947	-0.6118	-0.0018	-0.3458	-0.2480	-0.0027	-0.4249	-0.2247	-0.0028		
IP(H)	-0.7873	-0.7519	0.0003	-1.9032	-1.6546	0.0065	-0.5229	-0.4958	-0.0019	-2.3185	-1.6638	0.0069		
IP(Y)	0.0832	0.1106	-0.0029	-1.0140	-1.2557	0.0034	0.2314	0.3118	-0.0025	-1.4460	-1.5286	0.0060		
UEXIP(Q)	-0.0068	-0.8333	-0.0005	-0.0041	-0.3632	-0.0025	-0.0015	-0.1607	-0.0029	-0.0004	-0.0268	-0.0029		
UEXIP(H)	-0.0066	-0.8466	0.0005	-0.0105	-1.1348	0.0014	-0.0030	-0.3619	-0.0024	-0.0110	-0.9779	0.0004		
UEXIP(Y)	-0.0025	-0.3853	-0.0019	-0.0102	-1.4110	0.0053	-0.0009	-0.1372	-0.0028	-0.0140	-1.6266	0.0079		
IPM(M)	-0.6774	-0.6519	-0.0024	-0.7855	-0.5382	-0.0026	-0.1370	-0.1150	-0.0029	-0.5462	-0.2845	-0.0028		
IPM(Q)	-0.0081	-0.0062	-0.0029	-0.1258	-0.0826	-0.0029	0.6010	0.4496	-0.0022	-0.0397	-0.0228	-0.0029		
IPM(H)	-0.1210	-0.1190	-0.0028	-0.5979	-0.5560	-0.0018	0.0728	0.0745	-0.0029	-0.9552	-0.7763	-0.0009		
IPM(Y)	0.3222	0.5061	-0.0011	-0.2598	-0.3887	-0.0024	0.4057	0.6390	-0.0009	-0.5247	-0.6915	-0.0013		
UEXIPM(Q)	-0.0016	-0.1916	-0.0028	-0.0013	-0.1300	-0.0029	0.0020	0.2293	-0.0028	-0.0012	-0.0955	-0.0029		
UEXIPM(H)	-0.0011	-0.1387	-0.0028	-0.0029	-0.3228	-0.0026	0.0015	0.1918	-0.0028	-0.0048	-0.4637	-0.0022		
UEXIPM(Y)	0.0004	0.0725	-0.0029	-0.0035	-0.5408	-0.0017	0.0012	0.2128	-0.0027	-0.0067	-0.9029	0.0003		
M0 (M)	0.7065	0.1723	-0.0028	1.1823	0.2090	-0.0028	0.6645	0.0941	-0.0029	2.9002	0.3108	-0.0024		
M0 (Q)	1.4838	0.4331	-0.0016	1.6168	0.3479	-0.0021	-0.8328	-0.1514	-0.0026	-0.7774	-0.1105	-0.0028		
M0 (H)	1.5083	0.5257	0.0000	0.2381	0.0695	-0.0029	-0.2531	-0.0632	-0.0029	-3.2445	-0.6911	0.0018		
M0 (Y)	2.1757	1.0051	0.0121	0.9639	0.3608	-0.0015	1.4122	0.5123	0.0017	-0.5070	-0.1436	-0.0027		
UEXM0(Q)	0.0000	0.4583	-0.0010	0.0001	0.9522	0.0041	0.0000	0.0701	-0.0029	0.0001	0.5818	0.0012		
UEXM0 (H)	0.0000	0.4126	-0.0010	0.0001	0.8495	0.0037	0.0000	-0.0263	-0.0029	0.0000	0.3795	-0.0011		
UEXM0 (Y)	0.0000	0.4362	-0.0006	0.0000	0.7090	0.0021	0.0000	0.0459	-0.0029	0.0000	0.3160	-0.0017		
UNE (M)	-1.1803	-0.7226	0.0006	-1.3653	-0.7533	-0.0006	-1.8008	-1.1176	0.0030	-1.0714	-0.5373	-0.0019		
UNE (Q)	-0.5811	-0.7639	0.0023	-0.3759	-0.4428	-0.0019	-0.8804	-1.1926	0.0058	-0.3149	-0.3295	-0.0024		
UNE (H)	-0.3492	-0.8793	0.0036	-0.2211	-0.4859	-0.0016	-0.4133	-1.1193	0.0037	-0.0709	-0.1430	-0.0028		
UNE (Y)	-0.3155	-1.5311	0.0157	-0.2321	-0.9062	0.0021	-0.3327	-1.8040	0.0121	-0.1568	-0.5833	-0.0013		
UEXUNE(Q)	-0.0443	-0.6644	0.0015	-0.0187	-0.2631	-0.0025	-0.0624	-0.9313	0.0035	-0.0091	-0.1158	-0.0029		
UEXUNE(H)	-0.0277	-0.6953	0.0018	-0.0090	-0.2069	-0.0027	-0.0349	-0.9447	0.0025	0.0002	0.0043	-0.0029		
UEXUNE(Y)	-0.0216	-1.0022	0.0060	-0.0099	-0.3969	-0.0020	-0.0233	-1.2149	0.0046	-0.0025	-0.0969	-0.0029		
CPI (M)	-0.1675	-0.8967	0.0011	-0.4411	-1.7308	0.0109	-0.3386	-1.4985	0.0090	-0.6844	-1.8987	0.0204		
CPI (Q)	-0.1160	-0.8354	0.0042	-0.3129	-1.7912	0.0230	-0.2322	-1.3276	0.0179	-0.5265	-2.1626	0.0485		
CPI (H)	-0.0916	-0.8016	0.0081	-0.2211	-1.7242	0.0290	-0.1654	-1.2173	0.0232	-0.3581	-2.1437	0.0559		
CPI (Y)	-0.0215	-0.2808	-0.0015	-0.0620	-0.7504	0.0029	-0.0359	-0.4059	-0.0001	-0.1087	-1.0406	0.0097		
UEXCPI(Q)	0.0066	0.4742	-0.0017	-0.0111	-0.5510	-0.0011	0.0000	-0.0022	-0.0029	-0.0169	-0.7069	0.0000		

 Table 5.6 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 6/6)

Continued												
UEXCPI(H)	0.0074	0.6231	0.0003	-0.0079	-0.4955	-0.0011	0.0018	0.1371	-0.0028	-0.0153	-0.8257	0.0019
UEXCPI(Y)	0.0069	0.9419	0.0040	-0.0019	-0.1983	-0.0027	0.0045	0.5628	-0.0008	-0.0058	-0.5352	-0.0012
TBILL(M)	0.1925	1.4648	0.0040	0.0475	0.2723	-0.0027	0.1841	1.2504	0.0017	0.0887	0.4115	-0.0024
TBILL (Q)	0.1221	1.5927	0.0067	0.0512	0.4829	-0.0021	0.1074	1.1563	0.0025	0.0394	0.2778	-0.0026
TBILL (H)	0.0880	1.3236	0.0078	0.0794	1.0194	0.0014	0.0637	0.8207	0.0011	0.0510	0.4906	-0.0017
TBILL (Y)	0.0983	2.2526	0.0257	0.1115	1.9419	0.0154	0.0981	1.9223	0.0178	0.1253	1.8193	0.0133
UEXTBILL(Q)	30.1138	2.2078	0.0156	21.6084	1.1702	0.0018	31.2654	2.0476	0.0116	27.0071	1.2601	0.0023
UEXTBILL(H)	25.4710	2.2313	0.0200	21.5627	1.4575	0.0052	24.0763	1.8657	0.0120	21.7688	1.2394	0.0029
UEXTBILL(Y)	21.5149	2.3550	0.0280	20.7564	1.7963	0.0114	19.9467	2.0138	0.0164	19.6420	1.4529	0.0060
TERM(M)	-2.3172	-1.1841	0.0041	-3.4679	-1.4589	0.0049	-1.7874	-0.9272	0.0001	-3.3972	-1.3077	0.0023
TERM(Q)	-1.3980	-1.0302	0.0044	-1.4688	-0.9200	0.0011	-1.1264	-0.7779	0.0005	-1.4916	-0.7798	0.0000
TERM(H)	-0.1105	-0.1094	-0.0028	-0.3053	-0.2831	-0.0026	0.1436	0.1415	-0.0028	-0.2307	-0.1817	-0.0028
TERM(Y)	-0.5154	-0.8581	0.0014	-0.7491	-1.0338	0.0016	-0.6342	-1.0426	0.0018	-0.9832	-1.1677	0.0025
UEXTERM(Q)	-2.2652	-1.1408	0.0070	-2.7595	-1.1907	0.0044	-1.7892	-0.8575	0.0016	-2.7711	-1.0234	0.0023
UEXTERM(H)	-1.0030	-0.6197	0.0003	-1.2815	-0.7258	-0.0003	-0.6124	-0.3732	-0.0021	-1.1557	-0.5429	-0.0014
UEXTERM(Y)	-0.6178	-0.5472	-0.0006	-0.9397	-0.7586	-0.0003	-0.5036	-0.4483	-0.0018	-0.9590	-0.6497	-0.0010
DY(M)	0.4741	1.9774	0.0113	0.2930	1.0839	-0.0002	0.3924	1.2438	0.0041	0.3031	0.9171	-0.0009
DY(Q)	0.5710	2.3128	0.0623	0.5544	2.0052	0.0276	0.5726	1.7609	0.0447	0.6160	1.8328	0.0235
DY(H)	0.3695	2.1949	0.0602	0.3647	1.7384	0.0277	0.3367	1.7348	0.0352	0.4056	1.6756	0.0236
DY(Y)	0.0724	0.6752	0.0028	0.1090	0.8391	0.0035	-0.0050	-0.0464	-0.0029	0.0879	0.5786	0.0000
UEXDY(Q)	16.4675	3.0277	0.0479	13.6966	2.3026	0.0145	15.1404	2.1661	0.0283	13.3919	1.8674	0.0088
UEXDY(H)	14.8669	3.1306	0.0706	13.9029	2.4994	0.0290	13.4909	2.3860	0.0411	14.1701	2.1706	0.0204
UEXDY(Y)	8.2067	2.5706	0.0444	8.1492	2.0731	0.0203	6.0521	1.7996	0.0158	7.9613	1.7776	0.0126
EMR(-1)/VMR(-1)	-0.0867	-1.5930	0.0304	0.0370	0.2975	-0.0018	-0.0843	-1.1649	0.0200	0.0510	0.3230	-0.0015
EMR(-2)/VMR(-2)	-0.0123	-0.2889	-0.0013	0.0152	0.2063	-0.0024	-0.0057	-0.1172	-0.0027	0.0152	0.1585	-0.0026
EMR(-3)/VMR(-3)	-0.0018	-0.0616	-0.0029	0.0223	0.5170	0.0000	0.0008	0.0247	-0.0029	0.0282	0.5049	0.0003
EMR(-4)/VMR(-4)	-0.0069	-0.3869	-0.0005	0.0046	0.1820	-0.0027	-0.0115	-0.6274	0.0019	-0.0014	-0.0444	-0.0029
EMR(-5)/VMR(-5)	-0.0037	-0.3094	-0.0014	0.0140	0.7835	0.0023	-0.0074	-0.5967	0.0017	0.0050	0.2218	-0.0025
EMR(-6)/VMR(-6)	-0.0067	-0.9262	0.0069	-0.0012	-0.1057	-0.0029	-0.0082	-1.0859	0.0078	-0.0042	-0.3198	-0.0023
EMR(-7)/VMR(-7)	-0.0029	-0.5493	0.0003	0.0008	0.1017	-0.0029	-0.0040	-0.7131	0.0015	-0.0026	-0.2749	-0.0025
EMR(-8)/VMR(-8)	-0.0014	-0.3382	-0.0017	-0.0020	-0.3002	-0.0023	-0.0020	-0.4474	-0.0011	-0.0060	-0.8124	0.0011
EMR(-9)/VMR(-9)	0.0001	0.0410	-0.0029	0.0016	0.3304	-0.0022	0.0004	0.1101	-0.0028	-0.0004	-0.0782	-0.0029
EMR(-10)/VMR(-10)	-0.0008	-0.3577	-0.0016	0.0029	0.7813	0.0010	-0.0010	-0.3607	-0.0015	0.0015	0.3621	-0.0022

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with shares returns in past 6 months, and hold for 6 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return  $R_{wml}$ .

	Panel A Momentum coefficient $\gamma_{mom}$ Panel B Winner/Loser return $R_{wml}$ Panel A1 Equal-weighted       Panel A2 Value-weighted       Panel B1 Equal-weighted       Panel B2 Value-weighted											
	Panel A	A1 Equal-weig	ghted	Panel A	2 Value-wei	ghted	Panel I	31 Equal-weig	ghted	Panel H	32 Value-wei	ghted
Macroeconomic variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	-0.0259	-2.1111	0.0687	-0.0248	-2.0871	0.0364	-0.0282	-2.3645	0.0539	-0.0247	-2.1335	0.0231
UEXGDP(Y)	-0.0344	-1.9043	0.0514	-0.0375	-2.0085	0.0357	-0.0367	-2.0602	0.0381	-0.0355	-1.9231	0.0201
IP(M)	-1.9477	-1.1416	-0.0011	-1.1268	-0.4897	-0.0026	-1.0085	-0.4414	-0.0027	0.3851	0.1197	-0.0030
IP(Q)	-1.8315	-0.8511	0.0004	-1.4123	-0.5886	-0.0018	-1.1302	-0.4385	-0.0021	0.0734	0.0255	-0.0030
IP(H)	-1.0283	-0.5757	-0.0006	-2.1660	-1.2270	0.0032	-0.9162	-0.4503	-0.0018	-2.1781	-0.9991	0.0012
IP(Y)	0.0160	0.0137	-0.0030	-1.0126	-0.7600	0.0002	-0.1380	-0.1021	-0.0029	-1.0524	-0.7167	-0.0007
UEXIP(Q)	-0.0132	-1.0072	0.0009	-0.0094	-0.6019	-0.0018	-0.0073	-0.4556	-0.0022	0.0003	0.0153	-0.0030
UEXIP(H)	-0.0115	-0.8174	0.0014	-0.0140	-0.9527	0.0009	-0.0088	-0.5438	-0.0013	-0.0104	-0.5795	-0.0016
UEXIP(Y)	-0.0045	-0.3997	-0.0016	-0.0113	-0.9379	0.0021	-0.0051	-0.3986	-0.0018	-0.0120	-0.8392	0.0009
IPM(M)	-1.3892	-0.7719	-0.0020	-0.8757	-0.4339	-0.0028	-1.1914	-0.5276	-0.0025	0.3770	0.1381	-0.0030
IPM(Q)	0.1189	0.0606	-0.0030	0.2217	0.0976	-0.0030	0.9975	0.4562	-0.0022	1.5651	0.6395	-0.0018
IPM(H)	-0.1048	-0.0675	-0.0030	-0.4030	-0.2445	-0.0027	-0.0874	-0.0504	-0.0030	-0.1156	-0.0613	-0.0030
IPM(Y)	0.6856	0.7487	0.0005	0.3186	0.3006	-0.0025	0.6310	0.6036	-0.0010	0.5498	0.4933	-0.0021
UEXIPM(Q)	-0.0044	-0.3442	-0.0025	-0.0025	-0.1664	-0.0029	-0.0006	-0.0455	-0.0030	0.0059	0.3415	-0.0027
UEXIPM(H)	-0.0022	-0.1731	-0.0028	-0.0025	-0.1727	-0.0029	-0.0004	-0.0311	-0.0030	0.0023	0.1482	-0.0029
UEXIPM(Y)	0.0021	0.2227	-0.0026	0.0001	0.0053	-0.0030	0.0014	0.1338	-0.0029	0.0011	0.0979	-0.0029
M0 (M)	-1.0925	-0.1673	-0.0029	2.1382	0.2855	-0.0028	-1.7579	-0.1629	-0.0028	2.5555	0.1852	-0.0028
M0 (Q)	-0.1454	-0.0289	-0.0030	1.9601	0.3399	-0.0024	-5.5854	-0.7276	0.0024	-5.8861	-0.6046	0.0006
M0 (H)	0.2902	0.0587	-0.0029	-1.8638	-0.3965	-0.0019	-3.4026	-0.4739	0.0012	-9.2032	-1.2776	0.0155
M0 (Y)	2.3657	0.5914	0.0046	0.7460	0.1720	-0.0025	0.7610	0.1436	-0.0025	-3.0032	-0.5084	0.0019
UEXM0(Q)	0.0000	-0.1707	-0.0027	0.0001	0.6013	0.0001	-0.0001	-0.4172	-0.0006	0.0000	0.0843	-0.0029
UEXM0 (H)	0.0000	-0.3471	-0.0012	0.0000	0.2307	-0.0024	-0.0001	-0.6130	0.0039	0.0000	-0.3345	-0.0013
UEXM0 (Y)	0.0000	-0.2538	-0.0017	0.0000	0.0749	-0.0029	0.0000	-0.4388	0.0014	0.0000	-0.3846	-0.0004
UNE (M)	-0.6093	-0.2484	-0.0026	-3.4816	-1.2312	0.0047	-0.9610	-0.3562	-0.0023	-2.8573	-1.0278	0.0004
UNE (Q)	-0.4864	-0.4411	-0.0014	-1.2839	-0.9463	0.0035	-0.6798	-0.5681	-0.0010	-1.1860	-0.8862	0.0007
UNE (H)	-0.5070	-0.8555	0.0029	-0.7012	-0.9458	0.0037	-0.4866	-0.7710	0.0006	-0.5400	-0.7523	-0.0004
UNE (Y)	-0.5069	-1.4192	0.0175	-0.4876	-1.1648	0.0083	-0.4555	-1.2778	0.0080	-0.3560	-0.8935	0.0010
UEXUNE(Q)	-0.0396	-0.4105	-0.0015	-0.0923	-0.8154	0.0019	-0.0497	-0.4795	-0.0014	-0.0863	-0.8007	-0.0001
UEXUNE(H)	-0.0357	-0.6174	0.0004	-0.0516	-0.7474	0.0012	-0.0380	-0.6192	-0.0005	-0.0460	-0.7090	-0.0008
UEXUNE(Y)	-0.0352	-1.0538	0.0071	-0.0331	-0.8535	0.0024	-0.0319	-0.9358	0.0025	-0.0245	-0.6709	-0.0010
CPI (M)	-0.3468	-0.9318	0.0043	-0.6953	-1.5969	0.0145	-0.5751	-1.1720	0.0104	-0.9729	-1.5902	0.0199
CPI (Q)	-0.2294	-0.8029	0.0089	-0.4929	-1.5271	0.0299	-0.3963	-1.0667	0.0207	-0.7266	-1.6542	0.0447
CPI (H)	-0.1263	-0.6579	0.0059	-0.2745	-1.2900	0.0223	-0.2245	-0.9538	0.0158	-0.4327	-1.5930	0.0388
CPI (Y)	0.0149	0.1293	-0.0027	-0.0026	-0.0200	-0.0030	-0.0049	-0.0357	-0.0030	-0.0311	-0.1969	-0.0025
UEXCPI(Q)	0.0142	0.5174	-0.0005	-0.0213	-0.6552	0.0004	0.0100	0.3014	-0.0022	-0.0209	-0.5383	-0.0008

 Table 5.7 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 6/12)

Continued												
UEXCPI(H)	0.0166	0.7544	0.0040	-0.0107	-0.4202	-0.0013	0.0133	0.5009	0.0000	-0.0142	-0.4783	-0.0009
UEXCPI(Y)	0.0142	1.0492	0.0096	0.0005	0.0357	-0.0030	0.0132	0.8277	0.0042	-0.0005	-0.0287	-0.0030
TBILL(M)	0.2971	1.5024	0.0041	0.0971	0.4334	-0.0025	0.3364	1.3299	0.0030	0.1896	0.6731	-0.0018
TBILL (Q)	0.2357	1.8903	0.0123	0.1731	1.1468	0.0019	0.2179	1.2540	0.0057	0.1737	0.8278	0.0003
TBILL (H)	0.1762	1.7568	0.0151	0.2129	1.7656	0.0128	0.1296	1.0166	0.0035	0.1803	1.1821	0.0046
TBILL (Y)	0.1799	2.8764	0.0378	0.2213	2.7305	0.0340	0.1782	2.4132	0.0237	0.2533	2.6942	0.0292
UEXTBILL(Q)	50.7535	2.5972	0.0194	38.3818	1.5370	0.0047	57.0940	2.1547	0.0159	51.9209	1.6902	0.0063
UEXTBILL(H)	44.2036	2.4440	0.0264	42.2632	1.8662	0.0131	43.2962	1.8446	0.0158	45.7082	1.7298	0.0095
UEXTBILL(Y)	35.8685	2.4573	0.0336	38.8701	2.0362	0.0227	33.4115	1.9430	0.0182	40.5404	1.9976	0.0156
TERM(M)	-2.7061	-1.1689	0.0011	-3.2224	-1.0849	0.0005	-1.4223	-0.5549	-0.0022	-2.7330	-0.8205	-0.0013
TERM(Q)	-1.4073	-0.7911	0.0002	-1.2459	-0.5859	-0.0015	-0.7346	-0.3535	-0.0024	-1.1380	-0.4487	-0.0022
TERM(H)	0.0535	0.0377	-0.0030	-0.4663	-0.2834	-0.0026	0.6251	0.4438	-0.0021	-0.4383	-0.2499	-0.0027
TERM(Y)	-0.4208	-0.4524	-0.0018	-0.8368	-0.7884	-0.0001	-0.5877	-0.5710	-0.0014	-1.4038	-1.2001	0.0024
UEXTERM(Q)	-2.4480	-0.9879	0.0020	-2.3972	-0.8018	-0.0001	-1.5240	-0.5321	-0.0017	-2.1778	-0.6199	-0.0014
UEXTERM(H)	-0.9348	-0.4341	-0.0018	-1.1977	-0.4733	-0.0018	-0.0966	-0.0416	-0.0030	-0.9892	-0.3436	-0.0025
UEXTERM(Y)	-0.5302	-0.3401	-0.0023	-1.0795	-0.5936	-0.0012	-0.3327	-0.2142	-0.0028	-1.3477	-0.6982	-0.0011
DY(M)	0.4707	1.2527	0.0029	0.1584	0.3603	-0.0026	0.2531	0.5356	-0.0019	0.0844	0.1740	-0.0029
DY(Q)	0.5054	1.4052	0.0185	0.3609	0.7976	0.0035	0.4484	0.9932	0.0083	0.3883	0.7460	0.0020
DY(H)	0.3389	1.1482	0.0196	0.2016	0.5862	0.0018	0.2659	0.7989	0.0063	0.1867	0.4923	-0.0003
DY(Y)	0.0612	0.3112	-0.0013	-0.0074	-0.0336	-0.0030	-0.0560	-0.2709	-0.0020	-0.0922	-0.3947	-0.0014
UEXDY(Q)	16.6050	1.9317	0.0188	9.4587	0.9103	0.0012	12.6312	1.2241	0.0054	8.2751	0.7446	-0.0008
UEXDY(H)	14.4172	1.7828	0.0263	10.4949	1.1187	0.0063	11.1307	1.2084	0.0086	9.6734	0.9643	0.0023
UEXDY(Y)	7.2515	1.1554	0.0127	4.1790	0.6168	0.0001	3.5120	0.5338	-0.0005	2.4362	0.3501	-0.0023
EMR(-1)/VMR(-1)	-0.1484	-1.3260	0.0375	0.0783	0.3161	-0.0006	-0.1742	-1.2136	0.0342	0.0822	0.2870	-0.0012
EMR(-2)/VMR(-2)	0.0028	0.0328	-0.0030	0.0780	0.5399	0.0040	-0.0096	-0.1003	-0.0027	0.0778	0.4518	0.0017
EMR(-3)/VMR(-3)	0.0109	0.1969	-0.0018	0.0353	0.4445	0.0007	0.0048	0.0787	-0.0028	0.0411	0.4341	0.0004
EMR(-4)/VMR(-4)	-0.0041	-0.1373	-0.0026	0.0088	0.1954	-0.0025	-0.0201	-0.6127	0.0027	-0.0089	-0.1620	-0.0026
EMR(-5)/VMR(-5)	0.0000	0.0018	-0.0030	0.0234	0.7806	0.0045	-0.0106	-0.4927	0.0006	0.0091	0.2547	-0.0022
EMR(-6)/VMR(-6)	-0.0056	-0.4329	-0.0001	0.0037	0.1937	-0.0026	-0.0119	-0.8300	0.0057	-0.0030	-0.1401	-0.0028
EMR(-7)/VMR(-7)	-0.0019	-0.1841	-0.0024	0.0064	0.4537	-0.0011	-0.0072	-0.6407	0.0025	0.0011	0.0687	-0.0030
EMR(-8)/VMR(-8)	-0.0021	-0.2564	-0.0019	-0.0005	-0.0408	-0.0030	-0.0061	-0.6402	0.0032	-0.0058	-0.4630	-0.0012
EMR(-9)/VMR(-9)	0.0023	0.4255	-0.0005	0.0036	0.4017	-0.0012	0.0009	0.1440	-0.0027	0.0004	0.0430	-0.0030
EMR(-10)/VMR(-10)	0.0002	0.0516	-0.0030	0.0067	1.0287	0.0072	-0.0010	-0.1913	-0.0024	0.0055	0.7933	0.0017

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with shares returns in past 6 months, and hold for 12 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return  $R_{wml}$ .

	Pane	l A Momentu	m coefficient	$\gamma_{\rm mom}$				Pane	el B Winner/L	oser return R <sub>wml</sub>		
	Panel A	A1 Equal-weig	ghted	Panel A	A2 Value-wei	ghted	Panel I	31 Equal-weig	ghted	Panel I	32 Value-wei	ghted
Macroeconomic variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	-0.0411	-2.0985	0.1032	-0.0344	-2.4097	0.0530	-0.0455	-2.4277	0.0886	-0.0374	-2.7118	0.0408
UEXGDP(Y)	-0.0545	-1.9449	0.0770	-0.0530	-2.3587	0.0541	-0.0616	-2.2124	0.0689	-0.0555	-2.5813	0.0382
IP(M)	-2.8360	-1.3315	-0.0007	-2.1249	-0.8766	-0.0021	-1.4719	-0.5232	-0.0026	-0.0755	-0.0229	-0.0030
IP(Q)	-3.0599	-1.2306	0.0026	-3.0828	-1.2308	0.0012	-2.0236	-0.7459	-0.0013	-1.2953	-0.3880	-0.0026
IP(H)	-1.5714	-0.7767	0.0002	-2.1941	-1.2127	0.0017	-1.5796	-0.6811	-0.0008	-2.3763	-0.9399	0.0006
IP(Y)	-0.6307	-0.4366	-0.0018	-1.8734	-1.3016	0.0051	-1.2777	-0.7673	0.0005	-2.1409	-1.2904	0.0040
UEXIP(Q)	-0.0218	-1.3792	0.0030	-0.0205	-1.2142	0.0010	-0.0143	-0.8077	-0.0012	-0.0096	-0.4454	-0.0025
UEXIP(H)	-0.0166	-1.0045	0.0024	-0.0189	-1.2260	0.0022	-0.0129	-0.7014	-0.0008	-0.0163	-0.7795	-0.0005
UEXIP(Y)	-0.0084	-0.6324	-0.0003	-0.0152	-1.2341	0.0038	-0.0117	-0.7622	0.0007	-0.0177	-1.1051	0.0031
IPM(M)	-1.5767	-0.6465	-0.0023	-0.4888	-0.2349	-0.0030	-2.1164	-0.7097	-0.0021	0.8314	0.2982	-0.0029
IPM(Q)	-0.3683	-0.1479	-0.0030	0.4376	0.1723	-0.0029	0.0200	0.0077	-0.0030	1.9245	0.6442	-0.0018
IPM(H)	-0.0142	-0.0079	-0.0030	0.9755	0.5436	-0.0019	-0.4068	-0.2008	-0.0029	1.2187	0.5233	-0.0019
IPM(Y)	0.5556	0.5304	-0.0017	0.6431	0.5299	-0.0017	0.1287	0.1088	-0.0030	1.0062	0.7301	-0.0009
UEXIPM(Q)	-0.0070	-0.4255	-0.0023	-0.0003	-0.0207	-0.0030	-0.0080	-0.4615	-0.0024	0.0083	0.4271	-0.0026
UEXIPM(H)	-0.0027	-0.1735	-0.0029	0.0040	0.2641	-0.0028	-0.0043	-0.2596	-0.0027	0.0080	0.4262	-0.0023
UEXIPM(Y)	0.0018	0.1654	-0.0029	0.0053	0.4771	-0.0019	-0.0023	-0.1871	-0.0029	0.0066	0.4749	-0.0019
M0 (M)	2.1676	0.2363	-0.0028	8.3792	0.9669	-0.0006	-2.3739	-0.1601	-0.0029	2.8835	0.1802	-0.0029
M0 (Q)	1.9871	0.2699	-0.0024	7.5637	1.0771	0.0035	-6.3250	-0.5528	0.0012	-3.7506	-0.2964	-0.0020
M0 (H)	4.0699	0.4908	0.0022	3.0749	0.4715	-0.0008	-0.7905	-0.0699	-0.0029	-6.1814	-0.5965	0.0030
M0 (Y)	5.3711	0.8962	0.0198	5.2240	0.9526	0.0132	2.4880	0.3289	0.0004	0.8845	0.1133	-0.0027
UEXM0(Q)	0.0001	0.3763	-0.0013	0.0001	0.8722	0.0042	-0.0001	-0.2330	-0.0021	0.0000	0.0851	-0.0029
UEXM0 (H)	0.0000	0.3030	-0.0013	0.0001	0.5764	0.0014	-0.0001	-0.2692	-0.0014	0.0000	-0.1626	-0.0025
UEXM0 (Y)	0.0000	0.3866	0.0005	0.0000	0.4184	0.0001	0.0000	-0.0948	-0.0028	0.0000	-0.1553	-0.0025
UNE (M)	-1.0902	-0.3528	-0.0023	-3.4535	-1.0616	0.0025	-0.8637	-0.2664	-0.0027	-2.0448	-0.6272	-0.0018
UNE (Q)	-0.7905	-0.5785	-0.0006	-1.3979	-0.9018	0.0026	-0.7928	-0.5573	-0.0013	-1.0405	-0.6742	-0.0010
UNE (H)	-0.8094	-1.0486	0.0057	-1.0314	-1.1158	0.0076	-0.6733	-0.8468	0.0012	-0.7590	-0.8404	0.0008
UNE (Y)	-0.8883	-1.7683	0.0338	-0.8325	-1.4189	0.0213	-0.7790	-1.6106	0.0169	-0.6584	-1.2171	0.0070
UEXUNE(Q)	-0.0755	-0.6066	0.0002	-0.1166	-0.8770	0.0027	-0.0732	-0.5760	-0.0009	-0.0985	-0.7720	-0.0003
UEXUNE(H)	-0.0660	-0.8641	0.0037	-0.0792	-0.9372	0.0043	-0.0611	-0.7898	0.0010	-0.0690	-0.8726	0.0006
UEXUNE(Y)	-0.0661	-1.3980	0.0178	-0.0624	-1.1979	0.0110	-0.0585	-1.2868	0.0085	-0.0522	-1.0851	0.0034
CPI (M)	-0.2928	-0.6212	0.0000	-0.3810	-0.7559	0.0008	-0.6599	-1.0920	0.0078	-0.6649	-0.9164	0.0047
CPI (Q)	-0.1430	-0.4397	-0.0003	-0.2406	-0.6407	0.0027	-0.4316	-1.0437	0.0143	-0.4329	-0.8287	0.0093
CPI (H)	-0.0446	-0.2020	-0.0024	-0.0819	-0.3360	-0.0014	-0.2051	-0.8279	0.0067	-0.1598	-0.5132	0.0011
CPI (Y)	0.0963	0.7378	0.0040	0.1080	0.7349	0.0037	0.0612	0.4483	-0.0010	0.1508	0.8571	0.0056
UEXCPI(Q)	0.0335	0.8446	0.0052	0.0088	0.2363	-0.0026	0.0209	0.4888	-0.0008	0.0183	0.4135	-0.0018

 Table 5.8 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 6/18)

Continued												
UEXCPI(H)	0.0332	1.0345	0.0133	0.0116	0.3950	-0.0015	0.0234	0.6850	0.0027	0.0178	0.5240	-0.0007
UEXCPI(Y)	0.0243	1.2667	0.0185	0.0117	0.6384	0.0007	0.0201	1.0458	0.0073	0.0194	0.9739	0.0038
TBILL(M)	0.2474	0.9428	-0.0002	0.2807	1.0333	-0.0003	0.0930	0.3119	-0.0028	0.3676	1.1219	0.0001
TBILL (Q)	0.2390	1.3231	0.0061	0.3010	1.7004	0.0079	0.1340	0.6324	-0.0010	0.3262	1.3858	0.0054
TBILL (H)	0.2241	1.6261	0.0140	0.3359	2.3948	0.0259	0.1267	0.8100	0.0008	0.3481	2.1517	0.0174
TBILL (Y)	0.2099	2.6286	0.0289	0.2936	3.0767	0.0442	0.1974	2.1741	0.0169	0.3905	3.4647	0.0520
UEXTBILL(Q)	52.0882	1.6600	0.0108	60.8657	1.8534	0.0112	40.4386	1.2338	0.0028	77.7084	2.1340	0.0122
UEXTBILL(H)	52.2504	1.8029	0.0210	64.9372	2.2427	0.0249	39.4884	1.2854	0.0066	74.3968	2.4819	0.0212
UEXTBILL(Y)	41.2018	2.0426	0.0251	55.3388	2.3977	0.0353	32.7132	1.5619	0.0095	66.7363	2.9089	0.0337
TERM(M)	-3.1937	-0.8962	0.0003	-7.2605	-2.0172	0.0100	-0.7633	-0.2245	-0.0029	-6.6708	-1.6642	0.0042
TERM(Q)	-1.2198	-0.5039	-0.0017	-3.8735	-1.5872	0.0076	0.2310	0.0962	-0.0030	-3.9726	-1.3585	0.0043
TERM(H)	0.1294	0.0722	-0.0030	-2.1926	-1.2128	0.0041	0.8444	0.5199	-0.0021	-2.4619	-1.2931	0.0029
TERM(Y)	-0.3113	-0.2692	-0.0027	-1.5367	-1.1678	0.0042	-0.5926	-0.4519	-0.0020	-2.7147	-1.7675	0.0118
UEXTERM(Q)	-2.3206	-0.6559	-0.0004	-6.3601	-1.7875	0.0118	-0.1202	-0.0350	-0.0030	-6.3140	-1.5539	0.0066
UEXTERM(H)	-0.7954	-0.2745	-0.0025	-4.2973	-1.4993	0.0082	0.7320	0.2703	-0.0027	-4.3058	-1.3284	0.0044
UEXTERM(Y)	-0.3622	-0.1819	-0.0029	-3.1221	-1.4825	0.0081	-0.0841	-0.0465	-0.0030	-4.0403	-1.7687	0.0092
DY(M)	0.4905	1.1538	0.0007	0.2744	0.5672	-0.0022	0.3571	0.6702	-0.0016	0.3665	0.6152	-0.0020
DY(Q)	0.4552	1.2330	0.0072	0.3411	0.7212	0.0013	0.4518	0.9717	0.0040	0.5504	0.9641	0.0044
DY(H)	0.2473	0.8008	0.0040	-0.0115	-0.0303	-0.0030	0.1700	0.5185	-0.0007	0.0279	0.0680	-0.0030
DY(Y)	0.0569	0.2670	-0.0022	0.0153	0.0595	-0.0030	-0.0354	-0.1668	-0.0028	-0.0449	-0.1724	-0.0028
UEXDY(Q)	16.1666	1.6771	0.0091	9.6372	0.8293	0.0002	13.1006	1.1875	0.0026	12.8462	0.9808	0.0008
UEXDY(H)	12.0417	1.3284	0.0089	4.9018	0.4544	-0.0016	8.8502	0.9245	0.0015	7.0694	0.6307	-0.0010
UEXDY(Y)	5.3608	0.7369	0.0020	0.9040	0.1086	-0.0029	2.0577	0.2920	-0.0025	0.8229	0.1038	-0.0030
EMR(-1)/VMR(-1)	-0.1652	-1.2148	0.0263	0.0718	0.2653	-0.0015	-0.2089	-1.2156	0.0300	0.0090	0.0286	-0.0030
EMR(-2)/VMR(-2)	-0.0031	-0.0324	-0.0030	0.0809	0.5138	0.0025	-0.0183	-0.1810	-0.0024	0.0518	0.2745	-0.0015
EMR(-3)/VMR(-3)	0.0130	0.1980	-0.0021	0.0195	0.2119	-0.0022	0.0115	0.1738	-0.0025	-0.0008	-0.0069	-0.0030
EMR(-4)/VMR(-4)	-0.0076	-0.2148	-0.0023	-0.0024	-0.0432	-0.0030	-0.0187	-0.5187	0.0000	-0.0365	-0.5439	0.0012
EMR(-5)/VMR(-5)	-0.0076	-0.3319	-0.0014	0.0158	0.4366	-0.0005	-0.0170	-0.7105	0.0028	-0.0119	-0.2758	-0.0021
EMR(-6)/VMR(-6)	-0.0124	-0.7510	0.0053	0.0009	0.0369	-0.0030	-0.0161	-0.9243	0.0068	-0.0125	-0.4406	-0.0011
EMR(-7)/VMR(-7)	-0.0100	-0.7538	0.0059	0.0057	0.3093	-0.0020	-0.0154	-1.1329	0.0122	-0.0068	-0.3336	-0.0021
EMR(-8)/VMR(-8)	-0.0109	-0.9885	0.0140	-0.0059	-0.3722	-0.0011	-0.0134	-1.1512	0.0152	-0.0160	-0.9659	0.0065
EMR(-9)/VMR(-9)	-0.0008	-0.1357	-0.0029	0.0005	0.0484	-0.0030	-0.0010	-0.1473	-0.0029	-0.0063	-0.5446	-0.0004
EMR(-10)/VMR(-10)	-0.0020	-0.4711	-0.0010	0.0060	0.7080	0.0030	-0.0017	-0.3215	-0.0020	0.0034	0.3793	-0.0018

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with shares returns in past 6 months, and hold for 18 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return  $R_{wml}$ .

	Panel	l A Momentu	m coefficient y	γ <sub>mom</sub>				Pane	el B Winner/Lo	oser return R <sub>wml</sub>		
	Panel A	A1 Equal-weig	ghted	Panel A	2 Value-wei	ghted	Panel B	1 Equal-weig	ghted	Panel B	32 Value-wei	ghted
Macroeconomic variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	-0.0390	-2.0874	0.0842	-0.0246	-1.6750	0.0225	-0.0408	-2.3091	0.0600	-0.0329	-2.2642	0.0256
UEXGDP(Y)	-0.0530	-1.9797	0.0661	-0.0423	-1.9594	0.0294	-0.0575	-2.1785	0.0506	-0.0517	-2.5195	0.0273
IP(M)	-3.9776	-1.7578	0.0010	-2.2363	-0.8354	-0.0021	-2.8691	-0.9475	-0.0017	-0.0237	-0.0060	-0.0031
IP(Q)	-4.4698	-1.6155	0.0078	-4.3820	-1.7026	0.0046	-4.1351	-1.3711	0.0030	-3.1818	-0.8249	-0.0006
IP(H)	-2.3769	-1.1207	0.0036	-2.8390	-1.4893	0.0039	-2.9434	-1.1878	0.0037	-3.3682	-1.1926	0.0031
IP(Y)	-0.7444	-0.4768	-0.0015	-1.9468	-1.0366	0.0047	-1.2569	-0.6616	-0.0002	-2.5914	-1.3292	0.0056
UEXIP(Q)	-0.0311	-1.7649	0.0081	-0.0265	-1.4686	0.0029	-0.0275	-1.4064	0.0027	-0.0182	-0.7169	-0.0013
UEXIP(H)	-0.0246	-1.3772	0.0078	-0.0235	-1.4327	0.0042	-0.0254	-1.2553	0.0045	-0.0251	-1.0570	0.0021
UEXIP(Y)	-0.0113	-0.7867	0.0015	-0.0164	-1.1342	0.0040	-0.0164	-0.9496	0.0032	-0.0222	-1.2074	0.0050
IPM(M)	-3.4413	-1.3947	0.0002	-1.8872	-0.8336	-0.0024	-3.6828	-1.1723	-0.0006	-0.7933	-0.2583	-0.0030
IPM(Q)	-1.7243	-0.6273	-0.0012	-1.0760	-0.3906	-0.0026	-1.6669	-0.5327	-0.0019	-0.3714	-0.1029	-0.0031
IPM(H)	-0.8198	-0.4041	-0.0021	0.5848	0.2694	-0.0027	-1.4250	-0.5676	-0.0012	0.2220	0.0756	-0.0031
IPM(Y)	0.1579	0.1330	-0.0030	0.9860	0.6468	-0.0003	-0.1711	-0.1180	-0.0030	1.0413	0.6137	-0.0011
UEXIPM(Q)	-0.0178	-0.9909	0.0010	-0.0095	-0.5513	-0.0022	-0.0202	-1.0072	0.0004	-0.0047	-0.2047	-0.0030
UEXIPM(H)	-0.0107	-0.6131	-0.0007	-0.0002	-0.0140	-0.0031	-0.0143	-0.7060	-0.0003	-0.0018	-0.0781	-0.0031
UEXIPM(Y)	-0.0019	-0.1542	-0.0029	0.0066	0.4867	-0.0016	-0.0068	-0.4576	-0.0017	0.0038	0.2218	-0.0028
M0 (M)	-1.1688	-0.1060	-0.0030	5.1667	0.6083	-0.0023	-5.3841	-0.3101	-0.0023	1.1295	0.0684	-0.0031
M0 (Q)	-0.4839	-0.0569	-0.0031	6.4609	0.8975	0.0011	-9.0408	-0.7319	0.0043	-3.0476	-0.2321	-0.0025
M0 (H)	-1.3491	-0.1529	-0.0026	0.3093	0.0466	-0.0031	-8.0874	-0.6578	0.0092	-7.6106	-0.6690	0.0045
M0 (Y)	2.6666	0.4422	0.0020	4.2678	0.8182	0.0065	-1.5486	-0.1908	-0.0020	0.7583	0.0900	-0.0029
UEXM0(Q)	0.0000	0.1290	-0.0029	0.0001	0.8457	0.0033	-0.0001	-0.4367	-0.0002	0.0000	0.1002	-0.0030
UEXM0 (H)	0.0000	-0.0446	-0.0031	0.0001	0.5364	0.0004	-0.0001	-0.6315	0.0056	0.0000	-0.1715	-0.0026
UEXM0 (Y)	0.0000	0.0493	-0.0030	0.0000	0.4466	-0.0001	-0.0001	-0.4935	0.0034	0.0000	-0.1487	-0.0026
UNE (M)	0.6184	0.1857	-0.0029	-3.2355	-0.8892	0.0012	0.7904	0.2180	-0.0029	-1.6610	-0.4230	-0.0024
UNE (Q)	-0.0563	-0.0377	-0.0031	-1.4253	-0.8478	0.0021	-0.1245	-0.0760	-0.0031	-0.8031	-0.4343	-0.0021
UNE (H)	-0.4112	-0.4891	-0.0011	-1.0759	-1.1313	0.0072	-0.3162	-0.3450	-0.0023	-0.7246	-0.7180	-0.0002
UNE (Y)	-0.6620	-1.2495	0.0155	-0.8212	-1.4814	0.0180	-0.5865	-1.0984	0.0065	-0.6019	-1.1225	0.0040
UEXUNE(Q)	-0.0171	-0.1286	-0.0030	-0.1313	-0.9397	0.0034	-0.0212	-0.1485	-0.0030	-0.0995	-0.6812	-0.0008
UEXUNE(H)	-0.0309	-0.3785	-0.0018	-0.0923	-1.0652	0.0057	-0.0282	-0.3261	-0.0024	-0.0749	-0.8529	0.0005
UEXUNE(Y)	-0.0467	-0.9490	0.0064	-0.0690	-1.3406	0.0122	-0.0408	-0.8027	0.0017	-0.0552	-1.1272	0.0030
CPI (M)	-0.3097	-0.6299	0.0000	-0.1752	-0.3155	-0.0024	-0.7433	-1.0739	0.0086	-0.4083	-0.4975	-0.0006
CPI (Q)	-0.1538	-0.4409	-0.0003	-0.1704	-0.4059	-0.0005	-0.5092	-1.1107	0.0175	-0.3718	-0.6138	0.0046
CPI (H)	-0.0579	-0.2469	-0.0021	-0.0327	-0.1178	-0.0029	-0.2725	-1.0059	0.0114	-0.1327	-0.3548	-0.0007
CPI (Y)	0.1237	0.8761	0.0073	0.1470	0.8452	0.0077	0.0810	0.5237	-0.0002	0.1829	0.9031	0.0073
UEXCPI(Q)	0.0208	0.4502	-0.0002	0.0036	0.0828	-0.0030	0.0047	0.0940	-0.0030	0.0165	0.3068	-0.0023

Table 5.9 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 6/24)

Continued												
UEXCPI(H)	0.0219	0.6181	0.0033	0.0080	0.2485	-0.0025	0.0084	0.2194	-0.0025	0.0124	0.3032	-0.0022
UEXCPI(Y)	0.0201	0.9771	0.0102	0.0144	0.7020	0.0019	0.0148	0.6998	0.0017	0.0216	0.9550	0.0040
TBILL(M)	0.2007	0.7155	-0.0014	0.2097	0.7157	-0.0017	0.0195	0.0598	-0.0031	0.2443	0.6973	-0.0019
TBILL (Q)	0.2825	1.3839	0.0085	0.4044	2.1039	0.0145	0.1908	0.8115	0.0004	0.4369	1.7531	0.0097
TBILL (H)	0.2233	1.5460	0.0122	0.4211	2.7426	0.0369	0.1207	0.6966	-0.0002	0.4162	2.4039	0.0214
TBILL (Y)	0.2232	2.4248	0.0295	0.3346	3.2137	0.0509	0.2208	2.1252	0.0180	0.4172	3.6433	0.0494
UEXTBILL(Q)	52.5877	1.4427	0.0097	59.6912	1.6333	0.0091	41.8087	1.1438	0.0022	78.8869	1.9759	0.0102
UEXTBILL(H)	48.1948	1.5200	0.0155	70.9780	2.2872	0.0266	35.8831	1.0863	0.0037	81.3986	2.5989	0.0214
UEXTBILL(Y)	37.4806	1.6917	0.0181	64.6691	2.6504	0.0434	28.0427	1.2131	0.0047	76.1583	3.3676	0.0372
TERM(M)	-1.1307	-0.2809	-0.0027	-5.4313	-1.6327	0.0034	0.3249	0.0782	-0.0031	-5.3066	-1.3165	0.0008
TERM(Q)	-0.3389	-0.1193	-0.0030	-3.2684	-1.3351	0.0036	0.7113	0.2423	-0.0028	-3.1832	-1.0336	0.0009
TERM(H)	1.1288	0.5268	-0.0008	-2.6555	-1.4310	0.0062	1.5589	0.8510	-0.0002	-2.9891	-1.5611	0.0043
TERM(Y)	0.3744	0.2814	-0.0026	-1.6503	-1.1229	0.0043	-0.1035	-0.0698	-0.0031	-3.1253	-2.0118	0.0136
UEXTERM(Q)	-0.5822	-0.1436	-0.0030	-4.8302	-1.4264	0.0045	0.9347	0.2263	-0.0029	-4.6115	-1.0985	0.0012
UEXTERM(H)	0.6135	0.1799	-0.0028	-4.3742	-1.5785	0.0073	1.5930	0.4998	-0.0019	-4.3669	-1.3183	0.0034
UEXTERM(Y)	1.0023	0.4386	-0.0017	-3.3501	-1.5508	0.0083	1.0500	0.5257	-0.0021	-4.4891	-2.0147	0.0097
DY(M)	0.4521	1.0439	-0.0002	0.4326	0.8075	-0.0011	0.2694	0.5105	-0.0024	0.5366	0.8982	-0.0012
DY(Q)	0.4220	1.1065	0.0049	0.5311	1.0465	0.0062	0.4520	0.9849	0.0029	0.7580	1.3542	0.0088
DY(H)	0.2232	0.6655	0.0021	0.2194	0.5304	0.0006	0.1871	0.5178	-0.0007	0.2751	0.6545	0.0006
DY(Y)	0.1094	0.4810	-0.0001	0.1841	0.6701	0.0031	0.0513	0.2223	-0.0027	0.1481	0.5359	-0.0006
UEXDY(Q)	14.5590	1.3766	0.0059	13.5708	0.9804	0.0026	12.1295	1.0658	0.0010	15.9420	1.0151	0.0019
UEXDY(H)	10.5799	1.0429	0.0053	11.4934	0.9230	0.0042	8.6804	0.8372	0.0006	12.7384	1.0412	0.0025
UEXDY(Y)	5.4731	0.6741	0.0017	7.5912	0.8186	0.0037	3.6137	0.4592	-0.0017	7.4144	0.9052	0.0009
EMR(-1)/VMR(-1)	-0.1742	-1.1518	0.0267	0.0185	0.0624	-0.0030	-0.2211	-1.0853	0.0285	-0.0667	-0.1885	-0.0024
EMR(-2)/VMR(-2)	-0.0188	-0.1799	-0.0022	0.0151	0.0827	-0.0029	-0.0357	-0.3219	-0.0010	-0.0316	-0.1442	-0.0026
EMR(-3)/VMR(-3)	-0.0035	-0.0512	-0.0030	-0.0352	-0.3381	-0.0007	-0.0045	-0.0638	-0.0030	-0.0601	-0.4683	0.0013
EMR(-4)/VMR(-4)	-0.0097	-0.2689	-0.0020	-0.0350	-0.5614	0.0022	-0.0230	-0.6064	0.0009	-0.0739	-0.9486	0.0116
EMR(-5)/VMR(-5)	-0.0097	-0.4125	-0.0007	-0.0211	-0.5153	0.0008	-0.0194	-0.7704	0.0034	-0.0486	-0.9772	0.0100
EMR(-6)/VMR(-6)	-0.0107	-0.6074	0.0024	-0.0172	-0.5776	0.0017	-0.0143	-0.7388	0.0034	-0.0279	-0.7927	0.0049
EMR(-7)/VMR(-7)	-0.0108	-0.7669	0.0064	-0.0116	-0.5231	0.0007	-0.0176	-1.1332	0.0135	-0.0241	-0.9900	0.0071
EMR(-8)/VMR(-8)	-0.0129	-1.1277	0.0185	-0.0133	-0.7403	0.0056	-0.0154	-1.2784	0.0173	-0.0254	-1.2843	0.0169
EMR(-9)/VMR(-9)	-0.0013	-0.1922	-0.0027	-0.0056	-0.4270	-0.0004	-0.0004	-0.0593	-0.0031	-0.0130	-0.9201	0.0062
EMR(-10)/VMR(-10)	-0.0028	-0.5721	0.0005	0.0027	0.2574	-0.0020	-0.0017	-0.2923	-0.0022	-0.0004	-0.0379	-0.0031

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with shares returns in past 6 months, and hold for 24 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return  $R_{wml}$ .

		,				0	,	
		Panel A	3 months' l	holding lengt	th			
	N	Iometnum co	efficient $\gamma_{mo}$	om	V	Vinner/Lose	er return R <sub>wm</sub>	l
	Equal-v	weighted	Value-	weighted	Equal-w	eighted	Value-w	eighted
Lag length		4		3	2		3	
Adj-R <sup>2</sup>	0.0	854	0.0	0585	0.06	558	0.08	340
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
UEXTBILL(Y)	9.9444	1.5959	16.4070	1.9604	9.0180	1.1901	21.6687	2.0963
DY(H)	0.2387	2.0957	0.2668	1.9340	0.2947	2.0277	0.3059	1.7946
CPI(O)			-0.2441	-2.6133			-0.4384	-2.9670
		Panel B	6 months' l	holding lengt	h			
	M	ometnum coe	efficient v		W	inner/Loser	return R	
	Faual-we	eighted	Value-we	n sighted	Faual-we	ighted	Value-we	eighted
Lag length	Equal W	erginea		ighted	Equal we	igitted		nginea
$A di_{\rm P} P^2$	0.07	12	0.074	55	0.041	1	۳ ۵۰۰۵	03
Macroaconomic variables	1	12 t())	0.07. 1	+())	2	τ +()	3	+())
	14 8005	1 2 2 8 0	۸ 22 0212	$(\lambda)$	12 8407	1.0782	∧ 20.5227	2 2604
DV(I)	0.2254	1.3369	0.0792	2.1730	0.2055	1.0762	0 2016	2.3094
$DI(\mathbf{H})$	0.5254	1.7385	0.2782	1.2402	0.2933	1.3031	0.5010	2.9101
CPI(Q)			-0.4047	-2.4559			-0.7152	-2.8191
		Den 1 C	10	1	4			
		Panel C	12 months	notaing leng	,tn	· /T		
	M.	ometnum coe	efficient $\gamma_{mor}$	n • • •	w E	inner/Loser	return $R_{wml}$	• • • •
	Equal-we	eighted	Value-we	eighted	Equal-we	ighted	Value-we	eighted
Lag length	1		5		6	_	5	
Adj-R <sup>2</sup>	0.10	05	0.058	80	0.070	19	0.03	79
Macroeconomic variables	λ	t(λ)	λ	$t(\lambda)$	λ	t(λ)	λ	$t(\lambda)$
GDP(Y)	-0.0255	-2.4409	-0.0244	-2.3510	-0.0279	-2.5993	-0.0243	-2.3235
UEXTBILL(Y)	34.8205	2.9525	37.8669	2.2167	32.2650	2.2203	39.5416	2.0890
		Panel D	18 months'	holding leng	gth			
	M	ometnum coe	efficient $\gamma_{mor}$	n	W	inner/Loser	return R <sub>wml</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-we	ighted	Value-we	eighted
Lag length	8		5		6		5	
Adj-R <sup>2</sup>	0.12	86	0.080	67	0.097	'1	0.07	31
Information ariables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP(Y)	-0.0407	-2.3295	-0.0338	-2.8332	-0.0452	-2.5612	-0.0368	-3.1601
UEXTBILL(Y)	39.5986	2.5098	54.0065	2.6518	30.9327	1.8984	65.2874	3.1746
		Panel E	24 months'	holding leng	gth			
	M	ometnum coe	efficient v		W	inner/Loser	return R <sub>uml</sub>	
	Equal-we	eighted	Value-we	righted	Equal-we	ighted	Value-we	eighted
Lag length	9	0	7	0	6	0	5	0
$Adi-R^2$	0 10	08	0.064	46	0.064	0	0.06	16
Macroeconomic variables	Λ	$t(\lambda)$	λ	 t(λ)	λ.	t(λ)	λ.	t(λ)
GDP(Y)	-0 0386	_2 2000	-0 0230	-2 0087	-0.0405	-2 4004	-0.0321	-2 6236
UFXTRILL(V)	35 8703	1 8183	63 6713	2.0007	26 3515	1 2667	-0.0321 74 8187	2.0250
Veter This table means the	33.0703	1.0103	03.0/13	2.0301	20.3313	1.2007	/4.010/	5.51//
Note: This table presents	slope coeff	icient $\lambda$ , t-	stats, and a	aujusted R	of logged	anate reg	ression of 1	nomentum
Ianuary 1980 to December	$r 2008 M_{\odot}$	mentum no	rtfolios are	formed wi	th share ret	urns in na	st 6 months	and hold
		memorin po	uc			wanto ni pa	se o monun	,

# Table 5.10 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables (Momentum-6 months' formation length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return  $R_{wml}$  on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 6 months, and hold for 3, 6, 12, 18, and 24 months, respectively. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(H) is the change of DY in past 6 months. CPI(Q) is the change of CPI in past 3 months. GDP(Y) is the change of GDP in past 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to E report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  with 3, 6, 12, 18, and 24 months' holding lengths, respectively.

	iluci occon	Denal A		December 1		uiii ( <i>i</i> ,c)			
	M	Panel A	April 1980-1	December I	989	7° /T	( D		
	Mometnum coefficient $\gamma_{mom}$ Equal-weighted Value-weighted					vinner/Lose	r return $R_{wml}$	• • • •	
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-w	eighted	
Lag length	5		1		4		1		
Adj-R <sup>2</sup>	0.24	90	0.13	37	0.20	90	0.11	74	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	13.5924	2.5516	21.9173	3.0679	18.8571	3.0049	28.9324	3.3980	
DY(H)	0.1858	1.5681	0.1071	0.8095	0.1473	1.2512	0.0098	0.0698	
CPI(Q)			-0.1153	-1.2365			-0.1227	-1.2772	
		Panel B	January 1990	-December	1999				
	M	ometnum co	befficient $\gamma_{mon}$	n	v	Vinner/Lose	r return R <sub>wml</sub>		
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	2		2		2		2		
Adj-R <sup>2</sup>	0.14	64	0.03	20	0.12	24	0.05	46	
Macroeconomic variables	λ	$t(\lambda)$ $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$					λ	t(λ)	
UEXTBILL(Y)	12.8746	1.3057	-3.0066	-0.1789	7.3256	0.7209	-12.3422	-0.6929	
DY(H)	0.3974	3.6832	0.3997	1.9262	0.4043	3.2425	0.4901	2.2533	
CPI(O)			0.0602	0.4780			0.1155	0.7302	
		Panel C	January 2000	-December	2008				
	M	ometnum co	befficient $\gamma_{mon}$	n	V	Vinner/Lose	r return R <sub>wml</sub>		
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-w	eighted	
Lag length	2		2		2		3		
Adj-R <sup>2</sup>	0.02	01	0.12	98	0.01	00	0.26	09	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	-52,4636	-1 6794	-27 6507	-0 6457	-44 7272	-1 0466	-30 3292	-0.6602	
DY(H)	-0.0771	-0.3256	-0.0597	-0.2313	0.1160	0.3084	-0.0974	-0.2886	
CPI(O)	0.0771	0.5250	-0 5903	-3 8902	0.1100	0.5004	-1 0892	-4 9547	
			-0.5705	-5.6702			-1.0072	-4.7547	
		Panel D	) April 1980-I	December 20	008				
	M	ometnum co	befficient $\gamma_{mon}$	n	V	Vinner/Lose	r return R <sub>wml</sub>		
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	4		3		2		3		
Adj-R <sup>2</sup>	0.08	54	0.05	85	0.06	58	0.08	40	
Macroeconomic variables	es $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ ) $\lambda$				t(λ)	λ	t(λ)		
UEXTBILL(Y)	9,9444	1.5959	16,4070	1.9604	9.0180	1,1901	21.6687	2.0963	
DY(H)	0.2387	2.0957	0 2668	1 9340	0 2947	2.0277	0 3059	1 7946	
CPI(O)	0.2307	2.0757	-0 2441	-2 6133	0.2747	2.0211	-0 4384	-2.9670	
	-	<u></u>	0.2771	2.0155	2 2 1 1			2.7010	

# Table 5.11 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 6/3)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return  $R_{wml}$  on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 6 months, and hold for 3 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. UNTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(H) is the change of DY in past 6 months. CPI(Q) is the change of CPI in past 3 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

		Panel A	July 1980-E	December 19	989			
	M	fficient $\gamma_{mom}$			Winner/Los	er return R <sub>wml</sub>		
	Equal-w	eighted	Value-w	veighted	Equal-w	eighted	Value-we	eighted
Lag length	. 6	C	2	2	. 6	i	5	U
Adj-R <sup>2</sup>	0.27	41	0.18	813	0.21	196	0.16	75
Macroeconomic variables	λ	$t(\lambda)$	λ	t(λ)	λ	t(λ)	λ	t(λ)
UEXTBILL(Y)	27.1195	5.5584	40.5715	4.3627	35.1553	4.2596	48.4745	4.7192
DY(H)	0.1550	1.2367	-0.0549	-0.2961	0.0594	0.4072	-0.1526	-0.7225
CPI(Q)			-0.0611	-0.3722			0.0252	0.1389
		Panel B. J	anuary 1990	-December	1999			
	M	ometnum coe	fficient $\gamma_{mom}$	Determoti		Winner/Los	er return R	
	Equal-w	eighted	Value-w	veighted	Equal-w	eighted	Value-we	eighted
Lag length	4	0	3	3	4		3	U
Adj-R <sup>2</sup>	0.17	63	0.07	743	0.11	02	0.06	99
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
UEXTBILL(Y)	15.9571	0.9218	-4.6560	-0.1713	4.1309	0.2718	-12.6711	-0.4990
DY(H)	0.6853	3.8040	0.7976	2.2975	0.5543	3.4510	0.8187	2.6812
CPI(Q)			0.0924	0.3262			0.0395	0.1242
			2000	<b>D</b>	2000			
		Panel C J	anuary 2000	-December	2008	XX7' /T		
	M	ometnum coe	fficient $\gamma_{mom}$	• 1 / 1	E 1	Winner/Los	er return $R_{wml}$	. 1 . 1
T 1 4	Equal-we	Ignted	value-we	righted	Equal-v	vergnted	value-w	reighted
Lag length $A^{1}$ : $D^{2}$	3	2	6	20		) 272	0.20	2
Adj-K Maanaan andre andre andre ale	0.082	())	0.320	19	0.0	3/3	0.39	49
Macroeconomic variables	٨	$t(\lambda)$	λ	t(λ)	٨	τ(λ)	٨	$t(\lambda)$
UEATBILL(Y)	-124.2990	-2.4852	-29.9155	-0.5707	-123.1133	-1.8072	-19.6466	-0.3227
DY(H)	-0.3220	-0.8691	-0.4186	-1.0734	-0.2423	-0.4800	-0.4405	-1.0651
CPI(Q)			-1.1278	-3.7224			-1./136	-4.7578
		Panel D	July 1980-D	December 20	008			
	Mometnum coefficient $\gamma_{mom}$ Winner/Loser return $R_{wml}$							
	Equal-weighted Value-weighted Equal-weighted Value-weighted						eighted	
Lag length	8 4 6 4							
Adj-R <sup>2</sup>	0.0712 0.0755				0.04	411	0.09	93
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
UEXTBILL(Y)	14.8005	1.3389	32.0312	2.1736	13.8497	1.0782	39.5337	2.3694
DY(H)	0.3254	1.7585	0.2782	1.2462	0.2955	1.3631	0.3016	1.2251
CPI(Q)			-0.4647	-2.4359			-0.7132	-2.8191

# Table 5.12 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 6/6)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return R<sub>wml</sub> on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 6 months, and hold for 6 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(H) is the change of DY in past 6 months. CPI(Q) is the change of CPI in past 3 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

						/			
		Panel A Ja	anuary 1981-	December 1	989				
	Mometnum coefficient $\gamma_{mom}$ Winner/Loser return $R_{wml}$								
	Equal-we	ighted	Value-we	eighted	Equal-we	eighted	Value-w	eighted	
Lag length	4		4		4		2		
Adj-R <sup>2</sup>	0.230	00	0.17	90	0.20	29	0.18	87	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0108	-1.4825	-0.0063	-0.8620	-0.0170	-1.8707	-0.0077	-1.0586	
UEXTBILL(Y)	46.8892	6.4782	50.4248	3.0999	61.6909	5.0278	68.8114	3.8234	
		Panel B Ja	anuary 1990-	December 1	999				
	M	ometnum co	efficient $\gamma_{mo}$	m	v	Vinner/Lose	er return R <sub>wml</sub>		
	Equal-w	eighted	Value-w	eighted	Equal-w	eighted	Value-weighted		
Lag length	5	5 3 4							
Adj-R <sup>2</sup>	0.24	-25	0.17	22	0.19	37	-46		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0511	-3.1705	-0.0606	-3.2786	-0.0448	-3.6165	-0.0558	-3.2609	
UEXTBILL(Y)	27.8153	1.7224	27.2094	1.3074	5.4260	0.3871	13.8631	0.6324	
		Panel C Ja	anuary 2000-	December 2	008				
	M	ometnum co	efficient $\gamma_{mo}$	m	v	Vinner/Lose	r return R <sub>wml</sub>		
	Equal-w	eighted	Value-w	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	7		7		5		5		
Adj-R <sup>2</sup>	0.01	80	-0.00	)57	0.00	076	-0.00	)42	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0365	-0.8124	-0.0203	-0.3322	-0.0296	-0.5847	-0.0213	-0.2937	
UEXTBILL(Y)	-33.8342	-0.7713	-43.7111	-0.7055	-61.7565	-0.9530	-77.7833	-0.8237	
		Panel D Ja	nuary 1981-	December 2	008				
	M	ometnum co	efficient $\gamma_{mo}$	m	V	Vinner/Lose	r return R <sub>wml</sub>		
	Equal-w	eighted	Value-w	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	7 5 6 5								
Adj-R <sup>2</sup>	0.10	05	0.05	580	0.07	09	0.03	79	
Macroeconomic variables	λ	t(λ)	$(\lambda)$ $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$				t(λ)		
GDP(Y)	-0.0255 -2.4409 -0.0244 -2.3510 -0.0279 -2.5				-2.5993	-0.0243	-2.3235		

## Table 5.13 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 6/12)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return R<sub>wml</sub> on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 6 months, and hold for 12 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

		Panel A	July 1981-D	ecember 19	89	/		
	$\frac{1}{10000000000000000000000000000000000$						r return R <sub>wml</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-w	eighted	Value-w	eighted
Lag length	4		4		4		5	
Adj-R <sup>2</sup>	0.18	65	0.16	05	0.19	50	0.20	72
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP(Y)	-0.0309	-1.8998	-0.0266	-2.3587	-0.0408	-2.2343	-0.0342	-2.9800
UEXTBILL(Y)	30.5151	2.4733	37.4925	1.5110	39.2556	2.7926	67.1931	2.5553
		Panel B Ja	unuary 1990-	December 1	999			
	Mo	ometnum co	efficient $\gamma_{mor}$	n	V	Vinner/Lose	r return R <sub>wml</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-w	eighted	Value-w	eighted
Lag length	5		3		5		2	
Adj-R <sup>2</sup>	0.21	10	0.15	39	0.15	47	0.11	89
Macroeconomic variables	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	t(λ)
GDP(Y)	-0.0543	-2.6439	-0.0473	-2.0770	-0.0468	-2.7152	-0.0403	-1.9188
UEXTBILL(Y)	47.8457	2.2001	76.5314	2.9183	25.4993	1.2879	68.0717	2.6854
		Panel C Ja	nuary 2000-	December 2	008			
	M	ometnum co	efficient v	n	v	Vinner/Lose	r return R	
	Equal-we	eighted	Value-we	" eighted	Equal-w	eighted	Value-w	eighted
Lag length	7	0	7	0	5	0	5	0
$Adj-R^2$	0.00	40	-0.01	49	-0.00	)21	-0.01	178
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP(Y)	-0.0484	-0.8237	0.0051	0.0683	-0.0346	-0.5292	0.0023	0.0248
UEXTBILL(Y)	11.9980	0.2012	-46.3487	-0.7868	-45.7176	-0.5286	-35.2509	-0.3516
		Panel D	July 1981-D	ecember 20	08			
	Mometnum coefficient $\gamma_{mom}$ Winner/Loser return $R_{wml}$							
	Equal-weighted Value-weighted Equal-weighted Value-weig						eighted	
Lag length	8 5 6						5	
Adj-R <sup>2</sup>	0.12	86	0.08	67	0.09	71	0.07	31
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP(Y)	-0.0407	-2.3295	-0.0338	-2.8332	-0.0452	-2.5612	-0.0368	-3.1601
UEXTBILL(Y)	39.5986	2.5098	54.0065	2.6518	30.9327	1.8984	65.2874	3.1746

# Table 5.14 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 6/18)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return  $R_{wml}$  on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 6 months, and hold for 18 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wnl}$  in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

Panel A. January 1982-December 1989											
	M	return R <sub>uml</sub>									
	Equal-we	eighted	Value-w	eighted	Equal-w	eighted	Value-w	eighted			
Lag length	4	0	4	0	4	0	4				
$Adi-R^2$	0.13	36	0.04	48	0.13	03	0 1214				
Macroeconomic variables	$\lambda $ t( $\lambda$ )		λ	t(λ)	λ	t(λ)	λ	t(λ)			
GDP(Y)	-0.0292	-2 1906	-0.0168	-1 5126	-0.0356	-2 2514	-0.0302	-2 1432			
UEXTBILL(Y)	19 9703	1 2933	30 4610	1 120	33 3996	1 7499	69 6338	2.1432			
	17.7705	1.2755	30.1010	1.1270	00.0000	10.00	0,10000 211002				
	Panel B January 1990-December 1999										
	Mo	ometnum co	befficient $\gamma_{mon}$	m	v						
	Equal-we	eighted	Value-w	eighted	Equal-w	eighted	Value-weighted				
Lag length	7		3		6		2				
Adj-R <sup>2</sup>	0.20	52	0.18	310	0.16	03	0.1463				
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)			
GDP(Y)	-0.0518	-2.3500	-0.0271	-1.1781	-0.0478	-2.7329	-0.0275	-1.4322			
UEXTBILL(Y)	44.1549	1.7840	104.8900	3.8580	10.0216	0.4798	83.2529	3.4110			
Panel C January 2000-December 2008											
	Mometnum coefficient $\gamma_{mom}$ Winner/Loser return $R_{wml}$										
	Equal-we	eighted	Value-w	eighted	Equal-w	eighted	value-weighted				
Lag length	6	~ .	5		4		4				
Adj-R <sup>2</sup>	0.01	84	-0.00	084	-0.01	109	-0.0136				
Macroeconomic variables	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	t(λ)			
GDP(Y)	-0.0714	-1.2925	-0.0480	-0.7302	-0.0490	-0.7137	-0.0490	-0.5599			
UEXTBILL(Y)	113.0103	1.5206	62.4231	0.8367	75.6309	0.6603	79.6372	0.5999			
		Panel D Ia	anuary 1982-	December 2	008						
	M	metnum co	$efficient \gamma$		<u>v</u>	Vinner/Lose	r return R				
	Equal-we	eighted	Value-w	<sup>n</sup> eighted	Equal-w	eighted	Value-weighted				
Lag length	9	-8	7	8	6	8	5				
Adj-R <sup>2</sup>	0.10	08	0.0646		0.0640		0.0616				
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)			
GDP(Y)	-0.0386	-2.2909	-0.0239	-2.0087	-0.0405	-2.4004	-0.0321	-2.6236			
UEXTBILL(Y)	35.8703	1.8183	63.6713 2.6381		26.3515 1.2667		74.8187	3.3177			

## Table 5.15 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 6/24)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return  $R_{wml}$  on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 6 months, and hold for 24 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wnl}$  in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

Panel A Momentum coefficient $\gamma_{mom}$							Panel B Winner/Loser return R <sub>wml</sub>						
	Panel A	A1 Equal-wei	ghted	Panel A	2 Value-wei	ghted	Panel B1 Equal-weighted Panel B2 Value-weighted						
Macroeconomic variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	-0.0048	-1.8056	0.0241	-0.0072	-2.0154	0.0182	-0.0076	-2.0631	0.0189	-0.0145	-2.6409	0.0271	
UEXGDP(Y)	-0.0077	-1.8522	0.0264	-0.0118	-2.0768	0.0210	-0.0114	-1.9801	0.0184	-0.0222	-2.5299	0.0270	
IP(M)	-0.6190	-1.1845	-0.0008	-0.7163	-0.6190	-0.0019	0.1571	0.1496	-0.0029	0.2055	0.1104	-0.0029	
IP(Q)	-0.4661	-0.7862	-0.0004	0.0754	0.0710	-0.0029	0.2765	0.2696	-0.0026	1.4903	0.9094	0.0002	
IP(H)	-0.6155	-1.2978	0.0064	-0.8244	-1.1421	0.0029	-0.5704	-0.7451	-0.0003	-0.8668	-0.7611	-0.0007	
IP(Y)	-0.1405	-0.3950	-0.0018	-0.4805	-1.0700	0.0017	-0.0600	-0.1201	-0.0028	-0.8223	-1.1252	0.0018	
UEXIP(Q)	-0.0033	-0.8455	-0.0002	-0.0014	-0.1847	-0.0027	0.0026	0.3669	-0.0024	0.0072	0.6043	-0.0013	
UEXIP(H)	-0.0041	-1.1199	0.0034	-0.0040	-0.6086	-0.0008	-0.0011	-0.1774	-0.0028	-0.0002	-0.0231	-0.0029	
UEXIP(Y)	-0.0030	-0.9616	0.0037	-0.0046	-0.9929	0.0027	-0.0020	-0.4399	-0.0019	-0.0057	-0.7873	0.0001	
IPM(M)	-1.0081	-2.0838	0.0030	-0.9470	-0.9048	-0.0011	-0.7838	-0.7865	-0.0017	-0.6845	-0.3902	-0.0026	
IPM(Q)	-0.4731	-0.8291	0.0000	0.1753	0.1913	-0.0028	0.1400	0.1494	-0.0028	1.1361	0.7940	-0.0008	
IPM(H)	-0.3092	-0.6525	-0.0001	-0.4856	-0.7516	-0.0005	-0.0862	-0.1247	-0.0028	-0.5000	-0.5271	-0.0020	
IPM(Y)	0.0036	0.0117	-0.0029	-0.1428	-0.4177	-0.0023	0.0780	0.1902	-0.0028	-0.2447	-0.4475	-0.0023	
UEXIPM(Q)	-0.0046	-1.2667	0.0029	-0.0016	-0.2375	-0.0027	-0.0009	-0.1331	-0.0028	0.0037	0.3472	-0.0025	
UEXIPM(H)	-0.0031	-0.8853	0.0015	-0.0026	-0.4481	-0.0019	-0.0004	-0.0694	-0.0029	-0.0003	-0.0290	-0.0029	
UEXIPM(Y)	-0.0014	-0.5089	-0.0009	-0.0018	-0.4952	-0.0018	-0.0004	-0.1046	-0.0029	-0.0025	-0.4395	-0.0022	
M0 (M)	-1.6367	-0.6405	-0.0006	-1.4985	-0.3563	-0.0022	-0.4962	-0.0945	-0.0028	1.0041	0.1216	-0.0028	
M0 (Q)	-1.4332	-0.9870	0.0029	-1.1349	-0.4429	-0.0017	-4.4281	-1.3726	0.0153	-4.2061	-0.8164	0.0032	
M0 (H)	0.5889	0.4852	-0.0009	1.2549	0.6616	0.0003	0.4904	0.1941	-0.0024	1.3751	0.3741	-0.0015	
M0 (Y)	0.6421	0.7073	0.0031	0.3036	0.2669	-0.0024	0.4634	0.2793	-0.0019	-0.3071	-0.1372	-0.0027	
UEXM0(Q)	0.0000	-0.5393	0.0006	0.0000	0.1961	-0.0025	0.0000	-0.7008	0.0024	0.0000	0.0570	-0.0029	
UEXM0 (H)	0.0000	-0.1864	-0.0024	0.0000	0.5692	0.0004	0.0000	-0.3870	-0.0011	0.0000	0.2828	-0.0019	
UEXM0 (Y)	0.0000	0.1543	-0.0026	0.0000	0.8280	0.0034	0.0000	-0.0289	-0.0029	0.0000	0.5157	-0.0001	
UNE (M)	-0.2820	-0.4093	-0.0020	-0.2496	-0.2230	-0.0027	-0.7995	-0.7592	-0.0004	-0.4655	-0.3071	-0.0026	
UNE (Q)	-0.0533	-0.1679	-0.0027	-0.0834	-0.1887	-0.0027	-0.1963	-0.4422	-0.0020	-0.1725	-0.2624	-0.0026	
UNE (H)	-0.0183	-0.1059	-0.0028	0.0383	0.1815	-0.0028	-0.0605	-0.2595	-0.0026	0.0780	0.2262	-0.0027	
UNE (Y)	-0.1033	-1.3012	0.0063	-0.1031	-1.0342	0.0003	-0.1577	-1.4674	0.0041	-0.1381	-0.8196	-0.0009	
UEXUNE(Q)	-0.0112	-0.4180	-0.0016	-0.0092	-0.2741	-0.0026	-0.0265	-0.7015	-0.0005	-0.0089	-0.1809	-0.0028	
UEXUNE(H)	-0.0055	-0.3381	-0.0021	-0.0010	-0.0517	-0.0029	-0.0123	-0.5531	-0.0015	0.0017	0.0574	-0.0029	
UEXUNE(Y)	-0.0063	-0.7400	0.0006	-0.0044	-0.4515	-0.0023	-0.0103	-0.8999	0.0002	-0.0048	-0.3098	-0.0027	
CPI (M)	-0.0320	-0.3683	-0.0022	-0.1696	-1.2901	0.0037	-0.0770	-0.5136	-0.0016	-0.3942	-1.5889	0.0097	
CPI (Q)	-0.0058	-0.1024	-0.0028	-0.0780	-0.9384	0.0023	-0.0460	-0.4511	-0.0012	-0.2291	-1.3906	0.0129	
CPI (H)	-0.0311	-0.6527	0.0030	-0.0902	-1.3459	0.0142	-0.0769	-1.1326	0.0089	-0.2059	-1.7298	0.0286	
CPI (Y)	-0.0249	-0.7783	0.0058	-0.0659	-1.5182	0.0183	-0.0449	-0.9991	0.0064	-0.1430	-2.0239	0.0324	
UEXCPI(Q)	0.0062	0.8706	0.0024	0.0010	0.1157	-0.0029	0.0071	0.7100	-0.0006	-0.0020	-0.1516	-0.0028	

 Table 5.16 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/3)

Continued												
UEXCPI(H)	0.0047	0.8398	0.0031	-0.0004	-0.0625	-0.0029	0.0050	0.6884	-0.0006	-0.0046	-0.4301	-0.0022
UEXCPI(Y)	0.0034	1.0017	0.0051	-0.0012	-0.2919	-0.0026	0.0043	0.9569	0.0012	-0.0047	-0.6802	-0.0011
TBILL(M)	0.0910	1.4969	0.0043	0.1096	1.0815	0.0007	0.1637	1.6221	0.0047	0.1385	0.7969	-0.0009
TBILL (Q)	0.0693	1.6776	0.0115	0.0518	0.8527	-0.0001	0.1398	2.0446	0.0163	0.0506	0.4607	-0.0020
TBILL (H)	0.0540	1.8474	0.0157	0.0336	0.8432	-0.0004	0.0935	2.0951	0.0154	0.0366	0.5148	-0.0019
TBILL (Y)	0.0437	2.1822	0.0233	0.0436	1.5283	0.0061	0.0718	2.1752	0.0203	0.0593	1.0884	0.0030
UEXTBILL(Q)	15.1459	2.4388	0.0187	15.9840	1.8392	0.0054	29.7632	2.9585	0.0245	21.6404	1.4501	0.0025
UEXTBILL(H)	14.3029	2.7805	0.0305	12.6698	1.7849	0.0062	26.9656	3.3059	0.0360	18.1020	1.4707	0.0036
UEXTBILL(Y)	11.9311	3.3813	0.0410	9.9480	1.9661	0.0077	20.2282	3.5079	0.0384	13.4411	1.4806	0.0039
TERM(M)	-1.1303	-1.7915	0.0048	-1.5467	-1.6402	0.0021	-1.6319	-1.6744	0.0024	-1.7228	-1.1811	-0.0007
TERM(Q)	-0.9576	-1.9252	0.0130	-1.1173	-1.5760	0.0046	-1.5336	-2.1084	0.0105	-1.0630	-0.9172	-0.0005
TERM(H)	-0.4356	-1.2425	0.0040	-0.0934	-0.1987	-0.0028	-0.8040	-1.5583	0.0048	0.1994	0.2393	-0.0027
TERM(Y)	-0.3228	-1.3872	0.0049	-0.2484	-0.7460	-0.0013	-0.5544	-1.4414	0.0046	-0.2945	-0.4553	-0.0021
UEXTERM(Q)	-1.3329	-1.9587	0.0131	-1.6782	-1.7524	0.0059	-2.0058	-2.0263	0.0089	-1.7403	-1.1329	0.0004
UEXTERM(H)	-0.9692	-1.7709	0.0112	-0.7809	-1.0282	0.0003	-1.6033	-2.0177	0.0097	-0.4712	-0.3661	-0.0025
UEXTERM(Y)	-0.5666	-1.4131	0.0061	-0.3438	-0.6114	-0.0018	-0.9125	-1.5201	0.0047	-0.0678	-0.0675	-0.0029
DY(M)	0.1444	1.3503	0.0032	0.1421	0.8139	-0.0009	0.1159	0.6583	-0.0016	0.1350	0.4313	-0.0023
DY(Q)	0.2432	2.6152	0.0518	0.2513	1.9925	0.0173	0.3857	2.6180	0.0422	0.4275	1.8249	0.0177
DY(H)	0.2164	3.0348	0.0983	0.2467	2.2398	0.0427	0.3509	3.1624	0.0843	0.4589	2.5370	0.0527
DY(Y)	0.0857	1.6627	0.0342	0.0816	1.1827	0.0087	0.1060	1.6436	0.0157	0.1492	1.3408	0.0108
UEXDY(Q)	6.5836	2.9816	0.0347	5.7865	1.7755	0.0071	8.9435	2.5243	0.0198	7.8709	1.3906	0.0036
UEXDY(H)	7.2386	3.4824	0.0780	7.0434	2.4203	0.0236	10.7504	3.5709	0.0556	10.9226	2.1582	0.0196
UEXDY(Y)	4.9445	3.2239	0.0767	4.5050	2.1004	0.0200	6.6730	3.3355	0.0447	7.1455	1.9894	0.0174
EMR(-1)/VMR(-1)	-0.0545	-2.4760	0.0596	-0.0307	-0.4319	-0.0004	-0.0775	-1.9013	0.0386	-0.0567	-0.5274	0.0001
EMR(-2)/VMR(-2)	-0.0171	-0.9814	0.0123	-0.0255	-0.6236	0.0020	-0.0234	-0.9155	0.0064	-0.0492	-0.7715	0.0035
EMR(-3)/VMR(-3)	-0.0031	-0.2412	-0.0019	-0.0030	-0.1347	-0.0027	-0.0041	-0.2284	-0.0023	-0.0021	-0.0596	-0.0029
EMR(-4)/VMR(-4)	-0.0030	-0.4203	-0.0008	-0.0064	-0.5039	-0.0012	-0.0070	-0.6756	0.0009	-0.0135	-0.6414	-0.0003
EMR(-5)/VMR(-5)	-0.0003	-0.0629	-0.0029	0.0044	0.5576	-0.0013	-0.0033	-0.4601	-0.0010	0.0017	0.1267	-0.0028
EMR(-6)/VMR(-6)	-0.0015	-0.4671	-0.0007	-0.0023	-0.4482	-0.0021	-0.0040	-0.8940	0.0024	-0.0041	-0.4897	-0.0019
EMR(-7)/VMR(-7)	-0.0008	-0.3360	-0.0019	-0.0001	-0.0309	-0.0029	-0.0024	-0.7231	0.0003	-0.0003	-0.0520	-0.0029
EMR(-8)/VMR(-8)	-0.0008	-0.4112	-0.0013	-0.0017	-0.5648	-0.0014	-0.0023	-0.8597	0.0020	-0.0039	-0.8275	-0.0001
EMR(-9)/VMR(-9)	-0.0001	-0.0501	-0.0029	-0.0004	-0.1811	-0.0028	-0.0002	-0.1045	-0.0029	-0.0007	-0.2056	-0.0027
EMR(-10)/VMR(-10)	-0.0001	-0.1123	-0.0028	0.0006	0.3459	-0.0024	-0.0003	-0.1662	-0.0027	0.0006	0.2210	-0.0027

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with share returns in past 12 months, and hold for 3 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return R<sub>wml</sub>.

Panel A Momentum coefficient $\gamma_{mom}$						Panel B Winner/Loser return R <sub>wml</sub>						
	Panel A	A1 Equal-wei	ighted	Panel A	Panel A2 Value-weighted			B1 Equal-weig	ghted	Panel B2 Value-weighted		
Macroeconomic			2			<u> </u>						
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	-0.0111	-1.8301	0.0531	-0.0132	-1.8282	0.0348	-0.0162	-1.8888	0.0431	-0.0223	-2.0892	0.0338
UEXGDP(Y)	-0.0158	-1.6827	0.0463	-0.0188	-1.6524	0.0301	-0.0219	-1.6041	0.0329	-0.0314	-1.8273	0.0281
IP(M)	-1.6502	-1.9561	0.0029	-1.3279	-1.0527	-0.0011	-1.6325	-1.1603	-0.0008	-0.7700	-0.3260	-0.0027
IP(Q)	-1.9997	-1.8886	0.0146	-1.3033	-0.9775	0.0006	-2.4029	-1.4337	0.0067	-1.2553	-0.5405	-0.0018
IP(H)	-1.7148	-1.6910	0.0251	-1.7420	-1.5428	0.0108	-2.3941	-1.5665	0.0178	-2.5974	-1.3306	0.0074
IP(Y)	-0.6000	-0.8613	0.0051	-0.9441	-1.2759	0.0066	-0.8726	-0.8185	0.0036	-1.6385	-1.3075	0.0068
UEXIP(Q)	-0.0127	-1.9156	0.0123	-0.0086	-0.9831	0.0004	-0.0136	-1.2652	0.0037	-0.0077	-0.4883	-0.0020
UEXIP(H)	-0.0142	-1.9964	0.0263	-0.0126	-1.4650	0.0080	-0.0180	-1.6213	0.0150	-0.0169	-1.0995	0.0037
UEXIP(Y)	-0.0089	-1.3964	0.0201	-0.0101	-1.4218	0.0111	-0.0121	-1.2488	0.0131	-0.0162	-1.3352	0.0092
IPM(M)	-1.7163	-1.9968	0.0038	-1.9761	-1.7380	0.0013	-2.0115	-1.4945	0.0006	-2.0214	-0.9692	-0.0014
IPM(Q)	-1.2072	-1.0996	0.0045	-1.1113	-0.8530	0.0001	-1.1009	-0.6766	-0.0006	-1.1732	-0.5379	-0.0018
IPM(H)	-0.9445	-0.9624	0.0073	-0.9547	-0.9201	0.0020	-1.1754	-0.8207	0.0031	-1.5079	-0.8639	0.0012
IPM(Y)	-0.1921	-0.3436	-0.0018	-0.3715	-0.6540	-0.0009	-0.2874	-0.3432	-0.0019	-0.5456	-0.5720	-0.0014
UEXIPM(Q)	-0.0100	-1.4669	0.0077	-0.0098	-1.1796	0.0019	-0.0097	-0.9371	0.0009	-0.0106	-0.7310	-0.0010
UEXIPM(H)	-0.0089	-1.2226	0.0105	-0.0086	-1.0490	0.0031	-0.0098	-0.9289	0.0034	-0.0120	-0.8532	0.0010
UEXIPM(Y)	-0.0047	-0.8241	0.0055	-0.0054	-0.8871	0.0022	-0.0059	-0.7058	0.0021	-0.0083	-0.8247	0.0013
M0 (M)	-3.6474	-0.9660	0.0016	-3.7434	-0.7249	-0.0007	-4.7234	-0.7756	0.0000	-7.6401	-0.8127	0.0003
M0 (Q)	-1.8466	-0.7009	0.0009	-0.3228	-0.0848	-0.0029	-4.8930	-1.1351	0.0072	-3.6115	-0.5289	-0.0006
M0 (H)	0.2444	0.1020	-0.0028	1.0080	0.3456	-0.0018	-0.3694	-0.0947	-0.0028	0.1257	0.0235	-0.0029
M0 (Y)	0.4600	0.2466	-0.0017	0.2051	0.0968	-0.0028	-0.1002	-0.0322	-0.0029	-1.0279	-0.2653	-0.0020
UEXM0(Q)	0.0000	-0.5622	0.0015	0.0000	0.1594	-0.0026	-0.0001	-0.5704	0.0013	0.0000	-0.0607	-0.0029
UEXM0 (H)	0.0000	-0.3268	-0.0012	0.0000	0.4083	-0.0003	0.0000	-0.3181	-0.0012	0.0000	0.2050	-0.0022
UEXM0 (Y)	0.0000	-0.2202	-0.0020	0.0000	0.3855	-0.0006	0.0000	-0.2142	-0.0021	0.0000	0.1730	-0.0024
UNE (M)	0.5051	0.3662	-0.0018	-0.1346	-0.0971	-0.0029	0.5819	0.2775	-0.0023	0.7478	0.3073	-0.0025
UNE (Q)	0.3660	0.5551	0.0008	0.1686	0.2536	-0.0026	0.6150	0.6151	0.0011	0.4774	0.4083	-0.0019
UNE (H)	0.1198	0.3414	-0.0015	0.0060	0.0172	-0.0029	0.2052	0.4025	-0.0014	0.1070	0.1792	-0.0028
UNE (Y)	-0.1247	-0.7876	0.0024	-0.1669	-0.9154	0.0016	-0.1439	-0.6068	-0.0002	-0.2093	-0.6948	-0.0005
UEXUNE(Q)	0.0143	0.2580	-0.0021	0.0015	0.0287	-0.0029	0.0203	0.2446	-0.0023	0.0140	0.1624	-0.0028
UEXUNE(H)	0.0079	0.2394	-0.0022	0.0014	0.0462	-0.0029	0.0129	0.2625	-0.0022	0.0075	0.1445	-0.0028
UEXUNE(Y)	-0.0043	-0.2521	-0.0023	-0.0077	-0.4387	-0.0020	-0.0044	-0.1718	-0.0027	-0.0090	-0.3176	-0.0025
CPI (M)	-0.1174	-0.8625	0.0006	-0.2522	-1.3117	0.0049	-0.2580	-1.0824	0.0037	-0.4999	-1.4218	0.0075
CPI (Q)	-0.0998	-0.9034	0.0067	-0.2194	-1.4508	0.0192	-0.1908	-1.0126	0.0105	-0.4443	-1.5647	0.0279
CPI (H)	-0.1154	-1.1931	0.0290	-0.2387	-1.9709	0.0618	-0.2059	-1.3393	0.0357	-0.4730	-2.2139	0.0832
CPI (Y)	-0.0616	-0.9627	0.0182	-0.1199	-1.5582	0.0351	-0.0953	-0.9624	0.0163	-0.2287	-1.6683	0.0439
UEXCPI(Q)	0.0032	0.2397	-0.0024	-0.0021	-0.1216	-0.0028	0.0024	0.1181	-0.0028	-0.0018	-0.0629	-0.0029

 Table 5.17 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/6)
Continued												
UEXCPI(H)	0.0031	0.2672	-0.0019	-0.0045	-0.3235	-0.0019	0.0026	0.1500	-0.0027	-0.0093	-0.3997	-0.0014
UEXCPI(Y)	0.0038	0.5506	0.0010	-0.0028	-0.3403	-0.0019	0.0048	0.4330	-0.0006	-0.0060	-0.4134	-0.0014
TBILL(M)	0.1200	1.2888	0.0020	0.1114	0.8399	-0.0009	0.2377	1.5834	0.0044	0.2567	1.1165	0.0007
TBILL (Q)	0.0796	1.1981	0.0046	0.0490	0.5455	-0.0016	0.1604	1.4666	0.0087	0.0819	0.5126	-0.0017
TBILL (H)	0.0589	1.0713	0.0058	0.0346	0.4881	-0.0015	0.0931	1.0074	0.0054	0.0330	0.2609	-0.0025
TBILL (Y)	0.0587	1.6176	0.0157	0.0713	1.4421	0.0101	0.0945	1.6509	0.0154	0.1120	1.2581	0.0080
UEXTBILL(Q)	20.1913	1.8359	0.0122	19.4165	1.4537	0.0037	40.4623	2.3232	0.0203	36.2870	1.6340	0.0049
UEXTBILL(H)	18.0106	1.7912	0.0180	15.7914	1.3841	0.0047	32.7696	2.0758	0.0234	25.3031	1.3360	0.0037
UEXTBILL(Y)	15.6764	1.9440	0.0270	15.1137	1.6153	0.0103	26.0993	2.0906	0.0286	22.8881	1.4404	0.0073
TERM(M)	-1.5386	-1.2719	0.0027	-1.2554	-0.8379	-0.0011	-2.2203	-1.2738	0.0016	-2.1677	-0.8970	-0.0011
TERM(Q)	-0.9276	-1.0104	0.0030	-0.6934	-0.6274	-0.0014	-1.6753	-1.2704	0.0044	-1.2123	-0.6260	-0.0013
TERM(H)	-0.1485	-0.2104	-0.0026	0.4098	0.5130	-0.0018	-0.2136	-0.2079	-0.0027	0.9875	0.6964	-0.0007
TERM(Y)	-0.1674	-0.3559	-0.0021	-0.0488	-0.0883	-0.0029	-0.4193	-0.6115	-0.0010	-0.2445	-0.2403	-0.0027
UEXTERM(Q)	-1.4777	-1.1434	0.0048	-1.2011	-0.7706	-0.0005	-2.3882	-1.2929	0.0048	-2.0979	-0.7885	-0.0004
UEXTERM(H)	-0.7537	-0.6920	0.0004	-0.1102	-0.0866	-0.0029	-1.3112	-0.8391	0.0009	0.0395	0.0177	-0.0029
UEXTERM(Y)	-0.3137	-0.3888	-0.0019	0.1421	0.1495	-0.0028	-0.5928	-0.5108	-0.0015	0.4112	0.2365	-0.0026
DY(M)	0.3513	1.9591	0.0113	0.5353	2.0725	0.0127	0.5054	1.6756	0.0083	0.8116	1.8185	0.0093
DY(Q)	0.4536	2.8023	0.0720	0.5523	2.4446	0.0498	0.7371	2.4744	0.0725	0.9247	2.2341	0.0472
DY(H)	0.3185	2.4192	0.0826	0.3241	1.7927	0.0391	0.4753	2.3059	0.0696	0.6165	1.9019	0.0486
DY(Y)	0.1213	1.2267	0.0263	0.1110	0.8518	0.0087	0.1436	0.9843	0.0127	0.2009	0.9100	0.0100
UEXDY(Q)	12.7522	3.2146	0.0526	14.4553	2.5944	0.0310	18.6052	2.7810	0.0421	21.8858	2.2924	0.0234
UEXDY(H)	12.0511	3.2898	0.0851	12.2955	2.3626	0.0406	17.3431	2.8808	0.0665	19.7611	2.1746	0.0351
UEXDY(Y)	7.5894	2.6008	0.0708	7.2548	1.8616	0.0291	9.6789	2.2252	0.0427	12.0626	1.8057	0.0270
EMR(-1)/VMR(-1)	-0.1115	-2.4271	0.0976	-0.0461	-0.3251	0.0000	-0.1643	-2.1973	0.0803	-0.0779	-0.3316	-0.0001
EMR(-2)/VMR(-2)	-0.0331	-0.9271	0.0191	-0.0056	-0.0690	-0.0028	-0.0461	-0.8163	0.0134	-0.0170	-0.1229	-0.0025
EMR(-3)/VMR(-3)	-0.0054	-0.2124	-0.0017	0.0077	0.1710	-0.0023	-0.0100	-0.2557	-0.0013	0.0128	0.1690	-0.0024
EMR(-4)/VMR(-4)	-0.0071	-0.5203	0.0018	-0.0077	-0.3055	-0.0016	-0.0162	-0.7519	0.0063	-0.0192	-0.4455	-0.0002
EMR(-5)/VMR(-5)	-0.0014	-0.1500	-0.0025	0.0083	0.5234	0.0003	-0.0079	-0.5364	0.0020	0.0038	0.1405	-0.0027
EMR(-6)/VMR(-6)	-0.0033	-0.5656	0.0014	-0.0034	-0.3540	-0.0019	-0.0087	-0.9308	0.0085	-0.0079	-0.4916	-0.0010
EMR(-7)/VMR(-7)	-0.0016	-0.3686	-0.0011	0.0012	0.1682	-0.0027	-0.0053	-0.7266	0.0044	-0.0006	-0.0546	-0.0029
EMR(-8)/VMR(-8)	-0.0021	-0.5978	0.0020	-0.0028	-0.4939	-0.0008	-0.0060	-0.9911	0.0120	-0.0078	-0.8539	0.0027
EMR(-9)/VMR(-9)	-0.0001	-0.0299	-0.0029	0.0008	0.1750	-0.0027	-0.0008	-0.1930	-0.0025	0.0003	0.0420	-0.0029
EMR(-10)/VMR(-10)	-0.0001	-0.0776	-0.0029	0.0019	0.5853	0.0000	-0.0005	-0.1647	-0.0025	0.0022	0.4112	-0.0016

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with share returns in past 12 months, and hold for 6 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return R<sub>wml</sub>.

	Panel A Momentum coefficient $\gamma_{mom}$							Panel B Winner/Loser return R <sub>wml</sub>					
	Panel A	1 Equal-wei	ghted	Panel A	2 Value-wei	ghted	Panel E	1 Equal-weig	ghted	Panel B	2 Value-wei	ghted	
Macroeconomic												-	
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	-0.0230	-2.2841	0.0917	-0.0217	-2.1949	0.0497	-0.0376	-2.6357	0.0793	-0.0320	-2.1598	0.0302	
UEXGDP(Y)	-0.0332	-2.1443	0.0816	-0.0318	-1.8946	0.0454	-0.0518	-2.3756	0.0638	-0.0451	-1.7937	0.0252	
IP(M)	-2.6558	-1.9053	0.0029	-1.1004	-0.6280	-0.0024	-2.3829	-1.0331	-0.0015	-1.1190	-0.3694	-0.0028	
IP(Q)	-2.8952	-1.5004	0.0114	-1.0584	-0.5073	-0.0018	-3.6551	-1.2873	0.0044	-2.1188	-0.6049	-0.0016	
IP(H)	-2.7850	-1.6415	0.0258	-1.7136	-1.0409	0.0038	-4.3154	-1.7881	0.0194	-3.1869	-1.1351	0.0038	
IP(Y)	-0.9893	-0.9067	0.0056	-0.7316	-0.6799	-0.0001	-1.8064	-1.1044	0.0063	-1.4974	-0.7692	0.0006	
UEXIP(Q)	-0.0195	-1.6077	0.0109	-0.0089	-0.6571	-0.0012	-0.0202	-1.0949	0.0019	-0.0157	-0.6989	-0.0013	
UEXIP(H)	-0.0223	-1.7329	0.0252	-0.0135	-1.0132	0.0034	-0.0299	-1.5929	0.0133	-0.0255	-1.1343	0.0036	
UEXIP(Y)	-0.0141	-1.3179	0.0194	-0.0092	-0.8467	0.0030	-0.0215	-1.3844	0.0138	-0.0190	-1.0037	0.0043	
IPM(M)	-2.2750	-1.5223	0.0016	-0.8453	-0.4840	-0.0026	-2.4876	-1.1298	-0.0012	-0.8487	-0.3066	-0.0029	
IPM(Q)	-1.3283	-0.6703	0.0005	-0.1573	-0.0802	-0.0030	-1.0631	-0.3859	-0.0023	-0.3081	-0.0994	-0.0030	
IPM(H)	-1.3970	-0.8805	0.0057	-0.5959	-0.4250	-0.0020	-2.0810	-0.9622	0.0033	-0.9929	-0.4290	-0.0022	
IPM(Y)	-0.0951	-0.1146	-0.0029	0.2368	0.2880	-0.0026	-0.3504	-0.2643	-0.0025	0.5531	0.3706	-0.0023	
UEXIPM(Q)	-0.0130	-1.0266	0.0039	-0.0044	-0.3342	-0.0025	-0.0112	-0.6274	-0.0013	-0.0070	-0.3396	-0.0026	
UEXIPM(H)	-0.0122	-0.9561	0.0069	-0.0056	-0.4626	-0.0017	-0.0139	-0.7908	0.0011	-0.0102	-0.5211	-0.0017	
UEXIPM(Y)	-0.0056	-0.6166	0.0017	-0.0008	-0.0888	-0.0029	-0.0084	-0.6337	0.0004	-0.0015	-0.1028	-0.0029	
M0 (M)	-5.8471	-0.9285	0.0016	-6.4588	-0.9245	0.0005	-9.5578	-0.8194	0.0010	-15.1731	-1.0876	0.0026	
M0 (Q)	-3.3782	-0.7361	0.0020	-0.7200	-0.1519	-0.0029	-9.2503	-1.2308	0.0091	-7.3849	-0.8001	0.0013	
M0 (H)	0.0327	0.0076	-0.0030	1.0354	0.2577	-0.0024	-2.3184	-0.3251	-0.0014	-4.4150	-0.5405	0.0002	
M0 (Y)	0.6166	0.1809	-0.0021	0.5772	0.1636	-0.0025	-1.0550	-0.1705	-0.0022	-3.3179	-0.4641	0.0015	
UEXM0(Q)	-0.0001	-0.5832	0.0023	0.0000	-0.1548	-0.0027	-0.0001	-0.6722	0.0036	-0.0001	-0.4827	-0.0002	
UEXM0 (H)	0.0000	-0.4276	0.0007	0.0000	-0.0003	-0.0030	-0.0001	-0.5539	0.0029	-0.0001	-0.3883	-0.0005	
UEXM0 (Y)	0.0000	-0.3221	-0.0004	0.0000	-0.0098	-0.0030	0.0000	-0.4652	0.0024	-0.0001	-0.4302	0.0010	
UNE (M)	1.7383	0.7199	0.0023	-1.1572	-0.5250	-0.0015	3.3938	0.9220	0.0036	0.2054	0.0500	-0.0030	
UNE (Q)	0.6711	0.6059	0.0020	-0.4285	-0.4037	-0.0017	1.3685	0.8179	0.0037	0.0913	0.0489	-0.0030	
UNE (H)	0.1386	0.2466	-0.0023	-0.2953	-0.5228	-0.0009	0.4940	0.5736	0.0000	-0.1078	-0.1130	-0.0029	
UNE (Y)	-0.2299	-0.8370	0.0040	-0.3271	-0.8916	0.0059	-0.1386	-0.3116	-0.0022	-0.3124	-0.5478	-0.0006	
UEXUNE(Q)	0.0337	0.3649	-0.0012	-0.0424	-0.4978	-0.0012	0.0716	0.5126	-0.0003	-0.0270	-0.1887	-0.0028	
UEXUNE(H)	0.0110	0.2017	-0.0025	-0.0252	-0.4773	-0.0012	0.0328	0.3948	-0.0015	-0.0205	-0.2373	-0.0027	
UEXUNE(Y)	-0.0105	-0.3658	-0.0015	-0.0220	-0.6722	0.0011	-0.0027	-0.0590	-0.0030	-0.0234	-0.4571	-0.0016	
CPI (M)	-0.2294	-0.8399	0.0023	-0.4608	-1.5655	0.0104	-0.5122	-0.9745	0.0056	-1.1837	-1.8002	0.0227	
CPI (Q)	-0.1849	-0.8287	0.0100	-0.3753	-1.5060	0.0303	-0.3525	-0.8307	0.0123	-0.8691	-1.5515	0.0487	
CPI (H)	-0.1702	-0.9535	0.0241	-0.2998	-1.4886	0.0495	-0.3287	-1.0001	0.0298	-0.6317	-1.5125	0.0645	
CPI (Y)	-0.0592	-0.5365	0.0046	-0.1015	-0.8295	0.0110	-0.1077	-0.5749	0.0052	-0.1544	-0.6578	0.0064	
UEXCPI(Q)	0.0078	0.3058	-0.0017	-0.0074	-0.2876	-0.0023	0.0068	0.1597	-0.0027	-0.0243	-0.4933	-0.0008	

 Table 5.18 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/12)

Continued												
UEXCPI(H)	0.0077	0.3656	-0.0005	-0.0067	-0.3107	-0.0018	0.0075	0.2095	-0.0022	-0.0169	-0.4148	-0.0008
UEXCPI(Y)	0.0088	0.6691	0.0050	-0.0021	-0.1547	-0.0027	0.0113	0.4871	0.0013	-0.0005	-0.0218	-0.0030
TBILL(M)	0.1968	1.2535	0.0022	0.1314	0.8291	-0.0016	0.4085	1.4520	0.0042	0.2599	0.8359	-0.0014
TBILL (Q)	0.1248	1.0656	0.0042	0.1083	0.8411	0.0004	0.2491	1.1528	0.0062	0.1586	0.6110	-0.0009
TBILL (H)	0.1006	1.0018	0.0069	0.1359	1.2016	0.0082	0.1451	0.7903	0.0037	0.1893	0.8548	0.0033
TBILL (Y)	0.1045	1.7491	0.0200	0.1466	1.9569	0.0252	0.1684	1.6394	0.0164	0.2742	2.1201	0.0257
UEXTBILL(Q)	31.4699	1.5937	0.0114	30.3698	1.7246	0.0054	62.8451	1.8061	0.0156	51.8052	1.4801	0.0040
UEXTBILL(H)	27.9316	1.5150	0.0166	30.9205	1.7267	0.0120	49.3265	1.5408	0.0168	50.0608	1.5133	0.0084
UEXTBILL(Y)	24.3336	1.6350	0.0252	29.2735	1.8051	0.0224	40.0870	1.5535	0.0218	50.2796	1.7986	0.0187
TERM(M)	-1.6553	-0.9638	-0.0004	-0.9620	-0.5298	-0.0025	-2.4157	-0.8984	-0.0012	-0.8035	-0.2341	-0.0029
TERM(Q)	-0.7819	-0.5362	-0.0014	-0.4793	-0.3076	-0.0026	-1.2449	-0.5269	-0.0017	-0.1799	-0.0601	-0.0030
TERM(H)	0.1901	0.1627	-0.0028	0.4346	0.3210	-0.0023	0.8117	0.4497	-0.0018	1.4293	0.5777	-0.0009
TERM(Y)	0.0183	0.0230	-0.0030	-0.1008	-0.1085	-0.0029	-0.3435	-0.2872	-0.0026	-0.5411	-0.3589	-0.0024
UEXTERM(Q)	-1.4329	-0.7204	-0.0002	-0.8868	-0.4339	-0.0023	-2.1360	-0.6756	-0.0009	-0.5977	-0.1525	-0.0029
UEXTERM(H)	-0.4240	-0.2381	-0.0026	0.1603	0.0813	-0.0030	-0.3517	-0.1261	-0.0029	1.2054	0.3291	-0.0024
UEXTERM(Y)	0.0035	0.0025	-0.0030	0.1120	0.0737	-0.0030	0.0427	0.0208	-0.0030	0.5079	0.1915	-0.0028
DY(M)	0.4558	1.4188	0.0063	0.6079	1.5144	0.0073	0.6673	1.0988	0.0034	1.5133	2.0360	0.0155
DY(Q)	0.5179	1.7343	0.0346	0.5563	1.4173	0.0241	0.9127	1.5625	0.0349	1.2711	1.7060	0.0380
DY(H)	0.3577	1.4572	0.0390	0.3113	1.0070	0.0168	0.5409	1.2596	0.0281	0.7281	1.2616	0.0285
DY(Y)	0.1205	0.7026	0.0083	0.0496	0.2394	-0.0018	0.1227	0.4619	0.0008	0.1547	0.4481	0.0004
UEXDY(Q)	15.2544	2.0306	0.0278	14.3086	1.4464	0.0139	22.2821	1.6212	0.0183	33.2396	1.8950	0.0234
UEXDY(H)	14.0720	1.9966	0.0436	11.9929	1.3108	0.0181	19.8938	1.6000	0.0272	27.6315	1.7179	0.0295
UEXDY(Y)	8.4133	1.5265	0.0323	5.5409	0.8153	0.0066	10.2372	1.1595	0.0140	14.2636	1.2507	0.0153
EMR(-1)/VMR(-1)	-0.1791	-1.8859	0.0955	-0.0008	-0.0031	-0.0030	-0.2959	-1.8061	0.0842	-0.0666	-0.1570	-0.0021
EMR(-2)/VMR(-2)	-0.0371	-0.5323	0.0076	0.0491	0.3659	0.0019	-0.0703	-0.6491	0.0093	0.0740	0.3059	0.0002
EMR(-3)/VMR(-3)	-0.0015	-0.0317	-0.0030	0.0175	0.2318	-0.0014	-0.0082	-0.1153	-0.0026	0.0191	0.1409	-0.0025
EMR(-4)/VMR(-4)	-0.0081	-0.3389	-0.0007	-0.0020	-0.0488	-0.0029	-0.0215	-0.5658	0.0023	-0.0248	-0.3181	-0.0010
EMR(-5)/VMR(-5)	-0.0011	-0.0755	-0.0029	0.0148	0.5529	0.0022	-0.0101	-0.4146	-0.0003	0.0003	0.0052	-0.0030
EMR(-6)/VMR(-6)	-0.0047	-0.4452	0.0004	-0.0006	-0.0381	-0.0030	-0.0119	-0.6766	0.0040	-0.0087	-0.2958	-0.0020
EMR(-7)/VMR(-7)	-0.0028	-0.3594	-0.0009	0.0085	0.6816	0.0028	-0.0092	-0.6870	0.0042	0.0051	0.2324	-0.0024
EMR(-8)/VMR(-8)	-0.0053	-0.7894	0.0088	-0.0019	-0.1907	-0.0025	-0.0135	-1.1547	0.0215	-0.0110	-0.6488	0.0018
EMR(-9)/VMR(-9)	-0.0004	-0.0865	-0.0029	0.0025	0.3343	-0.0015	-0.0021	-0.2887	-0.0019	0.0005	0.0375	-0.0030
EMR(-10)/VMR(-10)	-0.0005	-0.1844	-0.0026	0.0039	0.6573	0.0029	-0.0014	-0.2637	-0.0020	0.0047	0.4749	-0.0004

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with share returns in past 12 months, and hold for 12 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return R<sub>wml</sub>.

	Panel A Momentum coefficient $\gamma_{mom}$						88	Pane	l B Winner/Lo	oser return R <sub>wml</sub>	- /	
	Panel A	1 Equal-wei	ghted	Panel A	2 Value-wei	ghted	Panel E	B1 Equal-weig	ghted	Panel B	2 Value-wei	ghted
Macroeconomic												
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	-0.0305	-2.2739	0.1061	-0.0250	-2.2500	0.0556	-0.0509	-2.6923	0.0955	-0.0425	-2.6080	0.0425
UEXGDP(Y)	-0.0438	-2.2010	0.0930	-0.0383	-2.0901	0.0559	-0.0727	-2.5541	0.0831	-0.0611	-2.2775	0.0372
IP(M)	-3.2266	-1.7177	0.0025	-1.9909	-1.0734	-0.0014	-3.0767	-1.0803	-0.0014	-3.5684	-1.1200	-0.0016
IP(Q)	-3.8792	-1.5694	0.0137	-1.8161	-0.8307	-0.0001	-5.0671	-1.5716	0.0062	-3.4286	-0.9637	-0.0002
IP(H)	-3.6131	-1.7749	0.0286	-2.0030	-1.1122	0.0048	-6.0173	-2.2301	0.0254	-4.5161	-1.4560	0.0076
IP(Y)	-1.4584	-1.1279	0.0091	-1.2358	-1.0639	0.0040	-3.2171	-1.6371	0.0162	-3.0229	-1.3869	0.0082
UEXIP(Q)	-0.0250	-1.5802	0.0118	-0.0141	-0.9839	0.0007	-0.0272	-1.2094	0.0027	-0.0271	-1.1533	0.0007
UEXIP(H)	-0.0285	-1.7711	0.0267	-0.0169	-1.1814	0.0054	-0.0401	-1.8567	0.0161	-0.0358	-1.4744	0.0070
UEXIP(Y)	-0.0183	-1.4224	0.0215	-0.0119	-1.0459	0.0052	-0.0320	-1.7458	0.0213	-0.0287	-1.3899	0.0099
IPM(M)	-2.0830	-1.0496	-0.0006	-1.0335	-0.5463	-0.0026	-2.0040	-0.7080	-0.0023	-2.5090	-0.7919	-0.0023
IPM(Q)	-1.3053	-0.5171	-0.0008	-0.2367	-0.1189	-0.0030	-1.1910	-0.3463	-0.0025	-0.4394	-0.1311	-0.0030
IPM(H)	-1.4351	-0.7780	0.0030	0.0565	0.0392	-0.0030	-2.4773	-0.9555	0.0028	-0.4424	-0.1700	-0.0029
IPM(Y)	-0.1415	-0.1460	-0.0029	0.3273	0.3310	-0.0024	-0.9731	-0.5795	-0.0006	0.1819	0.0949	-0.0030
UEXIPM(Q)	-0.0123	-0.7481	0.0010	-0.0048	-0.3578	-0.0026	-0.0100	-0.4322	-0.0022	-0.0093	-0.4180	-0.0025
UEXIPM(H)	-0.0122	-0.7752	0.0034	-0.0029	-0.2291	-0.0028	-0.0150	-0.6827	0.0001	-0.0080	-0.3737	-0.0024
UEXIPM(Y)	-0.0058	-0.5407	0.0002	0.0014	0.1596	-0.0029	-0.0120	-0.7283	0.0015	-0.0015	-0.0880	-0.0030
M0 (M)	-5.5479	-0.7478	-0.0004	-3.2524	-0.4814	-0.0023	-14.6767	-1.1035	0.0030	-17.0833	-1.1888	0.0024
M0 (Q)	-3.0711	-0.5280	-0.0004	1.6282	0.3240	-0.0024	-12.7001	-1.2895	0.0118	-8.5687	-0.7921	0.0014
M0 (H)	1.4916	0.2536	-0.0017	2.5976	0.5622	0.0002	-2.4632	-0.2436	-0.0019	-4.2205	-0.4125	-0.0008
M0 (Y)	1.1390	0.2489	-0.0011	1.8592	0.4394	0.0010	-2.5747	-0.3087	0.0001	-2.0406	-0.2286	-0.0017
UEXM0(Q)	0.0000	-0.3270	-0.0014	0.0000	0.1047	-0.0029	-0.0002	-0.8421	0.0086	-0.0002	-0.6057	0.0015
UEXM0 (H)	0.0000	-0.1173	-0.0028	0.0000	0.2459	-0.0020	-0.0001	-0.6085	0.0056	-0.0001	-0.4466	0.0006
UEXM0 (Y)	0.0000	-0.0405	-0.0030	0.0000	0.2216	-0.0020	-0.0001	-0.5180	0.0051	-0.0001	-0.4260	0.0013
UNE (M)	1.7618	0.5963	0.0005	-1.4995	-0.5738	-0.0010	4.8854	1.1091	0.0059	1.9800	0.4024	-0.0021
UNE (Q)	0.7303	0.5559	0.0008	-0.3949	-0.3281	-0.0022	2.1631	1.0865	0.0078	1.2693	0.5857	-0.0006
UNE (H)	0.1580	0.2365	-0.0024	-0.3308	-0.4793	-0.0009	0.9797	0.9442	0.0046	0.5302	0.4526	-0.0016
UNE (Y)	-0.3633	-1.0161	0.0084	-0.4500	-0.9743	0.0111	-0.0923	-0.1589	-0.0028	-0.1535	-0.2089	-0.0026
UEXUNE(Q)	0.0294	0.2597	-0.0021	-0.0541	-0.5276	-0.0006	0.1148	0.6797	0.0014	0.0341	0.1959	-0.0028
UEXUNE(H)	0.0084	0.1261	-0.0028	-0.0325	-0.5054	-0.0006	0.0635	0.6305	0.0007	0.0220	0.2065	-0.0028
UEXUNE(Y)	-0.0195	-0.5358	0.0003	-0.0330	-0.7921	0.0047	0.0058	0.1002	-0.0030	-0.0085	-0.1287	-0.0029
CPI (M)	-0.2280	-0.6770	0.0004	-0.3045	-0.9825	0.0018	-0.7327	-1.0998	0.0084	-1.0514	-1.4732	0.0126
CPI (Q)	-0.1800	-0.7142	0.0049	-0.2093	-0.8405	0.0056	-0.5596	-1.1297	0.0220	-0.6784	-1.1493	0.0214
CPI (H)	-0.1662	-0.8293	0.0138	-0.1553	-0.7798	0.0088	-0.5018	-1.3574	0.0469	-0.3948	-0.8820	0.0174
CPI (Y)	-0.0339	-0.2894	-0.0014	-0.0075	-0.0615	-0.0030	-0.1730	-0.8267	0.0107	0.0144	0.0587	-0.0030
UEXCPI(Q)	0.0128	0.4028	-0.0008	0.0078	0.2926	-0.0024	-0.0015	-0.0278	-0.0030	-0.0037	-0.0708	-0.0030

 Table 5.19 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/18)

Continued												
UEXCPI(H)	0.0112	0.4264	0.0004	0.0048	0.2153	-0.0025	-0.0042	-0.0928	-0.0029	0.0016	0.0373	-0.0030
UEXCPI(Y)	0.0107	0.6567	0.0047	0.0030	0.1994	-0.0026	0.0028	0.0953	-0.0029	0.0109	0.3881	-0.0013
TBILL(M)	0.2110	1.0460	0.0008	0.3015	1.5984	0.0033	0.3358	0.9542	0.0001	0.4754	1.2830	0.0012
TBILL (Q)	0.1157	0.7851	0.0010	0.2317	1.5995	0.0098	0.1539	0.5778	-0.0008	0.3723	1.1983	0.0059
TBILL (H)	0.0983	0.7853	0.0031	0.2073	1.6142	0.0188	0.0787	0.3395	-0.0018	0.3284	1.2908	0.0116
TBILL (Y)	0.1131	1.5319	0.0142	0.1998	2.2456	0.0403	0.1420	1.1510	0.0058	0.3743	2.6087	0.0376
UEXTBILL(Q)	30.3820	1.1414	0.0057	50.3954	2.2277	0.0163	48.9034	1.1008	0.0043	90.7217	2.0242	0.0137
UEXTBILL(H)	26.3070	1.0467	0.0083	45.4966	2.0133	0.0242	35.3051	0.8202	0.0036	81.6230	1.9127	0.0204
UEXTBILL(Y)	22.4869	1.1612	0.0126	38.2184	1.8497	0.0332	28.1732	0.8677	0.0049	71.4012	2.0980	0.0308
TERM(M)	-1.9074	-0.7860	-0.0008	-1.8753	-0.8479	-0.0013	-0.9390	-0.2622	-0.0029	-1.0729	-0.2672	-0.0029
TERM(Q)	-0.4333	-0.2354	-0.0027	-1.7303	-1.0038	0.0011	0.4948	0.1726	-0.0029	-2.0116	-0.5732	-0.0015
TERM(H)	0.8622	0.6053	-0.0003	-0.2072	-0.1461	-0.0029	2.4712	1.1378	0.0042	0.0744	0.0273	-0.0030
TERM(Y)	0.4582	0.4672	-0.0015	-0.3568	-0.3191	-0.0023	0.6576	0.4326	-0.0020	-1.1343	-0.5867	-0.0010
UEXTERM(Q)	-1.2557	-0.4764	-0.0016	-2.4404	-1.0434	0.0013	-0.0218	-0.0055	-0.0030	-2.3287	-0.5076	-0.0020
UEXTERM(H)	0.2481	0.1105	-0.0030	-1.1197	-0.5281	-0.0015	1.9979	0.5867	-0.0011	-0.8147	-0.1966	-0.0028
UEXTERM(Y)	0.7470	0.4300	-0.0015	-0.6490	-0.3777	-0.0021	2.0031	0.7763	0.0006	-0.9339	-0.3016	-0.0025
DY(M)	0.5148	1.3586	0.0047	0.7911	1.7795	0.0116	0.8236	1.0614	0.0034	2.2267	2.5123	0.0280
DY(Q)	0.5731	1.6810	0.0270	0.6203	1.4633	0.0253	1.0058	1.3940	0.0270	1.6639	2.0100	0.0515
DY(H)	0.3707	1.3353	0.0263	0.2587	0.7633	0.0085	0.4896	0.9432	0.0136	0.7441	1.2293	0.0224
DY(Y)	0.1362	0.7154	0.0063	0.1220	0.5584	0.0030	0.1102	0.3484	-0.0011	0.2979	0.8411	0.0066
UEXDY(Q)	16.7603	1.8846	0.0212	16.3226	1.4932	0.0154	23.6595	1.3788	0.0126	44.2619	2.0678	0.0334
UEXDY(H)	14.5494	1.7499	0.0295	10.3188	0.9922	0.0101	18.6570	1.2260	0.0143	30.0246	1.6099	0.0267
UEXDY(Y)	8.3306	1.2769	0.0195	4.9539	0.6300	0.0034	8.4139	0.7800	0.0044	15.5429	1.2149	0.0139
EMR(-1)/VMR(-1)	-0.1847	-1.5813	0.0651	-0.0611	-0.2725	-0.0009	-0.2985	-1.3701	0.0548	-0.2404	-0.5861	0.0060
EMR(-2)/VMR(-2)	-0.0291	-0.3502	0.0012	0.0083	0.0641	-0.0029	-0.0464	-0.3451	0.0004	-0.0127	-0.0532	-0.0030
EMR(-3)/VMR(-3)	0.0024	0.0431	-0.0030	-0.0144	-0.1912	-0.0021	0.0073	0.0845	-0.0029	-0.0599	-0.4228	0.0011
EMR(-4)/VMR(-4)	-0.0068	-0.2380	-0.0020	-0.0282	-0.6581	0.0046	-0.0115	-0.2552	-0.0021	-0.0871	-0.9942	0.0163
EMR(-5)/VMR(-5)	-0.0034	-0.1927	-0.0024	-0.0053	-0.1849	-0.0025	-0.0095	-0.3314	-0.0015	-0.0464	-0.8390	0.0085
EMR(-6)/VMR(-6)	-0.0069	-0.5342	0.0017	-0.0118	-0.6383	0.0022	-0.0096	-0.4293	-0.0001	-0.0328	-0.9253	0.0077
EMR(-7)/VMR(-7)	-0.0065	-0.6669	0.0041	0.0003	0.0174	-0.0030	-0.0126	-0.7528	0.0056	-0.0133	-0.4844	0.0000
EMR(-8)/VMR(-8)	-0.0092	-1.0772	0.0196	-0.0082	-0.6633	0.0046	-0.0180	-1.2194	0.0250	-0.0274	-1.2916	0.0196
EMR(-9)/VMR(-9)	-0.0016	-0.3301	-0.0018	-0.0008	-0.0977	-0.0029	-0.0024	-0.2900	-0.0021	-0.0090	-0.6083	0.0013
EMR(-10)/VMR(-10)	-0.0011	-0.3389	-0.0019	0.0023	0.3198	-0.0013	-0.0002	-0.0329	-0.0030	-0.0012	-0.0988	-0.0029

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with share returns in past 12 months, and hold for 18 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return R<sub>wml</sub>.

	Panel A Momentum coefficient $\gamma_{mom}$							Panel B Winner/Loser return R <sub>wml</sub>					
	Panel A	1 Equal-wei	ghted	Panel A	2 Value-wei	ghted	Panel E	B1 Equal-weig	ghted	Panel B	2 Value-wei	ghted	
Macroeconomic													
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	-0.0246	-1.8755	0.0587	-0.0177	-1.5679	0.0225	-0.0319	-1.6703	0.0277	-0.0360	-2.2176	0.0250	
UEXGDP(Y)	-0.0388	-2.0223	0.0626	-0.0302	-1.7325	0.0288	-0.0542	-2.0155	0.0348	-0.0528	-2.1433	0.0228	
IP(M)	-5.1010	-2.1931	0.0089	-2.8575	-1.4600	-0.0001	-6.8237	-1.7957	0.0033	-5.7626	-1.7595	0.0002	
IP(Q)	-6.2598	-2.0536	0.0347	-2.9873	-1.4057	0.0038	-10.3011	-2.2365	0.0272	-6.3348	-1.7925	0.0051	
IP(H)	-5.1836	-2.1518	0.0534	-2.2901	-1.3332	0.0058	-9.6553	-2.6815	0.0550	-5.7207	-1.9232	0.0116	
IP(Y)	-1.6152	-1.1455	0.0099	-1.0581	-0.8853	0.0014	-3.0545	-1.3069	0.0106	-3.2022	-1.3973	0.0077	
UEXIP(Q)	-0.0397	-2.0850	0.0294	-0.0223	-1.5759	0.0051	-0.0588	-1.9670	0.0180	-0.0462	-1.9737	0.0062	
UEXIP(H)	-0.0423	-2.2414	0.0537	-0.0212	-1.5193	0.0084	-0.0710	-2.4968	0.0443	-0.0497	-2.0832	0.0135	
UEXIP(Y)	-0.0247	-1.6805	0.0356	-0.0124	-1.0843	0.0047	-0.0451	-1.9645	0.0351	-0.0345	-1.6189	0.0130	
IPM(M)	-3.2338	-1.5676	0.0020	-1.7508	-0.9391	-0.0019	-3.9939	-1.2469	-0.0008	-4.0469	-1.3212	-0.0014	
IPM(Q)	-2.1705	-0.8318	0.0022	-0.8767	-0.4933	-0.0024	-3.0513	-0.8221	0.0000	-2.0124	-0.5952	-0.0021	
IPM(H)	-2.2147	-1.1106	0.0093	0.1067	0.0852	-0.0031	-4.1993	-1.3842	0.0101	-0.9607	-0.3643	-0.0026	
IPM(Y)	-0.1387	-0.1351	-0.0030	0.7546	0.8030	0.0001	-0.5721	-0.3001	-0.0024	0.3711	0.1839	-0.0029	
UEXIPM(Q)	-0.0189	-1.1047	0.0051	-0.0097	-0.7834	-0.0014	-0.0231	-0.9264	0.0005	-0.0199	-0.9038	-0.0012	
UEXIPM(H)	-0.0185	-1.1173	0.0097	-0.0043	-0.3926	-0.0025	-0.0283	-1.1765	0.0058	-0.0151	-0.7022	-0.0013	
UEXIPM(Y)	-0.0080	-0.7049	0.0023	0.0036	0.4340	-0.0022	-0.0159	-0.8580	0.0031	-0.0029	-0.1606	-0.0030	
M0 (M)	-6.1902	-0.7355	-0.0002	-2.8094	-0.4209	-0.0026	-15.4317	-0.9902	0.0022	-13.9832	-0.9270	0.0000	
M0 (Q)	-3.7446	-0.5647	0.0003	2.6061	0.5184	-0.0018	-14.2901	-1.3456	0.0116	-4.9285	-0.4262	-0.0018	
M0 (H)	-1.7858	-0.2896	-0.0015	1.1692	0.2459	-0.0026	-9.7335	-0.9683	0.0111	-5.0102	-0.4552	-0.0004	
M0 (Y)	-0.7582	-0.1735	-0.0024	1.4268	0.3554	-0.0010	-7.3207	-0.9011	0.0169	-1.4050	-0.1470	-0.0026	
UEXM0(Q)	0.0000	-0.2825	-0.0019	0.0001	0.3827	-0.0014	-0.0002	-0.8673	0.0075	-0.0001	-0.3043	-0.0020	
UEXM0 (H)	0.0000	-0.2058	-0.0023	0.0000	0.4112	-0.0005	-0.0002	-0.8403	0.0104	-0.0001	-0.3183	-0.0013	
UEXM0 (Y)	0.0000	-0.1678	-0.0024	0.0000	0.3732	-0.0006	-0.0001	-0.8108	0.0135	-0.0001	-0.3576	-0.0002	
UNE (M)	2.6727	0.8496	0.0039	-1.5246	-0.5876	-0.0013	6.6127	1.3815	0.0097	3.1537	0.6147	-0.0010	
UNE (Q)	1.1600	0.8214	0.0053	-0.4222	-0.3471	-0.0022	2.8690	1.3067	0.0120	1.9279	0.8424	0.0018	
UNE (H)	0.3415	0.4771	-0.0006	-0.4157	-0.6118	-0.0001	1.1858	1.0225	0.0059	0.6509	0.5381	-0.0012	
UNE (Y)	-0.3229	-0.9690	0.0047	-0.4826	-1.1322	0.0110	-0.0933	-0.1669	-0.0029	-0.0921	-0.1260	-0.0030	
UEXUNE(Q)	0.0601	0.5056	0.0002	-0.0682	-0.7187	0.0003	0.1628	0.9103	0.0040	0.0660	0.3769	-0.0023	
UEXUNE(H)	0.0267	0.3832	-0.0013	-0.0441	-0.7233	0.0008	0.0886	0.8257	0.0027	0.0328	0.3065	-0.0025	
UEXUNE(Y)	-0.0127	-0.3530	-0.0019	-0.0397	-1.0322	0.0067	0.0124	0.2163	-0.0028	-0.0056	-0.0850	-0.0031	
CPI (M)	-0.2762	-0.8692	0.0012	-0.1646	-0.4903	-0.0019	-0.9131	-1.3106	0.0109	-0.8455	-1.0530	0.0055	
CPI (Q)	-0.2467	-0.9539	0.0099	-0.0897	-0.3350	-0.0017	-0.7860	-1.4596	0.0359	-0.5281	-0.7826	0.0095	
CPI (H)	-0.2269	-1.0646	0.0239	-0.0830	-0.3798	-0.0002	-0.7019	-1.7078	0.0736	-0.3091	-0.6096	0.0076	
CPI (Y)	-0.0428	-0.3258	-0.0009	0.0140	0.0975	-0.0029	-0.2147	-0.8981	0.0133	0.0474	0.1657	-0.0025	
UEXCPI(Q)	-0.0010	-0.0311	-0.0031	0.0135	0.4612	-0.0014	-0.0315	-0.5607	0.0004	0.0069	0.1216	-0.0030	

 Table 5.20 Univariate regression of momentum effect on individual lagged macroeconomic variables (Momentum 12/24)

Continued												
UEXCPI(H)	-0.0001	-0.0042	-0.0031	0.0101	0.4306	-0.0012	-0.0296	-0.6268	0.0031	0.0092	0.1900	-0.0027
UEXCPI(Y)	0.0073	0.4420	0.0000	0.0070	0.4210	-0.0008	-0.0040	-0.1369	-0.0028	0.0193	0.6232	0.0015
TBILL(M)	0.1283	0.6860	-0.0019	0.2978	1.4929	0.0023	0.2830	0.8450	-0.0013	0.4411	1.0517	0.0000
TBILL (Q)	0.0209	0.1424	-0.0030	0.2605	1.5957	0.0110	0.0220	0.0813	-0.0031	0.4104	1.1823	0.0062
TBILL (H)	0.0134	0.1006	-0.0030	0.2465	1.6859	0.0234	-0.0605	-0.2332	-0.0025	0.3950	1.4030	0.0149
TBILL (Y)	0.0837	0.8945	0.0050	0.2224	2.0099	0.0429	0.1078	0.6628	0.0009	0.4096	2.4584	0.0383
UEXTBILL(Q)	15.5642	0.6061	-0.0011	49.3403	1.9475	0.0130	34.6673	0.8543	-0.0002	92.0272	1.8583	0.0117
UEXTBILL(H)	8.8164	0.3476	-0.0020	48.0005	1.9410	0.0231	11.1636	0.2648	-0.0026	88.6499	1.9460	0.0206
UEXTBILL(Y)	11.7891	0.5678	0.0006	42.0674	1.8484	0.0349	15.0228	0.4395	-0.0013	79.7407	2.2385	0.0331
TERM(M)	-0.6574	-0.2546	-0.0029	-1.0833	-0.5375	-0.0026	0.4798	0.1236	-0.0031	-0.9675	-0.2494	-0.0030
TERM(Q)	0.5041	0.2598	-0.0027	-1.5329	-0.8810	-0.0002	0.5066	0.1676	-0.0030	-2.0367	-0.5768	-0.0018
TERM(H)	2.1062	1.1410	0.0109	-0.4825	-0.3089	-0.0025	3.8680	1.2714	0.0109	-0.2188	-0.0777	-0.0031
TERM(Y)	1.0408	0.8066	0.0040	-0.3022	-0.2317	-0.0026	1.1189	0.5461	-0.0007	-1.1675	-0.5659	-0.0012
UEXTERM(Q)	0.0470	0.0171	-0.0031	-1.9804	-0.8639	-0.0006	0.4036	0.0958	-0.0031	-2.3133	-0.5073	-0.0022
UEXTERM(H)	1.8795	0.7198	0.0015	-1.2801	-0.5801	-0.0014	3.4126	0.8456	0.0014	-1.1966	-0.2846	-0.0027
UEXTERM(Y)	2.0946	0.9493	0.0076	-0.7517	-0.3870	-0.0020	3.2953	0.9734	0.0048	-1.0553	-0.3294	-0.0025
DY(M)	0.5464	1.4143	0.0044	0.8841	1.9472	0.0127	0.9645	1.1780	0.0039	2.5589	2.8877	0.0321
DY(Q)	0.6426	1.8244	0.0297	0.7080	1.5854	0.0289	1.2097	1.5942	0.0314	1.8131	2.2657	0.0525
DY(H)	0.4527	1.5473	0.0350	0.4053	1.1289	0.0214	0.6967	1.2585	0.0236	0.9470	1.5610	0.0324
DY(Y)	0.2286	1.1326	0.0199	0.2850	1.1716	0.0256	0.2882	0.8403	0.0077	0.5201	1.4183	0.0222
UEXDY(Q)	18.1246	1.8917	0.0216	18.7088	1.5109	0.0180	29.5487	1.6516	0.0163	49.4656	2.1416	0.0360
UEXDY(H)	17.0768	1.9103	0.0358	14.2257	1.2186	0.0186	26.7679	1.7041	0.0253	35.8389	1.8347	0.0334
UEXDY(Y)	11.4543	1.6543	0.0340	9.6006	1.0689	0.0178	16.6815	1.5011	0.0202	22.3566	1.6764	0.0270
EMR(-1)/VMR(-1)	-0.1939	-1.5247	0.0622	-0.1157	-0.5197	0.0037	-0.3271	-1.3095	0.0520	-0.3574	-0.8495	0.0140
EMR(-2)/VMR(-2)	-0.0390	-0.4353	0.0035	-0.0605	-0.4464	0.0023	-0.0569	-0.3873	0.0011	-0.1405	-0.5347	0.0046
EMR(-3)/VMR(-3)	-0.0112	-0.1899	-0.0019	-0.0561	-0.7203	0.0088	-0.0120	-0.1286	-0.0027	-0.1370	-0.8901	0.0157
EMR(-4)/VMR(-4)	-0.0114	-0.3969	-0.0005	-0.0523	-1.1614	0.0196	-0.0183	-0.3913	-0.0011	-0.1347	-1.4265	0.0368
EMR(-5)/VMR(-5)	-0.0099	-0.5707	0.0014	-0.0251	-0.8275	0.0077	-0.0182	-0.6188	0.0014	-0.0794	-1.3187	0.0255
EMR(-6)/VMR(-6)	-0.0086	-0.6665	0.0033	-0.0236	-1.1247	0.0144	-0.0106	-0.4651	-0.0003	-0.0507	-1.1926	0.0184
EMR(-7)/VMR(-7)	-0.0091	-0.8946	0.0087	-0.0086	-0.5119	0.0009	-0.0160	-0.9133	0.0078	-0.0277	-0.8649	0.0079
EMR(-8)/VMR(-8)	-0.0099	-1.1168	0.0195	-0.0103	-0.7338	0.0070	-0.0178	-1.1744	0.0186	-0.0342	-1.3853	0.0265
EMR(-9)/VMR(-9)	-0.0019	-0.3315	-0.0016	-0.0026	-0.2615	-0.0020	-0.0015	-0.1502	-0.0028	-0.0132	-0.7611	0.0047
EMR(-10)/VMR(-10)	-0.0019	-0.4975	0.0000	0.0020	0.2546	-0.0019	-0.0007	-0.0981	-0.0030	-0.0025	-0.1665	-0.0027

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Momentum portfolios are formed with share returns in past 12 months, and hold for 24 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$ . Panel B reports corresponding parameters for equal-(value-) weighted Winner/Loser return R<sub>wml</sub>.

		Panel	A 3 months	s' holding leng	gth			
	Ν	lometnum co	efficient $\gamma_{mor}$	n		Winner/Los	er return R <sub>wml</sub>	
	Equal-w	eighted	Value-v	weighted	Equal-we	eighted	Value-v	veighted
Lag length	3			2	2			3
Adj-R <sup>2</sup>	0.12	28	0.0	813	0.09	72	0.1	064
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP (Y)	-0.0042	-1.9358	-0.0065	-2.0271	-0.0068	-2.1333	-0.0129	-2.5043
UEXTBILL(Y)	9.6828	2.9657	21.1592	2.8077	16.6478	2.9570	36.4822	3.0549
DY(0)	0.1618	2.0106	0.2241	1.7694	0.2729	1.8105	0.3927	1.7035
CPI(H)			-0.1727	-2.2408			-0.3463	-2.6895
EMR(-1)	-0.0332	-1.4480	0.1727	2.2100	-0.0417	-0.9352	0.0100	2.0075
		Pane	B 6 months	s' holding leng	gth			
	М	ometnum coe	efficient $\gamma_{mom}$	1		Winner/Lose	er return $R_{wml}$	
	Equal-we	ighted	Value-we	eighted	Equal-we	ighted	Value-w	veighted
Lag length	5		5		5		4	5
Adj-R <sup>2</sup>	0.175	53	0.21	69	0.157	'4	0.24	405
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP (Y)	-0.0098	-2.2378	-0.0113	-1.9985	-0.0148	-2.9137	-0.0186	-2.1934
UEXTBILL(Y)	11.0404	1.8311	41.7750	2.9912	18.6212	1.8318	75.0935	3.4077
DY(Q)	0.3220	2.7186	0.5015	2.3244	0.5484	2.3767	0.8348	2.2404
CPI(H)			-0.4046	-2.7780			-0.7718	-3.1972
EMR(-1)	-0.0680	-1.4979			-0.0933	-1.2439		
			G 10 1					
	м	Panel	C 12 month	s' holding leng	gth	W/:		
	Equal wa	omethum co	$\gamma_{mon}$	n vaiahtad	Equal wa	winner/Los	Volue V	vaiabtad
T 1 41	Equal-we	eignied	v alue-w	eignied	Equal-we	righted	value-v	vergnied
Lag length	8	- 1	8		1			8
Adj-K	0.18	21	0.21	157	0.16	11	0.2	296
Macroeconomic variables	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	$t(\lambda)$
GDP (Y)	-0.0203	-2.6166	-0.0185	-2.2239	-0.0336	-2.8836	-0.0259	-1.9840
UEXTBILL(Y)	18.4719	1.6721	68.8807	2.7257	29.7813	1.4934	129.5681	2.9703
DY(Q)	0.3048	1.4829	0.4517	1.2245	0.5724	1.3327	1.0602	1.5924
CPI(H)			-0.5657	-2.4349			-1.1467	-2.3865
EMR(-1)	-0.1184	-1.3526			-0.1903	-1.2424		
		Panel	D 18 month	s' holding len	gth			
	М	ometnum co	efficient $\gamma_{mon}$	n	-	Winner/Los	er return R <sub>wml</sub>	
	Equal-we	eighted	Value-w	 veighted	Equal-we	ighted	Value-v	veighted
Lag length	. 8	C	9	)	8	C	(	9
$Adj-R^2$	0.16	21	0.15	579	0.14	16	0.1	852
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP (Y)	-0.0286	-2.4460	-0.0230	-2.4644	-0.0486	-2.7199	-0.0390	-2.5309
UEXTBILL(Y)	15.3726	1.0874	63.8265	2.3120	15.3012	0.6144	126.1555	2.7390
DY(0)	0.4220	1.7590	0.5192	1.3090	0.8247	1.6452	1.4598	1.9367
CPI(H)			-0.3979	-1.7883			-0.8910	-1.7935
EMR(-1)	-0.1022	-0.9540		2.1.900	-0.1523	-0.7347		
· /								

# Table 5.21 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables (Momentum-12 months' formation length)

		Pane	el E 24 mont	ns' holding le	ngth							
Mometnum coefficient $\gamma_{mom}$ Winner/Loser return $R_{wml}$												
	Equal-w	reighted	Value-w	eighted	Equal-we	ighted	Value-w	veighted				
Lag length	9	)	9	)	8		ç	)				
Adj-R <sup>2</sup>	0.11	110	0.10	051	0.077	71	0.14	492				
Macroeconomic variables	Λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)				
GDP (Y)	-0.0225	-1.9572	-0.0164	-1.7342	-0.0285	-1.5882	-0.0333	-2.0110				
UEXTBILL(Y)	3.9579	0.2176	60.9019	1.9711	0.9943	0.0307	126.7514	2.6215				
DY(Q)	0.4851	1.9481	0.5866	1.3122	0.9225	1.7902	1.5788	2.0831				
CPI(H)			-0.3230	-1.2777			-0.8174	-1.4216				
EMR(-1)	-0.1183	-0.9867			-0.2074	-0.8660						

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return R<sub>wml</sub> on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 12 months, and hold for 3, 6, 12, 18, and 24 months, respectively. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(Q) is the change of DY in past 3 months. CPI(H) is the change of CPI in past 6 months. EMR(-1) is equal-weighted market return in past 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to E report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> with 3, 6, 12, 18, and 24 months' holding lengths, respectively.

Panel A April 1980-December 1989	
Momentum coefficient $\gamma_{mom}$ Long/short retu	rn R <sub>wml</sub>
Equal-weighted Value-weighted Equal-weighted	Value-weighted
Lag length 1 2 4	2
Adj-R <sup>2</sup> 0.4903 0.1187 0.4195	0.1204
Macroeconomic variables $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ )	$\lambda$ t( $\lambda$ )
GDP(Y) 0.0010 0.7041 -0.0009 -0.2919 0.0007 0.2095	-0.0051 -1.0015
UEXTBILL(Y) 12.3718 4.8539 15.9565 3.3182 25.8601 4.1219	27.1335 3.0228
DY(O) 0.0935 1.3249 0.1256 1.0294 0.0834 0.5124	0.0698 0.2934
CPI(H) -0.0516 -1.0303	-0.0016 -0.0164
EMR(-1) -0.0648 -5.0949 -0.1292 -3.1124	0.0010 0.0101
Panal R. January 1000 December 1000	
Momentum coefficient u	D
Figure 1 weighted  Velve weighted  Figure 1  F	.III K <sub>lms</sub>
Equal-weighted Value-weighted Equal-weighted	value-weighted
Lag length $2$ I $2$	1
Adj-K 0.2392 0.0935 0.2252	0.0534
Macroeconomic variables $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ )	$\lambda$ $t(\lambda)$
GDP(Y) -0.0158 -3.7838 -0.0176 -2.9359 -0.0245 -3.8312	-0.0205 -2.4217
UEXTBILL(Y) 19.8662 2.8928 5.3491 0.3716 24.6615 2.5612	5.8236 0.2923
DY(Q) 0.1908 2.0191 0.2235 1.3828 0.2848 1.7226	0.3781 1.3547
CPI(H) 0.0994 0.9452	0.0717 0.4707
EMR(-1) 0.0508 1.1045 0.1424 2.0861	
Panel C January 2000-December 2008	
Momentum coefficient $\gamma_{mom}$ Long/short retu	ırn R <sub>lms</sub>
Equal-weighted Value-weighted Equal-weighted	Value-weighted
Lag length 2 4 2	5
Adj-R <sup>2</sup> 0.1498 0.2079 0.0919	0.2814
Macroeconomic variables $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ ) $\lambda$ t( $\lambda$ )	$\lambda$ t( $\lambda$ )
GDP(Y) -0.0261 -1.9537 -0.0140 -0.7578 -0.0345 -1.5171	-0.0286 -0.7922
UEXTBILL(Y) -34.3459 -1.7826 -14.5247 -0.3152 -59.8930 -1.2483 -	16.6304 -0.2141
DY(Q) 0.0998 0.6013 0.0309 0.1211 0.2384 0.7080	0.1663 0.3527
CPI(H) -0.3452 -2.3722	-0.7064 -3.5273
EMR(-1) 0.0278 0.5979 0.0430 0.4110	
Panel D April 1980-December 2008	D
Momentum coefficient $\gamma_{mom}$ Long/short retu Equal-weighted Value weighted Equal weighted	III K <sub>lms</sub> Value-weighted
Leg length 2 2 2 Equal-weighted	
Lag length     5     2     2 $Adi-R^2$ 0.1228     0.0813     0.0972	о 0 1064
0.0015 0.0012	0.1004
Macroeconomic $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$	$\lambda = t(\lambda)$
Macroeconomic $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$ GDP(Y)-0.0042-1.9358-0.0065-2.0271-0.0068-2.1333	λ t(λ) -0.0129 -2.5043

#### Table 5.22 Multivariate regression of momentum effect on the combination of laggedmacroeconomic variables in subperiods (Momentum 12/3)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return  $R_{wml}$  on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 12 months, and hold for 3 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(Q) is the change of DY in past 3 months. CPI(H) is the change of CPI in past 6 months. EMR(-1) is equal-weighted market return in past 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

1.7694

-2.2408

0.2729

-0.0417

1.8105

-0.9352

0.3927

-0.3463

1.7035

-2.6895

0.2241

-0.1727

0.1618

-0.0332

2.0106

-1.4480

DY(Q)

CPI(H)

EMR(-1)

				perious (n		12/0)		
		Panel	<u>A July 1980-</u>	December 19	89	T (1 )	. D	
	N.	Iomentum co	efficient $\gamma_{mom}$		<b>F</b> 1	Long/short i	return R <sub>wml</sub>	• • • •
	Equal-we	eighted	Value-we	eighted	Equal-we	ighted	Value-w	eighted
Lag length	2		3		4		2	
Adj-R <sup>2</sup>	0.52	24	0.21	78	0.446	52	0.22	210
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP(Y)	-0.0008	-0.2866	-0.0005	-0.1148	-0.0036	-0.5546	-0.0059	-0.8136
UEXTBILL(Y)	19.7516	5.1568	30.6954	4.1912	37.2772	4.1377	48.2459	4.0154
DY(Q)	0.2708	2.3740	0.1413	0.9484	0.4396	1.6477	0.2752	0.7970
CPI(H)			-0.0945	-1.1304			0.0105	0.0744
EMR(-1)	-0.0844	-3.9728			-0.1647	-2.9579		
	010011	017720			011017	20000		
		Panel B	January 1990	)-December 1	999			
	N	Iomentum co	efficient $\gamma_{max}$			Long/short	return R <sub>1ma</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-we	ighted	Value-w	eighted
Lag length	24444	Jigiitea	3	engineea	3	iginea	3	eiginea
$A di R^2$	0.30	03	0.18	83	0.220	3	0.08	272
Auj-K Maaraaaanamia yariahlaa	1	+())	1	+())	0.225	+())	2	+())
	۸ ۵.0251	ι(λ) 2.0511	λ	ι(λ) 2 0710	λ	ι(λ) 2 21 1 0	λ	( <i>L</i> )
GDP(Y)	-0.0251	-3.0544	-0.0317	-3.0719	-0.0356	-3.3118	-0.0332	-2.2186
UEXTBILL(Y)	11.4618	1.0923	-6.7839	-0.3092	9.6617	0.6608	-1.3927	-0.0386
DY(Q)	0.2835	2.0342	0.5031	2.2167	0.3538	1.8374	0.6059	2.0754
CPI(H)			0.1945	1.0526			0.1116	0.3916
EMR(-1)	-0.0116	-0.1479			0.0919	0.8578		
		Panel C	January 2000	)-December 2	2008			
	Μ	omentum coe	efficient $\gamma_{mom}$			Long/short	return R <sub>lms</sub>	
	Equal-we	ighted	Value-we	eighted	Equal-we	ighted	Value-w	eighted
Lag length	4		4		5		6	
Adj-R <sup>2</sup>	0.22	14	0.54	10	0.189	95	0.61	52
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
GDP(Y)	-0.0585	-2.4802	-0.0308	-1.4501	-0.0880	-2.2294	-0.0613	-1.3904
UEXTBILL(Y)	-26 0910	-1.0190	88 2552	1 4754	-2.0367	-0.0427	191.0603	1 8091
DY(0)	0.2967	1 2790	0 3799	1 3373	0.7958	1 7280	0.9086	1 7957
CPI(H)	0.2907	1.2790	-0.8136	-3 7452	0.7750	1.7200	-1 5944	-5 2715
FMR(-1)	0.0523	0 5506	-0.8150	-3.7432	0.0543	0 3264	-1.5744	-5.2715
	0.0323	0.5500			0.0545	0.3204		
		Panel	D July 1980-1	December 200	18			
	М	omentum cor	efficient v	beechiber 200	,0	Long/short	return R.	
	Equal-we	ighted	Value-we	vighted	Equal-we	ighted	Value-w	eighted
Laglongth	Equal-we	ignieu	v alue-we	righteu	Equal-we	igineu	v alue-w	eignieu
Lag length $A^{1}$ : $D^{2}$	0.17		3	<b>C</b> 0	0 157		3	05
Adj-R <sup>-</sup>	0.173	53	0.21	59	0.157	4	0.24	-05
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	$t(\lambda)$
GDP(Y)	-0.0098	-2.2378	-0.0113	-1.9985	-0.0148	-2.9137	-0.0186	-2.1934
UEXTBILL(Y)	11.0404	1.8311	41.7750	2.9912	18.6212	1.8318	75.0935	3.4077
DY(Q)	0.3220	2.7186	0.5015	2.3244	0.5484	2.3767	0.8348	2.2404
CPI(H)			-0.4046	-2.7780			-0.7718	-3.1972
EMR(-1)	0.0500	-1 4979			-0.0933	-1.2439		
	-0.0680	1.1777						
Note: This table presen	-0.0680 its slope coe	efficient $\lambda$ .	t-stats, and	adjusted R	$r^2$ for multiv	variate regr	ession of r	nomentum
Note: This table presen coefficient $\gamma_{mom}$ , and Wi	-0.0680 its slope coo nner/Loser r	efficient $\lambda$ , eturn $R_{wm}$	t-stats, and on the comb	adjusted R ination of la	<sup>2</sup> for multivagged macro	variate regr economic v	ession of r variables fro	nomentum m January
Note: This table presen coefficient $\gamma_{mom}$ , and Wi 1980 to December 2008.	-0.0680 its slope coo nner/Loser r Momentum	efficient $\lambda$ , eturn $R_{wml}$ of portfolios at	t-stats, and on the comb re formed w	adjusted R ination of la ith share retu	<sup>2</sup> for multivagged macro	variate regr economic v 2 months, a	ression of r variables fro and hold for	nomentum m January 6 months.
Note: This table presen coefficient $\gamma_{mom}$ , and Wi 1980 to December 2008. Momentum coefficient $\gamma_{r}$	-0.0680 its slope coo nner/Loser r Momentum nom is estima	efficient $\lambda$ , eturn $R_{wml}$ of portfolios at ted by mont	t-stats, and on the comb re formed with thy-by-mont	adjusted R ination of la ith share retu thly cross-se	t <sup>2</sup> for multivagged macro urns in past 1 actional regre	variate regreeconomic variate regreeconomic value va Value value v	ression of r variables fro and hold for rtfolios' equ	nomentum m January 6 months. al-(value-)

# Table 5.23 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 12/6)

weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(Q) is the change of DY in past 3 months. CPI(H) is the change of CPI in past 6 months. EMR(-1) is equal-weighted market return in past 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

Panel A January 1981-December 1989									
	м	Panel A	January 198	-December 1	.989	Long/short	roturn P		
	Equal wa	ighted	Volue w	nighted	Equal we	ighted	Value v	aightad	
Lag longth	Equal-we	ignicu	value-we	ligiticu	Equal-we	ligiticu	value-w	cigilica	
$Ad; \mathbf{P}^2$	0.294	51	0 13	66	0.25	41	0.24	27	
Auj-K	0.58.	1(2)	0.15	00 +())	0.554	+1	0.20	1007	
CDD(V)	λ	ι(λ)	λ	ι(λ)	٨	ι(λ)	λ	ι(λ)	
GDP(1)	-0.0128	-1.9113	-0.0025	-0.3440	-0.0300	-2.2834	-0.0077	-0.5876	
UEXTBILL(Y)	25.9150	2.9540	35.7892	1.8411	49.8927	2.9455	53.1343	1.5763	
DY(Q)	0.7010	2.3325	-0.0986	-0.4041	1.4067	2.2154	0.2415	0.5987	
CPI(H)			0.0645	0.4968			0.5835	1.6462	
EMR(-1)	-0.0604	-1.0746			-0.1578	-1.3777			
		Panel R	Ianuary 100(	)-December 1	999				
	М	omentum co	efficient v	-December 1	,,,,,	Long/short	return R.		
	Foual-we	ighted	Value-we	pighted	Equal-we	ighted	Value-w	reighted	
Lag length	24001 WC	iginea	7 unde we	ligitted	2quui we	ighted		eiginea	
$\Delta di R^2$	0.373	32	0.26	31	0.29	56	0.17	713	
Macroeconomic variables	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	$t(\lambda)$	λ	$t(\lambda)$	
GDP(Y)	0.0344	2 4702	0.0447	3 3180	0.0505	3 0400	0.0420	2 1712	
UEXTRILL(Y)	10.0511	-2.4792	-0.0447	-3.3189	-0.0303	-3.0409	-0.0429 53 7418	-2.1712	
DY(0)	0.2622	1 2246	0.6051	1 5200	9.8222	0.4049	1.0670	1.3283	
CPI(H)	0.2022	1.2340	0.0031	0.6249	0.3320	0.9958	0.1680	1.8107	
FMR(-1)	0 1566	1.0640	0.1388	0.0348	0.0458	0.2618	-0.1080	-0.4708	
	-0.1500	-1.0040			-0.0438	-0.2018			
		Panel C	January 2000	)-December 2	2008				
	М	omentum co	efficient $\gamma_{mom}$			Long/short	return R <sub>lms</sub>		
	Equal-we	ighted	Value-we	eighted	Equal-we	eighted	Value-w	reighted	
Lag length	7		7		7		8		
Adj-R <sup>2</sup>	0.130	)3	0.53	48	0.082	21	0.59	923	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0743	-1.9155	-0.0351	-0.9664	-0.0852	-1.1354	-0.0487	-0.7557	
UEXTBILL(Y)	-56.2767	-1.3326	161.3283	2.3506	-90.1294	-1.2185	387.9281	3.2166	
DY(Q)	0.1045	0.2670	0.2846	0.6672	0.4726	0.5451	1.2139	1.6873	
CPI(H)			-1.1905	-4.0484			-2.6110	-5.8173	
EMR(-1)	0.0767	0.4221			0.0020	0.0056			
		Danal F	100110rg 1081	December 2	008				
	м	omentum co	officient v	-Determoter 2	008	Long/short	roturn D		
	Equal-we	ighted	Value-we	eighted	Equal-we	zighted	Value-w	reighted	
Lag length	8	-Billea	8	-Billea	7		8	e-Billea	
Adi-R <sup>2</sup>	0.182	21	0.21	57	0.16	11	0.22	296	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0203	-2.6166	-0.0185	-2.2239	-0.0336	-2.8836	-0.0259	-1.9840	
UEXTBILL(Y)	18.4719	1.6721	68.8807	2.7257	29.7813	1.4934	129,5681	2.9703	
DY(0)	0.3048	1.4829	0.4517	1.2245	0.5724	1.3327	1.0602	1.5924	
CPI(H)	0.2010	1.1022	-0.5657	-2.4349	0.0721	1.0021	-1.1467	-2.3865	
EMR(-1)	-0.1184	-1.3526			-0.1903	-1.2424			
Note: This table present	ts slope coe	efficient $\lambda$	t-stats, and	adjusted R	$r^2$ for multi-	variate reg	ression of 1	nomentum	
coefficient y <sub>max</sub> and Wir	nner/Loser re	eturn R <sub>wm1</sub>	on the comb	ination of la	agged macro	economic v	ariables fro	m January	
coefficient mom, and with									

# Table 5.24 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 12/12)

roter time table procession of prostation in the prostation of the prostation prostation of the prostation of the prostation of the prostation of the prostation prostation of the prostation prostation of the prostation of the prostation of the prostation prostation of the prostation of the prostation of the prostation prostation of the prostation of the prostation of the prostation of the prostation prostation of the prostation prostation of the prostation prostation of the prostation of the prostation prostation of the prostation prostation of the prostation of the prostation prostation of the prostation of the prostation prostation prostation prostation of the prostation prostation of the prostation prostation of the prostation prostatio

macroeconomic variables in subperiods (Niomentum 12/18)												
		Panel	A July 1981-	December 19	89	- /-						
	М	omentum co	befficient $\gamma_{mom}$		- ·	Long/short	return R <sub>wml</sub>					
	Equal-we	ighted	Value-w	eighted	Equal-we	eighted	Value-w	reighted				
Lag length	3		6		4		1					
Adj-R <sup>2</sup>	0.257	70	0.10	23	0.284	48	0.23	384				
Macroeconomic variables	λ	$t(\lambda)$	λ	t(λ)	λ	t(λ)	λ	$t(\lambda)$				
GDP(Y)	-0.0255	-2.3739	-0.0122	-1.4035	-0.0525	-2.5721	-0.0307	-1.9392				
UEXTBILL(Y)	14.8199	1.1248	14.6500	0.6084	23.1169	0.9344	36.5113	0.8316				
DY(Q)	1.0638	2.2854	-0.0098	-0.0304	2.0363	2.2028	0.4303	0.6667				
CPI(H)			0.2238	1.3910			0.8627	1.6325				
EMR(-1)	0.0794	0.7045			0.0515	0.2234						
		Den al D	I 100	0 D 1	000							
	м	Panel B	January 199	0-December 1	999	T	nation D					
Equal-weighted Value-weighted Equal-weighted Value-weighted Equal-weighted Equal-weighted Value-weighted Value												
Lag length $7$ $5$ $6$ $2$												
Lag length     7     5     6     2 $Adi-R^2$ 0.2876     0.2996     0.2474     0.1615												
Adj-K <sup>-</sup> $0.28/6$ $0.2996$ $0.24/4$ $0.1615$ Macroeconomic variables $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$ $\lambda$ $t(\lambda)$												
CDD(V)	λ	ι(λ) 1.0026	λ	ι(λ) 2.20.52	λ	ι(λ) 2.2100	λ	ι(λ) 0. <b>5</b> 0.15				
GDP(1)	-0.0364	-1.9036	-0.0702	-3.3863	-0.0512	-2.3189	-0.0151	-0.5946				
UEXTBILL(Y)	9.1630	0.5155	-52.2884	-1.8113	3.6660	0.1523	117.1984	2.7403				
DY(Q)	0.4257	1.5360	0.8793	2.8348	0.6213	1.6053	1.3144	2.1557				
CPI(H)			0.5956	2.2216			-0.4590	-1.1931				
EMR(-1)	-0.2090	-1.0058			-0.1154	-0.5048						
		Panel C	January 200	0-December 2	2008							
	М	omentum co	efficient v <sub>mom</sub>			Long/short	return R <sub>lma</sub>					
	Equal-we	ighted	Value-w	eighted	Equal-we	eighted	Value-w	reighted				
Lag length	7	C	5	C	. 8	C	8	5				
Adi-R <sup>2</sup>	0.055	56	0.54	.06	0.03	19	0.49	925				
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)				
GDP(Y)	-0.0660	-1 6427	-0.0364	-0 7014	-0.0706	-0 7776	-0.0736	-1 0487				
UEXTBILL(Y)	-29 8846	-0 4472	462 9111	3 3498	-65 6622	-0 5194	502 8608	3 8529				
DY(0)	-0.0099	-0.0208	1 2912	1 7458	0 5004	0.4709	2 1607	2 4730				
CPI(H)	0.0077	0.0200	-2.3613	-6 4620	0.5001	0.1702	-2.4604	-6 1061				
EMR(-1)	0.0479	0.2564	2.3013	0.1020	-0.0503	-0.1067	2.1001	0.1001				
	0.0.77	012001			010000	011007						
		Panel	D July 1981-	December 200	)8							
	М	omentum co	efficient $\gamma_{mom}$			Long/short	return R <sub>lms</sub>					
	Equal-we	ighted	Value-w	eighted	Equal-we	eighted	Value-w	reighted				
Lag length	8		9		8		9	)				
Adj-R <sup>2</sup>	0.162	21	0.15	79	0.14	16	0.18	352				
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)				
GDP(Y)	-0.0286	-2.4460	-0.0230	-2.4644	-0.0486	-2.7199	-0.0390	-2.5309				
UEXTBILL(Y)	15.3726	1.0874	63.8265	2.3120	15.3012	0.6144	126.1555	2.7390				
DY(Q)	0.4220	1.7590	0.5192	1.3090	0.8247	1.6452	1.4598	1.9367				
CPI(H)			-0.3979	-1.7883			-0.8910	-1.7935				
EMR(-1)	-0.1022	-0.9540			-0.1523	-0.7347						
Note: This table present	ts slope coe	fficient λ,	t-stats, and	adjusted R	<sup>2</sup> for multiv	variate regi	ression of 1	nomentum				
coefficient $\gamma_{mom}$ , and Wir	nner/Loser re	eturn R <sub>wml</sub>	on the comb	ination of la	agged macro	economic v	ariables fro	m January				
1980 to December 2008. I	Momentum p	ortfolios a	re formed wi	th share retu	rns in past 12	2 months, a	nd hold for	18 months.				
Momentum coetticient M	10 Actimat	ad by mon	min hu mon	INIV CROSS SO	cuonal reare	seion of no	TTOULOS' AGU	( ALLEVALIA )				

# Table 5.25 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 12/18)

1980 to December 2008. Momentum portfolios are formed with share returns in past 12 months, and hold for 18 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return  $R_{wml}$  is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(Q) is the change of DY in past 3 months. CPI(H) is the change of CPI in past 6 months. EMR(-1) is equal-weighted market return in past 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$  in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

Panel A January 1982-December 1989											
	Ν	Iomentum c	$\alpha$ oefficient $\gamma_{mor}$	n		Long/short	t return R <sub>wml</sub>				
	Equal-w	reighted	Value-w	eighted	Equal-we	eighted	Value-w	reighted			
Lag length	4		5	e	4	e	7				
Adi-R <sup>2</sup>	0.04	587	0.00	)61	0.06	27	0.18	396			
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	 t(λ)	λ	t(λ)			
GDP(Y)	-0.0145	-1 5145	-0.0044	-0 5873	-0.0218	-1.0883	-0.0268	-1 4989			
UEXTBILLG(Y)	-6 5877	-0.3187	1 2262	0.0520	0.7289	0.0175	27 9906	0.6297			
DY(0)	1 2138	2 4816	0.2045	0.6499	2 3366	2 0976	0.7575	1 0029			
CPI(H)	1.2156	2.4010	0.2045	0.0499	2.5500	2.0770	0.7373	1 2904			
EMR(-1)	0.0682	0.6361	0.2100	0.7572	-0.0358	-0.1620	0.9747	1.2704			
	0.0082	0.0301			-0.0558	-0.1029					
		Panel	B January 19	90-December	1999						
	N	Iomentum c	oefficient $\gamma_{mon}$	n		Long/short	t return R <sub>lms</sub>				
	Equal-w	eighted	Value-we	eighted	Equal-we	ighted	Value-weighted				
Lag length	8		4		8		2				
Adj-R <sup>2</sup>	0.31	83	0.202	24	0.281	17	0.16	580			
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)			
GDP(Y)	-0.0393	-2.2924	-0.0137	-0.6780	-0.0563	-2.8806	0.0018	0.0787			
UEXTBILLG(Y)	6.0338	0.3805	96.5451	2.7664	-8.5234	-0.4138	135.8729	3.3175			
DY(Q)	0.3191	1.2949	9 0.1771 0.3486		0.5730	1.7761	1.0960	1.4884			
CPI(H)			-0.1035	-0.3620			-0.4148	-1.0360			
EMR(-1)	-0.2277	-1.0550			-0.1192	-0.5012					
		Panel	C January 20	00-December	2008						
	N	Iomentum c	oefficient $\gamma_{mon}$	n		Long/short	t return R <sub>lms</sub>				
	N Equal-w	Iomentum co eighted	oefficient γ <sub>mon</sub> Value-we	n eighted	Equal-we	Long/short ighted	t return R <sub>lms</sub> Value-w	reighted			
Lag length	N Equal-we 7	fomentum co eighted	oefficient $\gamma_{mon}$ Value-we 7	n eighted	Equal-we 7	Long/short ighted	t return R <sub>lms</sub> Value-w 7	reighted			
Lag length Adj-R <sup>2</sup>	N Equal-we 7 0.06	fomentum co eighted 668	oefficient γ <sub>mon</sub> Value-we 7 0.389	n bighted	Equal-we 7 0.018	Long/short ighted	t return R <sub>lms</sub> Value-w 7 0.43	eighted			
Lag length Adj-R <sup>2</sup> Macroeconomic variables	Ν Equal-we 7 0.06 λ	fomentum co eighted 68 $t(\lambda)$	oefficient $\gamma_{mon}$ Value-we 7 0.389 $\lambda$	$\frac{1}{2}$ sighted 92 $t(\lambda)$	Equal-we 7 0.018 λ	Long/short ighted 35 t(λ)	t return R <sub>lms</sub> Value-w 7 0.43 λ	reighted 320 t(λ)			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y)	Μ Equal-we 7 0.06 λ -0.0877	formentum consistent of the formatting formatting for the formatting formatting for the formatting formatting for the formatting formatting for the formatting formatting formatting formatting for the fo	oefficient $\gamma_{mon}$ Value-we 7 0.389 $\lambda$ -0.1115	$t_{\rm rest}^{\rm h}$ sighted 92 $t(\lambda)$ -3.1314	Equal-we 7 0.018 λ -0.0853	Long/short ighted $t(\lambda)$ -0.9576	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255	reighted 320 $t(\lambda)$ -1.9521			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y)	Μ Equal-wa 7 0.06 λ -0.0877 16.5707	fomentum ce eighted 68 t(λ) -2.5308 0.2118	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \end{array}$	$t(\lambda)$ -3.1314 -3.4032	Equal-we 7 0.018 λ -0.0853 -8.0204	Long/short ighted 35 t(λ) -0.9576 -0.0521	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799	reighted 320 t(λ) -1.9521 4.0463			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q)	Μ Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061	fomentum ce eighted 68 t(λ) -2.5308 0.2118 0.2376	$\begin{matrix} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \end{matrix}$	<sup>n</sup> pighted 92 t(λ) -3.1314 3.4032 1.6760	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008	reighted 320 t(λ) -1.9521 4.0463 2.9312			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H)	Ν Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061	fomentum ce eighted 68 t(λ) -2.5308 0.2118 0.2376	$\begin{matrix} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \end{matrix}$	<sup>a</sup> pighted 92 t(λ) -3.1314 3.4032 1.6760 -3.4945	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216	t return R <sub>Ims</sub> Value-w 7 0.43 λ -0.1255 608.4799 2.3008 -2.4943	reighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1)	Ν Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675	fomentum co eighted t(λ) -2.5308 0.2118 0.2376 0.3322	coefficient γ <sub>mon</sub> Value-we 7 0.389 λ -0.1115 290.2160 0.7283 -0.9629	<sup>2</sup> <sup>2</sup> <sup>2</sup> <sup>2</sup> <sup>1</sup> (λ) <sup>-3.1314</sup> <sup>3.4032</sup> <sup>1.6760</sup> <sup>-3.4945</sup>	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069 -0.0649	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231	t return R <sub>Ims</sub> Value-w 7 0.43 λ -0.1255 608.4799 2.3008 -2.4943	reighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1)	Ν Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675	fomentum ce eighted 68 t(λ) -2.5308 0.2118 0.2376 0.3322	coefficient γ <sub>mon</sub> Value-we 7 0.389 λ -0.1115 290.2160 0.7283 -0.9629	$t(\lambda)$ -3.1314 3.4032 1.6760 -3.4945	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069 -0.0649	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231	t return R <sub>Ims</sub> Value-w 7 0.43 λ -0.1255 608.4799 2.3008 -2.4943	eighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1)	Ν Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675	Iomentum ce           eighted           68 $t(\lambda)$ -2.5308           0.2118           0.2376           0.3322           Panel	coefficient γ <sub>mon</sub> Value-we 7 0.389 λ -0.1115 290.2160 0.7283 -0.9629 D January 198	<ul> <li>bighted</li> <li>92 <ul> <li>t(λ)</li> <li>-3.1314</li> <li>3.4032</li> <li>1.6760</li> <li>-3.4945</li> </ul> </li> <li>82-December</li> </ul>	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069 -0.0649 2008	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231	t return R <sub>Ims</sub> Value-w 7 0.43 λ -0.1255 608.4799 2.3008 -2.4943	eighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1)	M Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675	fomentum cr eighted $t(\lambda)$ -2.5308 0.2118 0.2376 0.3322 Panel fomentum cr	coefficient γ <sub>mon</sub> Value-we 7 0.389 λ -0.1115 290.2160 0.7283 -0.9629 D January 198 coefficient γ <sub>mon</sub>	$(\lambda)$ $(\lambda)$ -3.1314 3.4032 1.6760 -3.4945 82-December	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069 -0.0649 2008	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$	eighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1)	M Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675 M Equal-wa	1000000000000000000000000000000000000	coefficient $γ_{mon}$ Value-we 7 0.389 λ -0.1115 290.2160 0.7283 -0.9629 D January 198 coefficient $γ_{mon}$ Value-we	$(\lambda)$ $(\lambda)$ -3.1314 3.4032 1.6760 -3.4945 82-December a sighted	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w	reighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563 reighted			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1)	M Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675 M Equal-wa 9 0.11	fomentum co eighted (68) (1) (2.5308) (0.2118) (0.2376) (0.3322) Panel fomentum co eighted	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \\ D \text{ January } 198 \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.100 $	$\frac{1}{2}$ by the d $\frac{1}{2}$ $t(\lambda)$ -3.1314 3.4032 1.6760 -3.4945 <b>82-December</b> $\frac{1}{2}$ by the d	Equal-we 7 0.018 λ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w	reighted 320 t(λ) -1.9521 4.0463 2.9312 -4.8563 reighted			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1) Lag length Adj-R <sup>2</sup>	M Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675 M Equal-wa 9 0.11	fomentum cr eighted (68) (1) (2.5308) (0.2118) (0.2376) (0.3322) Panel fomentum cr eighted (10)	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \\ \hline \\ \text{D January 198} \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.100 \\ \gamma_{\text{mon}} \\ \gamma$	$(\lambda)$ $(\lambda)$ -3.1314 3.4032 1.6760 -3.4945 <b>82-December</b> $(\lambda)$	Equal-we 7 0.018 $\lambda$ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w 9 0.14	reighted $t(\lambda)$ -1.9521 4.0463 2.9312 -4.8563 reighted 192			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1) Lag length Adj-R <sup>2</sup> Macroeconomic variables	M Equal-wa 7 0.06 λ -0.0877 16.5707 0.1061 0.0675 M Equal-wa 9 0.11 λ 2 0.0222	fomentum cr eighted (68) (-2.5308) (0.2118) (0.2376) (0.3322) Panel fomentum cr eighted 10 $t(\lambda)$	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \\ D \text{ January } 199 \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.100 \\ \lambda \\ 0.0110 \\ \end{array}$	$^{h}$ pighted $^{92}$ t( $\lambda$ ) -3.1314 3.4032 1.6760 -3.4945 $^{h}$ pighted 51 t( $\lambda$ ) 1.5762 $^{h}$ t( $\lambda$ )	Equal-we 7 0.018 $\lambda$ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077 $\lambda$	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted 71 t(λ)	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w 9 0.14 $\lambda$	reighted $t(\lambda)$ -1.9521 4.0463 2.9312 -4.8563 reighted $t(\lambda)$ $t(\lambda)$			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1) Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y)	M Equal-wa 7 0.06 $\lambda$ -0.0877 16.5707 0.1061 0.0675 M Equal-wa 9 0.11 $\lambda$ -0.0225 -0.0225	Iomentum cr         eighted         68 $t(\lambda)$ -2.5308         0.2118         0.2376         0.3322         Panel         Iomentum cr         eighted         10 $t(\lambda)$ -1.9572         -2.5308	Defficient $γ_{mon}$ Value-we           7           0.389 $\lambda$ -0.1115           290.2160           0.7283           -0.9629           D January 198           0efficient $γ_{mon}$ Value-we           9           0.10: $\lambda$ -0.0164	$^{a}$ yighted $^{b}$ $^{c}$ $t(\lambda)$ $^{c}$ $^{c}$	Equal-we 7 0.018 $\lambda$ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077 $\lambda$ -0.0285	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted 71 t(λ) -1.5882	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w 9 0.14 $\lambda$ -0.0333	reighted $t(\lambda)$ -1.9521 4.0463 2.9312 -4.8563 reighted $t(\lambda)$ -2.0110			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1) Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y)	M Equal-way 7 0.06 $\lambda$ -0.0877 16.5707 0.1061 0.0675 M Equal-way 9 0.11 $\lambda$ -0.0225 3.9579	Iomentum cr           eighted           68 $t(\lambda)$ -2.5308           0.2118           0.2376           0.3322           Panel           Iomentum cr           eighted           10 $t(\lambda)$ -1.9572           0.2176	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \begin{array}{c} \text{D January 198} \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.103 \\ \lambda \\ -0.0164 \\ 60.9019 \\ \hline \end{array}$	$^{a}$ pighted $22$ t( $\lambda$ ) -3.1314 3.4032 1.6760 -3.4945 32-December $^{a}$ pighted $51$ t( $\lambda$ ) -1.7342 1.9711	Equal-we 7 0.018 $\lambda$ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077 $\lambda$ -0.0285 0.9943	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted 71 t(λ) -1.5882 0.0307	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w 9 0.14 $\lambda$ -0.0333 126.7514	reighted $t(\lambda)$ -1.9521 4.0463 2.9312 -4.8563 reighted $t(\lambda)$ -2.0110 2.6215			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1) Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q)	M Equal-wa 7 0.06 $\lambda$ -0.0877 16.5707 0.1061 0.0675 M Equal-wa 9 0.111 $\lambda$ -0.0225 3.9579 0.4851	Iomentum cr           eighted           68 $t(\lambda)$ -2.5308           0.2118           0.2376           0.3322           Panel           Iomentum cr           eighted           10 $t(\lambda)$ -1.9572           0.2176           1.9481	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \\ \hline \\ D \text{ January 199} \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.102 \\ \lambda \\ -0.0164 \\ 60.9019 \\ 0.5866 \\ \hline \end{array}$	$\lambda_{\rm prighted}^{\rm prighted}$ $\gamma_{\rm prighted}^{\rm p2}$ $t(\lambda)$ -3.1314 3.4032 1.6760 -3.4945 $\lambda_{\rm p2}^{\rm p2}$ $\lambda_{\rm p2}^{\rm p2}$ $\lambda_{\rm p2}^{\rm p2}$ $\lambda_{\rm p2}^{\rm p2}$ $\lambda_{\rm p2}^{\rm p2}$ $t(\lambda)$ -1.7342 1.9711 1.3122	Equal-we 7 0.018 $\lambda$ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077 $\lambda$ -0.0285 0.9943 0.9225	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted 71 t(λ) -1.5882 0.0307 1.7902	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w 9 0.14 $\lambda$ -0.0333 126.7514 1.5788	reighted $t(\lambda)$ -1.9521 4.0463 2.9312 -4.8563 reighted 492 $t(\lambda)$ -2.0110 2.6215 2.0831			
Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H) EMR(-1) Lag length Adj-R <sup>2</sup> Macroeconomic variables GDP(Y) UEXTBILLG(Y) DY(Q) CPI(H)	M Equal-way 7 0.06 $\lambda$ -0.0877 16.5707 0.1061 0.0675 M Equal-way 9 0.11 $\lambda$ -0.0225 3.9579 0.4851	Image: Image for the system         Image for the system <thi< td=""><td><math display="block">\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \\ \hline \\ D \text{ January } 198 \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.100 \\ \lambda \\ -0.0164 \\ 60.9019 \\ 0.5866 \\ -0.3230 \\ \hline \end{array}</math></td><td><math display="block">\frac{1}{2}</math> bighted <math display="block">\frac{92}{t(\lambda)}</math> -3.1314 3.4032 1.6760 -3.4945 <math display="block">\frac{82-\text{December}}{t(\lambda)}</math> bighted <math display="block">51</math> t(<math>\lambda</math>) -1.7342 1.9711 1.3122 -1.2777</td><td>Equal-we 7 0.018 <math>\lambda</math> -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077 <math>\lambda</math> -0.0285 0.9943 0.9225</td><td>Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted 71 t(λ) -1.5882 0.0307 1.7902</td><td>t return <math>R_{lms}</math> Value-w 7 0.43 <math>\lambda</math> -0.1255 608.4799 2.3008 -2.4943 t return <math>R_{lms}</math> Value-w 9 0.14 <math>\lambda</math> -0.0333 126.7514 1.5788 -0.8174</td><td>reighted <math>t(\lambda)</math> -1.9521 4.0463 2.9312 -4.8563 reighted <math>t(\lambda)</math> -2.0110 2.6215 2.0831 -1.4216</td></thi<>	$\begin{array}{c} \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 7 \\ 0.389 \\ \lambda \\ -0.1115 \\ 290.2160 \\ 0.7283 \\ -0.9629 \\ \hline \\ \hline \\ D \text{ January } 198 \\ \text{oefficient } \gamma_{\text{mon}} \\ \text{Value-we} \\ 9 \\ 0.100 \\ \lambda \\ -0.0164 \\ 60.9019 \\ 0.5866 \\ -0.3230 \\ \hline \end{array}$	$\frac{1}{2}$ bighted $\frac{92}{t(\lambda)}$ -3.1314 3.4032 1.6760 -3.4945 $\frac{82-\text{December}}{t(\lambda)}$ bighted $51$ t( $\lambda$ ) -1.7342 1.9711 1.3122 -1.2777	Equal-we 7 0.018 $\lambda$ -0.0853 -8.0204 0.6069 -0.0649 2008 Equal-we 8 0.077 $\lambda$ -0.0285 0.9943 0.9225	Long/short ighted 35 t(λ) -0.9576 -0.0521 0.6216 -0.1231 Long/short ighted 71 t(λ) -1.5882 0.0307 1.7902	t return $R_{lms}$ Value-w 7 0.43 $\lambda$ -0.1255 608.4799 2.3008 -2.4943 t return $R_{lms}$ Value-w 9 0.14 $\lambda$ -0.0333 126.7514 1.5788 -0.8174	reighted $t(\lambda)$ -1.9521 4.0463 2.9312 -4.8563 reighted $t(\lambda)$ -2.0110 2.6215 2.0831 -1.4216			

# Table 5.26 Multivariate regression of momentum effect on the combination of lagged macroeconomic variables in subperiods (Momentum 12/24)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of momentum coefficient  $\gamma_{mom}$ , and Winner/Loser return R<sub>wml</sub> on the combination of lagged macroeconomic variables from January 1980 to December 2008. Momentum portfolios are formed with share returns in past 12 months, and hold for 24 months. Momentum coefficient  $\gamma_{mom}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their formation returns. Winner/Loser return R<sub>wml</sub> is the return spread in 'Winner' portfolio and 'Loser' portfolio. GDP(Y) is the change of GDP in past 12 months. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. DY(Q) is the change of DY in past 3 months. CPI(H) is the change of CPI in past 6 months. EMR(-1) is equal-weighted market return in past 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return R<sub>wml</sub> in the 1980s, the 1990s, from 2000 to 2008, and the whole sample period, respectively.

		Panel A S	ize coefficient $\gamma_s$	ize				Pan	el B Small/Larg	ge return R <sub>sml</sub>		
	Panel	A1 Equal-we	eighted	Panel A	2 Value-wei	ghted	Panel B	1 Equal-weig	ghted	Panel B	2 Value-wei	ghted
Macroeconomic		•										_
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	0.0006	0.9040	0.0042	0.0005	0.8634	0.0040	0.0037	0.7425	0.0023	0.0034	0.7082	0.0021
UEXGDP(Y)	0.0008	0.7581	0.0028	0.0007	0.7383	0.0027	0.0054	0.6783	0.0018	0.0052	0.6766	0.0021
IP(M)	0.4048	3.0311	0.0142	0.3217	2.6506	0.0092	3.2028	3.0725	0.0152	2.4730	2.4573	0.0096
IP(Q)	0.3099	1.9601	0.0179	0.2384	1.5788	0.0109	2.7322	2.0892	0.0244	2.0708	1.6796	0.0153
IP(H)	0.1282	1.0511	0.0047	0.0802	0.6911	0.0004	1.4600	1.4126	0.0139	0.9945	0.9996	0.0061
IP(Y)	0.0057	0.0775	-0.0029	-0.0094	-0.1296	-0.0028	0.2231	0.3549	-0.0020	0.0689	0.1035	-0.0028
UEXIP(Q)	0.0026	2.6227	0.0297	0.0020	2.2058	0.0193	0.0218	2.6871	0.0351	0.0165	2.2130	0.0222
UEXIP(H)	0.0016	1.6811	0.0160	0.0011	1.2559	0.0077	0.0154	1.9251	0.0259	0.0109	1.4691	0.0140
UEXIP(Y)	0.0007	0.9454	0.0045	0.0004	0.6089	0.0002	0.0075	1.1778	0.0108	0.0047	0.7724	0.0033
IPM(M)	0.2735	1.8878	0.0053	0.1927	1.4382	0.0017	2.3775	2.1347	0.0076	1.6285	1.5485	0.0028
IPM(Q)	0.1573	1.0421	0.0033	0.0955	0.6440	-0.0004	1.3632	1.1320	0.0049	0.7461	0.6153	-0.0002
IPM(H)	0.0231	0.1939	-0.0026	-0.0129	-0.1143	-0.0028	0.3690	0.3766	-0.0016	-0.0098	-0.0100	-0.0029
IPM(Y)	-0.0441	-0.7941	0.0001	-0.0552	-1.0094	0.0023	-0.2963	-0.6555	-0.0006	-0.4260	-0.8889	0.0025
UEXIPM(Q)	0.0017	1.7073	0.0116	0.0012	1.2497	0.0049	0.0140	1.7835	0.0144	0.0092	1.2381	0.0057
UEXIPM(H)	0.0007	0.7141	0.0007	0.0003	0.3303	-0.0021	0.0063	0.8367	0.0028	0.0028	0.3744	-0.0016
UEXIPM(Y)	0.0001	0.1392	-0.0028	-0.0001	-0.2020	-0.0026	0.0014	0.2646	-0.0023	-0.0009	-0.1621	-0.0026
M0 (M)	0.2583	0.6058	-0.0018	0.1925	0.4609	-0.0022	2.4479	0.7972	-0.0013	1.7852	0.5957	-0.0019
M0 (Q)	0.1270	0.3404	-0.0021	0.1019	0.2915	-0.0023	1.0479	0.3695	-0.0019	1.0208	0.3959	-0.0018
M0 (H)	0.2694	1.1599	0.0052	0.2320	1.1503	0.0039	2.0753	1.0608	0.0053	1.7385	1.0731	0.0037
M0 (Y)	-0.0789	-0.4776	-0.0012	-0.0791	-0.5279	-0.0010	-0.5868	-0.4725	-0.0013	-0.5314	-0.4768	-0.0014
UEXM0(Q)	0.0000	-0.0472	-0.0029	0.0000	-0.0879	-0.0028	0.0000	-0.1325	-0.0028	0.0000	-0.1315	-0.0028
UEXM0 (H)	0.0000	0.0675	-0.0029	0.0000	0.0455	-0.0029	0.0000	-0.0468	-0.0029	0.0000	-0.0304	-0.0029
UEXM0 (Y)	0.0000	-0.1524	-0.0027	0.0000	-0.1867	-0.0026	0.0000	-0.2978	-0.0019	0.0000	-0.3222	-0.0020
UNE (M)	0.0683	0.4310	-0.0019	0.0945	0.6039	-0.0006	0.4121	0.3190	-0.0023	0.6316	0.4750	-0.0011
UNE (Q)	0.0551	0.7927	0.0013	0.0691	1.0336	0.0045	0.3967	0.6996	0.0008	0.5427	0.9595	0.0051
UNE (H)	0.0228	0.6328	-0.0005	0.0313	0.9077	0.0022	0.1662	0.5680	-0.0007	0.2627	0.9082	0.0034
UNE (Y)	0.0252	1.3525	0.0074	0.0276	1.5706	0.0111	0.2116	1.5006	0.0095	0.2404	1.7134	0.0156
UEXUNE(Q)	0.0046	0.7808	0.0014	0.0059	1.0410	0.0048	0.0321	0.6556	0.0006	0.0453	0.9537	0.0051
UEXUNE(H)	0.0026	0.7685	0.0009	0.0034	1.0446	0.0041	0.0186	0.6567	0.0003	0.0272	0.9911	0.0050
UEXUNE(Y)	0.0019	1.0389	0.0034	0.0023	1.2990	0.0070	0.0147	1.0026	0.0032	0.0189	1.3242	0.0088
CPI (M)	-0.0134	-0.8656	-0.0007	-0.0122	-0.8173	-0.0008	-0.0993	-0.9506	-0.0008	-0.0827	-0.8013	-0.0012
CPI (Q)	-0.0163	-1.4368	0.0095	-0.0146	-1.4482	0.0082	-0.1206	-1.5729	0.0086	-0.1051	-1.6497	0.0072
CPI (H)	-0.0153	-1.6576	0.0240	-0.0130	-1.5370	0.0188	-0.1174	-1.7570	0.0239	-0.0958	-1.6832	0.0178
CPI (Y)	-0.0146	-2.3151	0.0542	-0.0123	-2.3067	0.0426	-0.1087	-2.2628	0.0505	-0.0873	-2.2204	0.0370
UEXCPI(Q)	-0.0022	-1.8152	0.0093	-0.0019	-1.7235	0.0078	-0.0138	-1.6079	0.0054	-0.0109	-1.1192	0.0031

 Table 5.27 Univariate regression of size effect on individual lagged macroeconomic variables (Size-3)

Continued												
UEXCPI(H)	-0.0023	-2.2490	0.0238	-0.0022	-2.3353	0.0246	-0.0157	-2.2414	0.0187	-0.0150	-2.2091	0.0203
UEXCPI(Y)	-0.0019	-2.8702	0.0438	-0.0018	-2.9657	0.0435	-0.0138	-2.9885	0.0383	-0.0131	-3.0699	0.0402
TBILL(M)	-0.0361	-3.0302	0.0185	-0.0342	-2.8513	0.0187	-0.2581	-3.0093	0.0156	-0.2520	-2.7698	0.0176
TBILL (Q)	-0.0286	-3.4594	0.0435	-0.0260	-3.1783	0.0400	-0.2116	-3.7865	0.0400	-0.1959	-3.6894	0.0398
TBILL (H)	-0.0186	-3.0725	0.0387	-0.0165	-2.8767	0.0339	-0.1347	-3.1995	0.0340	-0.1173	-2.8923	0.0296
TBILL (Y)	-0.0215	-4.8180	0.1169	-0.0188	-4.5328	0.0995	-0.1586	-4.3967	0.1073	-0.1341	-4.2755	0.0885
UEXTBILL(Q)	-4.8231	-3.1896	0.0385	-4.4679	-2.9540	0.0370	-36.5712	-3.6439	0.0374	-34.8512	-3.3954	0.0396
UEXTBILL(H)	-3.7968	-3.0351	0.0415	-3.5061	-2.8683	0.0396	-28.6622	-3.4592	0.0399	-27.1160	-3.3265	0.0416
UEXTBILL(Y)	-3.8986	-4.4020	0.0856	-3.5530	-4.2089	0.0796	-28.8310	-4.6259	0.0790	-26.1355	-4.3807	0.0752
TERM(M)	0.2792	1.8288	0.0060	0.2675	1.7324	0.0063	2.0251	1.9057	0.0050	1.9773	1.7405	0.0059
TERM(Q)	0.3210	2.4299	0.0309	0.2955	2.3189	0.0292	2.5983	2.7688	0.0345	2.4244	2.6322	0.0350
TERM(H)	0.1901	2.0023	0.0219	0.1717	1.9516	0.0198	1.4775	2.0944	0.0224	1.3547	2.0765	0.0218
TERM(Y)	0.2174	3.1593	0.0641	0.1970	3.2694	0.0588	1.4958	2.7583	0.0507	1.3458	3.0117	0.0475
UEXTERM(Q)	0.3974	2.3320	0.0239	0.3694	2.2140	0.0231	3.1212	2.5800	0.0251	2.9318	2.3696	0.0258
UEXTERM(H)	0.3427	2.2433	0.0304	0.3124	2.1516	0.0282	2.7511	2.5306	0.0334	2.5510	2.4860	0.0333
UEXTERM(Y)	0.3243	2.8544	0.0527	0.2967	2.9654	0.0493	2.3464	2.8153	0.0463	2.1573	2.8815	0.0454
DY(M)	0.0047	0.1845	-0.0028	0.0082	0.3290	-0.0025	-0.0189	-0.0961	-0.0029	0.0135	0.0704	-0.0029
DY(Q)	0.0070	0.3255	-0.0021	0.0055	0.2644	-0.0023	0.0305	0.1790	-0.0026	0.0051	0.0327	-0.0029
DY(H)	-0.0046	-0.2862	-0.0020	-0.0086	-0.5897	0.0004	-0.0097	-0.0707	-0.0029	-0.0596	-0.5334	-0.0001
DY(Y)	-0.0045	-0.4194	-0.0010	-0.0055	-0.5415	0.0003	-0.0189	-0.2366	-0.0023	-0.0310	-0.4405	-0.0011
UEXDY(Q)	-0.2634	-0.3873	-0.0018	-0.2357	-0.3420	-0.0019	-3.0170	-0.5830	-0.0004	-3.0291	-0.6092	0.0000
UEXDY(H)	-0.4147	-0.7603	0.0021	-0.4685	-0.8654	0.0043	-3.4286	-0.7602	0.0029	-4.3297	-1.0309	0.0078
UEXDY(Y)	-0.4022	-1.0786	0.0070	-0.4299	-1.2260	0.0099	-2.8111	-0.9422	0.0053	-3.2864	-1.2429	0.0101
EMR(-1)/VMR(-1)	0.0086	1.8760	0.0266	0.0073	0.8605	0.0057	0.0911	2.6316	0.0529	0.0954	1.6050	0.0228
EMR(-2)/VMR(-2)	0.0029	0.8893	0.0054	0.0052	0.9248	0.0096	0.0308	1.2315	0.0128	0.0449	1.1761	0.0133
EMR(-3)/VMR(-3)	0.0010	0.3966	-0.0008	0.0037	1.1059	0.0131	0.0167	0.8989	0.0070	0.0349	1.5086	0.0218
EMR(-4)/VMR(-4)	0.0007	0.3938	-0.0007	0.0026	1.2286	0.0147	0.0102	0.7755	0.0050	0.0220	1.5034	0.0187
EMR(-5)/VMR(-5)	0.0001	0.0837	-0.0028	0.0018	1.1978	0.0139	0.0042	0.4841	0.0002	0.0145	1.3947	0.0164
EMR(-6)/VMR(-6)	0.0003	0.4310	-0.0009	0.0014	1.4511	0.0173	0.0047	0.8169	0.0042	0.0117	1.6955	0.0218
EMR(-7)/VMR(-7)	0.0003	0.5794	0.0007	0.0010	1.2003	0.0160	0.0043	0.9416	0.0074	0.0083	1.4028	0.0193
EMR(-8)/VMR(-8)	0.0003	0.7533	0.0019	0.0006	1.0599	0.0094	0.0037	1.1770	0.0095	0.0051	1.2803	0.0115
EMR(-9)/VMR(-9)	0.0003	0.8139	0.0038	0.0004	0.8695	0.0064	0.0031	1.2716	0.0132	0.0039	1.1512	0.0120
EMR(-10)/VMR(-10)	0.0003	1.2663	0.0155	0.0003	0.6700	0.0029	0.0028	1.4784	0.0218	0.0017	0.6788	0.0017

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of size coefficient  $\gamma_{mv}$  and Small/Large return R<sub>sml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Size decile portfolios are formed with shares' market values at the end of each month, and hold for 3 months. Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values. Small/Large return R<sub>sml</sub> is the return spread between 'Small' portfolio and 'Large' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$ . Panel B reports corresponding parameters for equal-(value-) weighted Small/Large return R<sub>sml</sub>.

	Panel A Size coefficient $\gamma_{size}$							Panel B Small/Large return R <sub>sml</sub>					
	Panel	A1 Equal-we	righted	Panel A	2 Value-wei	ghted	Panel H	B1 Equal-weig	ghted	Panel E	32 Value-wei	ghted	
Macroeconomic	1 41101	iii Equal iii			<u> </u>	Biiled		1 0				0	
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	0.0012	1.0654	0.0090	0.0010	0.8907	0.0072	0.0089	1.0276	0.0086	0.0070	0.7668	0.0056	
UEXGDP(Y)	0.0013	0.7394	0.0033	0.0011	0.6306	0.0024	0.0100	0.7342	0.0033	0.0079	0.5464	0.0018	
IP(M)	0.3080	1.7404	0.0009	0.1934	1.1700	-0.0012	2.8881	1.9996	0.0028	1.6801	1.2472	-0.0006	
IP(Q)	0.2242	0.9785	0.0012	0.1072	0.4923	-0.0018	2.9113	1.5225	0.0090	1.6439	0.8562	0.0016	
IP(H)	0.0900	0.4463	-0.0015	0.0010	0.0050	-0.0029	1.4046	0.8275	0.0031	0.3486	0.2057	-0.0025	
IP(Y)	-0.0276	-0.2077	-0.0026	-0.0619	-0.4860	-0.0011	0.1038	0.0987	-0.0029	-0.3106	-0.2884	-0.0021	
UEXIP(Q)	0.0017	1.2461	0.0025	0.0009	0.7185	-0.0012	0.0194	1.6914	0.0085	0.0109	0.9994	0.0014	
UEXIP(H)	0.0010	0.6722	-0.0002	0.0002	0.1675	-0.0028	0.0143	1.1597	0.0066	0.0057	0.4806	-0.0011	
UEXIP(Y)	0.0002	0.1374	-0.0028	-0.0003	-0.2557	-0.0024	0.0050	0.5089	-0.0007	-0.0002	-0.0226	-0.0029	
IPM(M)	0.1473	0.7280	-0.0020	0.0394	0.2071	-0.0029	1.4131	0.8727	-0.0015	0.2288	0.1458	-0.0029	
IPM(Q)	0.0276	0.1067	-0.0029	-0.0639	-0.2619	-0.0025	0.7732	0.3748	-0.0020	-0.2814	-0.1400	-0.0028	
IPM(H)	-0.0510	-0.2445	-0.0024	-0.1236	-0.6387	0.0009	-0.1104	-0.0658	-0.0029	-1.0319	-0.6523	0.0017	
IPM(Y)	-0.1115	-1.0524	0.0045	-0.1305	-1.3116	0.0088	-0.7338	-0.9390	0.0025	-0.9890	-1.3222	0.0090	
UEXIPM(Q)	0.0004	0.2882	-0.0026	-0.0002	-0.1444	-0.0028	0.0066	0.5355	-0.0015	-0.0004	-0.0366	-0.0029	
UEXIPM(H)	-0.0002	-0.1588	-0.0027	-0.0008	-0.5769	-0.0003	0.0013	0.1036	-0.0028	-0.0057	-0.4811	-0.0008	
UEXIPM(Y)	-0.0007	-0.7011	0.0009	-0.0010	-1.0579	0.0058	-0.0040	-0.4878	-0.0010	-0.0078	-0.9963	0.0060	
M0 (M)	1.0687	1.5613	0.0044	0.8652	1.4134	0.0026	8.2310	1.3876	0.0045	6.0782	1.1631	0.0019	
M0 (Q)	0.8478	1.5821	0.0122	0.7393	1.6383	0.0103	6.1356	1.2797	0.0105	5.1783	1.3435	0.0085	
M0 (H)	0.4016	0.9444	0.0042	0.3291	0.8878	0.0026	2.7938	0.7619	0.0029	2.1812	0.6957	0.0013	
M0 (Y)	-0.2633	-0.8396	0.0046	-0.2475	-0.8716	0.0047	-2.2736	-0.9684	0.0066	-1.9982	-0.9646	0.0059	
UEXM0(Q)	0.0000	0.5344	0.0002	0.0000	0.5560	0.0002	0.0000	0.3918	-0.0011	0.0000	0.4621	-0.0006	
UEXM0 (H)	0.0000	0.3995	-0.0007	0.0000	0.4420	-0.0004	0.0000	0.2452	-0.0020	0.0000	0.3320	-0.0013	
UEXM0 (Y)	0.0000	-0.1064	-0.0028	0.0000	-0.0767	-0.0029	0.0000	-0.2410	-0.0021	0.0000	-0.1563	-0.0026	
UNE (M)	0.1981	0.7449	0.0005	0.2721	1.0898	0.0044	1.1640	0.5377	-0.0010	2.0978	1.0160	0.0048	
UNE (Q)	0.0885	0.7227	0.0012	0.1249	1.0870	0.0066	0.5406	0.5531	-0.0003	1.0022	1.0790	0.0079	
UNE (H)	0.0631	0.9582	0.0044	0.0811	1.2958	0.0109	0.4613	0.9211	0.0037	0.6912	1.4380	0.0149	
UNE (Y)	0.0498	1.2512	0.0130	0.0558	1.4792	0.0200	0.4092	1.4000	0.0153	0.4846	1.7200	0.0278	
UEXUNE(Q)	0.0081	0.7881	0.0022	0.0110	1.1447	0.0078	0.0484	0.5846	0.0001	0.0852	1.0912	0.0085	
UEXUNE(H)	0.0053	0.8536	0.0032	0.0070	1.1903	0.0092	0.0346	0.7026	0.0014	0.0562	1.1987	0.0107	
UEXUNE(Y)	0.0043	1.1294	0.0090	0.0050	1.3973	0.0159	0.0313	1.1093	0.0080	0.0407	1.5034	0.0192	
CPI (M)	-0.0450	-1.4703	0.0070	-0.0379	-1.3484	0.0052	-0.3351	-1.5660	0.0064	-0.2700	-1.4874	0.0043	
CPI (Q)	-0.0397	-1.5484	0.0260	-0.0325	-1.3928	0.0192	-0.3125	-1.6749	0.0274	-0.2428	-1.5409	0.0190	
CPI (H)	-0.0321	-1.5099	0.0436	-0.0260	-1.3415	0.0322	-0.2457	-1.5394	0.0433	-0.1885	-1.3463	0.0297	
CPI (Y)	-0.0295	-2.0145	0.0889	-0.0246	-1.9452	0.0700	-0.2237	-2.1036	0.0863	-0.1767	-2.1377	0.0638	
UEXCPI(Q)	-0.0055	-2.3298	0.0273	-0.0053	-2.4702	0.0300	-0.0396	-2.2407	0.0241	-0.0399	-2.5611	0.0298	

 Table 5.28 Univariate regression of size effect on individual lagged macroeconomic variables (Size-6)

Continued												
UEXCPI(H)	-0.0051	-2.5594	0.0509	-0.0050	-2.6705	0.0557	-0.0383	-2.5497	0.0480	-0.0385	-2.7875	0.0589
UEXCPI(Y)	-0.0040	-2.9378	0.0782	-0.0038	-3.0282	0.0787	-0.0305	-3.0572	0.0763	-0.0288	-3.2834	0.0818
TBILL(M)	-0.0466	-2.0391	0.0112	-0.0420	-1.9364	0.0102	-0.3521	-2.1989	0.0107	-0.3267	-2.1351	0.0111
TBILL (Q)	-0.0409	-2.3750	0.0344	-0.0361	-2.2443	0.0305	-0.2916	-2.3856	0.0293	-0.2567	-2.2994	0.0270
TBILL (H)	-0.0457	-3.6427	0.0963	-0.0401	-3.6988	0.0846	-0.3299	-3.3275	0.0846	-0.2764	-3.3157	0.0707
TBILL (Y)	-0.0372	-4.3936	0.1385	-0.0326	-4.4354	0.1214	-0.2748	-4.1630	0.1276	-0.2323	-4.1452	0.1088
UEXTBILL(Q)	-6.1579	-2.1411	0.0237	-5.6850	-2.0593	0.0231	-44.9869	-2.2500	0.0212	-42.8132	-2.1830	0.0232
UEXTBILL(H)	-6.9404	-2.8183	0.0557	-6.3974	-2.8296	0.0541	-50.2434	-2.8237	0.0491	-46.5550	-2.8628	0.0506
UEXTBILL(Y)	-7.3228	-3.7671	0.1204	-6.6211	-3.8897	0.1126	-53.8241	-3.7680	0.1099	-47.6256	-3.9726	0.1029
TERM(M)	0.4492	1.5167	0.0062	0.4193	1.4957	0.0062	3.5715	1.6586	0.0068	3.3924	1.6437	0.0076
TERM(Q)	0.4554	1.8951	0.0239	0.4130	1.9119	0.0223	3.4796	1.8817	0.0236	3.2149	1.8959	0.0242
TERM(H)	0.4166	2.3166	0.0439	0.3840	2.4687	0.0427	2.8710	2.0390	0.0348	2.6655	2.2520	0.0360
TERM(Y)	0.3263	2.4151	0.0566	0.2999	2.4238	0.0547	2.1512	2.0834	0.0409	1.9453	2.3289	0.0400
UEXTERM(Q)	0.5875	1.7828	0.0202	0.5367	1.7706	0.0192	4.6352	1.8477	0.0214	4.3174	1.8436	0.0224
UEXTERM(H)	0.5892	2.0607	0.0359	0.5381	2.1418	0.0341	4.3754	2.0028	0.0333	4.0672	2.0994	0.0346
UEXTERM(Y)	0.5732	2.5477	0.0655	0.5274	2.7128	0.0635	3.9142	2.2582	0.0511	3.5925	2.4950	0.0516
DY(M)	0.0031	0.0607	-0.0029	-0.0047	-0.1051	-0.0029	0.0857	0.1981	-0.0027	-0.0086	-0.0224	-0.0029
DY(Q)	0.0020	0.0462	-0.0029	-0.0066	-0.1759	-0.0026	0.0728	0.2002	-0.0023	-0.0202	-0.0642	-0.0029
DY(H)	-0.0193	-0.6808	0.0030	-0.0233	-0.8999	0.0070	-0.0821	-0.3618	-0.0011	-0.1274	-0.6390	0.0023
DY(Y)	-0.0140	-0.6823	0.0045	-0.0139	-0.7037	0.0054	-0.0654	-0.4372	-0.0002	-0.0593	-0.4198	-0.0003
UEXDY(Q)	-0.8359	-0.7123	0.0016	-0.9694	-0.8730	0.0040	-5.7020	-0.5807	0.0006	-7.5103	-0.7900	0.0044
UEXDY(H)	-1.0170	-1.0924	0.0089	-1.1274	-1.3087	0.0137	-6.8116	-0.8645	0.0061	-8.1969	-1.1265	0.0127
UEXDY(Y)	-0.8013	-1.1833	0.0126	-0.8285	-1.3006	0.0161	-5.0014	-0.9372	0.0073	-5.3155	-1.0917	0.0109
EMR(-1)/VMR(-1)	0.0102	1.1119	0.0131	0.0076	0.4369	0.0007	0.1084	1.6078	0.0275	0.1110	0.8623	0.0109
EMR(-2)/VMR(-2)	0.0024	0.3637	-0.0007	0.0055	0.4939	0.0027	0.0322	0.6573	0.0038	0.0493	0.6228	0.0049
EMR(-3)/VMR(-3)	0.0000	-0.0058	-0.0029	0.0049	0.7216	0.0084	0.0144	0.3907	-0.0001	0.0499	0.9869	0.0175
EMR(-4)/VMR(-4)	0.0006	0.1606	-0.0024	0.0048	1.0847	0.0205	0.0117	0.4619	0.0011	0.0381	1.2485	0.0232
EMR(-5)/VMR(-5)	-0.0005	-0.2396	-0.0018	0.0023	0.7852	0.0085	0.0013	0.0799	-0.0028	0.0187	0.9350	0.0102
EMR(-6)/VMR(-6)	0.0003	0.2209	-0.0021	0.0022	1.0986	0.0165	0.0059	0.5410	0.0016	0.0178	1.2744	0.0203
EMR(-7)/VMR(-7)	0.0002	0.2143	-0.0022	0.0014	0.8019	0.0113	0.0046	0.5520	0.0018	0.0108	0.8956	0.0120
EMR(-8)/VMR(-8)	0.0005	0.5724	0.0019	0.0010	0.8038	0.0102	0.0063	0.9964	0.0108	0.0077	0.8805	0.0104
EMR(-9)/VMR(-9)	0.0004	0.5844	0.0030	0.0004	0.4396	0.0010	0.0045	0.9227	0.0103	0.0037	0.5530	0.0025
EMR(-10)/VMR(-10)	0.0006	1.0306	0.0217	0.0003	0.4138	0.0015	0.0051	1.2712	0.0285	0.0018	0.3160	-0.0008

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of size coefficient  $\gamma_{mv}$  and Small/Large return R<sub>sml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Size decile portfolios are formed with shares' market values at the end of each month, and hold for 6 months. Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values. Small/Large return R<sub>sml</sub> is the return spread between 'Small' portfolio and 'Large' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$ . Panel B reports corresponding parameters for equal-(value-) weighted Small/Large return R<sub>sml</sub>.

		Panel A Siz	ze coefficient $\gamma_{si}$	ze				Pan	el B Small/Larg	ge return R <sub>sml</sub>		
	Panel	A1 Equal-we	eighted	Panel A	2 Value-wei	ghted	Panel B	1 Equal-weig	ghted	Panel E	32 Value-wei	ghted
Macroeconomic												
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	0.0035	1.8897	0.0371	0.0029	1.6411	0.0313	0.0312	2.1453	0.0490	0.0247	1.7284	0.0394
UEXGDP(Y)	0.0041	1.4963	0.0201	0.0034	1.2778	0.0169	0.0361	1.7273	0.0268	0.0283	1.3313	0.0208
IP(M)	0.0353	0.1491	-0.0030	-0.1264	-0.5312	-0.0027	1.0906	0.5435	-0.0027	-0.8124	-0.3880	-0.0028
IP(Q)	-0.1325	-0.3880	-0.0024	-0.2891	-0.8813	0.0002	0.4133	0.1442	-0.0029	-1.5566	-0.5353	-0.0014
IP(H)	-0.0669	-0.2159	-0.0027	-0.1765	-0.6016	-0.0004	0.3603	0.1411	-0.0029	-1.0475	-0.4129	-0.0014
IP(Y)	-0.0839	-0.3799	-0.0019	-0.1517	-0.7266	0.0015	-0.2088	-0.1249	-0.0029	-1.1118	-0.7114	0.0012
UEXIP(Q)	-0.0007	-0.3503	-0.0027	-0.0017	-0.8903	-0.0005	0.0030	0.1794	-0.0029	-0.0096	-0.5642	-0.0017
UEXIP(H)	-0.0009	-0.4094	-0.0021	-0.0019	-0.8601	0.0013	0.0008	0.0417	-0.0030	-0.0108	-0.5792	-0.0004
UEXIP(Y)	-0.0010	-0.5026	-0.0011	-0.0016	-0.8777	0.0034	-0.0028	-0.1881	-0.0027	-0.0110	-0.7657	0.0023
IPM(M)	-0.1791	-0.7060	-0.0025	-0.3225	-1.2574	-0.0010	-0.7133	-0.3542	-0.0029	-2.4709	-1.1443	-0.0009
IPM(Q)	-0.5326	-1.5909	0.0073	-0.6255	-2.0034	0.0142	-3.4172	-1.3233	0.0039	-4.7711	-2.0035	0.0145
IPM(H)	-0.3997	-1.3626	0.0100	-0.4508	-1.6733	0.0171	-2.6916	-1.2265	0.0067	-3.5021	-1.8037	0.0183
IPM(Y)	-0.2958	-1.3562	0.0167	-0.3219	-1.6059	0.0254	-1.8499	-1.1379	0.0097	-2.2712	-1.5947	0.0218
UEXIPM(Q)	-0.0030	-1.4356	0.0038	-0.0037	-1.8793	0.0095	-0.0180	-1.1046	0.0010	-0.0273	-1.7662	0.0089
UEXIPM(H)	-0.0036	-1.5810	0.0124	-0.0040	-1.9336	0.0209	-0.0235	-1.3774	0.0079	-0.0304	-1.9795	0.0208
UEXIPM(Y)	-0.0029	-1.5272	0.0192	-0.0031	-1.8401	0.0292	-0.0188	-1.3666	0.0127	-0.0231	-1.9374	0.0276
M0 (M)	0.4666	0.5464	-0.0025	0.3962	0.5199	-0.0025	2.7360	0.3682	-0.0027	2.0587	0.3155	-0.0028
M0 (Q)	-0.0130	-0.0180	-0.0030	0.0533	0.0847	-0.0030	-1.4938	-0.2389	-0.0027	-0.5975	-0.1122	-0.0029
M0 (H)	-0.5184	-0.7776	0.0015	-0.4066	-0.6859	0.0003	-5.7710	-1.0432	0.0061	-4.2649	-0.8997	0.0035
M0 (Y)	-0.9231	-1.4045	0.0320	-0.6890	-1.1815	0.0207	-9.1113	-1.8221	0.0529	-6.1861	-1.4514	0.0305
UEXM0(Q)	0.0000	-0.2260	-0.0025	0.0000	-0.0759	-0.0029	-0.0001	-0.3301	-0.0018	0.0000	-0.1326	-0.0028
UEXM0 (H)	0.0000	-0.4062	-0.0008	0.0000	-0.2363	-0.0023	-0.0001	-0.5305	0.0010	0.0000	-0.3031	-0.0017
UEXM0 (Y)	0.0000	-0.6761	0.0048	0.0000	-0.4782	0.0009	-0.0001	-0.8346	0.0091	0.0000	-0.5460	0.0022
UNE (M)	0.6672	1.3571	0.0113	0.7738	1.6854	0.0204	4.5630	1.2149	0.0080	6.0926	1.7807	0.0224
UNE (Q)	0.3256	1.3486	0.0183	0.3711	1.6587	0.0306	2.3097	1.2588	0.0145	2.9913	1.8138	0.0353
UNE (H)	0.1765	1.2288	0.0185	0.2000	1.4979	0.0306	1.3026	1.1912	0.0162	1.6504	1.6675	0.0372
UNE (Y)	0.1014	1.1543	0.0219	0.1099	1.3369	0.0325	0.8414	1.2316	0.0251	0.9660	1.5587	0.0451
UEXUNE(Q)	0.0288	1.3756	0.0212	0.0323	1.6896	0.0340	0.1981	1.2390	0.0158	0.2505	1.7520	0.0361
UEXUNE(H)	0.0178	1.3109	0.0223	0.0198	1.5763	0.0351	0.1244	1.1974	0.0173	0.1549	1.6534	0.0380
UEXUNE(Y)	0.0108	1.2636	0.0257	0.0117	1.4751	0.0381	0.0806	1.2460	0.0234	0.0947	1.6032	0.0444
CPI (M)	-0.1076	-1.5306	0.0183	-0.0876	-1.3806	0.0142	-0.8083	-1.5570	0.0168	-0.6194	-1.3907	0.0121
CPI (Q)	-0.0944	-1.8924	0.0584	-0.0762	-1.6722	0.0456	-0.7205	-1.9615	0.0556	-0.5401	-1.6603	0.0399
CPI (H)	-0.0714	-1.8189	0.0837	-0.0576	-1.6027	0.0657	-0.5442	-1.8481	0.0797	-0.4086	-1.5534	0.0576
CPI (Y)	-0.0475	-1.9498	0.0863	-0.0399	-1.7901	0.0736	-0.3509	-1.8832	0.0769	-0.2796	-1.6898	0.0630
UEXCPI(Q)	-0.0113	-2.2114	0.0457	-0.0099	-2.2056	0.0423	-0.0891	-2.2362	0.0468	-0.0763	-2.3327	0.0444

Table 5.29 Univar	riate regression	of size effect on i	ndividual lagg	ed macroeconomi	c variab	les (Siz	ze-12)	
D 110'	CC			D	100	11 /7		

Continued												
UEXCPI(H)	-0.0098	-2.2126	0.0714	-0.0085	-2.2012	0.0652	-0.0778	-2.2520	0.0735	-0.0655	-2.3161	0.0676
UEXCPI(Y)	-0.0065	-2.0368	0.0767	-0.0056	-1.9871	0.0684	-0.0512	-2.0506	0.0783	-0.0422	-2.0151	0.0686
TBILL(M)	-0.1332	-2.6004	0.0401	-0.1148	-2.5975	0.0359	-1.0260	-2.4604	0.0389	-0.8642	-2.5250	0.0357
TBILL (Q)	-0.1128	-3.1146	0.1032	-0.0971	-3.0917	0.0926	-0.8567	-2.8755	0.0975	-0.7141	-2.9285	0.0878
TBILL (H)	-0.0836	-3.7712	0.1208	-0.0727	-3.6045	0.1109	-0.6218	-3.4124	0.1094	-0.5213	-3.2561	0.0998
TBILL (Y)	-0.0531	-3.2058	0.1049	-0.0473	-3.0337	0.1014	-0.3872	-2.9722	0.0912	-0.3398	-2.8184	0.0914
UEXTBILL(Q)	-17.9225	-2.9435	0.0818	-15.5997	-2.9078	0.0751	-139.3756	-2.8203	0.0811	-118.3855	-2.9010	0.0759
UEXTBILL(H)	-15.8837	-3.1006	0.1123	-13.8598	-3.0099	0.1037	-122.0271	-2.9684	0.1086	-103.2993	-2.9333	0.1010
UEXTBILL(Y)	-12.0556	-2.9416	0.1226	-10.6110	-2.8257	0.1153	-90.6167	-2.7727	0.1134	-76.9653	-2.6446	0.1062
TERM(M)	1.1143	2.4151	0.0181	1.0334	2.4775	0.0190	7.4476	1.9595	0.0124	6.7848	2.0159	0.0137
TERM(Q)	0.9819	2.2785	0.0438	0.8935	2.3242	0.0442	6.7093	1.8646	0.0329	6.0883	1.9453	0.0354
TERM(H)	0.6099	1.7728	0.0347	0.5687	1.7865	0.0369	3.7150	1.3370	0.0200	3.4863	1.3737	0.0233
TERM(Y)	0.3460	1.3002	0.0222	0.3402	1.3793	0.0266	2.0338	0.9891	0.0113	2.1048	1.1163	0.0169
UEXTERM(Q)	1.3343	2.3642	0.0418	1.2193	2.4536	0.0425	9.1718	1.9406	0.0317	8.3141	2.0275	0.0341
UEXTERM(H)	1.1058	2.1432	0.0483	1.0127	2.1674	0.0494	7.2917	1.7143	0.0336	6.6581	1.7704	0.0367
UEXTERM(Y)	0.7495	1.7791	0.0410	0.7092	1.8561	0.0449	4.5833	1.3685	0.0240	4.4034	1.4567	0.0294
DY(M)	-0.0244	-0.3643	-0.0025	-0.0149	-0.2392	-0.0028	-0.1421	-0.2678	-0.0027	-0.0375	-0.0771	-0.0030
DY(Q)	-0.0211	-0.3433	-0.0019	-0.0170	-0.2956	-0.0021	-0.0960	-0.2038	-0.0026	-0.0392	-0.0908	-0.0029
DY(H)	-0.0424	-0.8164	0.0077	-0.0382	-0.7910	0.0076	-0.2567	-0.6623	0.0035	-0.1956	-0.5568	0.0019
DY(Y)	-0.0239	-0.5854	0.0051	-0.0187	-0.4934	0.0030	-0.1462	-0.4822	0.0019	-0.0733	-0.2612	-0.0014
UEXDY(Q)	-1.0681	-0.6315	-0.0002	-0.8457	-0.5269	-0.0009	-6.9301	-0.5061	-0.0011	-4.2944	-0.3387	-0.0020
UEXDY(H)	-1.3663	-0.8666	0.0050	-1.2140	-0.8273	0.0047	-8.5380	-0.6937	0.0021	-6.4352	-0.5721	0.0008
UEXDY(Y)	-1.0558	-0.7975	0.0071	-0.8860	-0.7206	0.0057	-6.4400	-0.6362	0.0032	-4.0330	-0.4342	0.0002
EMR(-1)/VMR(-1)	0.0052	0.2943	-0.0015	0.0036	0.1150	-0.0027	0.0627	0.4426	0.0006	0.0569	0.2370	-0.0016
EMR(-2)/VMR(-2)	-0.0012	-0.0947	-0.0028	0.0098	0.4712	0.0039	0.0122	0.1219	-0.0027	0.0795	0.5108	0.0049
EMR(-3)/VMR(-3)	-0.0023	-0.2371	-0.0014	0.0086	0.7229	0.0106	-0.0013	-0.0181	-0.0030	0.0787	0.8664	0.0169
EMR(-4)/VMR(-4)	-0.0011	-0.1551	-0.0022	0.0091	1.1333	0.0305	0.0012	0.0223	-0.0030	0.0663	1.1633	0.0281
EMR(-5)/VMR(-5)	-0.0022	-0.5453	0.0041	0.0031	0.5915	0.0050	-0.0106	-0.3524	-0.0002	0.0216	0.5732	0.0039
EMR(-6)/VMR(-6)	-0.0003	-0.1131	-0.0027	0.0030	0.7947	0.0116	0.0016	0.0761	-0.0029	0.0213	0.7989	0.0099
EMR(-7)/VMR(-7)	-0.0006	-0.2732	-0.0014	0.0011	0.3368	0.0005	-0.0005	-0.0297	-0.0030	0.0073	0.3087	-0.0004
EMR(-8)/VMR(-8)	0.0007	0.3931	0.0006	0.0013	0.5509	0.0056	0.0101	0.7537	0.0097	0.0105	0.5830	0.0065
EMR(-9)/VMR(-9)	0.0007	0.5373	0.0049	0.0002	0.0835	-0.0028	0.0072	0.6843	0.0091	-0.0001	-0.0082	-0.0030
EMR(-10)/VMR(-10)	0.0010	0.9358	0.0225	0.0003	0.2101	-0.0015	0.0087	1.1192	0.0301	0.0015	0.1277	-0.0024

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of size coefficient  $\gamma_{mv}$  and Small/Large return R<sub>sml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Size decile portfolios are formed with shares' market values at the end of each month, and hold for 12 months. Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values. Small/Large return R<sub>sml</sub> is the return spread between 'Small' portfolio and 'Large' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$ . Panel B reports corresponding parameters for equal-(value-) weighted Small/Large return R<sub>sml</sub>.

	Panel A Size coefficient $\gamma_{size}$							Panel B Small/Large return R <sub>sml</sub>				
	Panel	A1 Equal-we	eighted	Panel A	A2 Value-wei	ghted	Panel F	31 Equal-weig	ghted	Panel E	32 Value-wei	ghted
Macroeconomic		•				-						
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	0.0054	2.1580	0.0552	0.0044	1.8383	0.0440	0.0489	2.4124	0.0787	0.0368	1.8775	0.0578
UEXGDP(Y)	0.0077	1.9745	0.0476	0.0062	1.6611	0.0374	0.0700	2.2758	0.0688	0.0520	1.7281	0.0489
IP(M)	0.1435	0.4052	-0.0029	-0.0152	-0.0457	-0.0030	1.6276	0.5754	-0.0026	-0.1075	-0.0414	-0.0030
IP(Q)	-0.0652	-0.1451	-0.0030	-0.2266	-0.5376	-0.0019	0.8682	0.2442	-0.0028	-1.0651	-0.3202	-0.0026
IP(H)	-0.0560	-0.1455	-0.0029	-0.2069	-0.5690	-0.0009	0.4184	0.1395	-0.0029	-1.5744	-0.5590	-0.0008
IP(Y)	0.0801	0.1906	-0.0024	-0.0520	-0.1402	-0.0027	1.1291	0.3580	-0.0009	-0.6019	-0.2296	-0.0023
UEXIP(Q)	-0.0004	-0.1436	-0.0030	-0.0014	-0.5341	-0.0020	0.0035	0.1550	-0.0030	-0.0085	-0.4141	-0.0024
UEXIP(H)	-0.0010	-0.3385	-0.0024	-0.0021	-0.7380	0.0002	-0.0007	-0.0316	-0.0030	-0.0142	-0.6548	-0.0002
UEXIP(Y)	-0.0004	-0.1377	-0.0028	-0.0014	-0.5186	-0.0001	0.0014	0.0641	-0.0030	-0.0115	-0.5889	0.0007
IPM(M)	-0.1756	-0.4487	-0.0027	-0.3078	-0.8360	-0.0019	-0.9651	-0.3239	-0.0029	-2.4761	-0.9050	-0.0017
IPM(Q)	-0.5850	-1.1512	0.0045	-0.6541	-1.4085	0.0086	-3.7630	-0.9772	0.0023	-4.7109	-1.4072	0.0079
IPM(H)	-0.5464	-1.2118	0.0118	-0.6148	-1.5098	0.0201	-3.4932	-1.0356	0.0074	-4.5818	-1.6038	0.0204
IPM(Y)	-0.3321	-0.9307	0.0122	-0.3880	-1.1877	0.0225	-2.0302	-0.7706	0.0067	-2.8753	-1.2317	0.0226
UEXIPM(Q)	-0.0035	-1.0912	0.0024	-0.0040	-1.3720	0.0060	-0.0225	-0.9372	0.0009	-0.0294	-1.3891	0.0058
UEXIPM(H)	-0.0046	-1.3448	0.0125	-0.0050	-1.6225	0.0199	-0.0298	-1.1669	0.0081	-0.0366	-1.6708	0.0191
UEXIPM(Y)	-0.0036	-1.2375	0.0188	-0.0040	-1.5091	0.0299	-0.0236	-1.0826	0.0127	-0.0299	-1.5864	0.0300
M0 (M)	-0.5880	-0.4759	-0.0025	-0.4126	-0.3726	-0.0027	-8.2585	-0.7977	-0.0013	-6.2714	-0.6993	-0.0017
M0 (Q)	-1.1712	-0.9683	0.0036	-0.7977	-0.7407	0.0007	-12.9279	-1.3148	0.0108	-8.3519	-0.9818	0.0045
M0 (H)	-1.5479	-1.2652	0.0212	-1.0720	-1.0056	0.0113	-16.4341	-1.6954	0.0438	-10.5922	-1.3159	0.0224
M0 (Y)	-1.4289	-1.4977	0.0482	-1.0249	-1.1948	0.0294	-14.4027	-1.9881	0.0862	-9.6097	-1.5126	0.0489
UEXM0(Q)	0.0000	-0.5300	0.0006	0.0000	-0.2796	-0.0021	-0.0002	-0.7221	0.0042	-0.0001	-0.3682	-0.0012
UEXM0 (H)	0.0000	-0.6489	0.0055	0.0000	-0.3741	-0.0004	-0.0002	-0.8665	0.0128	-0.0001	-0.4652	0.0012
UEXM0 (Y)	0.0000	-0.7767	0.0127	0.0000	-0.5037	0.0033	-0.0001	-1.0049	0.0236	-0.0001	-0.6010	0.0060
UNE (M)	0.9039	1.1970	0.0130	1.0284	1.4866	0.0224	7.0358	1.2238	0.0136	8.9393	1.7463	0.0321
UNE (Q)	0.4331	1.1558	0.0199	0.4842	1.4154	0.0322	3.5308	1.2357	0.0231	4.3489	1.7202	0.0488
UNE (H)	0.2434	1.0822	0.0219	0.2681	1.3024	0.0342	2.1012	1.2248	0.0288	2.4923	1.6476	0.0557
UNE (Y)	0.1399	1.0649	0.0260	0.1483	1.2125	0.0370	1.3311	1.3357	0.0419	1.4644	1.6291	0.0682
UEXUNE(Q)	0.0405	1.2665	0.0263	0.0445	1.5283	0.0404	0.3155	1.2986	0.0274	0.3793	1.7712	0.0546
UEXUNE(H)	0.0258	1.2347	0.0295	0.0280	1.4679	0.0440	0.2069	1.3107	0.0329	0.2429	1.7448	0.0618
UEXUNE(Y)	0.0160	1.2484	0.0357	0.0169	1.4397	0.0504	0.1357	1.4140	0.0448	0.1521	1.7775	0.0756
CPI (M)	-0.1224	-1.4165	0.0138	-0.1011	-1.2894	0.0110	-0.9190	-1.4125	0.0132	-0.7451	-1.2826	0.0109
CPI (Q)	-0.0960	-1.7781	0.0357	-0.0802	-1.5893	0.0303	-0.7088	-1.7562	0.0332	-0.5771	-1.5450	0.0284
CPI (H)	-0.0683	-1.8115	0.0457	-0.0577	-1.6234	0.0397	-0.4851	-1.7012	0.0390	-0.3984	-1.5000	0.0340
CPI (Y)	-0.0452	-1.9390	0.0464	-0.0399	-1.7947	0.0442	-0.3133	-1.7638	0.0376	-0.2729	-1.6131	0.0374
UEXCPI(Q)	-0.0130	-1.9466	0.0362	-0.0114	-1.9221	0.0343	-0.1019	-1.9123	0.0385	-0.0898	-1.9679	0.0392

 Table 5.30 Univariate regression of size effect on individual lagged macroeconomic variables (Size-18)

Continueu												
UEXCPI(H)	-0.0105	-1.9514	0.0494	-0.0093	-1.9392	0.0474	-0.0823	-1.9441	0.0518	-0.0725	-1.9995	0.0526
UEXCPI(Y)	-0.0068	-1.8215	0.0498	-0.0059	-1.7511	0.0461	-0.0525	-1.8389	0.0516	-0.0448	-1.7561	0.0490
TBILL(M)	-0.1182	-2.2666	0.0177	-0.1069	-2.2801	0.0178	-0.8489	-2.0151	0.0153	-0.7817	-2.1012	0.0173
TBILL (Q)	-0.0940	-2.5113	0.0420	-0.0854	-2.5160	0.0427	-0.6603	-2.1734	0.0350	-0.6106	-2.2922	0.0396
TBILL (H)	-0.0750	-2.7933	0.0577	-0.0679	-2.7260	0.0582	-0.5268	-2.4348	0.0483	-0.4778	-2.4256	0.0522
TBILL (Y)	-0.0423	-1.6457	0.0383	-0.0408	-1.7064	0.0442	-0.2934	-1.4960	0.0309	-0.2976	-1.6633	0.0427
UEXTBILL(Q)	-14.9444	-2.1943	0.0331	-13.5879	-2.1896	0.0336	-108.9772	-1.9831	0.0298	-100.1866	-2.0483	0.0333
UEXTBILL(H)	-13.8318	-2.2744	0.0504	-12.6115	-2.2667	0.0516	-100.3470	-2.0689	0.0452	-92.0578	-2.1210	0.0500
UEXTBILL(Y)	-10.3289	-1.9380	0.0532	-9.5857	-1.9365	0.0565	-74.4818	-1.7929	0.0471	-69.8052	-1.8299	0.0545
TERM(M)	0.7228	1.2216	0.0024	0.7226	1.3287	0.0036	4.7994	1.0513	0.0011	5.0464	1.2301	0.0029
TERM(Q)	0.6333	1.1476	0.0089	0.6409	1.2904	0.0120	3.8925	0.8854	0.0047	4.3610	1.1384	0.0096
TERM(H)	0.4587	0.9896	0.0100	0.4970	1.1985	0.0158	2.5474	0.7032	0.0039	3.2852	1.0444	0.0120
TERM(Y)	0.1496	0.3571	-0.0002	0.2043	0.5419	0.0036	0.8183	0.2629	-0.0016	1.5681	0.5769	0.0041
UEXTERM(Q)	0.8784	1.2059	0.0088	0.8820	1.3409	0.0117	5.6428	0.9769	0.0054	6.1218	1.2080	0.0099
UEXTERM(H)	0.7690	1.1091	0.0122	0.8020	1.2895	0.0173	4.6117	0.8437	0.0063	5.4816	1.1557	0.0143
UEXTERM(Y)	0.5105	0.8411	0.0095	0.5670	1.0385	0.0159	2.9977	0.6469	0.0043	3.9725	0.9809	0.0139
DY(M)	-0.0884	-0.9229	0.0008	-0.0684	-0.7713	-0.0002	-0.6026	-0.8419	0.0000	-0.3835	-0.5923	-0.0014
DY(Q)	-0.0732	-0.7961	0.0053	-0.0623	-0.7237	0.0044	-0.4650	-0.6878	0.0027	-0.3507	-0.5646	0.0013
DY(H)	-0.0540	-0.7459	0.0076	-0.0461	-0.6770	0.0065	-0.3345	-0.6298	0.0039	-0.2432	-0.4900	0.0018
DY(Y)	-0.0428	-0.8223	0.0128	-0.0363	-0.7333	0.0109	-0.2986	-0.7893	0.0101	-0.2197	-0.6054	0.0063
UEXDY(Q)	-2.5060	-1.0199	0.0062	-2.1312	-0.9275	0.0052	-16.3885	-0.8846	0.0037	-12.5751	-0.7428	0.0022
UEXDY(H)	-2.0342	-0.8676	0.0078	-1.7612	-0.8063	0.0069	-12.8176	-0.7323	0.0043	-9.7751	-0.6057	0.0026
UEXDY(Y)	-1.5354	-0.8055	0.0101	-1.2958	-0.7264	0.0084	-10.1023	-0.7196	0.0067	-7.2014	-0.5487	0.0034
EMR(-1)/VMR(-1)	0.0006	0.0259	-0.0030	0.0052	0.1325	-0.0026	0.0288	0.1619	-0.0026	0.0398	0.1340	-0.0026
EMR(-2)/VMR(-2)	-0.0043	-0.2555	-0.0015	0.0176	0.6413	0.0106	-0.0050	-0.0390	-0.0030	0.1318	0.6459	0.0109
EMR(-3)/VMR(-3)	-0.0046	-0.3364	0.0008	0.0145	0.8526	0.0208	-0.0142	-0.1355	-0.0024	0.1116	0.8717	0.0227
EMR(-4)/VMR(-4)	-0.0039	-0.4136	0.0030	0.0127	1.1402	0.0374	-0.0191	-0.2715	-0.0006	0.0818	1.0079	0.0274
EMR(-5)/VMR(-5)	-0.0032	-0.6040	0.0063	0.0050	0.6713	0.0099	-0.0173	-0.4406	0.0016	0.0307	0.5678	0.0059
EMR(-6)/VMR(-6)	-0.0009	-0.2364	-0.0016	0.0045	0.7549	0.0168	-0.0029	-0.1001	-0.0028	0.0279	0.6550	0.0108
EMR(-7)/VMR(-7)	-0.0007	-0.2131	-0.0018	0.0018	0.3535	0.0022	0.0007	0.0282	-0.0030	0.0092	0.2501	-0.0005
EMR(-8)/VMR(-8)	0.0011	0.4003	0.0025	0.0017	0.4465	0.0052	0.0133	0.6234	0.0107	0.0107	0.3773	0.0031
EMR(-9)/VMR(-9)	0.0015	0.6644	0.0160	0.0006	0.1872	-0.0014	0.0126	0.7376	0.0204	0.0000	0.0007	-0.0030
EMR(-10)/VMR(-10)	0.0013	0.8669	0.0252	0.0004	0.1701	-0.0018	0.0120	1.0490	0.0370	0.0009	0.0518	-0.0029

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of size coefficient  $\gamma_{mv}$  and Small/Large return R<sub>sml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Size decile portfolios are formed with shares' market values at the end of each month, and hold for 18 months. Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values. Small/Large return R<sub>sml</sub> is the return spread between 'Small' portfolio and 'Large' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$ . Panel B reports corresponding parameters for equal-(value-) weighted Small/Large return R<sub>sml</sub>.

	Panel A Size coefficient $\gamma_{size}$								Panel B Small/Large return R <sub>sml</sub>					
	Panel	A1 Equal-we	eighted '	Panel A	2 Value-wei	ghted	Panel E	B1 Equal-weig	ghted	Panel E	32 Value-wei	ghted		
Macroeconomic	1 41101					Bitted		1 0				0		
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>		
GDP(Y)	0.0045	1.7954	0.0255	0.0038	1.5812	0.0222	0.0366	1.7611	0.0283	0.0284	1.3951	0.0218		
UEXGDP(Y)	0.0076	2.0210	0.0324	0.0063	1.7015	0.0271	0.0656	2.1225	0.0402	0.0499	1.5796	0.0298		
IP(M)	0.3591	0.7162	-0.0023	0.0828	0.1911	-0.0031	3.5563	0.8697	-0.0018	0.3806	0.1140	-0.0031		
IP(Q)	0.5273	0.6500	0.0006	0.1470	0.2110	-0.0027	5.6396	0.8358	0.0040	1.2319	0.2215	-0.0027		
IP(H)	0.4096	0.5046	0.0018	0.0846	0.1195	-0.0028	4.3499	0.6425	0.0061	0.4383	0.0763	-0.0030		
IP(Y)	0.1060	0.1724	-0.0023	-0.0638	-0.1165	-0.0028	1.0860	0.2265	-0.0018	-1.0094	-0.2475	-0.0016		
UEXIP(Q)	0.0023	0.4969	-0.0016	0.0001	0.0299	-0.0031	0.0267	0.7070	0.0003	0.0011	0.0352	-0.0031		
UEXIP(H)	0.0020	0.3596	-0.0013	-0.0003	-0.0541	-0.0031	0.0245	0.5228	0.0013	-0.0030	-0.0773	-0.0030		
UEXIP(Y)	0.0010	0.1973	-0.0022	-0.0007	-0.1559	-0.0026	0.0127	0.3048	-0.0008	-0.0084	-0.2408	-0.0017		
IPM(M)	-0.3399	-0.6764	-0.0023	-0.5316	-1.1644	-0.0007	-2.0227	-0.5387	-0.0026	-4.3718	-1.3139	-0.0003		
IPM(Q)	-0.5717	-0.8431	0.0020	-0.7574	-1.2314	0.0079	-3.7462	-0.7470	0.0005	-6.2175	-1.4156	0.0100		
IPM(H)	-0.4466	-0.6736	0.0039	-0.6313	-1.0553	0.0142	-2.9705	-0.6117	0.0020	-5.4582	-1.3015	0.0197		
IPM(Y)	-0.3974	-0.7653	0.0124	-0.4911	-1.0334	0.0260	-2.8667	-0.7700	0.0102	-4.1236	-1.2425	0.0331		
UEXIPM(Q)	-0.0042	-0.9898	0.0024	-0.0053	-1.3915	0.0079	-0.0272	-0.8735	0.0008	-0.0420	-1.5389	0.0092		
UEXIPM(H)	-0.0047	-0.9760	0.0084	-0.0059	-1.3541	0.0191	-0.0322	-0.9141	0.0058	-0.0484	-1.5894	0.0234		
UEXIPM(Y)	-0.0039	-0.9243	0.0149	-0.0048	-1.2550	0.0303	-0.0277	-0.9034	0.0117	-0.0400	-1.5011	0.0375		
M0 (M)	-1.2913	-0.6570	-0.0014	-0.8073	-0.4741	-0.0023	-16.0640	-0.9746	0.0013	-11.2080	-0.8124	-0.0003		
M0 (Q)	-1.6605	-0.9721	0.0062	-1.1531	-0.7643	0.0024	-18.6724	-1.3639	0.0164	-13.2951	-1.1224	0.0099		
M0 (H)	-1.7687	-1.2597	0.0190	-1.2764	-1.0128	0.0110	-18.8884	-1.7554	0.0385	-13.4627	-1.4259	0.0247		
M0 (Y)	-1.7339	-1.6859	0.0498	-1.2775	-1.3427	0.0322	-16.5893	-2.1889	0.0769	-11.5138	-1.6748	0.0475		
UEXM0(Q)	0.0000	-0.5527	0.0022	0.0000	-0.2863	-0.0017	-0.0002	-0.7628	0.0072	-0.0001	-0.4183	-0.0001		
UEXM0 (H)	0.0000	-0.6362	0.0068	0.0000	-0.3613	0.0000	-0.0002	-0.8656	0.0149	-0.0001	-0.4964	0.0027		
UEXM0 (Y)	0.0000	-0.7403	0.0131	0.0000	-0.4715	0.0033	-0.0002	-0.9791	0.0234	-0.0001	-0.5942	0.0065		
UNE (M)	1.0535	1.0174	0.0122	1.2273	1.2860	0.0224	9.4541	1.2428	0.0172	11.7826	1.7401	0.0383		
UNE (Q)	0.5045	0.9971	0.0189	0.5827	1.2495	0.0331	4.6390	1.2503	0.0277	5.7203	1.7359	0.0584		
UNE (H)	0.2913	0.9828	0.0223	0.3272	1.1914	0.0364	2.7370	1.2645	0.0340	3.2454	1.6701	0.0654		
UNE (Y)	0.1973	1.1531	0.0378	0.2035	1.2803	0.0504	1.9215	1.5766	0.0609	2.0297	1.8087	0.0908		
UEXUNE(Q)	0.0519	1.2196	0.0310	0.0580	1.4869	0.0493	0.4435	1.4373	0.0380	0.5294	1.9427	0.0740		
UEXUNE(H)	0.0332	1.2082	0.0353	0.0367	1.4448	0.0545	0.2866	1.4436	0.0441	0.3363	1.8939	0.0824		
UEXUNE(Y)	0.0219	1.3167	0.0486	0.0231	1.4896	0.0678	0.1936	1.6449	0.0636	0.2132	1.9838	0.1033		
CPI (M)	-0.1267	-1.5033	0.0096	-0.1082	-1.3677	0.0083	-0.9146	-1.4857	0.0078	-0.7764	-1.3609	0.0072		
CPI (Q)	-0.0859	-1.5770	0.0188	-0.0758	-1.4434	0.0179	-0.5938	-1.4869	0.0142	-0.5251	-1.3792	0.0147		
CPI (H)	-0.0550	-1.3478	0.0191	-0.0502	-1.2737	0.0196	-0.3592	-1.1694	0.0125	-0.3346	-1.1288	0.0147		
CPI (Y)	-0.0457	-1.5034	0.0318	-0.0422	-1.4525	0.0335	-0.3134	-1.3767	0.0240	-0.2921	-1.3439	0.0279		
UEXCPI(Q)	-0.0123	-1.5640	0.0220	-0.0113	-1.5563	0.0229	-0.0992	-1.7152	0.0237	-0.0923	-1.7881	0.0274		

Table 5.31 Univariate regression of size effect on individual lagged macroeconomic variables (Size-24)

Continued												
UEXCPI(H)	-0.0096	-1.5240	0.0279	-0.0091	-1.5616	0.0308	-0.0769	-1.6731	0.0293	-0.0740	-1.8088	0.0364
UEXCPI(Y)	-0.0071	-1.5813	0.0379	-0.0065	-1.5174	0.0388	-0.0567	-1.7738	0.0400	-0.0514	-1.7066	0.0434
TBILL(M)	-0.1109	-1.8996	0.0098	-0.1060	-2.0521	0.0114	-0.8245	-1.8583	0.0087	-0.8244	-2.1950	0.0124
TBILL (Q)	-0.0760	-1.4727	0.0177	-0.0768	-1.7139	0.0230	-0.5285	-1.3119	0.0135	-0.5797	-1.7712	0.0232
TBILL (H)	-0.0484	-0.9999	0.0146	-0.0507	-1.2018	0.0209	-0.3008	-0.7770	0.0082	-0.3465	-1.0763	0.0166
TBILL (Y)	-0.0379	-0.9036	0.0201	-0.0396	-1.0719	0.0281	-0.2500	-0.7535	0.0136	-0.2844	-1.0116	0.0253
UEXTBILL(Q)	-13.9351	-1.6271	0.0192	-13.6123	-1.7643	0.0230	-104.1866	-1.5936	0.0174	-106.5719	-1.8999	0.0252
UEXTBILL(H)	-10.4623	-1.2042	0.0186	-10.6692	-1.3810	0.0246	-73.4229	-1.0790	0.0145	-79.9291	-1.3940	0.0243
UEXTBILL(Y)	-8.4587	-1.0506	0.0236	-8.6024	-1.1978	0.0308	-57.8387	-0.9111	0.0175	-61.8599	-1.1417	0.0279
TERM(M)	0.6967	1.0708	0.0005	0.7388	1.2557	0.0019	5.0791	1.0537	0.0000	6.0000	1.3989	0.0027
TERM(Q)	0.6180	0.9765	0.0050	0.6876	1.2344	0.0092	4.2297	0.8917	0.0031	5.4418	1.3610	0.0105
TERM(H)	0.1088	0.1612	-0.0026	0.2483	0.4252	0.0002	-0.0404	-0.0076	-0.0031	1.6637	0.3817	-0.0005
TERM(Y)	0.0413	0.0681	-0.0029	0.1176	0.2196	-0.0015	0.3066	0.0655	-0.0030	1.1673	0.2959	-0.0004
UEXTERM(Q)	0.8545	1.0491	0.0049	0.9326	1.2948	0.0086	6.1015	1.0020	0.0036	7.5337	1.4508	0.0104
UEXTERM(H)	0.4866	0.5511	0.0012	0.6386	0.8277	0.0061	2.8426	0.4216	-0.0007	4.9693	0.8874	0.0067
UEXTERM(Y)	0.1806	0.2045	-0.0020	0.3250	0.4192	0.0013	0.7831	0.1149	-0.0028	2.5498	0.4467	0.0017
DY(M)	-0.0799	-0.7090	-0.0009	-0.0650	-0.6127	-0.0013	-0.6136	-0.6929	-0.0009	-0.4723	-0.5713	-0.0014
DY(Q)	-0.0937	-0.8851	0.0066	-0.0849	-0.8491	0.0067	-0.6426	-0.7911	0.0045	-0.5723	-0.7472	0.0048
DY(H)	-0.0901	-1.1103	0.0179	-0.0808	-1.0457	0.0177	-0.6429	-1.0403	0.0146	-0.5465	-0.9358	0.0137
DY(Y)	-0.0789	-1.3495	0.0350	-0.0673	-1.1788	0.0310	-0.5827	-1.3549	0.0313	-0.4454	-1.0446	0.0233
UEXDY(Q)	-2.8897	-0.9076	0.0056	-2.5490	-0.8595	0.0053	-21.1867	-0.8579	0.0047	-18.1732	-0.7956	0.0044
UEXDY(H)	-3.1165	-1.0664	0.0150	-2.8014	-1.0202	0.0149	-23.0287	-1.0338	0.0132	-19.7955	-0.9486	0.0127
UEXDY(Y)	-2.8433	-1.2524	0.0288	-2.4461	-1.1258	0.0260	-21.6251	-1.2820	0.0274	-16.9631	-1.0528	0.0216
EMR(-1)/VMR(-1)	-0.0011	-0.0407	-0.0031	0.0170	0.3364	0.0000	0.0393	0.1973	-0.0025	0.1440	0.3847	0.0009
EMR(-2)/VMR(-2)	-0.0050	-0.2445	-0.0016	0.0269	0.7755	0.0197	0.0024	0.0154	-0.0031	0.2024	0.7938	0.0196
EMR(-3)/VMR(-3)	-0.0067	-0.3846	0.0028	0.0232	1.0138	0.0403	-0.0191	-0.1427	-0.0023	0.1693	0.9983	0.0378
EMR(-4)/VMR(-4)	-0.0068	-0.5931	0.0099	0.0150	1.0574	0.0368	-0.0371	-0.4325	0.0033	0.0951	0.9223	0.0252
EMR(-5)/VMR(-5)	-0.0043	-0.6694	0.0090	0.0067	0.6734	0.0135	-0.0233	-0.4856	0.0027	0.0406	0.5578	0.0075
EMR(-6)/VMR(-6)	-0.0022	-0.4475	0.0029	0.0046	0.5595	0.0109	-0.0100	-0.2683	-0.0011	0.0296	0.4979	0.0073
EMR(-7)/VMR(-7)	-0.0006	-0.1306	-0.0025	0.0023	0.3392	0.0031	0.0044	0.1355	-0.0025	0.0146	0.2923	0.0012
EMR(-8)/VMR(-8)	0.0018	0.4544	0.0074	0.0016	0.3213	0.0024	0.0184	0.6057	0.0149	0.0079	0.2087	-0.0009
EMR(-9)/VMR(-9)	0.0022	0.7279	0.0260	0.0007	0.1798	-0.0012	0.0191	0.8359	0.0330	0.0015	0.0491	-0.0030
EMR(-10)/VMR(-10)	0.0018	0.8128	0.0320	0.0005	0.1533	-0.0018	0.0163	0.9931	0.0468	0.0005	0.0217	-0.0031

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of size coefficient  $\gamma_{mv}$  and Small/Large return R<sub>sml</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Size decile portfolios are formed with shares' market values at the end of each month, and hold for 24 months. Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values. Small/Large return R<sub>sml</sub> is the return spread between 'Small' portfolio and 'Large' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$ . Panel B reports corresponding parameters for equal-(value-) weighted Small/Large return R<sub>sml</sub>.

	Panel A 3 months' holding length								
		Size coef	ficient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>		
	Equal-w	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	7		6		4		5		
Adj-R <sup>2</sup>	0.09	988	0.08	55	0.09	27	0.08	06	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
IP(M)	0.3913	3.0781	0.3138	2.6042	3.0960	2.9481	2.4282	2.5129	
CPI(Y)	-0.0046	-0.5906	-0.0027	-0.3868	-0.0353	-0.5910	-0.0140	-0.2732	
UEXTBILL(Y)	-2.9674	-2.0517	-2.8757	-2.1250	-22.6178	-2.0061	-22.1973	-2.1969	
UEXTERM(Y)	0.0440	0.3177	0.0452	0.3480	0.2121	0.2073	0.2915	0.3001	
		D 1	D (	1.11	- 41.				
		Size coef	$B \ 0 \ months$	notaing ten	gth	Small/Large	return R <sub>cml</sub>		
	Equal-w	reighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	13	3	10	8	10	0	9	0	
Adi-R <sup>2</sup>	0.12	272	0.11	31	0.11	92	0.10	31	
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	- t(λ)	λ	$t(\lambda)$	
CPI(Y)	-0.0136	-0.7275	-0.0088	-0 5320	-0.1129	-0.8252	-0.0642	-0 5349	
UFXTBILL(Y)	-5 6816	-1 7483	-5 4062	-1 7710	-44 9141	-1 9425	-41 8923	-1 9232	
UEXTERM(Y)	-0.0228	-0.0909	0.0075	0.0323	-0.8591	-0.4451	-0 3945	-0.2211	
	0.0220	0.0707	0.0075	0.0525	0.0271	0.2211			
		Panel	C 12 months	' holding ler	ngth				
		Size coeffi	cient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>		
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	10		10		10		10	)	
Adj-R <sup>2</sup>	0.1689		0.14	96	0.17	64	0.14	93	
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	$t(\lambda)$	
GDP(Y)	0.0034	2.0411	0.0029	1.8107	0.0290	2.3280	0.0233	1.9025	
CPI(Y)	-0.0250	-0.9071	-0.0185	-0.7415	-0.1888	-0.8871	-0.1253	-0.6690	
UEXTBILL(Y)	-10.3982	-2.1117	-8.8450	-1.8997	-92.8557	-2.5106	-75.6362	-2.1449	
UEXTERM(Y)	-0.2560	-0.6648	-0.1139	-0.3144	-4.0384	-1.3199	-2.3041	-0.7984	
		Panel	D 18 months	' holding ler	ngth				
		Size coef	ficient y <sub>mv</sub>		-	Small/Large	return R <sub>sml</sub>		
	Equal-w	reighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	- 14	4	15	i	14		15		
Adj-R <sup>2</sup>	0.11	181	0.10	75	0.13	26	0.11	58	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	0.0056	2.4797	0.0045	2.1269	0.0499	2.8110	0.0374	2.2134	
CPI(Y)	-0.0311	-1.1188	-0.0247	-0.9478	-0.2147	-0.9876	-0.1572	-0.7857	
UEXTBILL(Y)	-5.9877	-0.9851	-6.1316	-1.0683	-44.0507	-0.9096	-47.4844	-1.0616	
		Panel	E 24 months	' holding ler	nøth				
		Size coef	ficient v		J	Small/Large	return R.m.		
	Equal-w	reighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	1	8	20	)	18	5	18	3	
$Adi-R^2$	0.07	787	0.07	50	0.08	88	0.08	34	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXGDP(Y)	$\Lambda$ $l(\Lambda)$ 0.0084 2.2564		0.0070	1.9619	0.0724	2,4914	0.0554	1.9183	
UEXCPI(Y)	-0.0070	-1.7244	-0.0061	-1.5404	-0.0581	-2.0257	-0.0487	-1.7056	
UEXTBILL(0)	-4.8635	-0.7981	-5.7310	-1.0558	-28.8353	-0.6477	-43.7022	-1.1543	
Note: This table present	s slope coet	fficient ) t	state and a	$\frac{1.0000}{\text{directed } \mathbf{P}^2}$	for multivariate regression of size			poofficiant	

Table 5.32 Multivariate regression of size effect on the combination of lagged macroeconomic variables

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of size coefficient  $\gamma_{mv}$ , and Small/Large return  $R_{sml}$  on the combination of lagged macroeconomic variables from January 1980 to December 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for 3, 6, 12, 18, and 24 months, respectively. Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market value in the formation time. Small/Large return  $R_{sml}$  is the return spread between 'Small' portfolio and 'Large' portfolio. IP(M) is the previous monthly change of Industry Production (IP). CPI(Y)/UEXCPI(Y) is the previous/ununexpected annual change of CPI. UEXTBILL(Q,Y) is unexpected change of TBILL in previous three months, 12 months. UEXTERM(Y) is unexpected change of TERM in previous 12 months. GDP(Y)/UEXGDP(Y) is pervious annual/unexpected annual change of GDP. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to E report corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with 3, 6, 12, 18, and 24 months' holding lengths, respectively.

Panel A April 1980-December 1989										
		Size coeffi	cient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>			
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted		
Lag length	7		6		2		4			
Adj-R <sup>2</sup>	0.14	11	0.13	76	0.13	12	0.1155			
Macroeconomic variables	$\lambda$ t( $\lambda$ )		λ	t(λ)	λ	t(λ)	λ	t(λ)		
IP(M)	0.2492	1.6784	0.1866	1.3693	1.8744	1.3273	1.4088	1.1503		
CPI(Y)	-0.0153	-2.2624	-0.0137	-2.2346	-0.0740	-1.1299	-0.0516	-0.8931		
UEXTBILL(Y)	-1.6054	-1.8455	-1.2080	-1.4063	-19.9177	-2.0433	-16.3429	-1.8513		
UEXTERM(Y)	0.0339	0.2503	0.0804	0.6402	-0.1852	-0.1946	0.3809	0.3847		

Panel B January 1990-December 1999

Small/Large return R<sub>sml</sub>

Size coefficient  $\gamma_{mv}$ 

# Table 5.33 Multivariate regression of size effect on the combination of lagged macroeconomic variables in subperiods (Size: 3 months' holding length)

	Equal-weighted		Value-we	eighted	Equal-we	eighted	Value-weighted	
Lag length	2		2		2		2	
Adj-R <sup>2</sup>	0.13	91	0.12	61	0.12	94	0.12	95
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
IP(M)	0.6466	2.1220	0.6006	2.0247	3.8544	1.8984	3.5804	1.8205
CPI(Y)	-0.0093 -0.9557		-0.0079	-0.8416	-0.0658	-0.9097	-0.0531	-0.8029
UEXTBILL(Y)	-4.2773	-1.8178	-4.4181	-1.9791	-30.7559	-1.7362	-33.4199	-2.0469
UEXTERM(Y)	-0.0288	-0.0288 -0.1332		-0.2364	-0.3857	-0.2463	-0.5683	-0.3659
		Panel C January 2000-December 2						
		Size coeffi	cient $\gamma_{mv}$			Small/Large	e return R <sub>sml</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted
Lag length	2		2		1		2	
Adj-R <sup>2</sup>	0.1100		0.05	98	0.1458		0.07	25
Macroeconomic variables	$\lambda$ $t(\lambda)$		λ	t(λ)	λ	t(λ)	λ	t(λ)
IP(M)	0.6049 1.7809		0.4452	1.4983	6.2172	2.0955	4.6090	1.8323

Adj-R <sup>2</sup>	0.11	00	0.05	98	0.14	58	0.07	25	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	$t(\lambda)$	λ	t(λ)	
IP(M)	0.6049	1.7809	0.4452	1.4983	6.2172	2.0955	4.6090	1.8323	
CPI(Y)	0.0151	1.5797	0.0142	1.5153	0.1151	1.5469	0.1027	1.3180	
UEXTBILL(Y)	-5.2966	-0.9340	-4.5844	-0.7992	-19.4426	-0.4760	-12.6472	-0.2908	
UEXTERM(Y)	0.8996	1.3712	0.6008	1.0602	10.5112	1.9600	5 4.6090 9 0.1027 0 -12.6472 0 7.1853	1.5315	
		Panel D	April 1980-	December 20	008				
		Size coefficient $\gamma_{mv}$		v		Small/Large return R <sub>sml</sub>			
	Equal m	aightad	Volue w	aightad	Equal weighted Value weighted				

		Size coeffi	cient $\gamma_{mv}$		Small/Large return R <sub>sml</sub>				
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted	
Lag length	7		6		4		5		
Adj-R <sup>2</sup>	0.093	88	0.08	55	0.09	27	0.0806		
Macroeconomic variables	$\lambda$ $t(\lambda)$		λ	t(λ)	λ	t(λ)	λ	t(λ)	
IP(M)	0.3913	3.0781	0.3138	2.6042	3.0960	2.9481	2.4282	2.5129	
CPI(Y)	-0.0046	-0.5906	-0.0027	-0.3868	-0.0353	-0.5910	-0.0140	-0.2732	
UEXTBILL(Y)	-2.9674 -2.0517		-2.8757	-2.1250	-22.6178	-2.0061	-22.1973	-2.1969	
UEXTERM(Y)	0.0440 0.3177		0.0452	0.3480	0.2121	0.2073	0.2915	0.3001	

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of size coefficient  $\gamma_{mv}$ , and Small/Large return  $R_{sml}$  on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for 3 months, Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values in the formation time. Small/Large return  $R_{sml}$  is the return spread in 'Small' portfolio and 'Large' portfolio. IP(M) is the previous monthly change of Industry Production (IP). CPI(Y) is the previous annual change of CPI. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXTERM(Y) is unexpected change of TERM in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with three subperiods and overall period, respectively.

Panel A July 1980-December 1989										
		Size coeff	ficient $\gamma_{mv}$			Small/Large 1	return R <sub>sml</sub>			
	Equal-w	eighted	Value-w	eighted	Equal-we	ighted	Value-we	ighted		
Lag length	8		8		5		3			
Adj-R <sup>2</sup>	0.24	35	0.24	58	0.222	25	0.201	.6		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
CPI(Y)	-0.0515	-3.0372	-0.0452	-3.1131	-0.3671	-2.8026	-0.3033	-2.7685		
UEXTBILL(Y)	-0.8797	-0.2660	-0.5157	-0.1734	-15.9417	-0.6395	-11.6080	-0.5470		
UEXTERM(Y)	0.0272	0.0848	0.1070	0.3658	-1.1434	-0.5326	-0.2930	-0.1583		
		Panel	B January 19	90-December	1999					
Size coefficient $\gamma_{mv}$ Small/Large return $R_{sml}$										
	Equal-w	eighted	Value-w	eighted	Equal-we	ighted	Value-weighted			
Lag length	4		4		4		4			
Adj-R <sup>2</sup>	0.17	68	0.1574		0.183	80	0.179	91		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
CPI(Y)	-0.0205	-1.1540	-0.0156	-0.8886	-0.1342	-1.0152	-0.0833	-0.6648		
UEXTBILL(Y)	-7.3022 -1.8247		-7.6767 -1.8549		-59.9120	-2.0671	-66.4462	-2.2279		
UEXTERM(Y)	-0.1825	-0.4671	-0.1922	-0.4903	-1.9396 -0.6740		-1.8322	-0.6458		
		Panel	C January 20	00-December	2008					
	Size coefficient $\gamma_{mv}$ Small/Large return $R_{sml}$									
	Equal-we	ighted	Value-w	eighted	Equal-we	ighted	Value-wei	ighted		
Lag length	4		4	ŀ	4		4			
Adj-R <sup>2</sup>	0.298	57	0.23	341	0.320	)9	0.247	8		
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
CPI(Y)	0.0374	2.6722	0.0355	2.4566	0.2953	2.7462	0.2795	2.5688		
UEXTBILL(Y)	-21.3517	-2.1227	-17.6026	-1.6368	-157.2094	-2.2550	-115.9658	-1.5160		
UEXTERM(Y)	1.2554	1.6245	1.0004	1.3462	13.3003	2.1985	11.2484	2.0648		
		Pan	el D July 1980	0-December 2	008					
		Size coef	ficient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>			
	Equal-we	eighted	Value-w	reighted	Equal-we	ighted	Value-wei	ighted		
Lag length	13	3	1	0	10		9			
Adj-R <sup>2</sup>	0.12	72	0.11	131	0.119	92	0.103	1		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
CPI(Y)	-0.0136	-0.7275	-0.0088	-0.5320	-0.1129	-0.8252	-0.0642	-0.5349		
UEXTBILL(Y)	-5.6816	-1.7483	-5.4062	-1.7710	-44.9141	-1.9425	-41.8923	-1.9232		
UEXTERM(Y)	-0.0228	-0.0909	0.0075	0.0323	-0.8591	-0.4451	-0.3945	-0.2211		
Note: This table present	ts slope coeffi	icient λ, t-st	ats, and adju	sted R <sup>2</sup> for n	nultivariate re	gression of s	ize coefficien	It $\gamma_{mv}$ , and		
Small/Large return R <sub>sml</sub>	Small/Large return R <sub>sml</sub> on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to									
2008. Decile size portfo	olios are form	ed with sha	ires' market	values at the	end of each i	nonth, and h	iold for 6 mo	nths, Size		
coefficient $\gamma_{mu}$ is estim	nated by more	ntniv-bv-mo	onthe cross	-sectional re	gression of r	orttolios' e	dual-(value-)	weighted		

### Table 5.34 Multivariate regression of size effect on the combination of lagged macroeconomic variables in subperiods (Size: 6 months' holding length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of size coefficient  $\gamma_{mv}$ , and Small/Large return  $R_{sml}$  on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for 6 months, Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values in the formation time. Small/Large return  $R_{sml}$  is the return spread in 'Small' portfolio and 'Large' portfolio. CPI(Y) is the previous annual change of CPI. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXTERM(Y) is unexpected change of TERM in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with three subperiods and overall period, respectively.

Panel A January 1981-December 1989										
		Size coe	fficient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>			
	Equal-v	veighted	Value-w	eighted	Equal-we	eighted	Value-we	ighted		
Lag length	1	1	10	)	10		8			
Adj-R <sup>2</sup>	0.2	.095	0.18	891	0.22	11	0.173	38		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
GDP(Y)	0.0005	0.4046	0.0000	-0.0300	0.0060	0.6646	-0.0002	-0.0249		
CPI(Y)	-0.0876	-1.8044	-0.0719	-1.8672	-0.6764	-1.8914	-0.5114	-2.0833		
UEXTBILL(Y)	-1.7121	-0.3740	-0.5162	-0.1342	-32.6086	-1.0432	-17.9835	-0.7472		
UEXTERM(Y)	-0.2960	-0.6194	-0.1064	-0.2524	-5.8114	-1.6269	-3.8814	-1.2853		
		Danal I	a January 100	December	1000					
		Size coefficient v								
	Equal-v	veighted	Value-w	eighted	Faual-we	ighted	Value-we	ighted		
Lag length	Equal-V	5	v andw	ergineu	Equal-we	15mcu	v and-we	ighteu		
$Adi_{\rm R}P^2$	0.4	102	0.41	07	0.43	78	0.448	21		
Auj-K Macroeconomic variables	۲.0 ۲	τ(λ)	۰.41 ک	t(λ)	۰.43 ک	τ(λ)	۵.440 ک	τ(λ)		
GDP(V)	0 0092	4 2160	0.0088	1 2002	0.0683	1 9153	0.0657	5 1246		
$CPI(\mathbf{Y})$	0.0092	4.2100	0.0008	4.2992	0.0005	4.9133	0.1723	0.8707		
UEXTRUL(V)	-0.0443	-6.3463 $-1.2714$		-1.1823	-0.2802	-1.4787	-0.1723	-0.8797		
UEXTBILL(T) UEXTERM(Y)	-0.1602 -0.3478		-7.4444	-1.4332	-04.2207	-1.0221	-1 6574	-1.8850		
OLATER(II)	-0.1002	-0.3470	-0.1474	-0.5500	-2.2-20	-0.0302	-1.0374	-0.52+2		
		Panel (	C January 200	0-December 2	2008					
		Size coeff	icient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>			
	Equal-we	ighted	Value-we	eighted	Equal-we	ighted	Value-weighted			
Lag length	3		3		3		3			
Adj-R <sup>2</sup>	0.42	57	0.388	82	0.432	24	0.403	6		
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
GDP(Y)	0.0073	1.5160	0.0061	1.3742	0.0610	1.7708	0.0480	1.6343		
CPI(Y)	0.0298	1.3114	0.0245	1.0743	0.2591	1.4872	0.2076	1.2262		
UEXTBILL(Y)	-70.2552	-4.1313	-62.0499	-4.1842	-544.9059	-3.9265	-468.1832	-4.1554		
UEXTERM(Y)	-2.3767	-1.5280	-2.3532	-1.7040	-17.1842	-1.4006	-17.1327	-1.6697		
		Panel	D January 1981	1-December 2	2008					
		Size coeff	icient v			Small/Large	return R			
	Equal-we	ighted	Value-we	eighted	Equal-we	ighted	Value-we	ighted		
Lag length	. 10	-	10	-	. 10	-	10	-		
Adj-R <sup>2</sup>	0.16	39	0.149	96	0.176	54	0.1493			
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)		
GDP(Y)	0.0034	2.0411	0.0029	1.8107	0.0290	2.3280	0.0233	1.9025		
CPI(Y)	-0.0250	-0.9071	-0.0185	-0.7415	-0.1888	-0.8871	-0.1253	-0.6690		

### Table 5.35 Multivariate regression of size effect on the combination of laggedmacroeconomic variables in subperiods (Size: 12 months' holding length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of size coefficient  $\gamma_{mv}$ , and Small/Large return  $R_{sml}$  on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for 12 months, Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values in the formation time. Small/Large return  $R_{sml}$  is the return spread in 'Small' portfolio and 'Large' portfolio. GDP(Y) is the previous annual change of GDP. CPI(Y) is the previous annual change of CPI. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXTERM(Y) is unexpected change of TERM in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with three subperiods and overall period, respectively.

-1.8997

-0.3144

-92.8557

-4.0384

-2.5106

-1.3199

-75.6362

-2.3041

-2.1449

-0.7984

-8.8450

-0.1139

UEXTBILL(Y)

UEXTERM(Y)

-10.3982

-0.2560

-2.1117

-0.6648

Panel A July 1981-December 1989									
		Size coe	fficient $\gamma_{mv}$			Small/Large	ge return R <sub>sml</sub>		
	Equal-v	veighted	Value-w	eighted	Equal-we	eighted	Value-we	ighted	
Lag length	1	2	11	l	11		8		
Adj-R <sup>2</sup>	0.0	744	0.05	549	0.10	46	0.073	32	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	Λ	t(λ)	λ	t(λ)	
GDP(Y)	0.0034	2.0588	0.0022	1.5741	0.0351	2.8063	0.0202	1.8212	
CPI(Y)	-0.0558	-1.1997	-0.0475	-1.2117	-0.3783	-1.0753	-0.2929	-1.1375	
UEXTBILL(Y)	4.7440	0.8142	4.0164	0.7877	41.4778	0.9263	34.7442	0.9694	
		Danal	D. January 100	0 December	1000				
		Panel	B January 199	1999	C 11/T	( D			
	Equal wa	Size coeffi	Clent $\gamma_{mv}$	inhtad	Equal wa	Small/Large	Value wa	abtad	
I	Equal-we	righted	value-we	eignted	Equal-we	ignied	Value-weighted		
Lag length $A_1$ : $D^2$	0.22	7	0 220	20	0.26		0.279		
Adj-R <sup>-</sup>	0.33	91	0.338	88	0.366	52	0.378	12	
Macroeconomic	λ	$t(\lambda)$	λ	$t(\lambda)$	Λ	$t(\lambda)$	λ	$t(\lambda)$	
GDP(Y)	0.0089	0.0089 2.8687		0.0084 2.9225		3.1038	0.0677	3.0725	
CPI(Y)	-0.0326	-0.0326 $-0.839211 1332 -0.1472$		-0.5631	-0.2228	-0.7563	-0.1160	-0.4242	
UEXTBILL(Y)	-11.1332	-2.1472	-12.0757	-2.3902	-89.6820	-2.5369	-102.7167	-3.1108	
		Panel	C January 200	0-December	2008				
		Size coeffi	cient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>		
	Equal-we	ghted	Value-we	eighted	Equal-we	ighted	Value-we	ighted	
Lag length	- 7	-	7	-	- 6	-	6	-	
Adj-R <sup>2</sup>	0.16	52	0.158	39	0.205	51	0.204	.3	
Macroeconomic	λ	t(λ)	λ	t(λ)	Λ	t(λ)	λ	t(λ)	
GDP(Y)	0.0076	0.7453	0.0062	0.6802	0.0529	0.8276	0.0360	0.6733	
CPI(Y)	0.0309	0.9831	0.0272	0.9396	0.2952	1.3200	0.2593	1.2999	
UEXTBILL(Y)	-44.8596	-4.6299	-39.8856	-4.4076	-335.7370	-4.6783	-292.7927	-4.4400	
		Dom	1 D July 1091	December 2	000				
			Figiant w	-December 2	008	Small/Langa	notum D		
	Equal w	Size coel		ighted	Equal wa	Sman/Large	Velue we	abtad	
Loglanath	Equal-w	/eigineu	value-we	rigitted	Equal-we	igitteu	value-we	Igilieu	
Lag length $A \downarrow D^2$	1	4	13	75	14		13	0	
Adj-K	0.1	181	0.10	15	0.132	20	0.115	0	
wacroeconomic variables	λ	ι(λ) 2.4707	λ	ι(λ) 0.1000	Λ 0.0400	ι(λ) 2.0110	λ	ι(λ) 2.2124	
GDP(Y)	0.0056	2.4/9/	0.0045	2.1269	0.0499	2.8110	0.0374	2.2134	
CPI(Y)	-0.0311	-1.1188	-0.0247	-0.9478	-0.2147	-0.9876	-0.1572	-0.7857	
UEXTBILL(Y)	-5.9877	-0.9851	-6.1316	-1.0683	-44.0507	-0.9096	-47.4844	-1.0616	
Note: This table presents slope coefficient $\lambda$ , t-stats, and adjusted R <sup>2</sup> for multivariate regression of size coefficient $\gamma$ and Small/Large return R <sub>sml</sub> on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and fr 2000 to 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for							icient $\gamma_{mv}$ , and from old for 18		

## Table 5.36 Multivariate regression of size effect on the combination of lagged macroeconomic variables in subperiods (Size: 18 months' holding length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of size coefficient  $\gamma_{mv}$ , and Small/Large return  $R_{sml}$  on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for 18 months, Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values in the formation time. Small/Large return  $R_{sml}$ is the return spread in 'Small' portfolio and 'Large' portfolio. GDP(Y) is the previous annual change of GDP. CPI(Y) is the previous annual change of CPI. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with three subperiods and overall period, respectively.

		Panel A January 1982-December 1989							
		Size coe	fficient $\gamma_{mv}$			Small/Large	e return R <sub>sml</sub>		
	Equal-v	veighted	Value-w	reighted	Equal-we	eighted	Value-we	ighted	
Lag length	1	3	12	2	12	2	10		
Adj-R <sup>2</sup>	0.0	059	-0.0	016	0.01	45	0.002	24	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXGDP(Y)	0.0051	1.2718	0.0036	1.2540	0.0454	1.6117	0.0270	1.6695	
UEXCPI(Y)	0.0004	0.0770	0.0005	0.1095	-0.0008	-0.0161	0.0017	0.0523	
UEXTBILL(Q)	-1.0519	-0.1615	-1.8526	-0.3372	-6.6173	-0.1459	-20.6622	-0.5738	
		Panel	B January 190	0-December	1999				
		Size coefficient $\gamma_{my}$ Small/Large return $R_{em}$							
	Equal-v	veighted	Value-w	reighted	Equal-we	eighted	Value-we	ighted	
Lag length	Lequin ,	7	7	e-B-itea	8	Billea	8	18.1100	
Adi-R <sup>2</sup>	0.3	869	0.38	331	0.40	22	0.401	11	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXGDP(Y)	0.0164	3.5718	0.0153	3.5220	0.1350	4.0423	0.1238	3.8363	
UEXCPI(Y)	-0.0185 -4.5130		-0.0168	-4.1799	-0.1454	-4.5899	-0.1318	-4.1499	
UEXTBILL(Q)	-4.5592 -0.5615		-6.1669	-0.8032	-21.4485	-0.3520	-41.2093	-0.7086	
		Panel	C January 200	00-December	2008				
		Size coeffi	icient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>		
	Equal-we	ighted	Value-we	eighted	Equal-we	eighted	Value-weighted		
Lag length	13		15		10		11		
Adj-R <sup>2</sup>	0.087	78	0.07	82	0.113	32	0.111	0	
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXGDP(Y)	0.0012	0.0776	-0.0011	-0.0787	-0.0165	-0.1885	-0.0437	-0.5897	
UEXCPI(Y)	0.0094	0.7538	0.0092	0.7432	0.1042	1.3424	0.0976	1.2281	
UEXTBILL(Q)	-63.7850	-3.0623	-54.9866	-2.8716	-437.7071	-3.2386	-351.3304	-2.7569	
		Panel	D January 198	82-December	2008				
		Size coef	ficient $\gamma_{mv}$			Small/Large	return R <sub>sml</sub>		
	Equal-w	reighted	Value-we	eighted	Equal-we	eighted	Value-we	ighted	
Lag length	1	8	20		18		18		
Adj-R <sup>2</sup>	0.07	787	0.07	50	0.08	88	0.0834		
Macroeconomic variables	$\lambda_{IV}$	$t(\lambda_{IV})$	$\lambda_{IV}$	$t(\lambda_{IV})$	$\lambda_{IV}$	$t(\lambda_{IV})$	$\lambda_{IV}$	$t(\lambda_{IV})$	
UEXGDP(Y)	0.0084	2.2564	0.0070	1.9619	0.0724	2.4914	0.0554	1.9183	
UEXCPI(Y)	-0.0070	-1.7244	-0.0061	-1.5404	-0.0581	-2.0257	-0.0487	-1.7056	

### Table 5.37 Multivariate regression of size effect on the combination of lagged macroeconomic variables in subperiods (Size: 24 months' holding length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of size coefficient  $\gamma_{mv}$ , and Small/Large return  $R_{sml}$  on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile size portfolios are formed with shares' market values at the end of each month, and hold for 24 months, Size coefficient  $\gamma_{mv}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their log mean market values in the formation time. Small/Large return  $R_{sml}$  is the return spread in 'Small' portfolio and 'Large' portfolio. UEXGDP(Y) is the unexpected annual change of GDP. UEXCPI(Y) is the unexpected annual change of CPI. UEXTBILL(Q) is unexpected change of TBILL in previous 3 months. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$  with three subperiods and overall period, respectively.

-1.0558

-28.8353

-43.7022

-1.1543

-0.6477

-5.7310

-0.7981

-4.8635

UEXTBILL(Q)

Panel A Value coefficient $\gamma_{mv}$						Panel B Low/High return R <sub>lmh</sub>							
	Panel	A1 Equal-we	eighted	Panel A	2 Value-wei	ghted		Panel E	31 Equal-weig	ghted	Panel B2 Value-weighted		
Macroeconomic		•	•			-							
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>		λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	0.0000	0.0076	-0.0029	0.0002	0.3187	-0.0010		0.0019	0.5176	-0.0004	0.0052	1.2580	0.0063
UEXGDP(Y)	0.0001	0.1126	-0.0027	0.0003	0.3804	0.0002		0.0032	0.5413	-0.0001	0.0094	1.3740	0.0099
IP(M)	0.1084	1.0763	0.0019	0.1082	1.0343	0.0015		0.7240	0.8295	-0.0013	2.5662	2.1105	0.0077
IP(Q)	0.0765	0.5748	0.0021	0.0728	0.5431	0.0012		0.2603	0.2429	-0.0025	0.9341	0.7283	0.0000
IP(H)	0.1199	1.2191	0.0235	0.0966	1.1101	0.0126		0.6928	1.0039	0.0038	1.1968	1.4605	0.0074
IP(Y)	0.0324	0.4753	0.0016	0.0254	0.4144	-0.0004		0.3069	0.7180	0.0002	0.2907	0.5908	-0.0015
UEXIP(Q)	0.0004	0.5694	0.0006	0.0005	0.6602	0.0013		0.0006	0.0824	-0.0029	0.0082	0.9424	0.0020
UEXIP(H)	0.0006	0.8404	0.0080	0.0006	0.8581	0.0062		0.0026	0.4094	-0.0014	0.0082	1.1029	0.0047
UEXIP(Y)	0.0005	0.8011	0.0088	0.0004	0.8288	0.0066		0.0028	0.6475	0.0004	0.0051	1.0068	0.0028
IPM(M)	0.1223	1.0170	0.0036	0.1612	1.3674	0.0073		0.8105	0.9112	-0.0007	2.5470	2.0165	0.0081
IPM(O)	0.0473	0.3224	-0.0007	0.0952	0.6621	0.0052		0.1256	0.1195	-0.0028	0.9698	0.8407	0.0007
IPM(H)	0.0703	0.6749	0.0079	0.0643	0.7019	0.0053		0.4426	0.6634	0.0004	0.8957	1.1813	0.0040
IPM(Y)	0.0192	0.3151	-0.0007	0.0130	0.2350	-0.0020		0.3050	0.8295	0.0014	0.0992	0.2647	-0.0027
UEXIPM(Q)	0.0004	0.4362	0.0000	0.0008	0.9049	0.0078		0.0008	0.1117	-0.0028	0.0095	1.2466	0.0044
UEXIPM(H)	0.0004	0.5186	0.0029	0.0006	0.7651	0.0072		0.0019	0.3082	-0.0020	0.0076	1.1476	0.0047
UEXIPM(Y)	0.0003	0.4677	0.0021	0.0003	0.5616	0.0027		0.0022	0.5823	-0.0003	0.0033	0.7971	0.0001
M0 (M)	-0.1108	-0.4891	-0.0021	0.0841	0.2980	-0.0025		-2.5112	-0.7830	0.0002	-0.1332	-0.0288	-0.0029
M0 (O)	0.0861	0.3974	-0.0014	0.0975	0.4048	-0.0011		1.1047	0.4800	-0.0010	2.1470	0.8743	0.0009
M0 (H)	-0.0805	-0.4275	-0.0001	0.0060	0.0325	-0.0029		-0.7854	-0.3785	-0.0008	1.2206	0.6158	-0.0003
M0 (Y)	-0.1200	-0.8828	0.0128	-0.1152	-0.8120	0.0102		-0.5743	-0.4673	-0.0002	0.2056	0.1515	-0.0027
UEXM0(O)	0.0000	0.0701	-0.0029	0.0000	-0.1308	-0.0028		0.0000	0.1639	-0.0028	0.0000	-0.2517	-0.0024
UEXM0 (H)	0.0000	0.0148	-0.0029	0.0000	-0.1341	-0.0028		0.0000	0.1803	-0.0026	0.0000	0.0257	-0.0029
UEXM0 (Y)	0.0000	-0.1904	-0.0026	0.0000	-0.3916	-0.0015		0.0000	0.0562	-0.0029	0.0000	-0.1054	-0.0028
UNE (M)	-0.0750	-0.6413	0.0021	-0.0182	-0.1603	-0.0026		-0.4758	-0.5297	-0.0014	0.9042	0.6535	0.0000
UNE (O)	-0.0320	-0.5804	0.0027	-0.0087	-0.1672	-0.0025		-0.2016	-0.4622	-0.0012	0.3395	0.6155	-0.0004
UNE (H)	-0.0168	-0.5562	0.0023	-0.0084	-0.3002	-0.0017		-0.1502	-0.6422	0.0003	0.0756	0.2980	-0.0025
UNE (Y)	-0.0075	-0.5586	0.0007	-0.0012	-0.0936	-0.0028		-0.0776	-0.6918	0.0000	0.1395	1.0254	0.0020
UEXUNE(O)	-0.0019	-0.4528	-0.0001	0.0003	0.0674	-0.0029		-0.0156	-0.4495	-0.0015	0.0357	0.8255	0.0010
UEXUNE(H)	-0.0012	-0.4557	0.0000	-0.0001	-0.0363	-0.0029		-0.0120	-0.5659	-0.0006	0.0151	0.6391	-0.0010
UEXUNE(Y)	-0.0007	-0.5038	0.0000	0.0000	-0.0068	-0.0029		-0.0080	-0.7127	0.0003	0.0119	0.9205	0.0008
CPI (M)	-0.0020	-0.3116	-0.0027	0.0041	0.4700	-0.0022		0.0052	0.0632	-0.0029	0.0212	0.1502	-0.0028
CPI(O)	0.0016	0.3164	-0.0025	0.0009	0.1529	-0.0028		0.0664	0.9071	0.0033	0.0547	0.5853	-0.0008
CPI(H)	-0.0009	-0.2309	-0.0026	-0.0026	-0.6575	-0.0002		0.0596	0.9859	0.0094	0.0367	0.5273	-0.0005
$CPI(\mathbf{Y})$	-0.0021	-0.6473	0.0017	-0.0039	-1.2244	0.0114		0.0295	0.6497	0.0041	-0.0034	-0.0696	-0.0029
UEXCPI(Q)	0.0002	0.1932	-0.0023	-0.0001	-0.0767	-0.0028		0.0068	0.8151	0.0007	0.0026	0.2544	-0.0026

 Table 5.38 Univariate regression of value effect on individual lagged macroeconomic variables (Value-3)

UEXCPI(H)	-0.0004	-0.4161	-0.0003	-0.0007	-0.7280	0.0059	0.0047	0.6988	0.0006	-0.0004	-0.0464	-0.0029
UEXCPI(Y)	-0.0008	-1.2773	0.0300	-0.0010	-1.6218	0.0461	0.0010	0.2009	-0.0026	-0.0043	-0.8959	0.0007
TBILL(M)	-0.0037	-0.4466	-0.0020	-0.0019	-0.1992	-0.0027	-0.0964	-1.2614	0.0017	-0.0655	-0.5151	-0.0018
TBILL (Q)	-0.0063	-1.0728	0.0059	-0.0074	-1.0479	0.0081	-0.0830	-1.5570	0.0089	-0.0660	-0.8370	0.0009
TBILL (H)	-0.0087	-1.9832	0.0333	-0.0111	-2.5207	0.0504	-0.0792	-1.8824	0.0199	-0.0918	-1.6958	0.0128
TBILL (Y)	-0.0055	-1.4260	0.0274	-0.0080	-2.1455	0.0561	-0.0348	-0.9320	0.0065	-0.0849	-2.2965	0.0260
UEXTBILL(Q)	-0.9862	-0.7605	0.0039	-1.0881	-0.7232	0.0046	-12.8058	-1.2892	0.0059	-10.6889	-0.7801	0.0002
UEXTBILL(H)	-1.5118	-1.3855	0.0248	-1.8264	-1.5416	0.0338	-14.5614	-1.7607	0.0168	-15.1085	-1.4519	0.0080
UEXTBILL(Y)	-1.6079	-1.7580	0.0564	-2.0639	-2.2608	0.0857	-10.4805	-1.5405	0.0163	-16.2410	-2.2128	0.0209
TERM(M)	0.0779	0.8920	-0.0002	0.1164	1.1831	0.0026	0.7851	1.0142	-0.0008	1.1267	0.8366	-0.0007
TERM(Q)	0.0620	0.7899	0.0021	0.0923	1.0988	0.0071	0.5743	0.7139	0.0003	0.9206	0.8198	0.0014
TERM(H)	0.0576	0.9933	0.0060	0.1050	1.8073	0.0241	0.3675	0.5251	-0.0001	1.0780	1.4164	0.0094
TERM(Y)	0.0569	1.0766	0.0151	0.0865	1.6855	0.0350	0.2369	0.4705	-0.0005	0.7560	1.4644	0.0096
UEXTERM(Q)	0.0730	0.7211	0.0006	0.1155	1.0364	0.0052	0.7750	0.7708	0.0002	1.2496	0.7993	0.0012
UEXTERM(H)	0.0778	0.8768	0.0039	0.1334	1.4372	0.0151	0.6638	0.6616	0.0008	1.4178	1.1188	0.0059
UEXTERM(Y)	0.0833	1.0624	0.0115	0.1367	1.8173	0.0324	0.4522	0.5364	0.0003	1.2824	1.3749	0.0105
DY(M)	-0.0138	-1.2581	0.0012	-0.0203	-1.5521	0.0052	-0.0719	-0.6035	-0.0021	-0.2356	-1.3117	0.0018
DY(Q)	-0.0245	-2.0853	0.0383	-0.0261	-2.2814	0.0396	-0.2656	-2.0965	0.0342	-0.3484	-2.3501	0.0300
DY(H)	-0.0239	-2.3400	0.0891	-0.0251	-2.9625	0.0890	-0.2350	-2.0426	0.0650	-0.3260	-2.4702	0.0643
DY(Y)	-0.0130	-1.9107	0.0609	-0.0122	-1.9473	0.0479	-0.1163	-2.4196	0.0359	-0.0915	-1.0601	0.0095
UEXDY(Q)	-0.9068	-2.2554	0.0502	-1.0259	-2.4038	0.0587	-7.0890	-2.4377	0.0219	-10.3498	-2.7028	0.0243
UEXDY(H)	-1.0507	-2.6594	0.1239	-1.1108	-3.1069	0.1257	-8.0505	-2.7090	0.0540	-11.1505	-3.2521	0.0532
UEXDY(Y)	-0.7273	-2.3582	0.1254	-0.7520	-2.7896	0.1215	-5.4555	-2.5462	0.0522	-6.6946	-2.5203	0.0398
EMR(-1)/VMR(-1)	0.0024	1.5375	0.0063	0.0048	1.4675	0.0090	0.0248	0.9988	0.0045	0.0633	0.9352	0.0060
EMR(-2)/VMR(-2)	0.0015	1.7473	0.0062	0.0030	1.7268	0.0098	0.0255	1.7464	0.0164	0.0222	0.5099	0.0002
EMR(-3)/VMR(-3)	0.0001	0.0628	-0.0029	0.0015	1.5420	0.0056	0.0065	0.5875	-0.0002	-0.0033	-0.1381	-0.0027
EMR(-4)/VMR(-4)	0.0000	-0.0382	-0.0029	0.0018	2.7695	0.0247	0.0020	0.2692	-0.0023	0.0059	0.3984	-0.0017
EMR(-5)/VMR(-5)	-0.0003	-0.4435	0.0003	0.0007	1.7994	0.0064	0.0000	-0.0004	-0.0029	0.0005	0.0573	-0.0029
EMR(-6)/VMR(-6)	0.0002	0.5304	0.0007	0.0011	2.9379	0.0372	0.0018	0.4777	-0.0011	0.0065	0.8909	0.0030
EMR(-7)/VMR(-7)	0.0003	1.1345	0.0081	0.0008	2.8184	0.0385	0.0013	0.4357	-0.0013	0.0055	1.0201	0.0046
EMR(-8)/VMR(-8)	0.0004	2.2536	0.0348	0.0009	2.9338	0.0758	0.0036	1.8199	0.0173	0.0080	1.9133	0.0253
EMR(-9)/VMR(-9)	0.0003	2.1587	0.0279	0.0006	3.0454	0.0611	0.0028	1.9980	0.0208	0.0056	1.8985	0.0211
EMR(-10)/VMR(-10)	0.0001	2.1280	0.0125	0.0003	2.1187	0.0292	0.0017	1.8619	0.0134	0.0025	1.1179	0.0049

Link(10)0.0012.1200.01230.00032.11070.02320.00171.80190.01340.00251.11790.0049Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> onindividual lagged macroeconomic variables from 1980 to 2008, respectively. Decile valuee portfolios are formed with shares' price-to-book values at the end of each month, andhold for 3 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period ontheir mean price-to-book values. Low/High return R<sub>lmh</sub> is the return spread in 'Low' portfolio and 'High' portfolio. t-stats is adjusted by Newey-West (1987) approach withautomatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$ . Panel Breports corresponding parameters for equal-(value-) weighted Low/High return R<sub>lmh</sub>.

	Panel A Value coefficient $\gamma_{mv}$						Panel B Low/High return R <sub>lmh</sub>						
	Panel A1 Equal-weighted			Panel A	Panel A2 Value-weighted			Panel B1 Equal-weighted			Panel B2 Value-weighted		
Macroeconomic		•											
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	0.0000	0.0583	-0.0029	0.0003	0.3368	-0.0005	0.0040	0.6909	0.0011	0.0054	0.7731	0.0014	
UEXGDP(Y)	0.0000	0.0308	-0.0029	0.0003	0.2030	-0.0018	0.0057	0.5559	0.0005	0.0090	0.7468	0.0022	
IP(M)	0.1828	0.9303	0.0022	0.2321	1.0659	0.0037	1.5504	1.1328	-0.0002	2.9227	1.8775	0.0031	
IP(Q)	0.2173	0.9945	0.0120	0.1886	0.8333	0.0060	1.8577	1.3103	0.0052	2.0412	1.5547	0.0030	
IP(H)	0.2180	1.5055	0.0296	0.1862	1.1600	0.0160	1.5702	1.6244	0.0096	1.6035	1.5012	0.0051	
IP(Y)	0.0469	0.6169	0.0006	0.0145	0.1831	-0.0027	0.6953	1.2045	0.0029	-0.0050	-0.0073	-0.0029	
UEXIP(Q)	0.0011	0.8475	0.0060	0.0012	0.8726	0.0055	0.0090	0.9408	0.0012	0.0147	1.5362	0.0038	
UEXIP(H)	0.0014	1.2592	0.0177	0.0013	1.1075	0.0114	0.0111	1.3684	0.0066	0.0144	1.7268	0.0068	
UEXIP(Y)	0.0009	1.0611	0.0134	0.0007	0.8198	0.0063	0.0080	1.3473	0.0069	0.0057	0.8820	0.0002	
IPM(M)	0.1892	0.8224	0.0030	0.2901	1.0688	0.0081	1.6229	1.1415	0.0003	3.6395	2.3103	0.0070	
IPM(Q)	0.1092	0.4556	0.0014	0.1430	0.5701	0.0031	1.3087	0.9041	0.0018	1.7891	1.4152	0.0024	
IPM(H)	0.1061	0.8519	0.0063	0.0768	0.5615	0.0009	1.1067	1.2874	0.0046	0.7629	0.8670	-0.0008	
IPM(Y)	0.0378	0.5991	0.0003	0.0052	0.0761	-0.0029	0.7824	1.5844	0.0075	-0.0388	-0.0756	-0.0029	
UEXIPM(Q)	0.0008	0.5039	0.0017	0.0012	0.7462	0.0065	0.0078	0.7958	0.0005	0.0161	1.8493	0.0061	
UEXIPM(H)	0.0008	0.6793	0.0042	0.0008	0.6755	0.0036	0.0082	1.0775	0.0033	0.0107	1.4544	0.0035	
UEXIPM(Y)	0.0005	0.6424	0.0036	0.0004	0.4467	0.0003	0.0067	1.2980	0.0063	0.0030	0.5998	-0.0018	
M0 (M)	-0.0868	-0.1950	-0.0028	0.2738	0.5577	-0.0015	-2.5816	-0.4523	-0.0017	5.5204	1.0176	0.0005	
M0 (Q)	-0.1397	-0.4001	-0.0014	0.0790	0.2196	-0.0025	-1.6910	-0.3564	-0.0012	2.5782	0.6502	-0.0005	
M0 (H)	-0.2828	-0.9176	0.0105	-0.2070	-0.6146	0.0028	-1.6195	-0.4850	0.0004	1.0285	0.3058	-0.0021	
M0 (Y)	-0.3272	-1.3726	0.0417	-0.3504	-1.2935	0.0379	-1.6982	-0.7749	0.0060	-0.3715	-0.1351	-0.0027	
UEXM0(Q)	0.0000	-0.2367	-0.0025	0.0000	-0.1049	-0.0029	0.0000	-0.1910	-0.0025	0.0000	0.1556	-0.0026	
UEXM0 (H)	0.0000	-0.3379	-0.0019	0.0000	-0.2839	-0.0023	0.0000	-0.1198	-0.0028	0.0000	0.1404	-0.0026	
UEXM0 (Y)	0.0000	-0.4707	-0.0010	0.0000	-0.5629	0.0000	0.0000	-0.0703	-0.0029	0.0000	0.0577	-0.0029	
UNE (M)	-0.1398	-0.6968	0.0035	-0.0790	-0.3368	-0.0013	-1.0945	-0.6566	0.0000	1.9880	1.0195	0.0030	
UNE (Q)	-0.0583	-0.6604	0.0040	-0.0283	-0.2759	-0.0016	-0.5808	-0.7739	0.0022	0.5751	0.6605	0.0001	
UNE (H)	-0.0325	-0.7697	0.0045	-0.0112	-0.2279	-0.0022	-0.4269	-1.1907	0.0066	0.3356	0.6810	0.0006	
UNE (Y)	-0.0142	-0.7380	0.0020	-0.0019	-0.0818	-0.0029	-0.1743	-0.8641	0.0026	0.2728	0.9017	0.0054	
UEXUNE(Q)	-0.0035	-0.5472	0.0007	-0.0003	-0.0420	-0.0029	-0.0422	-0.6957	0.0010	0.0602	0.8254	0.0019	
UEXUNE(H)	-0.0023	-0.6202	0.0013	-0.0001	-0.0194	-0.0029	-0.0330	-0.9493	0.0037	0.0354	0.7859	0.0017	
UEXUNE(Y)	-0.0013	-0.6528	0.0010	0.0001	0.0551	-0.0029	-0.0182	-0.9255	0.0033	0.0266	0.9322	0.0051	
CPI (M)	0.0026	0.2061	-0.0028	-0.0011	-0.0766	-0.0029	0.1917	0.9252	0.0022	0.1499	0.6364	-0.0010	
CPI (Q)	-0.0016	-0.1593	-0.0028	-0.0085	-0.7908	0.0011	0.1509	0.8773	0.0089	0.0942	0.5365	-0.0001	
CPI (H)	-0.0034	-0.4492	-0.0009	-0.0104	-1.2976	0.0119	0.1152	0.8661	0.0141	0.0274	0.1996	-0.0024	
CPI (Y)	-0.0056	-0.8553	0.0098	-0.0110	-1.5853	0.0357	0.0381	0.4278	0.0014	-0.0464	-0.4566	0.0010	
UEXCPI(Q)	-0.0020	-0.8272	0.0120	-0.0031	-1.1393	0.0275	0.0048	0.2757	-0.0023	-0.0063	-0.3529	-0.0023	

 Table 5.39 Univariate regression of value effect on individual lagged macroeconomic variables (Value-6)

\_

UEXCPI(H)	-0.0025	-1.3296	0.0444	-0.0035	-1.7208	0.0754	0.0005	0.0355	-0.0029	-0.0120	-0.8512	0.0021
UEXCPI(Y)	-0.0023	-1.5384	0.0995	-0.0032	-2.0378	0.1507	-0.0026	-0.2647	-0.0020	-0.0144	-1.4775	0.0150
TBILL(M)	-0.0140	-1.0010	0.0019	-0.0216	-1.4456	0.0062	-0.2189	-1.5473	0.0059	-0.2454	-1.1479	0.0038
TBILL (Q)	-0.0188	-1.7278	0.0272	-0.0266	-2.0969	0.0452	-0.1760	-1.5652	0.0167	-0.2207	-1.6240	0.0159
TBILL (H)	-0.0179	-1.8751	0.0550	-0.0270	-2.7079	0.1027	-0.1092	-0.9923	0.0131	-0.2298	-2.1845	0.0404
TBILL (Y)	-0.0101	-1.5285	0.0369	-0.0161	-2.3866	0.0778	-0.0705	-1.0189	0.0114	-0.2084	-3.0865	0.0737
UEXTBILL(Q)	-3.5892	-1.5447	0.0316	-4.9637	-1.8889	0.0497	-32.8469	-1.8213	0.0186	-38.9625	-1.7079	0.0155
UEXTBILL(H)	-4.1693	-1.9581	0.0776	-5.8132	-2.3852	0.1221	-30.3645	-1.8426	0.0289	-43.4518	-2.3919	0.0368
UEXTBILL(Y)	-3.4004	-1.8512	0.0983	-4.9996	-2.4474	0.1718	-19.5624	-1.3742	0.0220	-39.3323	-2.7997	0.0586
TERM(M)	0.1048	0.6895	-0.0011	0.1986	1.0845	0.0025	1.0226	0.6040	-0.0016	2.5912	1.0923	0.0023
TERM(Q)	0.0940	0.7038	0.0014	0.2234	1.4626	0.0167	0.7650	0.4904	-0.0008	2.4133	1.2061	0.0101
TERM(H)	0.1054	1.0546	0.0085	0.2292	2.1745	0.0402	0.5825	0.4546	-0.0003	2.5284	1.7865	0.0269
TERM(Y)	0.0881	0.9329	0.0136	0.1668	1.6730	0.0444	0.3690	0.4579	-0.0008	1.5131	1.5987	0.0192
UEXTERM(Q)	0.1152	0.6484	0.0004	0.2729	1.3487	0.0122	1.1140	0.5369	-0.0006	3.3197	1.1880	0.0098
UEXTERM(H)	0.1366	0.8925	0.0050	0.3119	1.8404	0.0301	1.0106	0.5249	0.0003	3.5703	1.5290	0.0217
UEXTERM(Y)	0.1404	1.0121	0.0127	0.2902	1.8860	0.0504	0.6672	0.4518	-0.0003	2.8715	1.6323	0.0268
DY(M)	-0.0555	-2.0413	0.0226	-0.0632	-2.3192	0.0235	-0.5165	-1.4771	0.0135	-0.6867	-2.0121	0.0148
DY(Q)	-0.0500	-2.2488	0.0626	-0.0540	-2.8208	0.0582	-0.5210	-1.8531	0.0501	-0.6910	-2.3598	0.0539
DY(H)	-0.0375	-2.1048	0.0825	-0.0403	-2.7011	0.0758	-0.3108	-1.9476	0.0408	-0.3637	-1.5536	0.0335
DY(Y)	-0.0165	-1.6192	0.0362	-0.0169	-1.4664	0.0297	-0.1088	-1.1180	0.0097	-0.0324	-0.1904	-0.0023
UEXDY(Q)	-2.0293	-2.5922	0.0983	-2.3020	-3.2067	0.1011	-16.0792	-2.3669	0.0444	-22.3039	-3.1191	0.0526
UEXDY(H)	-1.8249	-2.5747	0.1424	-2.0357	-3.1112	0.1414	-13.3492	-2.4700	0.0550	-17.5234	-2.6487	0.0579
UEXDY(Y)	-1.1922	-2.1486	0.1280	-1.3093	-2.4880	0.1232	-8.0051	-2.1265	0.0411	-8.7101	-1.6229	0.0288
EMR(-1)/VMR(-1)	0.0049	1.8776	0.0111	0.0131	1.9712	0.0260	0.0673	1.6356	0.0167	0.0882	0.6348	0.0045
EMR(-2)/VMR(-2)	0.0024	1.5707	0.0053	0.0048	1.5700	0.0084	0.0467	1.8551	0.0206	-0.0134	-0.1571	-0.0024
EMR(-3)/VMR(-3)	-0.0003	-0.1149	-0.0027	0.0030	1.5641	0.0078	0.0106	0.4976	-0.0003	-0.0257	-0.5184	0.0017
EMR(-4)/VMR(-4)	-0.0001	-0.0828	-0.0028	0.0040	3.0745	0.0402	0.0030	0.2183	-0.0025	0.0005	0.0178	-0.0029
EMR(-5)/VMR(-5)	-0.0007	-0.5573	0.0042	0.0011	1.3195	0.0044	-0.0009	-0.0831	-0.0028	-0.0060	-0.3112	-0.0018
EMR(-6)/VMR(-6)	0.0005	0.7559	0.0034	0.0024	3.6615	0.0601	0.0039	0.5948	0.0004	0.0108	0.7525	0.0043
EMR(-7)/VMR(-7)	0.0005	0.9923	0.0078	0.0015	2.0057	0.0378	0.0024	0.4409	-0.0007	0.0076	0.7001	0.0035
EMR(-8)/VMR(-8)	0.0009	2.9985	0.0615	0.0018	3.1851	0.1086	0.0081	2.4672	0.0357	0.0145	1.7909	0.0375
EMR(-9)/VMR(-9)	0.0005	1.9795	0.0316	0.0011	2.7147	0.0693	0.0053	2.0601	0.0275	0.0086	1.5151	0.0219
EMR(-10)/VMR(-10)	0.0003	2.4180	0.0237	0.0007	2.2481	0.0476	0.0039	2.2835	0.0276	0.0052	1.1890	0.0120

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Decile value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 6 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values. Low/High return R<sub>lmh</sub> is the return spread in 'Low' portfolio and 'High' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$ . Panel B reports corresponding parameters for equal-(value-) weighted Low/High return R<sub>lmh</sub>.

	Panel A Va	lue coefficient v	Panel B Low/High return R <sub>Imb</sub>										
	Panel	A1 Foual-we	highted	Panel A	2 Value-wei	ohted	Panel B	1 Equal-weig	phted	Panel B2	Panel B2 Value-weighted		
Macroeconomic								raior 21 Equal Weighted					
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	0.0010	1.0812	0.0122	0.0020	1.8900	0.0574	0.0106	1.0924	0.0066	0.0151	1.4521	0.0115	
UEXGDP(Y)	0.0009	0.6420	0.0017	0.0023	1.4551	0.0306	0.0112	0.7461	0.0016	0.0180	1.1146	0.0058	
IP(M)	0.1829	1.0375	-0.0007	0.0369	0.2820	-0.0029	2.4609	1.3328	-0.0006	0.2663	0.1217	-0.0030	
IP(Q)	0.1285	0.5624	-0.0006	-0.0272	-0.1454	-0.0029	1.9862	0.8407	0.0002	-0.8358	-0.3320	-0.0026	
IP(H)	0.2637	1.3872	0.0186	0.1580	0.8450	0.0050	2.3342	1.2933	0.0067	0.2269	0.1167	-0.0029	
IP(Y)	0.0600	0.6811	-0.0004	0.0542	0.5176	-0.0008	1.0403	0.9351	0.0016	-0.6267	-0.3984	-0.0018	
UEXIP(Q)	0.0008	0.5932	-0.0012	-0.0002	-0.1702	-0.0029	0.0118	0.8732	-0.0005	-0.0043	-0.2757	-0.0027	
UEXIP(H)	0.0014	1.1417	0.0068	0.0006	0.5338	-0.0012	0.0156	1.1577	0.0036	-0.0013	-0.0864	-0.0030	
UEXIP(Y)	0.0009	1.1241	0.0046	0.0005	0.5889	-0.0006	0.0108	1.1233	0.0033	-0.0051	-0.3952	-0.0019	
IPM(M)	0.1376	0.8689	-0.0016	0.0978	0.7170	-0.0023	2.4189	1.5694	-0.0005	0.9027	0.5078	-0.0027	
IPM(Q)	-0.0843	-0.5043	-0.0018	-0.1258	-0.6992	-0.0003	0.8880	0.4870	-0.0022	-1.3935	-0.7008	-0.0016	
IPM(H)	0.1360	0.9313	0.0039	0.0617	0.4092	-0.0015	2.0927	1.3851	0.0064	-0.1477	-0.1002	-0.0030	
IPM(Y)	0.0588	0.7900	0.0006	0.0295	0.3768	-0.0021	1.5170	1.4909	0.0107	-0.3268	-0.2405	-0.0025	
UEXIPM(Q)	-0.0003	-0.2821	-0.0027	-0.0005	-0.4971	-0.0020	0.0080	0.7224	-0.0017	-0.0054	-0.4213	-0.0026	
UEXIPM(H)	0.0004	0.4663	-0.0021	0.0000	-0.0384	-0.0030	0.0120	1.1082	0.0016	-0.0048	-0.3937	-0.0025	
UEXIPM(Y)	0.0005	0.7154	-0.0005	0.0001	0.1711	-0.0028	0.0113	1.3452	0.0061	-0.0046	-0.4408	-0.0019	
M0 (M)	-0.0148	-0.0377	-0.0030	0.2003	0.5259	-0.0025	-0.3114	-0.0492	-0.0030	2.9526	0.4050	-0.0026	
M0 (Q)	-0.0641	-0.1584	-0.0028	-0.0001	-0.0002	-0.0030	0.9641	0.2079	-0.0028	2.4479	0.4408	-0.0021	
M0 (H)	-0.3834	-0.9401	0.0082	-0.3692	-0.8602	0.0077	-1.3695	-0.3320	-0.0022	1.4852	0.2888	-0.0023	
M0 (Y)	-0.5213	-1.6125	0.0483	-0.5634	-1.6402	0.0589	-3.2358	-0.7967	0.0083	-0.2239	-0.0472	-0.0030	
UEXM0(Q)	0.0000	-0.8288	0.0013	0.0000	-0.7256	0.0018	0.0000	-0.2730	-0.0025	0.0000	0.1475	-0.0027	
UEXM0 (H)	0.0000	-0.9746	0.0052	0.0000	-0.8809	0.0069	0.0000	-0.2358	-0.0025	0.0000	0.2635	-0.0017	
UEXM0 (Y)	0.0000	-1.0903	0.0104	0.0000	-1.0686	0.0160	0.0000	-0.2994	-0.0018	0.0000	0.3806	0.0003	
UNE (M)	-0.1709	-0.7289	0.0013	-0.0300	-0.1289	-0.0029	-2.2807	-0.8679	0.0014	2.5778	0.7241	0.0012	
UNE (Q)	-0.0689	-0.6841	0.0014	-0.0019	-0.0183	-0.0030	-1.1041	-0.8820	0.0034	0.7631	0.4440	-0.0007	
UNE (H)	-0.0421	-0.7325	0.0026	-0.0083	-0.1351	-0.0028	-0.6545	-0.8433	0.0048	0.3251	0.3123	-0.0016	
UNE (Y)	-0.0248	-0.7064	0.0038	-0.0078	-0.1817	-0.0023	-0.3062	-0.5700	0.0030	0.1086	0.1597	-0.0024	
UEXUNE(Q)	-0.0048	-0.5270	0.0000	0.0021	0.2097	-0.0024	-0.0910	-0.8039	0.0034	0.0943	0.6125	0.0021	
UEXUNE(H)	-0.0032	-0.5491	0.0007	0.0008	0.1263	-0.0027	-0.0592	-0.7952	0.0044	0.0501	0.4954	0.0009	
UEXUNE(Y)	-0.0021	-0.5742	0.0018	0.0001	0.0183	-0.0030	-0.0321	-0.6411	0.0037	0.0263	0.3983	0.0004	
CPI (M)	-0.0026	-0.1004	-0.0029	-0.0199	-0.8066	0.0005	0.2742	0.6302	0.0007	0.0192	0.0454	-0.0030	
CPI (Q)	-0.0054	-0.2623	-0.0021	-0.0188	-0.9963	0.0085	0.1952	0.5959	0.0039	0.0262	0.0739	-0.0029	
CPI (H)	-0.0091	-0.5641	0.0035	-0.0188	-1.2634	0.0256	0.1169	0.5084	0.0031	-0.0310	-0.1140	-0.0027	
CPI (Y)	-0.0113	-1.0255	0.0203	-0.0162	-1.5589	0.0464	-0.0282	-0.2157	-0.0022	-0.1084	-0.6039	0.0061	
UEXCPI(Q)	-0.0025	-0.9307	0.0083	-0.0046	-1.9752	0.0355	0.0045	0.1584	-0.0028	-0.0203	-0.6984	0.0001	

 Table 5.40 Univariate regression of value effect on individual lagged macroeconomic variables (Value-12)

UEXCPI(H)	-0.0029	-1.2113	0.0276	-0.0046	-2.2951	0.0733	0.0016	0.0678	-0.0029	-0.0188	-0.7480	0.0023
UEXCPI(Y)	-0.0028	-1.3971	0.0661	-0.0036	-1.9983	0.1139	-0.0030	-0.1671	-0.0025	-0.0156	-0.8328	0.0060
TBILL(M)	-0.0163	-0.6130	0.0000	-0.0455	-1.5655	0.0209	-0.1667	-0.4433	-0.0012	-0.5694	-1.4786	0.0124
TBILL (Q)	-0.0179	-0.9143	0.0093	-0.0415	-1.9391	0.0651	-0.1246	-0.4675	0.0004	-0.5042	-1.9527	0.0384
TBILL (H)	-0.0200	-1.2922	0.0295	-0.0349	-2.3373	0.0994	-0.1398	-0.7459	0.0061	-0.4391	-2.4767	0.0638
TBILL (Y)	-0.0111	-1.0059	0.0186	-0.0184	-1.8939	0.0583	-0.1065	-0.9563	0.0084	-0.3048	-2.6136	0.0665
UEXTBILL(Q)	-4.5149	-1.2414	0.0217	-8.5733	-2.1249	0.0890	-35.0742	-0.8563	0.0056	-80.2527	-1.8851	0.0302
UEXTBILL(H)	-4.6468	-1.4682	0.0423	-7.9143	-2.5256	0.1327	-32.6731	-0.9780	0.0098	-74.0509	-2.2999	0.0459
UEXTBILL(Y)	-3.6256	-1.3127	0.0492	-5.6138	-2.2586	0.1261	-23.6306	-0.9379	0.0097	-54.5198	-2.2032	0.0472
TERM(M)	0.2614	0.8693	0.0023	0.4796	1.6268	0.0155	2.3640	0.7252	-0.0005	4.3731	1.1557	0.0033
TERM(Q)	0.0958	0.4061	-0.0009	0.3530	1.5138	0.0257	1.1138	0.3773	-0.0014	3.6991	1.0565	0.0100
TERM(H)	0.0432	0.2472	-0.0021	0.2431	1.6354	0.0254	0.2701	0.1263	-0.0028	2.4478	0.9840	0.0089
TERM(Y)	0.0244	0.1536	-0.0024	0.0696	0.4456	0.0018	0.2840	0.1736	-0.0025	0.7128	0.3874	-0.0009
UEXTERM(Q)	0.1806	0.5513	0.0008	0.5058	1.5270	0.0275	1.9870	0.4989	-0.0004	5.1735	1.0867	0.0102
UEXTERM(H)	0.0990	0.3620	-0.0011	0.4178	1.6720	0.0317	1.0811	0.3170	-0.0017	4.3679	1.0825	0.0126
UEXTERM(Y)	0.0631	0.2640	-0.0016	0.2397	1.1401	0.0184	0.5382	0.2118	-0.0024	2.2829	0.7588	0.0050
DY(M)	-0.0499	-1.9725	0.0063	-0.0447	-1.9973	0.0047	-0.5001	-1.3697	0.0023	-0.4263	-0.9178	-0.0001
DY(Q)	-0.0444	-1.8444	0.0201	-0.0398	-1.7548	0.0162	-0.4705	-1.2838	0.0119	-0.3150	-0.6817	0.0019
DY(H)	-0.0343	-1.2489	0.0292	-0.0320	-1.2557	0.0261	-0.1625	-0.4806	0.0012	0.0003	0.0008	-0.0030
DY(Y)	-0.0073	-0.4393	0.0004	-0.0053	-0.3188	-0.0011	0.0108	0.0448	-0.0030	0.2959	1.0603	0.0211
UEXDY(Q)	-1.7552	-2.3248	0.0311	-1.6445	-2.2803	0.0279	-15.8700	-1.8326	0.0130	-10.4421	-0.8520	0.0021
UEXDY(H)	-1.7216	-1.9490	0.0553	-1.5926	-1.8763	0.0485	-11.4475	-1.2497	0.0118	-5.8620	-0.4978	-0.0001
UEXDY(Y)	-0.8763	-1.1349	0.0290	-0.7640	-0.9888	0.0221	-3.7997	-0.4613	0.0005	3.4227	0.3519	-0.0009
EMR(-1)/VMR(-1)	0.0100	1.9579	0.0226	0.0194	1.4267	0.0333	0.1720	1.7337	0.0405	-0.0012	-0.0045	-0.0030
EMR(-2)/VMR(-2)	0.0069	2.0648	0.0280	0.0103	1.3011	0.0263	0.1098	1.7593	0.0414	-0.1040	-0.6650	0.0094
EMR(-3)/VMR(-3)	0.0018	0.4948	0.0017	0.0067	1.4169	0.0291	0.0386	0.8033	0.0089	-0.0624	-0.6469	0.0085
EMR(-4)/VMR(-4)	0.0012	0.5202	0.0012	0.0076	2.5404	0.0877	0.0211	0.6579	0.0046	-0.0072	-0.1242	-0.0027
EMR(-5)/VMR(-5)	-0.0001	-0.0504	-0.0029	0.0025	1.1935	0.0175	0.0064	0.2599	-0.0014	-0.0106	-0.2779	-0.0015
EMR(-6)/VMR(-6)	0.0013	1.1220	0.0178	0.0041	2.9554	0.1044	0.0084	0.5553	0.0022	0.0113	0.4416	0.0003
EMR(-7)/VMR(-7)	0.0010	1.0644	0.0207	0.0022	1.5708	0.0476	0.0059	0.4420	0.0013	0.0027	0.1309	-0.0027
EMR(-8)/VMR(-8)	0.0021	3.4214	0.1541	0.0032	3.2291	0.1905	0.0198	2.8766	0.0747	0.0214	1.5401	0.0329
EMR(-9)/VMR(-9)	0.0012	2.8222	0.0960	0.0020	2.9899	0.1245	0.0140	2.4643	0.0701	0.0110	1.0488	0.0135
EMR(-10)/VMR(-10)	0.0008	3.2801	0.0811	0.0012	2.2828	0.0709	0.0106	2.8799	0.0759	0.0073	0.9290	0.0091

EXECUTE: For the table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> onNote: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> onindividual lagged macroeconomic variables from 1980 to 2008, respectively. Decile value portfolios are formed with shares' price-to-book values at the end of each month, andhold for 12 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period ontheir mean price-to-book values. Low/High return R<sub>lmh</sub> is the return spread in 'Low' portfolio and 'High' portfolio. t-stats is adjusted by Newey-West (1987) approach withautomatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted Low/High return R<sub>lmh</sub>.
Panel A Value coefficient y <sub>my</sub>						Panel B Low/High return R						
	Panel	A1 Foual-we	highted	Panel A	2 Value-wei	ohted	Panel B	1 Equal-weig	phted	Panel B2 Value-weighted		
Macroeconomic	T aner	TT Equal-we				giiteu		T Equal Weig				
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>
GDP(Y)	0.0028	2.3737	0.0612	0.0036	2.8658	0.1101	0.0263	1.9150	0.0267	0.0256	1.4971	0.0213
UEXGDP(Y)	0.0037	2.2471	0.0438	0.0047	2.6434	0.0787	0.0360	1.8015	0.0208	0.0330	1.3515	0.0142
IP(M)	0.2033	1.2497	-0.0015	0.0825	0.5091	-0.0028	3.0839	1.4929	-0.0012	-1.1252	-0.3538	-0.0028
IP(Q)	0.2817	1.4102	0.0031	0.0769	0.4036	-0.0026	2.6170	0.9776	-0.0003	-1.4689	-0.4133	-0.0023
IP(H)	0.4661	2.6469	0.0338	0.2974	1.5349	0.0129	3.5612	1.5096	0.0082	0.3190	0.0955	-0.0030
IP(Y)	0.3042	1.8408	0.0341	0.2852	1.5565	0.0316	3.3020	1.5752	0.0198	0.5311	0.1568	-0.0025
UEXIP(Q)	0.0013	1.1031	-0.0004	0.0002	0.1640	-0.0030	0.0136	0.8848	-0.0014	-0.0128	-0.5637	-0.0018
UEXIP(H)	0.0023	1.9583	0.0111	0.0011	0.8675	0.0004	0.0196	1.1845	0.0021	-0.0073	-0.2917	-0.0024
UEXIP(Y)	0.0023	2.0045	0.0251	0.0017	1.3314	0.0125	0.0220	1.3544	0.0099	-0.0035	-0.1395	-0.0028
IPM(M)	0.1916	1.1924	-0.0016	0.1473	0.8660	-0.0021	3.3863	1.6416	-0.0007	-0.7506	-0.2503	-0.0029
IPM(Q)	0.0914	0.4349	-0.0023	0.0182	0.0921	-0.0030	2.0938	0.7956	-0.0010	-1.1393	-0.3367	-0.0025
IPM(H)	0.3508	2.2326	0.0221	0.1951	1.0726	0.0052	4.1057	2.0430	0.0150	0.5484	0.1697	-0.0028
IPM(Y)	0.2374	1.7123	0.0290	0.1917	1.1998	0.0191	3.2510	1.8948	0.0283	0.7249	0.2440	-0.0017
UEXIPM(Q)	0.0005	0.3965	-0.0026	0.0001	0.1241	-0.0030	0.0138	0.8838	-0.0012	-0.0097	-0.4617	-0.0023
UEXIPM(H)	0.0015	1.3186	0.0037	0.0006	0.5229	-0.0017	0.0215	1.3892	0.0043	-0.0053	-0.2253	-0.0027
UEXIPM(Y)	0.0017	1.5920	0.0169	0.0011	0.9254	0.0055	0.0230	1.6741	0.0158	-0.0003	-0.0139	-0.0030
M0 (M)	-0.4035	-0.6496	-0.0021	-0.2504	-0.4180	-0.0026	-6.4440	-0.8113	-0.0017	1.1355	0.1134	-0.0030
M0 (Q)	-0.3744	-0.6659	-0.0003	-0.3605	-0.6690	-0.0003	-4.1813	-0.5820	-0.0012	2.7438	0.3203	-0.0024
M0 (H)	-0.6840	-1.3094	0.0164	-0.6737	-1.3268	0.0170	-6.8441	-1.0630	0.0071	-0.0558	-0.0070	-0.0030
M0 (Y)	-0.7370	-1.7799	0.0529	-0.7473	-2.0244	0.0580	-7.5480	-1.2473	0.0277	-2.1230	-0.3184	-0.0010
UEXM0(Q)	0.0000	-1.2535	0.0120	0.0000	-0.9504	0.0084	-0.0002	-1.0095	0.0066	0.0001	0.6245	0.0022
UEXM0 (H)	0.0000	-1.3517	0.0219	0.0000	-1.0718	0.0179	-0.0001	-1.0079	0.0108	0.0001	0.6130	0.0049
UEXM0 (Y)	0.0000	-1.4066	0.0326	0.0000	-1.2217	0.0322	-0.0001	-1.0483	0.0168	0.0001	0.5335	0.0046
UNE (M)	-0.2932	-0.7273	0.0039	-0.0648	-0.1587	-0.0027	-2.5010	-0.5270	-0.0004	1.9160	0.3128	-0.0017
UNE (Q)	-0.1411	-0.7781	0.0069	-0.0459	-0.2324	-0.0019	-1.3707	-0.6155	0.0019	0.0078	0.0026	-0.0030
UNE (H)	-0.0789	-0.7467	0.0077	-0.0295	-0.2432	-0.0015	-0.6781	-0.4844	0.0011	-0.0624	-0.0335	-0.0030
UNE (Y)	-0.0232	-0.3631	0.0002	-0.0048	-0.0629	-0.0029	0.0802	0.0794	-0.0028	0.1058	0.0920	-0.0027
UEXUNE(Q)	-0.0097	-0.5832	0.0038	-0.0008	-0.0437	-0.0030	-0.1047	-0.5179	0.0012	0.0598	0.2192	-0.0019
UEXUNE(H)	-0.0059	-0.5625	0.0040	-0.0007	-0.0553	-0.0030	-0.0605	-0.4582	0.0008	0.0320	0.1780	-0.0021
UEXUNE(Y)	-0.0023	-0.3622	0.0003	0.0003	0.0380	-0.0030	-0.0098	-0.1115	-0.0027	0.0291	0.2570	-0.0007
CPI (M)	-0.0094	-0.3107	-0.0026	-0.0254	-0.8768	0.0001	0.1771	0.3493	-0.0023	-0.0892	-0.1366	-0.0029
CPI (Q)	-0.0109	-0.4732	-0.0010	-0.0282	-1.3075	0.0115	0.0895	0.2377	-0.0023	-0.1475	-0.3178	-0.0014
CPI (H)	-0.0164	-0.8972	0.0085	-0.0254	-1.4979	0.0261	-0.0191	-0.0767	-0.0030	-0.1398	-0.4106	0.0007
CPI (Y)	-0.0180	-1.3188	0.0293	-0.0189	-1.5220	0.0347	-0.1459	-0.9724	0.0080	-0.1260	-0.5300	0.0041
UEXCPI(Q)	-0.0028	-0.7342	0.0047	-0.0053	-1.5710	0.0254	0.0027	0.0688	-0.0030	-0.0223	-0.4819	-0.0009

 Table 5.41 Univariate regression of value effect on individual lagged macroeconomic variables (Value-18)

#### Continued

\_

UEXCPI(H)	-0.0029	-0.8431	0.0135	-0.0048	-1.5487	0.0433	-0.0007	-0.0195	-0.0030	-0.0195	-0.4755	0.0003
UEXCPI(Y)	-0.0029	-0.9865	0.0375	-0.0033	-1.2150	0.0501	-0.0058	-0.2135	-0.0022	-0.0112	-0.3503	-0.0004
TBILL(M)	-0.0213	-0.6880	-0.0003	-0.0403	-1.4043	0.0074	-0.3380	-0.8382	0.0006	-0.6699	-1.4633	0.0093
TBILL (Q)	-0.0193	-0.8618	0.0048	-0.0355	-1.5688	0.0249	-0.2131	-0.7326	0.0019	-0.4935	-1.5121	0.0200
TBILL (H)	-0.0219	-1.0883	0.0182	-0.0326	-1.8976	0.0469	-0.1830	-0.7995	0.0047	-0.4088	-1.7597	0.0304
TBILL (Y)	-0.0140	-0.9833	0.0154	-0.0165	-1.3177	0.0242	-0.1871	-1.2646	0.0143	-0.2589	-1.6308	0.0256
UEXTBILL(Q)	-4.1122	-0.8852	0.0082	-6.7480	-1.5505	0.0290	-44.0584	-0.8906	0.0037	-73.2156	-1.3469	0.0130
UEXTBILL(H)	-4.4613	-1.0120	0.0198	-6.6198	-1.6927	0.0502	-39.2856	-0.8629	0.0062	-65.2839	-1.3838	0.0190
UEXTBILL(Y)	-3.7053	-0.9635	0.0267	-4.7727	-1.4275	0.0492	-31.2491	-0.8271	0.0080	-46.7509	-1.2320	0.0183
TERM(M)	0.0806	0.2333	-0.0028	0.1503	0.5120	-0.0020	2.9314	0.6807	-0.0011	2.9287	0.6162	-0.0014
TERM(Q)	-0.0575	-0.1730	-0.0026	0.0527	0.1737	-0.0027	1.2527	0.2851	-0.0020	1.3846	0.2795	-0.0020
TERM(H)	-0.0472	-0.1753	-0.0025	0.0202	0.0815	-0.0029	0.3331	0.0922	-0.0029	0.6898	0.1733	-0.0025
TERM(Y)	0.0236	0.1122	-0.0028	-0.0382	-0.1931	-0.0022	1.3720	0.4630	0.0022	-0.2852	-0.1024	-0.0029
UEXTERM(Q)	-0.0217	-0.0501	-0.0030	0.1111	0.2938	-0.0022	2.4430	0.4313	-0.0011	2.6290	0.4131	-0.0011
UEXTERM(H)	-0.0527	-0.1303	-0.0028	0.0670	0.1809	-0.0025	1.3770	0.2559	-0.0020	1.7306	0.2860	-0.0016
UEXTERM(Y)	0.0143	0.0417	-0.0030	0.0059	0.0186	-0.0030	1.5623	0.3382	-0.0005	0.3687	0.0766	-0.0029
DY(M)	-0.0394	-0.9450	0.0001	-0.0288	-0.8998	-0.0013	-0.0790	-0.1062	-0.0030	0.1014	0.1419	-0.0030
DY(Q)	-0.0266	-0.7419	0.0015	-0.0242	-0.7906	0.0009	-0.1213	-0.1789	-0.0026	0.0763	0.1107	-0.0029
DY(H)	-0.0283	-0.8173	0.0089	-0.0156	-0.5187	0.0008	-0.0397	-0.0796	-0.0029	0.3598	0.6336	0.0057
DY(Y)	-0.0126	-0.6235	0.0025	-0.0035	-0.1648	-0.0026	-0.0475	-0.1499	-0.0026	0.3829	0.9847	0.0204
UEXDY(Q)	-1.2326	-0.9790	0.0062	-1.0650	-0.9578	0.0042	-6.3192	-0.3737	-0.0018	0.6190	0.0324	-0.0030
UEXDY(H)	-1.2105	-0.9170	0.0127	-0.8266	-0.6863	0.0047	-3.9010	-0.2433	-0.0022	6.9750	0.3763	-0.0007
UEXDY(Y)	-0.7925	-0.7644	0.0113	-0.3484	-0.3284	-0.0001	-1.4999	-0.1209	-0.0028	11.3341	0.7667	0.0102
EMR(-1)/VMR(-1)	0.0165	2.1673	0.0351	0.0276	1.3264	0.0381	0.2665	2.0681	0.0490	-0.0311	-0.0817	-0.0028
EMR(-2)/VMR(-2)	0.0114	2.2552	0.0426	0.0164	1.2548	0.0390	0.1778	1.9607	0.0550	-0.1286	-0.5559	0.0079
EMR(-3)/VMR(-3)	0.0039	0.7720	0.0087	0.0125	1.5350	0.0597	0.0689	0.9615	0.0158	-0.0608	-0.4252	0.0033
EMR(-4)/VMR(-4)	0.0020	0.5428	0.0032	0.0109	2.2750	0.1024	0.0408	0.7698	0.0111	-0.0060	-0.0714	-0.0029
EMR(-5)/VMR(-5)	0.0008	0.2805	-0.0008	0.0053	1.6866	0.0486	0.0203	0.5890	0.0050	-0.0005	-0.0085	-0.0030
EMR(-6)/VMR(-6)	0.0019	0.9619	0.0226	0.0055	2.4004	0.1020	0.0157	0.6647	0.0060	0.0086	0.2260	-0.0020
EMR(-7)/VMR(-7)	0.0025	1.7425	0.0737	0.0038	2.0335	0.0850	0.0232	1.2732	0.0301	0.0100	0.3456	-0.0005
EMR(-8)/VMR(-8)	0.0038	3.8098	0.2643	0.0045	2.8305	0.2089	0.0406	3.6509	0.1582	0.0299	1.3724	0.0365
EMR(-9)/VMR(-9)	0.0024	4.1941	0.1994	0.0030	2.7602	0.1611	0.0294	3.5708	0.1557	0.0179	1.1073	0.0221
EMR(-10)/VMR(-10)	0.0013	3.2628	0.1172	0.0016	2.0076	0.0733	0.0194	3.3974	0.1286	0.0110	0.9073	0.0128

Link(-10)0.0150.1280.1120.0102.00700.01530.01943.59740.12800.01100.90730.0128<t

	Panel A Value coefficient $\gamma_{mv}$						Panel B Low/High return R <sub>lmh</sub>						
	Panel	A1 Equal-we	eighted	Panel A	A2 Value-wei	ghted	Panel B	Panel B1 Equal-weighted			Panel B2 Value-weighted		
Macroeconomic													
variable	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t (λ)	Adj-R <sup>2</sup>	λ	t(λ)	Adj-R <sup>2</sup>	
GDP(Y)	0.0030	1.6870	0.0405	0.0041	2.3608	0.0898	0.0304	1.7187	0.0264	0.0294	1.2430	0.0191	
UEXGDP(Y)	0.0044	1.8718	0.0365	0.0055	2.3641	0.0701	0.0468	1.8955	0.0269	0.0411	1.1650	0.0156	
IP(M)	0.7453	2.0409	0.0090	0.4958	1.5656	0.0031	7.5164	2.1778	0.0051	3.5219	0.7500	-0.0017	
IP(Q)	1.0541	2.7259	0.0473	0.6370	1.7236	0.0182	10.5996	2.4685	0.0308	3.6475	0.5951	0.0001	
IP(H)	1.0409	3.5097	0.1042	0.6489	2.0995	0.0451	10.5967	3.1874	0.0709	3.6046	0.6539	0.0038	
IP(Y)	0.5125	2.6424	0.0583	0.4363	1.8702	0.0484	6.2408	2.5208	0.0576	1.7840	0.3845	0.0009	
UEXIP(Q)	0.0055	2.1673	0.0265	0.0032	1.3555	0.0085	0.0560	2.1095	0.0171	0.0166	0.4428	-0.0017	
UEXIP(H)	0.0067	2.6415	0.0638	0.0039	1.5489	0.0234	0.0688	2.4921	0.0439	0.0163	0.3828	-0.0010	
UEXIP(Y)	0.0050	2.5709	0.0718	0.0034	1.5688	0.0360	0.0551	2.3819	0.0572	0.0095	0.2402	-0.0016	
IPM(M)	0.6188	1.6138	0.0058	0.5886	1.5145	0.0062	6.6211	1.8866	0.0036	3.6404	0.7069	-0.0015	
IPM(Q)	0.6097	1.8659	0.0165	0.5497	1.4069	0.0153	7.3984	2.0624	0.0161	3.3094	0.5233	0.0000	
IPM(H)	0.6559	2.4095	0.0481	0.4938	1.4994	0.0305	8.0672	2.7392	0.0485	3.2722	0.5916	0.0038	
IPM(Y)	0.2996	1.6756	0.0266	0.2897	1.2958	0.0291	4.4563	1.9872	0.0407	1.6216	0.4064	0.0016	
UEXIPM(Q)	0.0034	1.4932	0.0093	0.0030	1.1931	0.0083	0.0415	1.7782	0.0093	0.0158	0.4062	-0.0017	
UEXIPM(H)	0.0041	1.8403	0.0258	0.0031	1.1915	0.0166	0.0504	2.1810	0.0266	0.0141	0.3296	-0.0012	
UEXIPM(Y)	0.0029	1.7264	0.0295	0.0023	1.1438	0.0219	0.0387	2.0255	0.0362	0.0098	0.2766	-0.0011	
M0 (M)	-0.3629	-0.4819	-0.0026	-0.2076	-0.3451	-0.0029	-10.1772	-0.9884	-0.0007	-0.6543	-0.0704	-0.0031	
M0 (Q)	-0.4049	-0.5768	-0.0012	-0.3572	-0.6590	-0.0014	-6.0870	-0.6390	-0.0003	2.9492	0.3677	-0.0026	
M0 (H)	-0.6820	-1.0286	0.0080	-0.5817	-1.2335	0.0062	-9.1714	-1.0863	0.0102	2.1363	0.2549	-0.0025	
M0 (Y)	-1.0427	-1.9280	0.0615	-0.9309	-2.3715	0.0565	-11.2019	-1.5633	0.0466	-2.0070	-0.2715	-0.0018	
UEXM0(Q)	0.0000	-1.2772	0.0172	0.0000	-1.0214	0.0138	-0.0003	-1.0671	0.0139	0.0002	0.5669	0.0017	
UEXM0 (H)	0.0000	-1.3980	0.0301	0.0000	-1.0810	0.0245	-0.0002	-1.1350	0.0222	0.0002	0.6191	0.0063	
UEXM0 (Y)	0.0000	-1.5226	0.0462	0.0000	-1.2171	0.0414	-0.0002	-1.2245	0.0335	0.0001	0.5969	0.0084	
UNE (M)	-0.3300	-0.4982	0.0020	-0.0842	-0.1282	-0.0027	-2.0687	-0.2652	-0.0018	0.1895	0.0200	-0.0031	
UNE (Q)	-0.1146	-0.3914	0.0007	-0.0284	-0.0919	-0.0028	-0.4845	-0.1260	-0.0026	-0.1844	-0.0409	-0.0031	
UNE (H)	-0.0412	-0.2511	-0.0014	-0.0044	-0.0250	-0.0031	0.0520	0.0225	-0.0031	-0.0512	-0.0196	-0.0031	
UNE (Y)	0.0125	0.1409	-0.0026	0.0222	0.2186	-0.0011	0.5328	0.3805	0.0036	0.2385	0.1596	-0.0020	
UEXUNE(Q)	-0.0041	-0.1636	-0.0024	0.0033	0.1239	-0.0026	-0.0166	-0.0527	-0.0030	0.0760	0.1961	-0.0018	
UEXUNE(H)	-0.0008	-0.0523	-0.0030	0.0031	0.1800	-0.0018	0.0135	0.0656	-0.0030	0.0564	0.2279	-0.0011	
UEXUNE(Y)	0.0017	0.1801	-0.0021	0.0034	0.3306	0.0017	0.0395	0.3123	0.0007	0.0522	0.3530	0.0022	
CPI (M)	-0.0191	-0.5210	-0.0021	-0.0227	-0.6669	-0.0015	0.0529	0.1107	-0.0031	0.1014	0.1357	-0.0030	
CPI (Q)	-0.0224	-0.7746	0.0019	-0.0269	-1.0204	0.0053	0.0079	0.0217	-0.0031	-0.0219	-0.0393	-0.0031	
CPI (H)	-0.0303	-1.1681	0.0196	-0.0296	-1.3777	0.0220	-0.1156	-0.4184	-0.0009	-0.1480	-0.3494	-0.0002	
CPI (Y)	-0.0358	-1.5974	0.0693	-0.0289	-1.6652	0.0513	-0.3183	-1.6412	0.0350	-0.2049	-0.6806	0.0096	
UEXCPI(Q)	-0.0026	-0.4914	0.0006	-0.0030	-0.6426	0.0028	0.0126	0.2644	-0.0025	0.0103	0.1704	-0.0028	
. 2													

 Table 5.42 Univariate regression of value effect on individual lagged macroeconomic variables (Value-24)

#### Continued

\_

UEXCPI(H)	-0.0041	-0.8929	0.0156	-0.0037	-0.8842	0.0144	-0.0010	-0.0239	-0.0031	-0.0019	-0.0351	-0.0031
UEXCPI(Y)	-0.0054	-1.4828	0.0780	-0.0041	-1.2885	0.0492	-0.0230	-0.7188	0.0066	-0.0175	-0.4621	0.0014
TBILL(M)	-0.0282	-0.9523	-0.0003	-0.0315	-0.8502	0.0010	-0.5004	-1.3739	0.0028	-0.6539	-1.3689	0.0051
TBILL (Q)	-0.0266	-1.1514	0.0055	-0.0315	-1.1333	0.0109	-0.3389	-1.1505	0.0062	-0.4812	-1.3958	0.0120
TBILL (H)	-0.0291	-1.1645	0.0186	-0.0327	-1.4570	0.0285	-0.3042	-1.0755	0.0126	-0.4170	-1.5972	0.0208
TBILL (Y)	-0.0263	-1.1436	0.0347	-0.0262	-1.3880	0.0404	-0.3280	-1.4636	0.0360	-0.3515	-1.7020	0.0331
UEXTBILL(Q)	-5.5393	-1.2188	0.0088	-5.6594	-1.0685	0.0112	-66.2947	-1.3376	0.0082	-71.1897	-1.2046	0.0074
UEXTBILL(H)	-6.0255	-1.2576	0.0211	-6.2613	-1.2918	0.0272	-58.3488	-1.1501	0.0120	-67.1615	-1.3058	0.0131
UEXTBILL(Y)	-6.2633	-1.2800	0.0462	-6.1978	-1.4459	0.0528	-57.7557	-1.2519	0.0248	-63.7168	-1.5366	0.0243
TERM(M)	0.1599	0.4822	-0.0025	0.1768	0.6130	-0.0022	3.6276	0.7466	-0.0009	2.5411	0.4610	-0.0022
TERM(Q)	0.0727	0.2344	-0.0027	0.0537	0.1762	-0.0029	3.4387	0.7010	0.0025	1.6030	0.2977	-0.0021
TERM(H)	-0.0072	-0.0244	-0.0031	0.0229	0.0865	-0.0030	1.4556	0.3056	-0.0010	1.0829	0.2464	-0.0022
TERM(Y)	0.0706	0.2809	-0.0016	0.0377	0.1802	-0.0026	2.4214	0.6407	0.0090	1.3641	0.4364	0.0000
UEXTERM(Q)	0.1226	0.2969	-0.0026	0.1117	0.2903	-0.0026	4.7564	0.7501	0.0025	2.7267	0.3856	-0.0016
UEXTERM(H)	0.0580	0.1436	-0.0029	0.0722	0.1877	-0.0027	3.5084	0.5402	0.0019	2.1224	0.3186	-0.0016
UEXTERM(Y)	0.0741	0.1966	-0.0025	0.0490	0.1487	-0.0028	3.2398	0.5440	0.0049	1.6821	0.3151	-0.0014
DY(M)	-0.0759	-1.2932	0.0037	-0.0380	-0.7987	-0.0011	-0.4088	-0.5773	-0.0018	0.2074	0.2365	-0.0028
DY(Q)	-0.0681	-1.2865	0.0143	-0.0363	-0.8039	0.0026	-0.5066	-0.7930	0.0033	0.1089	0.1289	-0.0029
DY(H)	-0.0648	-1.4323	0.0337	-0.0298	-0.7944	0.0059	-0.4606	-0.9345	0.0093	0.2901	0.4142	0.0009
DY(Y)	-0.0398	-1.3433	0.0297	-0.0174	-0.6137	0.0042	-0.3174	-0.8196	0.0108	0.4202	0.8032	0.0165
UEXDY(Q)	-2.6890	-1.4141	0.0225	-1.3557	-0.8046	0.0044	-17.7547	-0.9282	0.0043	2.1539	0.0885	-0.0030
UEXDY(H)	-2.8373	-1.5743	0.0475	-1.4023	-0.8808	0.0112	-18.0334	-0.9975	0.0105	4.5720	0.1977	-0.0024
UEXDY(Y)	-2.2529	-1.6570	0.0645	-1.0735	-0.8435	0.0147	-14.9481	-1.0434	0.0167	7.1220	0.3812	0.0005
EMR(-1)/VMR(-1)	0.0234	1.9552	0.0415	0.0380	1.1732	0.0466	0.3487	2.1249	0.0631	-0.0424	-0.0845	-0.0028
EMR(-2)/VMR(-2)	0.0131	1.9425	0.0318	0.0246	1.1958	0.0574	0.2079	1.9177	0.0558	-0.1511	-0.4541	0.0075
EMR(-3)/VMR(-3)	0.0049	0.7110	0.0074	0.0201	1.6224	0.1000	0.0856	0.9318	0.0185	-0.0438	-0.2126	-0.0008
EMR(-4)/VMR(-4)	0.0021	0.4062	0.0010	0.0142	1.9222	0.1098	0.0462	0.6527	0.0103	0.0109	0.0888	-0.0028
EMR(-5)/VMR(-5)	0.0012	0.3219	-0.0001	0.0082	1.7221	0.0753	0.0183	0.4340	0.0017	0.0152	0.1940	-0.0019
EMR(-6)/VMR(-6)	0.0028	0.8961	0.0279	0.0064	1.7445	0.0847	0.0188	0.5790	0.0064	0.0043	0.0768	-0.0029
EMR(-7)/VMR(-7)	0.0042	2.0161	0.1179	0.0058	1.9922	0.1204	0.0381	1.5991	0.0622	0.0237	0.5632	0.0065
EMR(-8)/VMR(-8)	0.0055	3.4310	0.3195	0.0060	2.3812	0.2311	0.0576	3.5378	0.2365	0.0378	1.1540	0.0399
EMR(-9)/VMR(-9)	0.0033	3.5553	0.2185	0.0040	2.3790	0.1792	0.0415	3.9979	0.2296	0.0251	1.0278	0.0305
EMR(-10)/VMR(-10)	0.0016	2.8437	0.0961	0.0019	1.6387	0.0686	0.0247	3.1107	0.1541	0.0116	0.6585	0.0090

Note: The table reports parameters of slope coefficient  $\lambda$ , t-stats, and adjusted R-square estimated from univariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> on individual lagged macroeconomic variables from 1980 to 2008, respectively. Decile Value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 24 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values. Low/High return R<sub>lmh</sub> is the return spread in 'Low' portfolio and 'High' portfolio. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A reports corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$ . Panel B reports corresponding parameters for equal-(value-) weighted Low/High return R<sub>lmh</sub>.

		Panel A	3 months' h	nolding lengt	th			
		Value coeff	ficient $\gamma_{pb}$			Low/High re	eturn R <sub>lmh</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted
Lag length	8		4		7		7	
Adj-R <sup>2</sup>	0.16	53	0.21	09	0.06	80	0.07	33
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
UEXTBILL(Y)	-0.9744	-1.3988	-1.3099	-1.9929	-5.4310	-0.6991	-8.1685	-0.938
UEXDY(H)	-0.8601	-2.4861	-0.8300	-2.8089	-6.8654	-1.9560	-9.1683	-2.327
EMR/VMR(-8)	0.0004	2.4600	0.0007	3.1855	0.0032	1.8318	0.0066	1.678
		Panel P	6 months' h	olding lengt	ħ			
		Value coef	ficient $\gamma_{\rm ph}$	ionaling lenge	.11	Low/High re	eturn R <sub>lmh</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted
Lag length	8		8		7	0	12	0
Adj-R <sup>2</sup>	0.23	25	0.31	08	0.08	92	0.11	13
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
JEXTBILL(Y)	-2.3917	-1.5954	-3.7346	-2.4620	-11.6825	-0.7108	-28.0779	-1.667
JEXDY(H)	-1.3725	-2.5143	-1.3074	-2.8807	-10.7742	-1.6781	-11,9846	-1.524
EMR/VMR(-8)	0.0008	3.2004	0.0015	3.3929	0.0076	2.5169	0.0122	1.616
		Derest C	10	1	d.			
		Value coeff	ficient $\gamma_{\rm ph}$	notaing teng	un	Low/High re	eturn R <sub>lmb</sub>	
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted
Lag length	. 7	C	8	e	. 9	e	10	U
Adj-R <sup>2</sup>	0.22	36	0.29	69	0.08	69	0.07	23
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)
JEXTBILL(Y)	-2.8615	-1.3777	-4.6107	-2.5976	-19.1502	-0.7703	-54.2589	-2.004
IEXDY(H)	-1 0797	-1 5873	-0 5994	-0.9391	-6 6187	-0.7125	4 4072	0 339
EMR/VMR(-8)	0.0021	3.8881	0.0029	3.7004	0.0195	3.0229	0.0197	1.523
		Panel D	18 months?	holding leng	rth			
		Value coeff	ficient v .	norung ieng	,ui	Low/High r	aturn R.	
	Faual-we	value coell	Value-we	aighted	Faual-we	LOW/HIGH N	Value-wa	and
ag length	14	Ignicu	value-we	Ignicu	13	Ignicu	v anuc-we	Ignicu
Ad; $\mathbf{p}^2$	0.20	22	0.28	07	0.16	78	0.05	01
Macroeconomic variables	λ	τ(λ)	λ	t(λ)	λ	τ(λ)	λ	τ())
SDP(V)	0 0010	2 0420	0 0020	(A) 2 7470	0.0170	1 1/15	0 0213	1 221
EMR/VMR(-8)	0.0019	3.8351	0.0029	2.9723	0.0388	3.2194	0.0213	1.231
		Danal F	24 months?	holding long	rth			
		Value coeff	ficient v_	notunig telig	501	Low/High re	eturn R1	
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted
Lag length	10	-0	20	)	14		18	-5
Adi-R <sup>2</sup>	0.40	87	0.33	00	0.28	44	0.06	17
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ.	· t(λ)
GDP(Y)	0.0019	1 4245	0.0034	2 3180	0.0175	0.9134	0.0269	1 161
$MO(\mathbf{Y})$	-0 3050	-0 7060	-0 1388	-0 4251	-4 0653	-0 5764	4 0182	0.616
$PI(\mathbf{Y})$	-0.03/0	-0.7000	-0.1300	_1 5001	-0.2002	-0.5704	-0.2225	-0.834
	-0.0340	3 7060	0.0203	-1.J771 2 26/1	0.2902	3 3/86	0.2323	1 120
$\overline{\mathrm{SMR}}/\mathrm{VMR}(-8)$	1111151			2)UH I	0.0337	5.5400	0.0505	1.130

values in the formation time. Low/High return  $R_{lmh}$  is the return spread between 'Low' portfolio and 'High' portfolio. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXDY(H) is unexpected change of DY in previous 6 months. EMR/VMR(-8) is equal-(value-) weighted market return in past eight years. GDP(Y) is pervious annual change of GDP. M0(Y) is the previous annual change of M0. CPI(Y) is the previous annual change of CPI. tstats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to E report corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$  with 3, 6, 12, 18, and 24 months' holding lengths, respectively.

Table 5.44         Multivariate regression	of value effect on the	combination of lagged
macroeconomic variables in sub	periods (Value: 3 mon	ths' holding length)

mueroet					•						
	Panel A April 1980-December 1989										
		Value coef	ficient $\gamma_{pb}$			Low/High	return R <sub>lmh</sub>				
	Equal-we	eighted	Value-we	eighted	Equal-we	eighted	Value-we	eighted			
Lag length	4		4		6		1				
Adj-R <sup>2</sup>	0.34	58	0.3637		0.1990		0.209	90			
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)			
UEXTBILL(Y)	0.3150	0.3573	-0.5997	-0.6139	0.5518	0.0928	-5.5474	-0.7205			
UEXDY(H)	-1.9466	-4.2300	-1.7223	-3.5530	-9.3593	-2.6609	-10.0445	-2.8329			
EMR/VMR(-8)	0.0015	3.0155	0.0015	2.8586	0.0111	3.5900	0.0080	2.2018			
		Panel B	January 1990	December	1999						
		Value coeff	ficient $\gamma_{pb}$			Low/High 1	return R <sub>lmh</sub>				
	Equal-w	eighted	Value-w	reighted	Equal-w	veighted	Value-w	eighted			
Lag length	2		2	2	2	2	2				
Adj-R <sup>2</sup>	0.08	73	0.03	326	0.0	685	0.14	88			
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)			
UEXTBILL(Y)	-0.4720	-0.8079	-0.5600	-0.6916	-11.9451	-1.0729	-24.3219	-1.7851			
UEXDY(H)	-0.1774	-0.9180	-0.1579	-0.7210	-7.0221	-1.5921	-12.0919	-2.5011			
EMR/VMR(-8)	0.0006	2.5234	0.0005	2.1374	0.0063	1.7212	0.0123	2.4359			
				_							
		Panel C	January 2000	December 2	2008						
		Value coef	ficient $\gamma_{pb}$			Low/High	return R <sub>lmh</sub>				
	Equal-w	eighted	Value-w	reighted	Equal-w	veighted	Value-w	eighted			
Lag length	2		4	Ļ	2	2	5				
Adj-R <sup>2</sup>	0.01	64	0.01	183	0.0428		0.0887				
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)			
UEXTBILL(Y)	0.0138	0.0064	0.2529	0.2082	-0.4842	-0.0112	67.4495	1.5857			
UEXDY(H)	0.1216	0.3579	-0.0322	-0.1409	-3.2037	-0.4747	7.1942	0.6454			
EMR/VMR(-8)	0.0007	1.1125	0.0006	1.5395	0.0171	1.2084	0.0362	2.1570			

		Panel D	April 1980-I	December 200	)8				
		Value coef	ficient $\gamma_{pb}$		Low/High return R <sub>lmh</sub>				
	Equal-we	eighted	Value-weighted		Equal-we	eighted	Value-weighted		
Lag length	8		4		7		7		
Adj-R <sup>2</sup>	0.16	53	0.2109		0.0680		0.0733		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	-0.9744	-1.3988	-1.3099	-1.9929	-5.4310	-0.6991	-8.1685	-0.9382	
UEXDY(H)	-0.8601	-2.4861	-0.8300	-2.8089	-6.8654	-1.9560	-9.1683	-2.3279	
EMR/VMR(-8)	0.0004	2.4600	0.0007	3.1855	0.0032	1.8318	0.0066	1.6789	

Note: This table presents slope coefficient  $\lambda_i$  t-stats, and adjusted R<sup>2</sup> for multivariate regression of value coefficient  $\gamma_{pb}$ and Low/High return R<sub>lmh</sub> on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 3months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values in the formation time. Low/High return R<sub>lmh</sub> is the return spread between 'Low' portfolio and 'High' portfolio. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXDY(H) is unexpected change of DY in previous 6 months. EMR/VMR(-8) is equal-(value-) weighted market return in past eight years. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> with three subperiods and overall period, respectively.

# Table 5.45 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 6 months' holding length)

Note: This table presents slope coefficient  $\lambda_i$  t-stats, and adjusted R<sup>2</sup> for multivariate regression of value coefficient  $\gamma_{pb}$ and Low/High return  $R_{lmh}$  on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 6 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values in the formation time. Low/High return  $R_{lmh}$  is the return spread between 'Low' portfolio and 'High' portfolio. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXDY(H) is unexpected change of DY in previous 6 months. EMR/VMR(-8) is equal-(value-) weighted market return in past eight years. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return  $R_{lmh}$  with three subperiods and overall period, respectively.

		Panel A	January 198	1-December	1989	8	8 /		
		Value coef	ficient $\gamma_{pb}$	. 200011001	Low/High return R <sub>lmh</sub>				
	Equal-we	eighted	Value-w	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	. 4	-	6	-	- 7	-	7	-	
$Adj-R^2$	0.38	64	0.39	35	0.28	19	0.22	64	
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	-0.5862	-0.2034	-4.8487	-1.9684	4.4282	0.1616	-19.3486	-0.8402	
UEXDY(H)	-2.8549	-2.2947	-1.0944	-0.9992	-12.5707	-1.2877	0.3686	0.0340	
EMR/VMR(-8)	0.0052	3.5327	0.0045	2.4057	0.0465	3.0950	0.0360	2.1629	
		Panel B	January 1990	0-December	1999				
		Value coef	ficient v			Low/High	return R <sub>lmb</sub>		
	Equal-we	eighted	Value-w	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	5	-8	7	8	6	8	6	8	
$Adj-R^2$	0.1655 0.1421			21	0.10	75	0.26	82	
Macroeconomic	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	-3.1885	-1.9822	-3.6444	-1.6492	-70.2238	-2.1315	-	-2.6915	
UEXDY(H)	0.1844	0.4386	-0.1680	-0.2975	-4.1792	-0.3848	-16.1873	-1.5918	
EMR/VMR(-8)	0.0013	3.0410	0.0014	1.9473	0.0228	2.3391	0.0491	3.1700	
Panel C. January 2000-December 2008									
		Value coef	fficient $\gamma_{\rm ph}$			Low/High	return R <sub>lmb</sub>		
	Equal-w	eighted	Value-w	reighted	Equal-w	eighted	Value-w	eighted	
Lag length	. 9	U U	8				8	U	
Adj-R <sup>2</sup>	0.21	66	0.09	980	0.25	539	0.04	88	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	10.3643	1.4359	4.9549	1.3665	211.228	1.4866	118.903	1.2189	
UEXDY(H)	1.9151	1.3885	1.3112	1.4507	32.0804	1.1014	50.2162	1.4031	
EMR/VMR(-8)	0.0022	1.3341	0.0015	1.6219	0.0706	2.2188	0.0549	1.2510	
		Panel D	January 1981	1-December	2008				
		Value coef	fficient ypb			Low/High	return R <sub>lmh</sub>		
	Equal-w	eighted	Value-w	reighted	Equal-w	eighted	Value-w	eighted	
Lag length	7		8	5	9		10		
Adj-R <sup>2</sup>	0.22	.36	0.29	969	0.08	369	0.07	23	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
UEXTBILL(Y)	-2.8615	-1.3777	-4.6107	-2.5976	-19.1502	-0.7703	-54.2589	-2.0045	
UEXDY(H)	-1.0797	-1.5873	-0.5994	-0.9391	-6.6187	-0.7125	4.4072	0.3393	
EMR/VMR(-8)	0.0021	3.8881	0.0029	3,7004	0.0195	3.0229	0.0197	1.5231	

# Table 5.46 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 12 months' holding length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 12 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values in the formation time. Low/High return R<sub>lmh</sub> is the return spread between 'Low' portfolio and 'High' portfolio. UEXTBILL(Y) is unexpected change of TBILL in previous 12 months. UEXDY(H) is unexpected change of DY in previous 6 months. EMR/VMR(-8) is equal-(value-) weighted market return in past eight years. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> with three subperiods and overall period, respectively.

		Panel A	A July 1981-1	December 19	89		0 /		
		Value coef	ficient $\gamma_{pb}$			Low/High 1	eturn R <sub>lmh</sub>		
	Equal-we	eighted	Value-we	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	3		7		8		10	)	
Adj-R <sup>2</sup>	0.39	75	0.34	27	0.4005		0.3257		
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	0.0009	0.5524	0.0033	1.8139	0.0073	0.4357	0.0257	1.8931	
EMR/VMR(-8)	0.0084	4.4498	0.0052	2.2142	0.0768	4.2253	0.0490	1.8642	
Panel B. January 1990-December 1999									
		Value coef	ficient $\gamma_{ab}$		Low/High return R				
	Equal-we	eighted	Value-we	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	9	0	10	)	10	)	10	)	
Adj-R <sup>2</sup>	0.00	06	0.00	21	-0.01	133	-0.00	)20	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0002	-0.1754	0.0007	0.4773	0.0089	0.3677	-0.0010	-0.0253	
EMR/VMR(-8)	0.0005	0.9141	0.0010	1.2503	0.0021	0.1884	0.0219	0.8657	
Panel C. January 2000-December 2008									
		Value coef	ficient $\gamma_{ab}$			Low/High 1	eturn R <sub>1</sub>		
	Equal-we	eighted	Value-we	eighted	Equal-w	eighted	Value-w	eighted	
Lag length	. 11		11		10	)	10	)	
$Adj-R^2$	0.13	65	0.00	91	0.20	159	-0.00	010	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0039	-1.5463	-0.0014	-1.0057	-0.1100	-1.5563	-0.0363	-0.5154	
EMR/VMR(-8)	0.0043	1.9307	0.0006	0.3328	0.1493	2.9694	0.0075	0.1083	
		Danal	D Inte 1091 I	December 20	00				
		Value agef	D July 1981-1	December 20	08	Low/Ilioh	atum D		
	Equal w	value coel	Volue w	aightad	Equal w	LOW/HIGH I	Volue w	aightad	
Lag length	Equal-we	eigineu	value-we	angineu a	Equal-w	R	value-w		
Adi-R <sup>2</sup>	0.29	22	0.28	, 07	0.16	, 578	0.05	01	
Macroeconomic variables	λ	 t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	0.0019	2.0429	0.0029	2.7470	0.0170	1.1415	0.0213	1.2310	
EMR/VMR(-8)	0.0036	3.8351	0.0041	2.9723	0.0388	3.2194	0.0270	1.2306	

# Table 5.47 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 18 months' holding length)

Note: This table presents slope coefficient  $\lambda_i$  t-stats, and adjusted R<sup>2</sup> for multivariate regression of value coefficient  $\gamma_{pb}$ and Low/High return R<sub>lmh</sub> on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 18 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values in the formation time. Low/High return R<sub>lmh</sub> is the return spread between 'Low' portfolio and 'High' portfolio. GDP(Y) is the previous annual change of GDP. EMR/VMR(-8) is equal-(value-) weighted market return in past eight years. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub>, with three subperiods and overall period, respectively.

macroe	conomic va	riables in s	ubperiods	(value: 2	4 months'	notaing let	ngtn)		
		Panel A	January 1982	-December 1	989				
		Value coeff	icient $\gamma_{pb}$			Low/High re	eturn R <sub>lmh</sub>		
	Equal-we	ighted	Value-wei	ighted	Equal-we	eighted	Value-w	eighted	
Lag length	3		13		7		14		
$Adj-R^2$	0.591	8	0.476	2	0.48	32	0.51	.33	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0007	-0.4057	0.0043	2.2031	-0.0060	-0.2722	0.0419	2.4237	
M0(Y)	0.0267	0.0301	1.4848	1.5146	-2.5803	-0.3163	27.3738	3.8112	
CPI(Y)	-0.0915	-2.8903	-0.0642	-1.5858	-0.1501	-0.4186	-0.6020	-1.6815	
EMR/VMR(-8)	0.0122	5.5710	0.0087	2.6189	0.1162	4.5885	0.1068	3.1912	
Panel B January 1990-December 1999									
		Value coef	ficient $\gamma_{pb}$		Low/High return R <sub>lmh</sub>				
	Equal-w	reighted	Value-w	eighted	Equal-we	ighted	Value-we	eighted	
Lag length	5	5	6		5		7		
Adj-R <sup>2</sup>	0.32	245	0.43	75	0.415	58	0.403	37	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	0.0002	0.2113	0.0012	0.9536	0.0381	1.7433	0.0071	0.2239	
M0(Y)	-0.6383	-1.8897	-1.0298	-2.9588	-20.9065	-2.1538	-16.9163	-1.6260	
CPI(Y)	-0.0131	-2.1698	-0.0179	-1.8436	-0.3237	-2.2235	-0.5996	-2.8848	
EMR/VMR(-8)	0.0012	5.5131	0.0014	1.4464	0.0201	2.8125	0.0235	1.0288	
Panel C January 2000-December 2008									
		Value coef	ficient $\gamma_{pb}$			Low/High re	eturn R <sub>lmh</sub>		
	Equal-w	reighted	Value-w	eighted	Equal-we	ighted	Value-we	eighted	
Lag length	ç	)	2		8		5		
Adj-R <sup>2</sup>	0.53	351	0.57	39	0.480	02	0.43	74	
Macroeconomic variables	λ	t(λ)	λ	t(λ)	λ	t(λ)	λ	t(λ)	
GDP(Y)	-0.0072	-2.6175	-0.0067	-5.7135	-0.1899	-2.2890	-0.2048	-2.8089	
M0(Y)	0.6693	5.2425	0.9235	6.7369	11.1378	2.5525	31.6682	5.1409	
CPI(Y)	0.0178	3.4692	0.0209	4.9933	0.3835	2.6003	0.9557	4.6061	
EMR/VMR(-8)	0.0039	3.2997	-0.0009	-1.1581	0.1719	5.1685	-0.0753	-1.4971	
		Panel D 1	anuary 1982-	December 2	008				
		Value coef	ficient v .	December 2		Low/High re	eturn R.		
	Equal-w	value coel	Value-w	eighted	Equal-we	ighted	Value-we	ighted	
Lag length	Equal- w	n	v alue-w	)	14	ignica	18	agineu	
$\Delta di_{\rm e} R^2$	0.4	187	0.33	, :00	0.28/	1/1	0.06	17	
Macroeconomic variables	λ λ	t(λ)	λ	t(λ)	λ	τ t(λ)	λ	t(λ)	
GDP(V)	0.0010	1 4245	0.0034	2 3180	0.0175	0.013/	0 0260	1 1615	
$MO(\mathbf{V})$	-0.3050	-0.7060	-0.1388	-0.4251	-4 0653	-0 5764	1 01209	0.6161	
$CDI(\mathbf{V})$	-0.3030	-0.7000	-0.1300	1 5001	-4.0055	-0.5704 2 0017	4.0102	0.0101	
$\mathbf{U} \mathbf{I} \left( \mathbf{I} \right)$	-0.0340	-2.0703	-0.0203	-1.3991	-0.2902	-2.001/	-0.2323	-0.0330	
$EIVIK/VIVIK(-\delta)$	0.0051	5.7060	0.0053	2.3641	0.0539	3.3480	0.0363	1.1305	

# Table 5.48 Multivariate regression of value effect on the combination of lagged macroeconomic variables in subperiods (Value: 24 months' holding length)

Note: This table presents slope coefficient  $\lambda$ , t-stats, and adjusted R<sup>2</sup> for multivariate regression of value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> on the combination of lagged macroeconomic variables in the 1980s, the 1990s, and from 2000 to 2008. Decile value portfolios are formed with shares' price-to-book values at the end of each month, and hold for 24 months. Value coefficient  $\gamma_{pb}$  is estimated by monthly-by-monthly cross-sectional regression of portfolios' equal-(value-) weighted returns in the holding period on their mean price-to-book values in the formation time. Low/High return R<sub>lmh</sub> is the return spread between 'Low' portfolio and 'High' portfolio. GDP(Y) is the previous annual change of GDP. M0(Y) is the previous annual change of M0. CPI(Y) is the previous annual change of CPI. EMR/VMR(-8) is equal-(value-) weighted market return in past eight years. t-stats is adjusted by Newey-West (1987) approach with automatic lag length as the method in Ferson, Sarkissian, and Simin (2003). Panel A to D report corresponding parameters for equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub>. with three subperiods and overall period, respectively.

# Chapter 6 The Recursive out-of-sample Forecasting Model on Style Effects

In Chapter 5, the research investigates what factors offer explanation for the time-variation in style effects in the UK Stock Market by regressing style coefficient  $\gamma$  and Long/Short return R<sub>lms</sub> on lagged macroeconomic variables in the overall period from 1980 to 2008 and in three subperiods, respectively. It finds that time-varying style effects could be explained by some lagged macroeconomic variables, which are associated with economic conditions and market states. Based on the findings in Chapter 5, the purpose of this chapter is to test the performance of recursive in-sample regression and out-of-sample forecasting model to verify whether the time-variation in style effects is indeed predictable. Section 6.1 describes the background of the recursive out-of-sample forecasting model. Section 6.2 introduces research methods in terms of recursive in-sample regression model and out-of-sample forecasting model, and statistical test on the model in the measure of 'PIS', 'POOS', and 'PSS'. Section 6.3 and Section 6.4 conduct test analysis on the performance of recursive in-sample regression and out-of-sample forecasting model, respectively. Section 6.5 analyzes the performance of forecasting macroeconomic variables in recursive in-sample regression model. Section 6.6 concludes this chapter.

# 6.1 The recursive out-of-sample forecasting model

There are three reasons for investigating the performance of recursive out-of-sample forecasting model on time-varying style effects. Firstly, Bossaerts and Hillion (1999) show the pitfalls of in-sample predictability. They find the existence of in-sample predictability of dividend yield and short-term interest rate in 14 international stock markets, but the poor performance in the out-of-sample period. A large difference exists between in-sample and out-of-sample adjusted  $R^2$  in these markets. For instance, in-sample adjusted  $R^2$  in Japan and US Stock Market is 12.4% and 8.3%, but out-of-sample one falls to 1.8% and 0.5%. They mention that the non-stationary pattern may be related to the limits of linear forecasting models and learning in the market. Goyal and Welch (2003) argue that in-sample relations conceal a systematic failure of dividend-yield model in out-of-sample ones, and forecasting regression has out-of-sample poor performance. They suggest that studies on market-timing ability should use comparative out-of-sample sum-squared model residuals to diagnose the model performance. Welch and Goyal (2008) investigate predictor variables for US stock market returns and conclude that many of them, such as dividend yield, earnings price ratio, book-to-market ratio, and TBILL, perform poorly in out-of-sample tests although they have significant explanatory power in in-sample tests. They find that the models' forecasting power volatiles over time, and in particular most models perform worse from 1975 to 2005.

Secondly, Welch and Goyal (2008) mention that out-of-sample tests, as one natural and useful diagnostic approach, help to evaluate whether a model is stable and well specified, or changing over time either suddenly or gradually. Lettau and Ludvigson (2010) point out that out-of-sample tests provide one way of assessing whether there has been structural change in a forecasting relation. In Chapter 5, the research finds the variation in explanatory power of lagged macroeconomic variables for time-varying style effects over subperiods. For instance, as Table 5.23 and 5.24 indicate, adjusted  $R^2$  of the regression of time-varying momentum effect with 12 months' formation horizon and 6/12 months' holding horizon on lagged macroeconomic variables declines for equal-weighted case, but rises for value-weighted case over the 1980s, the 1990s, and from 2000 to 2008. The similar volatility exists in adjusted  $R^2$  of the regression of size effect and value effect. In addition, Chapter 5 shows the existence of the variation in significant macroeconomic variables to explain time-varying style effect through time. For instance, as shown in Table 5.23, UEXTBILL(Y) and EMR(-1), GDP(Y), respectively, combining with DY(Q), significantly capture the variation in equal-weighted momentum effect (12/6) in the 1980s and the 1990s. Table 5.35 shows that size coefficient  $\gamma_{mv}$  of size effect with 12 months' holding horizon is significant sensitive to CPI(Y), GDP(Y), UEXTBILL(Y), respectively, in the 1980s, the 1990s, and from 2000 to 2008. More importantly, the same macroeconomic variables offer opposite explanatory power for style effects over three decades. For example, Table 5.35 shows that size effect with 12 months' holding horizon is marginally negatively driven by CPI(Y) in the 1980s, but positive though unsignificant since the 2000. Table 5.48 reflects that value effect with 24 months'

281

holding horizon is significantly negatively associated to MO(Y) and CPI(Y) in the 1990s, but positive and significant since 2000. So, the research compares predicted style coefficient  $\gamma$  and Long/Short return R<sub>lms</sub> from recursive out-of-sample forecasting model with actual ones to evaluate the performance of predictive model, and details the variation in significant macroeconomic variables.

Thirdly, the signals from recursive out-of-sample model, based on macroeconomic variables at time t-1, can be applied in practice to time style rotation monthly-by-monthly. Campbell and Thompson (2005) argue that actual-world investors would not mechanically run the unrestricted predictive models that are tested in the literature. Instead, they would impose restrictions on the regression coefficient beta to require positive expected risk premium. Some scholars, such as Levis and Liodakis (1999), Cooper, Gulen, and Vassalou (2001), Levis and Tessaromatis (2004), use out-of-sample model to explore the performance of active trading strategy.

In this chapter, the research uses recursive out-of-sample forecasting model to verify the predictability of the model for the time-variation in style effects, to illustrate dynamic pattern of macroeconomic variables to explain the variation, which offers the foundation to explore the application of forecasting model in active trading strategies for timing style rotation in Chapter 7.

# 6.2 Research methods

The recursive forecasting methodology is similar in spirit to approaches used in Bossaerts and Hillion (1999), and Cooper, Gutierrez, and Hameed (2004) among others. Levis and Liodakis (1999) apply recursive out-of-sample forecasting model to explore the capability of timing small/large-cap rotation and value/growth rotation in the UK Stock Market. Cooper, Gulen, Vassalou (2001) show that the recursive out-of-sample forecast model offers the time of allocating the position in small/medium/large-cap portfolio and high/medium/low book-to-market portfolio in the US Stock Market.

#### 6.2.1 Recursive out-of-sample forecasting model

The chapter uses recursive out-of-sample forecasting model on time-varying style effect in terms of momentum effect (12/6, 12/12), size effect (12), and value effect (24) because their long/short returns are higher than corresponding ones based on 6 months' formation length (momentum) and other holding horizons, and macroeconomic variables offer stronger explanatory power for their time-variation than corresponding style effects based on other formation horizon or holding horizons as shown in Chapter 4 and Chapter 5. Momentum effect (12/6, 12/12) is based on decile portfolios sorted by past 12 months' share returns and hold for 6/12

months. Size effect (12) is based on decile portfolios sorted by shares' market values and hold for 12 months. Value effect (24) is based on decile portfolios sorted by shares' price-to-book ratios and hold for 24 months. The detailed procedures to test recursive out-of-sample forecasting model are described as below:

The first step: The research uses in-sample regression model (6.1) and (6.1') to estimate the intercept  $\lambda_0$  and slope coefficient  $\lambda_{EV}$  of forecasting macroeconomic variables (EV) in the initial in-sample period of ten years from January 1980 to December 1989, and obtains the first intercept  $\lambda_0$  and the first slope coefficient  $\lambda_{EV}$  in June 1990 (December 1990, December 1991) when portfolios are hold for 6 (12, 24) months.

$$\gamma_t = \lambda_{0,t} + \lambda_{EV,t} * EV_{t-L} + \varepsilon_{t,t}$$
(6.1)

and

$$R_{lms,t} = \lambda_{0,t} + \lambda_{EV,t} * EV_{t-L} + \mathcal{E}_{t,t}$$
(6.1')

Where  $\gamma$  (R<sub>lms</sub>) is style coefficient (long/short return) estimated in Chapter 4,  $\lambda_0$  is intercept, EV is (n\*1) vector representing macroeconomic variables at lagged length L, which are described in Chapter 5,  $\lambda_{EV}$  is the OLS slope coefficient from the regression of  $\gamma$  (R<sub>lms</sub>) on lagged macroeconomic variables EV.  $\varepsilon$  is an error.

The second step: The intercept  $\lambda_0$  and slope coefficient  $\lambda_{EV}$  estimated from in-sample regression model (6.1) and (6.1') reflect the relation between style coefficient  $\gamma$ , long/short return  $R_{Ims}$  and forecasting macroeconomic variables at time point t. It is assumed that this relation would not change in following 3 to 24 months' horizon, and that investors equipped with information from prior period form expectations on style effects at time point t+1, and construct long/short portfolio. The research uses out-of-sample forecasting model (6.2) and (6.2'), the intercept  $\lambda_0$  and slope coefficient  $\lambda_{EV}$  estimated in in-sample regression model (6.1) and (6.1'), and known lagged macroeconomic variables (EV) in advance in June 1990 (December 1990, December 1991) to calculate first predicted monthly step-ahead style coefficient  $\hat{\gamma}$  (long/short return  $\hat{R}_{Ims}$ ) when portfolios are hold for 6 (12, 24) months.

$$\hat{\gamma}_{t+1} = \lambda_{0,t} + \lambda_{EV,t} * EV_t$$
(6.2)

and

$$\widehat{R}_{ims.t+1} = \lambda_{0,t} + \lambda_{EV,t} * EV_t$$
(6.2')

The third step: The research repeats the procedure in the first and second step, increasing every time in-sample window by one month, until obtaining 216 (204, 180) series out-of-sample predicted style coefficient  $\hat{\gamma}$  (long/short return  $\hat{R}_{lms}$ ) that cover the period from July 1990 to June 2008 (from January 1991 to December 2007, from January 1992 to December 2006) when portfolios are hold for 6 (12, 24) months.

Table 6.1 describes the beginning date on in-sample period and out-of-sample period when long/short portfolios are hold for 6 months. The similar pattern is for portfolios with 12 and 24 months' holding horizons.

### [Table 6.1]

#### 6.2.2 Statistic test

The performance of recursive in-sample regression model and out-of-sample forecasting model should be quantitatively measured and compared to the benchmark, that is the historical mean model. Although adjusted R<sup>2</sup> from the regression model offers explanatory power of the forecasting variable for the variation in dependent variable, it can't be used to evaluate the outperformance of the model relative to the unconditional benchmark (the prevailing mean). The research uses the statistics 'PIS' to test the performance of recursive in-sample regression model, and the statistics 'POOS' and 'PSS' to test the performance of recursive out-of-sample forecasting model. The plot of 'PIS' and 'POOS' helps to capture when in-sample regression model and out-of-sample forecasting model perform well or poorly over the

historical mean model. 'PSS' reflects the extent that the estimate from the forecasting model is matched by the values actually observed.

The research uses the approach in Welch and Goyal (2008) to examine whether recursive in-sample regression model and out-of-sample forecasting model outperform historical mean model. Based on the root-mean-squared error (RMSE), Welch and Goyal (2008) approach compares cumulative RMSE between recursive in-sample regression model and the historical mean model, between recursive out-of-sample forecasting model and the historical mean model, to reflect the performance of two models relative to historical mean model as the benchmark. For the performance of recursive in-sample regression model, the test statistics, which is labeled 'PIS', is estimated by the cumulative squared demeaned style coefficient  $\gamma$ (Long/Short return R<sub>lms</sub>) minus the cumulative squared regression residual from in-sample regression model (6.1, 6.1'). For the performance of recursive out-of-sample forecasting model, the test statistics, which is labeled 'POOS', is estimated by the cumulative squared demeaned style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) minus the cumulative squared its prediction error from out-of-sample forecasting model (6.2, 6.2'). This measure is to compare two forecasting methods. One forecasting model is to compute the sample mean style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) from time 1 to time t, and use that sample mean to forecast style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) at time t+1. Another forecasting method is to run a

regression from time 1 to time t, and use estimated regression coefficients to forecast style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) at time t+1. The statistics of "POOS" compares the performance in forecasts that use the sample mean with the performance in predicted style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) from regression and forecasting model in the same period.

Welch and Goyal (2008) argue that a well-specified signal would inspire confidence in the model if 'PIS' and 'POOS' had a generally upward drift over time, which occurs not just in one short or unusual sample period. The research uses the first 10 years to estimate first statistics 'PIS', 'POOS' in December 1989, increases window by one month each time, and repeats the estimation of two statistics. It obtains series 'PIS' and 'POOS' over the sample period, and plots 'PIS' and 'POOS' to indicate the performance of recursive in-sample regression model and out-of-sample forecasting model, respectively, relative to the sample mean model. When the lines of 'PIS' and 'POOS' increase, in-sample regression model and out-of-sample forecasting model perform better relative to the historical mean model. When they decrease, the models perform poorly relative to the historical mean model. The lines of 'PIS' and 'POOS' allow diagnosis of time with good or poor performance of recursive in-sample regression and out-of-sample forecasting model.

The statistics 'POOS' is to evaluate the performance of out-of-sample forecasting

model relative to the historical mean model by comparing two models' cumulative squared forecasting errors. Additively, the research uses the statistics 'PSS' to measure the predictability of out-of-sample forecasting model by comparing the sign of predicted values and actually observed values. It uses the number (percentage) of observations with same sign of predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>), which is denoted as 'PSS', to indicate the extent that out-of-sample forecasting model offers successful forecasting signals for the time-variation in style *effects.* It is instructive for actual-world investors to apply the model for timing style rotation. The research uses recursive out-of-sample forecasting model (6.2, 6.2') to estimate style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>) from 1990 to 2008, and to calculate 'PSS', the percentage of observation with same sign between predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>). Actual style coefficient  $\gamma$ (Long/Short return R<sub>lmh</sub>) of momentum, size, and value effect are estimated in Chapter 4. When predicted style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>) from the recursive out-of-sample forecasting model has the same sign (positive or negative) as actual one, the model is successful in offering forecasting signals. When predicted style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>) from the recursive out-of-sample forecasting model has the opposite sign as actual one, the model fails to offer correct forecasting signals. The higher is the statistics 'PSS', the better is the performance of forecasting model. The statistics 'PSS' offers underlying parameter for active trading strategy on style effects, which is in detail analyzed in Chapter 7.

# 6.3 The test on performance of recursive in-sample regression

In Chapter 5, the research finds dynamic explanatory power of lagged macroeconomic variables for style effects over three decades from 1980 to 2008. For example, Table 5.24 Panel A to C show that the combination of lagged macroeconomic variables, which are GDP(Y), UEXBILL(Y), DY(Q), and CPI(H), can explain 26.37%, 17.13%, and 59.23% of the variation in the return spread between value-weighted 'Winner' and 'Loser' portfolios with 12 months' formation length and 12 months' holding horizon in the 1980s, the 1990s, and the period from 2000 to 2008, respectively. In this section, the research further investigates the performance of recursive in-sample regression on three style effects from 1980 to 2008, which highlights monthly-by-monthly fluctuation pattern of explanatory power of macroeconomic variables for time-varying style effects. The research uses series adjusted  $R^2$  and PIS to evaluate the performance of recursive in-sample regression of style coefficient  $\gamma$  (Long/short return  $R_{lms}$ ) on lagged macroeconomic variables.

# 6.3.1 Series adjusted $R^2$

The research uses in-sample regression model (6.1) and (6.1') to estimate series explanatory power of lagged macroeconomic variables for time-varying style effects.

The initial in-sample period is set from January 1980 to December 1989. The first adjusted- $R^2$  of the regression is estimated in June 1990 (December 1990, December 1991) when portfolios are hold for 6 (12, 24) months.

[Table 6.2]

# [Figure 6.1, 6.2]

Panel A and B of Table 6.2 detail the results on the performance of series adjusted  $R^2$ in recursive in-sample regression of momentum effect (12/6, 12/12) on macroeconomic variables from 1980 to 2008. When momentum portfolios are formed by previous 12 months' share returns and hold for 6 and 12 months, the combination of four lagged macroeconomic variables, which include GDP(Y), UEXTBILL(Y), DY(Q), and EMR(-1)/CPI(H), could explain average 19.14 to 32.75, and 19.75 to 35.15 percentage of the time-variation in equal-(value-) weighted momentum coefficient  $\gamma_{mom}$  and Winner/Loser return  $R_{wml}$ . The explanatory power maintains higher volatile with standard deviation of about 0.10 for equal-weighted case, low volatile with standard deviation of less than 0.05 for value-weighted case. Importantly, the highest (lowest) explanatory power of macroeconomic variables for two cases of momentum effect is 58.06% and 52.61% (11.29% and 15.62%), respectively. Figure 6.1 and Figure 6.2 plot series adjusted  $R^2$  estimated from recursive in-sample regression of momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) with 6 and 12 months' holding horizons on lagged macroeconomic variables from 1980 to 2008, respectively. Adjusted  $R^2$  in equal-weighted pattern exhibits the obvious decline in the early 1990s and the year 2002 and then the stability, but the rise in the early 2000s. Adjusted  $R^2$  in value-weighted pattern experiences three periods, which include the sustaining decline in the 1990s, the rise in the early 2000s, and the stability afterwards.

## [Figure 6.3]

Panel C of Table 6.2 reports the performance of series adjusted  $R^2$  in recursive in-sample regression of size effect (12) on lagged macroeconomic variables from 1980 to 2008. When size portfolios are hold for 12 months, the combination of four macroeconomic variables, which includes GDP(Y), CPI(Y), UEXTBILL(Y), and UEXTERM(Y), provides explanatory power with average 21.13 to 24.61 percentage of the time-variation in equal-(value-) weighted size coefficient  $\gamma_{mv}$  and Small/Large return  $R_{sml}$ . This explanatory power is stable over time, with low standard deviation of 0.0538 to 0.0557, and the highest (lowest) one of 31.52% to 35.47% (13.44% to 16.69%). Figure 6.3 shows that adjusted  $R^2$  in recursive in-sample regression of size effect (12) is in stable condition over most of time, except for the decline in 1991, 1996, and 2002.

## [Figure 6.4]

Panel D of Table 6.2 reports the performance of series adjusted  $R^2$  in recursive in-sample regression of value effect (24) on lagged macroeconomic variables from 1980 to 2008. When price-to-book portfolios are hold for 24 months, the combination of macroeconomic variables, which consists of GDP(Y), M0(Y), CPI(Y), and MR(-8), explains high up to average 25.91 to 48.11 percentage of time-varying equal-(value-) weighted value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub>. Except for vale-weighted Low/High return R<sub>lmh</sub>, the explanatory power has lower volatility over time, with low standard deviation of 0.0358 to 0.0593. The highest adjusted R<sup>2</sup> is between 0.4236 and 0.5728. Particularly, the lowest adjusted R<sup>2</sup> keeps higher level ranging between 0.2724 and 0.4002. Figure 6.4 graphs declining adjusted R<sup>2</sup> in the early 1990s and 2000s, rising one in the second half of the 1990s, and stable one after 2003.

In summary, recursive in-sample explanatory power of macroeconomic variables for time-varying momentum effect (12/6, 12/12), size effect (12), and value effect (24) are generally higher, but volatiles over time from 1980 to 2008, which consists of the declining, rising, and stable period. This volatility is consistent with the findings in Chapter 5 on the variation in adjusted  $R^2$  of multivariate regression for style effects in three subperiods of the 1980s, the 1990s, and 2000 afterwards. Few

macroeconomic variables persistently keep significant impact on the time-variation in style effects in three subperiods, which might lead to the volatility in their explanatory power over time. In addition, the limits of linear regression model and learning in the market, as Bossaerts and Hillion (1999) mention, might cause the unstable explanatory power of macroeconomic variables.

#### 6.3.2 Series test statistics, 'PIS'

The research uses the statistics, 'PIS', to test the performance of recursive in-sample regression of style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) on lagged macroeconomic variables from 1980 to 2008. Similar to the method in Welch and Goyal (2008), 'PIS' is measured by the cumulative squared demeaned style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) minus the cumulative squared regression residual from in-sample regression model (6.1, 6.1') each month. The research estimates series PIS of style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) from 1990 to 2008, and plots their 'PIS' performance.

The research uses the first 10 years to estimate information coefficients from the regression of momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) on lagged macroeconomic variables, which include GDP(Y), UEXTBILL(Y), DY(Q), and EMR(-1)/CPI(H), from January 1980 to December 1989, and calculates the first

statistics 'PIS' in December 1989. Increasing in-sample window by one month each time and repeating the regression, the research obtains series 'PIS' over in-sample period. Figure 6.5 and Figure 6.6 graph series 'PIS' for momentum effect (12/6, 12/12), the performance of recursive in-sample regression of momentum coefficient  $\gamma_{\rm mom}$  (Winner/Loser return R<sub>wml</sub>) on macroeconomic variables from December 1989 to June 2008/December 2007. For equal-weighted momentum portfolios, 'PIS' line (dashed line) increases from 1990 to 1991 and from 1999 to early 2002, indicating performance of in-sample regression in these two periods better than the historical mean model. After two increases, 'PIS' line faces the fall, especially sharp fall from early 2003 to early 2004, showing poor performance of in-sample regression in the period. In other periods, 'PIS' maintains stable condition. For value-weighted momentum portfolios, 'PIS' line exhibits a generally upward trend from 1999 to early 2003, and keeps high horizontal level. Therefore, Figure 6.5 and Figure 6.6 show that in-sample impact of the combination of four macroeconomic variables on time-varying momentum effect had neither good nor bad performance in most sample period, best performance at the end of 1990s and the beginning of 2000s, and poor performance for equal-weighted momentum after experiencing good one. Figure 6.1 and Figure 6.5, Figure 6.2 and Figure 6.6 show similar pattern in equal-weighted momentum effect (12/6, 12/12) between adjusted  $R^2$  and 'PIS'. For value-weighted momentum effect (12/6, 12/12), there is difference in the performance between adjusted  $R^2$  and 'PIS' in the 1990s. Although adjusted  $R^2$  experiences declining condition in the 1990s as shown in Figure 6.1 and Figure 6.2, 'PIS' performs in stable condition.

## [Figure 6.5, 6.6]

Using the same way in above case of momentum effect, the research estimates series 'PIS' for size effect (12), the performance of recursive in-sample regression of size coefficient  $\gamma_{mv}$ (Small/Large return  $R_{sml}$ ) on lagged macroeconomic variables, which include GDP(Y), CPI(Y), UEXTBILL(Y), and UEXTERM(Y), from 1980 to 2008 when size portfolios are hold for 12 months. Figure 6.7 shows the 'PIS' pattern of in-sample regression on size effect (12) is volatile. The line of 'PIS' increases during 1990, 1992, 1997 to mid-1999, and 2004 to 2008, indicating that the combination of macroeconomic variables explains time-varying size effect (12) well in these four periods. 'PIS' line exhibits a fall after its four increases. The line of 'PIS' experiences a fall in the periods, consistent with the periods in which adjusted  $R^2$  performs downward in Figure 6.3. But the line of 'PIS' rises in the periods when adjusted  $R^2$  keeps flat.

## [Figure 6.7]

The research estimates and graphs series 'PIS', in-sample predictive performance of

lagged macroeconomic variables of GDP(Y), MO(Y), CPI(Y), and MR(-8) for the time-variation in value coefficient  $\gamma_{pb}$ (Low/High return R<sub>lmh</sub>) from 1980 to 2008. As shown in Figure 6.8, the line of 'PIS' performs upward from 1994 to 1998 and then downward until the late 2003, showing superior performance and inferior performance of in-sample regression in two periods over the historical mean model. Especially, for value-weighted Low/High return R<sub>lmh</sub>, the line of 'PIS' experiences sharp fall from 1999 to early 2005, indicating that macroeconomic variables fail to offer predictive power in that period. The line of 'PIS' presents the similar change of adjusted R<sup>2</sup> as shown in Figure 6.4.

### [Figure 6.8]

In summary, the research uses the statistics 'PIS' to examine the performance of recursive in-sample regression of style effects on macroeconomic variables relative to the sample mean model in the UK Stock Market from 1980 to 2008. It finds the volatility in explanatory power of regression model through time, which exhibits its better/and poor performance at some periods. This reflects that time-varying style effects are not always captured by these macroeconomic variables all the time, and that they might be caused by other information in the market.

# 6.4The test on performance of recursive out-of-sample forecasting model

As introduced in Section 6.1, recursive out-of-sample forecasting model can evaluate whether the model is indeed predictable, and whether the model is stable or volatile, either suddenly or gradually. In this section, the research uses the statistics 'POOS' to test the performance of out-of-sample forecasting model for the time-variation in style effects relative to historical mean model. Meanwhile, it uses the statistics 'PSS' to examine the predictability of out-of-sample forecasting model, which is measured by the percentage of observations with same sign of predicted and actual Long/Short return  $R_{ims}$  (style coefficient  $\gamma$ ).

### 6.4.1 Series test statistics, 'POOS'

Similar to the method of estimating the statistics 'PIS' in subsection 6.3.2, the research uses the statistics 'POOS' to test the performance of out-of-sample forecasting model on expected style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>). 'POOS' is measured by the cumulative squared demeaned style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) minus the cumulative squared its prediction error from out-of-sample forecasting model (6.2, 6.2'). The research estimates series out-of-sample predicted

style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) and series 'POOS' from 1990 to 2008. The line of series 'POOS' is used to evaluate the performance of out-of-sample forecasting model relative to sample mean model. The increase in 'POOS' line reflects better performance of forecast model relative to sample mean model, and vice versa.

Figure 6.5 and Figure 6.6 graph solid line as the 'POOS' for predicted momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) from 1990 to 2008 in the case of momentum effect (12/6, 12/12). They show that 'POOS' line is similar to 'PIS' for two cases of momentum effect. For equal-weighted momentum portfolios, 'POOS' line exhibits stable pattern from 1991 until 1999, and increases from 1999 to early 2002, indicating the better out-of-sample forecasting performance relative to the prevailing sample mean. After the increase, 'POOS' line sharply falls from early 2003 to early 2004, showing poor performance of out-of-sample forecasting model during the period. Afterwards, the 'POOS' line restores its stable pattern as its performance in the beginning of the sample period. Different from equal-weighted case, 'POOS' line for value-weighted momentum portfolios exhibits a generally upward movement from 1999 to early 2003, and keeps at high level. Therefore, Figure 6.5 and Figure 6.6 show that out-of-sample forecasting model had stable performance in most sample period, better performance at the end of 1990s and the beginning of 2000s, and poor performance for equal-weighted momentum portfolios after experiencing good one.

The performance of out-of-sample forecasting model on momentum effect is almost consistent with the performance of its in-sample regression from 1990 to 2008.

Figure 6.7 plots solid line as the 'POOS' for predicted size coefficient  $\gamma_{mv}$ (Small/Large return  $R_{sml}$ ) from 1990 to 2008 when size portfolios are hold for 12 months. Similar to 'PIS' line, the 'POOS' line is volatile over the period. It exhibits rising or flat pattern from 1992 to 1995, from 1997 to mid-1999, and from 2004 to 2008, indicating that out-of-sample forecasting model offers better or stable predicting signal relative to prevailing sample mean during these periods. The line falls in other periods, showing the poor performance of the forecasting model. So, in-sample regression and out-of-sample forecasting model on size effect (12) keeps the same performance from 1990 to 2008.

Figure 6.8 graphs solid line as the 'POOS' for predicted value coefficient  $\gamma_{pb}$ (Low/Hihg return  $R_{lmh}$ ) from 1990 to 2008 when value portfolios are hold for 24 months. For predicted value coefficient  $\gamma_{pb}$ , the 'POOS' line exhibits flat pattern even if the 'PIS' line has rising performance from 1994 to 1998. For predicted equal-weigted Low/High return  $R_{lmh}$ , the 'POOS' line has similar pattern over sample period as the 'PIS' line, showing the forecasting model's superior performance from 1994 to 1998, inferior performance after that period until the late 2003, and stable performance afterwards. For predicted value-weighted Low/High return  $R_{lmh}$ , its

'POOS' line fails to follow the increase in 'PIS' line and maintains flat from 1994 to 1998. After this period, the POOS line has sharp downward trend until 2006. Except for equal-weighted Low/High return  $R_{lmh}$ , out-of-sample forecasting model on value effect (24) performs poorly from 1990 to 2008 although in-sample regression model has good performance in some periods.

#### 6.4.2 Test statistics, 'PSS'

The research uses the statistics, 'PSS', to test whether the recursive out-of-sample forecasting model offers the same signal as actual style effect. 'PSS' is measured as the percentage of observation with same sign (PSS) between predicted and actual style coefficient  $\gamma$  (long/short return R<sub>lms</sub>). Actual style coefficient  $\gamma$  (long/short return R<sub>lms</sub>) is estimated in Chapter 4. Panel A and B of Table 6.3 indicate '+PSS' and '-PSS', the percentage of observation with same positive and negative sign, respectively, between predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>), for momentum effect (12/6, 12/12), size effect (12), and value effect (24). Panel C reports 'PSS', the percentage of observation with the combination of same positive and same negative sign, for corresponding style effects.

# [Table 6.3]

Panel A shows that when the recursive out-of-sample forecasting model for momentum effect (12/6, 12/12) offers positive predicted momentum coefficient  $\gamma_{mom}$ (Winner/Loser return  $R_{wml}$ ), there are 68.07% to 83.50% of forecasts with the same positive sign as actual ones. For value effect (24), 61.54% to 90.0% of predicted positive value coefficient  $\gamma_{pb}$  (Low/High return  $R_{lmh}$ ) is matched by actual ones. For size effect (12), the forecasting model offers lower percentage in same positive sign between predicted and actual size coefficient  $\gamma_{mv}$  (Small/Large return  $R_{sml}$ ), ranging between 45.71% and 55.47%.

As indicated in Panel B, when out-of-sample forecasting model offers momentum effect (12/6, 12/12) and value effect (24) high percentage of observation with same positive sign between predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lmh</sub>), the model has lower percentage with the same negative sign, which is less than 56.25% and 28.95% for momentum effect (12/6, and 12/12) and value effect (24), respectively. Conversely, although the forecasting model offers size effect (12) with unsatisfied predictive signals with same positive sign, it has relatively higher percentage with the same negative one, which is more than 64%. Panel C reports the percentage with the combination of same positive and negative sign between predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) to indicate overall the predictability of the model. It is shown that the recursive out-of-sample forecasting model offers momentum effect (12/6, 12/12) higher percentage of observation with

the same sign between predicted and actual ones, followed by size effect (12) and value effect (24). The corresponding statistics 'PSS' ranges between 60.29% and 81.48%, between 51.47% and 60.29%, and between 45.0% and 58.33%, respectively.

The research further observes the relation between 'PIS' and 'PSS' from 1990 to 2008. It assigns 1 to 'Sign' when predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) has the same positive or negative sign, and assigns -1 to 'Sign' when predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) has opposite sign. When the research compares dashed 'PIS' and column graph 'Sign' for equal-weighted momentum effect (12/6, 12/12), it is found in the Figure 6.5 and Figure 6.6 that '1' sign is likely to appear when 'PIS' line is flat or increases. This indicates that the sign of predicted equal-weighted momentum coefficient  $\gamma_{mom}$ (Winner/Loser R<sub>wml</sub>) is consistent with actual ones when in-sample regression performs in stable or better condition. Conversely, most of '-1' sign appear later 1992 to early 1993, and later 2003 to early 2004 when 'PIS' has downward trends. This indicates that the signal from out-of-sample forecasting model is different from actual one when in-sample regression had poor performance. For value-weighted momentum effect, the sign of '1' usually appears when 'PIS' line is flat or rises, and the sign of '-1' does in 2002 and 2003 when 'PIS' line is volatile although no having obvious decline as one in equal-weighted case. So, the percentage having same sign ('PSS') for predicted and actual momentum coefficient  $\gamma_{mom}$  (Winner/Loser  $R_{wml}$ ) is closely correlated with the performance of in-sample regression ('PIS').

As shown in Figure 6.7, the sign of '1' for size effect (12) is likely to appear in a row from early 1992 to 1993 and from mid-1998 to 1999 when the 'PIS' line trends obviously upward. Conversely, the sign of '-1' happens in three periods when the 'PIS' line trends downward, from early 1991 to early 1992, from mid-1995 to mid-1997, and from 2001 to mid-2003. More than 50 percent of the sign of '-1' are in three periods that account for less than 40 percent of observations. This reflects that the percentage having same sign ('PSS') between predicted and actual size coefficient  $\gamma_{mv}$  (Small/Large  $R_{sml}$ ) is closely correlated with the performance of in-sample regression.

Similar to momentum effect (12/6, 12/12) and size effect (12), Figure 6.8 shows the sign of '-1', the opposite sign between predicted and actual value coefficient  $\gamma_{pb}$  (Low/High Return R<sub>lmh</sub>) based on 24 months' holding horizon, appears when in-sample regression performs worse from 2000 to 2003. But the sign of '1', the same sign between predicted and actual ones does not respond to the increase in the line of 'PIS' from 1994 to 1998 as the same way in momentum effect (12/6, 12/12) and size effect (12). Although the 'PIS' line rises during 1994 to 1998, the 'PSS' ranges between 40 and 53 percentage. This pattern is consistent with the poor performance of out-of-sample forecasting model during this period in which the
'POOS' line experiences slight decline or flat, except for equal-weighted Low/High Return  $R_{lmh}$ .

In summary, the research uses two statistics, 'POOS' and 'PSS', to investigate the performance of recursive out-of-sample forecasting model. Statistic tests show that the performance of out-of-sample forecasting model is volatile through time. The lines of 'POOS' and 'PIS' reflect that the out-of-sample forecasting model has the same volatile pattern as in-sample regression model for momentum effect (12/6, 12/12) and size effect (12), but difference for value effect (24). As Welch and Goyal (2008) argue, the out-of-sample forecasting model for value effect (24) performs poorly even though in-sample regression model offers high explanatory power for value coefficient  $\gamma_{pb}$  and Low/High Return R<sub>lmh</sub>. The statistics 'PSS' shows that out-of-sample forecasting model offers correct signal to capture momentum effect (12/6, 12/12) in 60.29% to 81.48% of all observations, followed by size effect (12) with 51.47% to 60.29%. However, the percentage ranges between 45.0% and 58.33% for value effect (24). The result in the statistics 'PSS' implies the possibility of out-of-sample forecasting model to capture the rotation in both momentum effect and size effect, but its failure for value effect.

# 6.5 The series performance of macroeconomic variables in recursive in-sample regression

The findings in Chapter 5 show the variation in significant macroeconomic variables on time-varying style effects in three decades from 1980 to 2008. The recursive in-sample regression offers more detailed description on their dynamic performance as sample length monthly extends ahead.

#### [Table 6.4]

Table 6.4 reports the percentage of observations in which macroeconomic variables are statistically significant in recursive in-sample regression of style coefficient  $\gamma$ (Long/Short return R<sub>lms</sub>) from 1980 to 2008. Panel A shows that UEXTBILL(Y) and DY(Q) keeps significant pattern in explaining time-varying momentum effect (12/6) in higher percentage of all observations, between 57.4% and 100%, between 59.6% and 91.5%, respectively. The percentage for GDP(Y) and EMR(-1)/CPI(H) is lower than 43% in some cases. Panel B shows that GDP(Y) is one stable significant macroeconomic variable for momentum effect (12/12) in more than 90 percent of observations. But the significance of other macroeconomic variables is volatile over time, with the percentage below 60%. Panel C indicates that although GDP(Y) and CPI(Y) are two most important variables among four macroeconomic variables to explain time-varying size effect (12), they are statistically significant in 45.6 to 78.8 and 49.8 to 70.5 percentage of observations, respectively. The explanatory power of UEXTBILL(Y) is much volatile, significant in less than 30 percent of observations. When combining other variables, UEXTERM(Y) losses significant impact on time-varying size effect (12) in all observations, as its performance in the regression in a whole period.

As Panel D shows, EMR(-8) is most stable explanatory variable except for value-weighted Low/High return  $R_{lmh}$ , and is significant in more than 95.6 percent of observations. This reflects the past long-term market performance is determinant factor in capturing the rotation between value shares and growth shares. CPI(Y) offers significant impact on time-varying value effect (24) in more than 71 percent of observations. GDP(Y) and M0(Y) lose significant explanatory power for value effect (24) in most cases.

Series performance of macroeconomic variables in recursive in-sample regression shows that macroeconomic variables, except GDP(Y) for momentum effect (12/12) and EMR(-8) for value effect (24), might lose explanatory power for time-varying style effects in the UK Stock Market at some time points although they are significant in overall period. This leads to the failure of recursive in-sample regression and out-of-sample forecasting model to capture correct signal for time-varying style effects in the real world.

### 6.6 Conclusion

Based on the findings in Welch and Goyal (2008) that most equity premium predictive models fail to statistically and economically outperform the unconditional benchmark (the prevailing mean), this chapter focuses on the performance of recursive in-sample regression and out-of-sample forecasting model on time-varying momentum effect (12/6, 12/12), size effect (12), and value effect (24) to verify whether the time-variation in style effects is indeed forecasted in the UK Stock Market from 1980 to 2008. Besides adjusted  $R^2$ , the research uses the statistics 'PIS', which is measured by the cumulative squared demeaned style coefficient  $\gamma$ (Long/Short return R<sub>lms</sub>) minus the cumulative squared regression residual, to test the performance of recursive in-sample regression model. It uses the statistics 'POOS', which is measured by the cumulative squared demeaned style coefficient  $\gamma$ (Long/Short return R<sub>lms</sub>) minus the cumulative squared its out-of-sample prediction error, to test the performance of recursive out-of-sample forecasting model. Meanwhile, it uses the statistics 'PSS', the percentage of observations with same sign of predicted and actual style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>), to evaluate the successful signal from out-of-sample forecasting model.

It is found that recursive in-sample explanatory power of the combination of lagged macroeconomic variables for time-varying momentum effect (12/6, 12/12), size

effect (12), and value effect (24) is generally higher, but volatile over time. The statistics 'PIS' can be used to measure the performance of recursive in-sample regression through time, and to capture the time with good and poor performance of regression model relative to the prevailing sample mean model.

The test of recursive out-of-sample forecasting model indicates that 'POOS' for momentum effect (12/6, 12/12) and size effect (21) keeps the similar performance in out-of-sample period as 'PIS', the statistics to test recursive in-sample regression, but exhibits quite difference for value effect (24). The percentage of same sign ('PSS') of predicted and actual style coefficient  $\gamma$  (Long/Short R<sub>lms</sub>) is higher for momentum effect (12/6, 12/12), followed by size effect (12) and value effect (24). The recursive out-of forecasting model offers same sign (positive and negative) between predicted and actual momentum coefficient  $\gamma_{mom}$  (Winner/Loser R<sub>wml</sub>) in 60 to 82 percent of observations. The percentage falls down to 51 to 61 percent, and 45 to 59 percent for size effect (12) and value effect (24), respectively. The diagnostic test on recursive out-of-sample forecasting model seems to be promising in timing momentum effect (12/6, 12/12) and size effect (12).

The test of recursive in-sample regression and out-of-sample forecasting model, measured by three statistics in terms of 'PIS', 'POOS', and 'PSS', paves the way to explore the outperformance of active trading strategy on style rotation relative to passive trading strategy on one fixed style all the time. The higher statistics 'PSS' in the empirical test of out-of-sample forecasting model is statistically and economically instructive for timing momentum effect (12/6, 12/12) and size effect (12), which is further investigated in the next chapter.

## Appendix: tables and figures in Chapter 6

Date	EV	Long/Short Portfolio	R <sub>lms</sub>	In-sample Reg	$\lambda_{\rm EV}$	Exp R <sub>lms</sub>
1980/1	EV <sub>1-L</sub>	$LSP_1$				
1980/2	EV <sub>2-L</sub>	$LSP_2$				
1980/3	EV <sub>3-L</sub>	LSP <sub>3</sub>				
1980/4	$EV_{4-L}$	$LSP_4$				
1980/5	EV <sub>5-L</sub>	LSP <sub>5</sub>				
1980/6	EV <sub>6-L</sub>	LSP <sub>6</sub>				
1980/7	EV <sub>7-L</sub>	LSP <sub>7</sub>	R <sub>lms,1</sub>			
1980/8	EV <sub>8-L</sub>	LSP <sub>8</sub>	R <sub>lms,2</sub>			
1980/9	EV <sub>9-L</sub>	LSP <sub>9</sub>	R <sub>lms,3</sub>			
1980/10	$EV_{10-L}$	$LSP_{10}$	R <sub>lms,4</sub>			
1980/11	EV <sub>11-L</sub>	LSP <sub>11</sub>	R <sub>lms,5</sub>			
1980/12	EV <sub>12-L</sub>	LSP <sub>12</sub>	R <sub>lms,6</sub>			
1990/1	EV <sub>121-L</sub>	LSP <sub>121</sub>	R <sub>lms,115</sub>			
1990/2	EV <sub>122-L</sub>	LSP <sub>122</sub>	R <sub>lms,116</sub>			
1990/3	EV <sub>123-L</sub>	LSP <sub>123</sub>	<b>R</b> <sub>lms,117</sub>			
1990/4	EV <sub>124-L</sub>	LSP <sub>124</sub>	R <sub>lms,118</sub>			
1990/5	EV <sub>125-L</sub>	LSP <sub>125</sub>	R <sub>lms,119</sub>			
1990/6	EV <sub>126-L</sub>	LSP <sub>126</sub>	R <sub>lms,120</sub>	IS Reg <sub>1</sub>	$\lambda_{EV,1}$	
1990/7	EV <sub>127-L</sub>	LSP <sub>127</sub>	<b>R</b> <sub>lms,121</sub>	IS Reg <sub>2</sub>	$\lambda_{EV,2}$	Exp R <sub>lms,1</sub>
1990/8	EV <sub>128-L</sub>	LSP <sub>128</sub>	R <sub>lms,122</sub>	IS Reg <sub>3</sub>	$\lambda_{EV,3}$	Exp R <sub>lms,2</sub>
1990/9	EV <sub>129-L</sub>	LSP <sub>129</sub>	R <sub>lms,123</sub>	IS Reg <sub>4</sub>	$\lambda_{\rm EV,4}$	Exp R <sub>lms,3</sub>
1990/10	EV <sub>130-L</sub>	LSP <sub>130</sub>	R <sub>lms,124</sub>	IS Reg <sub>5</sub>	$\lambda_{\rm EV,5}$	Exp R <sub>lms,4</sub>
1990/11	EV <sub>131-L</sub>	LSP <sub>131</sub>	R <sub>lms,125</sub>	IS Reg <sub>6</sub>	$\lambda_{EV,6}$	Exp R <sub>lms,5</sub>
1990/12	EV <sub>132-L</sub>	LSP <sub>132</sub>	R <sub>lms,126</sub>	IS Reg <sub>7</sub>	$\lambda_{EV,7}$	Exp R <sub>lms,6</sub>
1991/1	EV <sub>133-L</sub>	LSP <sub>133</sub>	<b>R</b> <sub>lms,127</sub>	IS Reg <sub>8</sub>	$\lambda_{\rm EV,8}$	Exp R <sub>lms,7</sub>
1991/2	EV <sub>134-L</sub>	LSP <sub>134</sub>	R <sub>lms,128</sub>	IS Reg <sub>9</sub>	$\lambda_{\rm EV,9}$	Exp R <sub>lms,8</sub>
1991/3	EV <sub>135-L</sub>	LSP <sub>135</sub>	R <sub>lms,129</sub>	IS Reg <sub>10</sub>	$\lambda_{EV,10}$	Exp R <sub>lms,9</sub>
1991/4	EV <sub>136-L</sub>	LSP <sub>136</sub>	R <sub>lms,130</sub>	IS Reg <sub>11</sub>	$\lambda_{EV,11}$	Exp R <sub>lms,10</sub>
1991/5	EV <sub>137-L</sub>	LSP <sub>137</sub>	<b>R</b> <sub>lms,131</sub>	IS Reg <sub>12</sub>	$\lambda_{EV,12}$	Exp R <sub>lms,11</sub>
1991/6	EV <sub>138-L</sub>	LSP <sub>138</sub>	R <sub>lms,132</sub>	IS Reg <sub>13</sub>	$\lambda_{EV,13}$	Exp R <sub>lms,12</sub>
1991/7	EV <sub>139-L</sub>	LSP <sub>139</sub>	<b>R</b> <sub>lms,133</sub>	IS Reg <sub>14</sub>	$\lambda_{EV,14}$	Exp R <sub>lms,13</sub>
1991/8	$EV_{140-L}$	$LSP_{140}$	<b>R</b> <sub>lms,134</sub>	IS Reg <sub>15</sub>	$\lambda_{EV,15}$	Exp R <sub>lms,14</sub>
1991/9	EV <sub>141-L</sub>	LSP <sub>141</sub>	R <sub>lms,135</sub>	IS Reg <sub>16</sub>	$\lambda_{EV,16}$	$Exp \; R_{lms,15}$
1991/10	EV <sub>142-L</sub>	LSP <sub>142</sub>	R <sub>lms,136</sub>	IS Reg <sub>17</sub>	$\lambda_{EV,17}$	$Exp \; R_{lms,16}$
1991/11	EV <sub>143-L</sub>	LSP <sub>143</sub>	R <sub>lms,137</sub>	IS Reg <sub>18</sub>	$\lambda_{EV,18}$	$Exp \; R_{lms,17}$

Table 6.1 The beginning date on in-sample period and out-of-sample period when long/short portfolios are hold for 6 months

Note: EV stands for macroeconomic variable at lagged length L, LSP stands for long/short portfolio,  $R_{lms}$  stands for long/short return, IS Reg stands for in-sample regression,  $\lambda_{EV}$  is the OLS slope coefficient from the regression (6.1') of long/short return  $R_{lms}$  on lagged macroeconomic variables. Exp  $R_{lms}$  stands for expected long/short return from recursive out-of-sample equation (6.2').

Panel A Momentum effect (12/6)								
	momentum coe	fficient $\gamma_{mom}$	Winner/Loser return R <sub>wml</sub>					
	Equal-weighted	Value-weighted	Equal-weighted	Value-weighted				
Average	0.3275	0.1914	0.2836	0.2057				
Std	0.1193	0.0379	0.1013	0.0499				
Max	0.5806	0.2656	0.5097	0.2761				
Min	0.1741	0.1234	0.1574	0.1129				
_	Panel B Momentum effect (12/12)							
	momentum coefficient $\gamma_{mom}$		Winner/Lose	er return R <sub>wml</sub>				
	Equal-weighted	Value-weighted	Equal-weighted	Value-weighted				
Average	0.3515	0.1975	0.3146	0.2284				
Std	0.1101	0.0252	0.0966	0.0443				
Max	0.5261	0.2464	0.4550	0.3558				
Min	0.1818	0.1562	0.1607	0.1636				
		Panel C Size effect	(12)					
	size coef	ficient $\gamma_{mv}$	Small/Larg	e return R <sub>sml</sub>				
	Equal-weighted	Value-weighted	Equal-weighted	Value-weighted				
Average	0.2314	0.2113	0.2461	0.2137				
Std	0.0541	0.0554	0.0557	0.0538				
Max	0.3443	0.3263	0.3547	0.3152				
Min	0.1542	0.1344	0.1669	0.1372				

Table 6.2 The performance of series adj-R <sup>2</sup> in recursive in-sample regression	ı of sty	le
coefficient $\gamma$ (Long/Short return $\mathbf{R}_{lms}$ ) on lagged macroeconomic variables from 1	1980 to	2008

	Panel D Value effect (24)								
	value coe	fficient $\gamma_{pb}$	Low/High return R <sub>lmh</sub>						
	Equal-weighted	Value-weighted	Equal-weighted	Value-weighted					
Average	0.4811	0.3826	0.3526	0.2591					
Std	0.0524	0.0358	0.0593	0.1107					
Max	0.5728	0.4236	0.4420	0.3622					
Min	0.4002	0.3169	0.2724	0.0493					

Note: The table offers summary statistics for series adjusted  $R^2$  in recursive in-sample regression of style coefficient  $\gamma$  (Long/Short return  $R_{lms}$ ) on lagged macroeconomic variables from 1980 to 2008, which include its mean, standard deviation, maximum, and minimum value. Panel A and Panel B include corresponding data for momentum effect based on 12 months' formation length and 6 and 12 months' holding length, respectively. Panel C reports corresponding data for size effect based on 12 months' holding length. Panel D reports corresponding data for value effect based on 24 months' holding length.

Figure 6.1 The performance of series adjusted  $R^2$  in recursive in-sample regression of momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) on lagged macroeconomic variables from December 1989 to June 2008: Momentum effect (12/6)



Note: The figure plots series adjusted  $R^2$  estimated from recursive in-sample regression of momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) on macroeconomic variables, including GDP(Y), UEXTBILL(Y), DY(Q), and EMR(-1)/CPI(H), from December 1989 to June 2008. The momentum portfolios are formed on previous 12 months' share returns and hold for 6 months. The initial in-sample period is from January 1980 to December 1989.

Figure 6.2 The performance of series adjusted  $R^2$  in recursive in-sample regression of momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) on lagged macroeconomic variables from December 1989 to December 2007: Momentum effect (12/12)



Note: The figure plots series adjusted  $R^2$  estimated from recursive in-sample regression of style coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) on lagged macroeconomic variables, including GDP(Y), UEXTBILL(Y), DY(Q), and EMR(-1)/CPI(H), from December 1980 to December 2007. The momentum portfolios are formed on previous 12 months' share returns and hold for 12 months. The initial in-sample period is from January 1980 to December 1989.

Figure 6.3 The performance of series adjusted  $R^2$  in recursive in-sample regression of size coefficient  $\gamma_{mv}$  (Small/Large return  $R_{sml}$ ) on lagged macroeconomic variables from December 1989 to December 2007: Size effect (12)\_



Note: The figure plots series adjusted  $R^2$  estimated from recursive in-sample regression of size coefficient  $\gamma_{mv}$  (Small/Large return  $R_{sml}$ ) on lagged macroeconomic variables, including GDP(Y), CPI(Y), UEXTBILL(Y), and UEXTERM(Y), from December 1980 to December 2007. The size portfolios are hold for 12 months. The initial in-sample period is from January 1980 to December 1989.

Figure 6.4 The performance of series adjusted  $R^2$  in recursive in-sample regression of value coefficient  $\gamma_{pt}$  (Low/High return  $R_{lmh}$ ) on lagged macroeconomic variables from December 1989 to December 2006: Value effect (24)



Note: The figure plots series adjusted  $R^2$  estimated from recursive in-sample regression of value coefficient  $\gamma_{pb}$  (Low/High return  $R_{lmh}$ ) on lagged macroeconomic variables, including GDP(Y), MO(Y), CPI(Y), MR(-8), from December 1980 to December 2006. The value portfolios are hold for 24 months. The initial in-sample period is from January 1980 to December 1989.











Note: The dashed and solid figures plot 'PIS' and 'POOS', the performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to June 2008: the case of Momentum effect (12/6). 'PIS' is the cumulative squared demeaned momentum coefficient  $\gamma_{mom}$  (Winner/Loser R<sub>wml</sub>) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared demeaned momentum coefficient  $\gamma_{mom}$  (Winner/Loser R<sub>wml</sub>) minus the cumulative squared prediction error from out-of-sample forecasting model. Momentum portfolios are formed by previous 12 months' share returns and hold for 6 months. The figure labeled as 'Sign' assigns 1 (-1) to the observation whose predicted and actual momentum coefficient  $\gamma_{mom}$  (Winner/Loser return R<sub>wml</sub>) have the same (opposite) sign.

Figure 6.6 'PIS', 'POOS', and 'Sign', the performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to 2008: Momentum effect (12/12)



Note: The dashed and solid figures plot 'PIS' and 'POOS', performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to December 2007: the case of Momentum effect (12/12). 'PIS' is the cumulative squared demeaned momentum coefficient  $\gamma_{mom}$  (Winner/Loser  $R_{wml}$ ) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared demeaned momentum coefficient  $\gamma_{mom}$  (Winner/Loser  $R_{wml}$ ) minus the cumulative squared prediction error from out-of-sample forecasting model. Momentum portfolios are formed by previous 12 months' share returns and hold for 12 months. The figure labeled as 'Sign' assigns 1 (-1) to the observation whose predicted and actual momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ ) have the same (opposite) sign.

Figure 6.7 'PIS', 'POOS', 'Sign', the performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to 2008: Size effect (12)













Note: The dashed and solid figures plot 'PIS' and POSS, performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to December 2007: the case of Size effect (12). 'PIS' is the cumulative squared demeaned size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared demeaned size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared from the cumulative squared productive squared by the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared productive squared productive squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared in-sample regression residual.' POOS' is the cumulative squared productive squared productive squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared productive squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared productive squared productive squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared productive squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared productive squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative squared size coefficient  $\gamma_{mv}$  (Small/Large R<sub>sml</sub>) minus the cumulative square demeaned size coefficient  $\gamma_{mv}$  (Small/Large  $R_{sml}$ ) minus the cumulative squared prediction error from out-of-sample forecasting model. Size portfolios are hold for 12 months. The figure labeled as 'Sign' assigns 1 (-1) to the observation whose predicted and actual size coefficient  $\gamma_{mv}$  (Small/Large return  $R_{sml}$  have the same (opposite) sign.

Figure 6.8 'PIS', 'POOS', and 'Sign', the performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to 2008: Value effect (24)





1

0

0.5

-0.5

-1

-1.5

Note: The dashed and solid figures plot 'PIS' and 'POOS', performance of recursive in-sample regression and out-of-sample forecasting model from December 1989 to December 2006: the case of Value effect (24). 'PIS' is the cumulative squared demeaned value coefficient  $\gamma_{pb}$  (Low/High R<sub>lmh</sub>) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared demeaned value coefficient  $\gamma_{pb}$  (Low/High R<sub>lmh</sub>) minus the cumulative squared in-sample regression residual. 'POOS' is the cumulative squared demeaned value coefficient  $\gamma_{pb}$  (Low/High R<sub>lmh</sub>) minus the cumulative squared prediction error from out-of-sample forecasting model. Value portfolios are hold for 24 months. The figure labeled as 'Sign' assigns 1 (-1) to the observation whose predicted and actual value coefficient  $\gamma_{pb}$  (Low/High R<sub>lmh</sub>) have the same (opposite) sign.

r							
Panel A '+PSS', the percentage having same positive sign							
	Style coe	efficient γ	Long/short	return R <sub>lms</sub>			
	Equal weighted	Value weighted	Equal weighted	Value weighted			
Momentum effect (12/6)	0.8350	0.7696	0.8115	0.7414			
Momentum effect (12/12)	0.8284	0.6807	0.7702	0.6977			
Size effect (12)	0.4775	0.5248	0.4571	0.5547			
Value effect (24)	0.8596	0.6154	0.9000	0.6812			

Table 6.3 The performance of 'PSS': the percentage having same sign between predicted and actual style coefficient  $\gamma$  (Long/Short return  $R_{ims}$ ) in out-of-sample period from 1990 to 2008.

Panel B '-PSS', the percentage having same negative sign

	Style coe	efficient y	Long/short	return R <sub>lms</sub>
	Equal weighted	Value weighted	Equal weighted	Value weighted
Momentum effect (12/6)	0.5625	0.4800	0.3600	0.4048
Momentum effect (12/12)	0.4571	0.2632	0.3721	0.4400
Size effect (12)	0.6667	0.6699	0.6406	0.6842
Value effect (24)	0.2602	0.2895	0.2667	0.2619

Panel C 'PSS', the percentage having same positive and same negative sign

	Style coe	efficient y	Long/short	return R <sub>lms</sub>
	Equal weighted	Value weighted	Equal weighted	Value weighted
Momentum effect (12/6)	0.8148	0.7361	0.7593	0.6759
Momentum effect (12/12)	0.7647	0.6029	0.6863	0.6029
Size effect (12)	0.5637	0.5980	0.5147	0.6029
Value effect (24)	0.4500	0.4778	0.5833	0.5833

Note: The table shows the percentage having same sign between predicted and actual style coefficient  $\gamma$  (Long/Short return  $R_{lms}$ ) for Momentum effect (12/6, 12/12), Size effect (12), and Value effect (24) from 1990 to 2008. Predicted style coefficient  $\gamma$  (Long/Short return  $R_{lms}$ ) is estimated from recursive out-of-sample forecasting model (6.2, 6.2'). Actual style coefficient  $\gamma$  (Long/Short return  $R_{lms}$ ) is estimated from the regression of decile portfolios' holding returns on firms' characteristics (the return spread between top/bottom portfolios) in Chapter 4.

Panel A Momentum effect (12/6) Momentum coefficient  $\gamma_{mom}$ Winner/Loser return R<sub>wml</sub> Equal-weighted Value-weighted Equal-weighted Value-weighted sign. + sign. sign. + sign. sign. + sign. sign. + sign. -GDP(Y) 0.0 42.2 0.0 65.0 0.0 36.3 0.0 89.2 UEXTBILL(Y) 0.0 100.0 100.0 61.4 0.0 57.4 0.0 0.0 91.5 DY(Q) 78.5 0.0 0.0 59.6 0.0 88.8 0.0 EMR(-1)/CPI(H) 0.0 71.3 0.0 36.8 0.0 43.0 0.0 42.6 Panel B Momentum effect (12/12) Momentum coefficient  $\gamma_{mom}$ Winner/Loser return R<sub>wml</sub> Equal-weighted Equal-weighted Value-weighted Value-weighted sign. + sign. + sign. sign. + sign. sign. sign. + sign. -GDP(Y) 0.0 91.2 0.0 90.3 0.0 100.0 0.0 90.8 55.3\*\* 0.0 UEXTBILL(Y) 57.1 0.0 50.7 0.0 47.9 0.0 DY(Q) 66.8 0.0 37.3 0.0 60.8 0.0 62.2 0.0 EMR(-1)/CPI(H) 0.0 0.0 30.9 0.0 50.7 0.0 28.1 46.1

Table 6.4 The percentage of observations with significant macroeconomic variables
in recursive in-sample regression of style coefficient $\gamma$ (Long/Short return $R_{\text{lms}})$ from
1980 to 2008

Panel C Size effect (12)								
	Size coefficient $\gamma_{mv}$					Small/Larg	e return R <sub>sml</sub>	
	Equal-weighted		Value-weighted		Equal-weighted		Value-weighted	
	sign. +	sign	sign. +	sign	sign. +	sign	sign. +	sign
GDP(Y)	45.6	0.0	63.6**	0.0	78.8	0.0	$66.8^{**}$	0.0
CPI(Y)	0.0	53.0	0.0	49.8	0.0	70.5	0.0	68.2
UEXTBILL(Y)	0.0	23.9	0.0	$25.8^{**}$	0.0	28.1	0.0	24.9
UEXTERM(Y)	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0

Panel D Value effect (24)									
	Value coefficient $\gamma_{pb}$				Low/High return R <sub>lmh</sub>				
	Equal-w	reighted	Value-weighted		Equal-weighted		Value-weighted		
	sign. +	sign	sign. +	sign	sign. +	sign	sign. +	sign	
GDP(Y)	0.0	0.0	100.0	0.0	0.0	0.0	26.7	0.5	
MO(Y)	0.0	0.0	0.0	0.0	0.0	0.0	1.8	0.0	
CPI(Y)	0.0	100.0	0.0	$77.1^{**}$	0.0	73.2	0.0	71.0	
EMR(-8)	100.0	0.0	95.6	0.0	98.5	0.0	60.4	0.0	

Note: This table presents the percentage of observations with significant macroeconomic variables in recursive in-sample regression of style coefficient  $\gamma$  (Long/Short return R<sub>lms</sub>) from 1980 to 2008. t-stats of macroeconomic variable's coefficient is adjusted with Newey-West (1987) approach with automatic lag length. The symbol \*\* indicates the 10% significance level. Panel A and Panel B include corresponding data for momentum effect based on 12 months' formation length and 6 and 12 months' holding length, respectively. Panel C reports corresponding data for size effect based on 12 months' holding length. Panel D reports corresponding data for value effect based on 24 months' holding length.

## **Chapter 7 Active Trading Strategy on Style Rotation**

In Chapter 4, the research constructs decile portfolios sorted by firms' characteristics, such as past share return, market value, and price-to-book ratio, and finds negative return spread on Winner/Loser portfolio, Small/Large portfolio, and Low/High portfolio at some time points. This means the failure of traditional zero investment strategy on style effects to achieve good performance all the time. This pattern is especially obvious for the size strategy with 12 months' holding length. Figure 7.1 plots the performance of Small/Large return  $R_{sml}$  when long/short position is allocated on equal-weighted Small/Large portfolio for 12 months. It shows that Small/Large return  $R_{sml}$  had two bad cycles from 1980 to 2008, each lasting for about three or four years. For example, besides short-lived negative return, there have been two bear markets for Small/Large return  $R_{sml}$  since 1980: from 1989 to 1992, and from 1997 to 1999. Given the fact that style-oriented investors could face the style drift in some periods, any forecasting indication of these unefficient periods would be rather useful in style investing.

#### [Figure 7.1]

The traditional absolute return strategy always goes long/short position on Winner/Loser, Small/Large, and Low/High portfolios each month, labeled as 'passive

trading strategy'. Different from passive trading strategy, active trading strategy pays attention to timing style rotation, and holds long/short position according to the signal from out-of-sample forecasting model. Firstly, the research focuses on three kinds of active trading strategy within individual style effect: two ones which go long/short position on the 1<sup>st</sup> or the 10<sup>th</sup> portfolio, another which allocates long/short position on any decile portfolio with predicted highest/lowest portfolio return. The third active trading strategy does not limit long/short position in top/bottom portfolio, and instead could go long/short position on Winner, Medium, or Loser (Small, Medium, or Large, and Low, Medium, or High price-to-book) portfolio. Secondly, the research explores active trading strategy within combined two/or three style effects to capture one style effect with predicted highest long/short return, relative to passive trading strategy which distributes average position within two/or three style effects all the time. In Chapter 4, the research finds negative correlation between momentum effect and size effect, momentum effect and value effect, and positive correlation between size effect and value effect. It is expected to yield the potential gains to timing the switch in the allocation of Winner/Loser portfolio, Small/Large portfolio, and Low/High portfolio.

This chapter is organized as four sections. Section 7.1 introduces research methods to develop active trading strategies on style effects. Section 7.2 examines active trading strategies based on top/bottom decile portfolios and any decile portfolio within

individual style effect. Section 7.3 explores active trading strategy within combined two/or three style effects. Section 7.4 comes to the conclusion of this chapter.

## 7.1 Research methods

Style consistency, sticking to a particular style, is likely to be a sub-optimal strategy in the market due to the time-variation in style effects. Style rotation creates a need for active trading strategies to allocate position on shares with special firm characteristics, and to benefit portfolio performance enhancement.

Levis and Liodakis (1999) use recursive out-of-sample logit regression to estimate the sign of the return spread in terms of size effect and value effect in the UK Stock Market. They classify each month as 1 or 0 according to the predicted sign of the return spread. If in a particular month small-cap shares (value shares) perform better than large-cap shares (growth shares), they set this month to 1; otherwise they classify it as 0. Their forecasting model is based on small-cap portfolio and large-cap portfolio for the size strategy, value portfolio and growth portfolio for the value strategy.

Different from the approach in Levis and Liodakis (1999), Cooper, Gulen, and Vassalou (2001) do not impose constraint that the long position should always only include small-cap portfolio or high book-to-price portfolio in the US Share Market, and that the short position should always only include large-cap portfolio or low book-to-price portfolio. Their strategies allocate long and short portfolios on any decile portfolio according to the decile size-sorted and value-sorted portfolio returns predicted by recursive out-of-sample forecasting model. Therefore, the long (short) position can include small-, medium- or large-cap portfolios for the size strategy, and low, medium or high book-to-price portfolios for the value strategy.

Campbell and Thompson (2005) argue that a rational investor would not use a model to forecast a negative equity premium, and apply truncation of such predictions at zero return. They find better out-of-sample forecasting performance when imposing restrictions on the signs of coefficients and return forecasts, and assume that investors rule out a negative predictive equity premium and set the forecast to zero whenever it is negative. Cooper (1999) shows that to impose thresholds for expected return is to boost signal-to-noise ratio of the portfolio. Cooper, Gulen, and Vassalou (2001) document that long (short) position might yield negative (positive) predicted return when not taking into account the magnitude of predicted returns from the forecasting model, and find the increase in the return of active trading strategy after imposing filter rules on return forecasts in the US Stock Market from 1953 to 1998.

Based on statistic tests of recursive out-of-sample forecasting model in Chapter 6, the chapter explores trading strategies on momentum effect (12/6, 12/12), size effect (12), and value effect (24), which has higher long/short return and can be captured by lagged macroeconomic variables much more than style effects with other

corresponding formation lengths and holding horizons. The research uses Levis and Liodakis (1999) approach, combining with filter threshold, to develop active trading strategy on 'Winner' portfolio and 'Loser' portfolio for the momentum strategy, 'Small' portfolio and 'Large' portfolio for the size strategy, 'Low' portfolio and 'High' portfolio for the value strategy. It uses Cooper, Gulen, and Vassalou (2001) approach to develop active trading strategy on any decile momentum, size, and value portfolio. In addition, the research investigates active trading strategy within combined two/or three style effects, which holds position with predicted highest long/short return on Winner/Loser portfolio, Small/Large portfolio, and Low/High portfolio.

#### 7.1.1 Active trading strategy on individual style effect

Active trading strategy on individual style effect includes three patterns in momentum effect (12/6, 12/12), size effect (12), and value effect (24), individually: Active trading strategy in single direction, Active trading strategy in double directions, and Active trading strategy on any decile portfolio.

<u>Active trading strategy in single direction</u> The research uses out-of-sample forecasting model (6.2) to predict long/short return for style effects. Using the similar

method in Cooper, Gulen, and Vassalou (2001), the research sets up 7 filter thresholds from 0 to 3.0% in increments of 0.5% in active trading strategy on style effects in terms of long/short return. When predicted long/short return from out-of-sample forecasting model meets filter ruler at a certain month, the research assumes allocation of long position on Winner/Small/Low portfolio and short position on Loser/Large/High portfolio. When predicted long/short return from out-of-sample forecasting model would not pass the constraint, the portfolio is gone to TBILL. Because filter threshold in this strategy is set to be only positive value, it is defined as 'Active trading strategy in single direction'. Because the strategy takes advantage of signal and magnitude from out-of-sample forecasting model, it is expected to hold long/short position with positive return and TBILL, which replaces for long/short position with predicted negative return, and further to enhance return performance of passive trading strategy and reduce the percentage with negative return.

<u>Active trading strategy in double directions</u> Levis and Tessaromatis (2004) propose growth/value rotation strategies that if return spread forecast is positive the strategy goes long (short) position on growth (value) stocks, and if return forecast is negative the strategy goes long (short) position on value (growth) stocks. Using similar method in Levis and Tessaromatis (2004), the research extends long/short trading in single direction to long/short trading in double directions by imposing positive and negative filter thresholds on predicted Long/Short return R<sub>lms</sub>. When predicted Long/Short return R<sub>lms</sub> from out-of-sample forecasting model is satisfied with one given positive filter threshold, long position is gone on Winner/Small/Low portfolio, and short position is gone on Loser/Large/High portfolio. When predicted Long/Short return R<sub>lms</sub> is satisfied with one given negative filter threshold, long position is gone on Loser/Large/High portfolio, and short position is gone on Loser/Large/High portfolio, and short position is gone on Satisfied with one given negative filter threshold, long position is gone on Loser/Large/High portfolio, and short position is gone on Winner/Small/Low portfolio. Otherwise, the position is allocated in TBILL. This strategy is named as 'Active trading strategy in double directions', and is expected to enhance the return performance of active trading strategy in single direction because it is expected capture opposite trading opportunity.

Active trading strategy on any decile portfolio Different from 'Active trading strategy in single/double direction' based on top/bottom portfolio, 'Active trading strategy on any decile portfolio' explores the strategy which allocates long/short positions on any decile portfolio, instead of top and bottom portfolios, according to predicted decile portfolio returns from out-of-sample forecasting model. This active trading strategy goes long position on the portfolio that has the highest predicted return from forecasting model, and short position on the portfolio that has the lowest predicted return. There is no constraint that long/short position always includes top and bottom decile portfolios. It chooses the allocation of long and short position according to predicted any decile portfolio return from out-of-sample forecasting model portfolios.

model. Therefore, long/short position can include Winner, Medium, or Loser portfolios for momentum strategy, Small, Medium, or Large portfolios for size strategy, and Low, Medium, or High price-to-book portfolios for value strategy. Predicted decile portfolio returns are estimated by recursive in-sample regression model (6.1) and out-of-sample forecasting model (6.2).

#### 7.1.2 Active trading strategy on combined style effects

In Chapter 4, the research finds negative correlation between momentum effect and size effect, between momentum effect and value effect, and positive correlation between size effect and value effect. The correlation among style effects is caused by their sensitivity to the same macroeconomic forces. In Chapter 5, the research finds that momentum effect (12/6, 12/12) and size effect (12) are commonly significantly captured by the previous annual change of GDP and unexpected annual change of TBILL with the opposite direction. Meanwhile, in the value-weighted case, momentum effect (12/6, 12/12) and value effect (12) are significant in the opposite TBILL condition, and size effect (12) and value effect (12) are significant in the same TBILL condition.

The existence of correlation among momentum effect, size effect, and value effect

motivates the exploration on active trading strategy with combined two/or three style effects. In one case, when one style effect is in favor in one certain economic condition and market state, another style effect might suffer. In another case, although two or three styles are effective at the same time, there is the difference in their strength. This means that to hold position with favorable style effect or style effect with highest long/short return can maximize the investment return.

Table 4.15 in Chapter 4 shows that negative correlation in equal-(value-) weighted long/short return with 12 months' holding horizon between momentum effect and size effect, and between momentum effect and value effect ranges from -0.4529 to -0.5893, the highest magnitude among the cases of five holding horizons. Positive correlation in equal-(value-) weighted long/short return with 12 month's holding horizon between size effect and value effect is 0.1560 (0.45), the third highest magnitude among the cases of five holding horizons. In addition, as found in Chapter 4 and Chapter 5, momentum effect (12/12) and size effect (12) have higher long/short return, and can be much explained by lagged macroeconomic variables relative to the case within corresponding other holding horizons. So, the examination on active trading strategy with combined two/or three style effects focuses on the case with 12 months' holding horizon from 1990 to 2008, which includes 204 observations.

The research sets up 7 filter thresholds from 0 to 3.0% in increments of 0.5% in active trading strategy with combined two/or three style effects. It compares predicted long/short returns of style effects from out-of-sample forecasting model (6.2) with the filter threshold, and makes a choice on position allocation. When predicted monthly long/short returns of two or three style effects from out-of-sample forecasting model are all lower than one filter threshold, the position is allocated on TBILL. Otherwise, the position is allocated on top/or bottom portfolio of one style effect with predicted highest return spread. This means that the position could be allocated in Winner/Loser portfolio, Small/Large portfolio, Low/High portfolio, and TBILL. As the benchmark, passive trading strategy on the combination of two or three style effects allocates average position on momentum effect (12/12), style effect (12), and value effect (12) all the time.

## 7.2 Active trading strategy on individual style effect

In this section, the research explores three patterns of active trading strategy on individual style effect, which allocates long/short position on top/bottom portfolio in single direction, top/bottom portfolio in double directions, and any decile portfolio, respectively.

## 7.2.1 Active trading strategy in single direction

Active trading strategy in single direction is to allocate long/short position on top/or bottom decile portfolios when predicted long/short return from recursive out-of-sample forecasting model meets one filter threshold from 0 to 3.0% in increments of 0.5%, or to hold TBILL. As the benchmark, passive trading strategy is to allocate long/short position on top/or bottom decile portfolios each month. The research compares active trading strategy with passive trading strategy in terms of trading return, hit rate, and trading cost.

[Table 7.1, 7.2, 7.3, and 7.4]

Table 7.1, 7.2, 7.3, and 7.4 report the result on the performance of active trading

strategy in single direction in terms of trading return, hit rate, and trading cost for momentum effect (12/6), momentum effect (12/12), size effect (12), and value effect (24), respectively. Each column of the table reports the result of the corresponding filter ruler in active trading strategy and passive trading strategy. The first five rows report statistics associated with trading return performance, that is average monthly excess return, standard deviation, Sharpe ratio, the percentage of observations with trading return less than zero (%<0) and monthly TBILL (%<monthly TBILL). The next five rows report the statistics associated with signal-to-noise ratio of trading strategy, that is the number of holding Long/Short portfolio and TBILL according to the signal and magnitude of predicted Long/short return from recursive out-of-sample forecasting model and filter rules, and their corresponding individual and combining percentage of successful implementation, which is labeled 'Hit rate'. 'Hit rate of holding long/short portfolio' indicates the percentage of trading satisfied with a given filter threshold in long/short trading. 'Hit rate of holding TBILL' indicates the percentage of actual return unsatisfied with a given filter threshold in holding TBILL. 'Hit rate of strategy' indicates the percentage of successful holding long/short portfolio and TBILL in observations. The last four rows report statistics, which are associated with trading costs, such as monthly turnover, net average monthly excess return (100bp), net average monthly excess return (200bp), and breakeven trading cost. Monthly turnover is the percentage of holding long/short position in the observations according to predicted long/short return from

out-of-sample forecasting model and filter threshold. Net average monthly excess return (100bp, 200bp) is trading return after reducing trading cost when assuming transaction cost of 100 and 200 basis points for every turnover. Similar to the method in Levis and Tessaromatis (2004), the research defines breakeven transaction cost as the fixed trading cost that makes monthly excess return of one strategy to zero. It is estimated by the following equation:  $R^{strategy}$ -T\*C=0.  $R^{strategy}$  is average monthly excess return of one strategy before trading cost, T is the monthly turnover (buys+sells), and C is the average breakeven transaction cost.

#### • *The trading return*

The empirical data shows that active trading strategy on top and bottom portfolios can improve investment performance in momentum effect (12/6, 12/12) and size effect (12). For example, Table 7.1 Panel A shows that for equal-weighted momentum effect (12/6), the active strategy with filter threshold of 0 and 0.5% earns 0.9135% and 0.8630% of average monthly excess return, respectively, from 1990 to 2008. They are higher than 0.7899% of average monthly excess return of passive trading strategy, which always invests in Winner/Loser portfolio each month. As indicated in Panel B, for value-weighted momentum effect (12/6), the breakeven point for the active Winner/Loser position in terms of average monthly excess return is the 1.5% filter threshold. Before that, the return of active trading strategy is higher than one of passive trading strategy. The second row in the table shows that standard

deviation of average monthly excess return of active trading strategy with all filter thresholds is lower than one of passive trading strategy, and that standard deviation decreases as filter threshold rises from 0 to 3.0%. The reason for low volatility is that active trading strategy uses signal and magnitude of predicted Winner/Loser return R<sub>wml</sub> from out-of-sample forecasting model to successfully time both Winner/Loser portfolio with positive return and TBILL, and reduces the number of long/short trading with negative return. As shown in the fourth and fifth row in the table, the percentages of observations with average monthly trading return less than zero and monthly TBILL in active trading strategy are lower than passive trading strategy. This means that the model with imposing constraints is able to successfully forecast the period with positive and negative Winner/Loser return R<sub>wml</sub>. Table 7.2 shows the similar findings on enhancement in active trading return with momentum effect (12/12). Relative to 0.1615 and -0.0967 of equal- and value-weighted Winner/Loser return  $R_{wml}$  by passive trading strategy, active trading strategy yields 0.2904 to 0.3643 of average monthly excess return when filter threshold is set to 0 and 0.5.

Like momentum effect (12/6, 12/12), the research finds the enhancement in return performance of active trading strategy with top and bottom decile size portfolios relative to passive trading strategy. Table 7.3 presents that passive trading strategy, which always goes long/short position on Small/Large portfolio for 12 months, yields negative monthly excess return from 1990 to 2008. When filter threshold of active trading strategy is set to zero, average monthly excess return is less than 0 due to two reasons. One reason is that predicted equal-(value-) weighted Small/Large return R<sub>sml</sub> from forecasting model is positive in only 140 (128) among 204 months, taking up 68.6 (62.7) percent of all observations, respectively. Another reason is that 45.71 (55.47) percentage of Small/Large position successfully yields positive  $R_{sml}$ , as shown by the statistics 'Hit rate of holding Small/Large portfolio'. Limited number of holding Small/Large position and its low hit rate lead to negative active trading return. If using active trading strategy with filter threshold of 0.5% to 1.5%, average monthly excess equal-(value-) weighted return ranges from 0.1064% to 0.1201% (from 0.0409% to 0.0737%), which increases equal-(value-) weighted return of passive trading strategy by more than 36 (33) basis points. Another advantage of active trading strategy on size effect is to low standard deviation of trading return, which indicates the return performance more stable and percentage of observations with trading return less than zero and TBILL lower than one of the passive strategy. As reported in the fourth row of the table, passive trading equal-(value-) weighted Small/Large portfolio yields negative excess return in 57.35 (53.43) percentage of observations. Correspondingly, active strategy with filter threshold 0.5% to 1.5% earns negative return in 4.41 to 17.16 (2.94 to 8.82) percentage of observations. The similar result is presented by Levis and Liodakis (1999). They find profitable small-cap/large-cap rotation strategy in the UK Stock Market from 1968 to 1997.

However, active trading strategy seems no benefit to the enhancement in passive trading return with top and bottom price-to-book portfolios. Table 7.4 Panel A and B show that average monthly excess return of active trading strategy with all filter thresholds is lower than one of passive trading strategy. One main reason is that the forecasting model offers 90 signals to hold equal-weighted Low/High portfolio in the low percentage (50 percent) of all months when imposing the filter threshold on 0. Another reason is that passive value strategy with 24 months' holding horizon, which always hold Low/High portfolio each month, has higher percentage of observations with positive return spread, 81.67% (69.44%) for equal-(value-) weighted case shown in the 7<sup>th</sup> row of Panel A and B of Table 7.4. It is difficult for limited number of active trading position to improve passive trading return with high hit rate. As indicated in Figure 6.8 in the previous chapter, the line of POOS on predicted equal-(value-) weighted Low/High return R<sub>lmh</sub> begins the fall since the year of 2000 and lasts for four (five) years, which indicates the poor performance of out-of-sample forecasting model. Similar findings on active trading strategy with value effect are in other academic work. Levis and Liodakis (1999) find the failure of value/growth rotation strategy in the UK Stock Market from 1968 to 1997. Levis and Tessaromatis (2004) document no benefit in using value/growth forecasting model in UK's large capitalization stocks from 1992 to 2001. Cooper, Gulen, and Vassalou (2001) find that active trading strategy on book-to-price ratio fails to improve the performance of passive trading strategy in the US Stock Market from 1953 to 1998.
It is obvious that recursive out-of-sample forecasting model makes active trading strategy on momentum effect (12/6, 12/12) and size effect (12) more profitable but less volatile as well than corresponding passive trading strategy. But long-term consistency of passive Low/High return  $R_{lmh}$  with 24 months' holding length, limited number of active trading position, and poor signal from out-of-sample forecasting model make an attempt to take advantage of the time-variation in value effect rather disappointed.

### • The hit rate

The research observes the hit rate of active trading strategy in two conditions: before breakeven point in which active trading strategy yields average monthly excess return higher than one of passive trading strategy, and at filter thresholds in which corresponding active trading strategy yields positive excess returns to some extent.

For the case of equal-weighted momentum effect (12/6), as shown in seventh, ninth, and tenth rows of Table 7.1 Panel A, the hit rate of holding Winner/Loser position at 0 and 0.5% filter threshold before breakeven point is 81.15 and 74.72 percent, the percentage of Winner/Loser trading with meeting a given filter threshold. The hit rate of holding TBILL is lower than 40 percent. Because the number of holding Winner/Loser position is much larger than the number of holding TBILL, such as 191 vs 25 and 178 vs 38, the hit rate of active trading strategy combining

Winner/Loser position and TBILL position is high up to 75.93 and 68.52 percent of the observations. The result is similar for the case of equal-weighted momentum effect (12/12) in Table 7.2 Panel A. Before breakeven point of 0.5% filter threshold, the hit rate of active trading strategy is more than 61.76 percent of observations.

For the case of value-weighted momentum effect (12/6) in Table 7.1 Panel B, before breakeven point of 1.5% filter threshold, 64.10 to 74.14 percent of holding Winner/Loser positions yields  $R_{wml}$  which meets a given filter threshold. Combining with the hit rate of holding TBILL which ranges between 40.48% and 46.38%, the hit rate of active trading strategy is between 52.78 and 67.59 percent of all observations. For value-weighted momentum effect (12/12) in Table 7.2 Panel B, the hit rate of active strategy before 1.5% filter threshold ranges between 56.37 and 61.27 percent of observations.

As shown in Table 7.3, active trading strategy on size effect (12) outperforms the passive strategy at all filter thresholds. At filter threshold from 0.5% to 1.5% in which average monthly excess equal-weighted return is more than 0.10 per month, 42.5% to 47.17% of holding Small/Large positions is satisfied with a given filter threshold, and more than 80% of actual return when holding TBILL is unsatisfied with a given filter threshold. Although the hit rate of holding Small/Large portfolio is less than 50 percent, the hit rate of active trading strategy is higher up to 0.6569 to

0.7843 due to large number of holding TBILL and its higher hit rate.

For the case of equal-weighted value effect (24), Table 7.4 Panel A shows that the hit rate of passive trading strategy is high up to 0.8167. To outperform passive trading strategy, active trading strategy needs to have large number of holding Low/High portfolio as well as its higher hit rate. Unfortunately, the forecasting model offers 90 signals, 50 percent of observations, to hold Low/High portfolio at 0% filter threshold although it has 90 percent of hit rate. The hit rate of holding TBILL in other 50 percent of observations is only 0.2667%. Two disadvantages lead to 0.5833 of hit rate of the active strategy, and further poor active trading return performance. For the case of value-weighted value effect (24) in Panel B, although having high number of holding TBILL is only 0.2619, which drags down overall hit rate of the active strategy and its return performance. Therefore, active value strategy with all filter thresholds underperforms the passive strategy.

### • The trading cost

It is rational to regard the trading cost as one measure to test one trading strategy, besides the trading return. The trading cost arises in two ways. Firstly, the trading cost is associated with the turnover, the number of rebalancing long/short position of the trading strategy according to return forecasts. The research estimates monthly turnover by calculating the percentage of holding long/short position in all months from 1990 to 2008. Secondly, the trading strategy needs to update the membership of stocks in long/short portfolio to maintain the characteristics of top and bottom portfolios. The constituents in top and bottom portfolios may change significantly at each rebalancing point.

Bogle (1994) and Carhart (1997) estimate the trading cost to be approximately 1.20 and 0.95 percent for every 100 percent turnover. Studying 10,922 equity funds in one-year's period of 2006, Unified Trust Company concludes that average annual turnover of all equity funds is 92.7 percent, and that average trading cost per 100 percent turnover is 1.47 percent. To maximize the effect of trading cost on return performance of trading strategies, the research assumes that all stocks be updated each rebalance on long/short position. Following Levis and Liodakis (1999), it estimates two kinds of return: trading return after assuming 100 or 200 basis points of transaction costs for each turnover, which is defined as 'net average monthly excess return (100bp, 200bp).

The performance of net average monthly excess return of active trading strategies on momentum effect (12/6, 12/12), size effect (12), and value effect (24) after reducing trading costs in the return spread of Long/Short position is consistent with the performance of average monthly excess return. As Table 7.1 shows, although 100

and 200 basis points of transaction cost per turnover is reduced, net average monthly excess return of active trading strategy on equal-(value-) weighted momentum effect (12/6) is more than 0.58 (0.72) before 1.0% (2.0%) of ruler threshold. Table 7.2 indicates that when the trading cost is 100 and 200 basis points per turnover, passive trading strategy on annual momentum effect (12/12) becomes unprofitable. But, active trading strategy yields more than 0.18 of net average monthly excess return when filter threshold is set up to 0 and 0.5%. As Table 7.3 shows, when the trading cost is 100 and 200 basis points per turnover, net average monthly excess return drops to below 0.10 at all filter thresholds. As Table 7.4 shows, after reducing trading costs 100bp and 200bp, passive trading strategy on equal- (value-) weighted top and bottom value portfolio yields net excess return of 0.1585 and 0.1169 (0.4843 and 0.4426) per month. The active strategy with 0 of filter threshold earns low net average monthly excess return, below 0.17. So, only active trading strategy with Winner/Loser portfolio maintains higher profitability after considering trading cost.

By comparing active trading strategy with the passive one in Table 7.1, 7.2, and 7.3, the research finds the increase in breakeven transaction cost of active momentum and size strategy with most filter thresholds. The breakeven transaction cost before the active momentum and size strategy becomes unprofitable is higher because they have higher monthly excess returns and lower turnovers than passive strategy. As shown in Table 7.4, active value strategy seems not to have impact on the increase in

breakeven trading cost as well as its excess return. Although active value strategy has low turnover than the passive one, the magnitude is much less than the decrease in corresponding excess return, which results in its low breakeven transaction cost.

In short, the empirical result is more promising for timing momentum effect (12/6, 12/12) and size effect (12) by using active trading strategy with top and bottom portfolio before reducing trading cost. It is obvious that recursive out-of-sample forecasting model makes active trading strategy with momentum effect and size effect more profitable but less risky than corresponding passive strategy. The active strategy in momentum effect and size effect succeeds in return enhancement of passive trading strategy due to its high hit rate. After reducing trading costs, only active trading strategy with momentum effect (12/6, 12/12) maintains higher profitability, but active strategy with size effect (12) becomes unprofitable. Long-term consistency of value stocks and poor out-of-sample signal from forecasting model lead to the failure of active trading strategy with value effect (24) to outperform passive strategy.

### 7.2.2 Active trading strategy in double directions

Based on empirical work in the previous subsection, the research further explores the performance of active trading strategy with top and bottom portfolios in double

directions by imposing positive and negative filter thresholds on expected Long/Short return  $R_{lms}$ . As the extension of the active strategy with positive filter thresholds, when predicted Long/Short return  $R_{lms}$  is satisfied with one given negative filter threshold, long position is gone on Loser/Large/High portfolio, and short position is gone on Winner/Small/Low portfolio. Otherwise, the position is allocated in TBILL.

# [Table 7.5, 7.6, 7.7, and 7.8]

The empirical work indicates that active trading strategy in double directions fails to enhance return of active trading strategy in single direction, except for the case of momentum effect (12/6). As indicated in Table 7.5, 7.6, 7.7, and 7.8, average monthly excess return of active trading strategy in double directions for momentum effect (12/12), size effect (12), and value effect (24) is lower than one of active trading strategy in single direction due to two reasons. Firstly, the number of long/short trading on positive filter threshold is much more than the number of long/short trading on negative filter threshold. Secondly, trading return meets a given negative filter threshold at less than 50 percent of trading except for size effect (12) with 0 filter threshold. Limited trading number and lower hit rate of long/short position with negative filter threshold prevent active trading strategy in double directions from enhancing return performance of active trading strategy in single direction.

For momentum effect (12/6), although active strategy in double directions raises trading return of active strategy in single direction, the enhancement is paid off by the increase in trading cost from the allocation of Loser/Winner portfolio. For example, as indicated in Table 7.5 Panel B, active trading strategy with filter threshold of >0.5%, <-0.5% on value-weighted momentum effect (12/6) yields average excess return of 1.3417 per month, only 15 basis points higher than return of active strategy with filter threshold of >0.5%. Because the strategy implements 26 Loser/Winner tradings, the increase in trading cost pays off the increase in trading return. It shortens net average monthly excess return spread from 15 to 13/11 basis points between two active strategies after reducing 100/200 basis points of trading cost every turnover.

The table presents that active trading strategy in double directions performs with breakeven transaction costs lower than corresponding ones in single direction due to two main reasons. Firstly, as mentioned before, most active trading strategies in double directions do not enhance return performance of active trading strategies in single direction. Secondly, due to the implementation of opposite long/short trading even if predicted return spread between top and bottom portfolios from recursive out-of-sample forecasting model is negative, active trading strategy in double directions has monthly turnover higher than active trading strategy in single direction. Therefore, the breakeven transaction cost required before the active strategy in double directions become unprofitable is lower.

In short, active trading strategy in double directions, which implements the opposite trading for predicted negative Long/Short return from forecasting model, fails to further improve return performance of active trading strategy in single direction.

### 7.2.3 Active trading strategy on any decile portfolio

Different from active trading strategy based on top and bottom portfolios discussed in previous two subsections, this subsection explores active strategy which goes long position on any decile portfolio with predicted highest return, and short position on any decile portfolio with predicted lowest return. This implies that long/short position can be constructed by any decile momentum portfolio, any decile size portfolio, and any decile value portfolio, respectively.

# • *The trading return*

The research compares the trading return, in terms of average monthly excess return and net average monthly excess return (100bp and 200bp), of active trading strategy

based on any decile portfolio with active trading strategy based on top and bottom decile portfolios and passive trading strategy. It finds that active trading strategy on any equal-weighted decile momentum portfolio significantly outperforms active trading strategy on 'Winner' and 'Loser' portfolios as well as passive trading strategy. Table 7.9 Panel A shows that the strategy on any equal-weighted momentum portfolio with 6 months' holding length, which goes long position on one portfolio with predicted highest return and short position on one portfolio with predicted lowest return, yields average monthly excess return ranging between 0.8039 and 1.2004 when filter thresholds are imposed between 0 to 1.5%, net average monthly excess return ranging between 0.7013 and 1.0338 (between 0.5987 and 0.8671) after reducing 100 (200) basis points of transaction cost every turnover. Three kinds of Long/Short return based on any equal-weighted decile momentum portfolio are much higher than corresponding Long/Short return based on top and bottom momentum portfolios by 31% to 70% as shown in Table 7.1 Panel A. Table 7.10 Panel A and Table 7.2 Panel A indicate the same advantage of active trading strategy on any equal-weighted decile momentum portfolio with 12 months' holding length, which increases Long/Short return of the active strategy on top and bottom momentum portfolios with filter thresholds 0 to 1.0% by 47% to 80%.

[Table 7.9, 7.10, 7.11, and 7.12]

Table 7.9 Panel B shows that active any value-weighted decile momentum portfolio with 6 months' holding length yields average monthly excess return and net average monthly excess return (100bp, 200bp) slightly higher than corresponding returns of passive strategy, but much lower than corresponding returns by active trading strategy based on top and bottom momentum portfolios as shown in Table 7.1 Panel B. For value-weighted portfolios with 12 months' holding length, Long/Short return based on any decile momentum portfolio is negative at most filter thresholds, lower than corresponding ones of passive trading strategy and active trading strategy based on top and bottom momentum portfolios. In short, active trading strategy based on any decile portfolio is no advantage for value-weighted momentum portfolios.

Different from the findings about return enhancement of active strategies on any decile size portfolio and any decial value portfolio in the US Stock Market by Cooper, Gulen, and Vassalou (2001), the empirical work shows that size strategy (12) and value strategy (24) in the UK Stock Market do not obtain the enhancement in Long/Short return based on any decile size and value portfolios. Table 7.11 indicates that active trading strategy on any decile size portfolio with most filter thresholds earns negative Long/Short return. Table 7.12 reports that active trading strategy on any decile value portfolio yields negative Long/Short return before filter thresholds 1.5%.

### • The hit rate

The performance of active trading strategy on any decile portfolio is associated with the hit rate of holding long/short portfolios with predicted highest/lowest return from forecasting model. Table 7.9 Panel A and Table 7.10 Panel A show that at 0 filter threshold the implementation of going long/short position on any equal-weighted decile momentum portfolio with predicted highest/lowest return in 6 and 12 months' holding length yields positive trading return in 84.72 and 75.49 percent of all implementations, respectively. The higher hit rate of the strategy leads to the good return performance of the active strategy on any equal-weighted decile momentum portfolio. But the condition is quite different for other cases whose hit rate of the strategy on any decile portfolio is no more than 68 percent. For instance, the hit rate of the strategy on any equal-(value-) weighted decile size portfolio with predicted highest/lowest return is 0.4559 (0.5343), as shown in Table 7.11.

The statistics 'PSS' on style coefficient  $\gamma$  in section 6.4 of the previous chapter gives the explanation for performance of active trading strategy on any decile portfolio. Because style coefficient  $\gamma$  includes information of all shares in decile portfolios, it reflects the correlation between decile portfolio returns and their firm characteristics. Table 6.3 Panel C shows that 'PSS', the percentage with same sign (positive and negative) between predicted equal-weighted momentum coefficient  $\gamma_{mom}$  and actual one for 6 and 12 months' holding length is 0.8148 and 0.7647, respectively, which is higher than corresponding ones for other style effects. This indicates that the signal from out-of-sample forecasting model has higher possibility to capture any equal-weighted decile momentum portfolio with highest and lowest return over time, which plays important role in return enhancement of active strategy on any equal-weighted momentum portfolio. But 'PSS' for value-weighted momentum portfolios (12/6, 12/12), size portfolios (12), and value portfolios (24) is below 0.74, 0.60, and 0.48, respectively. This reflects that forecasting model offers low signal-to-noise ratio to capture any value-weighted decile momentum portfolio (12/6, 12/12), any decile size portfolio (12), and any decile value portfolio (24) with highest and lowest return.

### • The return on long and short position

The research compares the performance in average monthly return on long and short position between active trading strategy on any decile momentum portfolio and passive holding 'Winner' and 'Loser' portfolios all the time. As shown in Table 7.13, average monthly return of the long position on any equal-weighted momentum portfolio (12/6) with predicted highest return is 1.9316, about 11 basis points (bp) per month higher than one of 'Winner' portfolio in passive trading strategy. Average monthly return of the short position on any equal-weighted momentum portfolio (12/6) with predicted lowest return is 0.2322, about 30bp per month lower than one of 'Loser' portfolio in passive trading strategy. The active long and short strategy on

any equal-weighted momentum portfolio (12/6) with predicted highest and lowest return creates return spread of 41bp (11bp+30bp) more than passive trading strategy. Similar pattern is in the case of equal-weighted momentum portfolio (12/12). The active long position on any equal-weighted momentum portfolio (12/12) with predicted highest return and short position on any portfolio with predicted lowest return yields 15.7bp and 21.9bp per month higher and lower than return on 'Winner' portfolio and return on 'Loser' portfolio in passive trading strategy, respectively, which enhances return spread of passive trading strategy by 37.6bp (15.7bp+21.9bp). In short, for equal-weighted momentum portfolio, both long position and short position in active trading strategy on any decile portfolio contribute to the increase in return spread of passive trading strategy.

### [Table 7.13]

For value-weighted momentum effect (12/6), active long position on any decile portfolio with predicted highest return earns average monthly return of 1.1595, 3.3bp lower than one from passive long position on 'Winner' portfolio. Active short position on any decile portfolio with predicted lowest return earns average monthly return of -0.1934, only 5.4bp lower than one from passive short position on 'Loser' portfolio. As compared to passive strategy of Winner/Loser portfolio, the strategy of active long and short any portfolio has poor and ordinary performance, which leads to the slight increase in return spread of passive trading strategy by 2.1bp (-3.3bp+5.4bp). For value-weighted portfolio (12/12), the active long and short positions yield average monthly return of 0.8543 and 0.5461, 19bp and 11.5bp lower than ones from passive 'Winner' and 'Loser' portfolios, respectively. The performance of active long and short positions on any decile portfolio, that the decline in long return is more than the decline in short return relative to passive trading strategy, results in its failure to enhance return spread of passive trading strategy on any value-weighted momentum portfolio not to outperform passive trading strategy.

As Table 7.13 presents, when active trading strategy on any decile size portfolio (12) is applied, to go long position on any equal-(value-) weighted size portfolio with predicted highest return enhances average monthly return of passive 'Small-cap' position by 1.5bp (16.7bp), from 1.0374 to 1.0527 (from 0.9344 to 1.1017). But to go short position on any equal-(value-) weighted size portfolio with predicted lowest return rises average monthly return of passive 'Large-cap' position by 2.3bp (15.4bp), from 0.8187 to 0.8421 (from 0.7426 to 0.8963). The increasing amount in active long return and active short return is almost same, which leads to no obvious difference in return spread between active trading strategy on any decile size portfolio and passive trading strategy on 'Small' and 'Large' portfolios.

As Table 7.13 reports, active long any equal-(value-) weighted price-to-book portfolio (24) with predicted highest return earns average monthly return of 0.9530 (1.2010), 21bp (63bp) lower than one of passive strategy on low price-to-book portfolio. Conversely, active short any value portfolio with predicted lowest return yields average return of 0.8370 (1.1258) per month, 33bp (27.5bp) higher than one of passive strategy on high price-to-book portfolio. The active strategy on any price-to-book portfolio shortens return spread of the passive strategy. Its failure is consistent with the performance of statistics 'POOS' and 'PSS' of value coefficient  $\gamma_{pb}$ , estimated in section 6.4 of the previous chapter. As Figure 6.8 indicates, 'POOS' of value coefficient  $\gamma_{pb}$  is in the decline in most sample period from 1990 to 2008, which reflects poor out-of-sample performance of forecasting model on any value portfolio return. Table 6.3 Panel C shows that its equal-(value-) weighted 'PSS' is 0.4500 (0.4778). The percentage with same sign (positive and negative) between actual and predicted  $\gamma_{pb}$ , which is estimated from the forecasting model, is the lowest among three style coefficients.

The above empirical analysis on active long and short any decile portfolio implies the profitability in the application of long-only position in momentum effect. By using out-of-sample forecasting model, active long any equal-weighted momentum portfolio (12/6, 12/12) with predicted highest return yields return higher than passive 'Winner' position by 11.0 (15.7) basic points per month. The similar advantage is for active long any value-weighted size portfolio (12) with predicted highest return against passive 'Small' portfilio by 16.7 basis points per month.

### • The inclusion number and frequency each decile portfolio

As mentioned above, in active trading strategy on any decile portfolio the long and short portfolios do not always include top and bottom portfolios all the time, but extends to any one with predicted highest and lowest returns from recursive out-of-sample forecasting model. The research observes active trading strategy on any equal-weighted momentum portfolios (12/6, 12/12) with the enhancement in trading returns against active trading strategy on 'Winner' and 'Loser' portfolios as well as passive trading strategy, to examine how inclusion number and frequency each decile momentum portfolio play role in return enhancement.

# [Table 7.14]

Table 7.14 Panel A reports the inclusion number and frequency for each equal-weighted momentum portfolio (12/6) when long and short positions are gone on any decile portfolio with predicted highest and lowest return from recursive out-of-sample forecasting model, respectively. Different from long on 'Winner' portfolio and short on 'Loser' portfolio in both active trading strategy and passive trading strategy all the time, the  $2^{nd}$  to  $9^{th}$  portfolios together with 'Winner' and

'Loser' portfolios share long and short positions. Besides 83.80 percent of long positions on 'Winner' portfolio, 6.48 and 4.63 percent of long positions is taken up by 'Loser' portfolio and the 9<sup>th</sup> portfolio, respectively. Similarly, 'Loser' portfolio, the 2<sup>nd</sup>, 3<sup>rd</sup>, 4<sup>th</sup> portfolios, and 'Winner' portfolio takes up 39.35, 31.94, 7.87, 10.19 and 7.41 percent of all short positions, respectively. Especially, the active strategy captures the second 'Loser' portfolio in 31.94 percent of all short positions. The inclusion of 'Loser' portfolio and the 9<sup>th</sup> portfolio in the long position, and the inclusion of 'Winner' portfolio and other three portfolios in the short position benefit return performance in active trading strategy on any decile momentum portfolio.

The similar findings for equal-weighted momentum portfolio (12/12) are shown in Panel B. Following 'Winner' portfolio with 78.43 percent, 'Loser' portfolio takes up 12.25 percent of all long positions. 'Loser' portfolio only accounts for 34.8 percent of all short positions. The 4<sup>th</sup>, 2<sup>nd</sup>, and 'Winner' portfolio accounts for 31.37, 14.71, and 14.22 percent of all short positions, respectively. Meanwhile, this reflects that active trading strategy on any decile momentum portfolio allocates 12.25 percent of long positions on 'Loser' portfolio and 14.22 percent of short positions on 'Winner' portfolio due to predicted price reversal in medium run from recursive out-of-sample forecasting model. This prediction is almost consistent with the findings in Table 4.4 in Chapter 4. Among decile momentum portfolios, 'Winner' portfolio yields the lowest return in 8.5 percent of all observations, and 'Loser' portfolio yields the highest return in 17.3 percent of all observations in the UK Stock Market from 1956 to 2008.

Figure 7.2 and 7.3 plot (6, 12) months' moving average of passive Winner/Loser return R<sub>wml</sub> (solid line), active Winner/Loser return R<sub>wml</sub> (dashed line), and active long/short return on any decile portfolio (dashed line) for equal-weighted momentum portfolio (12/6, 12/12), respectively. Panel A shows that dashed line overlaps solid line above X axis, which means that active strategy on Winner/Loser portfolio by using out-of-sample forecasting model correctly captures positive 'Winner/Loser' return R<sub>wml</sub> of passive strategy. Importantly, dashed line is on X axis in 2000 when solid line is below X axis. This reflects that forecasting model offers signal for allocating position on TBILL when passive Winner/Loser portfolio yields negative return. Panel B shows that active strategy on any decile momentum portfolio with expected highest/lowest return by using out-of-sample forecasting model with 0 filter threshold also seize the opportunity to earn positive long/short return as passive trading strategy. Meanwhile, different from active Winner/Loser strategy, it allocates position on non-'Winner' and non-'Loser' portfolios to yield positive long/short return in 2000. Therefore, active Winner/Loser strategy and active strategy on any decile momentum portfolio perform well during the period from 1990 to 2008 because they capture one of three failures of passive momentum strategy, and hold TBILL and allocate position on other long/short portfolios with positive return, respectively. Two active momentum strategies can capture as much upside performance as possible and minimize downside risk.

[Figure 7.2, 7.3]

# 7.3 Active trading strategy on combined two/or three style effects

In this section, the research further explores active trading strategy on combined two or three style effects with 12 months' holding horizon from 1990 to 2008, and compares it with corresponding passive trading strategy.

# 7.3.1 Active trading strategy on combined three style effects

The research compares the performance between active trading strategy and passive trading strategy with combining momentum effect (12/12), size effect (12), and value effect (12) from 1990 to 2008. Table 7.15 Panel A and Panel B show that active trading strategy based on the combination of three equal-(value-) weighted style effects yields 0.2092% to 0.3819% (0.1479% to 0.2437%) of average monthly excess return from 1990 to 2008 at the filter threshold of 0 to 1.0%, higher than one of passive trading strategy which always allocates average position on three style effects all the time. Active trading strategy yields return lower than TBILL rate at 16.18 to 36.76 (17.65 to 42.65) percent of observations, relative to 51.96 percent by passive trading strategy. After reducing trading cost (100bp, 200bp), active trading strategy yields positive average monthly excess return, but passive trading strategy becomes unprofitable.

### [Table 7.15]

The research further observes the allocation of individual style effect in active trading strategy with 0 of filter threshold. Table 7.16 Panel A shows that the strategy with combined equal-weighted three style effects allocates 54.41%, 28.43%, and 17.16% of long/short positions on Winner/Loser portfolio, Small/Large portfolio, and Low/High portfolio, respectively. Because recursive out-of-sample forecasting model predicts positive long/short return for three style effects, the allocation is gone on corresponding long/short position in all 204 observations all the time. Similar to the performance of individual style strategy, long/short position on value effect (12) yields the highest average monthly return (1.3858%) and negative return in the lowest percentage (14.29%) of observations, followed by long/short position on momentum effect (12/12) and size effect (12). Panel B shows that the strategy with combined value-weighted three style effects allocates 43.72%, 26.63%, and 29.65% of long/short positions on Winner/Loser portfolio, Small/Large portfolio, Low/High portfolio, respectively. Because the forecasting model predicts negative long/short returns of three style effects in 5 observations, the allocation is gone on corresponding long/short portfolios in 199 observations and TBILL in 5 observations. Long/short position on momentum effect (12/12) yields the highest average monthly return (1.1324%) and negative return in the lowest percentage (27.59%) of observations, followed by size effect (12) and value effect (12). Different from the equal-weighted case, active trading of long/short value-weighted value portfolios yields negative return in 49.15% of observations, much higher than the corresponding percentage of active trading of long/short value-weighted momentum portfolio and size portfolio.

# [Table 7.16]

### 7.3.2 Active trading strategy on combined two style effects

The research further explores active trading strategy on the combination of momentum effect (12/12) and value effect (12) from 1990 to 2008 for two reasons. Firstly, Table 7.16 in the previous subsection shows that active trading strategy on three equal-(value-) weighted style effects allocates 54.41 (43.72) percent of all long/short positions on Winner/Loser portfolio, much higher than percentage of positions on Small/Large portfolio and Low/High portfolio, and that long/short position on equal-(value-) weighted Winner/Loser portfolio yields average monthly return of 0.8330% (1.1324%), the second highest (highest) return among long/short position on equal-weighted Low/High portfolio in active trading strategy yields average monthly

return of 1.3858%, much higher than the return of long/short position on equal-weighted Winner/Loser portfolio and Small/Large portfolio.

### [Table 7.17]

Table 7.17 Panel A and Panel B show that active trading strategy on the combination of equal-(value-) weighted momentum effect (12/12) and value effect (12) yields 0.4445% and 0.3936% (0.4846% and 0.3456%) of average monthly excess return from 1990 to 2008 at the filter threshold of 0 and 0.5%, enhancing the return of passive trading strategy which always allocates average position on two style effects all the time by more than 60% (70%). The return of active trading strategy is lower than TBILL rate at 30.88 and 23.53 (39.22 and 32.84) percent of observations, relative to 29.90 (47.06) percent by passive trading strategy. Because active trading strategy holds TBILL according to signal and magnitude from forecasting model, it reduces the number (turnover) of long/short position on two style effects and trading cost relative to passive trading strategy which always holds long/short position all the time. Net average monthly excess return (100bp, 200bp) of active trading strategy on two equal-(value-) weighted style effects is more than 0.3229% and 0.2522% (0.2785% and 0.2115%), increasing corresponding return of passive trading strategy by more than 98% (134%).

# [Table 7.18]

As reported in Panel A and Panel B of Table 7.18, active trading strategy allocates 69.15 and 30.85 (48.74 and 51.26) percent of all long/short positions on equal-(value-) weighted Winner/Loser portfolio and Low/High portfolio, which yields 0.9050% and 0.9926% (1.1414% and 0.8332%) of average monthly return, respectively. Different from active trading strategy on the combination of three value-weighted style effects, the position on long/short value effect (12) in active trading strategy on the combination of two style effects takes up 51.26 percent of all long/short positions, higher than corresponding percentage of the position on long/short momentum effect (12/12). The position of long/short size effect (12) in active strategy on the combination of three style effects is allocated to the position of long/short value effect (12) more than the position of long/short momentum effect (12) in active strategy on the combination of two style effects. As reported in Panel B1 of Table 4.9 and 4.12, the research finds that average value-weighted return spread of size effect (12) and value effect (12) is 0.20 and 1.01 per month from 1979(1980) to 2008, respectively, The position on long/short value effect (12) with higher return spread replaces for the position on long/short size effect (12) with lower return spread, which leads to the large increase in average monthly excess return from 0.1969 to 0.4846 at filter threshold 0, as reported in Panel B of Table 7.17.

# 7.4 Conclusion

Higher 'PSS', the statistics to measure the performance of recursive out-of-sample forecasting model in Chapter 6, motivates the exploration of active trading strategy on style effects in the UK Stock Market from 1990 to 2008. The empirical result is promising for timing momentum effect and size effect by using active trading strategy on top and bottom portfolios. With signal and magnitude of out-of-sample forecasting model and positive filter threshold, active trading strategy on Winner/Loser portfolio and Small/Large portfolio can capture the time of significant momentum effect (12/6, 12/12) and small effect (12), and is more profitable but less risky than traditional passive strategy which always goes long/short position on Winner/Loser portfolio and Small/Large portfolio each month. The active strategy in momentum effect (12/6, 12/12) and size effect (12) succeeds in return enhancement of passive trading strategy due to its high hit rate. After reducing 100/200 basis points of trading cost each turnover, only active trading strategy on 'Winner/Loser' portfolio maintains advantage, but active trading strategy on 'Small/Large' portfolio becomes unprofitable. Active trading strategy fails to benefit value effect (24). High percentage of observations with positive passive Low/High return R<sub>lmh</sub>, 81.67% (69.44%) for equal-(value-) weighted case shown in Panel A and B of Table 7.4, offers small space to enhance the return of passive strategy. Meanwhile, poor out-of-sample signal from forecasting model results in disappointed performance of an attempt to capture the time of value effect (24).

Active trading strategy on top/bottom portfolio with imposing positive and negative filter thresholds, which implements opposite trading by using signal and magnitude from out-of-sample forecasting model, fails to improve return performance of active trading strategy with imposing positive filter threshold, except for momentum effect (12/6). This failure is due to the forecasting model not to offer high percentage of successful opposite trading signal (measured by hit rate) for long/short position on Loser/Winner portfolio (12/12), Large/Small portfolio (12), and High/Low portfolio (24). In addition, the inclusion of opposite trading leads to the increase in turnover and trading cost, and further weakens the outperformance of active strategy on momentum effect (12/6) in double directions over active strategy in single direction.

Active trading strategy on any equal-weighted decile momentum portfolio (12/6, 12/12) significantly outperforms active trading strategy on 'Winner' and 'Loser' portfolios as well as passive trading strategy. The implementation of active long/short position on any equal-weighted decile momentum portfolio with predicted highest/lowest return from out-of-sample forecasting model yields positive return at higher percentage of all observations. The active long position and short position on any equal-weighted decile momentum portfolio has more chance to capture one portfolio whose return is higher than one of passive long 'Winner' portfolio, and one

portfolio whose return is lower than one of passive short 'Loser' portfolio. This leads to the increase (decrease) in the return of active long (short) any momentum portfolio relative to passive long (short) 'Winner' ('Loser') portfolio. The inclusion of any equal-weighted decile momentum portfolio in the long and short positions outperforms active trading strategy on Winner/Loser portfolio. But return performance of the strategy on any value-weighted decile momentum portfolio (12/6, 12/12), any decile size portfolio (12), and any decile value portfolio (24) is disappointed. The empirical result on active long and short any decile portfolio implies the profitability in the application of long-only position on one style. By using out-of-sample forecasting model, active long any equal-weighted momentum portfolio (12/6, 12/12) and any value-weighted size portfolio (12) outperforms corresponding passive long 'Winner' portfolio and 'Small' portfolio, respectively.

The empirical work in Chapter 4 presents the negative correlation between momentum effect and size effect, between momentum effect and value effect, the positive correlation between size effect and value effect. It implies the possibility of active trading strategy with combined two/or three style effects to outperform corresponding passive trading strategy which always allocates average position on three/ or two style effects all the time. The research finds that active trading strategy based on the combination of momentum effect (12/12), size effect (12), and value effect (12) yields higher average monthly excess return at the filter threshold of 0 to 1.0%, and positive return at higher percentage of all observations than passive trading strategy from 1990 to 2008. Relative to active trading strategy with combined three style effects, active strategy on the combination of momentum effect (12/12) and value effect (12) performs better because the long/short position on Winner/Loser portfolio and Low/High portfolio with higher return spread replaces for the position on Small/Large portfolio with lower return spread.

In summary, the research explores the profitability of active trading strategy to combine top-down thematic ideas using economic and market series with bottom-up stock selection using specific share information. It finds that a variety of active trading strategies benefits passive absolute return strategy on momentum effect and size effect in the UK Stock Market from 1990 to 2008. For individual passive style investing, equal-(value-)weighted Winner/Loser portfolio in momentum effect (12/6) yields highest average monthly excess return of 0.7899 (0.8322), followed by Low/High portfolio in value effect (24). Active trading strategy has most significant impact on performance enhancement for momentum effect (12/6). Using recursive out-of-sample forecasting model with macroeconomic variables, such as GDP(Y), UEXTBILL(Y), DY(Q), EMR(-1)/CPI(H), active strategy on any equal-weighted decile momentum portfolio (12/6) with 0 filter threshold increases average monthly excess return of passive strategy from 0.7899 to 1.2004, and decreases the percentage with negative return in observations from 20.83% to 15.28%. Active strategy on

value-weighted Winner/Loser portfolio in momentum effect (12/6) with 0.5 filter threshold increases average monthly excess return from 0.8322 of passive strategy to 1.1901, and decreases the percentage with negative return in observations from 28.70% to 15.28%.

Recursive out-of-sample forecasting model has capability to enhance performance of passive trading strategy with average position allocation on three/or two style effects all the time. Although active trading strategy on combined two/or three style effects outperforms corresponding passive strategy, its performance is behind active strategy on momentum effect (12/6). Active trading strategy on combined equal-(value-) weighted momentum effect (12/12) and value effect (12) earns average excess return of 0.4445 (0.4846) per month, and negative return in 22.06 (32.84) percent of observations. Comparatively, active strategy on any equal-weighted decile momentum portfolio (12/6) with 0 filter threshold earns average excess return of 1.2004 per month, and negative return in 15.28 percent of observations. Active strategy on value-weighted Winner/Loser portfolio in momentum effect (12/6) with 0.5 filter threshold earns average excess return of 1.1901 per month, and negative return in 15.28 percent of observations. Therefore, the exploration of active trading strategy on style effects indicates that style-oriented investors in the UK Stock Market are better trying to time momentum effect (12/6), either any equal-weighted decile momentum portfolio or value-weighted Winner/Loser portfolio.

# **Appendix: tables and figures in Chapter 7**



Figure 7.1 Equal-weighted monthly Small/Large return  $R_{\mbox{\scriptsize sml}}$  from 1980 to 2008

Note: The figure plots series equal-weighted monthly Small/Large return  $R_{sml}$  from 1980 to 2008, which is estimated by the spread in 12 months' holding return between 'Small-cap' portfolio and 'Large-cap' portfolio.

		Panel A	A Equal-weighte	ed					
		Active trading strategy							
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan	
Average monthly excess return	0.9135	0.8630	0.7508	0.5252	0.2826	0.1524	0.0365	0.7899	
Standard deviation	2.4667	2.4512	2.3782	2.3066	2.0915	1.7600	1.6137	2.7472	
Sharpe ratio	0.3703	0.3521	0.3157	0.2277	0.1351	0.0866	0.0226	0.2875	
%<0	0.1667	0.1528	0.1296	0.1111	0.0741	0.0509	0.0370	0.2083	
% <monthly tbill<="" td=""><td>0.2222</td><td>0.2037</td><td>0.1759</td><td>0.1250</td><td>0.0833</td><td>0.0602</td><td>0.0509</td><td>0.2685</td></monthly>	0.2222	0.2037	0.1759	0.1250	0.0833	0.0602	0.0509	0.2685	
Number of holding Winner/Loser portfolio	191	178	155	112	66	43	27	216	
Hit rate of holding Winner/Loser portfolio	0.8115	0.7472	0.6839	0.6071	0.5455	0.4884	0.4815	0.7917	
Number of holding TBILL	25	38	61	104	150	173	189	Nan	
Hit rate of holding TBILL	0.3600	0.3947	0.4754	0.5192	0.6333	0.6994	0.7725	Nan	
Hit rate of strategy	0.7593	0.6852	0.6250	0.5648	0.6065	0.6574	0.7361	0.7917	
Monthly turnover	0.8843	0.8241	0.7176	0.5185	0.3056	0.1991	0.1250	1.0000	
Net average monthly excess return (100bp)	0.7661	0.7257	0.6312	0.4388	0.2317	0.1192	0.0157	0.6232	
Net average monthly excess return (200bp)	0.6187	0.5883	0.5116	0.3524	0.1807	0.0860	-0.0052	0.4566	
Breakeven transaction cost	0.5165	0.5236	0.5232	0.5065	0.4625	0.3828	0.1458	0.3949	
		Panel F	3 Value-weighte	ed					
				Passive trading strategy					
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan	
Average monthly excess return	0.9970	1.1901	1.1055	0.8673	0.7068	0.6051	0.2488	0.8322	
Standard deviation	4.0093	3.4293	3.1586	2.7957	2.4758	2.0884	1.5366	4.5252	
Sharpe ratio	0.2487	0.3470	0.3500	0.3102	0.2855	0.2897	0.1619	0.1839	
%<0	0.2083	0.1528	0.1111	0.0741	0.0417	0.0093	0.0093	0.2870	
% <monthly tbill<="" td=""><td>0.2315</td><td>0.1713</td><td>0.1296</td><td>0.0972</td><td>0.0602</td><td>0.0278</td><td>0.0278</td><td>0.3102</td></monthly>	0.2315	0.1713	0.1296	0.0972	0.0602	0.0278	0.0278	0.3102	
Number of holding Winner/Loser portfolio	174	145	111	78	46	26	12	216	
Hit rate of holding Winner/Loser portfolio	0.7414	0.7310	0.7027	0.6410	0.6957	0.8077	0.5833	0.713	
Number of holding TBILL	42	71	105	138	170	190	204	Nan	
Hit rate of holding TBILL	0.4048	0.4507	0.4286	0.4638	0.5529	0.6526	0.6814	Nan	
Hit rate of strategy	0.6759	0.6389	0.5694	0.5278	0.5833	0.6713	0.6759	0.7130	
Monthly turnover	0.8056	0.6713	0.5139	0.3611	0.2130	0.1204	0.0556	1.0000	
Net average monthly excess return (100bp)	0.8628	1.0783	1.0199	0.8072	0.6714	0.5851	0.2396	0.6656	
Net average monthly excess return (200bp)	0.7286	0.9664	0.9342	0.7470	0.6359	0.5650	0.2303	0.4989	
Breakeven transaction cost	0.6189	0.8864	1.0756	1,2009	1.6595	2,5134	2,2394	0.4161	

Table 7.1 The performance of active trading strategy on top and bottom momentum portfolios (12/6) with positive filter thresholds

Note: The table reports the performance of active trading strategy on top and bottom momentum portfolios based on 12 months' formation length and 6 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted Winner/Loser return  $R_{wnl}$  from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Winner/Loser return meets filter threshold, the portfolio is gone long position on 'Winner' portfolio and short position on 'Loser' one. Otherwise, it is invested in TBILL. The passive strategy goes long position on 'Winner' portfolio' reports the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Winner/Loser portfolio' reports the percentage of trading return statisfied with filter threshold in Winner/Loser trading according to signal and magnitude from forecasting model. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The passive strategy model. The row labeled 'Hit rate of holding Unner/Loser portfolio' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding trategy or 200 basis point transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

		Panel	A Equal-weight	ted				
			Acti	ve trading strategy	7			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.3643	0.3031	0.1241	0.0011	-0.0982	-0.1576	-0.0408	0.1615
Standard deviation	1.9269	1.8578	1.6884	1.5479	1.4706	1.1936	0.6293	2.3640
Sharpe ratio	0.1890	0.1631	0.0735	0.0007	-0.0668	-0.1320	-0.0648	0.0683
%<0	0.1814	0.1520	0.0833	0.0588	0.0588	0.0490	0.0098	0.2598
% <monthly tbill<="" td=""><td>0.2157</td><td>0.1863</td><td>0.0931</td><td>0.0637</td><td>0.0637</td><td>0.0539</td><td>0.0147</td><td>0.3235</td></monthly>	0.2157	0.1863	0.0931	0.0637	0.0637	0.0539	0.0147	0.3235
Number of holding Winner/Loser portfolio	161	137	71	39	27	16	3	204
Hit rate of holding Winner/Loser portfolio	0.7702	0.7153	0.6479	0.5897	0.4815	0.2500	0.3333	0.7402
Number of holding TBILL	43	67	133	165	177	188	201	Nan
Hit rate of holding TBILL	0.3721	0.4179	0.4812	0.6545	0.7571	0.8723	0.9104	Nan
Hit rate of strategy	0.6863	0.6176	0.5392	0.6422	0.7206	0.8235	0.9020	0.7402
Monthly turnover	0.7892	0.6716	0.3480	0.1912	0.1324	0.0784	0.0147	1.0000
Net average monthly excess return (100bp)	0.2985	0.2471	0.0951	-0.0149	-0.1093	-0.1642	-0.0420	0.0781
Net average monthly excess return (200bp)	0.2327	0.1911	0.0661	-0.0308	-0.1203	-0.1707	-0.0433	-0.0052
Breakeven transaction cost	0.2308	0.2257	0.1783	0.0029	-0.3710	-1.0046	-1.3857	0.0808
		Panel	B Value-weigh	ted				
			Active t	rading strategy				Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.2904	0.3148	0.2633	0.1579	0.1052	0.0190	0.0000	-0.0967
Standard deviation	2.5600	2.2460	1.9254	1.1725	0.8267	0.4793	0.0000	3.4190
Sharpe ratio	0.1134	0.1401	0.1368	0.1347	0.1273	0.0397	-0.0700	-0.0283
%<0	0.1912	0.1324	0.0686	0.0196	0.0098	0.0049	0.0000	0.3529
% <monthly tbill<="" td=""><td>0.2059</td><td>0.1422</td><td>0.0784</td><td>0.0245</td><td>0.0147</td><td>0.0098</td><td>0.0049</td><td>0.3725</td></monthly>	0.2059	0.1422	0.0784	0.0245	0.0147	0.0098	0.0049	0.3725
Number of holding Winner/Loser portfolio	129	92	61	19	10	4	0	204
Hit rate of holding Winner/Loser portfolio	0.6977	0.6957	0.6885	0.7368	0.7000	0.5000	Nan	0.6471
Number of holding TBILL	75	112	143	185	194	200	204	Nan
Hit rate of holding TBILL	0.4400	0.4554	0.5315	0.6000	0.6959	0.7400	0.8382	Nan
Hit rate of strategy	0.6029	0.5637	0.5784	0.6127	0.6961	0.7353	0.8382	0.6471
Monthly turnover	0.6324	0.4510	0.2990	0.0931	0.0490	0.0196	0.0000	1.0000
Net average monthly excess return (100bp)	0.2376	0.2771	0.2384	0.1501	0.1011	0.0173	0.0000	-0.1801
Net average monthly excess return (200bp)	0.1849	0.2395	0.2134	0.1424	0.0970	0.0157	0.0000	-0.2635
Breakeven transaction cost	0.2296	0.3490	0.4403	0.8478	1.0735	0.4852	Nan	-0.0484

Table 7.2 The performance of active trading strategy on top and bottom momentum portfolios (12/12) with positive filter thresholds

Note: The table reports the performance of active trading strategy on top and bottom momentum portfolios with 12 months' formation length and 12 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted Winner/Loser return  $R_{wnl}$  from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Winner/Loser return meets filter threshold, the portfolio is gone long position on 'Winner' portfolio and short position on 'Loser' one. Otherwise, it is invested in TBILL. The passive strategy goes long position on 'Winner' portfolio and short position on 'Loser' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Winner/Loser portfolio' reports the percentage of trading return satisfied with filter threshold in Winner/Loser trading according to signal and magnitude from forecasting model. The row labeled 'Hit rate of holding TBILL' reports the percentage of successful holding long/short portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position. 'Net average monthly turnover.' Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return (100bp,200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover.' Breakeven trading cost is defined as the fixed trading cost that makes monthly excess return of very terrors.' Net average to zero.'

		Panel	A Equal-weight	ted				
			Activ	ve trading strategy				Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	-0.1618	0.1064	0.1201	0.1117	0.0466	0.0192	0.0161	-0.2636
Standard deviation	1.9005	1.6974	1.1911	0.8207	0.6083	0.4699	0.2930	2.1711
Sharpe ratio	-0.0851	0.0627	0.1008	0.1360	0.0766	0.0409	0.0549	-0.1214
%<0	0.3725	0.1716	0.0735	0.0441	0.0294	0.0196	0.0049	0.5735
% <monthly tbill<="" td=""><td>0.4510</td><td>0.2206</td><td>0.0980</td><td>0.0588</td><td>0.0392</td><td>0.0294</td><td>0.0098</td><td>0.6716</td></monthly>	0.4510	0.2206	0.0980	0.0588	0.0392	0.0294	0.0098	0.6716
Number of holding Small/Large portfolio	140	86	53	40	23	17	6	204
Hit rate of holding Small/Large portfolio	0.4571	0.4535	0.4717	0.4250	0.2174	0.0588	0.0000	0.4265
Number of holding TBILL	64	118	151	164	181	187	198	Nan
Hit rate of holding TBILL	0.6406	0.8051	0.8477	0.8720	0.8619	0.8717	0.8889	Nan
Hit rate of strategy	0.5147	0.6569	0.7500	0.7843	0.7892	0.8039	0.8627	0.4265
Monthly turnover	0.6863	0.4216	0.2598	0.1961	0.1127	0.0833	0.0294	1.0000
Net average monthly excess return (100bp)	-0.2190	0.0712	0.0984	0.0953	0.0372	0.0122	0.0136	-0.3470
Net average monthly excess return (200bp)	-0.2762	0.0361	0.0768	0.0789	0.0278	0.0053	0.0111	-0.4303
Breakeven transaction cost	-0.1179	0.1262	0.2312	0.2847	0.2066	0.1153	0.2735	-0.1318
		Panel	B Value-weight	ted				
			Activ	ve trading strategy				Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	-0.1031	0.0409	0.0737	0.0425	-0.0044	0.0035	0.0000	-0.2905
Standard deviation	1.6758	1.1337	0.7406	0.6011	0.4296	0.2652	0.0000	1.9962
Sharpe ratio	-0.0615	0.0361	0.0995	0.0707	-0.0103	0.0130	-0.0700	-0.1455
%<0	0.2794	0.0882	0.0441	0.0294	0.0245	0.0049	0.0000	0.5343
% <monthly tbill<="" td=""><td>0.3971</td><td>0.1471</td><td>0.0833</td><td>0.0490</td><td>0.0392</td><td>0.0098</td><td>0.0049</td><td>0.6716</td></monthly>	0.3971	0.1471	0.0833	0.0490	0.0392	0.0098	0.0049	0.6716
Number of holding Small/Large portfolio	128	61	40	24	15	4	0	204
Hit rate of holding Small/Large portfolio	0.5547	0.5082	0.5250	0.2917	0.1333	0.0000	Nan	0.4657
Number of holding TBILL	76	143	164	180	189	200	204	Nan
Hit rate of holding TBILL	0.6842	0.7762	0.8415	0.8444	0.8624	0.8800	0.9020	Nan
Hit rate of strategy	0.6029	0.6961	0.7794	0.7794	0.8088	0.8627	0.9020	0.4657
Monthly turnover	0.6275	0.2990	0.1961	0.1176	0.0735	0.0196	0.0000	1.0000
Net average monthly excess return (100bp)	-0.1554	0.0160	0.0573	0.0326	-0.0106	0.0018	0.0000	-0.3738
Net average monthly excess return (200bp)	-0.2077	-0.0090	0.0410	0.0228	-0.0167	0.0001	0.0000	-0.4572
Breakeven transaction cost	-0.0822	0.0684	0.1880	0.1806	-0.0301	0.0882	Nan	-0.1452

Table 7.3 The performance of active trading strategy on top and bottom size portfolios (12) with positive filter thresholds

Note: The table reports the performance of active trading strategy on top and bottom size portfolios based on 12 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted Small/Large return  $R_{sml}$  from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Small/Large return meets filter threshold, the portfolio is gone long position on 'Small' portfolio and short position on 'Large' one. Otherwise, it is invested in TBILL. The passive strategy always goes long position on 'Small' portfolio and short position on 'Large' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return satisfied with filter threshold in Small/Large trading according to signal and magnitude from recursive out-of-sample forecasting model. The row labeled 'Hit rate of holding TBILL' reports the percentage of trading return satisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of holding Small/Large portfolio and TBILL' reports the percentage of successful holding Small/Large portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position in the observations. 'Net average monthly excess return (100bp,200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven trading cost is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

		Panel	A Equal-weight	ted				
		Passive trading strategy						
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.1414	0.0149	0.0000	0.0000	0.0000	0.0000	0.0000	0.2002
Standard deviation	0.9100	0.2599	0.0000	0.0000	0.0000	0.0000	0.0000	1.3939
Sharpe ratio	0.1554	0.0574	Nan	Nan	Nan	Nan	Nan	0.1437
%<0	0.0500	0.0167	0.0000	0.0000	0.0000	0.0000	0.0000	0.1833
% <monthly tbill<="" td=""><td>0.2000</td><td>0.0444</td><td>0.0000</td><td>0.0000</td><td>0.0000</td><td>0.0000</td><td>0.0000</td><td>0.4611</td></monthly>	0.2000	0.0444	0.0000	0.0000	0.0000	0.0000	0.0000	0.4611
Number of holding Low/High portfolio	90	23	0	0	0	0	0	180
Hit rate of holding low/High portfolio	0.9000	0.6087	Nan	Nan	Nan	Nan	Nan	0.8167
Number of holding TBILL	90	157	180	180	180	180	180	Nan
Hit rate of holding TBILL	0.2667	0.5223	0.6944	0.8000	0.8444	0.9000	0.9500	Nan
Hit rate of strategy	0.5833	0.5333	0.6944	0.8000	0.8444	0.9000	0.9500	0.8167
Monthly turnover	0.5000	0.1278	0.0000	0.0000	0.0000	0.0000	0.0000	1.0000
Net average monthly excess return (100bp)	0.1205	0.0095	0.0000	0.0000	0.0000	0.0000	0.0000	0.1585
Net average monthly excess return (200bp)	0.0997	0.0042	0.0000	0.0000	0.0000	0.0000	0.0000	0.1169
Breakeven transaction cost	0.1414	0.0584	Nan	Nan	Nan	Nan	Nan	0.1001
		Panel	B Value-weigh	ted				
			Acti	ve trading strategy	7			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.2009	0.0709	0.0852	0.0284	0.0082	0.0000	0.0000	0.5260
Standard deviation	1.3423	1.0014	0.4425	0.2505	0.1005	0.0000	0.0000	1.8534
Sharpe ratio	0.1497	0.0708	0.1925	0.1134	0.0811	Nan	Nan	0.2838
%<0	0.2444	0.1500	0.0167	0.0111	0.0000	0.0000	0.0000	0.3056
% <monthly tbill<="" td=""><td>0.3111</td><td>0.1889</td><td>0.0500</td><td>0.0389</td><td>0.0111</td><td>0.0000</td><td>0.0000</td><td>0.3778</td></monthly>	0.3111	0.1889	0.0500	0.0389	0.0111	0.0000	0.0000	0.3778
Number of holding Low/High portfolio	138	76	27	16	5	0	0	180
Hit rate of holding low/High portfolio	0.6812	0.5395	0.4444	0.1875	0.0000	Nan	Nan	0.6944
Number of holding TBILL	42	104	153	164	175	180	180	Nan
Hit rate of holding TBILL	0.2619	0.3173	0.4837	0.5976	0.7200	0.8056	0.8500	Nan
Hit rate of strategy	0.5833	0.4111	0.4778	0.5611	0.7000	0.8056	0.8500	0.6944
Monthly turnover	0.7667	0.4222	0.1500	0.0889	0.0278	0.0000	0.0000	1.0000
Net average monthly excess return (100bp)	0.1690	0.0532	0.0789	0.0247	0.0070	0.0000	0.0000	0.4843
Net average monthly excess return (200bp)	0.1370	0.0356	0.0727	0.0210	0.0058	0.0000	0.0000	0.4426
Breakeven transaction cost	0.1311	0.0839	0.2840	0.1598	0.1468	Nan	Nan	0.2630

Table 7.4 The performance of active trading strategy on top and bottom value portfolios (24) with positive filter thresholds

Note: The table reports the performance of active trading strategy on top and bottom price-to-book portfolios on 24 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted Low/High return R<sub>im</sub> from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Low/High return meets filter threshold, the portfolio is gone long position on 'Low' portfolio and short position on 'High' one. Otherwise, it is invested in TBILL. The passive strategy goes long position on 'Low' portfolio and short position on 'High' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Low/High portfolio' reports the percentage of realing return satisfied with filter threshold in Low/High trading according to signal and magnitude from recursive out-of-sample forecasting model. Hit rate of holding TBILL' reports the percentage of real return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of real return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of real return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding Low/High portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position in the observations. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

			Panel A Equal-v	veighted					
		Active trading strategy							
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan	
Average monthly excess return	0.9380	0.8980	0.7071	0.4943	0.2720	0.1524	0.0365	0.7899	
Standard deviation	2.6671	2.5761	2.4090	2.3273	2.0988	1.7600	1.6137	2.7472	
Sharpe ratio	0.3517	0.3486	0.2935	0.2124	0.1296	0.0866	0.0226	0.2875	
%<0	0.2407	0.1991	0.1574	0.1296	0.0787	0.0509	0.0370	0.2083	
% <monthly tbill<="" td=""><td>0.2963</td><td>0.2500</td><td>0.2037</td><td>0.1435</td><td>0.0880</td><td>0.0602</td><td>0.0509</td><td>0.2685</td></monthly>	0.2963	0.2500	0.2037	0.1435	0.0880	0.0602	0.0509	0.2685	
Number of holding Winner/Loser portfolio	191	178	155	112	66	43	27	216	
Hit rate of holding Winner/Loser portfolio	0.8115	0.7472	0.6839	0.6071	0.5455	0.4884	0.4815	0.7917	
Number of holding Loser/Winner portfolio	25	15	6	4	1	0	0	Nan	
Hit rate of holding Loser/Winner portfolio	0.3600	0.3333	0.0000	0.0000	0.0000	Nan	Nan	Nan	
Number of holding TBILL	0	23	55	100	149	173	189	Nan	
Hit rate of holding TBILL	Nan	0.1739	0.2727	0.4100	0.5570	0.6185	0.7090	Nan	
Hit rate of strategy	0.7593	0.6574	0.5602	0.5046	0.5509	0.5926	0.6806	0.7917	
Monthly turnover	1.00	0.89	0.75	0.54	0.31	0.20	0.13	1.00	
Net average monthly excess return (100bp)	0.7714	0.7491	0.5829	0.4048	0.2203	0.1193	0.0157	0.6232	
Net average monthly excess return (200bp)	0.6047	0.6002	0.4587	0.3153	0.1686	0.0861	-0.0052	0.4566	
Breakeven transaction cost	0.4690	0.5025	0.4743	0.4602	0.4384	0.3828	0.1458	0.3949	

Table 7.5 The performance of active trading strategy on top and bottom momentum portfolios (12/6) with positive and negative filter thresholds
Panel B Value-weighted										
		Active trading strategy								
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan		
Average monthly excess return	0.9835	1.3417	1.1683	0.8915	0.6883	0.6051	0.2488	0.8322		
Standard deviation	4.5004	3.8245	3.4035	2.8653	2.4961	2.0884	1.5366	4.5252		
Sharpe ratio	0.2185	0.3508	0.3433	0.3111	0.2758	0.2897	0.1619	0.1839		
%<0	0.3241	0.2083	0.1481	0.0787	0.0463	0.0093	0.0093	0.2870		
% <monthly tbill<="" td=""><td>0.3565</td><td>0.2315</td><td>0.1667</td><td>0.1019</td><td>0.0648</td><td>0.0278</td><td>0.0278</td><td>0.3102</td></monthly>	0.3565	0.2315	0.1667	0.1019	0.0648	0.0278	0.0278	0.3102		
Number of holding Winner/Loser portfolio	174	145	111	78	46	26	12	216		
Hit rate of holding Winner/Loser portfolio	0.7414	0.7310	0.7027	0.6410	0.6957	0.8077	0.5833	0.713		
Number of holding Loser/Winner portfolio	42	26	16	3	1	0	0	Nan		
Hit rate of holding Loser/Winner portfolio	0.4048	0.5000	0.4375	0.3333	0.0000	Nan	Nan	Nan		
Number of holding TBILL	0	45	89	135	169	190	204	Nan		
Hit rate of holding TBILL	Nan	0.0667	0.1573	0.2148	0.3669	0.4789	0.5294	Nan		
Hit rate of strategy	0.6759	0.5648	0.4583	0.3704	0.4352	0.5185	0.5324	0.7130		
Monthly turnover	1.00	0.79	0.59	0.38	0.22	0.12	0.06	1.00		
Net average monthly excess return (100bp)	0.8168	1.2098	1.0703	0.8291	0.6521	0.5851	0.2396	0.6656		
Net average monthly excess return (200bp)	0.6502	1.0778	0.9723	0.7666	0.6158	0.5650	0.2303	0.4989		
Breakeven transaction cost	0.4917	0.8474	0.9935	1.1887	1.5816	2.5134	2.2394	0.4161		

Note: The table reports the performance of active trading strategy with positive and negative filter thresholds on top and bottom momentum portfolios (12/6) based on 12 months' formation length and 6 months' holding length from 1990 to 2008. The strategy imposes positive/negative filter threshold on predicted Winner/Loser return  $R_{wml}$  from 0 to +/-3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Winner/Loser return meets positive filter threshold, the portfolio is gone long position on 'Winner' portfolio and short position on 'Loser' one. When predicted Winner/Loser return meets negative filter threshold, the portfolio is gone long position on 'Winner' one. Otherwise, it is invested in TBILL. The passive strategy always goes long position on 'Winner' portfolio and short position on 'Loser' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding the percentage of trading return satisfied with negative filter threshold in Loser/Winner trading. The row labeled 'Hit rate of holding the percentage of trading return satisfied with negative filter threshold in Loser/Winner trading. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding Winner/Loser portfolio, Loser/Winner portfolio, and TBILL in observations. 'Not hy turnover' is the percentage of holding long/short position in the observations. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven transaction cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

Panel A Equal-weighted										
		Active trading strategy								
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan		
Average monthly excess return	0.3855	0.2916	0.0708	-0.0287	-0.1159	-0.1653	-0.0408	0.1615		
Standard deviation	2.2812	2.1159	1.7825	1.5593	1.4772	1.1977	0.6293	2.3640		
Sharpe ratio	0.1690	0.1378	0.0397	-0.0184	-0.0785	-0.1380	-0.0648	0.0683		
%<0	0.3137	0.2598	0.1471	0.0833	0.0735	0.0539	0.0098	0.2598		
% <monthly tbill<="" td=""><td>0.3627</td><td>0.3039</td><td>0.1667</td><td>0.0980</td><td>0.0784</td><td>0.0588</td><td>0.0147</td><td>0.3235</td></monthly>	0.3627	0.3039	0.1667	0.0980	0.0784	0.0588	0.0147	0.3235		
Number of holding Winner/Loser portfolio	161	137	71	39	27	16	3	204		
Hit rate of holding Winner/Loser portfolio	0.7702	0.7153	0.6479	0.5897	0.4815	0.2500	0.3333	0.7402		
Number of holding Loser/Winner portfolio	43	31	16	7	3	1	0	Nan		
Hit rate of holding Loser/Winner portfolio	0.3721	0.2258	0.0625	0.0000	0.0000	0.0000	Nan	Nan		
Number of holding TBILL	0	36	117	158	174	187	201	Nan		
Hit rate of holding TBILL	Nan	0.1111	0.3077	0.5253	0.6667	0.7968	0.8209	Nan		
Hit rate of strategy	0.6863	0.5343	0.4069	0.5196	0.6324	0.7500	0.8137	0.7402		
Monthly turnover	1.0000	0.8235	0.4265	0.2255	0.1471	0.0833	0.0147	1.0000		
Net average monthly excess return (100bp)	0.3021	0.2229	0.0352	-0.0476	-0.1282	-0.1723	-0.0420	0.0781		
Net average monthly excess return (200bp)	0.2187	0.1543	-0.0004	-0.0664	-0.1405	-0.1792	-0.0433	-0.0052		
Breakeven transaction cost	0.1927	0.1770	0.0830	-0.0637	-0.3941	-0.9918	-1.3857	0.0808		

 Table 7.6 The performance of active trading strategy on top and bottom momentum portfolios (12/12) with positive and negative filter thresholds

Panel B Value-weighted									
			Acti	ve trading strategy	7			Passive trading strategy	
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan	
Average monthly excess return	0.2801	0.1542	0.0217	0.0916	0.0961	0.0190	0.0000	-0.0967	
Standard deviation	3.3978	2.7248	2.1376	1.2466	0.8382	0.4793	0.0000	3.4190	
Sharpe ratio	0.0824	0.0566	0.0101	0.0735	0.1146	0.0397	-0.0700	-0.0283	
%<0	0.3971	0.2549	0.1618	0.0539	0.0147	0.0049	0.0000	0.3529	
% <monthly tbill<="" td=""><td>0.4461</td><td>0.2941</td><td>0.1765</td><td>0.0588</td><td>0.0196</td><td>0.0098</td><td>0.0049</td><td>0.3725</td></monthly>	0.4461	0.2941	0.1765	0.0588	0.0196	0.0098	0.0049	0.3725	
Number of holding Winner/Loser portfolio	129	92	61	19	10	4	0	204	
Hit rate of holding Winner/Loser portfolio	0.6977	0.6957	0.6885	0.7368	0.7000	0.5000	Nan	0.6471	
Number of holding Loser/Winner portfolio	75	37	21	7	1	0	0	Nan	
Hit rate of holding Loser/Winner portfolio	0.4400	0.1892	0.0000	0.0000	0.0000	Nan	Nan	Nan	
Number of holding TBILL	0	75	122	178	193	200	204	Nan	
Hit rate of holding TBILL	0.0000	0.0667	0.2377	0.3596	0.4819	0.5800	0.7059	Nan	
Hit rate of strategy	0.6029	0.3725	0.3480	0.3824	0.4902	0.5784	0.7059	0.6471	
Monthly turnover	1.0000	0.6324	0.4020	0.1275	0.0539	0.0196	0.0000	1.0000	
Net average monthly excess return (100bp)	0.1968	0.1015	-0.0119	0.0809	0.0915	0.0173	0.0000	-0.1801	
Net average monthly excess return (200bp)	0.1134	0.0488	-0.0454	0.0703	0.0870	0.0157	0.0000	-0.2635	
Breakeven transaction cost	0.1401	0.1219	0.0270	0.3592	0.8908	0.4852	Nan	-0.0484	

Note: The table reports the performance of active trading strategy with positive and negative filter thresholds on top and bottom momentum portfolios (12/12) based on 12 months' formation length and 12 months' holding length from 1990 to 2008. The strategy imposes positive/negative filter threshold on expected Winner/Loser return R<sub>wml</sub> from 0 to +/-3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Winner/Loser return meets positive filter threshold, the portfolio is gone long position on 'Winner' portfolio and short position on 'Loser' one. When predicted Winner/Loser return meets negative filter threshold, the portfolio is gone long position on 'Winner' one. Otherwise, it is invested in TBILL. The passive strategy always goes long position on 'Winner' portfolio and short position on 'Loser' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Winner/Loser portfolio' reports the percentage of trading return satisfied with negative filter threshold in Loser/Winner trading. The row labeled 'Hit rate of holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of actual return satisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of holding Unner/Loser portfolio, Loser/Winner portfolio, and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position in the observations. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven transaction cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

		F	Panel A Equal-we	ighted					
		Active trading strategy							
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan	
Average monthly excess return	-0.3465	-0.0601	-0.0405	0.0501	0.0267	0.0192	0.0161	-0.2636	
Standard deviation	2.1402	1.9380	1.4765	0.9929	0.6725	0.4699	0.2930	2.1711	
Sharpe ratio	-0.1619	-0.0310	-0.0274	0.0505	0.0398	0.0409	0.0549	-0.1214	
%<0	0.4853	0.2304	0.1225	0.0588	0.0343	0.0196	0.0049	0.5735	
% <monthly tbill<="" td=""><td>0.6225</td><td>0.2990</td><td>0.1667</td><td>0.0784</td><td>0.0441</td><td>0.0294</td><td>0.0098</td><td>0.6716</td></monthly>	0.6225	0.2990	0.1667	0.0784	0.0441	0.0294	0.0098	0.6716	
Number of holding Small/Large portfolio	140	86	53	40	23	17	6	204	
Hit rate of holding Small/Large portfolio	0.4571	0.4535	0.4717	0.4250	0.2174	0.0588	0.0000	0.4265	
Number of holding Large/Small portfolio	64	27	18	7	1	0	0	Nan	
Hit rate of holding Large/Small portfolio	0.6406	0.4074	0.0556	0.0000	0.0000	Nan	Nan	Nan	
Number of holding TBILL	0	91	133	157	180	187	198	Nan	
Hit rate of holding TBILL	Nan	0.3187	0.5564	0.7325	0.7667	0.8128	0.8535	Nan	
Hit rate of strategy	0.5147	0.3873	0.4902	0.6471	0.7010	0.7500	0.8284	0.4265	
Monthly turnover	1.0000	0.5539	0.3480	0.2304	0.1176	0.0833	0.0294	1.0000	
Net average monthly excess return (100bp)	-0.4299	-0.1063	-0.0696	0.0309	0.0169	0.0122	0.0136	-0.3470	
Net average monthly excess return (200bp)	-0.5133	-0.1524	-0.0986	0.0117	0.0071	0.0053	0.0111	-0.4303	
Breakeven transaction cost	-0.1733	-0.0542	-0.0582	0.1088	0.1137	0.1153	Nan	-0.1318	

Table 7.7 The performance of active trading strategy on top and bottom size portfolios (12) with positive and negative filter thresholds

		Panel	B Value-weight	ed				
			Active	trading strategy				Passive trading strategy
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan
Average monthly excess return	-0.2584	-0.1389	-0.0625	0.0195	-0.0044	0.0035	0.0000	-0.2905
Standard deviation	1.9524	1.4417	1.0628	0.6794	0.4296	0.2652	0.0000	1.9962
Sharpe ratio	-0.1323	-0.0964	-0.0588	0.0287	-0.0103	0.0130	-0.0700	-0.1455
%<0	0.3971	0.1520	0.0833	0.0343	0.0245	0.0049	0.0000	0.5343
% <monthly tbill<="" td=""><td>0.5833</td><td>0.2500</td><td>0.1471</td><td>0.0588</td><td>0.0392</td><td>0.0098</td><td>0.0049</td><td>0.6716</td></monthly>	0.5833	0.2500	0.1471	0.0588	0.0392	0.0098	0.0049	0.6716
Number of holding Small/Large portfolio	128	61	40	24	15	4	0	204
Hit rate of holding Small/Large portfolio	0.5547	0.5082	0.5250	0.2917	0.1333	0.0000	Nan	0.4657
Number of holding Large/Small portfolio	76	28	14	2	0	0	0	Nan
Hit rate of holding Large/Small portfolio	0.6842	0.2143	0.0000	0.0000	Nan	Nan	Nan	Nan
Number of holding TBILL	0	115	150	178	189	200	204	Nan
Hit rate of holding TBILL	Nan	0.3826	0.6067	0.7022	0.7619	0.8250	0.8578	Nan
Hit rate of strategy	0.6029	0.3971	0.5490	0.6471	0.7157	0.8088	0.8578	0.4657
Monthly turnover	1.0000	0.4363	0.2647	0.1275	0.0735	0.0196	0.0000	1.0000
Net average monthly excess return (100bp)	-0.3418	-0.1753	-0.0846	0.0088	-0.0106	0.0018	0.0000	-0.3738
Net average monthly excess return (200bp)	-0.4251	-0.2117	-0.1066	-0.0018	-0.0167	0.0001	0.0000	-0.4572
Breakeven transaction cost	-0.1292	-0.1592	-0.1180	0.0765	-0.0301	0.0882	Nan	-0.1452

Note: The table reports the performance of active trading strategy with positive and negative filter thresholds on top and bottom size portfolios (12) based on 12 months' holding length from 1990 to 2008. The strategy imposes positive/negative filter threshold on predicted Small/Large return  $R_{sml}$  from 0 to +/-3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Small/Large return meets positive filter threshold, the portfolio is gone long position on 'Small' portfolio and short position on 'Large' one. When predicted Small/Large return meets negative filter threshold, the portfolio is gone long position on 'Large' portfolio and short position on 'Small' one. Otherwise, it is invested in TBILL. The passive strategy always goes long position on 'Small' portfolio and short position on 'Large' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Small/Large portfolio' reports the percentage of trading return satisfied with negative filter threshold in Large/Small trading. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in Large/Small trading. The row labeled 'Hit rate of holding Small/Large, Large/Small, and TBILL in observations. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven transaction cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

			Ac	tive trading strateg	у			Passive trading strategy
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan
Average monthly excess return	-0.3711	-0.1621	-0.0131	0.0000	0.0000	0.0000	0.0000	0.2003
Standard deviation	1.5269	0.6429	0.1226	0.0000	0.0000	0.0000	0.0000	1.3939
Sharpe ratio	-0.2430	-0.2522	-0.1068	Nan	Nan	Nan	Nan	0.143
%<0	0.4167	0.1278	0.0111	0.0000	0.0000	0.0000	0.0000	0.1833
% <monthly tbill<="" td=""><td>0.6500</td><td>0.1778</td><td>0.0167</td><td>0.0000</td><td>0.0000</td><td>0.0000</td><td>0.0000</td><td>0.4611</td></monthly>	0.6500	0.1778	0.0167	0.0000	0.0000	0.0000	0.0000	0.4611
Number of holding Low/High portfolio	90	23	0	0	0	0	0	180
Hit rate of holding low/High portfolio	0.9000	0.6087	Nan	Nan	Nan	Nan	Nan	0.8167
Number of holding High/Low portfolio	90	24	3	0	0	0	0	Nar
Hit rate of holding High/Low portfolio	0.2667	0.0000	0.0000	Nan	Nan	Nan	Nan	Nar
Number of holding TBILL	Nan	133	177	180	180	180	180	Nar
Hit rate of holding TBILL	0.0000	0.4436	0.6158	0.7389	0.7889	0.8611	0.9278	Nar
Hit rate of strategy	0.5833	0.4056	0.6056	0.7389	0.7889	0.8611	0.9278	0.8167
Monthly turnover	1.0000	0.2611	0.0167	0.0000	0.0000	0.0000	0.0000	1.0000
Net average monthly excess return (100bp)	-0.4128	-0.1731	-0.0138	0.0000	0.0000	0.0000	0.0000	0.158
Net average monthly excess return (200bp)	-0.4545	-0.1839	-0.0145	0.0000	0.0000	0.0000	0.0000	0.1169
Breakeven transaction cost	-0.1856	-0.3105	-0.3926	Nan	Nan	Nan	Nan	0.100

Table 7.8 The performance of active trading strategy on top and bottom value portfolios (24) with positive and negative filter thresholds

Panel D Value-weighted									
			Act	tive trading strates	gy			Passive trading strategy	
Filter threshold	>0, <0	>0.5, <-0.5	>1.0, <-1.0	>1.5, <-1.5	>2.0, <-2.0	>2.5, <-2.5	>3.0, <-3.0	Nan	
Average monthly excess return	-0.3184	-0.2382	-0.0299	-0.0003	0.0082	0.0000	0.0000	0.5260	
Standard deviation	2.0340	1.5496	0.9082	0.4618	0.1005	0.0000	0.0000	1.8534	
Sharpe ratio	-0.1565	-0.1537	-0.0329	-0.0007	0.0811	Nan	Nan	0.2838	
%<0	0.4167	0.2278	0.0389	0.0167	0.0000	0.0000	0.0000	0.3056	
% <monthly tbill<="" td=""><td>0.4833</td><td>0.2667</td><td>0.0722</td><td>0.0444</td><td>0.0111</td><td>0.0000</td><td>0.0000</td><td>0.3778</td></monthly>	0.4833	0.2667	0.0722	0.0444	0.0111	0.0000	0.0000	0.3778	
Number of holding Low/High portfolio	138	76	27	16	5	0	0	180	
Hit rate of holding low/High portfolio	0.6812	0.5395	0.4444	0.1875	0.0000	Nan	Nan	0.6944	
Number of holding High/Low portfolio	42	15	4	1	0	0	0	Nan	
Hit rate of holding High/Low portfolio	0.2619	0.0667	0.0000	0.0000	Nan	Nan	Nan	Nan	
Number of holding TBILL	0	89	149	163	175	180	180	Nan	
Hit rate of holding TBILL	Nan	0.0899	0.2617	0.4908	0.7086	0.8056	0.8500	Nan	
Hit rate of strategy	0.5833	0.2778	0.2833	0.4611	0.6889	0.8056	0.8500	0.6944	
Monthly turnover	1.0000	0.5056	0.1722	0.0944	0.0278	0.0000	0.0000	1.0000	
Net average monthly excess return (100bp)	-0.3601	-0.2593	-0.0371	-0.0043	0.0070	0.0000	0.0000	0.4843	
Net average monthly excess return (200bp)	-0.4018	-0.2804	-0.0443	-0.0083	0.0058	0.0000	0.0000	0.4426	
Breakeven transaction cost	-0.1592	-0.2356	-0.0867	Nan	Nan	Nan	Nan	0.2630	

Note: The table reports the performance of active trading strategy with positive and negative filter thresholds on top and bottom price-to-book portfolios based on 24 months' holding length from 1990 to 2008. The strategy imposes positive/negative filter threshold on predicted Low/High return  $R_{lmh}$  from 0 to +/-3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted Low/High return meets positive filter threshold, the portfolio is gone long position on 'Low' portfolio and short position on 'High' one. When predicted Low/High return meets negative filter threshold, the portfolio and short position on 'Low' portfolio and short position on 'High' portfolio each month. The row labeled '%<0' and '%<monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Low/High portfolio' reports the percentage of trading return satisfied with positive filter threshold in High/Low trading. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return satisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of holding Low/High portfolio, and TBILL in observations. 'Not average monthly postfolio Low/High portfolio, High/Low portfolio, and TBILL in observations. 'Monthly turnover' is the percentage of holding low/High portfolio, High/Low portfolio, and TBILL in observations. 'Monthly turnover' is the percentage of holding low/High portfolio, High/Low portfolio, and TBILL in observations. 'Not average monthly excess return (100bp, 200bp)' is trading return after reducin

		Panel A	Equal-weight	ed				
			Active	trading strateg	<u>y</u>		]	Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	1.2004	1.1962	1.0828	0.8039	0.4388	0.3423	0.1625	0.7899
Standard deviation	2.0097	2.0110	1.9939	1.8262	1.6007	1.3239	1.1406	2.7472
Sharpe ratio	0.5973	0.5949	0.5431	0.4402	0.2741	0.2585	0.1425	0.2875
%<0	0.1528	0.1528	0.1343	0.0972	0.0648	0.0324	0.0278	0.2083
% <monthly tbill<="" td=""><td>0.2407</td><td>0.2361</td><td>0.1991</td><td>0.1296</td><td>0.0787</td><td>0.0417</td><td>0.0417</td><td>0.2685</td></monthly>	0.2407	0.2361	0.1991	0.1296	0.0787	0.0417	0.0417	0.2685
Number of holding portfolio with predicted highest/Lowest	216	214	182	133	73	50	32	216
Hit rate of holding portfolio with predicted highest/Lowest return	0.8472	0.7617	0.7033	0.5940	0.5479	0.5400	0.4375	0.7917
Number of holding TBILL	0	2	34	83	143	166	184	Nan
Hit rate of holding TBILL	NaN	0.5000	0.4412	0.5301	0.6224	0.6988	0.7772	Nan
Hit rate of strategy	0.8472	0.7593	0.6620	0.5694	0.5972	0.6620	0.7269	0.7917
Monthly turnover	1.0000	0.9907	0.8426	0.6157	0.3380	0.2315	0.1481	1.00
Net average monthly excess return (100bp)	1.0338	1.0312	0.9425	0.7013	0.3825	0.3037	0.1378	0.6232
Net average monthly excess return (200bp)	0.8671	0.8660	0.8020	0.5987	0.3262	0.2651	0.1131	0.4566
Breakeven transaction cost	0.6002	0.6037	0.6426	0.6528	0.6491	0.7393	0.5484	0.3949
		Panel B	Value-weight	ed				
			Activ	e trading strateg	у			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.8539	0.8757	0.7781	0.8191	0.6841	0.5823	0.2213	0.8322
Standard deviation	3.9881	3.9432	3.6555	2.9920	2.5909	2.0083	1.3991	4.5252
Sharpe ratio	0.2141	0.2221	0.2129	0.2737	0.2640	0.2899	0.1582	0.1839
	0.3333	0.3194	0.2315	0.1296	0.0787	0.0139	0.0093	0.2870
% <monthly tbill<="" td=""><td>0.3796</td><td>0.3657</td><td>0.2546</td><td>0.1481</td><td>0.0926</td><td>0.0278</td><td>0.0231</td><td>0.3102</td></monthly>	0.3796	0.3657	0.2546	0.1481	0.0926	0.0278	0.0231	0.3102
Number of holding portfolio with predicted highest/Lowest	216	208	160	107	68	28	12	216
Hit rate of holding portfolio with predicted highest/Lowest return	0.6667	0.6154	0.5688	0.5234	0.5735	0.7857	0.5833	0.713
Number of holding TBILL	0	8	56	109	148	188	204	Nan
Hit rate of holding TBILL	NaN	0.3750	0.5357	0.5780	0.6419	0.7287	0.7353	Nan
Hit rate of strategy	0.6667	0.6065	0.5602	0.5509	0.6204	0.7361	0.7269	0.7130
Monthly turnover	1.0000	0.9630	0.7407	0.4954	0.3148	0.1296	0.0556	1.0000
Net average monthly excess return (100bp)	0.6873	0.7153	0.6547	0.7365	0.6317	0.5607	0.2121	0.6656
Net average monthly excess return (200bp)	0.5206	0.5548	0.5312	0.6540	0.5792	0.5391	0.2028	0.4989
Breakeven transaction cost	0.4270	0.4547	0.5252	0.8267	1.0865	2.2460	1.9919	0.4161

Table 7.9 The performance of active trading strategy on any decile momentum portfolio (12/6)

Note: The table reports the performance of active trading strategy on any decile momentum portfolio based on 12 months' formation length and 6 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted highest/lowest portfolio return from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted highest/lowest portfolio return meets filter threshold, the portfolio is gone long position on the portfolio with highest predicted return and short position on the portfolio with predicted lowest return. Otherwise, it is invested in TBILL. The row labeled "40" and "%<br/>Monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding portfolio with predicted highest/lowest return' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding long/short portfolio and TBILL in observations. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

		Panel A	Equal-weight	ted				
			Active	e trading strate	gy			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.5373	0.4639	0.2079	-0.0871	-0.1064	-0.1781	-0.0309	0.1615
Standard deviation	1.9991	1.9498	1.7810	1.3617	1.3240	1.1662	0.6456	2.3640
Sharpe ratio	0.2688	0.2379	0.1168	-0.0640	-0.0804	-0.1527	-0.0478	0.0683
%<0	0.2451	0.2255	0.1569	0.0931	0.0735	0.0539	0.0098	0.2598
% <monthly tbill<="" td=""><td>0.3137</td><td>0.2941</td><td>0.1863</td><td>0.1225</td><td>0.0882</td><td>0.0637</td><td>0.0196</td><td>0.3235</td></monthly>	0.3137	0.2941	0.1863	0.1225	0.0882	0.0637	0.0196	0.3235
Number of holding portfolio with predicted highest/Lowest	204	187	109	51	34	17	4	204
Hit rate of holding portfolio with predicted highest/Lowest return	0.7549	0.6684	0.5413	0.4118	0.3824	0.1176	0.2500	0.7402
Number of holding TBILL	0	17	95	153	170	187	200	Nan
Hit rate of holding TBILL	NaN	0.2353	0.4421	0.6078	0.7294	0.8235	0.8800	Nan
Hit rate of strategy	0.7549	0.6324	0.4951	0.5588	0.6716	0.7647	0.8676	0.7402
Monthly turnover	1.0000	0.9167	0.5343	0.2500	0.1667	0.0833	0.0196	1.0000
Net average monthly excess return (100bp)	0.4539	0.3875	0.1634	-0.1080	-0.1204	-0.1851	-0.0326	0.0781
Net average monthly excess return (200bp)	0.3706	0.3111	0.1188	-0.1289	-0.1343	-0.1921	-0.0342	-0.0052
Breakeven transaction cost	0.2687	0.2530	0.1946	-0.1743	-0.3193	-1.0687	-0.7871	0.0808
		Panel B	Value-weight	ted				
			Acti	ve trading strates	gy			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	-0.1740	-0.0548	-0.0973	-0.2444	-0.0093	0.0080	-0.0014	-0.0967
Standard deviation	2.7495	2.6487	2.2083	1.8576	1.1684	0.8260	0.0194	3.4190
Sharpe ratio	-0.0633	-0.0207	-0.0441	-0.1316	-0.0080	0.0097	-0.0700	-0.0283
%<0	0.4167	0.3382	0.2059	0.1471	0.0637	0.0196	0.0000	0.3529
% <monthly tbill<="" td=""><td>0.4951</td><td>0.4069</td><td>0.2696</td><td>0.1863</td><td>0.0833</td><td>0.0343</td><td>0.0098</td><td>0.3725</td></monthly>	0.4951	0.4069	0.2696	0.1863	0.0833	0.0343	0.0098	0.3725
Number of holding portfolio with predicted highest/Lowest	204	176	107	60	27	11	1	204
Hit rate of holding portfolio with predicted highest/Lowest return	0.5833	0.5341	0.3925	0.3000	0.2963	0.3636	0.0000	0.6471
Number of holding TBILL	0	28	97	144	177	193	203	Nan
Hit rate of holding TBILL	NaN	0.6786	0.5670	0.6458	0.7345	0.8187	0.8719	Nan
Hit rate of strategy	0.5833	0.5539	0.4755	0.5441	0.6765	0.7941	0.8676	0.6471
Monthly turnover	1.0000	0.8627	0.5245	0.2941	0.1324	0.0539	0.0049	1.0000
Net average monthly excess return (100bp)	-0.2574	-0.1268	-0.1410	-0.2690	-0.0204	0.0035	-0.0018	-0.1801
Net average monthly excess return (200bp)	-0.3407	-0.1987	-0.1848	-0.2935	-0.0314	-0.0010	-0.0022	-0.2635
Breakeven transaction cost	-0.0870	-0.0318	-0.0927	-0.4155	-0.0351	0.0746	-0.1383	-0.0484

Table 7.10 The performance of active trading strategy on any decile momentum portfolio (12/12)

Note: The table reports the performance of active trading strategy on any decile momentum portfolio based on 12 months' formation length and 12 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted highest/lowest portfolio return from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted highest/lowest portfolio return meets filter threshold, the portfolio is gone long position on the portfolio with highest predicted return and short position on the portfolio with predicted lowest return. Otherwise, it is invested in TBILL. The row labeled '%<0' and '%<Monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding portfolio with predicted highest/lowest return' reports the percentage of trading on predicted highest/lowest return according to signal from recursive out-of-sample forecasting model. The row labeled 'Hit rate of holding TBILL' reports the percentage of strategy on and monthly TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding long/short portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position in observations. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis points transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

		Panel A	Equal-weighte	ed				
			Activ	ve trading strateg	у			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	-0.2716	-0.2290	-0.0470	-0.0405	0.0232	-0.0041	-0.0228	-0.2636
Standard deviation	1.7624	1.7121	1.4483	1.0112	0.8753	0.7064	0.5419	2.1711
Sharpe ratio	-0.1541	-0.1338	-0.0324	-0.0400	0.0265	-0.0057	-0.0421	-0.1214
%<0	0.5441	0.4706	0.2108	0.1078	0.0637	0.0392	0.0196	0.5735
% <monthly tbill<="" td=""><td>0.6324</td><td>0.5539</td><td>0.2500</td><td>0.1275</td><td>0.0735</td><td>0.0490</td><td>0.0294</td><td>0.6716</td></monthly>	0.6324	0.5539	0.2500	0.1275	0.0735	0.0490	0.0294	0.6716
Number of holding portfolio with predicted highest/Lowest	204	177	94	54	36	21	11	204
Hit rate of holding portfolio with predicted highest/Lowest return	0.4559	0.3559	0.3298	0.3148	0.2778	0.1905	0.0000	0.4265
Number of holding TBILL	0	27	110	150	168	183	193	Nan
Hit rate of holding TBILL	NaN	0.6296	0.7636	0.8533	0.9345	0.9399	0.9534	Nan
Hit rate of strategy	0.4559	0.3922	0.5637	0.7108	0.8186	0.8627	0.9020	0.4265
Monthly turnover	1.0000	0.8676	0.4608	0.2647	0.1765	0.1029	0.0539	1.0000
Net average monthly excess return (100bp)	-0.3550	-0.3014	-0.0854	-0.0626	0.0084	-0.0127	-0.0273	-0.3470
Net average monthly excess return (200bp)	-0.4383	-0.3737	-0.1238	-0.0846	-0.0063	-0.0213	-0.0318	-0.4303
Breakeven transaction cost	-0.1358	-0.1320	-0.0510	-0.0765	0.0657	-0.0197	-0.2115	-0.1318
		Panel B	Value-weighte	ed				
			Activ	ve trading strateg	У			Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	-0.2768	-0.1889	-0.1476	-0.0782	-0.0513	-0.0154	-0.0215	-0.2905
Standard deviation	1.6742	1.5717	1.0913	0.9064	0.7295	0.4638	0.3075	1.9962
Sharpe ratio	-0.1653	-0.1202	-0.1352	-0.0863	-0.0703	-0.0331	-0.0700	-0.1455
%<0	0.4657	0.3382	0.1520	0.0833	0.0392	0.0098	0.0049	0.5343
% <monthly tbill<="" td=""><td>0.5490</td><td>0.4167</td><td>0.1814</td><td>0.0980</td><td>0.0490</td><td>0.0147</td><td>0.0098</td><td>0.6716</td></monthly>	0.5490	0.4167	0.1814	0.0980	0.0490	0.0147	0.0098	0.6716
Number of holding portfolio with predicted highest/Lowest	204	152	70	43	20	6	1	204
Hit rate of holding portfolio with predicted highest/Lowest return	0.5343	0.4276	0.2857	0.2558	0.2500	0.0000	0.0000	0.4657
Number of holding TBILL	0	52	134	161	184	198	203	Nan
Hit rate of holding TBILL	NaN	0.5962	0.7313	0.8323	0.9076	0.9495	0.9704	Nan
Hit rate of strategy	0.5343	0.4706	0.5784	0.7108	0.8431	0.9216	0.9657	0.4657
Monthly turnover	1.0000	0.7451	0.3431	0.2108	0.0980	0.0294	0.0049	1.0000
Net average monthly excess return (100bp)	-0.3602	-0.2510	-0.1762	-0.0958	-0.0595	-0.0179	-0.0220	-0.3738
Net average monthly excess return (200bp)	-0.4435	-0.3131	-0.2048	-0.1134	-0.0677	-0.0203	-0.0224	-0.4572
Breakeven transaction cost	-0.1384	-0.1267	-0.2150	-0.1855	-0.2615	-0.2611	-2.1963	-0.1452

 Table 7.11 The performance of active trading strategy on any decile size portfolio (12)

Note: The table reports the performance of active trading strategy on any decile size portfolio based on 12 months' holding length from 1990 to 2008. The strategy imposes filter threshold on predicted highest/lowest portfolio return from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted highest/lowest portfolio return meets filter threshold, the portfolio is gone long position on the portfolio and short position on the portfolio with predicted lowest return. Otherwise, it is invested in TBILL. The passive strategy always goes long position on 'Small' portfolio and short position on 'Large' portfolio each month. The row labeled '%<0' and '%<Monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding portfolio with predicted highest/lowest return according to signal from recursive out-of-sample forecasting model. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return unsatisfied with filter threshold in he trading of predicted highest/lowest return according to signal from recursive out-of-sample forecasting model. The row labeled 'Hit rate of strategy' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of actual return unsatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' reports the percentage of holding long/short portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short portfolio and trategy reports the percentage of holding long/short portfolio and trategy to zero.

	Panel A Equal-weighted									
			Activ	e trading strateg	У			Passive trading strategy		
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan		
Average monthly excess return	-0.3386	-0.2883	-0.0173	-0.0068	0.0000	0.0000	0.0000	0.2002		
Standard deviation	0.7585	0.6738	0.2125	0.0522	0.0000	0.0000	0.0000	1.3939		
Sharpe ratio	-0.4465	-0.4279	-0.0813	-0.1299	Nan	Nan	Nan	0.1437		
%<0	0.3278	0.2944	0.0389	0.0111	0.0000	0.0000	0.0000	0.1833		
% <monthly tbill<="" td=""><td>0.6944</td><td>0.6111</td><td>0.1222</td><td>0.0167</td><td>0.0000</td><td>0.0000</td><td>0.0000</td><td>0.4611</td></monthly>	0.6944	0.6111	0.1222	0.0167	0.0000	0.0000	0.0000	0.4611		
Number of holding portfolio with predicted highest/Lowest	180	157	35	4	0	0	0	180		
Hit rate of holding portfolio with predicted highest/Lowest return	0.6722	0.2611	0.1143	0.0000	Nan	Nan	Nan	0.8167		
Number of holding TBILL	0	23	145	176	180	180	180	Nan		
Hit rate of holding TBILL	Nan	0.8261	0.9517	0.9773	0.9889	1.0000	1.0000	Nan		
Hit rate of strategy	0.6722	0.3333	0.7889	0.9556	0.9889	1.0000	1.0000	0.8167		
Monthly turnover	1.0000	0.8722	0.1944	0.0222	0.0000	0.0000	0.0000	1.0000		
Net average monthly excess return (100bp)	-0.3803	-0.3247	-0.0254	-0.0077	0.0000	0.0000	0.0000	0.1585		
Net average monthly excess return (200bp)	-0.4220	-0.3610	-0.0335	-0.0087	0.0000	0.0000	0.0000	0.1169		
Breakeven transaction cost	-0.1693	-0.1653	Nan	Nan	Nan	Nan	Nan	0.1001		
		Panel B	Value-weighte	d						
			Activ	e trading strateg	у			Passive trading strategy		
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan		
Average monthly excess return	-0.3794	-0.3355	-0.0910	0.1019	0.0475	0.0156	0.0000	0.5260		
Standard deviation	1.4916	1.4615	1.1929	0.6071	0.2804	0.1219	0.0000	1.8534		
Sharpe ratio	-0.2544	-0.2296	-0.0763	0.1679	0.1695	0.1282	Nan	0.2838		
%<0	0.4611	0.4389	0.1889	0.0167	0.0056	0.0000	0.0000	0.3056		
% <monthly tbill<="" td=""><td>0.5667</td><td>0.5444</td><td>0.2444</td><td>0.0333</td><td>0.0222</td><td>0.0000</td><td>0.0000</td><td>0.3778</td></monthly>	0.5667	0.5444	0.2444	0.0333	0.0222	0.0000	0.0000	0.3778		
Number of holding portfolio with predicted highest/Lowest	180	175	97	36	18	5	0	180		
Hit rate of holding portfolio with predicted highest/Lowest return	0.5389	0.4286	0.3196	0.3889	0.1111	0.0000	Nan	0.6944		
Number of holding TBILL	0	5	83	144	162	175	180	Nan		
Hit rate of holding TBILL	Nan	0.8000	0.8313	0.8819	0.9383	0.9657	0.9944	Nan		
Hit rate of strategy	0.5389	0.4389	0.5556	0.7833	0.8556	0.9389	0.9944	0.6944		
Monthly turnover	1.0000	0.9722	0.5389	0.2000	0.1000	0.0278	0.0000	1.0000		
Net average monthly excess return (100bp)	-0.4212	-0.3761	-0.1135	0.0936	0.0433	0.0144	0.0000	0.4843		
Net average monthly excess return (200bp)	-0.4628	-0.4166	-0.1359	0.0852	0.0391	0.0133	0.0000	0.4426		
Breakeven transaction cost	-0.1897	-0.1725	-0.0844	0.2548	0.2376	Nan	Nan	0.2630		

Table 7.12 The performance of active trading strategy on any decile value portfolio (24)

Note: The table reports the performance of active trading strategy on any decile price-to-book portfolio based on 24 months' holding length from 1990 to 2008. The strategy imposes filter threshold, the portfolio is gone long position on the portfolio eturn from 0 to 3.0% in increments of 0.5%, from recursive out-of-sample forecasting model. When predicted highest/lowest portfolio return meets filter threshold, the portfolio is gone long position on the portfolio with highest predicted return and short position on the portfolio with predicted lowest return. Otherwise, it is invested in TBILL. The passive strategy always goes long position on low PE/BE portfolio and short position on high PE/BE portfolio each month. The row labeled '%<0' and '%<Monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding portfolio with predicted highest/lowest return' reports the percentage of trading return usatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of holding TBILL' reports the percentage of actual return usatisfied with filter threshold in holding TBILL. The row labeled 'Hit rate of strategy' is the percentage of successful holding IDBILL. The row labeled 'Hit rate of strategy' is the percentage of holding long/short position in observations. 'Net average monthly excess return (100/200bp)' is trading return after reducing 100/200 basis point transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

• •		Active trac	ling strategy	Passive tradi	ng strategy
	-	Long	Short	Long	Short
Momentum effect (12/6)	EW	1.9316	0.2322	1.8220	0.5333
	VW	1.1595	-0.1934	1.1923	-0.1390
Momentum effect (12/12)	EW	1.8017	0.7821	1.6451	1.0013
	VW	0.8543	0.5461	1.0467	0.6612
Size effect (12)	EW	1.0527	0.8421	1.0374	0.8187
	VW	1.1017	0.8963	0.9344	0.7426
Value effect (24)	EW	0.9530	0.8370	1.1616	0.5067
	VW	1.2010	1.1258	1.8312	0.8505

Table 7.13 Average monthly long return and short return of active trading strategy on any decile portfolio and passive trading strategy on top and bottom portfolio

Note: The table reports average monthly return on long position and short position for active trading strategy on any decile portfolio and passive trading strategy on top and bottom portfolio. Active trading strategy on any decile portfolio goes long/short position on any decile portfolio with expected highest/lowest return which is estimated by recursive out-of-sample forecasting model. The passive strategy always goes long/short on top/or bottom decile portfolios each month. Momentum effect (12/6, 12/12) represents decile momentum portfolios with 12 months' formation length and 6, 12 months' holding length, respectively. Size effect (12) represents decile size portfolios with 12 months' holding length. Value effect (24) represents decile value portfolios with 24 months' holding length

Panel A Momentum effect (12/6)											
	Portfolio	P <sub>1</sub> (Loser)	$P_2$	$P_3$	$P_4$	P <sub>5</sub>	$P_6$	$P_7$	P <sub>8</sub>	<b>P</b> <sub>9</sub>	P <sub>10</sub> (Wiinner)
Long position	Number	14	0	4	4	1	0	1	1	10	181
	Frequency	0.0648	0.0000	0.0185	0.0185	0.0046	0.0000	0.0046	0.0046	0.0463	0.8380
Short position	Number	85	69	17	22	4	2	0	1	0	16
	Frequency	0.3935	0.3194	0.0787	0.1019	0.0185	0.0093	0.0000	0.0046	0.0000	0.0741
				Panel	B Momentum	effect (12/12)					
	Portfolio	P <sub>1</sub> (Loser)	$P_2$	P <sub>3</sub>	$P_4$	P <sub>5</sub>	$P_6$	$P_7$	P <sub>8</sub>	P <sub>9</sub>	P <sub>10</sub> (Wiinner)
Long position	Number	25	3	2	2	11	0	0	0	1	160
	Frequency	0.1225	0.0147	0.0098	0.0098	0.0539	0.0000	0.0000	0.0000	0.0049	0.7843
Short position	Number	71	30	0	64	0	3	0	6	1	29
-	Frequency	0.3480	0.1471	0.0000	0.3137	0.0000	0.0147	0.0000	0.0294	0.0049	0.1422

Table 7.14 Inclusion number and frequency each portfolio in active trading strategy on any equal-weighted decile momentum portfolio (12/6, 12/12)

Note: This table reports the inclusion number and frequency each equal-weighted decile momentum portfolio which is included in the long and short positions by using the active trading strategy on any decile portfolio. Long position is gone on one portfolio with predicted highest return, and short position is gone on one portfolio with predicted lowest return according to estimation of portfolio returns from recursive out-of-sample forecasting model. Panel A reports corresponding data on equal-weighted decile portfolios with 12 months' formation length and 6 months' holding length. Panel B reports corresponding data on equal-weighted decile portfolios with 12 months' formation length.



Figure 7.2 The performance of 6 months' moving average passive and active long/short return for equal-weighted momentum effect (12/6) Panel A Active Winner/Loser return R<sub>wml</sub> and passive Winner/Loser return R<sub>wml</sub>

Panel B Active long/short return on any decile momentum portfolio and passive Winner/Loser return R<sub>wml</sub>



Note: The dashed line and solid line in Panel A plot series 6 months' moving average of active Winner/Loser return  $R_{wml}$  and passive Winner/Loser return  $R_{wml}$  from 1990 to 2008. The dashed line and solid line in Panel B plot series 6 months' moving average active long/short return on any decile momentum portfolio and passive Winner/Loser return  $R_{wml}$ . Equal-weighted momentum portfolios are formed with past 6 months' share returns and hold for 6 months. Passive Winner/Loser return  $R_{wml}$  is estimated by the return spread between 'Winner' portfolio and 'Loser' portfolio and 'Loser' portfolio by using recursive out-of-sample forecasting model with 0 filter threshold. Active long/short return momentum portfolio is estimated by the return spread between out-of-sample forecasting model with 0 filter threshold.



Figure 7.3 The performance of 12 months' moving average passive and active long/short return for equal-weighted momentum effect (12/12)

Panel B Active long/short return on any decile momentum portfolio and passive Winner/Loser return R<sub>wml</sub>



Note: The dashed line and solid line in Panel A plot series 12 months' moving average of active Winner/Loser return  $R_{wml}$  and passive Winner/Loser return  $R_{wml}$  from 1990 to 2008. The dashed line and solid line in Panel B plot series 12 months' moving average active long/short return on any decile momentum portfolio and passive Winner/Loser return  $R_{wml}$ . Equal-weighted momentum portfolios are formed with past 12 months' share returns and hold for 12 months. Passive Winner/Loser return  $R_{wml}$  is estimated by the return spread between 'Winner' portfolio and 'Loser' portfolio is estimated by the return spread between 'Winner' portfolio and 'Loser' portfolio and

			Panel A Equal	-weighted				
	_	Passive trading strategy						
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.3819	0.3459	0.2092	0.0996	-0.0521	-0.1384	-0.0247	0.0076
Standard deviation	2.2763	2.2562	2.0310	1.7444	1.5943	1.2852	0.6951	0.7332
Sharpe ratio	0.1678	0.1533	0.1030	0.0571	-0.0327	-0.1077	-0.0355	0.1040
%<0	0.2696	0.2549	0.1373	0.1029	0.0882	0.0686	0.0147	0.2108
% <monthly tbill<="" td=""><td>0.3676</td><td>0.3235</td><td>0.1618</td><td>0.1127</td><td>0.0931</td><td>0.0735</td><td>0.0147</td><td>0.5196</td></monthly>	0.3676	0.3235	0.1618	0.1127	0.0931	0.0735	0.0147	0.5196
Number of holding Long/Short portfolio	204	188	111	75	49	33	9	204
Hit rate of holding long/short portfolio	0.7304	0.6436	0.5946	0.5200	0.3673	0.1515	0.1111	0.7892
Number of holding TBILL	0	16	93	129	155	171	195	Nan
Hit rate of holding TBILL	NaN	0.6250	0.6129	0.7054	0.7613	0.8129	0.8359	Nan
Hit rate of strategy	0.7304	0.6422	0.6029	0.6373	0.6667	0.7059	0.8039	0.7892
Monthly turnover	1.0000	0.9216	0.5441	0.3676	0.2402	0.1618	0.0441	1.0000
Net average monthly excess return (100bp)	0.2985	0.2690	0.1638	0.0689	-0.0721	-0.1519	-0.0284	-0.0071
Net average monthly excess return (200bp)	0.2152	0.1922	0.1185	0.0383	-0.0921	-0.1654	-0.0321	-0.0905
Breakeven transaction cost	0.1910	0.1877	0.1923	0.1355	-0.1084	-0.4277	-0.2796	0.0038
			Panel B Value	-weighted				
			Acti	ve trading strategy				Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.1969	0.1479	0.2437	0.1789	0.1008	0.0225	0.0000	0.0380
Standard deviation	2.6099	2.5148	2.0408	1.2826	0.9321	0.5477	0.0000	1.0905
Sharpe ratio	0.0754	0.0588	0.1194	0.1395	0.1082	0.0411	-0.0700	0.0349
%<0	0.3431	0.2990	0.1275	0.0490	0.0343	0.0098	0.0000	0.3284
% <monthly tbill<="" td=""><td>0.4265</td><td>0.3578</td><td>0.1765</td><td>0.0637</td><td>0.0441</td><td>0.0049</td><td>0.0000</td><td>0.5196</td></monthly>	0.4265	0.3578	0.1765	0.0637	0.0441	0.0049	0.0000	0.5196
Number of holding Long/Short portfolio	199	170	101	41	25	8	0	204
Hit rate of holding long/short portfolio	0.6482	0.5647	0.5743	0.4878	0.3600	0.2500	NaN	0.6716
Number of holding TBILL	5	34	103	163	179	196	204	Nan
Hit rate of holding TBILL	0.2000	0.4706	0.6505	0.6810	0.7151	0.7602	0.8235	Nan
Hit rate of strategy	0.6373	0.5490	0.6127	0.6422	0.6716	0.7402	0.8235	0.6716
Monthly turnover	0.9755	0.8333	0.4951	0.2010	0.1225	0.0392	0.0000	1.0000
Net average monthly excess return (100bp)	0.1156	0.0784	0.2024	0.1621	0.0906	0.0192	0.0000	-0.0453
Net average monthly excess return (200bp)	0.0343	0.0089	0.1612	0.1454	0.0803	0.0159	0.0000	-0.1287
Breakeven transaction cost	0.1009	0.0887	0.2461	0.4451	0.4113	0.2867	Nan	0.0190

Table 7.15 The performance of active trading strategy on the combination of momentum, size, and value effect with 12 months' holding length

Note: The table reports the performance of active trading strategy based on the combination of momentum, size, value effect with 12 months' holding length from 1990 to 2008. The active strategy imposes filter threshold on the predicted long/short return from 0 to 3.0% in increments of 0.5%. When predicted monthly long/short returns of three style effects from out-of-sample forecasting model are all lower than one filter threshold, the position is allocated on TBILL. Otherwise, the position is allocated on top/or bottom portfolio of one style effect with predicted highest return spread. The passive strategy allocates average position on long/short portfolio's of three style effects all the time. The row labeled '%<0' and '%<Monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Long/Short portfolio' reports the percentage of strategy' reports the percentage of successful holding long/short portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position. 'Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

	Panel A Equal-weighted					Panel B Value-weighted			
	Overall	Winner/Loser	Small/Large	Low/High		Overall	Winner/Loser	Small/Large	Low/High
Number of Long/Short position	204	111	58	35	Number of Long/Short position	199	87	53	59
% Long/Short position	1.0000	0.5441	0.2843	0.1716	% Long/Short position	1.0000	0.4372	0.2663	0.2965
Average monthly return $R_{\text{lms}}$	0.8642	0.8330	0.6091	1.3858	Average monthly return $R_{lms}$	0.6885	1.1324	0.4537	0.2449
Standard deviation of $R_{lms}$	2.2914	2.4076	2.3643	1.6822	Standard deviation of $R_{lms}$	2.6380	3.3257	2.0404	1.7546
Sharp ratio	0.3771	0.3460	0.2576	0.8238	Sharp ratio	0.2610	0.3405	0.2224	0.1396
Number of R <sub>lms</sub> <0	55	27	23	5	Number of R <sub>lms</sub> <0	70	24	17	29
% R <sub>lms</sub> <0	0.2696	0.2432	0.3966	0.1429	% R <sub>lms</sub> <0	0.3518	0.2759	0.3208	0.4915

Table 7.16 The long/short position performance in active trading strategy on the combination of momentum, size, and value effect with 12 months' holding length (Filter threshold=0)

Note: The table reports the long/short position performance in active trading strategy on the combination of momentum, size, and value effect with 12 months' holding length when filter threshold is set to 0. The active strategy imposes filter threshold on the predicted long/short return. When predicted monthly long/short returns of three style effects from out-of-sample forecasting model are all lower than filter threshold, the position is allocated on TBILL. Otherwise, the position is allocated on top/or bottom portfolio of one style effect with predicted highest return spread. The row labeled '% Long/Short position' indicates the percentage of long/short position on one style effect with predicted highest return spread from out-of-sample forecasting model. The row labeled '%  $R_{lms} < 0$ ' indicates the percentage of trading return less than 0

			Panel A Ec	ual-weighted				
			Acti	ve trading strategy				Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.4445	0.3936	0.1372	0.0011	-0.0982	-0.1576	-0.0408	0.2462
Standard deviation	2.1369	2.0696	1.6915	1.5479	1.4706	1.1936	0.6293	1.1627
Sharpe ratio	0.2080	0.1902	0.0811	0.0007	-0.0668	-0.1320	-0.0648	0.2118
%<0	0.2206	0.1863	0.0833	0.0588	0.0588	0.0490	0.0098	0.1961
% <monthly tbill<="" td=""><td>0.3088</td><td>0.2353</td><td>0.0931</td><td>0.0588</td><td>0.0588</td><td>0.0490</td><td>0.0098</td><td>0.2990</td></monthly>	0.3088	0.2353	0.0931	0.0588	0.0588	0.0490	0.0098	0.2990
Number of holding Long/Short portfolio	201	173	74	39	27	16	3	204
Hit rate of holding Long/Short portfolio	0.7761	0.7168	0.6486	0.5897	0.4815	0.2500	0.3333	0.8049
Number of holding TBILL	3	31	130	165	177	188	201	Nan
Hit rate of holding TBILL	0.0000	0.5161	0.5000	0.6303	0.7345	0.8351	0.8657	Nan
Hit rate of strategy	0.7647	0.6863	0.5539	0.6225	0.7010	0.7892	0.8578	0.8049
Monthly turnover	0.9853	0.8480	0.3627	0.1912	0.1324	0.0784	0.0147	1.0000
Net average monthly excess return (100bp)	0.3623	0.3229	0.1069	-0.0149	-0.1093	-0.1642	-0.0420	0.1628
Net average monthly excess return (200bp)	0.2802	0.2522	0.0767	-0.0308	-0.1203	-0.1707	-0.0433	0.0795
Breakeven transaction cost	0.2256	0.2321	0.1891	0.0029	-0.3710	-1.0046	-1.3857	0.1231
			Panel B Va	lue-weighted				
			Acti	ve trading strategy				Passive trading strategy
Filter threshold	>0	>0.5	>1.0	>1.5	>2.0	>2.5	>3.0	Nan
Average monthly excess return	0.4846	0.3456	0.1642	0.1579	0.1052	0.0190	0.0000	0.2024
Standard deviation	2.6925	2.5355	1.9772	1.1725	0.8267	0.4793	0.0000	1.4018
Sharpe ratio	0.1800	0.1363	0.0830	0.1347	0.1273	0.0397	-0.0700	0.1434
%<0	0.3284	0.2892	0.1275	0.0196	0.0098	0.0049	0.0000	0.3137
% <monthly tbill<="" td=""><td>0.3922</td><td>0.3284</td><td>0.1569</td><td>0.0196</td><td>0.0098</td><td>0.0049</td><td>0.0000</td><td>0.4706</td></monthly>	0.3922	0.3284	0.1569	0.0196	0.0098	0.0049	0.0000	0.4706
Number of holding Long/Short portfolio	199	164	80	19	10	4	0	204
Hit rate of holding Long/Short portfolio	0.6633	0.5915	0.5375	0.7368	0.7000	0.5000	Nan	0.6863
Number of holding TBILL	5	40	124	185	194	200	204	Nan
Hit rate of holding TBILL	0.2000	0.3750	0.5242	0.6378	0.6856	0.7300	0.8039	Nan
Hit rate of strategy	0.6520	0.5490	0.5294	0.6471	0.6863	0.7255	0.8039	0.6863
Monthly turnover	0.9755	0.8039	0.3922	0.0931	0.0490	0.0196	0.0000	1.0000
Net average monthly excess return (100bp)	0.4032	0.2785	0.1314	0.1501	0.1011	0.0173	0.0000	0.1190
Net average monthly excess return (200bp)	0.3219	0.2115	0.0987	0.1424	0.0970	0.0157	0.0000	0.0356
Breakeven transaction cost	0.2484	0.2149	0.2093	0.8478	1.0735	0.4852	#DIV/0!	0.1012

Table 7.17 The performance of active trading strategy on the combination of momentum and value effect with 12 months' holding length

Note: The table reports the performance of active trading strategy on the combination of momentum and value effect with 12 months' holding length from 1990 to 2008. The active strategy imposes filter threshold on the predicted long/short return from 0 to 3.0% in increments of 0.5%. When predicted monthly long/short returns of two style effects from out-of-sample forecasting model are both lower than one filter threshold, the position is allocated on TBILL. Otherwise, the position is allocated on top/or bottom portfolio of one style effect with predicted highest return spread. The passive strategy always allocates average position on long/short portfolios of two style effects all the time. The row labeled '%<0' and '%<Monthly TBILL' indicate the percentage of trading return less than zero and monthly TBILL in observations. The row labeled 'Hit rate of holding Long/Short portfolio' reports the percentage of actual return unsatisfied with filter threshold in Long/Short portfolio and TBILL. The row labeled 'Hit rate of strategy' reports the percentage of successful holding long/short portfolio and TBILL in observations. 'Monthly turnover' is the percentage of holding long/short position. Net average monthly excess return (100bp, 200bp)' is trading return after reducing 100 or 200 basis point transaction cost every turnover. 'Breakeven trading cost' is defined as the fixed trading cost that makes monthly excess return of one strategy to zero.

	Pa	nel A Equal-weighte	ed		Panel B Value-weighted		
	Overall	Winner/Loser	Low/High		Overall	Winner/Loser	Low/High
Number of Long/Short position	201	139	62	Number of Long/Short position	199	97	102
% Long/Short position	1.0000	0.6915	0.3085	% Long/Short position	1.0000	0.4874	0.5126
Average monthly return $R_{\text{lms}}$	0.9320	0.9050	0.9926	Average monthly return $R_{lms}$	0.9834	1.1414	0.8332
Standard deviation of $R_{lms}$	2.1883	2.3376	1.8256	Standard deviation of $R_{lms}$	2.7279	3.3077	2.0354
Sharp ratio	0.4259	0.3871	0.5437	Sharp ratio	0.3605	0.3451	0.4093
Number of R <sub>lms</sub> <0	45	35	10	Number of R <sub>lms</sub> <0	68	28	39
$% R_{lms} < 0$	0.2239	0.2518	0.1613	% R <sub>lms</sub> <0	0.3417	0.2887	0.3824

Table 7.18 The long/short position performance in active trading strategy on the combination of momentum and value effect with 12 months' holding length (Filter threshold=0)

Note: The table reports the long/short position performance in active trading strategy on the combination of momentum and value effect with 12 months' holding length from 1990 to 2008 when filter threshold is set to 0. The active strategy imposes filter threshold on the predicted long/short return. When predicted monthly long/short returns of two style effects from out-of-sample forecasting model are both lower than filter threshold, the position is allocated on TBILL. Otherwise, the position is allocated on top/or bottom portfolio of one style effect with predicted highest return spread. The row labeled '% Long/Short position' indicates the percentage of long/short position on one style effect with predicted highest return spread from out-of-sample forecasting model. The row labeled '%  $R_{lms} < 0'$  indicates the percentage of trading return less than 0

# **Chapter 8 Conclusions to the Research**

The thesis investigates the time-variation in style effects in the UK Stock Market. It focuses on four research questions which are empirically analyzed in four chapters. In Chapter 4, the thesis uses two indicators, style coefficient  $\gamma$  and long/short return  $R_{lms}$ , to examine time-varying momentum effect, size effect, and value effect with 3, 6, 12, 18, and 24 months' holding horizon. It further examines the correlation among style effects. In Chapter 5, the thesis discusses whether the time-variation in style effects can be captured by lagged macroeconomic variables, which are associated with macroeconomic conditions and market states. In Chapter 6, with three statistics PIS, POOS, and PSS, the thesis tests the performance of recursive in-sample regression and out-of-sample forecasting model, and examines whether dynamic style effects can be indeed predicted by lagged macroeconomic variables. In Chapter 7, the thesis explores active trading strategy on individual style effect and combined two/or three style effects to time style rotation. As the ending part of the thesis, this chapter presents overall conclusions in section 8.1, contributions and implications of the research in section 8.2, and limitations and future research in the section 8.3.

### 8.1 Summary of findings

The overall conclusions of the thesis are described in the form of individual style effect and combined style effects. It proceeds with the empirical findings on timevarying pattern of style effect, the impact of macroeconomic variables on the variation, the test of recursive in-sample regression model and out-of-sample forecasting model, and active trading strategy on style effects.

#### • Momentum effect

Firstly, when decile momentum portfolios are formed by previous 6/12 months' share returns and hold for 6 and 12 months, momentum effect is strong from 1956 to 2008. As holding horizon grows more than 12 months, momentum effect becomes weaker and unsignificant. Two indicators, Winner/Loser return  $R_{wml}$  and momentum coefficient  $\gamma_{mom}$ , are volatile over time, and show the strongest momentum effect in the 1990s and the weakest in the 1970s. This volatility leads to positive (negative) Winner/Loser return  $R_{wml}$  in 53 to 70 (30 to 47) percent of all observations.

Secondly, lagged macroeconomic variables explain 15 to 25 percent of the timevariation in momentum effect (12/6, 12/12) with 12 months' formation length and 6/12 months' holding length from 1980 to 2008. The decline in annual change of GDP (GDP(Y)) and the increase in unexpected annual change of TBILL (UEXTBILL(Y)) implies stronger momentum effect (12/6, 12/12).

Thirdly, two statistics 'POOS' and 'PSS' in the test of recursive out-of-sample forecasting model for momentum effect (12/6, 12/12) keep the similar performance as the statistics 'PIS' in the test of recursive in-sample regression of momentum effect on lagged macroeconomic variables from 1990 to 2008. Series 'PSS' indicates that the recursive out-of forecasting model offers same sign (positive and negative) between predicted and actual momentum coefficient  $\gamma_{mom}$  (Winner/Loser return  $R_{wml}$ )

in 60 to 82 (60 to 76) percent of all observations. High 'PSS' implies promising capability of the forecast model to time momentum effect.

Fourthly, with the signal of recursive out-of-sample forecasting model and filter threshold, active trading strategy on momentum effect (12/6) outperforms passive trading strategy which always holds Winner/Loser portfolio all the time. Active trading strategy on any equal-weighted momentum effect (12/6), which holds long/short portfolio with predicted highest/lowest return, enhances average monthly excess return on passive Winner/Loser portfolio from 0.7899% to 1.2004% during 1990 to 2008. Active trading strategy on value-weighted Winner/Loser portfolio (12/6) enhances average monthly excess return on passive Winner/Loser portfolio from 0.7899% to 1.2004% during 1990 to 2008. Active trading strategy on value-weighted Winner/Loser portfolio (12/6) enhances average monthly excess return on passive Winner/Loser portfolio from 0.8322% to 1.1901%.

### • Size effect

Firstly, there is no significant size effect in the UK Stock Market from 1979 to 2008 due to size rotation over three subperiods, which exhibits (marginally) significant small-cap effect in the 1980s, marginally significant large-cap effect in the 1990s, and unsignificant small-cap effect afterwards. This rotation between small-cap effect and large-cap effect leads to positive (negative) Small/Large return  $R_{sml}$  in 44 to 62 (38 to 56) percent of all observations.

Secondly, the explanatory power of lagged macroeconomic variables for the timevariation in size effect with five holding horizons exhibits ' $\Lambda$ ' pattern from 1980 to 2008. When decile size portfolios are hold for 12 months, denoted as size effect (12), macroeconomic variables captures 14.93 to 17.64 percent of the variation in size effect from 1980 to 2008. The increase in GDP(Y) and the decline in UEXTBILL(Y) imply stronger small-cap effect.

Thirdly, two statistics 'POOS' and 'PSS' to test recursive out-of-sample forecasting model for size effect (12) have the similar performance of the statistics 'PIS' to test recursive in-sample regression of size effect on lagged macroeconomic variables. As 'PSS' shows, predicted and actual size coefficient  $\gamma_{mv}$  (Small/Large return  $R_{sml}$ ) have same sign in 51 to 61 percent of all observations, which is below the corresponding 'PSS' for momentum effect (12/6, 12/12).

Fourthly, active trading strategy on 'Small' and 'Large' portfolios with 12 months' horizon before reducing trading cost performs better than passive trading strategy which always holds 'Small/Large' position all the time. When filter threshold is set to 0.5% to 1.5%, active trading strategy improves passive strategy from negative average monthly excess return to positive one. However, after considering trading costs, active trading strategy on Small/Large portfolio becomes unprofitable.

#### • Value effect

Firstly, value effect is significant from 1980 to 2008, and stronger as the holding horizon extends from 5 month to 24 months. The outperformance of shares with low price-to-book ratio over shares with high price-to-book ratio is persistent in the 1980s, the 1990s, and 2000 afterwards. Persistent value premium leads to positive (negative) Low/High return  $R_{lmh}$  in 60 to 82 (18 to 40) percent of all observations.

399

Secondly, as holding horizon extends from 5 months to 24 months, the percentage of the time-variation in value effects driven by lagged macroeconomic variables increases to higher level, ranging 28.44 and 40.87, except for value-weighted Low/High return  $R_{lmh}$ . The decline in annual change of CPI (CPI(Y)) and higher lagged 8-year's market return imply strong value effect (24) with 24 months' holding horizon from 1980 to 2008.

Thirdly, the performance of two statistics 'POOS' and 'PSS' in the test of recursive out-of-sample forecasting model for value effect (24) is quite different from the performance of the statistics 'PIS' in the test of recursive in-sample regression of value effect on lagged macroeconomic variables. Out-of-sample forecasting model for value effect (24) performs poorly relative to the historical mean model even though its in-sample regression model does better. As 'PSS' indicates, the recursive out-of forecasting model offers same sign (positive and negative) between predicted and actual value coefficient  $\gamma_{pb}$  and Low/High return R<sub>lmh</sub> in 45 to 59 percent of all observations, which are lower than corresponding 'PSS' for momentum effect (12/6, 12/12) and size effect (12).

Fourthly, different from momentum effect and size effect, long-term persistent performance of return spread between 'Low' and 'High' portfolios and poor performance of out-of-sample forecasting model lead to the failure of active trading strategy on value effect (24) to outperform its passive trading strategy which always allocates long/short position on Low/High portfolio all the time.

#### • Combined style effects

The empirical work finds negative correlation between momentum effect and size effect, between momentum effect and value effect, and positive correlation between size effect and value effect from 1980 to 2008. It is found that momentum effect (12/6, 12/12) and size effect (12) are significantly sensitive to GDP(Y) and UEXTBILL(Y). The coefficient signs of GDP(Y) and UEXTBILL(Y) in the multivariate regression of momentum effect (12/6, 12/12) are opposite to ones in the regression of size effect (12). This means that momentum effect and size effect are significant in the opposite macroeconomic conditions. Similarly, momentum effect (12/6, 12/12) and value effect (12) are significant in the same TBILL condition, and size effect (12) and value effect (12) are significant in the same TBILL condition. The sensitivity of style effects to same economic forces offers reasonable explanation for the correlation among three style effects.

The correlation among style effects implies the possibility of active trading strategy on combined style effects to perform better than passive strategy which allocates average long/short position on style effects all the time. The research finds that active trading strategy with predicted highest return spread within momentum effect (12/12), size effect (12), and value effect (12) yields higher return than passive strategy from 1990 to 2008. Active strategy with predicted highest return spread within momentum effect (12/12) and value effect (12) performs better because the position on Small/Large portfolio with lower return spread is replaced by the position on Winner/Loser portfolio and Low/High portfolio with higher return spread.

## 8.2 Contributions and implications of the research

Based on the empirical findings in the previous chapters, the thesis deepens understanding on not only time-varying style effects but also forecasting factors for their dynamic patterns. It makes contributions to studies on style effects, and offers implications for researchers in the academic field and practitioners in the market.

#### 8.2.1 Contributions of the research

Firstly, different from most previous studies on individual style effect, or two style effects with one certain holding horizon, such as 6 or 12 months, the thesis examines three style effects in one empirical work, and finds negative correlation between momentum effect and size effect, between momentum effect and vale effect, and positive correlation between size effect and vale effect. In addition, the thesis investigates style effects with 3, 6, 12, 18, and 24 months' holding horizons, and sheds light on the pattern of style effects with short-/medium-/long-term horizon, and the variation in significant explanatory factors for style effects through holding horizons.

Secondly, besides traditional indicator long/short return  $R_{lms}$  based on top/bottom decile portfolios in much literature, the thesis uses style coefficient  $\gamma$  to examine time-varying style effects. Estimated by cross-sectional regression of decile portfolios' returns in the holding period on their average values of firm characteristics in the formation period, style coefficient  $\gamma$  contains the information of

402

all shares in decile portfolios. So, it could capture the relation between share return and firm characteristics more than long/short return  $R_{lms}$ . This pattern is salient in examining value effect, in which other decile value portfolios, following 'Low' portfolio and 'High' portfolio, are likely to yield highest/lowest return. Meanwhile, the test on the impact of lagged macroeconomic variables on the time-variation in style coefficient  $\gamma$  helps us to diagnose the efficiency of active trading strategy which allocates long position and short position on any decile portfolio.

Thirdly, all previous studies only investigate the time-variation in style effects and explain their dynamic property in the context of macroeconomic conditions and market states in the overall sample period. The thesis extends these empirical analysis in the overall period to ones in subperiods, which shows the stability/or volatility of significant macroeconomic variables to explain the time-variation in style effects. For example, UEXTBILL(Y) and EMR(-1), GDP(Y), respectively, combining with DY(Q), significantly explain the variation in equal-weighted momentum effect (12/6) in the 1980s and the 1990s.

Fourthly, although some previous studies find significant explanatory power of macroeconomic factors for the time-variation in style effects, they do not undertake out-of-sample test to observe the reliability/or pitfall of in-sample predictability on style effects. The thesis uses statistics PIS, POOS, and PSS to test the performance of recursive in-sample regression and out-of-sample forecasting model on style effects relative to historical mean model, and investigates whether macroeconomic factors

and market factors indeed offer predictive power for the time-variation in style effects.

#### 8.2.2 Implications of the research

Empirical findings offer implications for researchers in academic field. The thesis finds the variation in significant macroeconomic variables to explain style effects through holding horizons and over time. This implies that to use one macroeconomic variable in time-series forecasting regression might lead to biased and inconsistent estimators in different holding horizons and different sample periods. Researchers, who undertake studies on series return predictability, should examine empirical work over different holding horizons and subperiods. In addition, the thesis finds that out-of-sample forecasting model on value effect (24) underperforms the sample mean model although its in-sample regression model performs well. This means that researchers should conduct out-of-sample model test to examine the reliability/or pitfall of in-sample predictability.

Empirical findings have implications for style-oriented investors in the market. The thesis finds the existence of biased and inconsistent significant macroeconomic variables for the time-variation in style effects with short-/medium-/long-horizons and over subperiods. This implies that style-oriented investors should be cautious to use forecasting variables to capture style rotation according to their holding horizons and macroeconomic conditions. In addition, the thesis develops active trading

strategies on long/short portfolio for individual style effect and combined style effects. Active trading strategies, which combines top-down perspective using economic and market series with bottom-up view using specific share information, benefit style-oriented investors to time the switch among style effects, and enhance the performance of traditional absolute return alternative strategies with long/short position all the time. Active trading strategy on either any equal-weighted decile momentum portfolio (12/6) or value-weighted Winner/Loser portfolio (12/6) might be one good choice for style-oriented investors in the UK Stock Market.

## 8.3 Limitations and future research

As far as the author is aware, the investigation on time-varying style effects is at the beginning stage in academic field although some studies have focused on the dynamic pattern since the 2000. As overall research, the thesis is subject to some limitations in the empirical work on the time-variation in style effects in the UK Stock Market. It is these limitations, as well as empirical findings in the thesis, that drives me to conduct further relevant research in the future.

#### 8.3.1 Limitations of the research

In Chapter 3, the research constructs decile size portfolios and finds that 'Small' portfolio includes average 41.3 percent of all shares but accounts for only 1 percent of overall market value from 1980 to 2008. Returns of these small-cap shares might not reflect actual trading prices in the market due to the existence of their wider bid-ask spread and low trading volume. Size decile portfolio formation with the inclusion of these small-cap shares might lead to biased result on size effect.

In Chapter 4, the research establishes size portfolios and value portfolios at the beginning of 1979 and 1980, respectively, due to relevant data available from LSPD and Datastream, and estimates their style coefficient  $\gamma$  and long/short return R<sub>lms</sub>. In Chapter 5, the research examines the impact of lagged macroeconomic variables on the time-variation in style effects from 1980 to 2008. In Chapter 6, after excluding

first 10-year's initial in-sample period, the test on recursive in-sample regression model and out-of-sample forecasting model is conducted at the beginning of 1990 until 2008. The empirical result might be biased due to shorter sample period from 1980 to 2008, in which there is less style rotation and cyclical pattern in macroeconomic conditions and market states. This might lead to the similar limitation in the exploration of active trading strategy on style effects from 1990 to 2008 in Chapter 7.

In Chapter 5, the research finds the variation in significant macroeconomic variables for time-varying style effects over three subperiods. This means that significant/unsignificant macroeconomic variables screened in univariate regression with overall sample period from 1980 to 2008 might be biased and inconsistent through time. It is possible that one unsignificant variable in overall period might be significant in the initial 10-year's period but unsignificant in later periods, and that it might be unsigificant in the initial 10-year's period but significant afterwards. This might lead to the inclusion of biased significant variables and the exclusion of significant variables in the recursive regression model because investors in the market do not know unbiased forecasting factors at the end of the initial period.

In Chapter 5, the research finds some macroeconomic variables, such as M0(Y) and CPI(Y), have opposite explanatory power for style effects in different subperiods although they are significant in the period from 1980 to 2008. Therefore, the research with volatile forecasting factors might have biased and inconsistent result as the sample period expends.

#### 8.3.2 Future research

Based on limitations discussed in previous subsection and empirical findings in the research, there are several intriguing directions and possible avenues in which this study could be extended in future as followings:

Firstly, the empirical work in Chapter 5 shows significant impact of lagged long-term market return on the time-variation in value effect. It is worthwhile to explore the correlation between style effects and their lagged performance in measure of long/short return  $R_{lms}$ . This offers another dimension to document the time-variation in style effects, and exhibit the pattern of their persistence and reversal.

Secondly, the empirical analysis in Chapter 5 finds different macroeconomic variables to explain time-varying style effects over time. It implies that recursive forecasting model with macroeconomic variables might offers biased and inconsistent signal on style effects over time. In addition, the research finds that most adjusted  $R^2$  of the regression of style effects on lagged macroeconomic variables in 10-years' subperiods are higher than ones in the overall sample period. This means that the same macroeconomic variables have explanatory power in one subperiod stronger than in the overall period. So, in order to lower the biased effect due to time-variant forecasting macroeconomic variables and increase the forecasting power of macroeconomic variables, it is instructive to examine out-of-sample forecasting model and explore active trading strategy based on the regression with fixed in-sample window (for example 10-year's length) instead for the regression with

increasing in-sample window by one month each time, or lagged-weighted regression instead for equal-weighted regression.

Thirdly, the thesis explores active trading strategy on momentum effect (12/6, 12/12), size effect (12), and value effect (24), which has stronger style effect and can be much more explained by macroeconomic variables relative to style effects with other holding horizons. Because style-oriented investors in the market have different investing horizons, it is instructive to investigate active trading strategy on short-/medium-/long-term horizon. Meanwhile, economic conditions and market states are much volatile at some times. This means the variation in significant style effects with different horizons through time. To optimize return performance, it is necessary to explore active trading strategy on style effects within short-/medium-/long-term investing horizons.

Fourthly, it is found in Chapter 4 that equal-(value-) weighted  $2^{nd}$  size portfolio with the second smallest market share yields the lowest or second lowest average monthly return among decile size portfolios, and that equal-(value-) weighted  $8^{th}$  portfolio earns the highest or second highest return. In addition, because small-cap shares (the  $1^{st}$  size portfolio) always have wider bid-ask spread as well as low trading volume, their share returns might not actually reflect trading prices in the market. It is instructive to explore active trading strategy which allocates long and short positions on the  $8^{th}$  size portfolio and the  $2^{nd}$  size portfolio.

# **References:**

Abhyankar, A., Ho, Keng-Yu, and Zhao, H. (2008), 'Value versus Growth: Stochastic Dominance Criteria', *Quantitative Finance*, Vol.8, No.7, pp.693-704.

Ahmed, P., Lockwood, L. J., and Nanda, S. (2002), 'Multistyle Rotation Strategies', *The Journal of Portfolio Management*, Vol. 28, No.3, pp.17-29.

Antoniou, A., Galariotis, E. C., and Spyrou, S. I. (2003), 'Are Contrarian Investment Strategies Profitable in the London Stock Exchange? Where do These Profits Come from?' *Working paper, Durham University*.

Antonious, A., Lam, H. Y. T., and Paudyal, K. (2007), 'Profitability of Momentum Strategies in International Markets: The Role of Business Cycle Variables and Behavioural Biases', *Journal of Banking & Finance*, Vol.31, No.3, pp.955-972.

Asness, C. S. (1997), 'The Interaction of Value and Momentum Strategies', *Financial Analysts Journal*, Vol. 53, No.2, pp. 29-36.

Asness, C. S. (2011), 'Momentum in Japan: The Exception that Proves the Rule', *The Journal of Portfolio Management*, Vol.37, No.4, pp.67-75.

Asness, C. S., Friedman, J. A., Krail, R. J., and Liew, J. M. (2000), 'Style Timing: Value versus Growth', *The Journal of Portfolio Management*, Vol. 26, No. 3, pp.50-60.

Asness, C. S., Moskowitz, T. J., and Pedersen, L. H. (2009), 'Value and Momentum Everywhere', *Working paper, AQR Capital Management, LLC*.

Avramov, D., Chordia, T. (2006), 'Predicting Stock Returns', *Journal of Financial Economics*, Vol.82, No.2, pp.387-415.

Avramov, D., Chordia, T., Jostova, G., Philipov, A. (2007), 'Momentum and Credit Rating', *The Journal of Finance*, Vol.62, No.5, pp. 2503-2520.

Babameto, E., and Harris, R. D. F. (2008), 'Exploiting Predictability in the Returns to Value and Momentum Investment Strategies: A Portfolio Approach', *working paper, University of Exeter*.

Bansal, R., Dittmar, R. F., and Lundblad, C. T. (2005), 'Consumption, Dividends, and the Cross Section of Equity Returns', *The Journal of Finance*, Vol.60, No.4, pp.1639-1672.

Banz, R. (1981), 'The Relationship between Return and Market Values of Common Stock', *Journal of Financial Economics*, Vol.9, No.2, pp.3-18.

Barberis, N., Shleifer, A., and Vishny, R. W., (1998), 'A Model of Investor

Sentiment', Journal of Financial Economics, Vol. 49, No.3, pp.307-343.

Barberis, N., and Shleifer, A. (2003), 'Style Investing', *Journal of Financial Economics*, Vol.68, No.2, pp.161–199.

Barberis, N., Shleifer, A., and Wurgler, J. (2005), 'Comovement', *Journal of Financial Economics*, Vol.75, No.2, pp.283-317.

Basu, S. (1977), 'Investment Performance of Common Stocks in Relation to Their Price- Earnings Ratios: A Test of the Efficient Market Hypothesis', *The Journal of Finance*, Vol. 32, No.3, pp.663-682.

Basu, S. (1983), 'The Relationship between Earnings' Yield, Market Value and Return for NYSE Common Stocks: Further Evidence', *Journal of Financial Economics*, Vol. 12, No.1, pp.129-156.

Beenstock, M., and Chan, K. F. (1988), 'Economic Forces in the London Stock Market', *Oxford Bulletin of Economics and Statistics*, Vol.50, No.1, pp.27-39.

Berger, A. L., Israel, R., and Moskowitz, T. J. (2009), 'The Case for Momentum Investing', *Working paper, AQR Capital Management, LLC*.

Bird, R., and Whitaker, J. (2004), 'The Performance of Value and Momentum Investment Portfolios: Recent Experience in the Major European Markets', *Journal of Asset Management*, Vol.5, no. 3, pp.157-175.

Black, F. (1972), 'Capital Market Equilibrium with Restricted Borrowing', *The Journal of Business*, Vol.45, No.3, pp.444-455.

Blume, L., David, E., and Maureen O. (1994), 'Market Statistics and Technical Analysis: The Role of Volume', *The Journal of Finance*, Vol. 49, No.1, pp.153–181.

Bogle, J., Bogle on Mutual Funds: New Perspectives for the Intelligent Investor, Burr Ridge, IL: Irwin Professional Publishing, 1994.

Bordo, M. D., Wheelock, D. C. (2006), 'When Do Stock Market Booms Occur? The Macroeconomic and Policy Environments of 20<sup>th</sup> Century Booms', *working paper, Federal Reserve Bank of St. Louis*.

Bossaerts, P., and Hillion, P. (1999), 'Implementing Statistical Criteria to Select Return Forecasting Models: What Do We Learn?' *The Review of Financial Studies*, Vol.12, No.2, pp.405-428.

Brandt, M. W., Clara, P. S., and Valkanov, R. (2009), 'Parametric Portfolio Policies: Exploiting Characteristics in the Cross-Section of Equity Returns', The Review of Financial Studies, Vol.22, pp.3411-2447.

Brouwer, I., Put, J. V., and Veld, C. (1996) 'Contrarian Investment Strategies in a European Context', *working paper, SSRN*.

Brunner, K. (1961), 'Some Major Problems in Monetary Theory', *The American Economic Review*', Vol. 51, No.2, pp. 47-56.

Campbell, J. Y. (1990), 'Measuring the Persistence of Expected Returns', *American Economic Review*, Vol.80, No.2, pp.43-47.

Campbell, J. Y., and Shiller, R. J. (1988), 'Stock Prices, Earnings, and Expected Dividends', *The Journal of Finance*, Vol.43, No.3, pp661-676.

Campbell, J. Y., and Hamao, Y. (1992), 'Predictable Stock Returns in the United States and Japan: A Study of Long-Term Capital Market Integration', *The Journal of Finance*, Vol. 47, No.1, pp. 43-69.

Campbell, J. Y., Grossman, S. J., and Wang, J. (1993), 'Trading Volume and Serial Correlation in Stock Returns', *Quarterly Journal of Economics*, Vol.108, No.4, pp. 905–939.

Campbell, J. Y., and Cochrane, J. H. (1999), 'By Force of Habit: A Consumption-Based Explanation of Aggregate Stock Market Behaviour', *The Journal of Political Economy*, Vol.107, No.2, pp.205-251.

Campbell, J. Y., and Vuolteenaho, T. (2004), 'Bad Beta, Good Beta', *American Economic Review*, Vol.94, pp.1249-1275.

Campbell, J. Y., and Thompson, S. B. (2005), 'Predicting Excess Stock Returns Out of Sample: Can Anything Beat the Historical Average?' *The Review of Financial Studies*, Vol.21, No.4, pp.1509-1531.

Campbell, J. Y., Polk, C., and Vuolteenaho, T. (2010), 'Growth or Glamour? Fundamentals and Systematic Risk in Stock Returns', *The Review of Financial Studies*, Vol.23, No.1, pp.305-344.

Campbell, S. D., and Diebold, F. X. (2009), 'Stock Returns and Expected Business Conditions: Half a Century of Direct Evidence', *Journal of Business and Economic Statistics*, Vol.27, No.2, pp.266-278.

Carhart, M. M. (1997), 'On Persistence of Mutual Fund Performance', *The Journal of Finance*, Vol.52, No.1, pp.57-82.

Chan, L. K. C., Hamao, Y., and Lakonishok, J. (1991), 'Fundamentals and Stock Returns in Japan', *The Journal of Finance*, Vol.46, No.5, pp.1739-1764.

Chan, L. K. C., Jegadeesh, N., and Lakonishok, J. (1996), 'Momentum Strategies', *The Journal of Finance*, Vol. 51, No.5, pp.1681-1713.
Chan, L. K. C., Karceski, J., and Lakonishok, J. (2000), 'New Paradigm or Same Old Hype in Equity Investing?' *Financial Analysts Journal*, Vol.56, No.4, pp.23-36.

Chen, H. L., and De Bondt, W. (2004), 'Style Momentum within the S&P-500 Index', *Journal of Empirical Finance*, Vol.11, No.4, pp.483–507.

Chen, N. F. (1989), Financial Investment Opportunities and the Real Economy, working paper, Center for Research in Security Prices. University of Chicago.

Chen, N. F. (1991) 'Financial Investment Opportunities and the Macroeconomy', *The Journal of Finance*, Vol. 46, No. 2, pp. 529-554.

Chen, N. F., Roll, R., and Ross, S. A. (1986), 'Economic Forces and the Stock Market', *The Journal of Business*, Vol. 59, No. 3, pp.383-403.

Chordia, T., and Shivakumar, L. (2002), 'Momentum, Business Cycle, and Timevarying Expected Returns', *The Journal of Finance*, Vol.57, No.2, pp. 985-1019.

Christopherson, J. A., Ferson, W. E., and Glassman, D. A. (1998), 'Conditioning Manager Alphas on Economic Information: Another Look at the Persistence of Performance', *The Review of Financial Studies*, Vol. 11, No.1, pp.111-142.

Chui, A. C. W., Titman, S., and Wei, K. C. J. (2010), 'Individualism and Momentum around the World', *The Journal of Finance, Vol.* 65, No.1, 361-392.

Clare, A. D., and Thomas, S. H. (1994), 'Macroeconomic Factors, the APT and the UK Stock Market', *Journal of Business Finance & Accounting*, Vol.21, No.3, pp.309-330.

Cochrane, J. H. (2001), Asset Pricing, Princeton and Oxford: Princeton University Press.

Cochrane, J. H. (2008), 'The Dog That Did Not Bark: A Defense of Return Predictability', *The Review of Financial Studies*, Vol.21, No.4, pp.1533-1575.

Cochrane, J. H. (2011), 'Presidential Address: Discount Rates', *The Journal of Finance*, Vol. 66, No.4, pp.1047–1108.

Coggin, T. D. (1998), 'Long-Term Memory in Equity Style Indexes', *The Journal of Portfolio Management*, Vol.24, No.2, pp. 37-46.

Cohen, R. B., Polk, C., and Vuolteenaho, T. (2003), 'The Value Spread', *The Journal of Finance*, Vol.58, No.2, pp.609-642.

Cooper, M. J. (1999), 'Filter Rules Based on Price and Volume in Individual Security Overreaction', *The Review of Financial Studies*, Vol.12, No.4, pp.901-935.

Cooper, M. J., Gulen, H., and Vassalou, M. (2001), 'Investing in Size and Book-to-

market Portfolios Using Information About the Macroeconomy: Some New Trading Rules', *working paper, SSRN*.

Cooper, M. J., Gutierrez, JR., and Hameed, A. (2004), 'Market States and Momentum', *The Journal of Finance*, Vol.59, No.3, pp.1345-1365.

Cooper, M. J., Gulen, H., and Rau, P. R. (2005), 'Changing Names with Style: Mutual Fund Name Changes and Their Effects on Fund Flows', *The Journal of Finance*, Vol.60, No.6, pp. 2825-2858.

Cooper, R. V. L, (1974), 'Efficient Capital Markets and the Quantity Theory of Money', *The Journal of Finance*, Vol.29, No. 3, pp.887-908.

Daniel, K., Hirshleifer, D., Subrahmanyam, A. (1998), 'Investor Psychology and Security Market Under- and Over-reactions', *The Journal of Finance*, Vol.53, No.6, pp.1839-1886.

Datar, V. T., Naik, Y. N., and Radcliffe, R. (1998), 'Liquidity and Stock Returns: An Alternative Test', *Journal of Financial Markets, Vol.*1, No.2, pp.203–219.

Davis, J. L., Fama, E. F., and French, K. R. (2000), 'Characteristics, Covariances, and Average Returns: 1929 to 1997', *The Journal of Finance*, Vol. 55, No.1, pp.389-406.

De Bondt, W. F. M., and Thaler, R. (1985), 'Does the Stock Market Overreact?' *The Journal of Finance*, Vol. 40, No.3, pp.793-805.

De Bondt, W. F. M., and Thaler, R. (1987), 'Further Evidence on Investor Overreaction and Stock Market Seasonality', *The Journal of Finance*, Vol. 42, No.3, pp.557-581.

Dimson, E., and Marsh, P. (2001), 'U.K. Financial Market Returns: 1955–2000', *Journal of Business*, Vol.74, No.1, pp.1-31.

Dimson, E., and Nagel, S. (2001), 'Style Investment', working paper London Business School.

Dimson, E., Nagel, S., and Quigley, G. (2003), 'Capturing the Value Premium in the United Kingdom', *Financial Analysts Journal*, Vol.59, No.6, pp.35-45.

Ehrmann, M. (2000), 'Firm Size and Monetary Policy Transmission: Evidence from German Business Survey Data', *working paper, European Central Bank*.

Fama, E. F. (1970), 'Efficient Capital Markets: A Review of Theory and Empirical Work', *The Journal of Finance*, Vol.25, No.2, pp.383-417.

Fama, E. F. (1981), 'Stock Returns, Real Activity, Inflation, and Money', *The American Economic Review*, Vol.71 No.4, pp.545-565.

Fama, E. F. (1990), 'Term-Structure Forecasts of Interest Rates, Inflation, and Real Returns', Journal of Monetary Economics, Vol.25, No.1, pp.59-76.

Fama, E. F., and Schwert, G. W. (1977), 'Asset Returns and Inflation', *Journal of Financial Economics*, Vol.5, No.2, pp.113-269.

Fama, E. F., and Gibbons, M. R. (1984), 'A Comparison of Inflation Forecasts', *Journal of Monetary Economics*, Vol.13, pp.327-348.

Fama, E. F., and French, K. R. (1988a), 'Permanent and Temporary Components of Stock Prices', *Journal of Political Economy*, Vol.96, No.2, pp.246-273.

Fama, E. F., and French, K. R. (1988b), 'Dividend Yields and Expected Stock Returns', *Journal of Financial Economics*, Vol.22, No.1, pp.3-25.

Fama, E. F., and French, K. R. (1989), 'Business Conditions and Expected Returns on Stocks and Bonds', *The Journal of Financial Economics*, Vol.25, pp23-49.

Fama, E. F., and French, K. R. (1992), 'The Cross-Section of Expected Stock Returns', *The Journal of Finance*, Vol. 47, No.2, pp.427-465.

Fama, E. F., and French, K. R. (1993), 'Common Risk Factors in the Returns on Stocks and Bonds', *The Journal of Financial Economics*, Vol.33, pp.3-56.

Fama, E. F., and French, K. R. (1996), 'Multifactor Explanations of Asset Pricing Anomalies', *The Journal of Finance*, Vol.51, No.1, pp.55-84.

Fama, E. F., and French, K. R. (1998), 'Value versus Growth: the International Evidence', *The Journal of Finance*, Vol. 53, No.6, pp.1975-1999.

Fama, E. F., and French, K. R. (2011), 'Size, Value, and Momentum in International Stock Returns', *Journal of Financial Economics*, Vol.105, No.3, pp.457-472.

Fama, E. F., and MacBeth J. D. (1973), 'Risk, Return, and Equilibrium: Empirical Tests', *The Journal of Political Economy*, Vol.81, No.3, pp. 607-36.

Ferson, W. E., and Harvey, C. R. (1991), 'The Variation of Economic Risk Premiums', *The Journal of Political Economy*, Vol.99, No.2, pp.385-415.

Ferson, W. E., and Schadt, R. W. (1996), 'Measuring Fund Strategy and Performance in Changing Economic Conditions', *The Journal of Finance*, Vol.51, No.2, pp.425-462.

Ferson, W. E., Sarkissian, S., and Simin, T. (2003), 'Spurious Regressions in Financial Economics', *The Journal of Finance*, Vol.1, No.3, pp.1-10.

Ferson, W. E., Sarkissian, S., and Simin, T. (2006), 'Asset Pricing Models with

Conditional Betas and Alphas: The Effects of Data Snooping and Spurious Regression', *working paper, National Bureau of Economic Research*.

Figelman, I. (2007a), 'Stock Return Momentum and Reversal', *The Journal of Portfolio Management*, Vol.34, No.1, pp.51-67.

Figelman, I. (2007b), 'Interaction of Stock Return Momentum with Earnings Measures', *Financial Analysts Journal*, Vol.63, No.3, pp.71-79.

Fletcher, J. (2001), 'An Examination of Predictable Risk and Return in UK Stock Returns', *Journal of Economics and Business*, Vol.53, No.6, pp.527-546.

Fletcher, J. (2002), 'Examination of Conditional Asset Pricing in UK Stock Returns', *The Financial Review*, Vol.37, No.3, pp.447–468.

Fletcher, J. (2007), 'Can Asset Pricing Models Price Idiosyncratic Risk in U.K. Stock Returns?' *The Financial Review*, Vol.42, No.4, pp.507-535.

Fletcher, J. (2011), 'Do Optimal Diversification Strategies Outperform the 1/N Strategy in U.K. Stock Returns?' *International Review of Financial Analysis*, Vol.20, No.5, pp.376-385.

Fletcher, J., and Hillier, J. (2002), 'An Examination of the Economic Significance of Stock Return Predictability in UK Stock Returns', *International Review of Economics and Finance*, Vol.11, No.4, pp.373-392.

Fraser, P. (1995), 'Returns and Firm Size: A Note on the UK Experience 1970-1991', *Applied Economics Letters*, Vol. 2, No.10, pp,331-334.

Friedman, M., and Schwartz, A. J. (1963), 'Money and Business Cycles', *The Review* of Economics and Statistics, Vol.45, No.1, pp.32-64.

Gertler, M. L., and Gilchrist, S. (1994), 'Monetary Policy, Business Cycles, and the Behaviour of Small Manufacturing Firms', *The Quarterly Journal of Economics*, Vol.109, No.2, pp.309-340.

Goyal, A. (2012), 'Empirical Cross-sectional Asset Pricing: A Survey', *Financial Markets and Portfolio Management*, Vol.26, No.1, pp.3-38.

Goyal, A., and Welch, I. (2003), 'Predicting the Equity Premium with Dividend Ratios', *Management Science*, Vol.49, No.5, pp.639-654.

Granger, C. W. J., and Newbold, P. (1974), 'Spurious Regressions in Econometrics', *Journal of Econometrics*, Vol.2, No.2, pp.111-120.

Greenwood, R. (2005), 'Short- and Long-term Demand Curves for Stocks: Theory and Evidence on the Dynamics of Arbitrage', *Journal of Financial Economics*, Vol.75, No.3, pp.607-649.

Grieb, T., and Reyes, M. G. (2002), 'The Temporal Relationship between Large- and Small-capitalization Stock Returns: Evidence from the UK', *Review of Financial Economics*, Vol.11, No.2, pp.109-118.

Gunsel, N., and Cukur, S. (2007), 'The Effects of Macroeconomic Factors on the London Stock Returns: a Sectoral Approach', *International Research Journal of Finance and Economics*, Vol.10, pp.140-152.

Hamao, Y. (1988), 'An Empirical Examination of the Arbitrage Pricing Theory', *Japan and the World Economy*, Vol.1, pp.45-61.

Hansen, L. P., Heaton, J. C. and Li, N. (2008), 'Consumption Strikes Back? Measuring Long-Run Risk', *Journal of Political Economy*, Vol.116, No.2, pp.260-302.

Harrison, P., and Zhang, H. (1999), 'An Investigation of the Risk and Return Relation at Long Horizons', *Review of Economics and Statistics*, Vol.81, No.3, pp.1-10.

Heston, S. L., and Sadka R. (2008), 'Seasonality in the Cross-section of Stock Returns', *Journal of Financial Economics*, Vol. 87, No.2, pp.418-445.

Hodrick, R. J. (1992), 'Dividend Yields and Expected Stock Returns: Alternative Procedures for Inference and Measurement', *The Review of Financial Studies*, Vol.5, No. 3, pp.357-386.

Hodrick, R. J., and Zhang, X. (2001), 'Evaluating the Specification Errors of Asset Pricing Models', *Journal of Financial Economics*, Vol. 62, No.2, pp.327-376.

Hong, H., and Stein, J. C. (1999), 'A Unified Theory of Underreaction, Momentum Trading, and Overreaction in Asset Markets', *The Journal of Finance*, Vol.54, No.6, pp.2143-2184.

Hou, K. G., Karolyi, G. A., and Kho, B. C. (2011), 'What Factors Drive Global Stock Return?' *The Review of Financial Studies*, Vol 24, No.8, pp.2528-2574.

Humpe, A., and Macmillan, P. (2007), 'Can Macroeconomic Variables Explain Long Term Stock Market Movements? A Comparison of the US and Japan', *working paper*, *University of St Andrews*.

Jagannathan, R., and Wang, Z. (1996), 'The Conditional CAPM and the Crosssection of Expected Returns', *The Journal of Finance*, Vol.51, No.1, pp.3-53.

Jegadeesh, N. (1990), 'Evidence of Predictable Behavior of Security Returns', *The Journal of Finance*, Vol.45, No.3, pp.881-898.

Jegadeesh, N., and Titman, S. (1993), 'Returns to Buying Winners and Selling Losers: Implications for Stock Market Efficiency', *The Journal of Finance*, Vol.48,

No.1, pp.65-91.

Jegadeesh, N., and Titman, S. (2001), 'Momentum', Working paper, University of Illinois.

Jensen, G. R., Johnson, R. R., and Mercer, J. M. (1998), 'The Inconsistency of Small-firm and Value Stock Premiums', *The Journal of Portfolio Management*, Vol.24, No.2, pp.27-36.

Kasten, G. W. (2008), 'High Transaction Costs Negatively Affect 401(K) Participants', Journal of Pension Benefits, pp.50-64.

Kao, D. L., and Shumaker, R. D. (1999), 'Equity Style Timing', *Financial Analysts Journal*, Vol.55, No.1, pp.37-48.

Keim, D. B., and Stambaugh, R. F. (1986), 'Predicting Returns in the Stock and Bond Markets', *Journal of Financial Economics*, Vol.17, No.2, pp.357-390.

Kessel. R. A. (1965), 'The Cyclical Behavior of the Term Structure of Interest Rates'. *working paper, National Bureau of Economic Research.* 

Lakonishok, J., Shleifer, A., and Vishny, R. W. (1994), 'Contrarian Investment, Extrapolation, and Risk', *The Journal of Finance*, Vol.49, No.5, pp.1541-1578.

Lee, C. M. C., and Swaminathan, B. (2000), 'Price Momentum and Trading Volume', *The Journal of Finance*, Vol.55, No.5, pp.2017-2069.

Lehmann, B. N., (1990) 'Fads, Martingales, and Market Efficiency', *The Quarterly Journal of Economics*, Vol.105, No.1, pp.1-28.

Lettau, M., and Ludvigson, S. (2001), 'Consumption, Aggregate Wealth, and Expected Stock Returns', *The Journal of Finance*, Vol.56, No.3, pp.815-849.

Lettau, M., and Ludvigson, S. (2010), 'Measuring and Modeling Variation in the Risk-Return Trade-off', *working paper, University of California.* 

Levis, M., and Liodakis, M. (1999), 'The Profitability of Style Rotation Strategies in the United Kingdom', *The Journal of Portfolio Management*, Vol.26, No.1, pp. 73–86.

Levis, M., and Liodakis, M. (2001), 'Contrarian Strategies and Investor Expectations: The U.K. Evidence', *Financial Analysts Journal*, Vol.57, No.5, pp.43-56.

Levis, M., and Tessaromatis, N. (2004) 'Style Rotation Strategies', *The Journal of Portfolio Management*, Vol.30, No.4, pp.160-169.

L'Her, J. F., Mouakhar, T., and Roberge, M. (2007), 'Timing Small versus Large Stocks', The *Journal of Portfolio Management*, Vol. 34, No.1, pp.41-50.

Lintner, J. (1965), 'The Valuation of Risk Assets and the Selection of Risky Investments in Stock Portfolios and Capital Budgets', *The Review of Economics and Statistics*, Vol.47, No.1, pp.13-37.

Liu, N. P., and Zhang, L (2008), 'Is the Value Spread a Useful Predictor of Returns?' *Journal of Financial Markets*, Vol. 11, No.3, pp. 199-227.

Liu, W., and Strong, N. (2008), 'Biases in Decomposing Holding-Period Portfolio Returns', *The Review of Financial Studies*, Vol.21, No 5, pp.2243-2274.

Lucas, A., Dijk, R.V., and Kloek, T. (2002), 'Stock Selection, Style Rotation, and Risk', *Journal of Empirical Finance*, Vol. 9, No.1, pp. 1-34.

Maddala, G. S., (1977), *Econometrics*, New York: McGraw-Hill.

Maysami, R. C., and Koh, T. S. (2000), 'A Vector Error Correction Model of the Singapore Stock Market', *International Review of Economics and Finance*, Vol.9, No.2, pp.79-96.

Montier, J. (2007) *Behavioural Investing: A Practitioner's Guide to Applying Behavioural Finance*. Chichester, England; Hoboken, NJ: John Wiley & Sons, Ltd.

Mukherjee, T. K., and Naka, A. (1995), 'Dynamic Relations between Macroeconomic Variables and the Japanese Stock Market: An Application of a Vector Error Correction Model', *Journal of Financial Research*, Vol.18, No.2, pp.223-237.

Nasseh, A., and Strauss, J. (2000), 'Stock Prices and Domestic and International Macroeconomic Activity: a Cointegration Approach', *The Quarterly Review of Economics and Finance*, Vol. 40, No. 2, pp.229-245.

Newey, W. K., West, K. D. (1987), 'A Simple, Positive Semi-definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix', *Econometrica*, Vol.55, No.3, pp703-708.

Perez-Quiros, G., and Timmermann, A. (2000), 'Firm Size and Cyclical Variations in Stock Returns', *The Journal of Finance*, Vol. 55, No. 3, pp.1229-1262.

Petkova, R. (2002) 'Do the Fama-French Factors Proxy for Innovations in Predictive Variables' *The Journal of Finance*, Vol.61, No.2, pp.581-612.

Poon, S., and Taylor, S. J. (191), 'Macroeconomic Factors and the UK Stock Market', *Journal of Business Finance & Accounting*, Vol.18, No.5, pp.619-636.

Rosenberg, B., Reid, K., and Lanstein, R. (1985), 'Persuasive Evidence of Market Inefficiency', *Journal of Portfolio Management*, Vol.11, No.3, pp.9-16.

Ross, S. A. (1976), 'The Arbitrage Theory of Capital Asset Pricing', Journal of

Economic Theory, Vol.13, No.3, pp.341-360.

Rozeff, M. S. (1974), 'Money and Stock Prices', *Journal of Financial Economics*, Vol.1, No.3, pp.245-302.

Santos, T., and Veronesi, P. (2005), 'Cash-Flow Risk, Discount Risk, and the Value Premium', *working paper, National Bureau of Economic Research*.

Shanken, J. (1990), 'Intertemporal Asset Pricing: An Empirical Investigation', *Journal of Econometrics*, Vol.45, No.1, pp.99-120.

Sharpe, W. F. (1964), 'Capital Asset Prices: A Theory of Market Equilibrium under Conditions of Risk', *The Journal of Finance*, Vol.19, No.3, pp.425-442.

Shleifer, A. (2000) *Inefficient Markets: An Introduction to Behavioural Finance*, Oxford: Oxford University Press.

Shumway, T. (1997) 'The Delisting Bias in CRSP Data', *The Journal of Finance*, Vol.52, No.1, pp. 327-340.

Siganos, A. (2004), 'Momentum Profits in Alternative Stock Market Structures', *Working paper, University of Glasgow.* 

Smith, D. R., and Whitelaw, R. F. (2009), 'Time-Varying Risk Aversion and the Risk-Return Relation', *working paper, Simon Fraser University*.

Soufian, N. (2001), 'Empirical Content of Capital Asset Pricing Model (CAPM) and Arbitrage Pricing Theory (APT) Across Time', *working paper, Manchester Metropolitan University Business School.* 

Tamari, M. (1984), 'Effects of Changes in the Business Cycle on Small Firms in the United States', *International Small Business Journal*, Vol.2, pp.42-50.

Teo, M., and Woo, S. J. (2004), 'Style Effects in the Cross-section of Stock Returns', *Journal of Financial Economics*, Vol.74, No.2, pp.367–398.

Vassalou, M. (2003), 'News Related to Future GDP Growth as a Risk Factor in Equity Returns', *Journal of Financial Economics*, Vol.68, No.1, pp.47-73.

Welch, I., and Goyal, A. (2008), 'A Comprehensive Look at The Empirical Performance of Equity Premium Prediction', *The Review of Financial Studies*, Vol.21, No.4, pp.1455-1508.

Zhang, Q. J., Hopkins, P., Satchell, S., and Schwob, R. (2009), 'The Link Between Macroeconomic Factors and Style Returns', *Journal of Asset Management*, Vol.10, pp.338-355.